Femtosecond Multidimensional Electronic and Raman Spectroscopy towards the Study of Mn Complexes and the Oxygen Evolving Complex: Developing Understanding of Femtosecond Light Generation and Applying Femtosecond Multidimensional Electronic and Raman Spectroscopies to Model Systems

By

Paul C. Hebert

A dissertation submitted in partial fulfillment of the requirements for the degree of

> Doctor of Philosophy (Chemistry)

at the UNIVERSITY OF WISCONSIN-MADISON 2016

Date of final oral examination: 5/26/2016

The dissertation is approved by the following members of the Final Oral Committee:

John C. Wright, Professor, Chemistry

Timothy H. Bertram, Assistant Professor, Chemistry

Thomas Brunold, Professor, Chemistry

Randall H. Goldsmith, Assistant Professor, Chemistry

Deniz Yavuz, Professor, Physics

# Acknowledgements

Over the course of several years, one will undoubtedly interact with many people, especially within the confines of a graduate program. As such, before getting into the scientific content, I would like to acknowledge the contributions, assistance, and support many people and programs offered throughout my time here. In some semblance of an order, I would like to thank:

- the opportunity made available through funding by the National Science Foundation, both for the project encompassing the work described in this thesis as well as funding through the Graduate Research Fellowship Program;
- the various faculty members that have made time in their schedules to participate on the committees for my thesis background oral, original research proposal, and thesis defense, including: Tim Bertram (Thesis), Thomas Brunold (TBO, RP, and Thesis), Randy Goldsmith (Thesis), Frank Keutsch (TBO and RP), John Wright (TBO, RP, and Thesis), and Deniz Yavuz (Thesis);
- the faculty and staff that provided the opportunities to teach CHEM 524 and 621: Tim Bertram, Rob McClain, Lloyd Smith, and John Wright; these opportunities have been extremely valuable, providing not only the experience and opportunity to teach and interact with students more directly than I found was available through general chemistry assignments, but new outlooks and approaches towards thinking about chemical instrumentation and experimentation;
- Art Camire (Spectra-Physics), Joe Juenemann (Spectra-Physics), Gediminas Veitas (Light Conversion), Dave Virgillito (Spectra-Physics), and Dmitriy Vorobyev (Spectra-Physics), with whom my many interactions and communications have provided instructive and valuable insights into the fundamental processes at play in ultrafast lasers in addition to the experience and knowledge necessary for maintenance and alignment of these systems;
- the many current and former members of the academic, facilities, and support staff, without whom the challenges associated with research would not merely by challenging, but in many cases, impossible; I in no way, shape, or form attempt to pass this off as a complete list, but would mention specifically the contributions and assistance of: Phill Bloedow, Arrietta Clauss, Bill Goebel, Bruce Goldade, Brad Herring, Teresa Knudson, Sue Martin-Zernicke, Rob McClain, Steph Nagle, Jeff Nielsen, Dennis Reece, Kyle Roux, Matt Sanders, and Chad Skemp;
- the present and former graduate students and scientists of the Wright group, particularly: Stephen

Block, Lena (Yurs) Woodis, and Ross Lovely, who were among the first to welcome me into the group; Stephen, additionally for the valuable insights when first attempting to navigate the complex field of multidimensional spectroscopy and the initial instruction in picosecond OPA and system alignment; Andrei Pakoulev, for his insights, suggestions, and contributions throughout many discussions; Dan Kohler, for many intriguing and interesting conversations on numerous topics in nonlinear spectroscopy, as well as advice and insights into the capabilities and functionalities available in NumPy and SciPy; Skye Kain, for his extensive work on COLORS, without which the experiments described would not have been possible; Blaise Thompson, for his contributions maintaining and updating COLORS, and the advice and insights on the usage and functionalities of NumPy and SciPy; and Blaise, Dan, and Eric Hagee, with whom the experimental data in Ch. 4 was collected;

- John Wright, not only for the opportunity to pursue the research described herein, but the flexibility to pursue new directions as they presented themselves; additionally, for the many hours of instruction, both in class and the many interactions over the past several years, and without whose insights many of the connections and ideas described in this thesis would still be completely opaque;
- the many friends I have made during my time in Madison, and whose camaraderie and support provided a valuable network of advice, commiseration, surrogate family, and enjoyable times during our graduate careers and hopefully into the future; again, by no means claiming to invoke a complete list, including Shy Bhattacharya, Stephen Block, Jennifer Faust, Jaritza Gomez-Zayas, Elvin Morales-Perez, Leith Samad, and Kate Skog.
- my family, both immediate my parents, Chris and Deb, and sister, Sarah; extended especially my grandparents, Babci (Dorothy Cook), Dzia-Dzia (Robert Cook), Vó (Gloria), and Vô (Arthur); recently extended Greg, Lisa, Andrew, and Amy Christison and recently even more extended Roy and Minerva Christison and Betty Slater. The support provided by family both biological and by law is irreplaceable, and without the foundation provided by the upbringing and opportunities from my parents, I would not be where I am today;
- and, finally, my friend and wife, Kelly, without whose advice, company, encouragement, levelheadedness, patience, stubbornness, and support I would not have gotten through graduate school; now, we get to pursue the next stage of our lives.

# Table of Contents

	Ack	nowledgements	i
	Tabl	le of Contents	ii
	List	of Figures	ri
	List	of Tables	ii
	Abst	tract	ii
1	Intr	roduction	1
-			-
2	Equ	lipment	5
	2.1	Light Generation	6
	2.2	Chillers	8
	2.3	Light Manipulation and Detection	9
3	The	eory 12	2
	3.1	Description of Electromagnetic Radiation 13	3
		3.1.1 Light as Electromagnetic Waves	3
		3.1.2 Polychromatic Electric Fields	8
	3.2	Polarizations	4
		3.2.1 Phase Matching of Nonlinear Experiments	2
		3.2.2 Susceptibilities	6
	3.3	Maxwell's Equations and the Wave Equation	2
		3.3.1 Wave Equation	2
		3.3.2 Maxwell's Equations	3
		3.3.3 Derivation of the Generalized Wave Equation	5
	3.4	Pulse Propagation in Dispersive Media	7
		3.4.1 General Treatment of Pulse Propagation in Dispersive Media	7
		3.4.2 Relaxing the Slowly Varying Amplitude Approximation	7

		3.4.3 The Propagation of Pulsed Gaussian Beams	58
		3.4.4 Integration of the Nonlinear Polarization Outside the Plane Wave Limit $\ldots \ldots \ldots$	69
	3.5	Group Velocity, Group Velocity Mismatch, and Group Velocity Dispersion	74
		3.5.1 Effects of Group Velocity and Group Velocity Dispersion	75
		3.5.2 Calculating Group Velocity and Group Velocity Dispersion	79
	3.6	Numerical Integration of Ordinary Differential Equations	81
4	$\mathbf{Sim}$	ulation of Sum-Frequency Idler Generation	86
	4.1	Background	87
	4.2	Methodology	93
	4.3	Results and Discussion	100
5	Ma	nganese Tetraphenylporphine	124
	5.1	Background	125
	5.2	White Light Generation	127
	5.3	Experimental Deviations	129
	5.4	Results and Discussion	130
6	Bro	adband Coherent Raman Spectroscopy	136
	6.1	Background	137
	6.2	Experimental Deviations	147
	6.3	Results and Discussion	150
Α	Арр	pendix A	165
в	App	pendix B	213
	B.1	Dispersion of Benzene	214
	B.2	Dispersion of BK7 Glass	214
	B.3	Dispersion of Air	215
	B.4	Dispersion of $\beta$ -Barium Borate	215
$\mathbf{C}$	Арр	pendix C	219
	C.1	user_settings.py	221
	C.2	parameters.py	228
	C.3	material_params.py	233

iv

	C.4 nl_crystal.py	. 244
	C.5 phasematching.py	. 258
	C.6 efield.py	. 266
	C.7 controls.py	. 270
	C.8 algorithm.py	. 290
	C.9 gain_formulae.py	. 295
	C.10 plotting_routines.py	. 305
п	Appendix D	200
ν		022
	D.1 parameters.py	. 323
	D.2 operations.py	. 338
	D.3 main.py	. 356

# References

362

# List of Figures

3.1	Mode-locking of electric fields	25
3.2	Simulation of the effects of group velocity	77
3.3	Simulation of the effects of group velocity mismatch	78
3.4	Simulation of the effects of group velocity dispersion	80
4.1	Observed sum-frequency idler tuning in TOPAS-C	94
4.2	Observed signal tuning in TOPAS-C	95
4.3	Comparison of BBO group velocity mismatch as calculated from different dispersion formulae.	98
4.4	Results of second-harmonic test calculations.	104
4.5	First moment analysis of simulated signal tuning	107
4.6	Zeroth moment analysis of simulated signal tuning	108
4.7	Theoretical phase matching for Type I and Type II process in BBO. $\ldots \ldots \ldots \ldots \ldots$	109
4.8	Simulated tuning curves for sum-frequency idler simulations.	110
4.9	Simulated signal and idler spectra.	112
4.10	Simulated sum-frequency idler spectra	113
4.11	Comparison of the sum-frequency idler simulation to the pump-idler and pump-signal group	
	velocity mismatch.	116
4.12	Comparison of the sum-frequency idler simulation to phase matching bandwidth estimate	118
4.13	Analysis of sum-frequency idler simulation in the context of published theory	122
5.1	Manganese(III) Tetraphenylporphine Data Summary	132
6.1	Narrowband Raman probe temporal profile	149
6.2	Comparison of narrow- and broadband probing of broadband CARS	152
6.3	Near infrared broadband CARS of carbon tetrachloride	153
6.4	Near infrared broadband CARS of benzene	155
6.5	Near infrared broadband CARS of benzene and pyridine.	156

6.6	Visible broadband CARS of benzonitrile	159
6.7	Visible broadband CARS of benzene	160
6.8	Spectrally broader near infrared broadband CARS of benzene	161
6.9	Monochromator slices exploring modulations in broadband CARS.	162

# List of Tables

4.1	Parameters used for sum-frequency idler simulations	101
4.2	Parameters used for second-harmonic test simulations.	103

# Abstract

The oxygen evolving complex of photosystem II is a tetramanganese complex that catalyzes the oxidation of water to dioxygen. Many experimental methods have been employed to study this complex, including a variety of spectroscopic methods. The complexity of the surrounding protein environment challenges reliable interpretation of the spectral features and discrimination between changes in the complex and the many cofactors of photosystem II. We have proposed coherent multidimensional spectroscopy in the mixed time-/frequency-domain as a strategy for exploring the spectroscopy of this important system. Coherent multidimensional spectroscopy is performed by the temporal synchronization and spatial overlap of a sequence of fields to induce a polarization in the material. The fields induced by polarizations meeting certain energy- and momentum-criteria are sampled; the intensity of the fields as functions of frequency and time elucidate the spectral and temporal characteristics of the sample's quantum states. Here, we describe efforts to extend the spectroscopy of model transition metal complexes to the femtosecond regime. We present the implementation of white-light transient absorption and demonstrate our ability to reproduce the spectroscopic behavior of manganese (III) tetraphenylporphine with improved time resolution over previously published results; we discuss the limitations encountered when extending this spectroscopy to homodynedetected transient grating methods. We have implemented a strategy for broadband coherent anti-Stokes Raman spectroscopy, showing the results of these efforts. A theoretical background into the pulse propagation effects in the ultrafast regime of femtosecond spectroscopy has been compiled. We have found these propagation effects to have profound importance in the tuning and amplification of femtosecond pulses, and we present the results of simulations exploring how these effects manifest and influence the behavior of a particular process employed in light generation in a commercial traveling-wave optical parametric amplifier.

CHAPTER 1	
Introduction	

Photosynthesis in algae, cyanobacteria, and plants is a cooperative effort involving an array of proteins.[1, 2] As an assembly, these proteins accumulate energy from the sun, using this energy to drive a sequence of redox reactions with the ultimate goal of sequestering carbon in the form of sugars.[2] The conversion of carbon dioxide to sugar is also accompanied by the four-electron oxidation of water to form dioxygen.[1, 2] This ensemble of reactions is facilitated by a wide variety of cofactors, including: carotenoids and chlorophylls that facilitate the transfer of energy; redox active species such as quinones and pheophytins that chaperone the exchange of charge; and, inorganic cofactors such as chloride and calcium.[2] The amino acids of the proteins are also understood to contribute in these reactions,[1, 2] and the larger protein environment aids in stabilizing the charge separation required throughout these processes.[1]

The specificity of photosystem II towards the oxidation of water has attracted significant attention in the hopes of reproducing its capabilities in synthetic catalysts. Photosystem II's catalytic center is the oxygen evolving complex (OEC), an inorganic complex constituting four manganese and one calcium, bridged by oxygen, and coordinated by various residues.[1, 2] The role of the OEC in photosynthesis is the accumulation of charge and potential to facilitate the four-electron oxidation of water.[2] The necessary charge and potential is accumulated during the Kok cycle, a five-state model for the catalytic process of the OEC.[1, 2] Each step of the Kok cycle is initiated by the absorption of a photon by various light-harvesting proteins; this energy is transferred to P680 by various chlorophyll and carotenoid cofactors.[2] This energy electronically excites P680, permitting a series of redox reactions involving cofactors such as pheophytins and quinones and facilitating charge separation: the electron is chaperoned to photosystem I, while the OEC is the final donor in a chain of exchanges that replace P680's lost electron.[2] Donation of this electron oxidizes the OEC, advancing the catalyst to the next stage of the Kok cycle, which is denoted as a transition from the state S<sub>n</sub> to the state S<sub>n+1</sub>.[1, 2] Four sequential oxidations provide the requisite charge and potential for catalysis of water oxidation.[2]

The OEC is tailored specifically towards the oxidation of water to dioxygen: its potential is such that oxidation only occurs upon the accumulation of four oxidizing equivalents.[2] This specificity has attracted significant attention, with many different experimental methods employed towards studying the structure and function of photosystem II and the OEC. These methods include x-ray absorption,[3–10] fluorescence,[11] diffraction,[12] and scattering;[13, 14] electron paramagnetic resonance;[6, 15–33] electronic spectroscopy;[34–56] vibrational spectroscopy;[57–62] and theoretical calculations.[22, 63–75]

However, each of the aforementioned experimental methods has challenges associated with it. X-ray-based spectroscopies have been shown capable of inducing redox damage to the OEC,[8, 10] raising questions whether x-ray characterization is truly characterizing the OEC in one of the states of the Kok cycle.[65, 66] While electron paramagnetic resonance (EPR) provides direct sampling of the manganese cluster, this method has traditionally been considered to be sensitive only to the  $S_2$  state. More recently, EPR spectra associated with the  $S_0$ , [19, 28, 29]  $S_1$ , [21, 26] and  $S_3$  [20, 21, 32, 76] states have been reported, though some of these signals require chemical additives in order to be detected.[19, 28-30] Chemical additives and the removal of certain cofactors can also modify the EPR signature of the different states of the OEC.[15, 21–23] Care must be taken to account for the changes induced by optical excitation, as near infrared and visible excitation have been reported to modify various EPR signatures of the OEC; [15–18, 20, 21, 25, 27, 32] these optically-induced changes are impacted by both temperature and photosystem II source. [18] Finally, spectroscopic methods - both electronic and vibrational - seek to characterize a complex mixture with many contributing species. In addition to the protein environment, many of the cofactors of photosystem II have been shown to exhibit significant spectroscopic contributions, either from changes in redox state or from changes in the surrounding dielectric environment. [35–38, 40, 46, 48–50, 52, 54–56, 77–84] These changes in the spectroscopy of the cofactors will also manifest in the difference spectra obtained from advancement through the Kok cycle, [36, 38, 40, 44, 50, 55, 56, 77, 83] challenging the specific identification of spectral changes definitively associated with the OEC.[41-43] In order to obtain insights from less complex systems, a variety of mimics have been synthesized and characterized. Some of these samples are structural mimics, [60. 85–88] while others are intended as functional mimics exhibiting catalytic activity.[89–97] It is not always obvious that similar electronic states and behaviors are expected from these mimics as are observed from the OEC, however.

We have proposed the application of coherent multidimensional spectroscopy (CMDS) to the OEC as a means to improve spectral resolution and specificity. As a multidimensional method, CMDS distributes the spectral information across multiple dimensions, improving our ability to resolve these features. Signals derived from coupled quantum states increase the specificity of the method, permitting more reliable assignment of the observed spectral features. In the mixed time- and frequency-domain, CMDS is performed through the temporal synchronization and spatial overlap of multiple ultrafast (picosecond or femtosecond) pulses. The sample serves as a nonlinear medium, coupling the fields through induced polarizations. The magnitude of these polarizations scales with the effective nonlinear susceptibility of the material; the susceptibility may be related to the properties of the quantum states of the sample, exhibiting enhancement through self-heterodyning or may be spatially resolved from the input fields through homodyne-detection. Spectral and spatial discrimination permit the isolation of contributions from distinct polarizations. Variation in the relative temporal ordering and the frequencies of the excitation fields permits the characterization of the dynamics and lineshapes of the contributing quantum states. [98–100]

Here we describe efforts aimed towards the goal of characterizing the OEC by CMDS, reporting the implementation of femtosecond electronic spectroscopy of a representative transition metal complex and broadband coherent anti-Stokes Raman spectroscopy. We have found implementation of CMDS in the femto second domain to be complicated by various pulse propagation effects; we describe a theoretical framework for understanding these effects and their manifestation in light generation. These topics are organized as follows. In Ch. 2, we provide an overview of the Wright group's femtosecond laser system and standard operating procedure; a collection of notes on the maintenance and alignment of the lasers constituting this system are compiled as a supplement in Appendix A. In Ch. 3, we outline a theoretical framework in which various pulse propagation effects may be understood, building up from the fundamental behavior of electric fields and polarizations to detailing the derivation of formulae that explicitly account for the effects of dispersion and polarizations on the propagation of ultrafast pulses. We apply this theory in Ch. 4, in which we attempt to unravel unexpected behavior in the generation of ultrafast pulses within our traveling-wave optical parametric amplifiers. Finally, Ch. 5 and Ch. 6 present experimental results on the topics mentioned previously. In Ch. 5, we describe the setup and implementation of white light transient absorption, present our ability to reproduce the transient absorption of manganese (III) tetraphenylporphine with better temporal resolution than previously reported, and discuss the extension of spectroscopy on this complex to homodyne-detected transient grating. Finally, Ch. 6 details our efforts implementing broadband coherent anti-Stokes Raman spectroscopy. Appropriate and specific background information on the topics relevant to Ch. 4, Ch. 5, and Ch. 6 is provided in the introduction to each chapter.

CHAPTER 2	
Equipment	

## 2.1 Light Generation

Here we will provide a brief description of the design, layout, and operating procedures of the Wright group's femtosecond laser system at this time. Throughout this discussion, we will use a tilde ( $\sim$ ) to indicate that the reported value is approximate: this will alternately be used to indicate that the reported value is an optimum from which some deviation should be anticipated, or to indicate the actual value is unimportant and we wish to merely provide an estimate of the average observed under standard operation. The information is derived from the manuals, interactions with the technicians, or experience maintaining and aligning the equipment. Detailed descriptions of alignment and maintenance procedures for the lasers will not be provided here; rather, the notes collected during interactions with the service technician have been digitized and appended in Appendix A. The reader is advised to consult these notes for details as needed.

The system begins with the Millenia V (Spectra-Physics, Santa Clara). This is a cw laser with a Nd:YVO<sub>4</sub> gain medium; the rod is excited by two diode lasers with emission ~ 809 nm. The 1064 nm lasing of the Nd:YVO<sub>4</sub> undergoes intracavity frequency doubling in a temperature phase matched lithium triborate crystal. The Millenia V is capable of supplying up to 5.5 W of the resulting 532 nm radiation. Typically, the Millenia V is set to yield ~ 4 W to pump the Ti:Al<sub>2</sub>O<sub>3</sub> femtosecond oscillator, the Tsunami (Spectra-Physics, Santa Clara). The Tsunami oscillates with a repetition rate of ~ 80 MHz and yields pulses with ~ 60 nm FWHM and ~ 30-35 fs duration. Mode locking can be initiated through an acousto-optic modulator via the Model 3955 electronics module control; it is believed that mode locking is intended to be sustained through Kerr lensing, though the acousto-optic modulator can also be used. With ~ 4 W from the Millenia V, the output energy of the Tsunami is expected to be ~ 6 nJ per pulse (~ 500 mW); typically cw operation yields slightly higher (~ 30 mW) power than mode locked operation.

The pulse train from the Tsunami is split ( $\sim 35\%/65\%$ ) to seed both the femtosecond (Spitfire Pro XP, Spectra-Physics, Santa Clara) and the picosecond (Spitfire Ace, Spectra-Physics, Santa Clara) regenerative amplifiers. The regenerative amplifiers are pumped by Q-switched, Nd:YLF nanosecond lasers (Empower, Spectra-Physics, Santa Clara). The Empower is a multimode laser with  $\sim 5$  ns duration. The Nd:YLF rod is pumped by diodes and lases at 1053 nm. Temperature phase matched lithium triborate is again used for frequency doubling. The Spitfire is designed to be pumped with 20 W from the Empower; ideally, this should be attained with  $\sim 20$  A of current supplied to the diodes. At this time, the Empower pumping the femtosecond Spitfire Pro XP requires  $\sim 22$  A to achieve this power: it is believed that the shorting of a diode clip introduced smoke or debris that is reducing the efficiency of the cavity, and that proper cleaning may return operation closer to expected specifications. Amplification in the Spitfire is based upon chirped pulse

amplification: the pulses from the Tsunami are stretched from  $\sim 30-35$  fs duration to hundreds of picoseconds in the stretcher. A pair of Pockels cells isolates single pulses from the Tsunami's pulse train for amplification; the isolation of these pulses reduces the repetition rate from  $\sim 80$  MHz to the repetition rate of the Empower, which is generally operated at the maximum allowed repetition rate of 1 kHz. Optimum amplification can usually be achieved with only 10 passes through the rod, though the Spitfire is often operated at 11 passes for long-term stability; if longer build up times are required, it is advisable to check on the alignment of the amplifier, as additional passes through the cavity will result in the accumulation of nonlinear chirp. The Pockels cells also provide the Q-switch to release the amplifier pulse from the cavity. After release from the cavity, the mode of the amplified pulse is expanded by a telescope prior to recompression. The recompressed output exhibits a bandwidth of  $\sim 35$  nm, a duration of  $\sim 35-40$  fs, and pulse energies of  $\sim 4$  mJ ( $\sim 4$  W at the standard 1 kHz repetition rate).

Outside the Spitfire, a sequence of beamsplitters and mirrors split the output (~ 50%/45%/5%); the mirrors route the  $\sim 50\%$  and  $\sim 45\%$  portions to pump two traveling-wave optical parametric amplifiers (TOPAS-C, Light Conversion, Vilnius, Lithuania). Within each OPA, this pump beam is split 97.5%/2.5%. The smaller portion is split further (distribution unknown) and used to generate a seed pulse: a temporally dispersed white light continuum is produced using one fraction, and this continuum is overlapped in a  $\beta$ barium borate crystal (BBO, 2.5 mm,  $\theta = 28^{\circ}, \phi = 0^{\circ}$ ) with the other fraction. With proper angle tuning and delay compensation, the seed is amplified, ideally to yield  $\sim 1.5 \ \mu$ J. The seed is imaged onto a second BBO (2 mm,  $\theta = 28^{\circ}, \phi = 0^{\circ}$ ), along with the 97.5% fraction of the pump. Optimization of temporal overlap and phase matching angle result in significant amplification of the seed: the sum of the energies of the resulting signal and idler may be as high as 700 µJ, though 500-650 µJ appears to be common. The TOPAS-C incorporates multiple mixer options for upconversion of the signal and idler to frequencies throughout ultraviolet, visible, and near infrared. The second harmonic of the signal and idler provide spectral coverage from  $\sim$  570-800 nm and  $\sim$  800-1140 nm, respectively. The crystal used to phase match both second-harmonic generation processes is a 0.3 mm BBO ( $\theta = 23^{\circ}, \phi = 90^{\circ}$ ). Upconversion of signal and idler by sum-frequency generation with the residual pump provides coverage between  $\sim 470-530$  nm and  $\sim$  535-610 nm, respectively. The sum-frequency idler process is generated in the same crystal as secondharmonic generation (0.3 mm BBO,  $\theta = 23^{\circ}, \phi = 90^{\circ}$ ); signal is upconverted in a different 0.3 mm BBO ( $\theta$  $= 23^{\circ}, \phi = 90^{\circ}$ ). It is noted here that the sum-frequency process between pump and idler (SFI) has been observed to yield unusual spectral character; this is the topic of Ch. 4. It is suspected that similar effects cause the sum-frequency process for signal to also exhibit sub-optimal characteristics, though the features are more subtle and the necessary characterization has not been performed to date. Though the output is weak, the fourth harmonics of signal and idler can also be generated; these processes are phase matched in a 0.15 mm BBO ( $\theta = 35^{\circ}, \phi = 90^{\circ}$ ) and provide spectral coverage between ~ 285-400 nm and ~ 403-480 nm, respectively. Crystals for difference frequency generation have been purchased and tested; they are not installed on the system at this time.

The TOPAS-C provide two separate tunable colors; for some experiments, including some of those detailed in later chapters, it is necessary to have a third source available. The third fraction of the fundamental (~ 5% of the amplifier output) provides this third color. When used, the beam size is reduced by a Galilean telescope formed from a f = -50 mm and  $f \approx 100$  mm lens pair to better match the beam size of the TOPAS-C processes and to ensure compatibility with the 1" optics on the table.

## 2.2 Chillers

While the Spitfire rod is thermoelectrically cooled, it still requires an external cooling system. The Millenia V, Tsunami, and Empower also require the support of chillers. At present, the Millenia V, Tsunami, and Spitfires (Pro XP and Ace) share a chiller (Neslab Merlin M33), while the Empowers are each connected to a unique chiller (Polyscience P/N 6360T11SP20C and P/N 6360T11A120C for the femtosecond and picosecond system Empowers, respectively). Operating temperatures are adjusted to optimize performance of each laser. The most recent records indicate operating temperatures of 18° C, 24.5° C, and 21° C for Neslab Merlin M33, the femtosecond Empower Polyscience chiller, and the picosecond Empower Polyscience chiller, respectively. Maintenance on these chillers should be performed on a six-month cycle, with the procedure adapted from documentation; parts of this are also available in the dissertation of Nathan Matthew. It has been noted that Spectra-Physics updates the recommended chiller maintenance procedures from time to time; it is advisable to consult with Spectra-Physics representatives to be apprised up any changes. The current procedure is reproduced below for convenience:

- drain the existing chilling fluid. For Polyscience P/N 6360T11SP20C, this is easily done through the valve installed to the drain outlet on the back of the unit. Neither of the other units have such a valve installed, and can be drained by various methods, including syphoning and using the chiller's pump to drain the fluid. Care should be made to drain the fluid from the hoses and the filter reservoir;
- remove the existing filters;
- reconnect the hoses such that the flow path excludes the lasers, then fill the chillers with Nalco cleaning solution 460-CCL2567. The change in flow path is necessary to protect the lasers from the cleaning

solution. As the current hardware for the Polyscience P/N 6360T11SP20C does not support the exclusion of the femtosecond Empower cavity from the flow path, the usage of cleaning solution has been skipped in this unit, simply cycling distilled water instead. The chillers should be run as indicated for at least thirty minutes;

- drain the chillers, flushing the cleaning solution using distilled water. Distilled water should be added and drained from the chillers until suds are no longer observed. This usually takes a few gallons of distilled water per chiller, though more may be required;
- with the flow paths still excluding the lasers, fill the chillers with distilled water and cycle for at least fifteen minutes, then drain. Repeat this step, cycling the fresh distilled water for at least five minutes;
- drain the chillers once more, then reconnect the lasers, ensuring the flow direction is correct. Install fresh filters and refill the chillers with chilling fluid. All units can be run with distilled water as the chilling fluid. However, as the chilling fluid flows through the optical path of the Empower, Spectra-Physics recommends the usage of Nalco inhibitor 460-PCCL104.

Distilled water can be obtained from any source, but cannot be replaced by deionized water: deionization results in water that is too pure, and the usage of deionized water may cause leeching from the pump. Cleaning usually requires 8-10 gallons of distilled water. The cleaning solution is Nalco 460-CCL2567. This can be acquired from Spectra-Physics (P/N 1607-0547), though at the time of the most recent purchase, it could be purchased directly from Nalco for a lower price. Nalco 460-PCCL104 is the inhibitor solution, and it is also available from either Spectra-Physics (P/N 1607-0546) or Nalco; usually 1 gallon per chiller is necessary. The filters depend upon the chiller: the Neslab Merlin M33 uses a single Pentek P5-478 spun polypropylene filter, while a Harmsco PP-S-1 PolyPleat cartridge has been used in the Polyscience chillers.

#### 2.3 Light Manipulation and Detection

Many of the experiments described later were performed in ways that deviate from the standard methodology and configuration. Such differences will be described when appropriate. Here, a brief discussion will be presented on the standard operation at this time.

Following light generation, periscopes based upon reflection at the Brewster angle ensure all processes from the OPAs are vertically polarized; a  $90^{\circ}$  periscope rotates the polarization of the residual fundamental from horizontal to vertical. The two OPAs and the residual fundamental permit two- or three-color experiments: for two-color experiments, the output from one OPA is split, yielding a series of interactions labeled  $\omega_1, \omega_2$ , and  $\omega_{2'}$ ; for three-color experiments, rather than splitting the OPA, the fundamental is introduced, yielding interactions  $\omega_1, \omega_2$ , and  $\omega_{800}$ . One beam is blocked for transient absorption.

The beams needed for an experiment are routed by a series of mirrors. Three delay lines are available for temporal synchronization of the pulses: these delay lines are corner-cube retroreflectors mounted on motorized stages. Commonly, a chopper is introduced to one of the beam lines for background subtraction. From the delay lines, the beams are aligned to an appropriate mask to satisfy the phase matching for a given experiment. The table is designed around a standard BOXCARS geometry with angles of  $\sim 1^{\circ}$  into the sample. This mask is robust for most experiments with a phase matching condition of  $\mathbf{k}_4 = \mathbf{k}_1 - \mathbf{k}_2 + \mathbf{k}_{2'}$ ; a small phase mismatch occurs if the difference between  $\omega_1$  and  $\omega_2$  is large (for example, phase matching within the cw approximation predicts phase mismatch should attenuate the intensity less than  $\sim 1\%$  for  $\lambda_1 = 495$  nm and  $\lambda_2 = 615$  nm). The beams are focused into the sample with a f = 1 m mirror. Signal is collected, collimated, and focused into a monochromator. A variety of detectors are available for different frequency regions and different purposes. A homebuilt InGaAs array detector allows increased throughput for expedited signal tuning. For experiments, an InSb detector and a silicon photodiode are commonly used for near infrared and visible frequencies, respectively; both detectors are compatible with the same preamplifier. Photomultiplier tubes are available, but the amplification provided by the current preamplifier's gain limits reliable application: the gain of the preamplifier is sufficiently large that preventing saturation of the signal processing hardware requires operating the PMT with biases significantly below specifications.

The signal from the preamplifier is sampled by a boxcar integrator; synchronization of the integrator and aforementioned chopper allows active background subtraction. The settings on the boxcar are adjusted according to the detector and the sensitivity of the experiment. The output from the boxcar is collected on a National Instrument DAQ card. A LabVIEW program (COLORS) is used to collect the data and interface with the various hardware components. COLORS includes features that allow: tuning, setting, and scanning the OPAs; synchronization of the pulses through the motorized delay lines; and, data collection and processing.

The standard protocol for performing an experiment begins with optimizing light generation. After aligning the output of the regenerative amplifier into the OPAs, recompression is optimized, generally with respect to the white light in the TOPAS-C. Further alignment adjustments to the TOPAS-C are made as needed. The table is aligned according to the appropriate mask for an experiment. Tuning of the signal process in the OPAs can be completed through an automated routine in COLORS. After confirming the signal tuning is acceptable, upconversion processes can be tuned as needed: the recommended tuning procedure is to set the signal to the appropriate color and identify the crystal angle that maximizes the power of the upconverted process. Alignment through a pinhole is used to achieve spatial overlap of the modes; the pinhole is appropriate for the f = 1 m mirror, but it has been found that for shorter focal lengths it can be more accurate to optimize overlap by eye. Temporal overlap is determined through nonlinear processes in appropriate media. A sequence that has proven convenient is: find and optimize nonlinear signal in KTP, which provides a high nonlinearity and less sensitivity to crystal angle but limits temporal resolution due to available crystal lengths; find and optimize signal in BBO, which enables high temporal resolution due to its shorter length; and, finally, optimize temporal overlap based upon the nonresonant response of an organic solvent. Generally, the solvent is matched to the sample that will be studied, but at times it can be convenient to first find signal in carbon tetrachloride or carbon disulfide due to their high nonlinearities. Zero delay is determined as a function of frequency to permit automatic delay compensation.

CHAPTER 3
Theory

### 3.1 Description of Electromagnetic Radiation

#### 3.1.1 Light as Electromagnetic Waves<sup>i</sup>

In order to apply Maxwell's equations towards understanding the effects of a medium on the propagation of ultrafast pulses and the resulting consequences within nonlinear spectroscopic experiments, we will require both a qualitative and mathematical understanding of light. As we will see in § 3.3, Maxwell's equations predict that the propagation of electric and magnetic fields in free space will assume a functional form consistent with the classical wave equation:

$$\frac{\partial^2}{\partial z^2} f(z,t) = \frac{1}{v^2} \frac{\partial^2}{\partial t^2} f(z,t)$$
(3.1.1)

The solutions to the classical wave equation are expected to be wave functions, f(z, t), consistent with some functional form propagating at a constant phase velocity, v. Within this definition of a wave, it is trivial to show the functional dependence of f on z and t. For example, let us consider that we have some functional form corresponding to the wave function at t = 0, f(z, t = 0). If we want to know the functional form of f(z,t), then we simply have to consider: at time t, how much have we displaced the wave function? For a velocity of v, this corresponds to z = vt. Therefore, we would anticipate that  $f(z,t) = f(z \pm vt, 0)$ , where  $\pm vt$  describes displacement of the wave function through propagation in either direction along the z axis. Since both cases (+vt or by -vt) can be shown to be solutions to Eq. 3.1.1, we can generalize the solution f(z,t) as their linear combination:[101]

$$f(z,t) = g(z - vt) + h(z + vt)$$
(3.1.2)

While any function consistent with Eq. 3.1.2 could describe a wave, a common functional form - and one that will prove a convenient starting point as a description for light - is the sinusoid:<sup>ii</sup>

$$f(z,t) = A\cos[k(z - vt) + \phi]$$
(3.1.3)

Within the functional form of Eq. 3.1.3,<sup>iii</sup> k is the wavenumber of the sinusoid, describing the angular

<sup>&</sup>lt;sup>i</sup>This section is strongly influenced by Griffiths' text on electrodynamics.[101]

<sup>&</sup>lt;sup>ii</sup>While the functional dependence of the sinusoid is given by the cosine function in Eq. 3.1.3, this is functionally equivalent to the sine function, differing only in the initial phase,  $\phi$ . However, it is more straightforward to relate the complex wave function to the cosine function (see Eq. 3.1.13).

<sup>&</sup>lt;sup>iii</sup>Note that in Eq. 3.1.3 we have taken the dependence of f(z,t) upon z and t to be z - vt; we could have just as easily

frequency of the wave's spatial oscillations: over a distance of  $\Delta z = 2\pi/k$ , the sinusoid completes one cycle. We can recognize this distance as one period of the wave and assign it to be the wavelength,  $\lambda$ , yielding the following relationship between the wavelength and the wavenumber:

$$k\lambda = 2\pi \tag{3.1.4}$$

However, the functional form of Eq. 3.1.3 indicates that changes in time will also cause an evolution of the wave function, with a characteristic frequency of kv. We can assign this frequency as the angular frequency,  $\omega$ , yielding the following relationships between the wavelength ( $\lambda$ ), linear frequency ( $\nu$ ), and the velocity (v) of the wave or between the spatial (k) and angular ( $\omega$ ) frequencies and the velocity of the wave:

$$v = \frac{\omega}{k} = \nu\lambda \tag{3.1.5}$$

Finally,  $\phi$  in Eq. 3.1.3 describes the initial phase of the wave, defining the relative delay of the oscillations.

In addition to predicting wave-like behavior for both electric and magnetic fields, Maxwell's equations predict the co-propagation of electric and magnetic fields, yielding electromagnetic radiation: light. The propagation of each field is consistent with the form in Eq. 3.1.3. We will take the electric and magnetic fields as plane waves propagating along an arbitrary z-axis. For now, it is sufficient to assume both fields to be monochromatic - each consisting of a single frequency component - and to behave as plane waves - with minimal dependence upon the transverse coordinates, taken to be the x- and y-axes.<sup>i</sup> Within this model, Maxwell's equations can provide some insight regarding the manner in which the electric and magnetic fields will co-propagate. We will identify the key relationships from Maxwell's equations when appropriate; otherwise, the reader is directed to § 3.3.2, where the forms of Maxwell's equations are presented.

First, since Maxwell's equations describe the behavior of electric and magnetic fields through vector operations, we should assume it is most appropriate to consider the fields as vectors, with separate components oriented along the x, y, and z axes. We will assume all three components of each field oscillate in phase and

taken as convention that the field propagates in the opposite direction, z + vt. However, the direction of propagation being z - vt appears to be a common convention.

<sup>&</sup>lt;sup>i</sup>The plane wave limit is cited as a reasonable approximation as long as the radius of curvature is sufficiently larger than the wavelength of the fields.[101]

at the same frequency:<sup>i</sup>

$$\tilde{\mathbf{E}}(z,t) = \begin{bmatrix} E_{0x}\cos(k_E z - \omega_E t + \phi_E)\hat{\mathbf{x}} \\ E_{0y}\cos(k_E z - \omega_E t + \phi_E)\hat{\mathbf{y}} \\ E_{0z}\cos(k_E z - \omega_E t + \phi_E)\hat{\mathbf{z}} \end{bmatrix}$$
(3.1.6a)  
$$\tilde{\mathbf{B}}(z,t) = \begin{bmatrix} B_{0x}\cos(k_B z - \omega_B t + \phi_B)\hat{\mathbf{x}} \\ B_{0y}\cos(k_B z - \omega_B t + \phi_B)\hat{\mathbf{y}} \\ B_{0z}\cos(k_B z - \omega_B t + \phi_B)\hat{\mathbf{z}} \end{bmatrix}$$
(3.1.6b)

In Eq. 3.1.6a and 3.1.6b,  $\hat{\mathbf{x}}$ ,  $\hat{\mathbf{y}}$ , and  $\hat{\mathbf{z}}$  represent the unit vectors aligned with the *x*-, *y*-, and *z*-axis, respectively, specifying the contribution of each available polarization state to the total field. The magnitude of the contribution of a given polarization state is described as a sinusoid with an amplitude represented by the scalar quantities  $E_{0x}$ ,  $E_{0y}$ ,  $E_{0z}$ ,  $B_{0x}$ ,  $B_{0y}$ , and  $B_{0z}$ . As mentioned previously, each polarization state of a given field is presumed to exhibit the same sinusoidal oscillations as functions of *z* and *t*. For now, we will maintain a distinction between the frequencies and phases of the two fields, reflected in the notation  $k_E$  versus  $k_B$ ,  $v_E$  versus  $v_B$ , and  $\phi_E$  versus  $\phi_B$ . We will find that this distinction is unnecessary, however, as Maxwell's equations predict co-propagation of the electric and magnetic fields with matched frequency and phase.

The first simplification of Eq. 3.1.6a and 3.1.6b can be made by invoking Gauss' law and Gauss' law of magnetism. In free space, these laws define the divergence of both fields to be zero, i.e.  $\nabla \cdot \tilde{\mathbf{E}}(\mathbf{r}, z, t) =$  $\nabla \cdot \tilde{\mathbf{B}}(\mathbf{r}, z, t) = 0$ . Since the plane wave limit describes both fields propagating along an arbitrary axis (taken to be the z-axis) without dependence upon the transverse coordinates, the derivative of the  $\hat{\mathbf{x}}$ -polarized component of each field with respect to x must be zero; the same must also be true of the derivative of the  $\hat{\mathbf{y}}$ -polarized component with respect to y. Gauss' laws then leave  $E_{0z}\partial/\partial z \cos(k_E z - \omega_E t + \phi_E) = 0$  and  $B_{0z}\partial/\partial z \cos(k_B z - \omega_B t + \phi_B) = 0$ . As the derivative of a sinusoid is a sinusoid,<sup>ii</sup> these relationships can only

 $<sup>^{</sup>i}$ Note that we are introducing a notation consistent with Boyd[102] in which quantities oscillating rapidly (i.e. at optical frequencies) in time are denoted by a tilde.

<sup>&</sup>lt;sup>ii</sup>Recalling  $\partial/\partial x \cos(x) = -\sin(x)$  and  $\partial/\partial x \sin(x) = \cos(x)$ .

be satisfied if  $E_{0z} = 0$  and  $B_{0z} = 0$ . This conclusion allows us to simplify Eq. 3.1.6a and 3.1.6b, yielding:

$$\tilde{\mathbf{E}}(z,t) = \begin{bmatrix} E_{0x}\cos(k_E z - \omega_E t + \phi_E)\hat{\mathbf{x}} \\ E_{0y}\cos(k_E z - \omega_E t + \phi_E)\hat{\mathbf{y}} \\ 0\hat{\mathbf{z}} \end{bmatrix}$$
(3.1.7a)  
$$\tilde{\mathbf{B}}(z,t) = \begin{bmatrix} B_{0x}\cos(k_B z - \omega_B t + \phi_B)\hat{\mathbf{x}} \\ B_{0y}\cos(k_B z - \omega_B t + \phi_B)\hat{\mathbf{y}} \\ 0\hat{\mathbf{z}} \end{bmatrix}$$
(3.1.7b)

From these results, we must conclude that both the electric and magnetic fields are transverse waves, and can only be polarized perpendicular to the direction of propagation.

The other simplification to  $\tilde{\mathbf{E}}$  (z,t) and  $\tilde{\mathbf{B}}$  (z,t) can be made through either Faraday's law or Ampere's law. In free space, these laws parallel each other:  $\nabla \times \tilde{\mathbf{E}}(\mathbf{r}, z, t) = -\partial/\partial t \tilde{\mathbf{B}}(\mathbf{r}, z, t)$  and  $\nabla \times \tilde{\mathbf{B}}(\mathbf{r}, z, t) \propto \partial/\partial t \tilde{\mathbf{E}}(\mathbf{r}, z, t)$ .<sup>i</sup> Following Faraday's law ( $\nabla \times \tilde{\mathbf{E}}(\mathbf{r}, z, t) = -\partial/\partial t \tilde{\mathbf{B}}(\mathbf{r}, z, t)$ ), we can evaluate the expressions on both sides of the equality ( $\nabla \times \tilde{\mathbf{E}}(\mathbf{r}, z, t)$  and  $-\partial/\partial t \tilde{\mathbf{B}}(\mathbf{r}, z, t)$ ):

$$\nabla \times \tilde{\mathbf{E}}(z,t) = \begin{bmatrix} E_{0y}k_E \sin(k_E z - \omega_E t + \phi_E)\hat{\mathbf{x}} \\ -E_{0x}k_E \sin(k_E z - \omega_E t + \phi_E)\hat{\mathbf{y}} \\ 0\hat{\mathbf{z}} \end{bmatrix}$$
(3.1.8a)  
$$-\frac{\partial}{\partial t}\tilde{\mathbf{B}}(z,t) = \begin{bmatrix} -B_{0x}\omega_B \sin(k_B z - \omega_B t + \phi_B)\hat{\mathbf{x}} \\ -B_{0y}\omega_B \sin(k_B z - \omega_B t + \phi_B)\hat{\mathbf{y}} \\ 0\hat{\mathbf{z}} \end{bmatrix}$$
(3.1.8b)

Faraday's law requires the results of Eq. 3.1.8a and 3.1.8b to be equivalent; we therefore end up with the following pair of equalities:

$$E_{0y}k_E\sin(k_Ez - \omega_E t + \phi_E) = -B_{0x}\omega_B\sin(k_Bz - \omega_B t + \phi_B)$$
  
$$-E_{0x}k_E\sin(k_Ez - \omega_E t + \phi_E) = -B_{0y}\omega_B\sin(k_Bz - \omega_B t + \phi_B)$$
  
(3.1.9)

The equalities of Eq. 3.1.9 informs on two key properties of the co-propagating electric and magnetic fields. First, comparison of the sinusoidal terms on the opposing sides of each equality necessitates coherence between the fields: the equalities can only be satisfied if the electric and magnetic fields exhibit the same

<sup>&</sup>lt;sup>i</sup>For now, it is sufficient to leave the relationship of Ampere's law as a proportionality.

frequency and phase. With this conclusion, we can divide through both equalities in Eq. 3.1.9 by the sinusoidal terms, leaving  $E_{0y} \propto -B_{0x}$  and  $E_{0x} \propto B_{0y}$ . It is trivial to show that this pair of relationships is consistent with:

$$\tilde{\mathbf{B}}(z,t) \propto \hat{\mathbf{z}} \times \tilde{\mathbf{E}}(z,t)$$
 (3.1.10)

The relationship of Eq. 3.1.10, defining  $\tilde{\mathbf{B}}(z,t)$  by the cross-product of  $\hat{\mathbf{z}}$  - the direction of propagation and  $\tilde{\mathbf{E}}(z,t)$ , imposes a requirement on the relative polarization of  $\tilde{\mathbf{B}}(z,t)$  and  $\tilde{\mathbf{E}}(z,t)$ : not only must both fields be transverse, but the fields must be perpendicular to each other and satisfy the right-hand-rule with respect to the direction of propagation.

Despite light constituting both electric and magnetic fields, our current scope is exploring the interaction of light and matter within the context of optical spectroscopies. The frequencies relevant to such experiments will generally exclude regimes for which we need to consider interactions between the magnetic field and the material.[99] Consequently, while experiments for which the material response to the magnetic field needs to be considered do exist, we will proceed throughout the following sections considering only the electric field. We will also generally only sample the electric field, and therefore be limited to characterizing the polarizations of two electric fields relative to each other. Even then, while the relative polarizations of interacting fields can be important, we will often be able to make approximations permitting the treatment of the fields as scalar quantities (as we will show in § 3.2.2). Under such assumptions, we can simplify the system described by Eq. 3.1.7a and 3.1.7b to a single expression:

$$\tilde{E}(z,t) = E_0(\omega)\cos(kz - \omega t + \phi)$$
(3.1.11)

While Eq. 3.1.11 is consistent with the observed behavior of light and conveniently yields a real-valued function, the form of the wave equation illustrates that the theoretical treatment of field propagation and nonlinear interactions will involve differential equations. In such circumstances, trigonometric functions increase the difficulty in evaluating the relevant expressions. Fortunately, we can make the treatment more tractable through Euler's formula  $(\exp(i\theta) = \cos(\theta) + i\sin(\theta))$ , which demonstrates that complex exponentials are also oscillatory and - as linear combinations of solutions to the wave equation - valid descriptions of waves.

Therefore, we can consider the complex function:<sup>i</sup>

$$\tilde{E}(z,t) = E_0(\omega)e^{-i(kz-\omega t+\phi)}$$
(3.1.12)

as a valid representation for the behavior of an electric field. However, though exponentials are more compatible with differential equations, we still need to ensure that the electric field evaluates to be a real valued function.[101, 102] Some authors[101] address this by defining the real-valued field as the real part of the complex exponential, i.e.  $\tilde{E}(z,t) = \Re\{E_0(\omega) \exp[-i(kz - \omega t + \phi)]\}$ ; many other authors instead recognize the relationship:

$$\tilde{E}(z,t) \propto E_0(\omega)e^{-i(kz-\omega t+\phi)} + c.c.$$

$$\propto E_0(\omega)e^{-i(kz-\omega t+\phi)} + E_0^*(\omega)e^{i(kz-\omega t+\phi)}$$

$$\propto E_0(\omega)\left[\cos(kz-\omega t+\phi) - i\sin(kz-\omega t+\phi) + \cos(kz-\omega t+\phi) + i\sin(kz-\omega t+\phi)\right]$$

$$\propto E_0(\omega)\cos(kz-\omega t+\phi)$$
(3.1.13)

where c.c. represents the complex conjugate of the complex field,  $E_0^*(\omega) \exp[i(kz - \omega t + \phi)]$ . In the second step of the scheme in Eq. 3.1.13, we have assumed  $E_0(\omega) = E_0^*(\omega)$ : since we have assumed  $E_0(\omega) \cos(kz - \omega t + \phi)$ is real-valued,  $E_0(\omega)$  must be real-valued and therefore equal to its conjugate. In the scheme in Eq. 3.1.13, we emphasize the relationship as a proportionality to raise awareness that different authors will scale this relationship differently: while a coefficient of 1/2 will yield a result that matches Eq. 3.1.11,[103] other authors use a coefficient of 1 for each exponential.[102] In order to maintain consistency with Boyd,[102] we will employ the convention that the coefficient of each exponential is 1; when evaluating nonlinear polarizations (see § 3.2), this will result in degeneracy factors that differ from those of other authors by a factor of 2.[103, 110]

### 3.1.2 Polychromatic Electric Fields

Throughout § 3.1.1, we have limited our focus to monochromatic fields - electric fields consisting of a single frequency. While this is a convenient starting point, it is rarely relevant in laser spectroscopy: even in a HeNe laser, it is trivial to end up with multiple cavity modes contributing to the output.[105] However, if a monochromatic field of the form in Eq. 3.1.12 is a valid solution to the relevant wave equation, polychromatic

<sup>&</sup>lt;sup>i</sup>Note that we are taking the component with phase  $\exp[-i(kz - \omega t + \phi)]$  to be our complex wave function; this selection is made to ensure compatibility with our adopted mathematical conventions. This is consistent with some authors,[103–105] but the opposite of others.[99, 101, 102, 106–109]

fields defined as linear combinations of such monochromatic fields must also be valid solutions to the wave equation.[101] Such polychromatic fields can be defined as sums in the manner of Eq. 3.1.14:[102, 105, 111]

$$\tilde{E}(\mathbf{r}, z, t) = \sum_{n} \tilde{E}_{n}(\mathbf{r}, z, t) + \text{c.c.} \qquad \{n : \omega_{n} \ge 0\}$$
(3.1.14)

where  $\tilde{E}_n(\mathbf{r}, z, t)$  is the monochromatic electric field at frequency  $\omega_n$ . This is equivalent to Eq. 3.1.15, where we have substituted for  $\tilde{E}_n(\mathbf{r}, z, t)$  according to the form in Eq. 3.1.12:

$$\tilde{E}(\mathbf{r},z,t) = \sum_{n} E_0(\mathbf{r},z,\omega_n) e^{-i[k(\omega_n)z - \omega_n t + \phi(\omega_n)]} + E_0^*(\mathbf{r},z,\omega_n) e^{i[k(\omega_n)z - \omega_n t + \phi(\omega_n)]} \qquad \{n:\omega_n \ge 0\} \quad (3.1.15)$$

In Eq. 3.1.14 and 3.1.15, we have introduced the notation  $\mathbf{r}$  to generalize the transverse coordinates (x, y); as noted before, we can still treat the fields as plane waves as long as the functional dependence on the transverse coordinates involves changes at scales significantly larger than the wavelength of the field.[101] It should also be noted that we have allowed for the possibility that the amplitude,  $E_0(\mathbf{r}, z, t)$ , exhibits a functional dependence upon the spatial coordinates. This will facilitate our treatment of propagation effects (see § 3.4); we still require the amplitude to be real at this time. We can simplify Eq. 3.1.15 by introducing a complex amplitude that incorporates the phase factor,  $\exp[-i\phi(\omega)]$ .[101] Defining this complex amplitude function,  $\hat{a}(\mathbf{r}, z, \omega)$ :<sup>i</sup>

$$\hat{a}(\mathbf{r}, z, \omega) = E_0(\mathbf{r}, z, \omega) e^{-i\phi(\omega)}$$
(3.1.16)

we can simplify Eq. 3.1.15 to yield:

$$\tilde{E}(\mathbf{r},z,t) = \sum_{n} \hat{a}(\mathbf{r},z,\omega_n) e^{-i[k(\omega_n)z - \omega_n t]} + \hat{a}^*(\mathbf{r},z,\omega_n) e^{i[k(\omega_n)z - \omega_n t]} \qquad \{n : \omega_n \ge 0\}$$
(3.1.17)

However, we will once again give consideration to mathematical convenience. At the limit in which the summation in Eq. 3.1.17 is sampled over a continuous range of frequencies, [101-103] this summation tends towards the equivalent integral over  $\omega$ . Introducing the appropriate integration factor, <sup>ii</sup> [112, 113] the

<sup>&</sup>lt;sup>i</sup>The notation of the envelope function as  $\hat{a}$  will be clear when we consider the form of the electric field in the regime of continuously varying frequencies.

<sup>&</sup>lt;sup>ii</sup>The factor of  $2\pi$  introduced with the integration factor  $d\omega$  accounts for the variable of integration being angular frequency.[112, 113] We will ultimately show that the integral in Eq. 3.1.18 represents an inverse Fourier transform (see Eq. 3.1.26), and the one-dimensional Fourier transform between time and angular frequency requires this factor of  $2\pi$  for normalization. While a factor of  $\sqrt{2\pi}$  in both the Fourier transform and the inverse Fourier transform would ensure a unitary Fourier transform,[114] for now we will follow the example of Boyd and account for the normalization constant on the inverse Fourier transform only.[102]

summation approaches the limit of an integral with the following form:

$$\tilde{E}(\mathbf{r},z,t) = \int_0^\infty \hat{a}(\mathbf{r},z,\omega) e^{-i[k(\omega)z-\omega t]} + \hat{a}^*(\mathbf{r},z,\omega) e^{i[k(\omega)z-\omega t]} \frac{d\omega}{2\pi}$$
(3.1.18)

Eq. 3.1.18 is consistent with a Laplace transform from the angular frequency domain to a purely imaginary Fourier domain.[112] This restriction of Eq. 3.1.18 corresponding to a Laplace transform arises from the bounds of integration: as negative frequencies have no meaning, we can only allow the integral to span between the bounds of 0 and  $\infty$ .[102] One approach in the literature[102] is to invoke the relationship between k,  $\omega$ , and v (see Eq. 3.1.5), recognizing that we can define the wavenumber:  $k = \omega/v$ . With this definition for the wavenumber, we can consider the conjugate of the complex electric field:

$$\begin{bmatrix} \hat{a}(\mathbf{r}, z, \omega) e^{-i[k(\omega)z - \omega t]} \end{bmatrix}^* = \begin{bmatrix} \hat{a}(\mathbf{r}, z, \omega) e^{-i[(\omega/v)z - \omega t]} \end{bmatrix}^*$$
$$= \hat{a}^*(\mathbf{r}, z, \omega) e^{i[(\omega/v)z - \omega t]}$$
$$= \hat{a}^*(\mathbf{r}, z, \omega) e^{-i[((-\omega)/v)z - (-\omega)t]}$$
(3.1.19)

From this scheme, we see that if we assume  $\hat{a}(\mathbf{r}, z, \omega)$  is Hermitian, that is we assume the equality  $\hat{a}^*(\mathbf{r}, z, \omega) = \hat{a}(\mathbf{r}, z, -\omega)$ , we can simplify Eq. 3.1.18 by extending the integration to all frequencies - positive and negative:[102]

$$\tilde{E}(\mathbf{r}, z, t) = \int_{-\infty}^{\infty} \hat{a}(\mathbf{r}, z, \omega) e^{-i[k(\omega)z - \omega t]} \frac{d\omega}{2\pi}$$
(3.1.20)

Within the form of Eq. 3.1.20, we now ensure the resulting field,  $\tilde{E}(\mathbf{r}, z, t)$ , is real-valued by allowing the negative frequencies in the integral to correspond to the conjugate terms of the positive frequency components according to Eq. 3.1.19.[102]

Unfortunately, this strategy is not robust: previously (see Eq. 3.1.16), we defined the complex amplitude function,  $\hat{a}(\mathbf{r}, z, \omega)$ , in order to incorporate the phase factor,  $\exp[-i\phi(\omega)]$ . As this phase factor is the only imaginary contribution to  $\hat{a}(\mathbf{r}, z, \omega)$ ,<sup>i</sup> if we assume  $\hat{a}(\mathbf{r}, z, \omega)$  to be a Hermitian function, we are implicitly assuming  $\exp[-i\phi(-\omega)] = \exp[i\phi(\omega)]$ . This assumption restricts the functional forms of  $\phi(\omega)$  to only those functions that exhibit odd symmetry. As  $\phi(\omega)$  is not required to be a function with odd symmetry, we must find an alternative approach to manipulate Eq. 3.1.18.

In contrast to Boyd's approach of assuming  $\hat{a}(\mathbf{r}, z, \omega)$  exhibits Hermitian symmetry,[102] we will instead begin by considering the distribution defined by  $\hat{a}(\mathbf{r}, z, \omega)$ . If we consider that this distribution can be

<sup>&</sup>lt;sup>i</sup>Recalling Eq. 3.1.16, the only other contribution to  $\hat{a}(\mathbf{r}, z, \omega)$  is  $E_0(\mathbf{r}, z, \omega)$ . We have taken  $E_0(\mathbf{r}, z, \omega)$  to represent a scalar amplitude, thereby requiring it to be real-valued.

characterized by a distribution around a central frequency,  $\omega_0$ , we can find this characteristic frequency through the first moment of the distribution in the manner of Brabec and Krausz:[115]

$$\omega_0 = \frac{\int_0^\infty \omega |\hat{a}(\omega)|^2 d\omega}{\int_0^\infty |\hat{a}(\omega)|^2 d\omega}$$
(3.1.21)

where  $|\hat{a}(\omega)|^2 = \hat{a}(\omega)\hat{a}^*(\omega) = E(\omega)E^*(\omega)$  and  $E(\omega)$  is defined as the Fourier transform of E(t). While Fourier transformation of E(t) will remove the terms describing the temporal oscillations, E(w) will still include a complex exponential describing the spatial oscillations. However, the product of  $E(\omega)$  and its complex conjugate,  $E^*(\omega)$ , will remove the exponentials, leaving only the amplitude function  $\hat{a}(\omega)$ , as in Eq. 3.1.21.

Having defined the characteristic carrier frequency, let us now reconsider the form of the continuously defined electric field from Eq. 3.1.18. As the integral of the sum of two functions is equal to the sum of the integrals of the same two functions, [116] we can express the integral in Eq. 3.1.18 in the form:

$$\tilde{E}(\mathbf{r},z,t) = \int_0^\infty \hat{a}(\mathbf{r},z,\omega) e^{-i[k(\omega)z-\omega t]} \frac{d\omega}{2\pi} + \int_0^\infty \hat{a}^*(\mathbf{r},z,\omega) e^{i[k(\omega)z-\omega t]} \frac{d\omega}{2\pi}$$
(3.1.22)

From Eq. 3.1.22, we will consider two separate substitutions, introducing a new variable of integration,  $\Delta\omega$ . For the first integral of Eq. 3.1.22, we will consider this variable to be defined  $\Delta\omega = \omega - \omega_0$ , where  $\omega_0$  is the carrier frequency determined according to Eq. 3.1.21. Substitution for  $\Delta\omega$  will shift the lower bound of integration from  $\omega = 0$  to  $\Delta\omega = -\omega_0$ , but the upper bound will remain  $\infty$ . As for the integration factor, since  $\partial\Delta\omega/\partial\omega = 1$ ,  $d\omega$  can be directly replaced by  $d\Delta\omega$ . For the second integral in Eq. 3.1.22, we will define  $\Delta\omega$  to be  $\Delta\omega = \omega_0 - \omega$ . With this substitution, we will observe the opposite changes in our bounds of integration: instead of spanning  $\omega = 0$  to  $\omega = \infty$ , substitution according to  $\Delta\omega = \omega_0 - \omega$  will shift the bounds to span  $\Delta\omega = \omega_0$  to  $\Delta\omega = -\infty$ . Additionally, since  $\partial\Delta\omega/\partial\omega = -1$  for this second substitution, the integration factor  $d\omega$  will need to be replaced by  $-d\Delta\omega$ . Recognizing that these substitutions permit the definition of  $\omega$  as  $\omega = \omega_0 + \Delta\omega$  for the first integral and  $\omega = \omega_0 - \Delta\omega$  for the second, we can proceed with these substitutions to yield:

$$\tilde{E}(\mathbf{r}, z, t) = \int_{-\omega_0}^{\infty} \hat{a}(\mathbf{r}, z, \omega_0 + \Delta\omega) e^{-i[k(\omega_0 + \Delta\omega)z - (\omega_0 + \Delta\omega)t]} \frac{d\Delta\omega}{2\pi} + \int_{\omega_0}^{-\infty} \hat{a}^*(\mathbf{r}, z, \omega_0 - \Delta\omega) e^{i[k(\omega_0 - \Delta\omega)z - (\omega_0 - \Delta\omega)t]} \frac{(-d\Delta\omega)}{2\pi}$$
(3.1.23)

As the integral from  $\omega_0$  to  $-\infty$  just yields the negative value of the integral from  $-\infty$  to  $\omega_0$ , we can recall

our earlier definition of the wavenumber,  $k = \omega/v$ , permitting the simplification of Eq. 3.1.23 slightly:

$$\tilde{E}(\mathbf{r},z,t) = e^{-ik_0 z} e^{i\omega_0 t} \int_{-\omega_0}^{\infty} \hat{a}(\mathbf{r},z,\omega_0+\Delta\omega) e^{-i(\Delta\omega/v)z} e^{i\Delta\omega t} \frac{d\Delta\omega}{2\pi} + e^{ik_0 z} e^{-i\omega_0 t} \int_{-\infty}^{\omega_0} \hat{a}^*(\mathbf{r},z,\omega_0-\Delta\omega) e^{i[(-\Delta\omega)/v]z} e^{i\Delta\omega t} \frac{d\Delta\omega}{2\pi}$$
(3.1.24)

where  $k_0 = k(\omega_0) = \omega_0 / v$ .

From Eq. 3.1.24 we can make two further simplifications. First, if we assume that the behavior of  $\hat{a}(\mathbf{r}, z, \omega)$  is such that it does not evaluate to non-zero values beyond the allowed bounds of integration, i.e. for negative values of  $\omega$ , there is no functional difference between the integrals in Eq. 3.1.24 spanning  $\pm \omega_0$  and  $\pm \infty$  and the same integrals spanning the bounds of  $-\infty$  to  $\infty$ . Second, let us consider a second function,  $\hat{A}(\mathbf{r}, z, \Delta \omega)$ , that we will define relative to  $\hat{a}(\mathbf{r}, z, \omega)$  simply by shifting the center of the distribution:  $\hat{A}(\mathbf{r}, z, \Delta \omega) = \hat{a}(\mathbf{r}, z, \omega_0 + \Delta \omega)$ . In other words,  $\hat{A}(\mathbf{r}, z, \Delta \omega)$  is a function that exhibits the same functional behavior as  $\hat{a}(\mathbf{r}, z, \omega)$ , but is centered such that the distribution's carrier frequency,  $\omega_0$ , serves as the origin  $(\Delta \omega = 0)$ . With these two changes, we can modify Eq. 3.1.24 to define the electric field:

$$\tilde{E}(\mathbf{r},z,t) = e^{-ik_0 z} e^{i\omega_0 t} \int_{-\infty}^{\infty} \hat{A}(\mathbf{r},z,\Delta\omega) e^{-i(\Delta\omega/\nu) z} e^{i\Delta\omega t} \frac{d\Delta\omega}{2\pi} + e^{ik_0 z} e^{-i\omega_0 t} \int_{-\infty}^{\infty} \hat{A}^*(\mathbf{r},z,-\Delta\omega) e^{i[(-\Delta\omega)/\nu] z} e^{i\Delta\omega t} \frac{d\Delta\omega}{2\pi}$$
(3.1.25)

The forms of the integrals in Eq. 3.1.25 correspond to the inverse Fourier transforms with respect to the difference frequency,  $\Delta \omega$ .<sup>i</sup> We can therefore consider Eq. 3.1.25 to be equivalent to:

$$\tilde{E}(\mathbf{r},z,t) = e^{-ik_0 z} e^{i\omega_0 t} \mathcal{F}^{-1} \left\{ \hat{A}(\mathbf{r},z,\Delta\omega) e^{-i(\Delta\omega/\nu)z} \right\} + e^{ik_0 z} e^{-i\omega_0 t} \mathcal{F}^{-1} \left\{ \left[ \hat{A}(\mathbf{r},z,-\Delta\omega) e^{-i[(-\Delta\omega)/\nu]z} \right]^* \right\}$$
(3.1.26)

where  $\mathcal{F}^{-1}\{\}$  represents the inverse Fourier transform of the term in brackets.

If we define an envelope function  $A(\mathbf{r}, z, t) = \mathcal{F}^{-1}\{\hat{A}(\mathbf{r}, z, \Delta\omega) \exp[-i(\Delta\omega)/vz]\},^{ii}$  we can use the identity  $\mathcal{F}^{-1}\{\hat{f}^*(-\omega)\} = f^*(t)$  [112] to see that Eq. 3.1.26 predicts the behavior of a polychromatic field to be

<sup>&</sup>lt;sup>i</sup>Note that the form of our inverse Fourier transforms in Eq. 3.1.25 do not agree with the notation employed by Boyd.[102] Boyd takes the convention that the inverse Fourier transform is achieved by multiplying the frequency domain function by  $\exp(-i\omega t)$  and integrating over  $\omega$ . While this convention is consistent with some authors,[113, 117] we will follow the convention in which the Fourier transform is performed by multiplying the time domain function by  $\exp(-i\omega t)$  and the inverse Fourier transform by multiplying by  $\exp(i\omega t)$ .[112, 114]

<sup>&</sup>lt;sup>ii</sup>We can simplify the assignment of  $A(\mathbf{r}, z, t)$  by considering the contributions from  $\exp[-i(\Delta\omega/v)z]$  to be negligible. We will generally adopt this approximation going forward, i.e. we will generally consider  $\mathcal{F}\{A(\mathbf{r}, z, t)\} \approx \hat{A}(\mathbf{r}, z, \Delta\omega)$ . However, we will note when it is convenient to relax this approximation. It is also trivial to explicitly account for this additional phase component numerically in a manner similar to Wang and Wang.[118]

consistent with that of the time domain envelope,  $A(\mathbf{r}, z, t)$ , propagating with carrier frequency,  $\omega_0$ :

$$\tilde{E}(\mathbf{r}, z, t) = A(\mathbf{r}, z, t)e^{-i(k_0 z - \omega_0 t)} + \text{c.c.}$$
(3.1.27)

Ignoring the factor of 1/2 as previously discussed, [102, 103, 110] this result is identical to the solution reported by Akhmanov et al. [103] The form of Eq. 3.1.27 suggests that it is satisfactory to consider the propagation of the complex field,  $A(\mathbf{r}, z, t) \exp[-i(k_0 z - \omega_0 t)]$ , and determine the resulting real-valued field through the predicted result and its complex conjugate.

Akhmanov et al give some insight into the nature of this pulse envelope A(t). They treat the function as the product of a real-valued envelope,  $\rho(t)$ , and a phase factor,  $\exp[i\phi(t)]$ . Within this notation, a transformlimited pulse can be considered to be characterized by a pulse envelope that is real-valued, and thereby characterized by a phase dependence of  $\phi(t) = 0$ . Other functionalities of  $\phi(t)$  yield different behaviors associated with the resulting phase-modulation. For example, phase-modulation of the time-domain envelope according to  $\phi(t) = -(\alpha/2)t^2$  corresponds to frequency modulation of the form  $\omega(t) - \omega_0 = d\phi(t)/dt = -\alpha t$ , yielding a linear chirp.[103]

As we see in the derivation of Eq. 3.1.27, the formation of ultrashort pulses is predicated on forming polychromatic electric fields. However, not only does pulse formation require many contributing frequencies, it also requires a fixed phase relation between the various frequency components in order to permit its characterization as a single pulse envelope. This is demonstrated in Fig. 3.1. This figure illustrates how the linear combination of various monochromatic fields can introduce a beat pattern in the total field observed, resulting in a pulse train. More meaningfully, when there is a characteristic distribution applied to the amplitudes of the monochromatic fields, mode locking by defining a fixed phase relationship between the different frequency components results in the transformation of this distribution into the time-domain, as predicted by Eq. 3.1.26. On the other hand, the loss of a phase relation between the different frequency components prevents a well-defined pulse from forming.

This regime in which the time-dependent behavior of a pulse is adequately described as the product of an envelope function and a term describing the rapid oscillations at a central or carrier frequency is referred to as the slowly varying amplitude or envelope approximation. Unfortunately, there is a limit to the conditions for which this approximation is considered valid: if the pulse envelope is too short relative to the oscillations at the carrier frequency, it is no longer appropriate to consider the time evolution of the envelope distinct from the rapid oscillations at the carrier frequency. The limit to this approximation is therefore related to the pulse width,  $\tau$ , and the carrier frequency,  $\omega_0$ . However, different authors cite different limits to this

regime. Akhmanov et al cite the limiting bounds as  $\tau\omega_0 \approx 20\pi$ .[103] On the other hand, Brabec and Krausz present calculations that suggest the limit may even be as low as  $\tau\omega_0 \approx 2\pi$ .[115] For fields in the regime we commonly work on the Wright group's femtosecond system (in the visible and near infrared),  $\omega_0$  is in the range of  $1 - 4 \times 10^{15}$  Hz.<sup>i</sup> According to Akhmanov et al, the shortest pulse lengths at these frequencies for which the slowly varying envelope approximation would be considered valid are 16-60 fs; by Brabec and Krausz's conditions, though, the slowly varying envelope approximation should be reasonable down to pulse lengths as short as 2-6 fs.[103, 115]

#### 3.2 Polarizations

Throughout § 3.1, we have focused on electromagnetic radiation propagating through free space. In such a scenario, Maxwell's equations predict the propagation of electric and magnetic fields to behave perfectly as a wave: propagating without distortion to the wave function. This behavior is expected as the propagation predicted by Maxwell's equations exhibits a functional form consistent with Eq. 3.1.1:<sup>ii</sup>

$$\frac{\partial^2}{\partial z^2} \tilde{\mathbf{E}}(\mathbf{r}, z, t) = \frac{1}{v^2} \frac{\partial^2}{\partial t^2} \tilde{\mathbf{E}}(\mathbf{r}, z, t)$$
(3.2.1)

However, when light propagates through matter, the field can interact with the dipoles of the material, perturbing the potential energy of the system according to:[99, 109]

$$\tilde{V} = \mu \cdot \tilde{\mathbf{E}} \tag{3.2.2}$$

where  $\mu$  is the dipole operator, er. As the potential energy defined by Eq. 3.2.2 is of the form of a dot product, the relative orientation of the dipole and the field polarization is important; we will represent appropriate quantities as vectors until we note the assumptions that allow us to represent the system with scalar quantities.

The additional potential energy introduced by the interactions between the dipole and the field can be

<sup>&</sup>lt;sup>i</sup>Corresponding to wavelengths between approximately 500 nm and 1500 nm, which covers the bounds of the sum-frequency signal, sum-frequency idler, second-harmonic signal, second-harmonic idler, and signal processes.

<sup>&</sup>lt;sup>ii</sup>A result identical to Eq. 3.2.1 can also be predicted for magnetic fields.



Figure 3.1: Formation of a polychromatic electric field through summation of monochromatic fields. (A) and (B) demonstrate the beating that arises in the polychromatic field (B) from the summation of a discrete set of monochromatic fields (A). The beats in (B) correspond to a pulse train with pulses occurring at points where all contributing fields are in phase. (C) shows a Gaussian amplitude envelope over a distribution of frequencies, while (D) shows two scenarios for the phase relation between the different frequency components: all frequencies have the same initial phase (dashed line), or the frequencies exhibit random phases (solid line). The polychromatic field resulting from the mode-locked frequencies exhibits the Gaussian temporal profile in (E): the solid line is the polychromatic field amplitude, and the dashed line outlines the Gaussian profile predicted by Fourier transformation of the frequency domain envelope in (C), scaled to match the peak of the solid trace. The randomly phased frequencies yield the polychromatic field in (F).

treated as a perturbation to the Hamiltonian,  $\mathcal{H}$ , describing the system:<sup>i</sup>

$$\begin{aligned} \mathcal{H} &= \mathcal{H}_0 + \tilde{V} \\ &= \mathcal{H}_0 + \mu \cdot \tilde{\mathbf{E}} \end{aligned} \tag{3.2.3}$$

where  $\mathcal{H}_0$  is the unperturbed Hamiltonian of the system. The effects of this perturbed Hamiltonian are described by the Liouville equation:

$$\frac{\partial}{\partial t}\rho = \frac{1}{i\hbar}[\mathcal{H},\rho] \tag{3.2.4}$$

The resulting density matrix,  $\rho$ , permits prediction of the properties of the system, including the expectation value for the electric polarization induced by the field:

$$\langle \tilde{\mathbf{P}} \rangle = N \operatorname{tr}(\rho \mu)$$
 (3.2.5)

where N is the number density of the source of the dipole and tr is the trace.[99, 109] Going forward, we will simply express the polarization as  $\tilde{\mathbf{P}}$ .

As we have implied by the notation of  $\tilde{\mathbf{P}}$ , this resulting polarization will exhibit an oscillatory character, which we can consider to be of the form:[102, 109]

$$\tilde{\mathbf{P}}(t) = \mathbf{P}(\omega)e^{i\omega t} + \text{c.c.}$$
(3.2.6)

where  $\mathbf{P}(\omega)$  is the amplitude of the polarization at the frequency of oscillation,  $\omega$ . This polarization can couple to the electric field oscillating at the same frequency, modifying the manner in which the field propagates. The effect of this induced polarization upon the electric field manifests in the introduction of a new quantity: the displacement,  $\tilde{\mathbf{D}}(t)$ . The displacement describes the coupling of the electric field and induced polarization according to the relationship:

$$\dot{\mathbf{D}}(t) = \epsilon_0 \dot{\mathbf{E}}(t) + \dot{\mathbf{P}}(t) \tag{3.2.7}$$

in the mks system of units, or:

$$\tilde{\mathbf{D}}(t) = \tilde{\mathbf{E}}(t) + 4\pi \tilde{\mathbf{P}}(t) \tag{3.2.8}$$

in the cgs or esu systems. In Eq. 3.2.7,  $\epsilon_0$  is the permittivity of free space.[101, 102, 109]

Thus far, we have only allowed for linear interactions between the field and the material. However, it is

<sup>&</sup>lt;sup>i</sup>Note that an additional perturbation can be introduced corresponding to the interactions that provide relaxation pathways for the density matrix predicted by the Hamiltonian. Such a relaxation matrix is not explicitly noted in Eq. 3.2.3.

possible to expand the perturbation to higher terms in the electric field, resulting in a series of higher order perturbations to the density matrix. The  $n^{\text{th}}$ -order density matrix,  $\rho(n)$ , will contribute a corresponding  $n^{\text{th}}$ -order polarization,  $\tilde{P}^{(n)}(t)$ . The total polarization will be the sum of all such resulting polarizations:[109]

$$\tilde{\mathbf{P}}(t) = \tilde{\mathbf{P}}^{(1)}(t) + \tilde{\mathbf{P}}^{(2)}(t) + \tilde{\mathbf{P}}^{(3)}(t) + \cdots$$

$$= \tilde{\mathbf{P}}^{\mathrm{L}}(t) + \tilde{\mathbf{P}}^{\mathrm{NL}}(t)$$
(3.2.9)

In the second line of Eq. 3.2.9, we have introduced the notations  $\tilde{\mathbf{P}}^{L}(t)$  and  $\tilde{\mathbf{P}}^{NL}(t)$  to describe the linear and nonlinear polarizations, respectively: the linear polarization,  $\tilde{\mathbf{P}}^{L}(t)$ , corresponds to the first order term,  $\tilde{\mathbf{P}}^{(1)}(t)$ , and the nonlinear polarization,  $\tilde{\mathbf{P}}^{NL}(t)$ , generalizes the higher order terms. This reduction will prove useful going forward: we will show that the linear term describes absorptive and dispersive effects on the propagation of an ultrashort pulse, while we will generally only need to consider one term from the series represented by  $\tilde{\mathbf{P}}^{NL}(t)$ .

Since the  $n^{\text{th}}$ -order polarization requires an expansion of the Hamiltonian to provide an  $n^{\text{th}}$ -order perturbation, we can also express the series in Eq. 3.2.9 as an expansion in increasingly higher powers of the field. This expansion describes the increasingly higher order of interactions between the total field and the material:[99, 102, 109]

$$\tilde{\mathbf{P}}(t) = \epsilon_0 \chi^{(1)} \cdot \tilde{\mathbf{E}}(t) + \epsilon_0 \chi^{(2)} \colon \tilde{\mathbf{E}}^2(t) + \epsilon_0 \chi^{(3)} \colon \tilde{\mathbf{E}}^3(t) + \cdots$$
(3.2.10)

in the mks system, or:

$$\tilde{\mathbf{P}}(t) = \chi^{(1)} \cdot \tilde{\mathbf{E}}(t) + \chi^{(2)} \colon \tilde{\mathbf{E}}^2(t) + \chi^{(3)} \colon \tilde{\mathbf{E}}^3(t) + \cdots$$
(3.2.11)

when the quantities are in cgs or esu units. Note that the : operator in Eq. 3.2.10 and 3.2.11 provides an operation analogous to the dot product between the tensor,  $\chi^{(n)}$ , and the electric field product,  $\tilde{\mathbf{E}}^{n}(t)$ .[109] Within the expansions of Eq. 3.2.10 and 3.2.11, the coefficient  $\chi^{(n)}$  is the  $n^{\text{th}}$ -order susceptibility. These susceptibilities are appropriately considered to be  $(n + 1)^{\text{th}}$ -order tensors. However, we will generally be able to apply approximations or manipulations that allow us to consider the susceptibility as an effective scalar value representing a linear combination of the appropriate elements of the tensor as they are sampled according to the experimental conditions.[102, 106, 119] Under such approximations, it is sufficient to consider the following alternatives to Eq. 3.2.10 and 3.2.11, in which the relationship between the polarization and
the field is treated as a scalar relationship:

$$\tilde{P}(t) = \epsilon_0 \chi^{(1)} \tilde{E}(t) + \epsilon_0 \chi^{(2)} \tilde{E}^2(t) + \epsilon_0 \chi^{(3)} \tilde{E}^3(t) + \cdots$$
(3.2.12a)

$$\tilde{P}(t) = \chi^{(1)}\tilde{E}(t) + \chi^{(2)}\tilde{E}^2(t) + \chi^{(3)}\tilde{E}^3(t) + \cdots$$
(3.2.12b)

We will discuss how this can be approached in birefringent media, as well as provide a brief discussion of the contributions to the susceptibility, in § 3.2.2. However, in the context of later sections, it will be sufficient to consider the susceptibility at the phenomenological level of Eq. 3.2.10 and 3.2.11, or Eq. 3.2.12a and 3.2.12b.

Unfortunately, the expansion of the polarization into a series in the manner of Eq. 3.2.10 - 3.2.12b introduces another inconsistency in the treatment of different authors. The forms that have been presented are consistent with the convention of Boyd,[102] in which the polarization as an expansion of increasing powers in the electric field is considered to be a time-domain description of the interaction. In contrast, Shen expands the polarization as increasing powers of the field in the frequency domain.[109] The significance of this distinction relates to the behavior of multiplication between the time and frequency domains. For two functions f(t) and g(t) with Fourier transforms  $\mathcal{F}{f(t)} = \hat{f}(\omega)$  and  $\mathcal{F}{g(t)} = \hat{g}(\omega)$ , we can define the following relationships:[112]

$$\mathcal{F}\{f(t)g(t)\} = \frac{1}{2\pi} (\hat{f} * \hat{g})(\omega)$$
(3.2.13a)

$$\mathcal{F}\left\{(f*g)(t)\right\} = \hat{f}(\omega)\hat{g}(\omega) \tag{3.2.13b}$$

where \* denotes the convolution of the functions.<sup>i</sup>

As can be seen from the relationships of Eq. 3.2.13a and 3.2.13b, Boyd and Shen are implying specific behaviors in their respective conventions. By taking the expansion of the polarization in powers of the field as the frequency domain description, Shen imposes behavior in the time-domain corresponding to the convolution of the fields and the material response, which allows the polarization to exhibit transient behavior corresponding to relaxation properties in the material.[99, 109] In contrast, Boyd's treatment describes an instantaneous response of the medium, with a response - the polarization - that requires the presence of the electric field to be sustained; such behavior is consistent with a nonresonant interaction, in which the field

<sup>&</sup>lt;sup>i</sup>It should be noted that the factor of  $2\pi$  in Eq. 3.2.13a is necessary to compensate for the additional normalization constant that arises from the pair of Fourier transforms necessary to yield  $\hat{f}$  and  $\hat{g}$ . This normalization constant should be changed according to the normalization convention being taken. Upon a change in normalization convention, it is also necessary to consider whether a normalization coefficient needs to be introduced to Eq. 3.2.13b.[114]

is not resulting in the formation of long-lived polarizations. Within this treatment, the polarization must be able to be defined in the frequency domain as a convolution; this is consistent with Boyd's treatment of the polarization for a  $\chi^{(2)}$  experiment corresponding to the summation:[102]

$$P_i(\omega_n + \omega_m) = \sum_{jk} \sum_{(nm)} \chi^{(2)}_{ijk}(\omega_n + \omega_m, \omega_n, \omega_m) E_j(\omega_n) E_k(\omega_m)$$
(3.2.14)

where i, j, and k correspond to the available polarization states and the summation over (nm) corresponds to summing over m and n in such a manner so as to maintain a constant sum,  $\omega_n + \omega_m$ . Ignoring the symmetry (the terms corresponding to summation of i, j, and k in Eq. 3.2.14) and extending the summation to the form of a continuous integral, this is consistent with the convolution:

$$P(\omega) = \int \chi^{(2)} E(\omega') E(\omega - \omega') d\omega' \qquad (3.2.15)$$

In contrast to Shen's approach, this treatment allows us to explicitly consider the permutations resulting from the frequency content of ultrafast pulses. Since the problems we will be seeking to address predominantly involve interactions reasonably considered nonresonant, we will adopt Boyd's convention moving forward. This convention has also been adopted by a variety of other authors concerned with systems involving both resonant and nonresonant response from the medium.[120–124]

If we wish adopt this treatment in order to consider fields with a broad frequency distribution, we will also need to consider the possibility of the polarization exhibiting a frequency distribution. Examining Eq. 3.2.6, it is clear that this mathematical description can be considered to describe a polarization exhibiting a single frequency, thereby coupling to a monochromatic field. To extend to the scenario in which multiple frequencies are present, we can treat the polarization defined by Eq. 3.2.6 in a manner similar to our extension of Eq. 3.1.13 to Eq. 3.1.14 in § 3.1.2, yielding the summation:[102]

$$\tilde{P}(\mathbf{r}, z, t) = \sum_{n} P(\mathbf{r}, z, \omega_n) e^{i\omega_n t} + \text{c.c.} \qquad \{n : \omega_n \ge 0\}$$
(3.2.16)

Note that we have also introduced the spatial dependence  $(\mathbf{r}, z)$  we previously allowed the field to exhibit (see Eq. 3.1.15). Given our previous treatment of polychromatic fields, though, we expect that we might be able to extend this summation in a manner similar to the result of Eq. 3.1.27, allowing us to treat the polarization as the product of some envelope function,  $p(\mathbf{r}, z, t)$ , and rapid oscillations around the carrier frequency:

$$\tilde{P}(\mathbf{r}, z, t) = p(\mathbf{r}, z, t)e^{i\omega_0 t} + \text{c.c.}$$
(3.2.17)

If such a treatment holds, we would predict that the envelope function would contain information on the frequency distribution in a manner analogous to the field envelope,  $A(\mathbf{r}, z, t)$ :

$$\hat{p}(\mathbf{r}, z, \Delta \omega) = P(\mathbf{r}, z, \omega_0 + \Delta \omega)$$
(3.2.18)

where  $P(\mathbf{r}, z, \omega_0 + \Delta \omega) = P(\mathbf{r}, z, \omega)$  is the coefficient in Eq. 3.2.16 and  $\hat{p}(\mathbf{r}, z, \Delta \omega)$  is defined as the Fourier transform of  $p(\mathbf{r}, z, t)$ ,  $\hat{p}(\mathbf{r}, z, \Delta \omega) = \mathcal{F}\{p(\mathbf{r}, z, t)\}$ .

If we consider a representative nonlinear process, we can show that the form of Eq. 3.2.17 is just a consequence of the pulse-like propagation of ultrafast electric fields, with  $p(\mathbf{r}, z, t)$  inheriting a form corresponding to the envelope functions of the driving fields,  $A(\mathbf{r}, z, t)$ . We will show how this might be evaluated for a representative example - second-harmonic generation - though extending this treatment to other nonlinear experiments simply requires an appropriate expansion of the total electric field.

In the case of second-harmonic generation, we are concerned with a total electric field consisting of contributions from two fields: the fundamental, characterized by a carrier frequency  $\omega_0 = \omega_1$ , and the second harmonic, which will exhibit a carrier frequency  $\omega_0 = \omega_2 = 2\omega_1$ . If we consider these fields to be defined in a manner consistent with Eq. 3.1.27, we expect the fields to propagate in the manner:

$$\tilde{E}_1(\mathbf{r}, z, t) = A_1(\mathbf{r}, z, t)e^{-ik_1 z}e^{i\omega_1 t} + A_1^*(\mathbf{r}, z, t)e^{ik_1 z}e^{-i\omega_1 t}$$
(3.2.19a)

$$\tilde{E}_{2}(\mathbf{r}, z, t) = A_{2}(\mathbf{r}, z, t)e^{-ik_{2}z}e^{i\omega_{2}t} + A_{2}^{*}(\mathbf{r}, z, t)e^{ik_{2}z}e^{-i\omega_{2}t}$$
(3.2.19b)

where Eq. 3.2.19a describes the fundamental field (noted  $\tilde{E}_1$ ) and Eq. 3.2.19b represents the second-harmonic field (noted  $\tilde{E}_2$ ). We can then define the total field,  $\tilde{E}$ :

$$\tilde{E}(\mathbf{r}, z, t) = \tilde{E}_{1}(\mathbf{r}, z, t) + \tilde{E}_{2}(\mathbf{r}, z, t)$$

$$= A_{1}(\mathbf{r}, z, t)e^{-ik_{1}z}e^{i\omega_{1}t} + A_{1}^{*}(\mathbf{r}, z, t)e^{ik_{1}z}e^{-i\omega_{1}t}$$

$$+ A_{2}(\mathbf{r}, z, t)e^{-ik_{2}z}e^{i\omega_{2}t} + A_{2}^{*}(\mathbf{r}, z, t)e^{ik_{2}z}e^{-i\omega_{2}t}$$
(3.2.20)

Substituting the total field of Eq. 3.2.20 into the appropriate term of the polarization expansion in Eq. 3.2.12a,  $\tilde{P}^{(2)}(\mathbf{r}, z, t) = \epsilon_0 \chi^{(2)} (\tilde{E}(\mathbf{r}, z, t))^2$ , and expanding the quadratic dependence on the total field yields

<sup>&</sup>lt;sup>i</sup>Note that the appropriate term from Eq. 3.2.12b,  $\tilde{P}^{(2)}(\mathbf{r}, z, t) = \chi^{(2)}(\tilde{E}(\mathbf{r}, z, t))^2$ , would be treated in an identical manner,

the result of Eq. 3.2.21:

$$\tilde{P}^{(2)}(\mathbf{r},z,t) = \epsilon_0 \chi^{(2)} \left[ A_1(\mathbf{r},z,t) A_1(\mathbf{r},z,t) e^{-i2k_1 z} e^{i2\omega_1 t} + 2A_1(\mathbf{r},z,t) A_2(\mathbf{r},z,t) e^{-i(k_1+k_2)z} e^{i(\omega_1+\omega_2)t} \right. \\ \left. + 2A_1(\mathbf{r},z,t) A_1^*(\mathbf{r},z,t) + 2A_1(\mathbf{r},z,t) A_2^*(\mathbf{r},z,t) e^{i(k_2-k_1)z} e^{-i(\omega_2-\omega_1)t} \right. \\ \left. + A_2(\mathbf{r},z,t) A_2(\mathbf{r},z,t) e^{-i2k_2 z} e^{i2\omega_2 t} + 2A_2(\mathbf{r},z,t) A_1^*(\mathbf{r},z,t) e^{-i(k_2-k_1)z} e^{i(\omega_2-\omega_1)t} \right.$$
$$\left. + 2A_2(\mathbf{r},z,t) A_2^*(\mathbf{r},z,t) + A_1^*(\mathbf{r},z,t) A_1^*(\mathbf{r},z,t) e^{i2k_1 z} e^{-i2\omega_1 t} \right. \\ \left. + 2A_1^*(\mathbf{r},z,t) A_2^*(\mathbf{r},z,t) e^{i(k_1+k_2)z} e^{-i(\omega_1+\omega_2)t} + A_2^*(\mathbf{r},z,t) A_2^*(\mathbf{r},z,t) e^{i2k_2 z} e^{-i2\omega_2 t} \right]$$

Examining this expression, we observe five frequency contributions: two DC components describing optical rectification; the second harmonics of  $\omega_1$  and  $\omega_2$ ; the sum frequency,  $\omega_1 + \omega_2$ ; and, the difference frequency,  $\omega_2 - \omega_1$ . Considering the oscillatory terms, Eq. 3.2.21 consists of two terms at each frequency that simply differ in phase; we will consider these terms to reflect four complex polarizations - the terms oscillating according to  $\exp(i\omega t)$  - and their complex conjugates, the inclusion of which ensures the polarization is real valued. Thus, we can consider the four oscillating polarizations to be of a form analogous to Eq. 3.1.27:

$$\tilde{P}_{(2\omega_1)}^{(2)}(\mathbf{r}, z, t) = \epsilon_0 \chi^{(2)} \left[ A_1(\mathbf{r}, z, t) A_1(\mathbf{r}, z, t) e^{-i2k_1 z} e^{i2\omega_1 t} + \text{c.c} \right]$$
(3.2.22a)

$$\tilde{P}_{(2\omega_2)}^{(2)}(\mathbf{r}, z, t) = \epsilon_0 \chi^{(2)} \left[ A_2(\mathbf{r}, z, t) A_2(\mathbf{r}, z, t) e^{-i2k_2 z} e^{i2\omega_2 t} + \text{c.c.} \right]$$
(3.2.22b)

$$\tilde{P}_{(\omega_2+\omega_1)}^{(2)}(\mathbf{r},z,t) = 2\epsilon_0 \chi^{(2)} \left[ A_1(\mathbf{r},z,t) A_2(\mathbf{r},z,t) e^{-i(k_1+k_2)z} e^{i(\omega_1+\omega_2)t} + \text{c.c.} \right]$$
(3.2.22c)

$$\tilde{P}_{(\omega_2-\omega_1)}^{(2)}(\mathbf{r},z,t) = 2\epsilon_0 \chi^{(2)} \left[ A_2(\mathbf{r},z,t) A_1^*(\mathbf{r},z,t) e^{-i(k_2-k_1)z} e^{i(\omega_2-\omega_1)t} + \text{c.c} \right]$$
(3.2.22d)

Therefore, our assumption in treating the polarization in the manner of Eq. 3.2.17 appears justified. Within this treatment, we can consider the envelope function of the polarization,  $p(\mathbf{r}, z, t)$ , exhibiting contributions from:  $\epsilon_0$ , if appropriate;  $\chi^{(n)}$ , in the general case; an appropriate product of the temporal envelope functions of the driving fields,  $A(\mathbf{r}, z, t)$ ; and, a term describing the cooperation of the wave vectors, k, of the driving fields. We will see in § 3.4 that coupling between a field and polarization introduces a wave vector associated with the field, which together with this sum of wave vectors yields a term,  $\Delta k$ . The quantity  $\Delta k$  describes the quality of momentum conservation in the nonlinear process and, under conditions of non-zero  $\Delta k$ , can result in a decrease of the efficiency of the nonlinear process (see § 3.2.1)

yielding the same form as the result in Eq. 3.2.21, only omitting the factor of  $\epsilon_0$ .

### 3.2.1 Phase Matching of Nonlinear Experiments

As shown in Eq. 3.2.21 and 3.2.22a - 3.2.22d, the characteristic frequencies of polarizations induced by an electric field must satisfy energy conservation: namely, the frequency of the polarization - and the field to which it couples - must correspond to the appropriate sum of the carrier frequencies of the interacting pulses. For example, the generation of the second harmonic of  $\omega_1$  (Eq. 3.2.21 and 3.2.22a) must initiate a polarization at the energy-conserving frequency,  $\omega_2 = \omega_1 + \omega_1$ . Similarly, it can be shown that optical parametric generation/amplification and sum-frequency generation must satisfy the energy conservation corresponding to  $\omega_3 = \omega_2 + \omega_1$ : in optical parametric generation/amplification,  $\omega_3$ ,  $\omega_2$ , and  $\omega_1$  would correspond to the pump, signal, and idler (assuming  $\omega_2 > \omega_1$ ), while in sum-frequency generation,  $\omega_3$  would be the sum frequency generated from  $\omega_1$  and  $\omega_2$ .

However, Eq. 3.2.21 and 3.2.22a - 3.2.22d illustrate that the interaction of the frequencies in the nonlinear medium also imposes a momentum upon the induced polarization: for example, the polarization oscillating at  $2\omega_1$  also exhibits spatial oscillations at  $2k_1$ . We will see in § 3.4.1 that the field coupling to the polarization at  $2\omega_1 = \omega_2$  propagates with a dependence upon  $\exp(ik_2z)\tilde{P}_{(2\omega_1)}$ .<sup>i</sup> Together, the spatial oscillation of the field,  $\exp(ik_2z)$  and the oscillation of the polarization,  $\exp(-i2k_1z)$ , determine the efficiency of the nonlinear process: when perfect spatial resonance exists between the field and the corresponding polarization,[106] corresponding to the equality condition  $k_2 = 2k_1$ , enhancement of the field is optimized.[106, 125–129] This condition of momentum conservation can also be described by the quantity referenced previously,  $\Delta k$ , the phase mismatch; in our example of second-harmonic generation,  $\Delta k$  would be defined as the quantity:  $\Delta k = k_2 - 2k_1$ .<sup>ii</sup>[102, 125, 127]

The phase mismatch,  $\Delta k$ , can have a wide variety of consequences on nonlinear experiments. Within the limit of low depletion, it has been shown that the efficiency of the nonlinear process scales according to  $\frac{\sin(\Delta kL/2)}{(\Delta kL/2)}$  on the amplitude (field) level and  $\frac{\sin^2(\Delta kL/2)}{(\Delta kL/2)^2}$  on the intensity level, where L is the length of the nonlinear crystal; this dependence is also reported as the sinc function,  $\frac{\Delta kL}{2}$  or  $\frac{\sin^2(\Delta kL/2)}{[99, 102, 130]}$  The fields generated by poorly phase matched processes have also been shown to exhibit phase modulation,[131, 132] the magnitude of which scales according to  $\frac{\Delta kL}{2}$ .[132] The final consequence of the phase mismatch we will mention is the limit imposed upon the phase matching bandwidth for a given process: the phase matching bandwidth is commonly taken to be the bandwidth over which  $\Delta k$  is

<sup>&</sup>lt;sup>i</sup>In § 3.4.1 we will actually end up with a result in the frequency domain, such that the evolution of the field at  $\omega_2$  would depend upon  $\exp(ik_2z)P_{(2\omega_1)}$ , rather than  $\exp(ik_2z)\tilde{P}_{(2\omega_1)}$ . However, within the CW limit upon which this discussion is based, the distinction between the two is irrelevant.

<sup>&</sup>lt;sup>ii</sup>Note that some authors define  $\Delta k$  as the negative value of the definition here; however, as the notation of Barnes and Corcoran shows, [125] it is generally the magnitude of  $\Delta k$  that matters.

less than  $\pi/L$ .[125, 127] This limit is imposed in both collinear[126, 127, 133] and noncollinear[120, 125, 127, 133–138] geometries. There are a variety of strategies that have been introduced to bypass phase-matching considerations; these include thin samples[139] and epi-directional detection,[140–142] pulse-shaping[143–145] to control phase in direct analogy to nuclear magnetic resonance,[144, 145] and high numerical aperture objectives to introduce a wide angular distribution to the excitation fields.[139, 140, 146–148] Otherwise, phase matching must be considered in both collinear and noncollinear experiments.[106]

In the collinear geometry, it is sufficient to consider a scalar representation of the phase matching condition.[106] We will show in § 3.4.1 that the wave vector can be defined  $k = n(\omega)\omega/c$  (Eq. 3.4.14), where  $n(\omega)$ is the refractive index of the material at frequency  $\omega$ . Generalizing our previous example, we could consider the frequency conversion process  $\omega_3 = \omega_2 + \omega_1$ ,  $\omega_3 \ge \omega_2 \ge \omega_1$ . For such a process, the phase matching condition would take the form  $k_3 = k_2 + k_1$ . By our definition of k, we can then frame the momentum conservation as  $n(\omega_3)\omega_3 = n(\omega_2)\omega_2 + n(\omega_1)\omega_1$ . In the manner of Boyd,[102] we can manipulate this relationship to show that this momentum conservation relation would require:  $n(\omega_3) - n(\omega_2) = [n(\omega_1) - n(\omega_2)](\omega_1/\omega_3)$ . Recognizing  $\omega_1/\omega_3$  must always be positive, the only way this equality can hold true is for  $n(\omega_3) > n(\omega_2)$ and  $n(\omega_1) > n(\omega_2)$  or for  $n(\omega_3) < n(\omega_2)$  and  $n(\omega_1) < n(\omega_2)$ ; under our assignment of  $\omega_3 \ge \omega_2 \ge \omega_1$ , this condition cannot be satisfied under normal dispersion, for which  $n(\omega)$  increases monotonically with  $\omega$ .[102, 106] While anomalous dispersion could permit momentum conservation,[102, 106] its application is not practical as it would require sufficiently reduced transparency that conversion is inefficient.[106] Instead, the birefringence of anisotropic materials is generally used as an alternative strategy for satisfying momentum conservation in collinear geometries.[102, 106]

Anisotropic crystals may be either biaxial or uniaxial.[106] In biaxial crystals, there are three unique refractive indices available, making consideration of effective index and phase matching more complicated.[106] However, there are a variety of biaxial crystals employed for different frequency conversion processes, such as bismuth triborate (BiBO),[149–151] lithium triborate (LBO),[152–157], and potassium titanyl phosphate (KTP).[158, 159] In contrast, uniaxial crystals exhibit only two refractive indices, the ordinary index  $(n_o)$ and the extraordinary index  $(n_e)$ ; the relative values of these refractive indices determine the nature of the uniaxial crystal: for negative uniaxial crystals,  $n_o > n_e$ , while  $n_e > n_o$  for positive uniaxial crystals. Many common uniaxial crystals - including potassium dihydrogen phosphate (KDP),[152, 160–163]  $\beta$ -barium borate (BBO), lithium niobate,[128, 133, 161, 162, 164, 165] silver gallium sulfide,[106, 166] and silver gallium selenide[106, 166] - are negative uniaxial crystals; BBO in particular has achieved wide-spread usage due to its high birefringence, low dependence of index upon temperature,[167] high damage threshold,[167, 168] low group velocity mismatch, and large nonlinearity.[126, 131, 168] In addition to the broad usage of BBO, the behavior of uniaxial crystals is much more straightforward; we will consequently focus our discussion as related to birefringent media upon uniaxial crystals.

The indices of uniaxial crystals can be described according to the relation between the polarization state of the field, the wave vector (**k**) of the field, and the optic axis of the crystal (**Z**).[106] The orientation of the wave vector with respect to the crystalline axes is described by two angles: the phase matching angle,  $\theta$ , is the angle between the optic axis, **Z**, and the wave vector, while the azimuthal angle,  $\phi$  is the angle between the **X**-axis of the crystal and the projection of the wave vector in the crystal's **XY** plane.[106] The field can be described as an ordinary or extraordinary beam, meanwhile, based upon the alignment of its polarization state with the principal plane formed by the optic axis and the wave vector. If the polarization of the field is perpendicular to this plane, the field is an ordinary beam and experiences the ordinary refractive index,  $n_o$ . On the other hand, if the polarization of the field is parallel to the principal plane, it is an extraordinary beam; in contrast to the ordinary polarization, extraordinary beams experience an index that exhibits a dependence upon  $\theta$ :

$$n(\theta) = n_o \sqrt{\frac{1 + \tan^2 \theta}{1 + (n_o/n_e)^2 \tan^2 \theta}}$$
(3.2.23)

It should be noted that for uniaxial crystals, the azimuthal angle,  $\phi$ , does not play a role in phase matching considerations.[106] This angle is important, however, in determining the symmetry of the crystal's nonlinear susceptibility; this is discussed in § 3.2.2.

The dependence of the effective refractive index of an extraordinary beam on  $\theta$  permits one method of phase matching, angle tuning. This strategy seeks to optimize  $\theta$  so as to minimize the phase mismatch associated with a particular process involving a certain set of frequencies. Due to the monotonic increase in index with increasing frequency, it is necessary to polarize the fields such that the highest frequency involved in the process is polarized such that it experiences the lower refractive index of the crystal:[102, 126, 169] in a negative uniaxial crystal, this requires extraordinary polarization for the highest frequency, while for a positive uniaxial crystal the highest frequency would be an ordinary beam. Phase matching requires the polarization of at least one of the two lower frequencies to be perpendicular to the polarization of the highest frequency; however, it is possible to have the two lower frequencies have either parallel (Type I) or perpendicular (Type II) polarization.[102, 106]

In BBO, these two options each present different advantages and disadvantages. Under parallel polarization, the signal and idler exhibit group velocities that cause both pulses to move away from the pump in the same direction.[126] As a consequence, Type I phase matching exhibits a finite pulse splitting length: when the crystal length is extended beyond the pulse splitting length, reduced gain is observed; gain ceases when the length of the crystal is approximately twice the pulse splitting length.[170, 171] However, Type I phase matching also exhibits lower group velocity mismatches than Type II, which together with the higher effective nonlinearity, yields higher gain.[168, 172] This permits usage of lower pump energies.[173] Type I also exhibits one disadvantage associated with the lower group velocity mismatches: since the signal and idler exhibit a lower group velocity mismatch, Type I typically yields significantly greater amplification bandwidths, particularly when tuned close to degeneracy ( $\omega_2 \approx \omega_1$ ).[173, 174] Unfortunately, the increased bandwidth is not accompanied by a decrease in pulse duration, and the generated pulses are generally not near the transform limit.[175, 176]

In contrast, the perpendicular polarization of signal and idler in Type II phase matching causes the signal and idler to move in opposite directions with respect to the pump across much of the near infrared.[126, 171, 177] Within this regime, the gain represented by the pump pulse envelope synchronizes the propagation of the idler and signal pulses: this synchronization causes amplification similar to steady state behavior,[103] with exponential gain continuing even past the pulse splitting length.[126, 177] The steady state operation of this process yields consistent bandwidths across the tuning range.[173] While the bandwidths tend to be smaller than those obtained through Type I phase matching,[126] the pulses generated tend to be closer to the transform limit.[174, 175] While neither broadening nor gain saturation is expected within this regime,[170] pump depletion has been reported to have effects on the temporal characteristics of the generated pulses.[168, 171] As noted before, Type II phase matching is possible with either the signal or the idler polarized as an extraordinary beam; however, polarizing the idler (the lower frequency) parallel to the pump generally proves more feasible as it requires less birefringence for successful phase matching.[102]

One complication associated with angle tuning is spatial walk-off: due to the birefringence experienced by extraordinary beams when  $\theta \neq 0^{\circ}$ , 90°, these beams exhibit a walk-off angle that is dependent upon the phase matching angle,  $\theta$ .[102, 106] This walk-off angle describes the relative angle between the wave vector and the Poynting vector,[106] the direction of flow for the energy density of the wave.[101, 106] This walk-off reduces the efficiency of the nonlinear process.[102] Temperature tuning is a strategy that has been employed to overcome this limitation.[102, 126] The strategy is based upon the dependence of refractive index upon temperature: while maintaining a constant phase matching angle,  $\theta = 90^{\circ}$ , temperature is adjusted to tune the refractive indices so as to achieve phase matching.[102] By operating at  $\theta = 90^{\circ}$ , walk off is avoided. This has been demonstrated in crystals such as LBO[153, 155, 170] and KDP.[127]

In contrast to collinear geometries, phase matching in a noncollinear geometry requires consideration of the directionality associated with the wave vector, **k**. While this method is implemented in secondorder frequency mixing processes, [125, 129, 152, 154, 171, 178–184] it is also of relevance in four-wave mixing processes that access the third-order nonlinear susceptibility. In these third-order experiments, the BOXCARS geometry - in which the three excitation fields are aligned so as to occupy three of the four corners of a rectangle - is a common starting point for determining appropriate phase matching conditions.[120, 121, 135, 147, 185–189] Four-wave mixing experiments of this manner can be described according to the generalized process:

$$\omega_4 = c_1 \omega_1 + c_2 \omega_2 + c_3 \omega_3 \tag{3.2.24}$$

where the set  $\{c_i\}$  describes the number and phase of interactions of each field in the experiment. For a given alignment, each field would have a characteristic wave vector  $\mathbf{k}_i$ ; according to the phase matching of the experiment, these wave vectors would cooperate to yield emission with a wave vector,  $\mathbf{k}'_4$ , satisfying the equation:

$$\mathbf{k}_{4}' = c_1 \mathbf{k}_1 + c_2 \mathbf{k}_2 + c_3 \mathbf{k}_4 \tag{3.2.25}$$

However, the magnitude of  $\mathbf{k}'_4$  may not necessarily be consistent with a field of frequency  $\omega_4$ . If the frequency  $\omega_4$  consistent with the energy conservation condition of Eq. 3.2.24 corresponds to a wave vector magnitude  $|\mathbf{k}_4| = n(\omega_4)\omega_4/c$ , the phase mismatch,  $\Delta k$ , for the process corresponds to the magnitude of the difference between  $\mathbf{k}'_4$  and the projection of  $\mathbf{k}_4$  on  $\mathbf{k}'_4$ :

$$\Delta k = \left| \mathbf{k}_4' - \frac{|\mathbf{k}_4|}{|\mathbf{k}_4'|} \mathbf{k}_4' \right|$$
(3.2.26)

Proper phase matching can be determined by identifying the angles that minimize the difference on the right hand side of Eq. 3.2.26. It should be noted that these angles correspond to the internal angles; in order to consider a mask for external alignment, it is necessary to consider the effects of refraction at the interface of the sample.

### 3.2.2 Susceptibilities

As we noted above, we will generally find it sufficient to consider the susceptibilities at the phenomenological level of Eq. 3.2.10 and 3.2.11. However, in this section we will introduce the various factors that contribute to the scale of the nonlinear susceptibility, as well as introduce the method of Midwinter and Warner for reducing the tensor character of uniaxial birefringent materials to an effective nonlinearity.

There are four key factors that contribute to the form of the susceptibility. The first two features we will discuss can be ascertained from Eq. 3.2.4 and 3.2.5. First, Eq. 3.2.5 illustrates that the nonlinear susceptibility scales with the number density of the oscillator; this property has a profound impact on the

contrasting behavior of (self-)heterodyned techniques - in which the signal is proportional to  $\chi^{(n)}$  - and homodyne-detected methods - in which the signal is proportional to  $|\chi^{(n)}|^2$ : in the former, the signal scales as N, while the latter scales as  $N^2$ . This difference will give differences in the detection level and sensitivity of heterodyne- and homodyne-detected techniques.[99, 190] Second, 3.2.4 and 3.2.5 illustrate that the nonlinear susceptibility provides insight into quantum mechanical properties of the oscillator; it is possible to show that Eq. 3.2.4 and 3.2.5 predict the nonlinear susceptibility to provide insight into the dipole strength, central frequency, and dephasing rate of the transitions.[99, 102, 109]

The third contribution to the susceptibilities is the local field enhancement factor. While we will generally know the macroscopic field,  $\tilde{\mathbf{E}}(t)$ , being applied, the interaction between the field and the material will be determined by the local field,  $\tilde{\mathbf{E}}_{\text{loc}}(t)$ , which will be influenced by induced polarizations. In this context, if we denote the molecular polarizability,  $\alpha$ , we can describe the linear polarization in terms of either the macroscopic field (Eq. 3.2.27a) or the local field (Eq. 3.2.27b):<sup>i</sup>

$$\tilde{\mathbf{P}}(t) = \epsilon_0 \chi^{(1)} \tilde{\mathbf{E}}(t) \tag{3.2.27a}$$

$$\tilde{\mathbf{P}}(t) = N\alpha \tilde{\mathbf{E}}_{\text{loc}}(t) \tag{3.2.27b}$$

In the mks system of units, the effect of the polarization on the local electric field in an isotropic medium is described by the Lorentz model:[107]

$$\tilde{\mathbf{E}}_{\text{loc}}(t) = \tilde{\mathbf{E}}(t) + \frac{1}{3\epsilon_0} \tilde{\mathbf{P}}(t)$$
(3.2.28)

Note that this contrasts with the Lorentz model for quantities in cgs or esu units: [102, 107, 109]

$$\tilde{\mathbf{E}}_{\text{loc}}(t) = \tilde{\mathbf{E}}(t) + \frac{4\pi}{3}\tilde{\mathbf{P}}(t)$$
(3.2.29)

By requiring the descriptions of the polarization in Eq. 3.2.27a and 3.2.27b to be equivalent, we can introduce the Lorentz model to yield:

$$\epsilon_0 \chi^{(1)} \tilde{\mathbf{E}}(t) = N \alpha \tilde{\mathbf{E}}_{\text{loc}}(t)$$

$$= N \alpha \left( \tilde{\mathbf{E}}(t) + \frac{1}{3\epsilon_0} \epsilon_0 \chi^{(1)} \tilde{\mathbf{E}}(t) \right)$$
(3.2.30)

<sup>&</sup>lt;sup>i</sup>Note that the form of Eq. 3.2.27a implies that the forms of these equations are appropriate for a system in which the quantities are reported in mks units. The following derivation of the local field effect can be reproduced with the appropriate modifications to describe quantities in cgs or esu units. However, as this derivation is presented by authors such as Boyd, Kittel, and Shen,[102, 107, 109] we will present the derivation of the appropriate correction for a system described in mks units.

Combining terms associated with  $\chi^{(1)}$  allows us to define the susceptibility:

$$\chi^{(1)} = \frac{N\alpha}{\epsilon_0 - N\alpha/3} \tag{3.2.31}$$

To proceed, we can invoke the definition of the dielectric constant,  $\epsilon = \epsilon_0(1 + \chi^{(1)}), [101]$  allowing substitution for  $\chi^{(1)}$ :

$$\epsilon_r - 1 = \frac{N\alpha}{\epsilon_0 - N\alpha/3} \tag{3.2.32}$$

where  $\epsilon_r$  is the relative permittivity,  $\epsilon_r = \epsilon/\epsilon_0$ .[101] If we manipulate Eq. 3.2.32 to isolate a quantity proportional to  $N\alpha$ , we arrive at the Clausius-Mossotti relation:[107]

$$\frac{\epsilon_r - 1}{\epsilon_r + 2} = \frac{N\alpha}{3\epsilon_0} \tag{3.2.33}$$

which has the analogous result for cgs/esu units:[102, 107]

$$\frac{\epsilon - 1}{\epsilon + 2} = \frac{4\pi N\alpha}{3} \tag{3.2.34}$$

with  $\epsilon$  defined in the cgs or esu unit systems as  $\epsilon = 1 + 4\pi\chi^{(1)}$ . From Eq. 3.2.33, we may multiply each side of the equation by -1, then add 1; the result of these operations can then be shown to yield the relationship:

$$\frac{\epsilon_r + 2}{3\epsilon_0} = \frac{1}{\epsilon_0 - \frac{N\alpha}{3}} \tag{3.2.35}$$

Recognizing that Eq. 3.2.31 relates the susceptibility to the product of the quantities  $N\alpha$  and  $1/(\epsilon_0 - N\alpha/3)$ , we can substitute according to Eq. 3.2.35 to define the susceptibility:

$$\chi^{(1)} = \frac{\epsilon_r + 2}{3\epsilon_0} N\alpha \tag{3.2.36}$$

where the term  $(\epsilon_r+2)/3\epsilon_0$  is the local field correction factor. This contrasts with the correction factor appropriate for cgs or esu units:  $(\epsilon+2)/3$ .[102, 109] However, in most materials, we will be able to relate  $\epsilon_r$  (mks units) or  $\epsilon$  (cgs/esu units) to the complex index of refraction,  $\bar{n}$ , according to the relationships:

$$\epsilon_r = \bar{n}^2 \tag{3.2.37a}$$

$$\epsilon = \bar{n}^2 \tag{3.2.37b}$$

The relationships of Eq. 3.2.37a and 3.2.37b allow us to substitute for the complex index of refraction in the local field correction factor, yielding factors of  $(\bar{n}^2+2)/3\epsilon_0$  and  $(\bar{n}^2+2)/3$  for mks and cgs/esu units, respectively.[102, 107]

This analysis can be extended to higher order susceptibilities.[102, 109] However, when this treatment is extended to consider the form of  $\chi^{(n)}$ , the result exhibits (n + 1) enhancement terms. In other words, the local field correction factor will assume the form  $((\bar{n}^2+2)/3\epsilon_0)^{n+1}$  (mks units) or  $((\bar{n}^2+2)/3)^{n+1}$  (cgs/esu units). This form arises due to the need to consider the enhancement arising at the frequency of the polarization (1 term) and the enhancement arising at each of the fields contributing to the polarization (*n* terms).[99, 102, 109]

Finally, the form of the nonlinear susceptibility can provide information regarding the symmetry of the medium. This behavior is particular pronounced in uniaxial birefringent media, for which it is possible to derive closed form solutions predicting the symmetry of the effective nonlinear susceptibility for experiments involving different combinations for the polarization states of the incident fields. The observed symmetry can have value in characterizing the structure of birefringent materials[191–193] and is an important consideration in determining the appropriate alignment[194] and efficiency of nonlinear crystals. The symmetry is defined according to the angles formed between the optical axes ( $\mathbf{X}$ ,  $\mathbf{Y}$ , and  $\mathbf{Z}$ ) and the wave vectors ( $\mathbf{k}$ ) of the electric fields.[106, 119, 195] We recall the definitions of the relevant angles ( $\theta$  and  $\phi$ ) from § 3.2.1:  $\theta$  reports the angle between  $\mathbf{k}$  and  $\mathbf{Z}$ , while  $\phi$  characterizes the angle between  $\mathbf{X}$  and the projection of  $\mathbf{k}$  on the  $\mathbf{XY}$  plane.[106] From these angles, the projections of the field onto each axis,  $E_{\mathbf{X}}$ ,  $E_{\mathbf{Y}}$ , and  $E_{\mathbf{Z}}$ , can be defined for ordinary beams:[119]

$$\begin{bmatrix} E_{\mathbf{X}}^{o} \\ E_{\mathbf{Y}}^{o} \\ E_{\mathbf{Z}}^{o} \end{bmatrix} = \begin{bmatrix} \sin(\phi) \\ -\cos(\phi) \\ 0 \end{bmatrix} E^{o}$$
(3.2.38)

where  $E^{o}$  is the amplitude of the electric field for the ordinary beam, and for extraordinary beams:[119]

$$\begin{bmatrix} E_{\mathbf{X}}^{e} \\ E_{\mathbf{Y}}^{e} \\ E_{\mathbf{Z}}^{e} \end{bmatrix} = \begin{bmatrix} -\cos(\theta)\cos(\phi) \\ -\cos(\theta)\sin(\phi) \\ \sin(\theta) \end{bmatrix} E^{e}$$
(3.2.39)

where  $E^e$  is the amplitude of the electric field for extraordinary beam.<sup>i</sup> While the symmetry of the tensor  $\chi^{(2)}$  is rigorously described by three polarization indices,  $\chi^{(2)}_{ijk}$ ,[102, 106] under conditions in which Kleinman

<sup>&</sup>lt;sup>i</sup>We note here that there appears to be an inconsistency in Dmitriev et al. For the effective electric field vectors, Dmitriev

symmetry is valid, the three dimensional representation of  $\chi_{ijk}^{(2)}$  or  $d_{ijk} = 1/2\chi_{ijk}^{(2)}$  can be reduced to a planar representation. This notation reduces  $d_{ijk}$  to  $d_{il}$ , where l = 1 represents jk = 11 (or  $jk = \mathbf{XX}$ ), l = 2corresponds to jk = 22 (or  $jk = \mathbf{YY}$ ), l = 3 arises from jk = 33 (or  $jk = \mathbf{ZZ}$ ), l = 4 replaces jk = 23and jk = 32 (or  $jk = \mathbf{YZ}$  and  $jk = \mathbf{ZY}$ ), l = 5 is a substitute for jk = 31 or jk = 13 (or  $jk = \mathbf{XZ}$  and  $jk = \mathbf{ZX}$ ), and l = 6 derives from jk = 12 or jk = 21 (or  $jk = \mathbf{YX}$  and  $jk = \mathbf{XY}$ ).[102, 106] In addition to these substitutions, we also have to redefine  $P_i = 2d_{ijk}E_jE_k$ ; in the planar representation, the analogous form is  $P_i = 2d_{il}E_l^2$ .[102, 119] Alternatively, we can consider that we are sampling a particular polarization state, in which case the effective polarization is  $P = P_i \cdot (2d_{il}E_l^2)$ . Making substitutions in  $d_{il}$  appropriate to Kleinman symmetry, this yields:

$$P = 2 \begin{bmatrix} P_{\mathbf{X}} \\ P_{\mathbf{Y}} \\ P_{\mathbf{Z}} \end{bmatrix} \cdot \begin{bmatrix} d_{11} & d_{12} & d_{13} & d_{14} & d_{15} & d_{16} \\ d_{16} & d_{22} & d_{23} & d_{24} & d_{14} & d_{12} \\ d_{15} & d_{24} & d_{33} & d_{23} & d_{13} & d_{14} \end{bmatrix} \begin{bmatrix} E_{\mathbf{X}}^{2} \\ E_{\mathbf{Z}}^{2} \\ E_{\mathbf{Y}}E_{\mathbf{Z}} + E_{\mathbf{Z}}E_{\mathbf{Y}} \\ E_{\mathbf{X}}E_{\mathbf{Z}} + E_{\mathbf{Z}}E_{\mathbf{X}} \\ E_{\mathbf{X}}E_{\mathbf{Z}} + E_{\mathbf{Z}}E_{\mathbf{X}} \\ E_{\mathbf{X}}E_{\mathbf{X}} + E_{\mathbf{Y}}E_{\mathbf{X}} \end{bmatrix}$$
(3.2.40)

Population of  $d_{il}$  can be achieved according to the the crystal class of the material.

To demonstrate how this might be applied, we will consider the case of BBO. As BBO belongs to the 3m point group,[106] it exhibits nonzero tensor elements  $\mathbf{XZX} = \mathbf{YZY}$ ,  $\mathbf{XXZ} = \mathbf{YYZ}$ ,  $\mathbf{ZXX} = \mathbf{ZYY}$ ,  $\mathbf{ZZZ}$ ,  $\mathbf{YYY} = -\mathbf{YXX} = -\mathbf{XXY} = -\mathbf{XYX}$ .[102] In terms of  $d_{il}$ , this corresponds to 15 = 24 ( $\mathbf{XZX} = \mathbf{YZY}$  and  $\mathbf{XXZ} = \mathbf{YYZ}$ ), 31 = 32 ( $\mathbf{ZXX} = \mathbf{ZYY}$ ), 33, 22 = -21 = -16 ( $\mathbf{YYY} = -\mathbf{YXX} = -\mathbf{XXY} = -\mathbf{XXY} = -\mathbf{XYX}$ ).[106] Substituting according to these relationships and considering the two major processes for

et al use the phase: [106]

$$\begin{bmatrix} E_{\mathbf{X}}^{o} \\ E_{\mathbf{Y}}^{o} \\ E_{\mathbf{Z}}^{o} \end{bmatrix} = \begin{bmatrix} -\sin(\phi) \\ \cos(\phi) \\ 0 \end{bmatrix} E^{o}$$

for the ordinary beam, and for the extraordinary beam:[106]

$$\begin{bmatrix} E_{\mathbf{x}}^{\mathbf{x}} \\ E_{\mathbf{y}}^{\mathbf{x}} \\ E_{\mathbf{z}}^{e} \end{bmatrix} = \begin{bmatrix} \cos(\theta)\cos(\phi) \\ \cos(\theta)\sin(\phi) \\ -\sin(\theta) \end{bmatrix} E^{e}$$

Note that these phases are opposite those employed by Midwinter and Warner.[119] However, the formulae for the effective nonlinearities of BBO reported by Dmitriev et al are consistent with the notation of Midwinter and Warner, as well as Boyd.[102, 106, 119] At the intensity level, both results will be identical when only the nonlinearity of BBO is relevant - and it should be noted that Dmitriev et al acknowledge this by reporting only the relative signs of  $d_{22}$  and  $d_{31}[106]$  - but this inconsistency bears mentioning. As it appears to yield results consistent with all other authors,[102, 106, 119] we will adopt the phase reported by Midwinter and Warner.[119]

BBO (Type I:  $e \leftarrow oo$ ; Type II:  $e \leftarrow oe$ ), we can evaluate Eq. 3.2.40 for Type I phase matching:

$$P = 2 \begin{bmatrix} -\cos(\theta)\cos(\phi) \\ -\cos(\theta)\sin(\phi) \\ \sin(\theta) \end{bmatrix} \cdot \begin{bmatrix} 0 & 0 & 0 & 0 & d_{15} & -d_{22} \\ -d_{22} & d_{22} & 0 & d_{15} & 0 & 0 \\ d_{15} & d_{15} & d_{33} & 0 & 0 & 0 \end{bmatrix} \begin{bmatrix} \sin^2(\phi) \\ \cos^2(\phi) \\ 0 \\ 0 \\ 0 \\ -2\cos(\phi) \\ \sin(\phi) \end{bmatrix} E_{sig}^o E_{idl}^o$$

$$= 2 \begin{bmatrix} -\cos(\theta)\cos(\phi) \\ -\cos(\theta)\sin(\phi) \\ \sin(\theta) \end{bmatrix} \cdot \begin{bmatrix} 2d_{22}\sin(\phi)\cos(\phi) \\ -d_{22}\sin^2(\phi) + d_{22}\cos^2(\phi) \\ d_{15} \end{bmatrix} E_{sig}^o E_{idl}^o$$

$$= 2 [-d_{22}\cos(\theta)\sin(3\phi) + d_{15}\sin(\theta)] E_{sig}^o E_{idl}^o$$

$$= 2 [-d_{22}\cos(\theta)\sin(3\phi) + d_{15}\sin(\theta)] E_{sig}^o E_{idl}^o$$

and for Type II phase matching:

$$P = 2 \begin{bmatrix} -\cos(\theta)\cos(\phi) \\ -\cos(\theta)\sin(\phi) \\ \sin(\theta) \end{bmatrix}$$

$$\cdot \begin{bmatrix} 0 & 0 & 0 & d_{15} & -d_{22} \\ -d_{22} & d_{22} & 0 & d_{15} & 0 & 0 \\ d_{15} & d_{15} & d_{33} & 0 & 0 & 0 \end{bmatrix} \begin{bmatrix} -\cos(\theta)\cos(\phi)\sin(\phi) \\ \cos(\theta)\cos(\phi)\sin(\phi) \\ 0 \\ -\sin(\theta)\cos(\phi) \\ \sin(\theta)\sin(\phi) \\ \cos(\theta)\cos(2\phi) \end{bmatrix}$$

$$= 2 \begin{bmatrix} -\cos(\theta)\cos(\phi) \\ -\cos(\theta)\sin(\phi) \\ \sin(\theta) \end{bmatrix} \cdot \begin{bmatrix} d_{15}\sin(\theta)\sin(\phi) - d_{22}\cos(\theta)\cos(2\phi) \\ 2d_{22}\cos(\theta)\cos(\phi)\sin(\phi) - d_{15}\sin(\theta)\cos(\phi) \\ 0 \end{bmatrix} E_{sig}^{o}E_{idl}^{o}$$

$$= 2 \begin{bmatrix} -\cos(\theta)\cos(\phi) \\ -\cos(\theta)\sin(\phi) \\ \sin(\theta) \end{bmatrix} \cdot \begin{bmatrix} d_{15}\sin(\theta)\sin(\phi) - d_{22}\cos(\theta)\cos(2\phi) \\ 2d_{22}\cos(\theta)\cos(\phi)\sin(\phi) - d_{15}\sin(\theta)\cos(\phi) \\ 0 \end{bmatrix} E_{sig}^{o}E_{idl}^{o}$$

$$= 2 [d_{22}\cos^{2}(\theta)\cos(3\phi)]E_{sig}^{o}E_{idl}^{o}$$

Considering the effective polarization to be of the form  $P = 2d_{\text{eff}}E^2$ , we can use the results of the schemes in Eq. 3.2.41 and 3.2.42 to define  $d_{\text{eff}} = d_{15}\sin(\theta) - d_{22}\cos(\theta)\sin(3\phi)$  for Type I phase matching and  $d_{\text{eff}} = d_{22}\cos^2(\theta)\cos(3\phi)$  for Type II phase matching. This latter result is consistent with the published formulae for  $d_{\text{eff}}$  for Type II phase matching;[102, 106, 119] for Type I phase matching, we note that Kleinman symmetry causes  $d_{15} = d_{31}$ ,[195] permitting substitution into our result in Eq. 3.2.41 to yield the more standard form,  $d_{\text{eff}} = d_{31}\sin(\theta) - d_{22}\cos(\theta)\sin(3\phi)$ .[102, 106, 119]

### 3.3 Maxwell's Equations and the Wave Equation

In § 3.1.1, we claimed that Maxwell's equations predict that the free space propagation of electric and magnetic fields are consistent with a wave equation of the form Eq. 3.1.1. We also introduced the form of Maxwell's equations as they apply to free space in order to explore the expected relationship between the electric and magnetic fields in light. In the following sections, we will outline the derivation of the wave equation from Maxwell's equations with the intention of arriving at a generalized result that can be applied to systems described in either cgs/esu or mks units. Throughout this section, we will continue to focus exclusively on the form of these equations as they relate to the electric field; a similar derivation can be performed to describe the magnetic field.[101] Throughout this section, we will end up alternating between noting quantities as vector and scalar values, but will explain at the appropriate point in § 3.3.3 why this is necessary. Various presentations of the wave equation, the final result of the derivation in § 3.3.3, are presented in § 3.3.1 for convenience.

# 3.3.1 Wave Equation

As we will show in the succeeding sections, Maxwell's equations show that the propagation of an electric field,  $\tilde{E}(\mathbf{r}, z, t)$ , under the influence of an induced polarization,  $\tilde{P}(\mathbf{r}, z, t)$ , in a medium is predicted:

$$\nabla^{2}\tilde{E}(\mathbf{r},z,t) - \epsilon_{0}\mu \frac{\partial^{2}}{\partial t^{2}}\tilde{E}(\mathbf{r},z,t) = \mu \frac{\partial^{2}}{\partial t^{2}}\tilde{P}(\mathbf{r},z,t)$$
(3.3.1)

when  $\tilde{E}(\mathbf{r}, z, t)$  and  $\tilde{P}(\mathbf{r}, z, t)$  are reported in mks units, with  $\epsilon_0 = 8.85 \times 10^{-12} F/m$  describing the permittivity of free space and  $\mu$  describing the permeability of the material. When  $\tilde{E}(\mathbf{r}, z, t)$  and  $\tilde{P}(\mathbf{r}, z, t)$  are represented by cgs units, the wave equation assumes the form:

$$\nabla^2 \tilde{E}(\mathbf{r}, z, t) - \frac{1}{c^2} \frac{\partial^2}{\partial t^2} \tilde{E}(\mathbf{r}, z, t) = \frac{4\pi}{c^2} \frac{\partial^2}{\partial t^2} \tilde{P}(\mathbf{r}, z, t)$$
(3.3.2)

where c is the speed of light. We will also adopt the generalization:

$$\nabla^{2}\tilde{E}(\mathbf{r},z,t) - b^{2}de\frac{\partial^{2}}{\partial t^{2}}\tilde{E}(\mathbf{r},z,t) = ab^{2}e\frac{\partial^{2}}{\partial t^{2}}\tilde{P}(\mathbf{r},z,t)$$
(3.3.3)

where we use a, b, d, and e to stand in for the various constants. In the generalized form of Eq. 3.3.3, the values of a and b are a = b = 1 for the mks unit system or  $a = 4\pi$  and  $b = c^{-1}$  for cgs and esu units. Meanwhile, d and e are used to represent the permittivity and permeability: in cgs or esu units, d = e = 1, while mks units require  $d = \epsilon_0$  and  $e = \mu$ .

In a similar manner to § 3.1, we will generally define a coordinate system such that the field is taken to propagate along the z direction. Within such a coordinate system, we can consider that  $\nabla^2$ , defined according to Eq. 3.3.4:

$$\nabla^2 = \frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + \frac{\partial^2}{\partial z^2}$$
(3.3.4)

has two key contributions: a term corresponding to the direction of propagation  $(\partial^2/\partial z^2)$ , and two terms corresponding to evolution of the field perpendicular to the direction of propagation (the transverse components,  $\partial^2/\partial x^2$  and  $\partial^2/\partial y^2$ ). As we introduced in § 3.1.2, we will generally not be concerned with differentiating x from y, and have accordingly been using the generalized transverse coordinate,  $\mathbf{r}$ . In a similar manner, we can introduce the operator  $\nabla_{\mathrm{T}}^2$  to generalize the transverse components of  $\nabla^2$ , defining this operator:

$$\nabla_{\rm T}^2 = \frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} \tag{3.3.5}$$

With this operator, we can simplify Eq. 3.3.3 by splitting  $\nabla^2$  accordingly:

$$\nabla_{\mathrm{T}}^{2}\tilde{E}(\mathbf{r},z,t) + \frac{\partial^{2}}{\partial z^{2}}\tilde{E}(\mathbf{r},z,t) - b^{2}de\frac{\partial^{2}}{\partial t^{2}}\tilde{E}(\mathbf{r},z,t) = ab^{2}e\frac{\partial^{2}}{\partial t^{2}}\tilde{P}(\mathbf{r},z,t)$$
(3.3.6)

However, in the limit that the field can be considered to be a plane wave, we do not anticipate the field exhibiting any dependence upon the transverse coordinates; at this limit, we can assume  $\nabla_{\rm T}^2 \tilde{E}(z,t) = 0$ . This allows us to simplify Eq. 3.3.6 further, yielding:

$$\frac{\partial^2}{\partial z^2}\tilde{E}(z,t) - b^2 de \frac{\partial^2}{\partial t^2}\tilde{E}(z,t) = ab^2 e \frac{\partial^2}{\partial t^2}\tilde{P}(z,t)$$
(3.3.7)

#### 3.3.2 Maxwell's Equations

Here we will briefly list Maxwell's equations and discuss appropriate assumptions that might be made for common material systems. The dependence of the wave equation on the different unit systems indicated in § 3.3.1 stems from the forms of Maxwell's equations in each unit system. However, the variables a and b allow these equations to be generalized, yielding the forms in Eq. 3.3.8a - 3.3.8d:

$$\nabla \cdot \tilde{\mathbf{D}}(t) = a\rho \tag{3.3.8a}$$

$$\nabla \cdot \tilde{\mathbf{B}}(t) = 0 \tag{3.3.8b}$$

$$\nabla \times \tilde{\mathbf{E}}(t) = -b \frac{\partial}{\partial t} \tilde{\mathbf{B}}(t)$$
(3.3.8c)

$$\nabla \times \tilde{\mathbf{H}}(t) = ab\tilde{\mathbf{J}}(t) + b\frac{\partial}{\partial t}\tilde{\mathbf{D}}(t)$$
(3.3.8d)

As before, we will allow  $a = 4\pi$  and  $b = c^{-1}$  for cgs or esu unit systems, while we will require a = b = 1 for the mks unit system.

Gauss' law (Eq. 3.3.8a) relates the changes in the displacement,  $\mathbf{\tilde{D}}(t)$ , to the free charges,  $\rho$ , in the medium. Generally, we will be able to assume that the medium contains no free charges ( $\rho = 0$ ), under which conditions Gauss' law reduces to  $\nabla \cdot \mathbf{\tilde{D}}(t) = 0$ . We have previously introduced the displacement (see § 3.2), through which we relate the dependence of the propagating field upon polarizations existing in the medium (see Eq. 3.2.7 and 3.2.8). However, the forms of Eq. 3.2.7 and 3.2.8 illustrate that we will also need to consider a generalized form for this quantity. This is achieved through our previous assignment of a and d,  $a = 4\pi$  and d = 1 for cgs and esu units, or a = 1 and  $d = \epsilon_0$  for mks units. The introduction of these generalized constants yields:

$$\mathbf{D}(t) = d\mathbf{E}(t) + a\mathbf{P}(t) \tag{3.3.9}$$

While Gauss' law of magnetism (Eq. 3.3.8b) and Faraday's law (Eq. 3.3.8c) do not require further simplification, we will be able to make some modifications to Ampere's law (Eq. 3.3.8d). First, it will generally be reasonable to assume the absence of free currents ( $\tilde{\mathbf{J}}(t) = 0$ ), leaving:

$$\nabla \times \tilde{\mathbf{H}}(t) = b \frac{\partial}{\partial t} \tilde{\mathbf{D}}(t)$$
(3.3.10)

Second, we will often be able to assume that the medium is nonmagnetic or nearly so  $(\tilde{\mathbf{M}}(t) \approx 0)$ , in which case the magnetic field,  $\tilde{\mathbf{B}}(t)$ , is simply:  $\tilde{\mathbf{B}}(t) = e(\tilde{\mathbf{H}}(t) + \tilde{\mathbf{M}}(t)) \approx e\tilde{\mathbf{H}}(t)$ , where e = 1 for cgs or esu units, but  $e = \mu$  for mks units. This then simplifies Ampere's law to yield:

$$\nabla \times \tilde{\mathbf{B}}(t) = be \frac{\partial}{\partial t} \tilde{\mathbf{D}}(t)$$
(3.3.11)

We will also note that at optical frequencies, we will generally be able to neglect the magnetic response of the material, permitting the assumption that  $\mu \approx \mu_0$  when we are concerned with mks units.

# 3.3.3 Derivation of the Generalized Wave Equation

An outline of the derivation of Eq. 3.3.3 is given by Boyd[102] for cgs/esu units. Powers [196] demonstrates how the wave equation would be derived for mks units, but restricts the derivation to the case of the linear optical response. These derivations are generalized and reproduced below.

We have previously outlined (§ 3.3.2) a handful of approximations and simplifications that we can make to Maxwell's equations, Eq. 3.3.8a - 3.3.8d. We start the derivation with Faraday's law:

$$\nabla \times \tilde{\mathbf{E}}(t) = -b \frac{\partial}{\partial t} \tilde{\mathbf{B}}(t)$$
(3.3.12)

and proceed to take the curl  $(\nabla \times)$  of both sides. This results in:

$$\nabla \times \nabla \times \tilde{\mathbf{E}}(t) = -b\nabla \times \frac{\partial}{\partial t}\tilde{\mathbf{B}}(t)$$
(3.3.13)

However, since the curl only applies to the spatial variables, while the derivative in Eq. 3.3.13 is with respect to time, we can reverse the order in which these operations are applied:

$$\nabla \times \nabla \times \tilde{\mathbf{E}}(t) = -b \frac{\partial}{\partial t} \nabla \times \tilde{\mathbf{B}}(t)$$
(3.3.14)

However, from Eq. 3.3.11, we know  $\nabla \times \tilde{\mathbf{B}}(t) = be\partial/\partial t \tilde{\mathbf{D}}(t)$ ; therefore, we can substitute for  $\nabla \times \tilde{\mathbf{B}}(t)$  in Eq. 3.3.14, yielding:

$$\nabla \times \nabla \times \tilde{\mathbf{E}}(t) = -b\frac{\partial}{\partial t} \left( be\frac{\partial}{\partial t} \tilde{\mathbf{D}}(t) \right)$$
  
=  $-b^2 e \frac{\partial^2}{\partial t^2} \tilde{\mathbf{D}}(t)$  (3.3.15)

At this point, it is common to use the identity:

$$\nabla \times \nabla \times \tilde{\mathbf{E}}(t) = \nabla [\nabla \cdot \tilde{\mathbf{E}}(t)] - \nabla^2 \tilde{\mathbf{E}}(t)$$
(3.3.16)

The advantage gained from this identity is that we have previously indicated that Gauss' law can generally be taken to be of the form:  $\nabla \cdot \tilde{\mathbf{D}}(t) = 0$ . Using the fact that  $\tilde{\mathbf{D}}(t)$  is of the form in Eq. 3.3.9, we can distribute the dot product to yield:[197]

$$\nabla \cdot \tilde{\mathbf{D}}(t) = d\nabla \cdot \tilde{\mathbf{E}}(t) + a\nabla \cdot \tilde{\mathbf{P}}(t) = 0$$
(3.3.17)

Since the polarization is a function of the electric field, it is often assumed that the relationship in Eq. 3.3.17 implies that  $\nabla \cdot \tilde{\mathbf{E}}(t) = 0.[102]$  This then allows us to reduce  $\nabla \times \nabla \times \tilde{\mathbf{E}}(t)$  to the second term on the right hand side of Eq. 3.3.16,  $-\nabla^2 \tilde{\mathbf{E}}(t)$ , yielding:

$$\nabla^2 \tilde{\mathbf{E}}(t) = b^2 e \frac{\partial^2}{\partial t^2} \tilde{\mathbf{D}}(t)$$
(3.3.18)

Splitting the displacement according to Eq. 3.3.9 and combining the terms associated with the electric field then yields the wave equation:

$$\nabla^2 \tilde{\mathbf{E}}(t) - b^2 de \frac{\partial^2}{\partial t^2} \tilde{\mathbf{E}}(t) = ab^2 e \frac{\partial^2}{\partial t^2} \tilde{\mathbf{P}}(t)$$
(3.3.19)

From Eq. 3.3.19, it is trivial to verify our claim that light behaves as a wave in free space: under such conditions,  $\tilde{\mathbf{P}}(t) = 0$ , leaving only the left hand side. Since the Laplacian operator contains the propagation and transverse evolution, we have a form of the wave equation in Eq. 3.3.19, with  $b^2 de$  equal to  $1/v^2$ . Using our definitions for these generalized variables (b = 1,  $d = \epsilon_0$ , and  $e = \mu \approx \mu_0$ , or  $b = c^{-1}$ , d = 1, and e = 1), we can then show that the speed of the wave predicted by Eq. 3.3.19 is c, the speed of light, in both systems of units.

Lax et al,[108] however, indicate that substitution according to the identity in Eq. 3.3.16 is not necessarily rigorous if we are are later splitting the Laplacian operator in the manner of Eq. 3.3.6. Rather, they argue that it is more accurate to consider the separation of the longitudinal and transverse components in Eq. 3.3.15, taking the curl to be  $(\partial/\partial z + \nabla_T) \times$ . They justify this treatment by showing that the form of the wave equation resulting from applying the identity of Eq. 3.3.16 prior to separating the transverse and longitudinal components results in an inconsistency in the behavior of polarized fields. In their treatment - in which they apply the curl as  $(\partial/\partial z + \nabla_T) \times$  - it is relatively straightforward, but tedious, to show that Maxwell's equations predict a pair of coupled differential equations, one describing transverse propagation and the other describing longitudinal propagation, of the form:<sup>i</sup>

$$\nabla_{\mathrm{T}} \left[ \nabla_{\mathrm{T}} \cdot \tilde{E}_{\mathrm{T}}(t) + \frac{\partial}{\partial z} \tilde{E}_{z}(t) - ik_{0}\tilde{E}_{z}(t) \right] - \left[ \nabla_{\mathrm{T}}^{2} + \frac{\partial^{2}}{\partial z^{2}} - i2k_{0}\frac{\partial}{\partial z} - k_{0}^{2} \right] \tilde{E}_{\mathrm{T}}(t) = -\frac{1}{c^{2}} \frac{\partial^{2}}{\partial t^{2}} \tilde{D}_{\mathrm{T}}^{(1)}(t) \quad (3.3.20a)$$
$$\frac{\partial}{\partial z} \left[ \nabla_{\mathrm{T}} \cdot \tilde{E}_{\mathrm{T}}(t) \right] - ik_{0}\nabla_{\mathrm{T}} \cdot \tilde{E}_{\mathrm{T}}(t) - \nabla_{\mathrm{T}}^{2}\tilde{E}_{z}(t) = -\frac{1}{c^{2}} \frac{\partial^{2}}{\partial t^{2}} \tilde{D}_{z}^{(1)}(t) \quad (3.3.20b)$$

<sup>&</sup>lt;sup>i</sup>It should be noted that Lax et al use the convention that the electric field accumulates phase according to  $\exp(ik_0z)$ , rather than  $\exp(-ik_0z)$ . The forms of Eq. 3.3.20a and 3.3.20b are derived in a manner appropriate to our convention; these only differ from the forms presented by Lax et al by the sign of the imaginary terms (substituting  $\pm ik_0$  by  $\mp ik_0$ ).[108]

Unfortunately, these coupled differential equations prove difficult to implement, with Lax et al needing approximations in order to describe the propagation of the  $\text{TEM}_{00}$  mode. Additionally, their result does not exhibit significant deviation from the model predicted by Eq. 3.3.6. Nevertheless, some authors[104] acknowledge the simplification introduced within the derivation shown above and consider the resulting wave equation to more accurately serve as a description for the scalar representation of the wave:

$$\nabla^2 \tilde{E}(t) - b^2 de \frac{\partial^2}{\partial t^2} \tilde{E}(t) = ab^2 e \frac{\partial^2}{\partial t^2} \tilde{P}(t)$$
(3.3.21)

We will adopt this convention proceeding forward.

#### 3.4 Pulse Propagation in Dispersive Media

In this section, we will apply the wave equation (§ 3.3) to describe the effects of propagation on ultrafast pulses. This section will be divided into four subsections. In the first, we will derive the general treatment of ultrafast pulse propagation, concluding in the second section in which we consider the ramifications of the slowly varying amplitude approximation. The third and fourth sections will consider applications of the general treatment towards two specific scenarios: a description of  $TEM_{00}$  mode propagation, and the integration of an arbitrary polarization to predict the emitted field.

#### 3.4.1 General Treatment of Pulse Propagation in Dispersive Media

While Akhmanov et al[103] imply a strategy for deriving the results of the following treatment in the time domain, we will follow the lead of Boyd [102], who appears influenced by Brabec and Krausz,[115] with a derivation that proceeds mostly in the frequency domain. This approach relies upon (inverse) Fourier transformation between the time and frequency domains at multiple points during the derivation; this is done to make the application of different effects mathematically more tractable. While we will generally not explicitly show the introduction of the exponential and integration associated with these transforms, we will note when the introduction of the appropriate quantities has ramifications on the result. We will continue to follow the conventions we introduced previously: normalization required for transformation into and from the angular frequency domain will be incorporated during inverse Fourier transformation, and we will apply the exponentials  $\exp(-i\omega t)$  and  $\exp(i\omega t)$  for Fourier transformation and inverse Fourier transformation, respectively. The former convention is consistent with Boyd and others,[102, 112–114] while the latter is not consistent with Boyd,[102] but has precedence from other authors.[112, 114]. These conventions will result in a deviation from the derivation shown here and the derivation presented by Boyd;[102] we will note the point at which the derivations begin deviating significantly. While we will work with general functions to note the electric field, polarization, and their respective envelope functions, we will assume these functions are compatible with Fourier transformation.[114] We will also assume the electric field and polarization exhibit forms compatible with Eq. 3.1.27 and 3.2.17 but will also assume it is sufficient to consider the propagation of only the complex field and the associated term of the polarization.

We will begin with the result of Eq. 3.3.7: while we will introduce the transverse propagation in later sections (see § 3.4.3 and § 3.4.4), it is sufficient to consider the following derivation within the plane wave limit. From Eq. 3.3.7, we will invoke the behavior of Eq. 3.2.9 and split the polarization on the right hand side into the linear and nonlinear contributions,  $\tilde{P}^{L}(z,t) = \tilde{P}^{(1)}(z,t)$  and  $\tilde{P}^{(NL)}(z,t)$ . Substituting the sum of these terms into Eq. 3.3.7 yields:

$$\frac{\partial^2}{\partial z^2}\tilde{E}(z,t) - b^2 de \frac{\partial^2}{\partial t^2}\tilde{E}(z,t) - ab^2 e \frac{\partial^2}{\partial t^2}\tilde{P}^{(1)}(z,t) = ab^2 e \frac{\partial^2}{\partial t^2}\tilde{P}^{\rm NL}(z,t)$$
(3.4.1)

In Eq. 3.4.1, we have also subtracted the term in  $\tilde{P}^{(1)}(z,t)$  from both sides; the advantage to doing so is that it allows us to introduce the linear displacement,  $\tilde{D}^{(1)}(z,t)$ . This quantity is defined in a manner analogous to Eq. 3.3.9, with the exception that the contributing polarization is restricted to the first-order term,  $\tilde{P}^{(1)}(z,t)$ , of the series. The generalized form of the linear displacement is then:

$$\tilde{D}^{(1)}(z,t) = d\tilde{E}(z,t) + a\tilde{P}^{(1)}(z,t)$$
(3.4.2)

The definition of  $\tilde{D}^{(1)}(z,t)$  in Eq. 3.4.2 allows us to combine the second and third terms on the left hand side of Eq. 3.4.1:

$$\frac{\partial^2}{\partial z^2}\tilde{E}(z,t) - b^2 e \frac{\partial^2}{\partial t^2} \tilde{D}^{(1)}(z,t) = ab^2 e \frac{\partial^2}{\partial t^2} \tilde{P}^{\rm NL}(z,t)$$
(3.4.3)

In order to understand the convenience of substituting  $\tilde{D}^{(1)}(z,t)$  into Eq. 3.4.3, it is necessary to introduce the definition of this quantity. The linear displacement relates the permittivity of the medium,  $\epsilon$ , to the electric field:[101–103]

$$\tilde{D}^{(1)}(z,t) = \int \epsilon(t')\tilde{E}(t-t',z)dt'$$
(3.4.4a)

$$D^{(1)}(z,\omega) = \epsilon(\omega)E(\omega)$$
(3.4.4b)

where  $D^{(1)}(z,\omega) = \mathcal{F}\{\tilde{D}^{(1)}(z,t)\}$ . Eq. 3.4.4a provides the time-domain description of the linear displacement, and Eq. 3.4.4b is the frequency-domain description. In Eq. 3.4.4a and 3.4.4b, the dependence of

$$\epsilon = \epsilon_0 (1 + \chi^{(1)}) \tag{3.4.5a}$$

$$\epsilon = 1 + 4\pi\chi^{(1)} \tag{3.4.5b}$$

In the mks unit system, an additional quantity - the relative permittivity,  $\epsilon_r$  - is also defined, and it is allowed to be the ratio of  $\epsilon$  and  $\epsilon_0$ ,  $\epsilon_r = \epsilon/\epsilon_0$ . The relative permittivity (in mks units) and the permittivity (in cgs/esu units) are significant, as both quantities are generally<sup>i</sup> equal to the square of the complex refractive index,  $\bar{n}^2$  (see Eq. 3.2.37a and 3.2.37b).[99, 101, 102, 107]

Based upon the forms of Eq. 3.4.4a and 3.4.4b, it will clearly be easier to apply the effects of linear displacement in the frequency domain. To achieve this, we can simply multiply both sides of Eq. 3.4.3 by the necessary exponential and integration factor, then integrate. We can determine the result of this transform by invoking the Fourier relations associated with the derivative:[112]

$$\mathcal{F}\{t^n f(t)\} = (-i)^n \frac{\partial^n}{\partial \omega^n} \hat{f}(\omega)$$
(3.4.6a)

$$\mathcal{F}\{\frac{\partial^n}{\partial t^n}f(t)\} = (i\omega)^n \hat{f}(\omega) \tag{3.4.6b}$$

where f(t) is an arbitrary function with  $\mathcal{F}{f(t)} = \hat{f}(\omega)$ . With these relationships, we can show the Fourier transform of Eq. 3.4.3 to be:

$$\frac{\partial^2}{\partial z^2} E(z,\omega) + b^2 e \omega^2 D^{(1)}(z,\omega) = -ab^2 e \omega^2 P^{\mathrm{NL}}(z,\omega)$$
(3.4.7)

where  $D^{(1)}(z,\omega)$  is defined as before, and  $E(z,\omega) = \mathcal{F}\{\tilde{E}(z,t)\}$  and  $P^{\mathrm{NL}}(z,\omega) = \mathcal{F}\{\tilde{P}^{\mathrm{NL}}(z,t)\}$ . Substituting for  $D^{(1)}(z,\omega)$  in Eq. 3.4.7 according to the frequency-domain definition in Eq. 3.4.4b yields Eq. 3.4.8:

$$\frac{\partial^2}{\partial z^2} E(z,\omega) + b^2 e \omega^2 \epsilon(\omega) E(z,\omega) = -ab^2 e \omega^2 P^{\rm NL}(z,\omega)$$
(3.4.8)

We can simplify Eq. 3.4.8 by spending some time considering the constants on both sides of the equation; in so doing, we can retain generalization, while reducing the number of terms to keep track of. Unfortunately,

<sup>&</sup>lt;sup>i</sup>The caveat is explicitly included here, as the formal definition of the complex refractive index is  $\bar{n}^2 = (\epsilon \mu)/(\epsilon_0 \mu_0)$ . However, as  $\mu \approx \mu_0$  in most materials, we can generally allow  $\bar{n}^2 \approx \epsilon_r$ .[101]

even though we arrive at the same result, we will need to consider the two unit systems (mks and cgs/esu) separately.

Starting with mks units, we recall that we have defined a = b = 1,  $d = \epsilon_0$ , and  $e = \mu$ . Starting with the coefficient of the second term on the left hand side of Eq. 3.4.8, we can evaluate  $b^2 e \epsilon(\omega)$  as follows:

$$b^{2}e\epsilon(\omega) = \mu(\omega)\epsilon(\omega)$$

$$= \mu_{r}(\omega)\epsilon_{r}(\omega)\mu_{0}\epsilon_{0}$$

$$= \frac{\mu_{r}(\omega)\epsilon_{r}(\omega)}{c^{2}}$$

$$= \frac{\left[n(\omega) - i\kappa(\omega)\right]^{2}}{c^{2}}$$
(3.4.9)

where we have used the relationship  $\epsilon_0\mu_0 = c^{-2}$  between the second and third lines and have directly introduced  $n(\omega) - i\kappa(\omega)^i$  in place of the complex index of refraction,  $\bar{n}$ .[99, 101, 102] Meanwhile, on the right hand side of Eq. 3.4.8, we have a coefficient of  $ab^2e = \mu(\omega) \approx \mu_0$ . However, we can proceed further by recognizing that, when defined in mks units (see Eq. 3.2.10 or 3.2.12a),  $P^{\rm NL}(z,\omega)$  will contain a factor of  $\epsilon_0$ . Therefore, if we define a new quantity,  $\bar{P}^{\rm NL}(z,\omega) = [P^{\rm NL}(z,\omega)]/\epsilon_0$ , we can redefine the right hand side of Eq. 3.4.8 as  $\mu_0\epsilon_0\omega^2\bar{P}^{\rm NL}(z,\omega)$ . With both  $\mu_0$  and  $\epsilon_0$ , we can use the identity above,  $\mu_0\epsilon_0 = c^{-2}$ , to yield a right hand side of  $\omega^2/c^2\bar{P}^{\rm NL}(z,\omega)$ .

On the other hand, when we are in the cgs or esu unit systems, we have defined  $a = 4\pi$ ,  $b = c^{-1}$ , and d = e = 1. With these values, we can show that the coefficients  $b^2 e\epsilon(\omega)$  of the second term on the left hand side can be simplified:

$$b^{2}e\epsilon(\omega) = \frac{\epsilon(\omega)}{c^{2}}$$

$$\approx \frac{\epsilon(\omega)\mu(\omega)}{c^{2}}$$

$$\approx \frac{[n(\omega) - i\kappa(\omega)]^{2}}{c^{2}}$$
(3.4.10)

where we have again directly introduced the quantity  $n - i\kappa$  to represent the complex index of refraction. We note that in the second line of Eq. 3.4.10, we have retained the approximate relation to emphasize that we assume  $\mu$  is negligible for most media. Meanwhile, for the right hand side of Eq. 3.4.8, we have  $ab^2e\omega^2 = (4\pi\omega^2)/c^2$ . We have seen that we retain a factor of  $\omega^2c^{-2}$  in mks units, but the factor of  $4\pi$  is unique

<sup>&</sup>lt;sup>i</sup>Note that this definition for the complex index of refraction differs from most sources.[99, 101, 107] However, since we have taken as our convention that the field accumulates phase as  $\exp(-ikz)$ , it is necessary for the index of refraction and extinction coefficient to have opposite sign (see Eq. 3.4.13). Though they appear to work strictly with the linear susceptibility, Akhmanov et al, also treating the phase accumulation of the complex field as  $\exp(-ikz)$ , take the imaginary component of the susceptibility to be of opposite sign as the real component.[198] Given  $\bar{n}^2 = [n \pm i\kappa]^2 = 1 + a\chi^{(1)}$  ( $a = 4\pi$  for cgs/esu units, and a = 1 for mks units),  $\Re\{a\chi^{(1)}\} = (n^2 - \kappa^2) - 1$  and  $\Im\{a\chi^{(1)}\} = \pm 2n\kappa$  (see Eq. 3.4.17). Since n and  $\kappa$  must both be real, the real and imaginary components of  $\chi^{(1)}$  will always have opposite signs only for  $\bar{n} = \pm n \mp i\kappa$ .

to cgs and esu units. However, we can use our new quantity,  $\bar{P}^{\rm NL}(z,\omega)$ , to mask this term by defining  $\bar{P}^{\rm NL}(z,\omega)$  in cgs or esu units:  $\bar{P}^{\rm NL}(z,\omega) = 4\pi P^{\rm NL}(z,\omega)$ . This then leaves  $\omega^2/c^2 \bar{P}^{\rm NL}(z,\omega)$ .

We see then that we can generalize Eq. 3.4.8 to be:

$$\frac{\partial^2}{\partial z^2} E(z,\omega) + \frac{\left[n(\omega) - i\kappa(\omega)\right]^2 \omega^2}{c^2} E(z,\omega) = -\frac{\omega^2}{c^2} \bar{P}^{\rm NL}(z,\omega)$$
(3.4.11)

with the behavior of  $\bar{P}^{\mathrm{NL}}(z,\omega)$  defined to be  $\bar{P}^{\mathrm{NL}}(z,\omega) = [P^{\mathrm{NL}}(z,\omega)]/\epsilon_0$  in mks units or  $\bar{P}^{\mathrm{NL}}(z,\omega) = 4\pi P^{\mathrm{NL}}(z,\omega)$ in cgs or esu units.

Before continuing, let us consider the nature of the complex index of refraction in Eq. 3.4.9 and 3.4.10. The quantity consists of a real component,  $n(\omega)$ , and an imaginary component,  $\kappa(\omega)$ . While the index of refraction is routinely encountered, the imaginary component, the extinction coefficient, can be related to more recognizable quantities. To do so, let us consider the propagation of a plane wave in the absence of a nonlinear polarization. This is consistent with Eq. 3.4.11 when  $\bar{P}^{\rm NL} = 0$ :

$$\frac{\partial^2}{\partial z^2} E(z,\omega) = -\left(\frac{[n(\omega) - i\kappa(\omega)]\omega}{c}\right)^2 E(z,\omega)$$
(3.4.12)

Recognizing Eq. 3.4.12 as a second-order ordinary differential equation, we might consider a trial solution for E of the form  $E(z, \omega) = E_0 \exp(-iaz)$ , where a is an arbitrary coefficient. Introducing this trial solution, we would find that  $a^2 = ([n(\omega) - i\kappa(\omega)]\omega/c)^2$ , which yields the solution:

$$E(z,\omega) = E_0 e^{-i[(n(\omega)\omega)/c]z} e^{-[(\kappa(\omega)\omega)/c]z}$$
(3.4.13)

We can see that the result in Eq. 3.4.13 exhibits two exponential terms: an oscillatory term with a wavenumber  $k = \frac{[n(\omega)\omega]}{c}$  and a term corresponding to exponential decay at a rate of  $\frac{[\kappa(\omega)\omega]}{c}$ .

Starting with the first term, we can recall our earlier definition of the wave number,  $k = \omega/v$  (see Eq. 3.1.5). Comparing these expressions, we see that the real part of the complex index of refraction behaves consistently with the index of refraction: the refraction of the medium reduces the velocity of light relative to the vacuum, yielding a speed,  $v(\omega) = c/n(\omega)$ .[101] This term also allows us to generalize our definition for the wavenumber of light:

$$k = \frac{n(\omega)\omega}{c} \tag{3.4.14}$$

Here we will note that the notation  $n(\omega)$  explicitly allows for a frequency dependence to the observed index of refraction.[99] In order to describe  $\kappa(\omega)$ , it will be more convenient to consider the intensity of the solution in Eq. 3.4.13. Generally, the intensity of the electric field is proportional to the square magnitude of the field; the proportionality constant is either  $[cn(\omega)]/(8\pi)$  (in cgs or esu units) or  $[c\epsilon_0 n(\omega)]/2$  (in mks units).[99] However, it is sufficient to identify that the intensity of the field in Eq. 3.4.13 exhibits a spatial dependence:

$$I(z) \propto E_0^2 e^{-2[(\kappa(\omega)\omega)/c]z}$$
 (3.4.15)

Comparing the intensity in Eq. 3.4.15 with the Beer-Lambert law,  $I(z) = I_0 \exp[-\alpha(\omega)z]$ , we can see that  $\kappa(\omega)$  is related to the absorption coefficient. If we set  $\alpha(\omega)$  equal to the coefficient in Eq. 3.4.15, we can see that  $\kappa(\omega)$  is defined:

$$\kappa(\omega) = \frac{c}{2\omega}\alpha(\omega) \tag{3.4.16}$$

where the notation  $\kappa(\omega)$  and  $\alpha(\omega)$  explicitly allow a potential frequency dependence for these quantities.[99]

Applying these insights, let us continue from Eq. 3.4.11. The first thing we will consider is expressing the coefficient to the second term on the left hand side,  $[n(\omega)-i\kappa(\omega)]^2\omega^2/c^2$ , in a more convenient manner. If we evaluate the term  $[n(\omega) - i\kappa(\omega)]^2$ :

$$[n(\omega) - i\kappa(\omega)]^2 = n^2(\omega) - i2n(\omega)\kappa(\omega) - \kappa^2(\omega)$$
(3.4.17)

we can see that the coefficient to this term can be evaluated:

$$\frac{\left[n(\omega) - i\kappa(\omega)\right]^2 \omega^2}{c^2} = \frac{\omega^2}{c^2} \left[n^2(\omega) - i2n(\omega)\kappa(\omega) - \kappa^2(\omega)\right] = \left[\frac{n(\omega)\omega}{c}\right]^2 - i2\left[\frac{n(\omega)\omega}{c}\right]\frac{\kappa(\omega)\omega}{c} - \frac{\kappa^2(\omega)\omega^2}{c^2}$$
(3.4.18)

However, we have previously defined the wavenumber,  $k(\omega) = \frac{[n(\omega)\omega]}{c}$  (see Eq. 3.4.14), allowing us to substitute  $k(\omega)$  for both terms in brackets in Eq. 3.4.18. We can also consider our result for  $\kappa(\omega)$  in Eq. 3.4.16. From  $\kappa(\omega) = \frac{c}{(2\omega)\alpha(\omega)}$ , we can see that  $\frac{[2\kappa(\omega)\omega]}{c}$  in the second term of Eq. 3.4.18 is equivalent to  $\alpha(\omega)$ . Finally, if we assume  $\alpha(\omega)$  is sufficiently small, we can consider the third term, proportional to  $\alpha^2(\omega)$ , to be negligible relative to the first two terms due to their proportionality to  $k^2(\omega)$  and  $k(\omega)$ . This then leaves us:

$$\frac{\partial^2}{\partial z^2} E(z,\omega) + \left[k^2(\omega) - ik(\omega)\alpha(\omega)\right] E(z,\omega) = -\frac{\omega^2}{c^2} \bar{P}^{\rm NL}(z,\omega)$$
(3.4.19)

From Eq. 3.4.19, we will make two substitutions: first, we will introduce the form of the electric field and

evaluate  $\partial^2/\partial z^2 E(z,\omega)$ ; second, we will consider  $k(\omega)$  as a Taylor expansion around the carrier frequency,  $\omega = \omega_0$ .

We indicated that we will consider the field to be of the form in Eq. 3.1.27, but consider it sufficient to track the propagation of the complex field, i.e.  $\tilde{E}(z,t) = A(r,t) \exp[-i(k_0 z - \omega_0 t)]$ . If we take the Fourier transform of this equation, we find  $E(z,\omega) = \hat{A}(z,\omega-\omega_0) \exp(-ik_0 z)$ ,[112] consistent with our result in § 3.1.2 within the limit that the contribution of  $\exp[-i(\Delta \omega/v)z]$  is negligible; it will be trivial to relax this assumption later. If we then evaluate the first term of Eq. 3.4.19, we find the second derivative to yield:

$$\frac{\partial^2}{\partial z^2} E(z,\omega) = \frac{\partial}{\partial z} \left[ \frac{\partial}{\partial z} \hat{A}(z,\Delta\omega) e^{-ik_0 z} \right] 
= \frac{\partial}{\partial z} \left[ -ik_0 \hat{A}(z,\Delta\omega) e^{-ik_0 z} + e^{-ik_0 z} \frac{\partial}{\partial z} \hat{A}(z,\Delta\omega) \right] 
= \left[ (-ik_0)^2 e^{-ik_0 z} - i2k_0 e^{-ik_0 z} \frac{\partial}{\partial z} + e^{-ik_0 z} \frac{\partial^2}{\partial z^2} \right] \hat{A}(z,\Delta\omega)$$
(3.4.20)

where we have re-introduced our variable  $\Delta \omega = \omega - \omega_0$ . We can substitute the result of the scheme in Eq. 3.4.20 into Eq. 3.4.19. Since the evaluation in Eq. 3.4.20 removes the exponential term from the derivatives, we can also divide through the resulting expression by  $\exp(-ik_0 z)$ , leaving:

$$\left(\frac{\partial^2}{\partial z^2} - i2k_0\frac{\partial}{\partial z}\right)\hat{A}(z,\Delta\omega) + \left[k^2(\omega) - k_0^2 - ik(\omega)\alpha(\omega)\right]\hat{A}(z,\Delta\omega) = -\frac{\omega^2}{c^2}e^{ik_0z}\bar{P}^{\rm NL}(z,\omega)$$
(3.4.21)

Meanwhile, we can define the Taylor series for the function f(x) around x = a by the general formula:[116]

$$f(x) = \sum_{n=0}^{\infty} \frac{1}{n!} \left. \frac{\partial^n}{\partial x^n} f \right|_{x=a} (x-a)^n \tag{3.4.22}$$

From the formula of Eq. 3.4.22, we can therefore evaluate  $k(\omega)$  around  $\omega = \omega_0$  as the sum:

$$k(\omega) = k_0 + \left. \frac{\partial k(\omega)}{\partial \omega} \right|_{\omega = \omega_0} (\omega - \omega_0) + \frac{1}{2} \left. \frac{\partial^2 k(\omega)}{\partial \omega^2} \right|_{\omega = \omega_0} (\omega - \omega_0)^2 + \cdots$$
(3.4.23)

We will adopt the notation of Akhmanov et al and note the derivatives as  $k_1 = \frac{\partial k(\omega)}{\partial \omega}|_{\omega=\omega_0}$  and  $k_2 = \frac{\partial^2 k(\omega)}{\partial \omega^2}|_{\omega=\omega_0}$ ;[103] Boyd uses a similar notation for the first-order derivative, but generalizes the summation over all higher order terms into a single variable, D.[102] We can also relate these first two derivatives to meaningful quantities with respect to pulse propagation. The first-order derivative,  $k_1$ , is defined as the inverse of the group velocity,  $u_g$ ,[101, 103] while the second-order derivative is the group velocity dispersion.[103] The form and behavior of these quantities will be discussed in a later section (see § 3.5).

The sum of Eq. 3.4.23 permits evaluation of the terms associated with  $k(\omega)$  in Eq. 3.4.21. We can apply some approximations to simplify the treatment, though. First, we will assume that the non-zeroth order terms are sufficiently small that their products with the absorption coefficient are negligible, leaving  $k(\omega)\alpha(\omega) \approx k_0\alpha(\omega)$ . Second, we will recognize that the Fourier relationships of derivatives (see Eq. 3.4.6a and Eq. 3.4.6b) cause terms in  $(\omega - \omega_0)^n$  to correspond to the derivatives  $\partial^n/\partial t^n$ ; therefore, we will only keep terms up to the second order in  $(\omega - \omega_0)$ . Under this second approximation, we can evaluate  $k^2(\omega)$  to be:

$$k^{2}(\omega) = k_{0}^{2} + 2k_{0}k_{1}\Delta\omega + \left(k_{0}k_{2} + k_{1}^{2}\right)\Delta\omega^{2}$$
(3.4.24)

where we have again introduced  $\Delta \omega$  in place of  $(\omega - \omega_0)$ . Introducing these substitutions into Eq. 3.4.21 results in the form:

$$\left(\frac{\partial^2}{\partial z^2} - i2k_0\frac{\partial}{\partial z}\right)\hat{A}(z,\Delta\omega) + \left[2k_0k_1\Delta\omega + \left(k_0k_2 + k_1^2\right)\Delta\omega^2 - ik_0\alpha(\omega)\right] \\ \times \hat{A}(z,\Delta\omega) = -\frac{\omega^2}{c^2}e^{ik_0z}\bar{P}^{\rm NL}(z,\omega)$$
(3.4.25)

At this point, it is common to invoke the slowly varying amplitude approximation, which assumes the effects of  $2k_0\partial/\partial z$  are significantly greater than the effects of  $\partial^2/\partial z^2$ , allowing us to neglect the second-order derivative.[102, 103] We will consider the case in which we do not apply this assumption at this point in § 3.4.2, but for now we will apply the slowly varying amplitude approximation, leaving:

$$-i2k_0\frac{\partial}{\partial z}\hat{A}(z,\Delta\omega) + \left[2k_0k_1\Delta\omega + \left(k_0k_2 + k_1^2\right)\Delta\omega^2 - ik_0\alpha(\omega)\right]\hat{A}(z,\Delta\omega) = -\frac{\omega^2}{c^2}e^{ik_0z}\bar{P}^{\rm NL}(z,\omega) \quad (3.4.26)$$

Unfortunately, within the context of simulating ultrafast propagation, group velocities tend to be sufficiently large that it is not feasible to define a sufficiently conservative grid that accommodates the delay introduced by the group velocity over practical distances. To overcome this limit, it is convenient to transfer to a moving coordinate system  $(\xi, \eta)$ , defined relative to the laboratory coordinates (z, t) as:

$$\xi = z \tag{3.4.27a}$$

$$\eta = t \pm k_{1r}z \tag{3.4.27b}$$

In Eq. 3.4.27b,  $k_{1r}$  is the reciprocal of the reference group velocity, usually taken as the group velocity of one of the pulses in the simulation; we will show that this form of  $\eta$  results in the reduction of the effective group velocity of the pulses. The  $\pm$  in Eq. 3.4.27b generalizes this expression, as the use of different conventions may require the sign of the term  $k_{1r}z$  to be either positive or negative. We will ultimately end up needing the negative sign, i.e.  $\eta = t - k_{1r}z$ .

Recalling that terms in  $\omega^n$  correspond to the derivatives  $\partial^n/\partial t^n$  (see Eq. 3.4.6a and Eq. 3.4.6b), it is clear from examining Eq. 3.4.26 that migrating to the moving coordinate system  $(\xi, \eta)$  will require us to identify the operators equivalent to  $\partial/\partial z$  and  $\partial/\partial t$ . To do so, let us consider an arbitrary function, f, expressed as a function of  $\xi$  and  $\eta$ :  $f(\xi(z,t),\eta(z,t))$ . We can then use the chain rule to find the derivatives of f with respect to z and t, informing on the forms of the operators  $\partial/\partial z$  and  $\partial/\partial t$  in the moving frame:[116]

$$\frac{\partial}{\partial z}f\left(\xi(z,t),\eta(z,t)\right) = \left[\frac{\partial}{\partial z}\xi(z,t)\frac{\partial}{\partial \xi} + \frac{\partial}{\partial z}\eta(z,t)\frac{\partial}{\partial \eta}\right]f\left(\xi(z,t),\eta(z,t)\right)$$
(3.4.28a)

$$\frac{\partial}{\partial t}f\left(\xi(z,t),\eta(z,t)\right) = \left[\frac{\partial}{\partial t}\xi(z,t)\frac{\partial}{\partial\xi} + \frac{\partial}{\partial t}\eta(z,t)\frac{\partial}{\partial\eta}\right]f\left(\xi(z,t),\eta(z,t)\right)$$
(3.4.28b)

From the forms of Eq. 3.4.27a and 3.4.27b (recalling we will be using the case of  $-k_{1r}z$ ), it is trivial to evaluate  $\partial/\partial z\xi = 1$ ,  $\partial/\partial z\eta = -k_{1r}$ ,  $\partial/\partial t\xi = 0$ , and  $\partial/\partial t\eta = 1$ . With these derivatives, we can use Eq. 3.4.28a and 3.4.28b to find  $\partial/\partial z$  and  $\partial/\partial t$  in the moving frame:

$$\frac{\partial}{\partial z} = \frac{\partial}{\partial \xi} - k_{1r} \frac{\partial}{\partial \eta} \tag{3.4.29a}$$

$$\frac{\partial}{\partial t} = \frac{\partial}{\partial \eta} \tag{3.4.29b}$$

In order to bring Eq. 3.4.26 to the moving frame, we will need to return to the time domain. In contrast to Boyd,[102] however, we will consider the form of the polarization before applying the inverse Fourier transform. We assumed that the polarization is of the form in Eq. 3.2.17, i.e.  $\tilde{P}^{\rm NL}(z,t) = \bar{p}^{\rm NL}(z,t) \exp(i\omega_0 t)$ , where  $\bar{p}^{\rm NL}(z,t)$  is an envelope function defined in the same manner as  $\tilde{P}^{\rm NL}(z,t)$ .<sup>i</sup> In the same manner as the electric field, the Fourier transform of this yields  $\bar{P}^{\rm NL}(z,\omega) = \hat{p}^{\rm NL}(z,\omega-\omega_0)$ .[112] Substituting for  $\bar{P}^{\rm NL}(z,\omega)$ then allows us to perform the Fourier transform of Eq. 3.4.26 with respect to  $\Delta\omega$  without having to consider residual exponentials or the expansion of  $\partial^2/\partial t^2 \tilde{P}^{\rm NL}(z,t)$ :[102]

$$\begin{bmatrix} -i2k_0 \left(\frac{\partial}{\partial z} + k_1 \frac{\partial}{\partial t}\right) - (k_0 k_2 + k_1^2) \frac{\partial^2}{\partial t^2} \end{bmatrix} A(z, t) - \mathcal{F}^{-1} \left\{ ik_0 \alpha(\omega) \hat{A}(z, \Delta \omega) \right\}$$

$$= \frac{\omega^2}{c^2 \Delta \omega^2} e^{ik_0 z} \frac{\partial^2}{\partial t^2} \bar{p}^{\mathrm{NL}}(z, t)$$

$$(3.4.30)$$

where the notation of  $\mathcal{F}^{-1}\{ik_0\alpha(\omega)\hat{A}(z,\Delta\omega)\}$  denotes the inverse Fourier transform of the term in brackets.

<sup>&</sup>lt;sup>i</sup>Recall we have defined  $\bar{P}^{\mathrm{NL}}(z,\omega)$  to be  $[P^{\mathrm{NL}}(z,\omega)]/\epsilon_0$  in mks units and  $4\pi P^{\mathrm{NL}}(z,\omega)$  in cgs or esu units.  $\tilde{P}^{\mathrm{NL}}(z,t)$  is the time domain form of  $\bar{P}^{\mathrm{NL}}(z,\omega)$ , i.e.  $\tilde{P}^{\mathrm{NL}}(z,t) = \mathcal{F}^{-1}\{\bar{P}^{\mathrm{NL}}(z,\omega)\}.$ 

As  $\alpha(\omega)$  and  $\hat{A}(z, \Delta\omega)$  both exhibit frequency dependence, the inverse Fourier transform of this term is likely to correspond to a convolution in the time domain,[112] but the exact form of the result is of no consequence at this time; it is sufficient to leave this term in the form presented in Eq. 3.4.30.

From Eq. 3.4.30, we can transition to the moving frame. We will substitute for  $\partial/\partial z$  and  $\partial/\partial t$  according to Eq. 3.4.29a and 3.4.29b and can directly replace dependence upon z for dependence upon  $\xi$ . Since  $\eta$  is only shifted relative to t, we will also assume that A(z,t) and  $\bar{p}^{\rm NL}$  can be mapped directly onto  $\eta$ . Making these substitutions and combining like terms yields:

$$\begin{bmatrix} -i2k_0 \left( \frac{\partial}{\partial \xi} + (k_1 - k_{1r}) \frac{\partial}{\partial \eta} \right) - (k_0 k_2 + k_1^2) \frac{\partial^2}{\partial \eta^2} \end{bmatrix} A(\xi, \eta) - \mathcal{F}^{-1} \left\{ ik_0 \alpha(\omega) \hat{A}(\xi, \Delta \omega) \right\}$$

$$= \frac{\omega^2}{c^2 \Delta \omega^2} e^{ik_0 \xi} \frac{\partial^2}{\partial \eta^2} \bar{p}^{\mathrm{NL}}(\xi, \eta)$$
(3.4.31)

The form of Eq. 3.4.31 is not of great utility: it is not feasible to apply the effects of group velocity dispersion in the time domain in simulations.[118, 199–201] Therefore, we will need to return to the frequency domain by performing the Fourier transform of Eq. 3.4.31 with respect to the reduced frequency,  $\Delta \omega$ , yielding:

$$\left[-i2k_0\left(\frac{\partial}{\partial\xi}+i\Delta\omega\left(k_1-k_{1r}\right)\right)+\left(k_0k_2+k_1^2\right)\Delta\omega^2-ik_0\alpha(\omega)\right]\hat{A}(\xi,\Delta\omega)=-\frac{\omega^2}{c^2}e^{ik_0\xi}\hat{p}^{\rm NL}(\xi,\Delta\omega)\quad(3.4.32)$$

Rearranging the result in Eq. 3.4.32:

$$\frac{\partial}{\partial\xi}\hat{A}(\xi,\Delta\omega) = \left[-i\Delta\omega\left(k_1 - k_{1r}\right) - i\frac{1}{2}\Delta\omega^2\left(k_2 + \frac{k_1^2}{k_0}\right) - \frac{1}{2}\alpha(\omega)\right]\hat{A}(\xi,\Delta\omega) 
- i\frac{\omega^2}{2c^2k_0}e^{ik_0\xi}\hat{p}^{\mathrm{NL}}(\xi,\Delta\omega)$$
(3.4.33)

Finally, from Eq. 3.4.33, it is possible to address a couple approximations:

$$\frac{\partial}{\partial\xi}\hat{A}(\xi,\Delta\omega) = \left[-i\Delta\omega\left(k_1 - k_{1r}\right) - i\frac{1}{2}\Delta\omega^2 k_2 - \frac{1}{2}\alpha(\omega)\right]\hat{A}(\xi,\Delta\omega) - i\frac{\omega^2}{2c^2k_0}e^{ik(\omega)\xi}\hat{p}^{\rm NL}(\xi,\Delta\omega) \qquad (3.4.34)$$

The first approximation we introduce in Eq. 3.4.34 is that  $k_1^2$  will be small relative to  $k_0$ , allowing us to neglect this term; while it is possible that close to the limit of the slowly varying amplitude approximation, the contribution from  $\Delta \omega^2$  may weaken this assumption, it should generally be reasonable.[103] The second change in Eq. 3.4.34 is that we have recognized that  $\mathcal{F}\{A(\xi,t)\} = \hat{A}(\xi,\Delta\omega) \exp[-i(\Delta\omega/v)\xi]$ . Throughout the derivation, it was sufficient to mask the contribution of the exponential within the envelope function,  $\hat{A}(z,\Delta\omega)$ ; in Eq. 3.4.34, however, we have reintroduced this contribution explicitly: if we assume  $\Delta\omega$  is sufficiently small, the expansion of  $\partial/\partial\xi \hat{A}(\xi, \Delta\omega) \exp[-i(\Delta\omega/v)\xi]$  according to the product rule will be approximately  $\exp[-i(\Delta\omega/v)\xi]\partial/\partial\xi \hat{A}(\xi, \Delta\omega)$ . Dividing through the result by  $\exp[-i(\Delta\omega/v)\xi]$  yields  $\exp[i(\Delta\omega/v)\xi]$  on the right hand side; combining this term with  $\exp(ik_0\xi)$  yields the exponential  $\exp[ik(\omega)\xi]$  (see Eq. 3.1.24).

# 3.4.2 Relaxing the Slowly Varying Amplitude Approximation

In § 3.4.1, we traced through the derivation of treating pulse propagation effects, but introduced the slowly varying amplitude approximation prior to shifting to the moving frame. This neglects the term in  $\partial^2/\partial z^2$ ; as  $\partial/\partial z$  includes two contributions (see Eq. 3.4.29a), it should be expected that retaining this term into the moving frame will impact the resulting differential equation. We will apply this treatment here.

Starting from Eq. 3.4.25, we will directly shift this expression to the moving frame using Eq. 3.4.29a and 3.4.29b. As before, we will need to begin by performing an inverse Fourier transform to take Eq. 3.4.25 into the time domain. Substituting for  $\hat{p}(z, \Delta \omega)$  and performing this transformation yields a result similar to Eq. 3.4.30, only retaining the term associated with  $\partial^2/\partial z^2$ :

$$\begin{bmatrix} \frac{\partial^2}{\partial z^2} - i2k_0 \left(\frac{\partial}{\partial z} + k_1 \frac{\partial}{\partial t}\right) - (k_0 k_2 + k_1^2) \frac{\partial^2}{\partial t^2} \end{bmatrix} A(z, t) - \mathcal{F}^{-1} \left\{ ik_0 \alpha(\omega) \hat{A}(z, \Delta \omega) \right\}$$
  
$$= \frac{\omega^2}{c^2 \Delta \omega^2} e^{ik_0 z} \frac{\partial^2}{\partial t^2} \bar{p}^{\mathrm{NL}}(z, t)$$
(3.4.35)

Much like Eq. 3.4.30, we have left the term associated with absorption noted as simply the inverse Fourier transformation of the term in brackets,  $\mathcal{F}^{-1}$ {}, for simplicity.

We can now substitute according to Eq. 3.4.29a and 3.4.29b, yielding:

$$\begin{bmatrix} \frac{\partial^2}{\partial\xi^2} - 2k_{1r}\frac{\partial}{\partial\xi}\frac{\partial}{\partial\eta} - i2k_0\left(\frac{\partial}{\partial\xi} + (k_1 - k_{1r})\frac{\partial}{\partial\eta}\right) - \left(k_0k_2 + (k_1^2 - k_{1r}^2)\right)\frac{\partial^2}{\partial\eta^2}\end{bmatrix}A(\xi,\eta) - \mathcal{F}^{-1}\left\{ik_0\alpha(\omega)\hat{A}(\xi,\Delta\omega)\right\} = \frac{\omega^2}{c^2\Delta\omega^2}e^{ik_0\xi}\frac{\partial^2}{\partial\eta^2}\bar{p}^{\mathrm{NL}}(\xi,\eta)$$
(3.4.36)

Performing the Fourier transform on Eq. 3.4.36 to return to the frequency domain yields:

$$\begin{bmatrix} \frac{\partial^2}{\partial \xi^2} & -i2 \left[ (k_0 + k_{1r} \Delta \omega) \frac{\partial}{\partial \xi} + ik_0 \Delta \omega (k_1 - k_{1r}) \right] + (k_0 k_2 + (k_1^2 - k_{1r}^2)) \Delta \omega^2 \right] \hat{A}(\xi, \Delta \omega) - ik_0 \alpha(\omega)$$

$$\times \hat{A}(\xi, \Delta \omega) = -\frac{\omega^2}{c^2} e^{ik_0 \xi} \hat{p}(\xi, \Delta \omega)$$
(3.4.37)

From Eq. 3.4.37, we could take two approaches. The first would be to rearrange Eq. 3.4.37 in the manner of a second-order differential equation; though more difficult to evaluate numerically than a first-order differential

equation, it is still possible to arrive at a solution (see  $\S$  3.6).

Alternatively, it would also be possible to delay making the slowly varying amplitude approximation, only ignoring the second-order derivative after transferring to the moving frame. Eq. 3.4.38 shows the analogous expression to Eq. 3.4.34 within this treatment:

$$\frac{\partial}{\partial\xi}\hat{A}(\xi,\Delta\omega) = \left[-i\frac{k_0}{k_0+k_{1r}\Delta\omega}\Delta\omega(k_1-k_{1r}) - i\frac{1}{2}\frac{k_0}{k_0+k_{1r}\Delta\omega}k_2\Delta\omega^2 - \frac{1}{2}\frac{k_0}{k_0+k_{1r}\Delta\omega}\alpha(\omega)\right] \\
\times \hat{A}(\xi,\Delta\omega) - i\frac{\omega^2}{2(k_0+k_{1r}\Delta\omega)c^2}e^{ik(\omega)\xi}\hat{p}(\xi,\Delta\omega)$$
(3.4.38)

where we have assumed the term associated with  $(k_1^2 - k_{1r}^2)$  is negligible when divided by  $(k_0 + k_{1r}\Delta\omega)$  and have reintroduced the term  $\exp[-i(\Delta\omega/v)\xi]$ . Comparing Eq. 3.4.38 to the earlier equation, Eq. 3.4.34, we see that this treatment retains the first order term of the Taylor expansion of  $k(\omega)$  (see Eq. 3.4.23) when considering the derivative of the field. At the limit of narrow bandwidth pulses, the contributions of the first-order term  $(k_{1r}\Delta\omega)$  are negligible, and Eq. 3.4.38 approaches the same limiting behavior as Eq. 3.4.34. However, this first-order term in Eq. 3.4.38 allows for the deviation from the limiting behavior predicted by Eq. 3.4.34 as the pulse bandwidth broadens.

# 3.4.3 The Propagation of Pulsed Gaussian Beams

In § 3.4.1, we considered the effects of a medium upon the propagation of a pulsed electric field within the plane wave limit. The standard model for a  $\text{TEM}_{00}$  mode arises from treating the field as a distortion to the plane wave limit,[104, 105] imposing the transverse coordinate dependence demonstrated by the  $\text{TEM}_{00}$  mode. In this section, we will extend the treatments of Svelto[104] and Verdeyen[105] to account for the propagation effects of a medium upon a  $\text{TEM}_{00}$  mode.

Starting from the plane wave limit, i.e.  $E(z) = E_0 e^{-ik_0 z}$ ,<sup>i</sup> Verdeyen and Svelto consider TEM<sub>00</sub> mode propagation to reflect a functional distortion describing the transverse dependence exhibited by the field: whereas Svelto considers the distortion to reflect a spatially-dependent amplitude,  $U(\mathbf{r}, z)$ , in place of the constant  $E_0$ ,[104] Verdeyen treats the distortion as a function,  $\psi(r, z)$ , with the full field defined E(z) = $E_0\psi(\mathbf{r}, z)\exp(-ik_0 z)$ .[105] For clarity, we will adopt the style of Verdeyen; in this manner, we will assume we can consider the total field in a manner similar to Eq. 3.1.27, consisting of a frequency- or time-dependent envelope and oscillations at the carrier frequency, but with an additional function describing the distortion to the plane wave. However, as we will be incorporating dispersive effects into this distortion, we will not be

<sup>&</sup>lt;sup>i</sup>Svelto and Verdeyen both treat the electric field amplitude as a scalar and consider propagation at the limit that the field is monochromatic. [104, 105] We will retain the previous approximation as noted in § 3.3.3, but will extend the treatment of  $\text{TEM}_{00}$  mode propagation within the behavior of a pulsed electric field as described by Eq. 3.1.27.

able to consider the envelope function and the distortion separable in both the time and frequency domains: due to the Fourier relation between time and frequency, if these functions are separable in one domain, they must be convolved in the other. Considering the derivation in § 3.4.1, we will consider the envelope function and distortion to be separable in the frequency domain, which gives us the following descriptions for the field:

$$\tilde{E}(\mathbf{r},z,t) = A(\mathbf{r},z,t)e^{-ik_0z}e^{i\omega_0t}$$
(3.4.39a)

$$E(\mathbf{r}, z, t) = \hat{A}(\Delta\omega)e^{-i(\Delta\omega/v)z}\psi(\mathbf{r}, z, \Delta\omega)e^{-ik_0z}$$
(3.4.39b)

The time-domain model of the field in Eq. 3.4.39a is based upon the treatment of the field in Eq. 3.1.27, where we have allowed the envelope function,  $A(\mathbf{r}, z, t)$ , to exhibit the spatial dependence characteristic of the field. The frequency-domain model in Eq. 3.4.39b, meanwhile, is the Fourier transform with respect to  $\omega$  of Eq. 3.4.39a, with  $\mathcal{F}\{A(\mathbf{r}, z, t) \exp(i\omega_0 t)\} = \hat{A}(\Delta \omega) \exp[-i(\Delta \omega/v)z]\psi(\mathbf{r}, z, \Delta \omega)$ . The first two terms  $(\hat{A}(\Delta \omega)$  and  $\exp[-i(\Delta \omega/v)z])$  arise in a similar manner as the treatment of Eq. 3.1.26 yields Eq. 3.1.27, though in the case of the Fourier transform, we recognize the relation  $\mathcal{F}\{f(t) \exp(\pm \omega_0 t)\} = \hat{f}(\omega \mp \omega_0)[112]$ . Meanwhile, the function  $\psi(\mathbf{r}, z, \Delta \omega)$  reflects the distortion to the plane wave, isolating the spatial dependence of  $\hat{A}(\mathbf{r}, z, \Delta \omega)$  in Eq. 3.1.26 into a separate function.

In contrast to § 3.4.1, we will begin our treatment with Eq. 3.3.6. In a manner similar to § 3.4.1, though, we will consider  $\tilde{P}(\mathbf{r}, z, t)$  as the sum of the linear and nonlinear contributions,  $\tilde{P}^{(1)}(\mathbf{r}, z, t)$  and  $\tilde{P}^{\mathrm{NL}}(\mathbf{r}, z, t)$ , respectively. We will assume that the field being considered has sufficiently low intensity that it does not experience any self-induced nonlinear effects, allowing us to let  $\tilde{P}^{\mathrm{NL}}(\mathbf{r}, z, t)$  equal zero. Combining the derivatives of  $\tilde{E}(\mathbf{r}, z, t)$  and  $\tilde{P}^{(1)}(\mathbf{r}, z, t)$  with respect to time yields the derivative with respect to the linear displacement,  $\tilde{D}^{(1)}(\mathbf{r}, z, t)$ ; in the same manner as Eq. 3.4.3, we obtain:

$$\frac{\partial^2}{\partial z^2}\tilde{E}(\mathbf{r},z,t) + \nabla_{\mathrm{T}}^2\tilde{E}(\mathbf{r},z,t) - b^2 e \frac{\partial^2}{\partial t^2}\tilde{D}^{(1)}(\mathbf{r},z,t) = 0$$
(3.4.40)

From Eq. 3.4.40, we will next evaluate  $\partial^2/\partial z^2 \tilde{E}(\mathbf{r}, z, t)$  in the manner of the scheme in Eq. 3.4.20; this yields:

$$e^{-ik_0z} \left[ -k_0^2 - i2k_0\frac{\partial}{\partial z} + \frac{\partial^2}{\partial z^2} \right] A(\mathbf{r}, z, t) e^{i\omega_0 t} + \nabla_{\mathrm{T}}^2 \tilde{E}(\mathbf{r}, z, t) - b^2 e \frac{\partial^2}{\partial t^2} \tilde{D}^{(1)}(\mathbf{r}, z, t) = 0$$
(3.4.41)

Next, let us consider the Fourier transformation of the result in Eq. 3.4.41. We will also make two simplifications: multiplying through the equation by  $\exp(ik_0 z)$ , and substituting for the linear displacement according to Eq. 3.4.4b. These changes yield:

$$\left[-k_0^2 - i2k_0\frac{\partial}{\partial z} + \frac{\partial^2}{\partial z^2} + e^{ik_0z}\nabla_{\mathrm{T}}^2 e^{-ik_0z} + b^2 e\omega^2 \epsilon(\omega)\right]\hat{A}(\Delta\omega)e^{-i(\Delta\omega/\nu)z}\psi(\mathbf{r}, z, \Delta\omega) = 0$$
(3.4.42)

From Eq. 3.4.42, we can achieve a couple further simplifications. First, since the transverse Laplacian operator  $(\nabla_{\rm T}^2)$  only involves the transverse coordinates, we can swap the order of  $\nabla_{\rm T}^2$  and  $\exp(-ik_0z)$ ; we can then multiply  $\exp(ik_0z)$  and  $\exp(-ik_0z)$  to remove both exponentials. Second, we can substitute for  $b^2 e \omega^2 \epsilon(\omega)$  according to Eq. 3.4.9, 3.4.10, and 3.4.18; as before, these terms reduce to  $k^2(\omega) - ik(\omega)\alpha(\omega)$ , and can be combined with the leading term of  $k_0^2$ . Third, since  $\hat{A}(\Delta\omega)$  does not exhibit any dependence on the spatial coordinates, we can move it outside each derivative and divide Eq. 3.4.42 by this term. These changes yield:

$$\left[\frac{\partial^2}{\partial z^2} - i2k_0\frac{\partial}{\partial z} + \nabla_{\mathrm{T}}^2 + \left(k^2(\omega) - k_0^2 - ik(\omega)\alpha(\omega)\right)\right]e^{-i(\Delta\omega/v)z}\psi(\mathbf{r}, z, \Delta\omega) = 0$$
(3.4.43)

From Eq. 3.4.43, we can achieve a few more simplifications. First, let us consider the remaining exponential. Once again recalling that the transverse Laplacian,  $\nabla_{\rm T}^2$ , has no longitudinal dependence, we can again switch the order of the Laplacian and the exponential, namely:  $\nabla_{\rm T}^2 \exp[-i(\Delta\omega/v)z] = \exp[-i(\Delta\omega/v)z]\nabla_{\rm T}^2$ . We can also consider the derivatives with respect to z. Considering these derivatives to be generalized in the form of the  $n^{\rm th}$ -order derivative with respect to z,  $\partial^n/\partial z^n \exp[-i(\Delta\omega/v)z]\psi(\mathbf{r}, z, \Delta\omega)$ , each operation of one of the n derivatives should yield two terms due to the product rule:[116] one corresponding to the derivative of the exponential  $(\partial/\partial z \exp[-i(\Delta\omega/v)z] = -i(\Delta\omega/v)\exp[-i(\Delta\omega/v)z])$ , and one corresponding to the derivative of the function  $\psi(\mathbf{r}, z, \Delta\omega)$ . However, if we take  $\Delta\omega$  to be reasonably small, we can neglect the terms arising from the derivative of the exponential, thereby assuming:  $\frac{\partial^n}{\partial z^n} \exp[-i(\Delta\omega/v)z]\psi(\mathbf{r}, z, \Delta\omega) \approx$  $\exp[-i(\Delta\omega/v)z][\partial^n/\partial z^n\psi(\mathbf{r}, z, \Delta\omega)]$ . This allows us to divide through the equation by this remaining exponential. We can then also invoke the slowly varying amplitude approximation,  $2k_0\partial/\partial z \gg \partial^2/\partial z^2$ , and ignore the second-order derivative. Finally, we can also substitute for  $k^2(\omega)$  and  $k(\omega)$  according to Eq. 3.4.23 and 3.4.24. These changes and assumptions yield:

$$\left[-i2k_0\frac{\partial}{\partial z} + \nabla_{\mathrm{T}}^2 + 2k_0k_1\Delta\omega + (k_0k_2 + k_1^2)\Delta\omega^2 - ik_0\alpha(\omega)\right]\psi(\mathbf{r}, z, \Delta\omega) = 0$$
(3.4.44)

The last consideration we will introduce before determining the nature of  $\psi(\mathbf{r}, z, \Delta \omega)$  is to bring Eq. 3.4.44 to the moving frame. Assuming an inverse Fourier transform of  $\psi(\mathbf{r}, z, \Delta \omega)$  exists, we can perform

the inverse Fourier transform of Eq. 3.4.44 to yield:

$$\left[-i2k_0\left(\frac{\partial}{\partial z}+k_1\frac{\partial}{\partial t}\right)+\nabla_{\mathrm{T}}^2-(k_0k_2+k_1^2)\frac{\partial^2}{\partial t^2}\right]\mathcal{F}^{-1}\{\psi(\mathbf{r},z,\Delta\omega)\}-\mathcal{F}^{-1}\{ik_0\alpha(\omega)\psi(\mathbf{r},z,\Delta\omega)\}=0 \quad (3.4.45)$$

In Eq. 3.4.45, it is sufficient to consider the inverse Fourier transforms of  $\psi(\mathbf{r}, z, t)$  and of  $ik_0\alpha(\omega)\psi(\mathbf{r}, z, \Delta\omega)$ to be noted as,  $\mathcal{F}^{-1}\{\psi(\mathbf{r}, z, \Delta\omega)\}$  and  $\mathcal{F}^{-1}\{ik_0\alpha(\omega)\psi(\mathbf{r}, z, \Delta\omega)\}$ , respectively. With Eq. 3.4.45, we can directly apply the substitutions (Eq. 3.4.29a and 3.4.29b) necessary to migrate to the moving frame:

$$\begin{bmatrix} -i2k_0 \left( \frac{\partial}{\partial \xi} + (k_1 - k_{1r}) \frac{\partial}{\partial \eta} \right) + \nabla_{\mathrm{T}}^2 - (k_0 k_2 + k_1^2) \frac{\partial^2}{\partial \eta^2} \end{bmatrix}$$

$$\times \mathcal{F}^{-1} \{ \psi(\mathbf{r}, \xi, \Delta \omega) \} - \mathcal{F}^{-1} \{ i k_0 \alpha(\omega) \psi(\mathbf{r}, \xi, \Delta \omega) \} = 0$$
(3.4.46)

With Eq. 3.4.46 now describing propagation within the moving frame, we can return to the frequency domain. We will also take the opportunity to divide through Eq. 3.4.46 by the term  $-i2k_0$  at the beginning; in addition to reducing the number of terms dependent upon  $k_0$ , it will also allow us to invoke the assumption that  $k_1^2/k_0$  is likely small and can be neglected. This leaves us with:

$$\left[\frac{\partial}{\partial\xi} + i(k_1 - k_{1r})\Delta\omega + i\frac{1}{2k_0}\nabla_{\mathrm{T}}^2 + i\frac{1}{2}k_2\Delta\omega^2 + \frac{1}{2}\alpha(\omega)\right]\psi(\mathbf{r},\xi,\Delta\omega) = 0$$
(3.4.47)

Moving forward, we will also introduce a new variable,  $\delta k(\Delta \omega)$ :

$$\delta k(\Delta \omega) = i(k_1 - k_{1r})\Delta \omega + i\frac{1}{2}k_2\Delta \omega^2 + \frac{1}{2}\alpha(\omega)$$
(3.4.48)

Introducing  $\delta k(\Delta \omega)$  simplifies Eq. 3.4.47 further, leaving:

$$\left[\frac{\partial}{\partial\xi} + i\frac{1}{2k_0}\nabla_{\rm T}^2 + \delta k(\Delta\omega)\right]\psi(\mathbf{r},\xi,\Delta\omega) = 0$$
(3.4.49)

In Eq. 3.4.49, we now have a compact description of the effects of field propagation on the distortion function,  $\psi(\mathbf{r}, \xi, \Delta \omega)$ , within a moving coordinate system. Now, we will seek to develop a closed-form solution for this distortion function. We begin by considering the form of the transverse Laplacian,  $\nabla_{\mathrm{T}}^2$ . We previously considered this operator in the form of Eq. 3.3.5, in which it is described in (x,y) coordinates. Since we expect the TEM<sub>00</sub> mode to exhibit Gaussian symmetry, though, it will prove more convenient to begin with

<sup>&</sup>lt;sup>i</sup>Recall  $\Delta \omega$  has been introduced as a compact variable for  $\omega - \omega_0$ . Therefore, we could reasonably consider  $\delta k$  to be a function of either  $\omega$  or  $\Delta \omega$ , but it would be redundant to consider  $\delta k$  to be a function of both. As we have been describing the distortion as a function of  $\Delta \omega$ , we have opted to treat  $\delta k$  as a function of this variable rather than of  $\omega$ .

 $\nabla_{\rm T}^2$  in cylindrical coordinates. In this coordinate system, the full Laplacian operator is defined:[101]

$$\nabla^2 = \frac{1}{r} \frac{\partial}{\partial r} r \frac{\partial}{\partial r} + \frac{1}{r^2} \frac{\partial^2}{\partial \phi^2} + \frac{\partial^2}{\partial \xi^2}$$
(3.4.50)

However, we have already defined the transverse Laplacian,  $\nabla_{\rm T}^2$ , as encompassing the terms of the Laplacian operator not associated with the longitudinal direction,  $\xi$ ; we can therefore neglect the last term in Eq. 3.4.50. We can also neglect the angular dependence in Eq. 3.4.50: since the distortion to the plane wave,  $\psi(r,\xi,\Delta\omega)$ ,<sup>i</sup> is expected to describe a two-dimensional Gaussian, we can invoke the cylindrical symmetry of this function to assume  $\partial^2/\partial\phi^2\psi(r,\xi,\Delta\omega) = 0.[104, 105]$  We are then left only with the first term, the dependence upon the radial coordinate, r. Through evaluation by the product rule, we can show that this remaining term leaves the following form of the transverse Laplacian operator:

$$\nabla_{\rm T}^2 = \frac{1}{r} \frac{\partial}{\partial r} + \frac{\partial^2}{\partial r^2} \tag{3.4.51}$$

Substitution for the transverse Laplacian according to Eq. 3.4.51 then leaves:

$$\left[\frac{\partial}{\partial\xi} + i\frac{1}{2k_0}\left(\frac{1}{r}\frac{\partial}{\partial r} + \frac{\partial^2}{\partial r^2}\right) + \delta k(\Delta\omega)\right]\psi(r,\xi,\Delta\omega) = 0$$
(3.4.52)

With Eq. 3.4.52, we can now determine the functional behavior of  $\psi(r,\xi,\Delta\omega)$ , beginning by defining a trial solution as well as appropriate boundary conditions. Starting with the latter, we will define the position  $\xi = 0$  as the focus of the mode. In the absence of dispersive effects, we will expect the radial dependence of  $\psi(r,\xi = 0,\Delta\omega)$  to be consistent with a two-dimensional Gaussian of width  $w_0$ ; in other words, we will expect a functional form of  $\exp(-r^2/w_0^2)$ . Meanwhile, when we introduce dispersive effects, we will assume that dispersion from the medium functionally depends upon the pathlength through the medium; in other words, we will require no dispersive effects at a point  $\xi = \xi'$  that corresponds to the beginning of the medium. Meanwhile, for a trial solution, we have two examples to consider. Verdeyen assumes a specific form for  $\psi(r,\xi)$ , taking the model  $\exp[-i(P(\xi) + (kr^2)/[2q(\xi)])]$ ;[105] while this trial solution is convenient when considering the final result for  $\psi(r,\xi)$  for free space propagation, it may not prove adequately general when dispersion is incorporated. Consequently, we will instead adopt a trial solution similar to that of Svelto, in which  $U(r,\xi)^{ii}$  is taken as a functional form incorporating two functions  $\alpha(\xi)$ 

<sup>&</sup>lt;sup>i</sup>It is noted here that since we are now considering the functional form of  $\psi(r, \xi, \Delta \omega)$  within the cylindrical coordinate system, we no longer have to consider the generalized vector **r** for the transverse coordinates; it is sufficient to consider the functional dependence upon the scalar radial coordinate, r.

<sup>&</sup>lt;sup>ii</sup>Recall, Svelto's model for distortion to the plane wave adopts the function  $U(r,\xi)$  in the same manner that Verdeyen

and  $\beta(\xi)$ ,  $U(r,\xi) = \exp[\alpha(\xi) - \beta(\xi)r^2]$ .[104] Using Svelto's treatment as a starting point, we also need to allow for frequency dependence due to our generalized dispersion variable  $\delta k(\Delta \omega)$ ; we will therefore tentatively allow  $\alpha$  and  $\beta$  to both exhibit a frequency dependence, yielding:

$$\psi(r,\xi,\Delta\omega) = e^{\alpha(\xi,\Delta\omega) - \beta(\xi,\Delta\omega)r^2}$$
(3.4.53)

Introducing this trial solution:

$$\left[\frac{\partial}{\partial\xi} + i\frac{1}{2k_0}\left(\frac{1}{r}\frac{\partial}{\partial r} + \frac{\partial^2}{\partial r^2}\right) + \delta k(\Delta\omega)\right]\exp^{\alpha(\xi,\Delta\omega) - \beta(\xi,\Delta\omega)r^2} = 0$$
(3.4.54)

From Eq. 3.4.54, we see that we will need to evaluate the derivatives of our trial solution with respect to  $\partial/\partial r$ ,  $\partial^2/\partial r^2$ , and  $\partial/\partial \xi$ . It can be shown that these derivatives evaluate to yield:

$$\frac{\partial}{\partial r}e^{\alpha(\xi,\Delta\omega)-\beta(\xi,\Delta\omega)r^2} = -2\beta(\xi,\Delta\omega)re^{\alpha(\xi,\Delta\omega)-\beta(\xi,\Delta\omega)r^2}$$
(3.4.55a)

$$\frac{\partial^2}{\partial r^2} e^{\alpha(\xi,\Delta\omega) - \beta(\xi,\Delta\omega)r^2} = \left[-2\beta(\xi,\Delta\omega) + 4\beta^2(\xi,\Delta\omega)r^2\right] e^{\alpha(\xi,\Delta\omega) - \beta(\xi,\Delta\omega)r^2}$$
(3.4.55b)

$$\frac{\partial}{\partial\xi}e^{\alpha(\xi,\Delta\omega)-\beta(\xi,\Delta\omega)r^2} = \left[\left(\frac{\partial}{\partial\xi}\alpha(\xi,\Delta\omega)\right) - r^2\left(\frac{\partial}{\partial\xi}\beta(\xi,\Delta\omega)\right)\right]e^{\alpha(\xi,\Delta\omega)-\beta(\xi,\Delta\omega)r^2} \tag{3.4.55c}$$

If we consider the forms of Eq. 3.4.55a, 3.4.55b, and 3.4.55c, we see that evaluation of the derivatives leaves the exponential unchanged and no longer within derivative expressions. Upon substituting the results of Eq. 3.4.55a, 3.4.55b, and 3.4.55c into Eq. 3.4.54, we can then divide through the resulting expression by the exponential, leaving:

$$\left[\left(\frac{\partial}{\partial\xi}\alpha(\xi,\Delta\omega)\right) - r^2\left(\frac{\partial}{\partial\xi}\beta(\xi,\Delta\omega)\right)\right] + i\frac{1}{2k_0}\left[-4\beta(\xi,\Delta\omega) + 4\beta^2(\xi,\Delta\omega)r^2\right] + \delta k(\Delta\omega) = 0 \qquad (3.4.56)$$

Examining Eq. 3.4.56, we see that the five terms can be divided into two groups: two terms with a dependence upon  $r^2$ , and three terms with no dependence upon r. Separating the terms accordingly yields:

$$r^{2}\left[-\left(\frac{\partial}{\partial\xi}\beta(\xi,\Delta\omega)\right)+i\frac{2}{k_{0}}\beta^{2}(\xi,\Delta\omega)\right]+\left[\left(\frac{\partial}{\partial\xi}\alpha(\xi,\Delta\omega)\right)-i\frac{2}{k_{0}}\beta(\xi,\Delta\omega)+\delta k(\Delta\omega)\right]=0$$
(3.4.57)

However, we still require the left hand side of Eq. 3.4.57 to equal zero. In Eq. 3.4.57, though, we have terms dependent upon  $r^2$  as well as terms independent of r; the only way to maintain the required equality is for adopts  $E_0\psi(r,\xi)$ .[104, 105]
both terms in brackets to individually equal zero.[104, 105] Eq. 3.4.57 then is consistent with the following pair of differential equations:

$$\frac{\partial}{\partial\xi}\beta(\xi,\Delta\omega) = i\frac{2}{k_0}\beta^2(\xi,\Delta\omega) \tag{3.4.58a}$$

$$\frac{\partial}{\partial\xi}\alpha(\xi,\Delta\omega) = i\frac{2}{k_0}\beta(\xi,\Delta\omega) - \delta k(\Delta\omega)$$
(3.4.58b)

This system of differential equations can be evaluated to determine the functional forms of  $\alpha(\xi, \Delta\omega)$  and  $\beta(\xi, \Delta\omega)$ .

Of the differential equations in Eq. 3.4.58a and 3.4.58b, the former will be the more straightforward to solve, as it exhibits a dependence only upon the function  $\beta(\xi, \Delta \omega)$ . However, we note that the only dependence upon  $\Delta \omega$  in Eq. 3.4.58a is the dependence implied by our assumed function,  $\beta(\xi, \Delta \omega)$ . We can assume, therefore, that it is sufficient to consider  $\beta(\xi)$ , a function of  $\xi$  only. Framing Eq. 3.4.58a to be of the form  $\partial/\partial\xi\beta(\xi) = ia\beta^2(\xi)$ , with  $a = 2/k_0$ , it can be shown that the general solution is of the form:<sup>i</sup>

$$\beta(\xi) = (C - ia\xi)^{-1} \tag{3.4.59}$$

where C is a constant of integration. In order to determine the value of C, let us consider the evaluation of this general solution at  $\xi = 0$ . We previously indicated that we require the mode at the focus ( $\xi = 0$ ) to be consistent with a Gaussian function with size  $w_0$ ,  $\exp(-r^2/w_0^2)$ . Since our trial function is of the form  $\exp[\alpha(\xi, \Delta \omega) - \beta(\xi)r^2]$ , we therefore require  $\beta(\xi = 0) = w_0^{-2}$ . Given the general solution in Eq. 3.4.59,  $\beta(\xi = 0) = C^{-1}$ , allowing us to assign  $C = w_0^2$  and to define the solution for  $\beta(\xi)$ :

$$\beta(\xi) = \left(w_0^2 - i\frac{2}{k_0}\xi\right)^{-1} \tag{3.4.60}$$

However, we can follow the example of Verdeyen[105] and make the solution in Eq. 3.4.60 more meaningful by relating it to the quantities  $w(\xi)$  and  $R(\xi)$ , which describe the beam waist and radius of curvature, respectively, as a function of longitudinal position. The functional forms of these parameters are:

$$w(\xi) = w_0 \sqrt{1 + (\frac{\xi}{\xi_0})^2}$$
(3.4.61)

<sup>&</sup>lt;sup>i</sup>The general differential equation was evaluated with the calculator at wolframalpha.com.

65

$$R(\xi) = \begin{cases} \xi(1 + (\frac{\xi_0}{\xi})^2), & \text{for } z \neq 0, \pm \infty \\ 0, & \text{for } z = 0, \pm \infty \end{cases}$$
(3.4.62)

where the quantity  $\xi_0$  in Eq. 3.4.61 and 3.4.62 is the Rayleigh length of the mode;<sup>i</sup> this quantity is defined:

$$\xi_0 = \frac{\pi w_0^2}{\lambda} \tag{3.4.63}$$

Considering Eq. 3.4.61, we can see  $\xi_0$  corresponds to the distance over which the beam waist expands by a factor of  $\sqrt{2}$  from the focus. Returning to the beam waist and radius of curvature, the first step to relating  $\beta(\xi)$  to these quantities is to show that the solution for  $\beta(\xi)$  (Eq. 3.4.60) is equivalent to:

$$\beta(\xi) = i \frac{k_0}{2} (\xi + i\xi_0)^{-1} \tag{3.4.64}$$

We can simplify the form for  $\beta(\xi)$  in Eq. 3.4.64 by multiplying this result by  $(\xi - i\xi_0)/(\xi - i\xi_0)$ , from which  $\beta(\xi)$  is equivalent to the sum:

$$\beta(\xi) = \frac{k_0}{2} \frac{\xi_0}{\xi^2 + \xi_0^2} + i \frac{k_0}{2} \frac{\xi}{\xi^2 + \xi_0^2}$$
(3.4.65)

From Eq. 3.4.65, the next step is to multiply the first term by  $\xi_0^{-2}/\xi_0^{-2}$  and the second term by  $\xi^{-2}/\xi^{-2}$ . This yields:

$$\beta(\xi) = \frac{k_0}{2\xi_0} \frac{1}{(\xi/\xi_0)^2 + 1} + i\frac{k_0}{2} \frac{1}{\xi[1 + (\xi_0/\xi)^2]}$$
(3.4.66)

Considering the definitions of  $w(\xi)$ ,  $R(\xi)$ , and  $\xi_0$  (Eq. 3.4.61, 3.4.62, and 3.4.63), it is now straightforward to show Eq. 3.4.66 reduces to:

$$\beta(\xi) = \frac{1}{w^2(\xi)} + i\frac{k_0}{2R(\xi)} \tag{3.4.67}$$

With a solution for  $\beta(\xi)$ , it is now possible to consider Eq. 3.4.58b to determine the solution for  $\alpha(\xi, \Delta\omega)$ . It will turn out most convenient to use the form of Eq. 3.4.64 for substitution into Eq. 3.4.58b; introducing this solution yields:

$$\frac{\partial}{\partial\xi}\alpha(\xi,\Delta\omega) = i\frac{2}{k_0} \left[ i\frac{k_0}{2} (\xi + i\xi_0)^{-1} \right] - \delta k(\Delta\omega)$$
(3.4.68)

In a similar manner to our treatment of  $\beta(\xi)$ , we can consider Eq. 3.4.68 to be of the generalized form  $\partial/\partial\xi\alpha(\xi,\Delta\omega) = -(\xi+ia)^{-1} - b(\Delta\omega)$ , with  $a = \xi_0$  and  $b(\Delta\omega) = \delta k(\Delta\omega)$ . A differential equation of this form

<sup>&</sup>lt;sup>i</sup>Note that it is more common to describe the Rayleigh length as  $z_0$ ; however, because we are using the coordinate of the moving frame,  $\xi$ , we are noting this length as  $\xi_0$  in an analogous manner. Note that because  $\xi = z$  (see Eq. 3.4.27a),  $\xi_0 = z_0$ .

will have a solution of the form:<sup>i</sup>

$$\alpha(\xi, \Delta\omega) = -\ln(a - i\xi) - ib(\Delta\omega)(a - i\xi) + C \tag{3.4.69}$$

Considering the form of the general solution in Eq. 3.4.69, we see that it involves two terms with dependence upon  $\xi$ : the first term,  $\ln(a - i\xi)$ , depends only upon  $\xi$ ; the second,  $-ib(\Delta\omega)(a - i\xi)$ , on the other hand exhibits a dependence upon  $\Delta\omega$  in addition to a dependence upon  $\xi$ . Therefore, let us consider  $\alpha(\xi, \Delta\omega)$  to represent the sum of two functions,  $\alpha_1(\xi)$  and  $\alpha_2(\xi, \Delta\omega)$ , corresponding to the first and second terms in Eq. 3.4.69, respectively:

$$\alpha_1(\xi) = -\ln(a - i\xi) + C_1 \tag{3.4.70a}$$

$$\alpha_2(\xi, \Delta\omega) = -ib(\Delta\omega)(a - i\xi) + C_2 \tag{3.4.70b}$$

where the sum of  $C_1$  (Eq. 3.4.70a) and  $C_2$  (Eq. 3.4.70b) is C (Eq. 3.4.69). The division of  $\alpha(\xi, \Delta\omega)$  into  $\alpha_1(\xi)$ and  $\alpha_2(\xi, \Delta\omega)$  isolates the frequency-dependence of  $\alpha(\xi, \Delta\omega)$  to  $\alpha_2(\xi, \Delta\omega)$ . The advantage of doing so is clear if we consider the scenario of free space propagation, for which  $\delta k(\Delta\omega) = 0$ : recalling  $b(\Delta\omega) = \delta k(\Delta\omega)$ ,  $\delta k(\Delta\omega) = 0$  causes  $\alpha_2(\xi, \Delta\omega)$  to be constant, allowing it to be incorporated into  $C_1$ . Therefore,  $\alpha_1(\xi)$  should be the solution for  $\alpha(\xi, \Delta\omega)$  when considering free-space propagation of the TEM<sub>00</sub> mode, with  $\alpha_2(\xi, \Delta\omega)$ providing the correction necessary to account for the dispersion introduced by a medium.

Within the treatment of  $\alpha(\xi, \Delta\omega)$  as the sum of  $\alpha_1(\xi)$  and  $\alpha_2(\xi, \Delta\omega)$ , we can address  $\alpha_1(\xi)$  and  $\alpha_2(\xi, \Delta\omega)$ separately. Starting with  $\alpha_1(\xi)$ , and recalling our assignment of the temporary variable  $a = \xi_0$ , we can show the formulation in Eq. 3.4.70a is identical to:

$$\alpha_1(\xi) = -\ln\left(1 - i\frac{\xi}{\xi_0}\right) - \ln(\xi_0) + C_1 \tag{3.4.71}$$

As  $\ln(\xi_0)$  will just have a constant value, we can incorporate it into the constant,  $C_1$ , leaving:

$$\alpha_1(\xi) = -\ln\left(1 - i\frac{\xi}{\xi_0}\right) + C_1 \tag{3.4.72}$$

We must then determine the appropriate value of  $C_1$ , first recalling that we have taken  $\alpha_1(\xi)$  to describe the behavior of  $\alpha(\xi, \Delta \omega)$  for free-space propagation of the TEM<sub>00</sub> mode. Previously, we indicated that in

<sup>&</sup>lt;sup>i</sup>The general differential equation was evaluated with the calculator at wolframalpha.com; when evaluating, it is sufficient to consider  $b(\Delta\omega)$  as a constant as the differential equation is associated with the derivative with respect to  $\xi$ .

the absence of dispersion, or for free-space propagation, we expect a form consistent with  $\exp(-r^2/w_0^2)$  at  $\xi = 0$ . This necessitates  $\alpha_1(\xi = 0) = 0$ . Given the general solution of Eq. 3.4.72,  $\alpha_1(\xi = 0) = -\ln(1) + C_1$ . However,  $\ln(1) = 0$ , so  $\alpha_1(\xi = 0) = C_1$ , requiring  $C_1 = 0$  in order to satisfy our assignment of  $\alpha_1(\xi = 0) = 0$ . The solution for  $\alpha_1(\xi)$  must then be:

$$\alpha_1(\xi) = -\ln\left(1 - i\frac{\xi}{\xi_0}\right) \tag{3.4.73}$$

However, we can simplify the solution in Eq. 3.4.73 by considering  $(1 - i\xi/\xi_0)$  as a value in the complex plane: namely, as the product of its magnitude and a phase factor,  $\exp(i\phi)$ , where  $\phi = \arctan \Im\{1 - i\xi/\xi_0\}/\Re\{1 - i\xi/\xi_0\}$ . In this manner, we can find  $(1 - i\xi/\xi_0)$  to be equivalent to:<sup>i</sup>

$$\left(1 - i\frac{\xi}{\xi_0}\right) = \sqrt{1 + \left(\frac{\xi}{\xi_0}\right)^2} e^{-i\arctan(\xi/\xi_0)}$$
(3.4.74)

Substitution for  $(1 - i\xi/\xi_0)$  according to Eq. 3.4.74 into Eq. 3.4.73 then yields:

$$\alpha_1(\xi) = -\ln\sqrt{1 + \left(\frac{\xi}{\xi_0}\right)^2} + i\arctan\left(\frac{\xi}{\xi_0}\right)$$
(3.4.75)

Continuing with  $\alpha_2(\xi, \Delta \omega)$ , we can substitute according to our assignments of  $a = \xi_0$  and  $b(\Delta \omega) = \delta k(\Delta \omega)$ and distribute the term corresponding to  $-ib(\Delta \omega)$  to yield:

$$\alpha_2(\xi, \Delta\omega) = -i\xi_0 \delta k(\Delta\omega) - \delta k(\Delta\omega)\xi + C_2 \tag{3.4.76}$$

In the same manner that we incorporated  $\ln(\xi_0)$  into  $C_1$  in Eq. 3.4.72, we can consider  $-i\xi_0\delta k(\Delta\omega)$  as a constant relative to mode propagation and combine it with  $C_2$  to yield the simplified result:

$$\alpha_2(\xi, \Delta\omega) = -\delta k(\Delta\omega)\xi + C_2 \tag{3.4.77}$$

From this simplified general solution for  $\alpha_2(\xi, \Delta\omega)$ , the last remaining step is to determine the exact solution by evaluating  $C_2$ . In contrast to  $\beta(\xi)$  and  $\alpha_1(\xi)$ , however, for  $\alpha_2(\xi, \Delta\omega)$ , it will not prove convenient to consider the behavior at  $\xi = 0$ ; we instead must consider what  $\delta k(\Delta\omega)$  represents. In Eq. 3.4.48, we defined  $\delta k(\Delta\omega)$  to represent the various contributions of the medium's dispersion upon the propagation of the field, and we expect these effects to scale with the propagation distance. Recalling that we have defined the point

<sup>&</sup>lt;sup>i</sup>Note that tan is an odd function  $(\tan(-\theta) = -\tan(\theta))$ .[116] From this, it is possible to show  $\arctan(-\xi/\xi_0) = -\arctan(\xi/\xi_0)$ , i.e. that arctan must also be odd.

 $\xi = \xi'$  to represent the point at which the field enters the medium, we therefore consider that at  $\xi = \xi'$ , we have accumulated no effects of dispersion, and at all other points within the medium, we accumulate the effects of dispersion according to  $\xi - \xi'$ . Therefore, we will need to define  $C_2$  such that  $\alpha_2(\xi, \Delta \omega)$  assumes the form:

$$\alpha_2(\xi, \Delta\omega) = -\delta k(\Delta\omega)(\xi - \xi') \tag{3.4.78}$$

Having obtained solutions for  $\alpha_1(\xi)$ ,  $\alpha_2(\xi, \Delta \omega)$ , and  $\beta(\xi)$ , we can substitute these results into our trial solution (Eq. 3.4.53), yielding:

$$\psi(r,\xi,\Delta\omega) = \frac{1}{\sqrt{1 + (\xi/\xi_0)^2}} e^{i\arctan\xi/\xi_0} e^{-\delta k(\Delta\omega)(\xi-\xi')} e^{-r^2/w^2(\xi)} e^{-ik_0/[2R(\xi)]r^2}$$
(3.4.79)

Incorporating the definition of  $\delta k(\Delta \omega)$  (Eq. 3.4.48), then substituting this solution for  $\psi(r, \xi, \Delta \omega)$  into our original frequency-domain model for the field (Eq. 3.4.39b), we arrive at a result very consistent with the results of Svelto and Verdeyen:[104, 105]

$$E(r,\xi,\Delta\omega) = \hat{A}(\Delta\omega) \frac{1}{\sqrt{1 + (\xi/\xi_0)^2}} \exp\left[-\frac{r^2}{w^2(\xi)}\right]$$
(3.4.80a)

$$\times \exp(-ik_0\xi) \exp\left(-i\frac{\Delta\omega}{v}\xi\right) \exp\left(i\arctan\frac{\xi}{\xi_0}\right)$$
(3.4.80b)

$$\times \exp\left(-i\frac{k_0}{2R(\xi)}r^2\right) \tag{3.4.80c}$$

$$\times \exp\left[-i(k_{1}-k_{1r})\Delta\omega(\xi-\xi')-i\frac{1}{2}k_{2}\Delta\omega^{2}(\xi-\xi')\right]$$
(3.4.80d)

$$\times \exp\left[-\frac{1}{2}\alpha(\omega)\left(\xi-\xi'\right)\right] \tag{3.4.80e}$$

The presentation of the field across Eq. 3.4.80a - 3.4.80e allows us to separately consider the various contributions to the field in the manner of Verdeyen.[105] Eq. 3.4.80a contains the three terms that combine to describe the amplitude of the field independent of the medium (i.e. neglecting absorptive effects) and as a function of the spatial coordinates (r or  $\xi$ ) or of detuning ( $\Delta\omega$ ). Eq. 3.4.80b encompasses the contributions to the longitudinal phase of the field; these contributions may arise from either the spatial oscillations of the field,  $\exp(-ik_0\xi)$  and  $\exp[-i(\Delta\omega/v)\xi]$ , or from the Guoy phase,  $\exp[i \arctan(\xi/\xi_0)]$ . Eq. 3.4.80c describes the phase accumulated by the field due to the radius of curvature of the phase front. Finally, Eq. 3.4.80d and Eq. 3.4.80e provide the corrections necessary to account for the effects of a dispersive medium, introducing the changes arising from dispersion and absorption, respectively.

## 3.4.4 Integration of the Nonlinear Polarization Outside the Plane Wave Limit

In this section, we will discuss the adaptation of the method presented by Moosmüller et al[202] for integration of the polarization induced by Gaussian beams, extending the approach to ultrashort pulses and crossed beam geometries. A variety of authors have addressed the integration of the nonlinear polarization induced by Gaussian beams in the literature.[139, 202–205] Some of these authors emphasize the importance of incorporating the effects of the Guoy phase,[139, 203] particularly in experiments that do not involve oppositely phase interactions of the same frequency. When two oppositely phase interactions have the same frequency, the Guoy phases of the interactions cancel; otherwise, the Guoy phase introduces additional interference effects in the same manner as phase mismatch. However, many of these authors focus on collinear experiments;[139, 202, 203] those that consider crossed beams appear to make approximations[204] or adopt similar methods that are less straightforward or less thoroughly documented than the method presented by Moosmüller et al.[205]

In a manner similar to § 3.4.3, we will allow the electric field to exhibit a transverse dependence. However, in contrast to the treatment in § 3.4.3, we cannot expect a closed-form solution for the transverse dependence; therefore, we will consider it sufficient to treat the field in the same manner as Eq. 3.1.27, with the envelope functions explicitly exhibiting dependence upon the spatial coordinates:

$$\tilde{E}(\mathbf{r},z,t) = A(\mathbf{r},z,t)e^{-ik_0 z}e^{i\omega_0 t}$$
(3.4.81a)

$$E(\mathbf{r}, z, \omega) = \hat{A}(\mathbf{r}, z, \Delta\omega) e^{-i(\Delta\omega/v)z} e^{-ik_0 z}$$
(3.4.81b)

In Eq. 3.4.81a and 3.4.81b, we return to our notation of  $\mathbf{r}$  to generalize the transverse coordinates (x, y). We make a similar assumption as in § 3.4.3, taking the relationship between the time- and frequency-domain models to be similar to Eq. 3.1.26, namely  $\mathcal{F}\{A(\mathbf{r}, z, t) \exp(i\omega_0 t)\} = \hat{A}(\mathbf{r}, z, \Delta\omega) \exp[-i(\Delta\omega/v)z]$  when the Fourier transform is with respect to  $\omega$ . In contrast to § 3.4.3, however, we will consider the field described by Eq. 3.4.81a or 3.4.81b as it couples to a nonlinear polarization oscillating with central frequency  $\omega_0$ . In the same manner as Eq. 3.2.17 and 3.2.18, and in analogy with Eq. 3.4.81a and 3.4.81b, we will consider this nonlinear polarization to be described:

$$\tilde{P}^{\mathrm{NL}}(\mathbf{r}, z, t) = p^{\mathrm{NL}}(\mathbf{r}, z, t)e^{i\omega_0 t}$$
(3.4.82a)

$$P^{\rm NL}(\mathbf{r}, z, \omega) = \hat{p}^{\rm NL}(\mathbf{r}, z, \Delta\omega) \tag{3.4.82b}$$

In the same manner as the field exhibits dependence upon both the transverse and longitudinal coordinates, the polarization is also allowed to do so through the envelope function,  $p^{\text{NL}}(\mathbf{r}, z, t)$  or  $\hat{p}^{\text{NL}}(\mathbf{r}, z, \Delta \omega)$ . Whereas the spatial dependence of the field will be assumed to arise from the spatial dependence of the polarization - which is to assume the field is created from the initiated nonlinear polarization - the polarization will be assumed to exhibit spatial dependence due to the manner of excitation.

We will begin our treatment in a very similar manner to § 3.4.3, starting from Eq. 3.3.6 and considering two contributions to the polarization: a linear polarization,  $\tilde{P}^{(1)}(\mathbf{r}, z, t)$ , and a nonlinear polarization,  $\tilde{P}^{\mathrm{NL}}(\mathbf{r}, z, t)$ . However, whereas we were satisfied considering  $\tilde{P}^{\mathrm{NL}}(\mathbf{r}, z, t) = 0$  in § 3.4.3, we will now consider there to be a previously induced nonlinear polarization existing within the medium, described according to Eq. 3.4.82a. From these assumptions, we can arrive at a result very similar to Eq. 3.4.40:

$$\frac{\partial}{\partial z^2}\tilde{E}(\mathbf{r},z,t) + \nabla_{\mathrm{T}}^2\tilde{E}(\mathbf{r},z,t) - b^2 e \frac{\partial^2}{\partial t^2}\tilde{D}^{(1)}(\mathbf{r},z,t) = ab^2 e \frac{\partial^2}{\partial t^2}\tilde{P}^{\mathrm{NL}}(\mathbf{r},z,t)$$
(3.4.83)

where we have again combined the derivatives of the field and of the linear polarization with respect to time according to Eq. 3.4.3 to yield the linear displacement. Evaluating the derivative with respect to z yields:

$$e^{-ik_0 z} \left[ -k_0^2 - i2k_0 \frac{\partial}{\partial z} + \frac{\partial^2}{\partial z^2} \right] A(\mathbf{r}, z, t) e^{i\omega_0 t} + \nabla_{\mathrm{T}}^2 \tilde{E}(\mathbf{r}, z, t) - b^2 e \frac{\partial^2}{\partial t^2} \tilde{D}^{(1)}(\mathbf{r}, z, t) = ab^2 e \frac{\partial^2}{\partial t^2} \tilde{P}^{\mathrm{NL}}(\mathbf{r}, z, t)$$
(3.4.84)

Again following a similar strategy to § 3.4.3, we will Fourier transform Eq. 3.4.84 to yield the frequencydomain equivalent. We will proceed to substitute for the resulting frequency-domain terms  $E(\mathbf{r}, z, \omega)$  and  $P^{\text{NL}}(\mathbf{r}, z, \omega)$  according to Eq. 3.4.81b and 3.4.82b, though for convenience we will currently make the approximation that the term  $\exp[-i(\Delta \omega/v)z]$  in Eq. 3.4.81b is negligible as in § 3.4.1; as before, we will be able to relax this assumption later. Finally, we will multiply through the result by  $\exp(ik_0z)$ . These changes yield:

$$\left[\frac{\partial^2}{\partial z^2} - i2k_0\frac{\partial}{\partial z} - k_0^2 + e^{ik_0z}\nabla_{\mathrm{T}}^2 e^{-ik_0z} + b^2e\omega^2\epsilon(\omega)\right]\hat{A}(\mathbf{r}, z, \Delta\omega) = -ab^2e\omega^2e^{ik_0z}\hat{p}^{\mathrm{NL}}(\mathbf{r}, z, \Delta\omega) \qquad (3.4.85)$$

From Eq. 3.4.85, we can apply a handful of simplifications that have previously been introduced. First, since the transverse Laplacian,  $\nabla_{\rm T}^2$ , is only associated with the transverse coordinates,  $\nabla_{\rm T}^2 \exp(-ik_0 z) = \exp(-ik_0 z)\nabla_{\rm T}^2$ ; this identity allows us to remove the exponentials from this term in Eq. 3.4.85. Second, Eq. 3.4.9, 3.4.10, and 3.4.18 have shown we can introduce  $k^2(\omega) - ik(\omega)\alpha(\omega)$  as a substitute for  $b^2 e \omega^2 \epsilon(\omega)$ ; we can combine this result with the term  $k_0^2$  in Eq. 3.4.85, and use the expansion of Eq. 3.4.24 to yield  $2k_0k_1\Delta\omega + (k_0k_2 + k_1^2)\Delta\omega^2 - ik_0\alpha(\omega)$ . In § 3.4.1, we also showed that, by introducing  $\hat{p}^{\rm NL}(\mathbf{r}, z, \Delta\omega)$ ,<sup>i</sup> we can substitute for  $ab^2e\omega^2$ , yielding  $\omega^2/c^2$  in both unit systems. Finally, we will also make the slowly varying amplitude approximation, assuming  $2k_0\partial/\partial z \gg \partial^2/\partial z^2$ . The result of applying these substitutions and approximations is:

$$\left[-i2k_0\frac{\partial}{\partial z} + \nabla_{\mathrm{T}}^2 + 2k_0k_1\Delta\omega + (k_0k_2 + k_1^2)\Delta\omega^2 - ik_0\alpha(\omega)\right]\hat{A}(\mathbf{r}, z, \Delta\omega) = -\frac{\omega^2}{c^2}e^{ik_0z}\hat{p}^{\mathrm{NL}}(\mathbf{r}, z, \Delta\omega) \quad (3.4.86)$$

As in § 3.4.1 and § 3.4.3, we will want to transfer Eq. 3.4.86 to the moving frame as defined by Eq. 3.4.27a and 3.4.27b. Performing the inverse Fourier transform to bring Eq. 3.4.86 back to the time domain yields:

$$\begin{bmatrix} -i2k_0 \left( \frac{\partial}{\partial z} + k_1 \frac{\partial}{\partial t} \right) + \nabla_{\mathrm{T}}^2 - (k_0 k_2 + k_1^2) \frac{\partial^2}{\partial t^2} \end{bmatrix} A(\mathbf{r}, z, t) - \mathcal{F}^{-1} \{ ik_0 \alpha(\omega) \hat{A}(\mathbf{r}, z, \Delta \omega) \} = \frac{\omega^2}{c^2 \Delta \omega^2} e^{ik_0 z} \frac{\partial^2}{\partial t^2} \bar{p}^{\mathrm{NL}}(\mathbf{r}, z, t)$$
(3.4.87)

With Eq. 3.4.87 in the time domain, we can substitute according to Eq. 3.4.29a and 3.4.29b to yield:

$$\begin{bmatrix} -i2k_0 \left( \frac{\partial}{\partial \xi} + (k_1 - k_{1r}) \frac{\partial}{\partial \eta} \right) + \nabla_{\mathrm{T}}^2 - (k_0 k_2 + k_1^2) \frac{\partial^2}{\partial \eta^2} \end{bmatrix} A(\mathbf{r}, \xi, \eta) - \mathcal{F}^{-1} \{ ik_0 \alpha(\omega) \hat{A}(\mathbf{r}, \xi, \Delta \omega) \} = \frac{\omega^2}{c^2 \Delta \omega^2} e^{ik_0 \xi} \frac{\partial^2}{\partial \eta^2} \bar{p}^{\mathrm{NL}}(\mathbf{r}, \xi, \eta)$$
(3.4.88)

Once again, now that Eq. 3.4.88 represents propagation within the moving frame, we will return to the frequency domain; the Fourier transform of Eq. 3.4.88 yields:

$$\begin{bmatrix} -i2k_0 \left( \frac{\partial}{\partial \xi} + i(k_1 - k_{1r})\Delta \omega \right) + \nabla_{\mathrm{T}}^2 + (k_0k_2 + k_1^2)\Delta \omega^2 - ik_0\alpha(\omega) \end{bmatrix} \times \hat{A}(\mathbf{r}, \xi, \Delta \omega) = -\frac{\omega^2}{c^2} e^{ik_0\xi} \hat{p}^{\mathrm{NL}}(\mathbf{r}, \xi, \Delta \omega)$$
(3.4.89)

We can simplify the result of Eq. 3.4.89 slightly by dividing the equation by  $-i2k_0$ . As before, we will assume the resulting term  $k_1^2/k_0$  is sufficiently small so as to be negligible. This leaves:

$$\left[\frac{\partial}{\partial\xi} + i(k_1 - k_{1r})\Delta\omega + i\frac{1}{2k_0}\nabla_{\mathrm{T}}^2 + i\frac{1}{2}k_2\Delta\omega^2 + \frac{1}{2}\alpha(\omega)\right]\hat{A}(\mathbf{r},\xi,\Delta\omega) = -i\frac{\omega^2}{2c^2k_0}e^{ik_0\xi}\hat{p}^{\mathrm{NL}}(\mathbf{r},\xi,\Delta\omega) \quad (3.4.90)$$

Note that at this point, we can adopt the approach of § 3.4.1 and relax the assumption that  $\exp[-i(\Delta\omega/v)\xi]$  is negligible; if we again allow  $\Delta\omega$  to be sufficiently small that the expansion of  $\partial/\partial\xi \hat{A}(\mathbf{r},\xi,\Delta\omega)\exp[-i(\Delta\omega/v)\xi]$ 

<sup>&</sup>lt;sup>i</sup>Recall that the quantity  $\bar{P}^{\rm NL}(z,\omega)$  was defined  $\bar{P}^{\rm NL} = [P^{\rm NL}(z,\omega)]/\epsilon_0$  for mks units and  $\bar{P}^{\rm NL} = 4\pi P^{\rm NL}(z,\omega)$  for cgs or esu units. Here, we take  $\hat{p}^{\rm NL}(\mathbf{r}, z\Delta\omega)$  in the same manner, only including the transverse dependence neglected in the plane wave limit of § 3.4.1.

is approximately  $\exp[-i(\Delta\omega/v)\xi]\partial/\partial\xi \hat{A}(\mathbf{r},\xi,\Delta\omega)$ , we can divide through the result by the exponential. Recombination of  $\exp[i(\Delta\omega/v)\xi]$  with  $\exp(ik_0\xi)$  yields  $\exp[ik(\omega)\xi]$  on the right hand side. We will apply this consideration going forward. We will also take the opportunity to introduce two variables to simplify the form of Eq. 3.4.90. Similar to § 3.4.3, we will introduce the variable  $\delta k(\Delta\omega)$ :

$$\delta k(\Delta \omega) = i(k_1 - k_{1r})\Delta \omega + i\frac{1}{2}k_2\Delta \omega^2 + \frac{1}{2}\alpha(\omega)$$
(3.4.91)

Second, in a manner similar to Moosmüller et al,[202] we will introduce a function representing the right hand side of Eq. 3.4.90:<sup>i</sup>

$$\rho(\mathbf{r},\xi,\Delta\omega) = -i\frac{\omega^2}{2c^2k_0}e^{ik(\omega)\xi}\hat{p}^{\mathrm{NL}}(\mathbf{r},\xi,\Delta\omega)$$
(3.4.92)

With these variables, Eq. 3.4.90 becomes:

$$\left[\frac{\partial}{\partial\xi} + i\frac{1}{2k_0}\nabla_{\mathrm{T}}^2 + \delta k(\Delta\omega)\right]\hat{A}(\mathbf{r},\xi,\Delta\omega) = \rho(\mathbf{r},\xi,\Delta\omega)$$
(3.4.93)

From Eq. 3.4.93, we will begin applying the approach outlined by Moosmüller et al.[202] Their method is predicated upon performing a two-dimensional Fourier transform with respect to the transverse coordinate, masking the transverse propagation of the field by converting from the spatial domain to the spatial frequency domain. For example, if we consider the generalized transverse coordinate  $\mathbf{r}$  to correspond to a (x, y)coordinate system, this two-dimensional Fourier transform takes  $(x, y) \rightarrow (k_x, k_y)$ , where  $k_x$  and  $k_y$  are the spatial frequencies associated with x and y, respectively. We will generalize this transform as  $\mathbf{r} \rightarrow \mathbf{k}_{\mathrm{T}}$ . However, given Eq. 3.4.93, we can see that we will need to understand how the transverse Laplacian,  $\nabla_{\mathrm{T}}^2$ , transforms. By considering our definition for this operator (Eq. 3.3.5) and its application to an arbitrary function,  $f(\mathbf{r})$ , we can use the appropriate Fourier relation<sup>ii</sup> (see Eq. 3.4.6b) to determine the transform of

<sup>&</sup>lt;sup>i</sup>Moosmüller et al notate this function as f; as we have consistently used f throughout this chapter as an arbitrary, representative function, we have chosen to use a different notation in this capacity ( $\rho$ ). Additionally, in a manner similar to how we previously assumed it was sufficient to consider  $\delta k$  in § 3.4.3 as a function of just  $\Delta \omega$  while it contained both terms dependent upon  $\omega$  and upon  $\Delta \omega$ , we will consider it sufficient to treat  $\rho$  in Eq. 3.4.92 as a function of  $\Delta \omega$ .

<sup>&</sup>lt;sup>ii</sup>It should be noted that the form of Eq. 3.4.6b is specific for Fourier transformation to the angular frequency domain.[112] When the Fourier transform is being taken to the linear frequency domain, the transform is instead  $\mathcal{F}\{\partial^n/\partial t^n f(t)\} = (i2\pi\nu)^n \hat{f}(\nu)$ , where  $\nu$  is the linear frequency.[114] We note here that we are using this relationship because we will be taking the generalized frequency  $\mathbf{k}_T$  to be a linear spatial frequency. Unfortunately, Moosmüller et al do not explicitly indicate whether they take their version of the generalized transverse spatial frequency to be angular or linear,[202] but by omitting the factor of  $(2\pi)^2$  that appears in Eq. 3.4.94, it appears to have been their intention to consider the frequency to be angular.

 $\nabla^2_{\mathrm{T}}$ :

$$\mathcal{F}\left\{\nabla_{\mathrm{T}}^{2}f(\mathbf{r})\right\} = \mathcal{F}\left\{\left(\frac{\partial^{2}}{\partial x^{2}} + \frac{\partial^{2}}{\partial y^{2}}\right)f(\mathbf{r})\right\}$$
$$= \left[(i2\pi)^{2}k_{x}^{2} + (i2\pi)^{2}k_{y}^{2}\right]\hat{f}(\mathbf{k}_{\mathrm{T}})$$
$$= -4\pi^{2}|\mathbf{k}_{\mathrm{T}}|^{2}\hat{f}(\mathbf{k}_{\mathrm{T}})$$
(3.4.94)

where we have used the relationship,  $|\mathbf{k}_{T}|^{2} = k_{x}^{2} + k_{y}^{2}$ , appropriate for  $\mathbf{k}_{T}$  providing generalization of the  $(k_{x}, k_{y})$  coordinates.

Using the result of Eq. 3.4.94, the two-dimensional Fourier transform of Eq. 3.4.93 with respect to the transverse coordinates yields:

$$\left[\frac{\partial}{\partial\xi} - i\frac{2\pi^2}{k_0}|\mathbf{k}_{\rm T}|^2 + \delta k(\Delta\omega)\right]\hat{A}(\mathbf{k}_{\rm T},\xi,\Delta\omega) = \rho(\mathbf{k}_{\rm T},\xi,\Delta\omega) \tag{3.4.95}$$

The next step is to define a function,  $\hat{a}(\mathbf{k}_{\mathrm{T}}, \xi, \Delta \omega)$ :

$$\hat{a}(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega) = \hat{A}(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega) \exp\left[\left(-i\frac{2\pi^{2}}{k_{0}}|\mathbf{k}_{\mathrm{T}}|^{2} + \delta k(\Delta\omega)\right)\xi\right]$$
(3.4.96)

The advantage of defining  $\hat{a}(\mathbf{k}_{\mathrm{T}}, \xi, \Delta \omega)$  is in recognizing that the derivative of  $\hat{a}(\mathbf{k}_{\mathrm{T}}, \xi, \Delta \omega)$  with respect to  $\xi$  yields:

$$\begin{aligned} \frac{\partial}{\partial\xi} \hat{a}(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega) &= \frac{\partial}{\partial\xi} \left[ \hat{A}(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega) \exp\left[ \left( -i\frac{2\pi^{2}}{k_{0}} |\mathbf{k}_{\mathrm{T}}|^{2} + \delta k(\Delta\omega) \right) \xi \right] \right] \\ &= \exp\left[ \left( -i\frac{2\pi^{2}}{k_{0}} |\mathbf{k}_{\mathrm{T}}|^{2} + \delta k(\Delta\omega) \right) \xi \right] \frac{\partial}{\partial\xi} \hat{A}(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega) + \hat{A}(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega) \\ &\qquad \times \exp\left[ \left( -i\frac{2\pi^{2}}{k_{0}} |\mathbf{k}_{\mathrm{T}}|^{2} + \delta k(\Delta\omega) \right) \xi \right] \frac{\partial}{\partial\xi} \left[ \left( -i\frac{2\pi^{2}}{k_{0}} |\mathbf{k}_{\mathrm{T}}|^{2} + \delta k(\Delta\omega) \right) \xi \right] \\ &= \exp\left[ \left( -i\frac{2\pi^{2}}{k_{0}} |\mathbf{k}_{\mathrm{T}}|^{2} + \delta k(\Delta\omega) \right) \xi \right] \\ &\qquad \times \left\{ \left[ \frac{\partial}{\partial\xi} - i\frac{2\pi^{2}}{k_{0}} |\mathbf{k}_{\mathrm{T}}|^{2} + \partial k(\Delta\omega) \right] \hat{A}(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega) \right\} \end{aligned}$$

If we examine the term in curly brackets in the last line of the scheme in Eq. 3.4.97, we see that this quantity is identical to the left hand side of Eq. 3.4.95. Therefore, we can substitute according to Eq. 3.4.95 to relate  $\hat{a}(\mathbf{k}_{\mathrm{T}}, \xi, \Delta \omega)$  to the polarization function,  $\rho(\mathbf{k}_{\mathrm{T}}, \xi, \Delta \omega)$ :

$$\frac{\partial}{\partial\xi}\hat{a}(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega) = \exp\left[\left(-i\frac{2\pi^2}{k_0}|\mathbf{k}_{\mathrm{T}}|^2 + \delta k(\Delta\omega)\right)\xi\right]\rho(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega)$$
(3.4.98)

Integrating both sides of Eq. 3.4.98 from the beginning of the sample,  $\xi'$ , to the end of the sample,  $\xi_{end}$ ,

yields:

$$\hat{a}(\mathbf{k}_{\mathrm{T}},\xi_{\mathrm{end}},\Delta\omega) - \hat{a}(\mathbf{k}_{\mathrm{T}},\xi',\Delta\omega) = \int_{\xi'}^{\xi_{\mathrm{end}}} \exp\left[\left(-i\frac{2\pi^2}{k_0}|\mathbf{k}_{\mathrm{T}}|^2 + \delta k(\Delta\omega)\right)\xi\right]\rho(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega)d\xi \qquad (3.4.99)$$

Recalling we have defined  $\hat{a}(\mathbf{k}_{\mathrm{T}}, \xi, \Delta \omega)$  as proportional to the field envelope  $\hat{A}(\mathbf{k}_{\mathrm{T}}, \xi, \Delta \omega)$  (see Eq. 3.4.96), Eq. 3.4.99 describes the regime in which we are concerned with the changes induced in an existing field by the polarization. The opposite limit is the scenario in which the polarization creates the field, in which case  $\hat{a}(\mathbf{k}_{\mathrm{T}}, \xi', \Delta \omega) = 0$ . This scenario is described:

$$\hat{a}(\mathbf{k}_{\mathrm{T}},\xi_{\mathrm{end}},\Delta\omega) = \int_{\xi'}^{\xi_{\mathrm{end}}} \exp\left[\left(-i\frac{2\pi^2}{k_0}|\mathbf{k}_{\mathrm{T}}|^2 + \delta k(\Delta\omega)\right)\xi\right]\rho(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega)d\xi$$
(3.4.100)

We will continue forward considering the regime described by Eq. 3.4.100.

However, we will again recall that  $\hat{a}(\mathbf{k}_{\mathrm{T}}, \xi, \Delta \omega)$  is only proportional to the field amplitude envelope,  $\hat{A}(\mathbf{k}_{\mathrm{T}}, \xi, \Delta \omega)$ , which is the quantity with which we are actually concerned. Using Eq. 3.4.96, we can relate the integral over the polarization in Eq. 3.4.100 back to the field envelope at  $\xi = \xi_{\mathrm{end}}$ :

$$\hat{A}(\mathbf{k}_{\mathrm{T}},\xi_{\mathrm{end}},\Delta\omega) = \int_{\xi'}^{\xi_{\mathrm{end}}} \exp\left[\left(i\frac{2\pi^2}{k_0}|\mathbf{k}_{\mathrm{T}}|^2 - \delta k(\Delta\omega)\right)(\xi_{\mathrm{end}} - \xi)\right]\rho(\mathbf{k}_{\mathrm{T}},\xi,\Delta\omega)d\xi$$
(3.4.101)

To obtain the form presented in Eq. 3.4.101, both sides have been divided by the exponential at  $\xi_{end}$  arising from the definition of  $\hat{a}(\mathbf{k}_{T}, \xi, \Delta \omega)$  (see Eq. 3.4.96); this term has been incorporated into the integral, yielding the difference between  $\xi_{end}$  and  $\xi$ . Eq. 3.4.101 can be considered to attribute the final field envelope as the summation of "slices" of  $\rho(\mathbf{k}_{T}, \xi, \Delta \omega)$  along  $\xi$ , incorporating the phase effects associated with the transverse frequency and material dispersion based upon the distance that "slice" has needed to propagate through the material.

### 3.5 Group Velocity, Group Velocity Mismatch, and Group Velocity Dispersion

In § 3.4.1, we presented the expansion of  $k(\omega)$  as a Taylor series in Eq. 3.4.23. From this series and the resulting expansion of  $k^2(\omega)$  (see Eq. 3.4.24), our retention of terms with  $\Delta \omega^n$ ,  $n \leq 2$  left the first and second derivatives of  $k(\omega)$  with respect to  $\omega$  remaining. While there are circumstances in which the thirdorder term can become important,[206, 207] it is generally sufficient to terminate the expansion after the second-order term.[103] These remaining terms correspond to the group velocity,  $u_g = 1/k_1$ , and the group velocity dispersion,  $k_2$ . In this section, we will consider the effects of these two quantities upon ultrafast pulse propagation and present formulae that allow the calculation of these quantities under various circumstances.

# 3.5.1 Effects of Group Velocity and Group Velocity Dispersion

We will begin considering the effects of group velocity and group velocity dispersion through the lens of § 3.4.1. In § 3.4.1, we derived Eq. 3.4.34, which describes the changes in the envelope function of a plane wave as it is influenced by dispersion or absorption by the medium (the term in brackets) or by an induced nonlinear polarization within the medium. If we ignore absorption ( $\alpha(\omega) = 0$ ) and assume no polarizations have been induced ( $\hat{p}^{\text{NL}} = 0$ ), Eq. 3.4.34 reduces to:

$$\frac{\partial}{\partial\xi}\hat{A}(\xi,\Delta\omega) = \left[-i(k_1 - k_{1r})\Delta\omega - i\frac{1}{2}k_2\Delta\omega^2\right]\hat{A}(\xi,\Delta\omega)$$
(3.5.1)

As Eq. 3.5.1 is a differential equation, it would not be surprising to find an exponential to serve as a valid solution; taking the trial solution to be  $\hat{A}(\xi, \Delta \omega) = \exp(\alpha \xi)$ , it is trivial to show that Eq. 3.5.1 yields a solution of the form:

$$\hat{A}(\xi, \Delta\omega) = \hat{A}(\xi = \xi', \Delta\omega) \exp\left[\left(-i(k_1 - k_{1r})\Delta\omega - i\frac{1}{2}k_2\Delta\omega^2\right)\xi\right]$$
(3.5.2)

where  $\xi'$  corresponds to some initial position in space.

Considering the form of Eq. 3.5.2, it is straightforward to gain insight into the effects of group velocity. In the absence of group velocity dispersion ( $k_2 = 0$ ), the general solution in Eq. 3.5.2 reduces to:

$$\hat{A}(\xi, \Delta\omega) = \hat{A}(\xi = \xi', \Delta\omega) \exp[-i(k_1 - k_{1r})\Delta\omega\xi]$$
(3.5.3)

From the form of Eq. 3.5.3, it is clear that propagation through the medium results in a frequency-dependent accumulation of phase governed by  $(k_1 - k_{1r})$ . However, the effects become more obvious in the time domain. If we consider taking the inverse Fourier transform of the result in Eq. 3.5.3, we can identify the following transform relationship:[112]

$$\mathcal{F}\{f(t\pm t_0)\} = \hat{f}(\omega)\exp(\pm i\omega t_0)$$
(3.5.4)

where f(t) and  $\hat{f}(\omega)$  are defined such that  $\mathcal{F}{f(t)} = \hat{f}(\omega)$  and  $t_0$  represents some delay in the time domain. Applying this relationship shows that the phase introduced by the exponential in the frequency domain is equivalent to the time-domain representation:

$$A(\xi, t) = A(\xi = \xi', t - (k_1 - k_{1r})\xi)$$
(3.5.5)

Therefore, the group velocity introduces a distance-dependent delay,  $(k_1 - k_{1r})\xi$ , relative to the pulse envelope at the initial position.

This behavior is exhibited in Fig. 3.2. In this figure, we show the effects of group velocity on the pulse envelope of a 35 fs pulse centered at 400 nm as a function of propagation distance, both on the intensity level - (A) and (D) - and on the real and imaginary components of the amplitude - (B), (C), (E), and (F). The consider how these effects manifest in both the time - (A), (B), and (C) - and frequency - (D), (E), and (F) - domains. (A) illustrates our conclusion from Eq. 3.5.5: we observe a linear change in the delay of the pulse envelope with increasing propagation distance. In Fig. (B) and (C), it is clear that this shift in delay does not influence the phase of the envelope: the envelope retains a purely real value at all distances. The frequency-domain profile, however, exhibits strong phase modulation, as shown in (E) and (F). This phase modulation does not create new frequencies: as shown in (D), the spectrum of the pulse is invariant.

Looking at the scale of the propagation distance in Fig. 3.2, though, it is clear that considering the group velocity referenced only to the laboratory frame is limiting: in these calculations, we observe the pulse envelope begin leaving the range of our time variable over only 50  $\mu$ m. This is the utility of considering the group velocity mismatch: the value of  $k_1$  for a given field relative to some other reference value. In the scenario of a single pulse, as in Fig. 3.2, it would be far more convenient to consider the pulse itself to be the reference, which would make the pulse appear to exhibit a constant center for all propagation distances. On the other hand, for the copropagation of multiple pulses, we might consider defining one to be the reference pulse, as is illustrated in Fig. 3.3.

In this figure, we consider the copropagation of two 35 fs pulses: one centered at 400 nm and the other at 800 nm. In (A), we display the total intensity of the two pulses: in this case, it is clear that we are observe the pulses beginning to split after approximately 100 µm with essentially complete resolution between the pulses after approximately 200 µm. This is in good agreement with the calculated pulse-splitting length of 107 µm ( $L_w = \tau/|\Delta u^{-1}|$ , where  $\tau$  is the pulse length and  $|\Delta u^{-1}|$  is the group velocity mismatch) for 35 fs pulses at these colors. Once again, however, we see that the group velocity mismatch does not impact the spectral content of the pulses, shown in (B) and (C).

The functional form for the effects of group velocity dispersion is much less straightforward to evaluate, as there is not a tabulated result for the Fourier transformation of the exponential,  $\exp(-ik_2\Delta\omega^2/2)$ . However,



Figure 3.2: Simulations showing the effects of group velocity on the propagation of a 35 fs pulse centered at 400 nm when polarized along the ordinary axis of  $\beta$ -barium borate ( $k_1 = 5942 \text{ fs/mm}[208]$ ). (A), (B), and (C) show the effects on the time-domain envelope, A(t): (A) shows the intensity of the field,  $|A(t)|^2$ ; (B) shows the amplitude of the real component,  $\Re\{A(t)\}$ ; and (C) shows the amplitude of the imaginary component,  $\Im\{A(t)\}$ . (D), (E), and (F) show the effects on the frequency-domain envelope,  $\hat{A}(\Delta\omega)$ : (D) shows the intensity of the field,  $|\hat{A}(\Delta\omega)|^2$ , (E) shows the amplitude of the real component,  $\Re\{\hat{A}(\Delta\omega)\}$ ; and, (F) shows the amplitude of the imaginary component,  $\Im\{\hat{A}(\Delta\omega)\}$ . In the complex representation of (B), (C), (E), and (F), white corresponds to zero amplitude, while red and blue represent positive and negative amplitudes, respectively.



Figure 3.3: Simulation showing the effects of group velocity mismatch on the co-propagation of two 35 fs pulses, each polarized along the ordinary axis of  $\beta$ -barium borate: one centered at 400 nm ( $k_1 = 5942$  fs/mm[208]) and the other centered at 800 nm ( $k_1 = 5615$  fs/mm[208]), corresponding to a group velocity mismatch of 327 fs/mm. The 800 nm pulse is taken as the reference. (A), (B), and (C) all report the intensity of the field: (A) shows  $|A_{400}(t)|^2 + |A_{800}(t)|^2$  in the time domain, where  $A_{400}(t)$  and  $A_{800}(t)$  are the temporal pulse envelopes of the 400 nm and 800 nm pulses, respectively. (B) and (C) show  $|\hat{A}(\Delta\omega)|^2$  for the 400 nm (B) and 800 nm (C) pulses.

it is still trivial to simulate its effects, as shown in Fig. 3.4. Once again considering a 35 fs pulse at 400 nm, we see that the effects of group velocity dispersion are a little more complicated. (E) and (F) illustrate that the group velocity dispersion influences the phase of the envelope in the frequency domain; given the dependence of the effects of group velocity dispersion upon  $\Delta \omega^2$ , it is also not surprising that the effects are symmetric about  $\Delta \omega = 0$ . Unlike group velocity, however, the phase applied in the frequency domain does manifest in the time domain, as seen in (B) and (C). This phase applied to the time domain envelope worsens the time-bandwidth product of the pulse, and we consequently see broadening of the pulse in the time domain in (A). However, this broadening is only a consequence of the phase added to the envelope: the frequency content of the pulse, shown in (D), exhibits no changes as a function of propagation distance.

## 3.5.2 Calculating Group Velocity and Group Velocity Dispersion

In order to calculate the group velocity and group velocity dispersion in an arbitrary medium, we must first recall the Taylor expansion in Eq. 3.4.23. In this expansion, we assigned the coefficients  $k_1$  and  $k_2$  as:

$$k_1 = \left. \frac{\partial k(\omega)}{\partial \omega} \right|_{\omega = \omega_0} \tag{3.5.6a}$$

$$k_2 = \left. \frac{\partial^2 k(\omega)}{\partial \omega^2} \right|_{\omega = \omega_0} \tag{3.5.6b}$$

With the definition we obtained for  $k(\omega)$  (Eq. 3.4.14), it is trivial to find the result of the derivative defining  $k_1$ , and only slightly more difficult to evaluate the derivative for  $k_2$ . One form for these derivatives is:

$$k_1 = \frac{1}{c} \left[ n(\omega_0) + \omega_0 \left. \frac{\partial n(\omega)}{\partial \omega} \right|_{\omega = \omega_0} \right]$$
(3.5.7a)

$$k_{2} = \frac{1}{c} \left[ 2 \frac{\partial n(\omega)}{\partial \omega} \bigg|_{\omega = \omega_{0}} + \omega_{0} \left. \frac{\partial^{2} n(\omega)}{\partial \omega^{2}} \right|_{\omega = \omega_{0}} \right]$$
(3.5.7b)

However, we describe Eq. 3.5.7a and 3.5.7b as one form of the derivatives, as it will generally be more convenient to consider the dispersion of materials as  $\partial n(\lambda)/\partial \lambda$ . The reason for this is that the dispersion of many materials is reported either in the form Cauchy's formula[101] or in the manner of the Sellmeier formula;[209] in both models, the index is considered to be a function of wavelength,  $\lambda$ , rather than frequency. However, we can recognize that wavelength and frequency are related: recalling Eq. 3.1.5 and recognizing  $\omega = 2\pi\nu$ ,  $\lambda\omega = 2\pi c$ . Therefore, we can use the chain rule to expand the derivatives in Eq. 3.5.7a and 3.5.7b



Figure 3.4: Simulations showing the effects of group velocity dispersion on the propagation of a 35 fs pulse centered at 400 nm when polarized along the ordinary axis of  $\beta$ -barium borate ( $k_2 = 215.17 \text{ fs}^2/\text{mm}[208]$ ). (A), (B), and (C) show the effects on the time-domain envelope, A(t): (A) shows the intensity of the field,  $|A(t)|^2$ ; (B) shows the amplitude of the real component,  $\Re\{A(t)\}$ ; and (C) shows the amplitude of the imaginary component,  $\Im\{A(t)\}$ . (D), (E), and (F) show the effects on the frequency-domain envelope,  $\hat{A}(\Delta\omega)$ : (D) shows the intensity of the field,  $|\hat{A}(\Delta\omega)|^2$ , (E) shows the amplitude of the real component,  $\Re\{\hat{A}(\Delta\omega)\}$ ; and, (F) shows the amplitude of the imaginary component,  $\Im\{\hat{A}(\Delta\omega)\}$ . In the complex representation of (B), (C), (E), and (F), white corresponds to zero amplitude, while red and blue represent positive and negative amplitudes, respectively.

to leave a wavelength-dependence:

$$k_1 = \frac{1}{c} \left[ n(\lambda_0) - \lambda_0 \left. \frac{\partial n(\lambda)}{\partial \lambda} \right|_{\lambda = \lambda_0} \right]$$
(3.5.8a)

$$k_2 = \frac{\lambda_0^3}{2\pi c^2} \left. \frac{\partial^2 n(\lambda)}{\partial \lambda^2} \right|_{\lambda = \lambda_0} \tag{3.5.8b}$$

With Eq. 3.5.8a and 3.5.8b, determining the group velocity or group velocity dispersion of a material requires only the derivatives of  $\partial^{n}(\lambda)/\partial\lambda$  or  $\partial^{2}n(\lambda)/\partial\lambda^{2}$ . These quantities can be determined from the appropriate dispersion curves, with several examples worked out in Appendix B.

## 3.6 Numerical Integration of Ordinary Differential Equations

The treatment of pulse propagation in § 3.4 focused upon the consideration of a single field, with that field coupling in some cases (§ 3.4.1 and § 3.4.4) to an induced nonlinear polarization. However, as we introduced in § 3.2, the induction of nonlinear polarizations can often involve the interaction of multiple pulses, thereby coupling the evolution and propagation of the field envelopes. In some circumstances, it is reasonable to approximate that the nonlinear interaction within the medium does not significantly deplete some of the fields, and that it is reasonable to consider the changes induced in a single field; under such circumstances, we are left with a single differential equation. However, there are also many conditions under which this approximation is not appropriate, such as simulation of the parametric processes within optical parametric amplifiers or oscillators.[102, 103, 118, 199–201, 208, 210–217] In such scenarios, it is necessary to consider the propagation of the fields as a system of coupled differential equations.

Fortunately, algorithms exist for numerical integration of systems of both first-order and second-order differential equations. While it is straightforward to consider a method as straightforward as the Euler method, there are fortunately more accurate methods, such as the Runge-Kutta algorithm. In the following sections, a background of such methods is provided with significant influence from the works of Press et al;[113, 218] for more information and information on computational implementation, the reader is directed to these sources.

Most methods of numerical integration are designed to evaluate a set of first order ordinary differential equations, described generically:

$$\frac{d}{dx}y_i(x) = f_i\{x, y_1, \dots, y_N\}$$
(3.6.1)

where i spans the range 1, 2, ..., N and  $f_i$  is of a known functional form and describes the coupling between

the set of functions  $\{y_1, y_2, \ldots, y_N\}$ . However, while most methods are intended towards evaluation of firstorder differential equations, systems of higher-order ordinary differential equations can generally be reduced to an equivalent set of first-order differential equations that can be evaluated by these methods.

Systems of first-order differential equations can be reduced to two generic categories based upon the known information: initial-value and two-point boundary problems. In the former, the entire set of functions is defined at an initial point, and the goal is to project the evolution of the set of functions according to the system of differential equations. Meanwhile, two-point boundary problems involve knowledge of the values of the functions at more than one point. Two-point boundary problems are consequently more complicated to solve, and require more advanced algorithms. Fortunately, the evolution of electric fields - where the fields are known for some initial condition and the goal is to calculate their propagation - is an example of an initial-value problem, for which the methods of evaluation are simpler.

The most simplistic approach to numerical integration of initial-value problems is the Euler method. In this method, the appropriate derivative,  $\frac{dy}{dx}$  is removed from the limit of infinitely small step sizes, yielding the case described by discrete steps,  $\frac{\Delta y}{\Delta x}$ . The change in the function y(x) can therefore be projected from its value at some position x, y(x), to its value at some position  $x + \Delta x$ ,  $y(x + \Delta x)$ , according to Eq. 3.6.2:

$$y(x_{n+1}) = y(x_n) + \Delta x f\{x_n, y(x_n)\}$$
(3.6.2)

Unfortunately, in this simple form, numerical integration of a system of differential equations can be unreliable. However, the reliability of the method can be improved by increasing the amount of information used to guide the evaluation of the differential equation.

This improvement is the basis of Runge-Kutta methods, in which a series of steps identical to the Euler method are evaluated; the results are then used to reconstruct a Taylor series expansion. These expansions can be evaluated to any arbitrary order, but the most common is the fourth-order Runge-Kutta method, which is outlined below. Runge-Kutta methods are not always the most efficient method for evaluating a system of differential equations, but the advantages associated with these methods are reliability, reasonable accuracy, and simple implementation. Two more efficient approaches to numerical integration - Richardson extrapolation, such as Bulirsch-Stoer methods, and predictor-corrector methods - are available, but both can be more difficult to implement and may not be robust to all problems. In methods based on Richardson extrapolation, the goal is to numerically evaluate the result that would be achieved at the limit of infinitely small step sizes. Predictor-corrector methods, on the other hand, base the value of the next point in the integration on the results of the prior point, using evaluation of the derivative to adjust the results While these methods for numerical integration are oriented towards solving systems of first-order differential equations, it may be necessary to consider systems of second-order differential equations, generically described:

$$\frac{\partial^2}{\partial\xi^2}A_n(\xi) + C\frac{\partial}{\partial\xi}A_n(\xi) = F_n\{A_m(\xi)\}$$
(3.6.3)

where Eq. 3.6.3 describes the evolution of the envelope  $A_n(\xi)$  according to its functional dependence  $(F_n)$ on a set of envelopes  $\{A_m(\xi)\}$ . However, if a new function,  $B_n(\xi)$ , is defined:

$$B_n(\xi) = \frac{\partial}{\partial \xi} A_n(\xi) \tag{3.6.4}$$

It is possible to define the second order differential equation in Eq. 3.6.3 as a pair of first-order differential equations:

$$\frac{\partial}{\partial \xi} A_n(\xi) = B_n(\xi) \tag{3.6.5a}$$

$$\frac{\partial}{\partial\xi}B_n(\xi) = -CB_n(\xi) + F_n\{A_m(\xi)\}$$
(3.6.5b)

While this treatment doubles the number of functions that need to be considered, we have now represented our previous system of second-order differential equations as a system of first-order differential equations. By reducing the system to first-order differential equations, it is now trivial to apply any desired method to evaluate the system.

The Euler method, which we introduced previously, serves as the starting point for the Runge-Kutta method. While using a single evaluation to project the evolution of the function can be unreliable, the Runge-Kutta method introduces additional evaluations to improve the projection of the function. At the lowest order, the Runge-Kutta method relies upon an additional evaluation of the derivative at the midpoint of the steps; for example, the derivative appropriate for the Euler method would be evaluated at x, then this information would be used to evaluate the derivative at  $x + \Delta x/2$ . These results would then be combined to project  $y(x + \Delta x)$ . As this improvement to the Euler method involves an evaluation at the midpoint, it is also called the midpoint method.

In order to combine the results of the evaluations at x and  $x + \Delta x/2$ , the model used mimics the Taylor series. However, just as the Taylor series can be extended to higher orders, so too can the Runge-Kutta methods. A common approach is to combine function evaluations to mimic the fourth-order Taylor series, yielding the fourth-order Runge-Kutta method. This method is accomplished by evaluating the derivative of the function for four different points (Eq. 3.6.6a - 3.6.6d), then the results are used to project the propagation of the function according to Eq. 3.6.6e.

$$k_1 = \Delta x f\{x_n, y(x_n)\} \tag{3.6.6a}$$

$$k_2 = \Delta x f\{x_n + \frac{\Delta x}{2}, y(x_n) + \frac{k_1}{2}\}$$
(3.6.6b)

$$k_3 = \Delta x f\{x_n + \frac{\Delta x}{2}, y(x_n) + \frac{k_2}{2}\}$$
(3.6.6c)

$$k_4 = \Delta x f\{x_n + \Delta x, y(x_n) + k_3\}$$
(3.6.6d)

$$y(x_{n+1}) = y(x_n) + \frac{k_1}{6} + \frac{k_2}{3} + \frac{k_3}{3} + \frac{k_4}{6}$$
(3.6.6e)

One limitation of Runge-Kutta methods that should be noted is the importance of step-size. Though Runge-Kutta methods are generally reliable, care should be taken when implementing these methods to ensure that appropriate step sizes are used in the calculation. In order to do so, the integration should be repeated with at least two step sizes (such as  $\Delta x$  and  $\Delta x/2$ ) and the results compared to ensure consistent behavior. Alternatively, controlling the algorithm with a program that monitors the results and adapts the step size as appropriate can achieve the same result: guaranteeing that the numerical integration predicts the proper result, rather than a result influenced by the step size. Introducing a routine that facilitates adaptive stepping should allow the integration to be performed slightly more efficiently, as larger steps can be taken when the functional behavior is characterized by slow changes and smaller steps when faster changes occur.[218]

In § 3.4, we derived the wave equation describing the propagation of a field, leaving the result in the frequency domain. The reason is clear in the context of the Runge-Kutta method: while the algorithm can be directly implemented to solve the wave equation in the frequency domain to any desired order of dispersion, [118, 200, 201] the standard Runge-Kutta algorithm accommodates only derivates with respect to one dimension, preventing direct consideration of dispersive effects in the time domain. However, Bakker et al [199] demonstrated a modified Runge-Kutta algorithm that directly incorporates first-order dispersive effects in the time domain. While we will consider the above discussion regarding the Runge-Kutta algorithm sufficient to discuss solution of the wave equation in the frequency-domain, we will outline the method presented by Bakker et al.

The modified Runge-Kutta algorithm described by Bakker et al treats the differential equation of each

field to be of the form:

$$\frac{\partial}{\partial z}f^n(z,\eta) = C_n \frac{\partial}{\partial \eta}f^n(z,\eta) + F^n\{f^m(z,\eta)\}$$
(3.6.7)

where  $\eta$  is a time coordinate in a moving frame of reference traveling at the group velocity of one of the fields;  $C_n$  corresponds to the GVM between the field described by  $f^n(z,\eta)$  and the field to which  $\eta$  is referenced; and  $F^n\{f^m(z,\eta)\}$  describes the functional dependence of the propagation of the field corresponding to  $f^n(z,\eta)$ on the set of fields  $\{f^m(z,\eta)\}$ . In this form, Bakker et al claim that the fourth-order Runge-Kutta method can be reproduced using the scheme described in Eq. 3.6.8:

$$\begin{split} K_1^n(z,\eta) &= F^n \{ f^m(z,\eta) \} \\ K_2^n \left( z + \frac{\Delta z}{2}, \eta \right) &= F^n \left\{ f^m \left( z, \eta + \frac{C_m \Delta z}{2} \right) + \frac{\Delta z}{2} K_1^m \left( z, \eta + \frac{C_m \Delta z}{2} \right) \right\} \\ K_3^n \left( z + \frac{\Delta z}{2}, \eta \right) &= F^n \left\{ f^m \left( z, \eta + \frac{C_m \Delta z}{2} \right) + \frac{\Delta z}{2} K_2^m \left( z + \frac{\Delta z}{2}, \eta + \frac{C_m \Delta z}{2} \right) \right\} \\ K_4^n(z + \Delta z, \eta) &= F^n \left\{ f^m(z, \eta + C_m \Delta z) + \Delta z K_3^m \left( z + \frac{\Delta z}{2}, \eta + \frac{C_m \Delta z}{2} \right) \right\} \\ f^n(z + \Delta z, \eta) &= f^n(z, \eta + C_n \Delta z) + \frac{h}{6} K_1^n(z, \eta + C_n \Delta z) \\ &+ \frac{h}{3} K_2^n \left( z + \frac{\Delta z}{2}, \eta + \frac{C_n \Delta z}{2} \right) + \frac{h}{3} K_3^n \left( z + \frac{\Delta z}{2}, \eta + \frac{C_n \Delta z}{2} \right) \\ &+ \frac{h}{6} K_4^n(z + \Delta z, \eta) \end{split}$$
(3.6.8)

While it is valuable to recognize this algorithm for treating the effects of dispersion on the propagation of electric fields in the time-domain, it is also important to note the limitations that arise in the method. First, examination of the scheme in Eq. 3.6.8 demonstrates that evaluation of  $K_1^n$  is required for six different time scales:  $\eta$ ;  $\eta + (\Delta z/2)(u_{g1}^{-1} - u_{g2}^{-1})$ ;  $\eta + (\Delta z/2)(u_{g1}^{-1} - u_{g2}^{-1})$ ;  $\eta + (\Delta z/2)(u_{g1}^{-1} - u_{g3}^{-1})$ ;  $\eta + \Delta z(u_{g1}^{-1} - u_{g2}^{-1})$ ;  $\eta + (\Delta z/2)(2u_{g1}^{-1} - u_{g2}^{-1})$ ;  $\eta + (\Delta z/2)(2u_{g1}^{-1}$ 

<sup>&</sup>lt;sup>i</sup>It should be noted that the particular points noted here are predicated on following the model of Bakker et al in which the field noted by the subscript 1 is the field to which  $\eta$  is referenced. In their example, this corresponds to either the signal or idler field in parametric amplification.

— CHAPTER 4 –

# Simulation of Sum-Frequency Idler Generation

# 4.1 Background

When doped in sapphire (Al<sub>2</sub>O<sub>3</sub>), the electronic structure of  $Ti^{3+}$  is interpreted as a consequence of a cubic environment with distortions that result in splitting of the <sup>2</sup>T<sub>2</sub> and <sup>2</sup>E degeneracies. The otherwise triply degenerate <sup>2</sup>T<sub>2</sub> ground state is split by a combination of a trigonal field and spin-orbit coupling, while the two <sup>2</sup>E excited states are split by the Jahn-Teller effect.[219] The resulting system exhibits both absorption and emission characterized by broad absorption and emission bands: both are approximately 150 nm full width at half maximum, with the absorption centered near 500 nm and emission peaking near 750 nm. Under lasing conditions, the gain is slightly red-shifted to 800 nm, and the gain profile exhibits an approximately 200 nm full width at half maximum.[219]

The broad gain bandwidth of Ti:sapphire easily provides the breadth necessary to achieve ultrafast pulses, with sub-10 fs pulses having been reported.[220–224] These ultrashort pulses are described as "soliton-like", exhibiting propagation very similar to solitons within optical fibers: the pulse is maintained by balancing the accumulation of linear phase and nonlinear phase from self-phase modulation with appropriate negative dispersion compensation.[222, 225–227] It is possible to optimize dispersion such that the pulse is limited by third- or fourth-order dispersion,[226, 228] but optimum performance is often observed with a slightly negative total group delay dispersion.[223, 226, 229] While other methods have been described,[224] the introduction of a prism pair is a common strategy for introducing negative dispersion.[159, 207, 220, 222, 226, 228, 230–232] However, as the integrated group delay dispersion is the quantity that matters,[222] some degree of flexibility exists in the functional dependence of the dispersion upon wavelength. However, pushing the limits of this tolerance is attributed as one source for the copropagation of multiple pulses within the cavity: this complication arises when different frequencies within the bandwidth experience dramatically different dispersion, incorrect applied dispersion,[226, 228] or a sufficiently low dispersion that third-order effects become dominant.[233] Copropagation of multiple pulses may also be observed in systems that sustain mode locking through Kerr lensing when over-pumping results in saturation of the Kerr effect.[229, 234]

Soliton-like propagation within an oscillator is both initiated and maintained through mode locking. Various methods of active and passive mode locking have been reported, [231, 232, 235] though regenerative methods [236] and external feedback [237] have also achieved the desired behavior. Active and passive mode locking introduce periodic modulations to the gains or losses of the cavity, inducing pulsed behavior. [238] Active mode locking yields temporal pulse envelopes that are Gaussian, but the use of active methods will generally limit the achievable pulse duration. [238] In contrast, passive mode locking schemes are capable of yielding shorter pulses [238] and exhibit very stable pulse trains with minimal timing jitter. [239] However,

the pulse envelope resulting from passive mode locking is a hyperbolic secant, [238, 240, 241] which exhibits broader wings than the Gaussian envelope of actively mode locked pulses. [242] Some authors have reasonably described passively mode locked systems within a Gaussian approximation, though. [243] While saturable absorbers can be used for passive mode locking, the broad bandwidth required for ultrashort pulses can make this a poor choice to achieve the shortest pulse durations. [221] The Kerr lens provides a much more convenient method for sustaining an ultrashort mode locked pulse: [220, 243] since it is based upon the intensity-dependent Kerr lens to reduce the losses or increase the gain of a pulse over cw operation during the pulse duration, [224, 238] it achieves the periodic gains and losses necessary for mode locking without the bandwidth limitations inherent to saturable absorbers. [221] Kerr lens mode locking can be achieved through either a hard or soft aperture. [224] Despite the advantages, one drawback to Kerr lens mode locking is initiation of mode locking: while the Kerr lens is an effective method for sustaining pulse propagation, it is less effective for self-starting. [221, 243] However, there are reports of self-starting oscillators based upon Kerr lens mode locking; [222, 231, 244] the key to such systems is that they must be designed such that the pulse experiences sufficiently greater gain than cw lasing. [244] As mentioned previously, systems based on Kerr lensing can also be subject to saturation of the Kerr effect, which allows copropagation of longer pulses or cw modes in addition to the desired ultrashort pulse. [229, 234]

The soliton-like propagation of Ti:sapphire oscillators provides a convenient avenue for the stable, reliable generation of ultrashort pulses. However, the oscillation of these lasers tend to be at high repetition rates, around 100 MHz.[221, 243] Such a high repetition rate results in a very low pulse energy, with values as low as 1-10  $\mu$ J.[245] Additionally, while the Ti:sapphire gain medium does exhibit a very broad gain bandwidth, allowing some tunability, the bandwidth still restricts oscillation and amplification to wavelengths around 800 nm. Fortunately, both of these limits can be addressed.

To overcome the pulse energy limits, regenerative amplification has been adapted to Ti:sapphire systems. Through a separate pump and cavity, regenerative amplification selects individual pulses from the oscillator's pulse train and amplifies these pulses by several orders of magnitude.[246] Such high pulse energies would result in catastrophic self-focusing in the cavity,[247] so these systems are generally based upon chirped pulse amplification.[245] This strategy is predicated upon introducing linear frequency modulation (chirp) to the pulses from the oscillator in order to generate longer pulses; the degree of stretching can vary based upon application: whereas it may be sufficient to stretch the pulses to a few picoseconds for low gain systems,[248] systems yielding higher energy may require significantly greater chirp.[249, 250] While a variety of strategies for pulse stretching and recompression exist,[248–251] care must be taken to minimize the introduction of effects such as pulse front tilt.[179] Regenerative amplification will often result in some degree of broadening and residual chirp regardless of the stretching/recompression strategy.[245, 252, 253] Gain narrowing contributes part of the limit by introducing spectral narrowing and temporal broadening.[246, 248, 253] Additionally, the number of round trips required for optimum amplification allow the accumulation of both linear and nonlinear phase. Compressors generally cannot reduce any third-order dispersion accumulated.[245, 252] Meanwhile, the nonlinear phase accumulation described by the *B*-integral will exhibit a radial dependence due to the radial distribution of the mode's intensity, making compensation challenging.[247, 254]

Increasing the tuning range of Ti:sapphire based solid state systems is generally accomplished through a variety of parametric processes. If we consider the fundamental frequency of these Ti:sapphire systems to be  $\omega_{\rm F}$ , harmonic generation makes the second,  $2\omega_{\rm F}$ , and third,  $3\omega_{\rm F}$ , harmonics accessible. However, the discretization of these available frequencies can be overcome through optical parametric generation (OPG) and optical parametric amplification (OPA).[126] In these nonlinear processes, a strong pump field at  $\omega_{\rm P}$ interacts with a weak field at the signal frequency,  $\omega_{\rm S}$ , or the idler frequency,  $\omega_{\rm I}$ . The convention is for  $\omega_{\rm S} > \omega_{\rm I}$ . In a nonlinear medium, the pulse envelopes of these fields can be shown to evolve along the direction of propagation, z, according to a system of equations;[102, 103, 118, 126, 199–201, 210, 211, 215, 216] when incorporating effects of pulse propagation, the equations of this system adopt the form of Eq. 3.4.34:

$$\frac{\partial}{\partial z}\hat{A}_{\mathrm{P}} = \left[-i\frac{1}{2}\Delta\omega^{2}k_{2,\mathrm{P}} - \frac{1}{2}\alpha(\omega_{\mathrm{P}})\right]\hat{A}_{\mathrm{P}} - i\frac{\omega_{\mathrm{P}}^{2}}{2c^{2}k_{0,\mathrm{P}}}\chi_{\mathrm{eff}}^{(2)}\mathcal{F}\{A_{\mathrm{S}}A_{\mathrm{I}}\}$$
(4.1.1a)

$$\frac{\partial}{\partial z}\hat{A}_{\rm S} = \left[-i\Delta\omega(k_{1,\rm S} - k_{1,\rm P}) - i\frac{1}{2}\Delta\omega^2 k_{2,\rm S} - \frac{1}{2}\alpha(\omega_{\rm S})\right]\hat{A}_{\rm S} - i\frac{\omega_{\rm S}^2}{2c^2k_{0,\rm S}}\chi_{\rm eff}^{(2)}\mathcal{F}\{A_{\rm P}A_{\rm I}^*\}$$
(4.1.1b)

$$\frac{\partial}{\partial z}\hat{A}_{\mathrm{I}} = \left[-i\Delta\omega(k_{1,\mathrm{I}} - k_{1,\mathrm{P}}) - i\frac{1}{2}\Delta\omega^2 k_{2,\mathrm{I}} - \frac{1}{2}\alpha(\omega_{\mathrm{I}})\right]\hat{A}_{\mathrm{I}} - i\frac{\omega_{\mathrm{I}}^2}{2c^2k_{0,\mathrm{I}}}\chi_{\mathrm{eff}}^{(2)}\mathcal{F}\{A_{\mathrm{P}}A_{\mathrm{S}}^*\}$$
(4.1.1c)

where we have taken the pump to be the reference pulse and retained the definitions of all quantities as in Ch. 3. We also mask the importance of symmetry in the nonlinear susceptibility through the use of the effective nonlinear susceptibility,  $\chi_{\text{eff}}^{(2)}$  (see § 3.2.2).[106, 119] For compactness, we have omitted the contributions of phase matching and do not explicitly note the frequency, space, or time dependence of the envelope functions.

The system of equations in Eq. 4.1.1a-4.1.1c illustrates a few key characteristics of OPA and OPG. First, in order for the polarizations corresponding to the product of pulse envelopes on the right hand side to feed the corresponding fields, the frequencies must satisfy the equality,  $\omega_{\rm P} = \omega_{\rm S} + \omega_{\rm I}$ . Additionally, it illustrates the need for the initial weak field at either  $\omega_{\rm S}$  or  $\omega_{\rm I}$ : if both fields have no initial amplitude, the only changes predicted by the system of equations are the effects of dispersion on the pump. It is adequate to provide a white light continuum as a broadband source for this initial interaction. [255] However, with a sufficiently strong pump, the vacuum field can be sufficient to initiate OPG; this superfluorescence requires much higher pump energies for stability[170] and may be limited by the presence of on- and off-axis components. [255] It is generally inadvisable to simultaneously seed the process with both signal and idler fields, [150, 170, 256] as group velocity differences can result in the formation of multiple pulses [170] and differences in phase can cause interference. [256] In the absence of pump depletion and dispersion, the system of equations predicts exponential growth of the amplitude of the signal and idler fields. [126] At high conversion rates, though, reconversion of the signal and idler according to Eq. 4.1.1a can become meaningful; [257–259] to avoid these effects, a conversion limit of ~ 20% has been cited. [258] Group velocity mismatch also has important consequences on the behavior of different processes. [125, 126, 129, 138, 154, 163, 168, 170, 171, 173, 178, 179, 182, 260–269]

Frequency conversion by parametric generation can be achieved through many different strategies. A weaker pump can be used by enhancing the conversion through oscillation in a cavity. This strategy, optical parametric oscillation, can be achieved with different pump sources, including Ti:sapphire lasers;[126, 127, 164, 181, 211, 270, 271] however, while this achieves the goal of tuning the field, it still results in low pulse energy at high repetition rates,[126] and the oscillator may require some considerations for stability.[266] Alternatively, optical parametric amplification requires higher pump energy in order to achieve appreciable conversion; however, since the optical parametric amplifiers (OPAs) do not require a cavity, they can be more straightforward to operate. When OPAs are taken to the limit of high gain, amplification can be achieved in only a few passes through the nonlinear medium; such systems are described as traveling-wave optical parametric generators (TOPG).[170, 173, 174, 177, 272]

While reported conversion efficiencies of TOPGs suggest reconversion can be relevant, there are many potential advantages. At high gain, the TOPG operates in saturation: at this limit, the performance is relatively insensitive to fluctuations in the seed, improving operating stability.[170, 178, 273] While OPA ideally yields tunable pulses at approximately the duration of the pump pulse,[173, 174, 269] this should be more feasible by minimizing the number of passes: each pass through the nonlinear crystal reduces the amplification bandwidth,[178] limiting the achievable pulse duration.[271] However, there will be a limit as to how few passes are feasible. Decreasing the number of passes requires higher gain to achieve saturation,[178] but this increased gain also increases the chance of superfluorescence, which can negatively impact the performance of the amplifier.[269] Besides the effects of amplification bandwidth, the generated signal and idler are rarely at the transform limit,[133, 266] especially for femtosecond pulses.[176] The chirp accumulated by the signal and idler is attributed to group velocity dispersion and group velocity mismatch between the

pump and signal; [266] it is not uncommon for the signal and idler to exhibit similar bandwidths, but to exhibit opposite chirp. [172]

In contrast to lasers, in which the bandwidth is restricted by the resonances of the gain medium, OPA is usually achieved in a nonresonant medium.[154, 179] However, while resonances are generally not considered relevant in determining the bandwidth of the OPA, absorption associated with resonances can limit tuning ranges; for example, in BBO, the tuning range is limited in the infrared by absorption[131, 166, 170, 173, 177, 208, 263] by phonon modes[208] and in the UV by two-photon absorption.[138] Rather, the bandwidth of the OPA is generally a consequence of the need to satisfy momentum conservation by phase matching.[125, 127, 129, 138, 154] With broadband pulses, it may prove impossible to simultaneously satisfy phase matching for all relevant frequencies,[161, 162] introducing the limits described above upon the number of passes.[178, 271] The importance of the number of passes can be understood by considering the phase matching bandwidth. By expanding the phase mismatch in a Taylor series in a manner similar to Eq. 3.4.23, it is possible to show that the phase matching bandwidth will be restricted by both increasing crystal length[106, 126] and by increasing group velocity mismatch between the signal and the idler fields.[125, 126, 129, 138, 154, 170, 178, 179, 182, 267–269]

The effect of increasing group velocity mismatch proves a limit when trying to generate visible pulses: [168, 260, 265] in a collinear geometry, the group velocity mismatch becomes restrictive, [263, 272] especially with decreasing pulse duration. [168, 173, 260, 265] In addition to the limits imposed by the phase-matching bandwidth, collinear geometries also exhibit a lower range of acceptance angles into the crystal. [125] While the range of acceptance angles does not prove limiting if beam divergence is minimized, [181] noncollinear geometries have been found to exhibit broader ranges of acceptance angles, [125, 128, 167, 181] improving efficiency. [128, 167] Noncollinear geometries can also demonstrate improved phase matching bandwidth through group velocity compensation. [125, 152, 182] When the phase matching angle  $\theta \neq 0^{\circ}$ , 90°, the birefringence causes a mismatch between the Poynting vector (the direction of energy) and the wave vector,  $\mathbf{k}$ ;[106] in a noncollinear geometry, this walk-off angle yields an effective change in group velocity. [274] This walk-off can have consequences, [128, 181, 270, 275] particularly with respect to mode structure [275] and interaction length, [128] though some authors have reported strategies that achieve some compensation. [164, 276] However, walk-off is utilized in a noncollinear geometry as the change in effective group velocity it causes can help match group velocities in the crystal. [125, 129, 152, 154, 171, 178–184] Phase matching in such noncollinear geometries is facilitated by the formation of an angularly dispersed idler; [126, 154, 269] additionally, it has been demonstrated that there exists a noncollinear angle such that the proper phase matching angle of BBO is nearly independent of wavelength.[152, 277] However, with a noncollinear geometry, changing

92

the desired signal frequency can also change the output direction.[255] Some authors have achieved similar group velocity mismatch compensation with tilted pulse-fronts[255, 260, 261, 274] or chirped pulses.[278] An appropriate amount of negative chirp has also been reported to improve efficiency.[131] Meanwhile, while tilted pulse-fronts can compensate for changes in output direction,[255] it can result in other distortions that also manifest in a noncollinear geometry.[154, 171, 182, 183]

An alternative approach for achieving a broad tuning range is to couple OPA in the near infrared with appropriate upconversion processes, [265, 272] the strategy employed in the TOPAS-C. While absorption of the idler is attributed as the limit for this tuning range, [126, 131, 166, 170, 173, 177, 208, 263] near infrared conversion can be pumped by the fundamental of Ti:sapphire lasers directly. Avoiding harmonic generation makes higher pump energies available; when coupled with the lower group velocities in the near infrared, significantly higher energies can be achieved. [126, 263] Upconversion is accomplished by a variety of processes between the pump, signal, and idler from the initial near infrared OPA: [272] the combination of harmonics and sum-frequency generation with the pump provides tuning throughout the visible and near infrared. It has been reported that upconversion can restrict the available bandwidth; [184] group velocity mismatch can also prove limiting. [171, 262] However, a wide variety of reports have demonstrated that upconversion processes can have pronounced consequences on the spectral or temporal character, [133, 154, 156, 163, 170, 176, 258, 264, 265, 273, 274, 279, 280] particularly under conditions of significant conversion or depletion [154, 170] and pronounced effects of group velocity. [156, 163, 262, 264] The behavior can also change according to the initial delay between the involved fields. [264]

We believe that such behaviors are to account for unusual behavior observed when generating the sumfrequency between the idler and residual pump (SFI), as shown in Fig. 4.1.<sup>i</sup> Exploring the effects of changing the delay between the pump and the signal seed in the OPA showed that the spectral characteristics observed can be changed with this adjustment, and that optimized signal and idler conversion did not necessarily contribute to optimum sum-frequency generation. Despite the unusual spectral character observed in the SFI process, the signal process appeared consistent with normal operation (see Fig. 4.2); it is difficult to reliably characterize the idler, but the literature suggests the bandwidth of the signal and idler processes should be similar.[172] The unexpected behavior of SFI is consistent with the observations of Danielius et al.[272] When using a very similar system, but with longer pulses (~ 100 fs rather than ~ 30 fs) and slightly longer (4 mm and 0.6 mm rather than 2 mm and 0.3 mm BBO for OPA and SFI, respectively), Danielius et al report very low conversion for the sum-frequency generation process between the idler and the pump

<sup>&</sup>lt;sup>i</sup>Experimental data collected with Eric Hagee, Daniel Kohler, and Blaise Thompson.

except near 600 nm, where they observe  $\sim 50\%$  higher efficiency.[272] They do not observe similar behavior for sum-frequency generation between the pump and signal.[272] Their explanation for these observations is that the group velocity mismatch between signal and pump is nearly independent of signal wavelength and non-zero in the Type II phase matching of the OPA;[126, 172] when sum-frequency generation is performed, however, this reverses the group velocity mismatch, limiting the effects of the accumulated group delay between the pump and the signal on sum-frequency generation. In contrast, as Type II phase matching results in parallel polarizations for the pump and idler, the accumulated group delay between the pump and the idler is not reversed during sum-frequency generation; rather the group delay is increased by continued co-propagation through additional crystals. Unlike signal, though, the pump-idler group velocity mismatch exhibits significant dependence on idler wavelength, crossing zero near 2400 nm (signal wavelength, 1200 nm).[126, 172] The sum frequency of 2400 nm and the fundamental occurs at 600 nm, where Danielius et al observe the unusually high efficiency.[272] This is also near the wavelength at in which we observe the unusual spectral characteristics of Fig. 4.1. Unfortunately, Danielius et al do not provide spectral characterization of their system. The simulations reported here seek to confirm that the features in Fig. 4.1 are expected behavior for our system and improve our understanding of the underlying mechanisms.

# 4.2 Methodology

A wide variety of authors have attempted to explore the simulation of parametric processes.[118, 199–201, 212–217] Bakker et al appear to have described one of the earlier methods towards simulating ultrafast pulse propagation, introducing a modified Runge-Kutta algorithm that allowed solving a system exhibiting both spatial and time dependence (see § 3.6).[199] However, there are multiple drawbacks to their approach. First, the algorithm can only handle behaviors associated with the first derivative with respect to time.[199, 201] Second, while this algorithm can be coded and implemented, the number of evaluations and the density of data points necessary to incorporate the temporal dependence make the calculation prohibitively expensive. Gale et al have documented an alternative approach for evaluating the system of differential equations describing OPA, which permits evaluation to an arbitrary order of dispersion.[200, 201] This is possible through operating in both the time and frequency domains. At each step of the numerical integration, the polarization at each frequency is calculated, the results are Fourier transformed to the frequency domain, and the evolution of the fields projected. Evaluation of the field changes in the frequency domain permits direct application of the Runge-Kutta algorithm:[218] as shown in § 3.4.1, Fourier transformation of the relevant system of equations masks the differential behavior associated with dispersion. While Fourier transformation



Figure 4.1: Results of sum-frequency idler process tuning. (A) shows the two dimensional representation of the tuning data, plotted as the detuning,  $\lambda_m - \lambda_0$ , versus assigned sum-frequency idler center color,  $\lambda_0$ , where  $\lambda_m$  is the wavelength of the monochromator at each point. (B), (C), (D), and (E) show the differences in spectral profile along different slices in the two-dimensional data set; the color and style of the line in each plot provide color- and style-coding to the slices in (A).



Figure 4.2: Results of signal process tuning prior to sum-frequency idler tuning of Fig. 4.1, plotted as the detuning,  $\lambda_m - \lambda_0$ , versus assigned signal center color,  $\lambda_0$ , where  $\lambda_m$  is the wavelength of the monochromator at each point.

permits consideration of dispersion to an arbitrary order, we follow the assumption in § 3.4.1 and only consider effects up to the second order in time/frequency. While this method follows the example of Gale et al[200, 201] and later authors employing the same approach,[118] we note that the system of equations we employ based upon the derivation in § 3.4.1 deviates slightly from the equations used by these authors.

The Python code developed for these calculations - with some modifications for presentation - is presented in Appendix C. It consists of a series of scripts containing discrete parts of the calculation. It should be noted that during development of the code, the architecture necessary for several features - such as support for isotropic media and adaptive stepping - were incorporated; however, as these features were not needed, they have not been developed. In earlier versions, the code necessary for evaluating second-order differential equations was maintained; however, testing has indicated it is reasonable to operate within the slowly varying amplitude approximation. Maintenance of this portion of the code has since been discontinued, and much of these portions of the code have been omitted in the appendix.

Before discussing the strategy we employed for simulating the SFI process, we will first outline the approximations implicit within the code. First, we adopt the plane wave limit for each field and neglect effects such as walk-off, phase effects of a TEM<sub>00</sub> mode, refraction, and imaging. The effects of spatial walk-off are neglected for convenience: due to the short crystal lengths in the crystals in the TOPAS-C, the effects of walk-off that would be expected are comparable to the resolution we adopt in radial sampling. As it is common to put the crystal just after the focus of the pump,[126] the propagation of the TEM<sub>00</sub> mode has the potential to be an important effect, especially when considering the imaging of the pump and the idler modes on the sum-frequency crystal. However, as we do not know the exact engineering of the TOPAS-C units, we neglect the phase terms associated with the TEM<sub>00</sub> mode and simply consider the amplification and sum-frequency crystals to be in direct sequence. While we neglect the phase terms of the TEM<sub>00</sub> mode, we do consider a radial intensity distribution consistent with a Gaussian profile; the handling of this is functionally equivalent to appropriately weighting the contributions of plane waves with different fluence. The beam sizes employed in the calculations are reported with the other characteristic parameters below; these have been adjusted within seemingly reasonable bounds so as yield conversion efficiencies comparable to those observed in the TOPAS-C.

In contrast with Gale et al and Wang and Wang, [118, 200, 201] we do not incorporate the effects of third-order nonlinear polarizations. While some authors suggest that the effects of third-order nonlinear polarizations during OPA may be important, [131, 281] others question the magnitude of their effects. [259] We also point out that some authors indicate TOPGs may be operated at lower gain levels in the later amplification stages; [170] the correspondingly lower pump intensities would be expected to reduce the sig-

nificance of any third-order effects. There also appear to be questions whether perceived third order effects are the result of third-order polarizations or cascaded effects; [118] see for instance that cascaded side bands can exhibit phase matching consistent with "self-diffraction" without clearly requiring a third-order polarization. [282–284] In addition to third-order polarizations, we also do not account for the possibility of parasitic processes, [172, 275, 285, 286] though it is clear that these can have consequences on the behavior of a system, either through spectral distortions or inefficiencies. [275]

While the Selmeier coefficients provided by Kato and Eimerl et al appear to be the standard description for the dispersion of BBO[106, 287, 288] and other authors have indicated they are reasonably accurate in the near infrared, [289] we have opted to employ the modified dispersion formulae reported by Zhang et al. [208] We note here that the group velocity mismatch curves projected by all three sources (Fig. 4.3) exhibit similarities, including a crossing point at which the pump-idler group velocity mismatch  $\Delta u_{pi}^{-1} = 0$  fs/mm; however, as Fig. 4.3 shows, this crossing point is projected at signal wavelengths as short as  $\lambda_s = 1108$  nm  $(\lambda_i = 2878 \text{ nm})[288]$  or as long as  $\lambda_s = 1330 \text{ nm} (\lambda_i = 2008 \text{ nm}).[287]$  Consequently, while we hope to observe similar behavior as seen in Fig. 4.1 in our simulations, we do not anticipate an exact match for the wavelength at which the features might occur. Absorption data has been simulated to reproduce available information. Data describing the absorbance of BBO has been published by Eimerl et al. [287] but a clearer set of data are available online.<sup>i</sup> While no crystal length was reported for the online reference, comparison with the reported absorbance by Eimerl et al allowed approximate scaling. We estimated absorption coefficients at various colors and interpolated as necessary. The results of these interpolations yield a higher coefficient at 532 nm ( $\sim 0.05 \text{ cm}^{-1}$  instead of 0.01 cm<sup>-1</sup>) and a lower coefficient near 2.55  $\mu$ m ( $\sim 0.3 \text{ cm}^{-1}$  instead of  $0.5 \text{ cm}^{-1}$ ) than those reported by Dmitriev et al. [106] As the interpolated values are of the correct order of magnitude, our interpolated absorbance should provide a reasonable approximation, especially given the short crystal lengths employed in the TOPAS-C.

Because the generation of the seed in the TOPAS-C is accomplished in a noncollinear geometry, we could not accurately model this. Some testing was done approximating the noncollinear interaction using an effective group velocity as calculated according to the formula reported by Dubietis et al.[274] In this treatment, we used the model for white light described by Reisner and Gutmann.[217] The calculation of the white light field proved computationally expensive, but in these tests we found that the resulting seed was comparable to an envelope with a phase of  $\sim 40^{\circ}$ . We therefore have assumed the seed envelope to be described by a Gaussian profile of 35 fs duration with an initial phase of  $40^{\circ}$ . This neglects residual

<sup>&</sup>lt;sup>i</sup>Located at u-oplaz.com.



Figure 4.3: A comparison of the group velocity mismatch curves projected by the BBO dispersion curves from different authors, from top to bottom: Eimerl et al;[287] Kato;[288] and Zhang et al.[208] Magenta curves are pump-idler group velocity mismatch,  $\Delta u_{pi}^{-1} = 1/u_p - 1/u_i$ , and black curves are pump-signal group velocity mismatch,  $\Delta u_{ps}^{-1} = 1/u_p - 1/u_s$ . Dashed lines represent the group velocity mismatch curves for Type I ( $e \leftarrow oo$ ) phase matching, and solid lines are the curves resulting for Type II ( $e \leftarrow oe$ ) phase matching. The phase matching angle was determined within the CW limit. The green lines identify the point at which each source for BBO dispersion data predicts  $\Delta u_{pi}^{-1} = 0$  fs/mm.

chirp, which is also neglected in the pump. It has been reported that a slight negative chirp can compensate for dispersion in the crystal and improve performance accordingly;[131] as we optimize recompression in the regenerative amplifier to TOPAS-C performance, this is undoubtedly meaningful, but we have no good estimate for the effects.

In attempting to reproduce the behavior of Fig. 4.1, we have sought to emulate the tuning procedure employed on the femtosecond system as closely as possible. According to our approximation regarding the seed, we have assumed that the first amplification stage has already been well-tuned and that a signal seed at any desired frequency can be generated. We proceed to simulate the acquisition of multidimensional tuning data at a distribution of signal colors: two of these dimensions arise from changing the phase matching angle and the temporal delay between the seed and the pump, while the third represents the distribution of frequencies within the fields. This is directly analogous to the acquisitions of D2 versus C2 versus  $\lambda_m$ collected during tuning of the TOPAS-C. The resulting data set is reduced to two two-dimensional data sets that track the functional dependence upon phase matching angle and delay of the zeroth (integrated energy) and the first (mean, or central color) moments of the signal field. These metrics were selected as the data in Fig. 4.1 question the validity of the parameters of Gaussian fits as reliable characterization metrics. Tuning curves were generated by picking the optimum point in the two dimensional space of phase matching angle and delay; the corresponding tuning curve point was obtained by assigning these values to the central color as determined by the first moment. The first metric for determining the optimum point was to require the first moment to be within  $\sim 1.5$  nm of the central color of the seed; of points satisfying this first metric, the appropriate tuning position was then assigned on the basis of maximum integrated energy.

The resulting tuning curves (phase matching angle versus signal color and delay versus signal color) were interpolated using the UnivariateSpline function from scipy.interpolate. The knots/smoothing factor were set to 5/10000 for the phase matching angle and 3/1000 for the delay. These interpolation functions were used to tune the sum-frequency process. For a distribution of SFI colors, the appropriate signal color was calculated on the basis of energy conservation; the generation of this signal color and its corresponding idler were calculated using the phase matching angle and delay interpolated by the splined tuning curves. The pump and idler fields from this simulation were retained and assigned as the pump and idler fields for sum-frequency generation. The sum-frequency generation of these fields was simulated for a distribution of phase matching angles. For each SFI color, the optimum phase matching angle was chosen by maximizing the SFI energy, and the first moment of the corresponding sum-frequency spectrum was calculated. The functional dependence of phase matching angle and energy-conserving SFI color upon these first moments provided the SFI tuning curves: for a desired SFI color, the latter curve permitted identification of the
appropriate signal color for simulating pump and idler generation, while the former indicated the proper phase matching angle for sum-frequency generation. Both tuning curves were interpolated using the same UnivariateSpline function (5/10000 knots/smoothing factor).

With these tuning curves, both the base OPA and the SFI upconversion processes were simulated with an even distribution of assigned colors. These simulations are reported in the following section. The parameters used for these simulations are reported in Table 4.1. The field parameters for the SFI simulation were inherited from the appropriate OPA simulation.

# 4.3 Results and Discussion

In order to test the performance of the script, we simulated the second-harmonic generation of 2 ps, 800 nm pulses. Within the framework of § 3.4.1, this process would occur according to the system of equations relating the fundamental (F) and second harmonic (SH):

$$\frac{\partial}{\partial z}\hat{A}_{\rm F} = \left[-i\frac{1}{2}\Delta\omega^2 k_{2,\rm F} - \frac{1}{2}\alpha(\omega_{\rm F})\right]\hat{A}_{\rm F} - i\frac{\omega_{\rm F}^2}{2c^2k_{0,\rm F}}\chi_{\rm eff}^{(2)}\mathcal{F}\{A_{\rm F}A_{\rm SH}^*\}$$
(4.3.1a)

$$\frac{\partial}{\partial z}\hat{A}_{\rm SH} = \left[-i\Delta\omega(k_{1,\rm SH} - k_{1,\rm F}) - i\frac{1}{2}\Delta\omega^2 k_{2,\rm SH} - \frac{1}{2}\alpha(\omega_{\rm SH})\right]\hat{A}_{\rm SH} - i\frac{\omega_{\rm SH}^2}{2c^2k_{0,\rm SH}}\chi_{\rm eff}^{(2)}\mathcal{F}\{A_{\rm F}A_{\rm F}\}$$
(4.3.1b)

However, with such long pulse durations, dispersive effects should be negligible for reasonable crystal lengths.<sup>i</sup> Additionally, we can approximate the pulse envelope as the amplitude at the central frequencies,  $A_{0,\text{F}}$  and  $A_{0,\text{SH}}$ ; this allows us to also treat the frequencies of each field  $\omega_{\text{F}}$  and  $\omega_{\text{SH}}$  as the central frequencies,  $\omega_{0,\text{F}}$  and  $\omega_{0,\text{SH}}$ . Within this limit of narrowband pulses, we should also be able to assume phase matching is reasonably well satisfied across the spectrum. Finally, as we are dealing with visible frequencies, the absorbance of BBO should be minimal. This reduces the system of Eq. 4.3.1a and 4.3.1b to:

$$\frac{\partial}{\partial z}A_{0,\mathrm{F}} = -i\frac{\omega_{0,\mathrm{F}}^2}{2c^2k_{0,\mathrm{F}}}\chi_{\mathrm{eff}}^{(2)}A_{0,\mathrm{F}}A_{0,\mathrm{SH}}^*$$
(4.3.2a)

$$\frac{\partial}{\partial z}A_{0,\rm SH} = -i\frac{\omega_{0,\rm SH}^2}{2c^2k_{0,\rm SH}}\chi_{\rm eff}^{(2)}A_{0,\rm F}A_{0,\rm F}$$
(4.3.2b)

Note that as we are operating within the narrowband limit, we are no longer retaining a distinction between the frequency-domain amplitude  $(\hat{A})$  and the time-domain amplitude (A). In the regime of low conversion, the effects of Eq. 4.3.2a will be perturbative to the fundamental field, allowing us to adopt the assumption

 $<sup>^{</sup>i}$ For the dispersion curves given by Zhang et al.[208] the predicted group velocity mismatch for the second harmonic generation of 800 nm is 232 fs/mm. For 2 ps pulses, then, neglecting group velocity mismatch should be a reasonable approximation up to 8-9 mm.

Parametric Amplification Parameters					
Process	Parametric Amplification				
Phase Matching	Type II				
Crystal Length	0.2 cm				
Azimuthal Angle	0°				
Steps	250				
Sum-Frequency	Sum-Frequency Generation Parameters				
Process	Sum-Frequency Generation				
Phase Matching	Type I				
Crystal Length	0.03 cm				
Azimuthal Angle	90°				
Steps	150				
Shared Parameters					
Dispersion Curve	Zhang et al[208]				
Time Samples	128				
Radial Samples	32				

Parameter	Pump	Signal	Idler
Wavelength	800 nm	(variable)	(variable)
Center	0  fs	(variable)	0  fs
Gaussian Width	35  fs	$35 \mathrm{~fs}$	35  fs
Energy	1500 μJ	1 μJ	0 μJ
$1/e^2$ Size	1.0 cm	$0.5~\mathrm{cm}$	$0.5~\mathrm{cm}$
Envelope Phase	0°	40°	0°

(b) Field Parameters

Table 4.1: Parameters used for sum-frequency idler simulations.

that  $\partial/\partial z A_{0,F} = 0$ . This leaves only Eq. 4.3.2b with a constant source,  $A_{0,F}$ . The solution to this equation is straightforward to evaluate, yielding:

$$A_{0,\rm SH} = -i \frac{\omega_{0,\rm SH}^2}{2c^2 k_{0,\rm SH}} \chi_{\rm eff}^{(2)} A_{0,\rm F}^2 z$$
(4.3.3)

under the assumption that no second harmonic is present at z = 0. In terms of energy, this result is equivalent to:

$$\frac{E_{\rm SH}}{f(w_0)} = \frac{\omega_{0,\rm SH}^4}{4c^4k_{0,\rm SH}^2} |\chi_{\rm eff}^{(2)}|^2 \frac{E_{\rm F}^2}{f(w_0)^2} z^2$$
(4.3.4)

where  $E_{\rm SH}$  and  $E_{\rm F}$  are the energy of the second harmonic and fundamental pulses, respectively;  $f(w_0)$  is a beam size dependent distribution of that energy, and other variables are defined as before. From Eq. 4.3.4, we therefore anticipate that in the low conversion limit, the energy of the second harmonic should scale with a quadratic dependence upon length (z) and fundamental energy ( $E_{\rm F}$ ) and a reciprocal dependence upon the energy distribution function.<sup>i</sup> The parameters employed for these calculations are summarized in Table 4.2, and the results are summarized in Fig. 4.4.

In these simulations, we found the behavior of Eq. 4.3.4 to be observed until approximately 10% conversion; at this point, deviations from the expected dependence began to become significant. This is comparable to reported bounds at which reconversion begins to be expected in OPA (~ 20%).[258] As indicated by the result of Eq. 4.3.4, the energy of the second harmonic is expected to exhibit quadratic dependence upon the fundamental energy and the crystal length, in very good agreement with the results in (A) and (C) of Fig. 4.4, where we observe high correlation with models that scale according to  $E_{\rm F}^{1.9717}$  and  $L^{1.9183}$ . Meanwhile, the energy distribution we employ in the calculations is  $[(\pi/2)w_0^2]^{-1} \exp[-2(r/w_0)^2]$ , where  $w_0$  is half the provided  $1/e^2$  beam size. Within this model, we should expect the functional dependence upon beam size to scale as the non-exponential term,  $[(\pi/2)w_0^2]^{-1}$ . In (D), we see that the low conversion regime of second harmonic generation exhibits a very strong match to this model, scaling as  $w_0^{-1.9682}$ . The one limitation that is apparent in the script comes in (B): ideally, we should observe energy conservation in the process. It is clear that at low energies this is not the case. We attribute this limitation at small second-harmonic energies to the limited resolution of our sampling grid (32 spatial points, 128 temporal points). It is clear that the script converges to the correct behavior with respect to energy conservation at higher energies; from this, we take the sensitivity of the script to be at the level of  $10^{-2} \cdot 10^{-1} \ \mu$ J. As we will be simulating TOPG, we

<sup>&</sup>lt;sup>i</sup>We note, too, that Eq. 4.3.4 exhibits a dependence upon central frequency, and that we have previously noted the omission of a phase matching dependence. Neither of these prove straightforward to incorporate cleanly, however. As the frequency influences  $\chi^{(2)}_{\text{eff}}$  through the phase matching angle  $\theta$  and  $k_0$  through both  $\omega_0$  and the phase velocity, the dependence upon  $\omega_0$ ends up more complicated than the quadratic dependence that first appears to be evident. Meanwhile, exploring the influence of  $\Delta k$  is computationally challenging as the phase modulation changes the characteristic step size between runs.

Dispersion Curve	Zhang et al.[208]
Process	Second-Harmonic Generation
Phase Matching	Type I
Crystal Length	$0.5 \mathrm{~cm}$
Azimuthal Angle	90°
Step Size	10 µm
Time Samples	128
Radial Samples	32

Parameter	Fundamental	Second Harmonic
Wavelength	800 nm	400 nm
Center	0  fs	0  fs
Gaussian Width	2  ps	2  ps
Energy	5 μJ	0 µJ
$1/e^2$ Size	1.0 cm	$0.5~{ m cm}$
Envelope Phase	0°	0°

(a) Simulation Parameters

(b) Field Parameters

Table 4.2: Parameters used for second-harmonic test simulations.



Figure 4.4: Results of simulations of second-harmonic generation as a function of fundamental energy (A), crystal length (C), and beam size (D). Simulations performed according to the parameters in Table 4.2. (B) shows the change in energy predicted for the fundamental and second harmonic at each initial fundamental energy tested in (A).

should generally be operating safely beyond this regime.

Having demonstrated that the behavior of the script is consistent with expectations and recognizing the sensitivity limits, we will present the results of the simulations of the sum-frequency idler process; after presenting the results, we will attempt to explain them in the context of published treatments of OPA and sum-frequency generation.

As noted in our outline of the methodology, the first step in these simulations was to emulate the tuning procedure employed on the femtosecond system. With the assumption that the preamplification stage has already been "well-tuned" and that nearly Gaussian seed profiles can be generated at any desired signal color, we explored the space corresponding to the pump-seed delay and the amplification crystal angle for a distribution of signal colors. As the resulting data forms a four-dimensional data set (amplitude/intensity versus frequency versus crystal angle versus delay), it is not convenient to analyze in its original state. However, in Fig. 4.5 and 4.6, we have reduced this four-dimensional data set to two three-dimensional data sets, characterizing the amplitude/intensity versus frequency dimensions according to their zeroth and first moments.

For seed colors with wavelengths longer than 1180 nm, we observe behavior in these results consistent with expectations. When examining the first moment (central color) as a function of angle versus delay in Fig. 4.5, we observe a dependence upon angle only. In the model example of these simulations, this is completely reasonable, as neither the seed nor pump have any time dependence; if we introduced chirp into the seed, which is likely more consistent with the character of the seed in the TOPAS-C, we would expect to observe some curvature or tilt to the contours appropriate to the functional form of the frequency modulation. The relationship between color and crystal angle is made clear by examining the phase matching angles predicted for BBO (as calculated from the dispersion curves described by Zhang et al [208]) within the cw approximation in Fig. 4.7: in the phase matching curve predicted for Type II phase matching, there is a monotonic trend of increasing phase matching angle with increasing signal color for all wavelengths longer than 1150 nm. This matches the behavior observed in Fig. 4.5, where a blue (red) shift is observed for decreasing (increasing) crystal angle. Meanwhile, at first glance, the results of the zeroth moment (integrated energy) in Fig. 4.6 for wavelengths longer than 1180 nm seem unusual. While it is not clear from Fig. 4.6 (the tuning curves shown in Fig. 4.8 illustrate this more clearly), there is a clear trend that the optimum delay is slightly negative; it seems reasonable that a slight delay would be optimum in order to compensate for the group velocity mismatch between the pump and the seed. However, the unusual trend is that optimum amplification appears to occur at smaller crystal angles than the angle assigned as the optimum tuning point. However, recall from § 3.2.2 that, in Type II phase matching,  $d_{\text{eff}}$  scales with respect to  $d_{22}$  according to  $d_{22}\cos^2(\theta)\cos(3\phi)$ . Consequently, we would anticipate  $d_{\text{eff}}$  to actually increase as we decrease the phase matching angle; this would be expected to increase the gain at these smaller angles. We attribute the optimum amplification at smaller phase matching angles to this effect.

In contrast to wavelengths longer than 1180 nm, the results for 1140-1180 nm appear less straightforward; each of these wavelengths exhibits less straightforward behavior as a function of crystal angle and delay, both in terms of the trends observed and in the rate of these changes. However, recalling Fig. 4.3, we note that this region is where the dispersion curves of Zhang et al predict the pump-idler group velocity to reach zero.[208] As a consequence, this is where we anticipate deviations from ideal behavior to begin manifesting; under the high gain of the TOPG, reconversion between the pump and the signal and idler is expected, likely exacerbating any deviations already occurring.

The tuning curves projected by these optimum points are shown in Fig. 4.8 (A). As these tuning curves show, the optimum delay appears quite dynamic at lower wavelengths, likely due to the pump-idler group velocity mismatch crossing zero; otherwise, the behavior of the delay tuning curve appears to mostly be limited by the resolution of our sampling in the simulations. The combination of these two does appear to limit the accuracy of the tuning curve in predicting the optimum delay: this is apparent in Fig. 4.5 and 4.6, where the tuning point assigned by the tuning curve is predominantly shifted from the optimum point horizontally - along the delay axis. In contrast to delay, the tuning curve as a function of crystal angle is much more monotonic. At wavelengths longer than 1250 nm, the curve exhibits a trend very similar to the theoretical curves (Fig. 4.7) predicted using the dispersion curves of Zhang et al.[208] At shorter wavelengths, deviation from the theoretical curve is observed; in contrast to the behavior of delay, though, this deviation is relatively smooth and can be interpolated quite well.

The signal tuning curves permitted simulated tuning of the sum-frequency idler; the results of these simulations are much less insightful as they are interpreted solely on the basis of maximizing the zeroth moment (integrated energy). The resulting tuning curves from these simulations are shown in Fig. 4.8 (B). In the same manner as (A), there are two axes of relevance when trying to simulate a desired sum-frequency idler color: a phase matching angle and a color corresponding to the energy conservation condition of an appropriate signal point. The latter of these curves is quite monotonic and can be well described by an interpolated curve. On the other hand, the results for the phase matching angle are reasonably well interpolated as the sum-frequency idler wavelength increases until  $\sim 605$  nm, where a deviation from monotonic behavior is observed. This deviation occurs at frequencies corresponding to signal wavelengths of 1180 nm; from the group velocity mismatch curves in Fig. 4.3, it seems reasonable to attribute this lack of monotonicity to the pump-idler group velocity mismatch crossing from positive to negative values.



Figure 4.5: Results of simulations of signal tuning, presenting detuning of the first moment as a function of angle detuning and delay detuning. Angle (delay) detuning reflects the detuning of the phase matching angle (initial pump-signal delay) relative to the angle (delay) position assigned as optimum. Detuning of the first moment is measured relative to the first moment of the optimum point; contours reflect 5 nm increments to the blue (dashed) or the red (solid). The unfilled marker at (0,0) corresponds to the assigned optimum point; the filled marker indicates the delay and angle assigned by interpolation of the tuning curve. The label identifies the center wavelength of the seed used for the simulations.



Figure 4.6: Results of simulations of signal tuning, presenting the zeroth moment (integrated signal pulse energy) as a function of angle detuning and delay detuning. Angle (delay) detuning reflects the detuning of the phase matching angle (initial pump-signal delay) relative to the value assigned as optimum. The contours represent steps corresponding to 10% of the maximum energy observed in the range of simulations for that seed wavelength. The unfilled marker at (0,0) corresponds to the assigned optimum point; the filled marker indicates the delay and angle assigned by interpolation of the tuning curve. The label identifies the center wavelength of the seed used for the simulations.



Figure 4.7: Theoretical tuning curves predicted for Type I and Type II phase matching in a BBO OPA according to the dispersion curves reported by Zhang et al[208] within the cw approximation.



Figure 4.8: The tuning curves for the simulated optical parametric amplification (A) and sum-frequency idler (B) processes.

The results of simulating signal and idler amplification are shown in Fig. 4.9. (A) shows the spectrum predicted for the signal process; the results is reasonably symmetric and is consistent with some of the cleaner data sets collected of a well-tuned signal process. The idler, meanwhile, is also quite symmetric at shorter wavelengths. As the idler wavelength increases, though, the spectrum clearly becomes broader. However, due to the scaling of frequency and wavelength, it is difficult from visual examination to directly attribute the breadth to spectral broadening or to the scaling of a constant bandwidth at longer wavelengths. We will leave a more detailed analysis for later in this discussion of the simulation results. Unfortunately, we are not aware of a good data set exhibiting experimental analysis of the idler process: past efforts to do so have proven challenging due to a variety of software and hardware limitations.

The final set of results is the simulated sum-frequency idler spectrum. However, before discussing the results in detail, two details bear mentioning. First, we recall our conclusion from Fig. 4.3 that all models of BBO dispersion predict a different wavelength where pump-idler group velocity mismatch crosses from positive to negative; consequently, we will avoid making direct comparison of the wavelengths at which features are observed. We will instead focus on the qualitative observations and the relative ordering of these features. Second, we will note that we have assigned wavelengths according to the first moment, whereas COLORS frequently employs Gaussian fits for the assignment of center; this will cause different definitions of zero detuning for slices in the simulated data relative to slices in the experimental data.

Qualitatively, the two-dimensional data, given in (A) in both Fig. 4.1 and 4.10, exhibit many similarities. At short sum-frequency idler wavelengths, both appear to exhibit relatively symmetric line shapes, with a broadened feature extending to longer wavelengths (positive detunings) beginning to occur at longer wavelengths. After this broadening, the spectrum narrows again, and also reaches its most intense spectral features as the tuning curve reaches the longest wavelengths.

The progression of spectral distortion is shown more clearly by the slices in (B)-(E) of Fig. 4.1 and (B)-(G) in Fig. 4.10. Initially, the spectrum is relatively symmetric (Fig. 4.10, B), until a shoulder begins to develop at longer wavelengths (Fig. 4.1, B and Fig. 4.10, C). This shoulder grows in amplitude until it nearly matches the higher energy peak (Fig. 4.1, C and Fig. 4.10, D). As the sum-frequency idler color is tuned further to the red, though, what had been the higher wavelength shoulder becomes the main peak of the feature (Fig. 4.1, D and Fig. 4.10, E), and the shorter wavelength feature becomes a gradually weakening shoulder (Fig. 4.10, F). After this shoulder becomes sufficiently weak, all that remains is a skewed spectrum (Fig. 4.1, E and Fig. 4.10, G).

Before getting into the analysis, we will briefly discuss the metrics we will use throughout the following discussion. These metrics are based upon quantities analogous to the moments about the origin of a prob-



Figure 4.9: The results of simulations of optical parametric amplification of signal (A) and idler (B) processes under traveling-wave operation. Both provide the spectral content at each central frequency of the signal, presented as intensity versus detuning  $(\lambda_m - \lambda_0)$  versus central wavelength,  $\lambda_0$ , of signal (based upon chosen wavelength) or idler (as calculated by energy conservation). The spectrum in (A) is provided in comparison to the signal tuning results in Fig. 4.2.



Figure 4.10: The results of simulations of sum-frequency idler produced by upconversion of the simulated idler in Fig. 4.9 (B) with the depleted pump. (A) presents a two-dimensional representation of the spectral content at each sum-frequency idler central wavelength. This is analogous to (A) in Fig. 4.1, presenting intensity as a function of detuning  $(\lambda_m - \lambda_0)$  versus sum-frequency idler central wavelength  $(\lambda_0)$ . (B)-(G) present color- and style-coded slices from (A), presented as normalized intensity versus detuning  $(\lambda_m - \lambda_0)$ , in analogy to (B)-(E) of Fig. 4.1.

ability distribution; however, since the simulated spectra are not normalized, we will add a normalization factor in the manner of Eq. 3.1.21. This normalization results in a calculation of the  $n^{\text{th}}$  moment about the origin,  $\mu'_n$ :[290]

$$\mu_n' = \frac{\int_0^\infty \Delta \omega^n |\hat{A}(\Delta \omega)|^2 d\Delta \omega}{\int_0^\infty |\hat{A}(\Delta \omega)|^2 d\Delta \omega}$$
(4.3.5)

We will focus particularly upon the second and third moments,  $\mu'_2$  and  $\mu'_3$ . From the second moment, we will calculate the standard deviation describing the spectral width according to the first and second moments:[290, 291]

$$\sigma = \sqrt{\mu_2' - \mu^2} \tag{4.3.6}$$

In Eq. 4.3.6,  $\mu$  is the mean and is calculated as  $\mu'_1$  according to Eq. 4.3.5.[290] For the purpose of this discussion, we will consider the standard deviation as calculated according to Eq. 4.3.6 to be an adequate representation of the spectral bandwidth. This bandwidth will be reported in units of energy (cm<sup>-1</sup>) to allow direct comparison within and across the different tuning regions. The third moment will be used to calculate the skew of the distribution, reported by  $\alpha_3$ . This quantity is calculated from the third moment about the mean,  $\mu_3$ :[290]

$$\alpha_3 = \frac{\mu_3}{\sigma^3} \tag{4.3.7}$$

where  $\sigma$  is the standard deviation calculated according to Eq. 4.3.6 and  $\mu_3$  is calculated from  $\mu'_3$ ,  $\mu'_2$ , and  $\mu$ :[290]

$$\mu_3 = \mu'_3 - 3\mu'_2\mu + 2\mu^3 \tag{4.3.8}$$

The value  $\alpha_3$  describes the skew of the distribution, the degree of asymmetry.[290] We generally will not be concerned with the actual value or sign of  $\alpha_3$ , but rather whether it is zero or nonzero (symmetric or asymmetric) and its relative magnitude.

Danielius et al[272] attribute the increased efficiency of idler upconversion in their OPG system near 600 nm to the minimization of the pump-idler group velocity mismatch. This is consistent with the understanding that the pump-idler and pump-signal group velocity mismatch can limit the gain of an OPA,[126] and that pre-existing group delay can impede sum-frequency generation.[103] This initially led us to suspect that the pump-idler group velocity mismatch may be responsible for the distortions observed in Fig. 4.1 and reproduced in Fig. 4.10.

In Fig. 4.11, we reproduce the two-dimensional representation of our SFI simulations from Fig. 4.10 in (A); we have also overlaid the simulated pulse energies. We have applied the statistical metrics in Eq. 4.3.6 and 4.3.7 to provide measurements of the bandwidth ( $\sigma$ , shown in (B) of Fig. 4.11) and skew ( $|\alpha_3|$ , shown in

(C) of Fig. 4.11). These metrics are compared to the pump-idler and pump-signal group velocity mismatch as calculated according to the dispersion curves of Zhang et al. [208]

A comparison of the pump-idler group velocity mismatch and the simulated pulse energy does show that the conversion is optimized when the group delay from the optical parametric amplification is minimized by the pump and idler exhibiting identical group velocities. This is consistent with the conclusions of Danielius et al.[272] In contrast, the group velocity mismatch is not minimized when we observe distortion to the spectral content: rather, as measured by the bandwidth ( $\sigma$ ) and skew ( $|\alpha_3|$ ), the distortion to the spectral character is maximized when mismatch still exists between the pump and idler group velocities.

The disagreement we observe in Fig. 4.11 between the frequencies at which we observe distortions to the sum-frequency idler spectral character and minimization of the pump-idler group velocity mismatch required us to consider other explanations for the behavior. We first considered the traditional approach in the literature for describing the bandwidth of phase matched processes. Within this approach, the phase matching bandwidth is defined according to the Taylor series expansion of the phase mismatch.[125, 129, 134, 170, 179, 265, 277, 292] Such an expansion is typically considered to predict that the phase matching bandwidth depends solely upon the signal-idler group velocity mismatch,[125, 129, 134, 170, 179, 265, 277, 292] though it has been noted that this approximation is only valid within the confines of a monochromatic or nearly monochromatic pump.[170, 172] Despite concerns that the accuracy of such approaches is limited in the regime of broad bandwidth pulses,[175] within our treatment, we find that the bandwidth predicted by a similar method yields reasonable results for some regions in the sum-frequency idler process (Fig. 4.12).

As noted above, the traditional view of the phase matching bandwidth is to perform the Taylor series expansion of the phase mismatch,  $\Delta k = k_P - k_I - k_S$ , with respect to the signal and idler frequencies:

$$\Delta k = \Delta k_0 + \left(\frac{\partial k_S}{\partial \omega} - \frac{\partial k_I}{\partial \omega}\right) \Delta \omega + \frac{1}{2} \left(\frac{\partial^2 k_S}{\partial \omega^2} - \frac{\partial^2 k_I}{\partial \omega^2}\right) \Delta \omega^2 + \cdots$$
(4.3.9)

where the difference in the group velocities or group velocity dispersions arises because an increase (decrease) in the frequency of the signal must be accompanied by a decrease (increase) in the idler frequency. With the decrease in signal according to  $\operatorname{sinc}^2(\Delta kL/2)$ , the phase matching bandwidth is typically defined as the bandwidth,  $\Delta \omega$ , for which  $|\Delta k| = \pi/L$ , where L is the crystal length.[125] Traditionally, this is only considered to the first order (only considering group velocities), with the group velocity dispersion taken into consideration when the group velocity mismatch is zero.[126, 170] As a first approximation, we extended this approach to consider all possible permutations of the fields in both processes (pump and signal, pump and idler, and signal and idler in optical parametric amplification and sum-frequency and pump, sum-frequency



Figure 4.11: The results of simulations of sum-frequency idler generation and comparison to the group velocity mismatch curves calculated by the dispersion curves from Zhang et al. [208] (A) reproduces the two dimensional presentation of the simulated sum-frequency idler (see (A) in Fig. 4.10). (A) also contains an overlay (white markers) indicating the sum-frequency idler pulse energy predicted by the simulations. (B) and (C) show standard deviation ( $\sigma$ ) and skew ( $|\alpha_3|$ ) calculated from the simulation in (A); these values are calculated according to Eq. 4.3.6 and 4.3.7, respectively. (D) displays the calculated group velocity mismatch between the pump and either signal or idler based upon the dispersion formulae of Zhang et al. [208] The faded red line in (C) emphasizes zero, identifying the point at which symmetric spectra should be observed. The faded green lines in (B), (C), and (D) mark where the pump-idler group velocity mismatch is zero.

and idler, and pump and idler in sum-frequency generation), generalized in Eq. 4.3.10 to any i,j pair:

$$\Delta k = \Delta k_0 + \left(\frac{\partial k_i}{\partial \omega} - \frac{\partial k_j}{\partial \omega}\right) \Delta \omega + \frac{1}{2} \left(\frac{\partial^2 k_i}{\partial \omega^2} - \frac{\partial^2 k_j}{\partial \omega^2}\right) \Delta \omega^2$$
(4.3.10)

Explicitly considering the expansion to the second order, we defined the phase matching bandwidth according to the maximum bandwidth allowed in any given process (optical parametric amplification or sum frequency generation) based upon the limit of  $|\Delta k| = \pi/L$ .

From an analysis of the predicted phase matching bandwidth, we found that the predicted bandwidth was limited by the optical parametric amplification process; based on similar situations described in the literature, we would therefore expect the bandwidth of the paired processes to be limited by the bandwidth predicted for optical parametric amplification.[170, 176, 285] In this first process, we found that the bandwidth permitted based upon the group velocity mismatch between the pump and either the signal or the idler was actually significantly greater than the limit imposed by the signal-idler group velocity mismatch. At shorter sumfrequency idler wavelengths, the maximum bandwidth was permitted by the pump-signal group velocity mismatch; this eventually gave way to the bandwidth allowed by the expansion of the phase mismatch with respect to the pump and idler. In Fig. 4.12 (B), we compare the maximum allowed bandwidth within this approximation to the bandwidth predicted by the second moment of the simulated spectra. We find that this approximation actually describes the bandwidth at shorter sum-frequency idler wavelengths quite well; however, while the maximum bandwidth predicted by this analysis is comparable to the maximum bandwidth observed in our simulations, we still observe that the distortion to the sum-frequency idler spectra occurs at different wavelengths than would be expected.

The theoretical treatment of Akhmanov et al for optical parametric amplification and sum-frequency generation appears to provide the best qualitative match to the behavior we observe in our simulations.[103] Unfortunately, we are restricted to a qualitative comparison. In their analysis of optical parametric amplification, Akhmanov et al assume equal pump-idler and pump-signal group velocity mismatch, while it is not clear whether their treatment is robust in the traveling-wave regime of high conversion.[103] On the other hand, their treatment of sum frequency generation is explicitly in the context of low conversion and does not incorporate the effects of pre-existing group delay.[103] Nevertheles, their models have been previously used as qualitative descriptions for the behavior of femtosecond parametric processes,[175, 176] as we shall do here.

Akhmanov et al describe optical parametric amplification within the typical regime of Type II phase



Figure 4.12: A comparison of the sum-frequency idler simulation to the estimated phase matching bandwidth. (A) reproduces the two-dimensional presentation of the simulation shown in Fig. 4.10 (A). (B) compares the bandwidth as calculated according to the second moment (see Eq. 4.3.6) to the maximum phase matching bandwidth as estimated by considering all permutations of Eq. 4.3.10.

matching - in which  $\Delta u_{i,p}^{-1} \Delta u_{s,p}^{-1} < 0, [126, 171]$  where  $\Delta u_{j,k}^{-1}$  is the group velocity mismatch between j and k,  $u_j^{-1} - u_k^{-1}$  - compatible with permitting the "mode amplification" regime.[103] They consider amplification under such conditions to be a steady state mode of operation, though the behavior of the process changes based upon the characteristic lengths. The lengths of relevance are the pulse-splitting (or walk-off) length,  $L_w$ , and the amplification length,  $L_a$ . The pulse-splitting length can be defined in terms of the pulse duration of the pump:  $L_w = \tau_P / |\Delta u^{-1}|$ , where  $\tau_P$  is the pump pulse duration and  $\Delta u^{-1}$  is its group velocity mismatch with respect to another pulse. However, they indicate that it is more robust to consider the pulse-splitting length with respect to the bandwidth  $(\Delta \omega)$  of the interacting pulses:

$$L_w = (|\Delta u^{-1}|\Delta \omega)^{-1} \tag{4.3.11}$$

Meanwhile, the amplification length is related to the exponential gain observed in the amplification process: for exponential gain of the signal on the amplitude level according to  $\exp(\Gamma z)$ , the amplification length is  $L_a = \Gamma^{-1}.[103]$ 

From the ratio of  $L_w$  and  $L_a$ , Akhmanov et al define a quantity,  $m = L_w/L_a$ . When m < 1/2, exponential gain of the signal and idler is observed in the amplification process; they also predict that this steady state operation contributes to generating pulses with durations  $\tau = \tau_P/\sqrt{\Gamma L}$ , where  $\tau_P$  is the pump pulse duration and  $\Gamma$  is the gain.[103] Such behavior is consistent with the slight pulse shortening that has been reported for Type II phase matching.[168] On the other hand, when m > 1/2 and the crystal is longer than the pulse-splitting width, exponential gain is still observed; however, the amplified pulses exhibit a decrease in duration,  $\tau = \tau_P/m$ . In both steady state regimes, the group delay accumulated is  $\tau_P/2m$ .[103]

Meanwhile, for sum-frequency generation, Akhmanov et al provide two different regimes in which sumfrequency generation can occur, with different results for the spectrum of the upconverted field. The difference between the regimes stems from the pulse-splitting length of the fields being summed; Akhmanov et al label these fields 1 and 2, with  $u_1 > u_2$ . The pulse-splitting length is defined according to the bandwidth of field 1,  $L_w = (|\Delta u_{2,1}^{-1}|\Delta \omega_1)^{-1}$ . When the crystal length is less than the pulse splitting length, the separation of fields 1 and 2 is considered inconsequential, and the spectrum depends upon the group velocity of the sum frequency with respect to these fields,  $\Delta u_{s,0}^{-1}$ :

$$s(\Delta\omega) = \operatorname{sinc}^{2}\left(\frac{\Delta u_{s0}^{-1}\Delta\omega L}{2}\right)s^{qss}(\Delta\omega)$$
(4.3.12)

where  $\Delta \omega$  is the detuning and  $s^{qss}(\Delta \omega)$  is the steady-state spectrum. We take that this spectrum should correspond to the ideal spectrum of the sum frequency, namely the convolution of the spectra of fields 1 and 2. Meanwhile, when the crystal length is longer than the pulse splitting length, a noticeably different result is predicted:

$$s(\Delta\omega) = \left[1 + \left(\frac{\Delta\omega\Delta u_{s,2}^{-1}}{\Delta\omega_1\Delta u_{2,1}^{-1}}\right)^2\right]^{-1} s^{qss}(\Delta\omega)$$
(4.3.13)

where all terms in Eq. 4.3.13 are defined as before.[103]

From Eq. 4.3.12 and 4.3.13, it is trivial to show that the sum frequency process narrows the steady-state spectrum according to a bandwidth<sup>i</sup> of  $\Delta \omega_{40\%} = \pi/L \Delta u_{s,0}^{-1}$  when  $z < L_w$  or  $\Delta \omega_{40\%} = \frac{1.5 \Delta \omega_1 \Delta u_{2,1}^{-1}}{\Delta u_{s,2}^{-1}}$  when  $z > L_w$ . Alternatively, for  $z > L_w$ , this result is identical to  $\Delta \omega_{40\%} = \frac{1.5}{L_w \Delta u_{s,2}^{-1}}$ .

In Fig. 4.13, we consider the sum-frequency idler simulations in the context of the theoretical treatment by Akhmanov et al. In (A), we see that the bandwidth of the sum-frequency idler process ( $\sigma_{SFI}$ ), is poorly described by the convolution of the idler and pump bandwidths:<sup>ii</sup> the convolution exhibits a broader width at short sum-frequency idler wavelengths ( $\lambda < 575$  nm), a narrower bandwidth at intermediate wavelengths ( $\sim 575$  nm  $< \lambda < 595$  nm), and reaches a peak at longer wavelengths than is observed for the sumfrequency idler. However, in (B), we see that the bandwidth that is expected to be imposed according to Eq. 4.3.12 (magenta markers) and Eq. 4.3.13 (cyan markers) exhibit very similar behavior to the sumfrequency idler bandwidth at shorter sum-frequency idler wavelengths. At longer wavelengths, though, their behaviors diverge: whereas the bandwidth predicted by Eq. 4.3.12 continues to increase with increasing sumfrequency idler wavelength, the bandwidth predicted by Eq. 4.3.13 exhibits a peak at approximately the same sum-frequency idler color as the peak observed in the bandwidth of the sumfrequency idler simulation in (A).

This change in the behavior predicted by Eq. 4.3.13 is readily apparent from the calculated pulse-splitting lengths in (E): at the same sum-frequency idler wavelength that we observe the peaks in the bandwidth of the simulated sum-frequency idler and the bandwidth predicted by Eq. 4.3.13, the pulse-splitting length is beginning to increase dramatically. This increase in the pulse-splitting length occurs despite the increased pump and idler bandwidths due to the rapid change in pump-idler group velocity mismatch near these frequencies (see (D) in Fig. 4.11). The increase in  $L_w$  also causes a sharp rise in m, as shown in (C); in calculating m, we have estimated  $L_a$  using a value of 1.8 mm<sup>-1</sup> for  $\Gamma$ . A value of 1.8 mm<sup>-1</sup> for  $\Gamma$  corresponds roughly to amplifying a 1 µJ seed to 400 µJ; the value of  $\Gamma$  changes relatively slowly, though, for different

<sup>&</sup>lt;sup>i</sup>Here, we are defining bandwidth as the detuning that results in a decrease in intensity to 40%, the losses induced by the sinc<sup>2</sup> function when the argument in brackets is equal to  $\pi/2$ .[125]

<sup>&</sup>lt;sup>ii</sup>Here, we are restricted to considering the bandwidth of the convolved spectra within the limit that the spectra are Gaussian, namely that the variance of the convolution is the sum of the variances of the initial spectra. However, we note that both the pump and idler exhibit some degree of skew at longer sum-frequency idler wavelengths (not shown), so the bandwidth predicted in this manner is not expected to be exact.

final signal pulse energies.<sup>i</sup> Finally, we see in (D) that the lower values and gradual changes associated with m at shorter sum-frequency idler wavelengths result in the largest values and only slight changes in the group delay,  $\tau_g$ , between the pump and idler. When the pulse-splitting length increases dramatically at longer sum-frequency idler wavelengths, we see a correspondingly fast decrease in  $\tau_g$ .

Combining these results, we interpret the distortions observed in the sum-frequency idler spectra in the following manner. At short sum-frequency idler colors, we do not observe any unexpected behavior during parametric amplification, yielding reasonably symmetric signal and idler pulses (see Fig. 4.9). However, by operating at lower values for m at these wavelengths, the pump and idler accumulate the greatest initial group delay. This large initial group delay reduces the efficiency of the conversion, yielding lower output energy for the sum-frequency idler process. As we tune to longer sum-frequency idler colors, the decreasing group velocity mismatch between the sum frequency and the idler/pump (see the behavior for  $z < L_w$  in (B) of Fig. 4.13) increases the bandwidth of the sum-frequency idler according to Eq. 4.3.13. However, while we interpret the trends in Fig. 4.13 (B) as arising predominantly from the changes in the group velocity mismatch between the sum frequency and the pump/idler, there are also more gradual changes in the pumpidler group velocity mismatch. This gradually increases  $L_w$ : while not enough to overcome the changes in the sum frequency-pump/idler group velocity mismatch, it does result in a decrease in the group delay by increasing m, improving the conversion efficiency. Eventually, the pump-idler group velocity mismatch does change sufficiently quickly to overcome the decreasing sum frequency-pump/idler group velocity mismatch. At this point, we simultaneously observe: a sudden decrease in the initial group delay, improving the efficiency of the sum frequency process; and, a sudden decrease in the bandwidth predicted by Eq. 4.3.13, due to the increase in  $L_w$ . Eventually, the increase in  $L_w$  is sufficient to transition from the  $z > L_w$  regime to the  $z < L_w$  regime: this results in the sum-frequency idler bandwidth being limited by the convolution of the idler and pump bandwidths, which appears to be the trend at the longest sum-frequency idler wavelengths in Fig. 4.13 (A).

The proposal from Light Conversion to address the observed issues with sum-frequency idler generation is an upgrade that replaces the "depleted pump" upconversion strategy with one based upon a "fresh pump." [194] Within the "fresh pump" strategy, the existing pump is split prior to parametric amplification: one fraction is used for amplification of the signal and the idler, while the other is routed through an external delay and recombined with the signal and idler for upconversion. Based upon the results of these simulations and previous reports in the literature, it seems reasonable to conclude that this would improve

<sup>&</sup>lt;sup>i</sup>For example, a final pulse energy of only 180  $\mu$ J corresponds to  $\Gamma \approx 1.4 \text{ mm}^{-1}$ .



Figure 4.13: Interpretation of the sum-frequency idler simulation results in the context of the theory presented by Akhmanov et al.[103] (A) Comparison of the bandwidth observed for the simulated sum-frequency idler, the simulated pump, the simulated idler, and the square root of the sum of the variances of the simulated pump and idler. (B) Calculated bandwidth (at the 40% intensity level) of sum frequency idler generation for  $z < L_w$  (magenta markers) and for  $z > L_w$  (cyan markers) according to the theory of Akhmanov et al. See Eq. 4.3.12 and 4.3.13. (C) Estimation of  $m = L_w/L_a$  according to the bandwidths observed from the sumfrequency idler simulations.  $L_w$  is the pulse-splitting length and  $L_a$  is the amplification length for  $\Gamma = 1.8$ mm<sup>-1</sup>. (D) Estimated group delay between the pump and idler accumulated during simulated parametric amplification, as calculated  $\tau_g = 35/2m$  fs. (E) Estimated pulse-splitting length in the sum frequency crystal according to the bandwidths calculated in sum-frequency idler simulations.

the performance of the sum-frequency idler process with respect to conversion efficiency.[103, 126] However, from the conclusions described above, it is not clear that compensating for the group delay would improve the spectral quality of the sum-frequency idler: while it seems reasonable to attribute the inconsistent conversion efficiency to the accumulated group delay, it appears more likely that the spectral character of the sum-frequency idler is controlled by the interplay of the group velocities of the sum-frequency, the pump, and the idler. It is not clear that the delay compensation incorporated in this "fresh pump" strategy would improve the spectral quality. While simulations could be performed to explore this behavior, it would also warrant communication with Light Conversion to determine what behavior they observe and anticipate from the proposed upgrade.

Additionally, while the sum-frequency signal process generally does not exhibit the asymmetric line shapes that are observed from the sum-frequency idler process, our experimental work with the sum-frequency signal process has led us to suspect that similar effects impact its time-bandwidth product. At this time, there has not been the opportunity to experimentally explore these suspicions, but it should be straightforward and more convenient to take the time to explore this possibility *in silico*.

- CHAPTER 5 -

Manganese Tetraphenylporphine

### 5.1 Background

In various efforts to characterize photosystem II, a variety of authors have reported evidence of activity from the protein under near infrared (NIR) excitation. [15, 17–21, 34, 39, 44, 45, 61] Early detection of this signature was reported by Dismukes and Mathis, [39] but this signature has since been characterized further. Boussac et al published a sampling of reports in which they observe a change in the electron paramagnetic resonance (EPR) signal associated with the oxygen evolving center upon NIR exposure.[15, 17–21] By determining the yield of signal conversion, they have been able to trace out an activity spectrum exhibiting a peak near 820 nm. Baxter et al extended this to directly detect the absorbance of photosystem II, tracking the NIR transient absorbance upon the  $S_2 \rightarrow S_1$  dark-adaptation of a flash-excited sample.[34] The spectral profile they determine is very consistent with the activity spectrum reported by Boussac et al. Though their data set is limited, they find their data to be consistent with a Gaussian fit with center of 12300 cm<sup>-1</sup> (813 nm) and half-width of approximately 1000 cm<sup>-1</sup>. From estimating the chlorophyll content of their sample, Baxter et al estimate a transient absorptivity of 50 cm<sup>-1</sup> M<sup>-1</sup>. Cua et al provide additional evidence for NIR activity through the observation of enhancement to low-frequency Raman modes with excitation near 820 nm; this enhancement is reduced when the excitation color is changed and is absent upon illumination or manganese depletion.[61] Most recently, Morton et al have published the results of NIR absorption and magnetic circular dichroism of photosystem II.[44] The combination of absorbance and MCD data allow Morton et al to make the strongest case for the origin of this feature, attributing the weak, with absorptivity  $< 30 \text{ cm}^{-1} \text{ M}^{-1}$ , absorbance found near 770 nm to d-to-d transitions localized on a manganese of the oxygen evolving center. [44]

The literature on the spectroscopy of manganese (III) provides support for this assignment. Many high-spin, monovalent coordination complexes of manganese (III) exhibit transitions in the NIR region, with different coordinating ligands resulting in NIR transitions at transition energies as low as 5500 cm<sup>-1</sup> (1800 nm) and as high as 14000 cm<sup>-1</sup> (714 nm).[293, 294] These transitions are characterized by very low absorptivities ( $\leq 100 \text{ cm}^{-1} \text{ M}^{-1}$ ), with the absorptivity generally decreasing with increasing transition energy.[294] The origin of these transitions is traced to the Jahn-Teller effect: due to its d<sup>4</sup> electron configuration, six-coordinate manganese (III) centers are inclined to exhibit a distortion from octahedral symmetry in order to reduce the overall energy of the ground state.[293–299] The high resolution crystal structure of photosystem II has been interpreted to support the presence of Jahn-Teller distortions in the oxygen-evolving complex.[64]

As a representative system, manganese (III) trisacetylacetonate,  $Mn(acac)_3$ , was selected as an initial test sample due to the convenient position of its NIR transition at the shorter wavelengths of the signal process in the TOPAS-C optical parametric amplifiers and its availability from commercial sources. Unfortunately, experiments with this sample proved unsuccessful, with no signal clearly distinguishable in either transient absorption or transient grating experiments. Coming across the work of McCusker and co-workers, it became clear that a priori knowledge of the photophysics, or the ability to ascertain such information through methods such as white-light transient absorption, would be key in guiding future experiments.[300–302] These publications also traced the photophysics of a species,  $Cr(acac)_3$ , with an absorptivity of very similar magnitude to the Jahn-Teller-allowed transition of manganese (III) coordination complexes. However, even in efforts to reproduce the reported results, we proved unable to replicate them: while very weak signal may have been present in transient absorption geometries, transient grating experiments yielded no signal.

At this point, it seemed probable that the low transition dipoles indicated by the low absorptivity of  $Mn(acac)_3$  and  $Cr(acac)_3$  may be proving limiting. Porphyrins and phthalocyanines seemed a convenient starting point for samples with stronger transitions. While manganese phthalocyanines seemed promising due to the presence of a NIR transition and evidence that the macrocycle and transition metal states exhibited significant interaction, [303–306] it became clear that manganese phthalocyanines also did not exhibit satisfactory stability in solution.[307] However, manganese (III) porphyrins also exhibit strong interaction between the transition metal and macrocycle states: the manganese d-orbitals are believed to be close in energy to the  $\pi$  and  $\pi^*$  states usually responsible for the Soret and Q-band in normal porphyrins. [306, 308, 309] The coupling between the metal and macrocycle states yields several transitions throughout the visible; [308] while it is not clear a definitive assignment of these transitions exist, it seems to be accepted that the states of the manganese (III) are involved in the spectroscopy. [308–311] While manganese (III) porphyrins generally exhibit little to no luminescence, [312] a variety of reports have explored the photophysics of manganese (III) tetraphenylporphine (MnTPP) following excitation with both pico- and femtosecond pulses. [313–316] The behavior observed in these experiments has been attributed to excitation of the  ${}^{5}S_{0}$ ground state to one of various  ${}^{5}S_{n}$  excited states, followed by an intersystem crossing to a "tripmultiplet" manifold including <sup>3</sup>T<sub>1</sub>, <sup>5</sup>T<sub>1</sub>, and <sup>7</sup>T<sub>1</sub>.[313, 315, 316] The features observed in transient absorption upon excitation consistently include bleaching signals associated with the ground state features near 480 nm, 560 nm, and 600 nm, and excited state absorption near 500 nm.[314–316] While some reports suggest MnTPP can undergo photoreduction in tetrahydrofuran or be coordinated by solvents such as pyridine, [308, 317] there did not appear to be other concerns regarding stability; repeated measurements of solutions prepared from commercially-obtained MnTPP exhibited no spectral evidence of changes over time when dissolved in dichloromethane.

Given the exhibited stability and documented spectral signatures, we proceeded to adopt MnTPP as a

test sample. We hoped to be able to reproduce the transient absorbance results - with white light if feasible - and to extend the study of this molecule to transient grating geometries. We successfully reproduced the white light transient absorption for MnTPP. While we were able to observe transient grating signal, we found the nonresonant response to provide a competitively strong background. We also were unable to extend the use of white light to transient grating: though we could detect some signal in a nonresonant medium, signal levels were insufficient to detect with the existing system configuration. A sampling of results are presented in this chapter.

## 5.2 White Light Generation

White light generation provides a convenient method for obtaining broadband radiation from an ultrafast laser pulse; here we will provide a brief overview of the background on white light generation and describe the procedure used to generate white light for the reported experiments.

When generated from the fundamental of a  $Ti:Al_2O_3$  laser (near 800 nm), the white light will include new frequencies in both the near infrared and the visible, potentially spanning the entire visible region. [318, 319] The introduction of these new frequencies is attributed to self-phase-modulation and other nonlinear processes that arise due to the high intensities within filaments; [318, 320] these processes maintain coherence between the white light and the original field.[132, 321–326] Filamentation of the beam appears necessary to achieve the sustained intensity needed for white light generation. The formation of a stable filament is attributed to the interplay of focusing and defocusing mechanisms. The initial focusing of the laser arises through the Kerr effect, in which the intensity distribution of the laser modulates the refractive index, thereby causing the beam to be lensed: the significance of this contribution is clear by the threshold of white-light generation matching the threshold for Kerr lensing. [318, 320, 327–335] However, the effects of self-focusing eventually result in an intensity sufficiently high so as to induce defocusing; from this point, the filament is maintained by a sustained balance between the Kerr lensing and this defocusing. The traditional view for the mechanism appears to attribute defocusing to plasma generation; more recently, other authors have suggested that defocusing through plasma generation is specific to longer pulses, while the focusing of shorter pulses is arrested by higher order contributions to the Kerr effect. [234, 320, 326, 327, 330, 334–336] While condensed media provide a convenient route to reliably generate white light, [318–320, 325, 328, 337] some authors have reported filamentation and white light generation in air.[330-332, 338-341]

Broadening of the pulse during white light generation results in new frequencies at both low (Stokes broadening) and high (anti-Stokes broadening) frequencies; the breadth of the broadening can be described

by the derivative of the accumulated nonlinear phase due to self-phase-modulation with time (see § 3.1.2).[320] Due to the relationship between phase- and frequency-modulation, the Stokes and anti-Stokes components are attributed to the accumulated nonlinear phase at the leading and trailing edges of the pulse, respectively.[320, 342] The accumulation of nonlinear phase can be described by a quantity called the *B*-integral.[172, 247, 254, 320] Since they experience different contributions from the temporal envelope of the laser, the Stokes and anti-Stokes components are generally not symmetric. The anti-Stokes component is generally more enhanced and exhibits a characteristic plateau - an even energy distribution across the observed frequencies.[320, 328] Within the approximation that the driving field can be described by a sech field envelope, a closed form solution for the breadth of the anti-Stokes and Stokes components has been published.[320] While even energy distributions of the Stokes component have been reported, this behavior appears to be specific to particular media and is only observed at higher energies.[328] Such high energies can also induce modulations in the spectra.[328]

In addition to the differences between the Stokes and anti-Stokes components, the medium and experimental approach can also impact the breadth of the white light. In condensed media, correlation has been reported between the breadth of the anti-Stokes broadening and the band gap of the material: larger band gaps tend to support additional broadening. This behavior also translates to the wavelength of the initial beam, as higher frequency lasers exhibit enhanced broadening.[319] This band gap dependence results in halide crystals, such as LiF and  $CaF_2$ , yielding far broader white light spectra than other condensed media.[319, 320] However, this advantage comes at a limitation, as these materials have been reported to be susceptible to damage when used for white light generation.[138, 178, 337] While the fluoride materials can be particularly susceptible to damage at the intensities in filaments, care should be taken regardless of the medium: at high numerical aperture (or short focal lengths), the threshold for optical breakdown can be lower than the threshold for filamentation.[333, 343]

In order to apply white light in experiments on the femtosecond system, the system described by Nikolaitchik et al was used as a model.[344] Like the Wright group's femtosecond system, the system described is seeded by a Millenia-pumped Tsunami, seeding a Spitfire that yields 100 fs (FWHM) pulses. They report using a fraction, approximately 0.2-0.3 mJ, of the fundamental to generate white light in a 3 mm sapphire plate or a 1 cm flow cell of ethylene glycol.[344] It is noted here that this number seems suspiciously high, as most sources cite threshold energies for white light generation in condensed media at hundreds of nJ to single  $\mu$ J.[322, 323, 328, 333, 343] More consistent with the literature, they report generating the white light by focusing the mode of the fundamental into the white light medium with a 100 mm focal length lens.[344] The documentation on the Spitfire Pro XP employed on the femtosecond system reports an expected  $1/e^2$  beam diameter of 7-9 mm. If this is a reasonable estimate for the Spitfire employed by Nikolaitchik et al, their 100 mm focal length lens should result in a half-width waist within their white light medium of approximately 3  $\mu$ m,[105] and a numerical aperture of approximately 0.03.[209] This value is consistent with the regime reported by Ashcom et al for which the threshold for continuum generation is lower than the threshold for damage to the medium.[343]

At the time of these experiments, the mirrors and beamsplitters dividing the fundamental from the Spitfire Pro XP also isolate a small fraction of the fundamental. The spot size of this beam was reduced by a Galilean telescope consisting of a f = -50 mm and  $f \approx 100$  mm lens pair. The resulting beam was routed down the table by a series of mirrors; it could be configured to pass through one of the delay lines for use in three color experiments. After the telescope, the beam diameter is approximately 6 mm, and the telescope and other optics introduce losses that reduce the energy to approximately 20 µJ. Starting from the values reported by Nikolaitchik et al, we expected a 50-75 mm lens to yield the correct imaging properties, but we unfortunately found ourselves limited by the lenses available. While the 35 mm lens employed yields a numerical aperture of approximately 0.15, exceeding the numerical aperture for which the damage threshold is below the threshold for continuum generation in silica, [343] we found we could consistently generate and maintain a white light continuum in sapphire. In order to attain stable white light, we found we needed to attenuate the fundamental with an optical density of approximately 3.<sup>i</sup> We also attempted to generate the continuum in  $CaF_2$ : consistent with the literature,  $CaF_2$  yielded a much broader continuum than sapphire.[178, 319, 328] Unfortunately, the continuum could only be sustained for a short period, which we attribute to damage: at this point, we suspect that the stability of sapphire over the  $CaF_2$ may be consistent with the behavior of halide crystals as reported by Huber et al, Kohl-Landgraf et al, and Tzankov et al. [138, 178, 337] White light generation was performed just before the appropriate beam line; the sapphire plate was followed by a 25 mm lens, which was adjusted to yield the proper spot size at the sample.

### 5.3 Experimental Deviations

Experiments with MnTPP were performed by dissolving the sample in dichloromethane to achieve an appropriate absorbance,  $A \approx 1$ , in the spectral region of interest; this was often the frequency of the pump (in transient absorption) or  $\omega_2/\omega_{2'}$  (in transient grating). Samples were contained in a 1 mm glass cuvet

 $<sup>^{</sup>i}$ It should be noted that this optical density is based upon the calibration reported by the manufacturer. Calibration has shown that the optical density of the filters exhibits some wavelength dependence, so the actual optical density is likely slightly different.

(Precision Cells).

For experiments not involving white light, standard operating procedures were employed for the experiments described in this section. Experiments were conducted using the second-harmonic and sum-frequency signal processes of the TOPAS-C. The supplied wavelength selector was used for the sum-frequency signal process. For second-harmonic signal, the wavelength selector was not used, but a heat filter was introduced to block the near infrared (wavelengths longer than  $\sim$  700-740 nm).

For experiments involving white light, white light was generated as described before, and the table was aligned as described for two color experiments: the fundamental and white light served as  $\omega_1$ , and one of the OPAs was used to provide  $\omega_2$  and  $\omega_{2'}$ . For transient absorption experiments, signal could be detected when using the standard f = 1 m mirror; however, the beamsplitter for the OPA had to be removed in order to provide sufficient pump energy. The silicon photodiode was used for detection, and the monochromator was scanned across the wavelength range of the white light (the probe). Zero delay compensation was implemented to reduce the effects of the chirp in the white light. Due to the exploratory nature of these scans, the chirp in the white light was not determined with sufficient resolution to characterize the nonlinearities in the chirp rate; however, the compensation was sufficient to ensure zero delay for all colors in the probe occurred within  $\sim 100$  fs of the assigned zero delay. For transient grating with the white light, no signal could be detected with the f = 1 m mirror; however, by decreasing the focal length to f = 500 mm, some signal was detected. This signal was strongest when  $\omega_1 \approx \omega_2$ ; the decrease in signal at longer wavelengths is believed to be a consequence of phase mismatch and changes in output direction. With the current detector (silicon photodiode), signal could only be detected in solvent. It is suspected that a more sensitive detector and modifications to the table layout to provide more a more intense white light and retain more energy from the OPA would be needed to make this method routine.

# 5.4 Results and Discussion

A summary of experimental data exploring the nonlinear electronic spectroscopy of MnTPP is provided in Fig. 5.1. (A)-(C) provide insights into the behavior of MnTPP in transient absorption along two different slices of the three-dimensional space: (A) and (B) present the change in optical density as the pump is tuned and the monochromator is scanned across the wavelengths of the white light probe; and, (C) explores the dynamics along delay at different probe wavelengths. (A) and (B) provide two alternative views of the same data set, while the data of (C) is shown only with the more dynamic color scale of (B). (D)-(H) explore the response of MnTPP in frequency-frequency and delay-delay space for transient grating/TriEE, the frequencydomain analogue of 2D-ES,[345] corresponding to the phase-matched output  $\mathbf{k}_4 = \mathbf{k}_1 - \mathbf{k}_2 + \mathbf{k}_{2'}$ . In the frequency-frequency domain of (D), the detected intensity at  $\omega_m = \omega_1$  is presented as a function of  $\omega_2 = \omega_{2'}$ and  $\omega_1 = \omega_m$ . The time-time domain of (E)-(H) show the detected intensity at  $\omega_m = \omega_1$  as a function of the relative delays  $\tau_{21} = \tau_{2'1}$  and  $\tau_{2'2}$ ; in this domain, signal along  $\tau_{21} = \tau_{2'1} \leq 0$  tracks the population dynamics of the system, while signal along other directions track the dynamics of coherences under the conditions of these experiments ( $\omega_1 \neq \omega_2$  and  $\omega_2 = \omega_{2'}$ ).

The spectroscopy of MnTPP observed in Fig. 5.1 is consistent with previously described transient absorption of this compound.[314–316] In solution,<sup>i</sup> all authors<sup>ii</sup> report a strong excited state absorption (ESA) feature peaking near 490-500 nm and a weak ESA feature near 550 nm.[314–316] The broader probe of Kim et al allow them to also report an ESA feature between 400 and 450 nm;[315] they also show agreement with Yan et al in reporting a broad ESA feature starting near 650 nm and falling off into the near infrared.[315, 316] The ground state absorption of MnTPP is bleached, with a strong feature near 480 nm[314, 315] and weaker features near 580 nm and 620 nm.[315, 316]

Within the sensitivity of the white light transient absorption measurements, we observe each of these features. The bleach near 480 nm is the most pronounced feature of the transient absorption spectrum, consistent with Kim et al.[315] The contours in (A) hint at features corresponding to the ESA features reported between 400 and 470 nm and 495 and 530 nm and the bleaches of the ground state transitions near 580 nm and 620 nm. The alternative color bar in (B) accentuates these features. The scale also provides sufficient contrast to observe features consistent with the ESA near 550 nm and from 650 nm into the near infrared. The consistency of these features in (A), (B), and the cited works indicate that the transient absorption of MnTPP is relatively insensitive to excitation wavelength: Yan et al tune the pump to 355 nm or 532 nm to observe features between 500 and 650 nm or 650 and 900 nm, respectively;[316] Kim et al use a frequency-doubled Ti:sapphire amplifier at 390 nm;[315] Irvine et al appear to use a 597 nm dye laser;[314] and (A) and (B) illustrate the changes in transient absorption while scanning across the sum-frequency signal process (470-515 nm). This is consistent with previous assignments in the literature: excitation of the singlet states is followed by rapid relaxation to the lowest excited singlet state, which undergoes rapid intersystem crossing to yield states in the "tripmultiplet" manifold.[313, 315, 316] Consequently, despite the different

 $<sup>^{</sup>i}$ It should be noted that Kim et al also study the transient absorption of MnTPP in molecular sieves; encapsulation in the molecular sieves results in changes to the ground state absorbance, attributed to interactions between the electronic states of MnTPP and the hydroxyl groups of the sieves.[315] When encapsulated, MnTPP also exhibits different spectral features in transient absorption.

<sup>&</sup>lt;sup>ii</sup>It is noted that the wavelength scale for the data reported by Irvine et al is nearly illegible.[314] Their reported transient absorption spectra exhibit qualitative consistency with the results of Yan et al and Kim et al,[314–316] and from those numbers that can be made out, the features they observe appear to occur at the same wavelengths.



Figure 5.1: (A) and (B): White light transient absorption,  $\lambda_{\text{Pump}}$  versus  $\lambda_{\text{Probe}}$  at  $\tau \approx -500$  fs. B is the same data as A, but with a repeating color scheme to emphasize the weaker features. (C): White light transient absorption,  $\tau$  versus  $\lambda_{\text{Probe}}$  at  $\lambda_{\text{Pump}} = 475$  nm. This is the same repeating color scheme as (B). Optical density of the samples for (A), (B), and (C) are  $A \approx 1$  near 480 nm and  $A \approx 0.1 - 0.2$  near 610 nm. (D): Transient grating,  $\omega_2 = \omega_{2'}$  versus  $\omega_1 = \omega_m$ ,  $\tau_{21} = \tau_{2'1} = -500$  fs. Data in the side bars is absorbance (red) and integrated signal (blue). Optical density of the sample for (D) is  $A \approx 0.3$  near 16100 cm<sup>-1</sup> and  $A \approx 2.5$  near 20900 cm<sup>-1</sup>. (E), (F), (G), and (H): Transient grating,  $\tau_{21}$  versus  $\tau_{2'1}$ ,  $\lambda_2 = \lambda_{2'} \approx 605$  nm and  $\lambda_1 \approx 495$  nm. (E) and (F) provide a comparison of signal from dichloromethane (E) and MnTPP in dichloromethane (F),  $A \approx 0.7$  at both frequencies, near zero delay. (G) and (H) provide a comparison of signal from MnTPP in dichloromethane at two concentrations:  $A \approx 0.7$  (G) and  $A \approx 0.15$  (H). Signal in (E), (F), (G), and (H) are each normalized according to the maximum amplitude in each spectrum.

excitation frequencies, each of these reported experiments probe similar dynamics.

Panel (C) in Fig. 5.1 shows the reported configuration for white light transient absorption is also amenable to collecting the dynamics associated with the features in (A) and (B). Consistent with the literature, [314– 316] the transients of MnTPP are very long lived, exhibiting very little change within the bounds of (C). Unfortunately, it is not possible to compare the dynamics at early times with the literature: the resolution of these experiments is much higher than the > 1 ps time resolution of Irvine et al and Yan et al., [314, 316] and still appreciably higher than the 170 fs resolution reported by Kim et al. [315] The profile of the signal near zero delay illustrates that routine usage of white light transient absorption may be complicated by the challenge in correctly compensating for the chirp in the continuum. Based upon the behavior of MnTPP observed in transient grating, however, some of the complicated dynamics near zero delay in (C) may be attributable to the coherent artifact, which is observed in transient absorption for many different systems. [346–351]

The spectroscopy of MnTPP has been extended to transient grating, in which the signal consistent with the phase matching condition ( $\mathbf{k}_4 = \mathbf{k}_1 - \mathbf{k}_2 + \mathbf{k}_{2'}$ ) is homodyne detected. As shown in (D), the response observed in transient grating is consistent with the behavior in transient absorption and absorption. In the side-plot on the right of (D), the integrated signal as a function of pump frequency ( $\omega_2 = \omega'_2$ ) closely matches the absorbance of the sample: this can be seen by comparing the blue and red traces, corresponding to the integrated transient grating signal and the absorption spectrum, respectively. This behavior is to be expected.[185] Meanwhile, along the pump, we are observing signal corresponding to the ESA feature in (A) and (B); the peak of this signal is near 490 nm, consistent with the peak of the ESA in transient absorption. The sharp drop off in signal is likely the consequence of M-factors,[352] as achieving an appreciable absorbance at  $\omega_2$  required  $A \approx 2.5$  near 480 nm: the decrease in signal from the peak near 20300 cm<sup>-1</sup> matches the increase in absorption from  $A \approx 0.3$  to  $A \approx 1-1.2$ .

Unfortunately, we were limited in the transient grating experiments of MnTPP to only longer delays; for example, the scan in (D) was collected at  $\tau_{21} = \tau_{2'1} = -500$  fs. This limit was imposed by the nonresonant background, as shown in (E)-(H). Between (E) and (F), we compare the dynamics observed at early times for a purely nonresonant system, dichloromethane (E), and a concentrated solution of MnTPP,  $A \approx 0.7$  at  $\omega_1$  and  $\omega_2$  (F). As this comparison shows, MnTPP exhibits a delay in the evolution of signal relative to the purely nonresonant response; we attribute this to the time required for singlet relaxation and intersystem crossing.[313, 315, 316] In contrast, when the concentration of MnTPP is reduced, the transient characteristic of MnTPP, (G), is nearly completely overwhelmed by the nonresonant background (H).

The nonresonant background manifests in transient grating due to the use of homodyne detection. Un-

der such conditions, the signal is proportional to  $|\chi^{(3)}|^2$ ,[190] rather than  $\chi^{(3)}$ . As a consequence, the signal depends upon the real and imaginary components of  $\chi^{(3)}$ , rather than just the imaginary component as in transient absorption;[190, 349] since the nonresonant response of the medium is predominantly real,[353] it is suppressed in transient absorption. Competition with resonant signals arise from scaling: since  $\chi^{(3)}$ is proportional to the number density, N,[190] homodyne-detected signals scale with  $N^2$ ,[148] which can limit the detection limit for such methods.[122, 190, 354, 355] While it has been reported that nonresonant background may be enhanced under conditions of femtosecond excitation,[139, 140, 146, 356, 357] the magnitude of the problem was surprising: previous experiments on the Wright group's femtosecond system with near infrared excitation have been robust at much lower sample absorbance.[358] However, it has also been reported that nonresonant signal can be influenced by excitation wavelength,[140, 148, 359] as well as pulse duration;[359] specifically, nonresonant background has been reported to be much worse under conditions of visible excitation,[140, 148] particularly when two-photon resonances are available.[139, 140]

In order to understand the limit imposed by the nonresonant background, we made measurements aimed at estimating and comparing the nonresonant response at near infrared and visible wavelengths. Beam energies were measured with a thermopile, and nonresonant signal was generated in a variety of transparent solvents. The response of the detector to the nonresonant signal and to one of the excitation beams were compared to provide an estimate of the energy emitted by the nonresonant process; we assumed the sampling efficiency was comparable for both color combinations. These measurements were consistent with a hundredfold increase in  $\chi^{(3)}$  from near infrared excitation ( $\lambda_1 = \lambda_2 = \lambda_{2'} = 1300$  nm) to visible excitation ( $\lambda_1 = 495$ nm and  $\lambda_2 = \lambda_{2'} = 615$  nm). However, all solvents tested<sup>i</sup> exhibited nonlinear susceptibilities on the same order of magnitude. Very low correlation was observed between the measured susceptibilities ( $R^2 = 0.46$ to a linear trendline). Higher correlation ( $R^2 = 0.75$ ) was observed when the aromatic molecules (benzene and pyridine) were excluded: this may be consistent with the observation that the nonresonant nonlinear susceptibility is enhanced when two-photon resonances are available,[139, 140] as the aromatic compounds would be expected to exhibit the lowest transition energies.

Beyond this limit from higher nonresonant susceptibilities, the previous electronic experiments were able to achieve similar absorbance over the same pathlength at much lower concentrations than our MnTPP samples: these experiments would be performed at concentrations of tens of nM[358] rather than hundreds of  $\mu$ M. It can be shown that this 10<sup>4</sup> reduction in concentration would be expected to increase the nonlinear susceptibility by the same factor. If we generalize this factor as  $\phi$ , this relationship can be shown by

 $<sup>^{\</sup>rm i}$  The solvents tested were acetone, acetonitrile, benzene, carbon tetrachloride, dichloromethane, methanol, octane, and pyridine.

recognizing that the same absorbance over the same pathlength implies  $\alpha$  to be the same for both samples. However,  $\alpha$  can be related to both the optical cross-section,  $\sigma = \alpha/N$ , and to the optical properties of a hypothetical ba transition,  $\alpha \propto N\mu_{ba}^2/\Gamma_{ba}^2$ ,<sup>i</sup> where  $\mu_{ba}$  and  $\Gamma_{ba}$  are the transition dipole and dephasing rate of the transition, respectively.[99] From the former relationship, we can see that a reduction in N by a factor of  $\phi$  must be accompanied by an increase in  $\sigma$  by a factor of  $\phi$  to maintain constant  $\alpha$ . Second, by combining the two relationships,  $\sigma \propto \mu_{ba}^2/\Gamma_{ba}^2$ , and the factor of  $\phi$  increase in  $\sigma$  implies  $\mu_{ba}/\Gamma_{ba}$  must increase by a factor of  $\sqrt{\phi}$ . However,  $\chi^{(3)}$  is proportional to  $N\mu_{ba}^4/\Gamma_{ba}^4$ ,<sup>ii</sup>[99] the factor of  $\phi$  decrease in N and factor of  $\sqrt{\phi}$  increase in  $\mu_{ba}/\Gamma_{ba}$  then suggests  $\chi^{(3)}$  should be increased by a factor of  $\phi$ . Thus, since the homodynedetected signal scales with  $|\chi^{(3)}|^2$ ,[99, 190] these visible experiments with MnTPP not only require resolving against a background that is 10<sup>4</sup> times stronger, but seek to detect signals that are weaker by a factor of  $\phi^2$ , corresponding to a factor of 10<sup>8</sup> when comparing MnTPP to these prior systems.

In Fig. 5.1, we have demonstrated one of the strategies that is employed for suppression of nonresonant background: introducing a delay between the pulses to time resolve the resonant signal.[121, 165, 360] However, this limits access to the initial dynamics, such as those visible in panels (F) and (G). Another strategy for suppressing nonresonant background is to rotate the polarization of the excitation fields in order to interfere different symmetry components of the nonresonant response.[146, 361] Experiments performed with different polarization combinations (not shown) indicate that there could be promise to this strategy, though care would need to be taken to ensure that the consequences of the rotated polarizations are understood with respect to the analyte's symmetry. Heterodyne detection is another strategy that has been employed in electronic spectroscopy,[362–371] permitting discrimination between the nonresonant and resonant response on the basis of resolving the phase of  $\chi^{(3)}$ .[122, 123, 353–355] However, such a strategy is poorly suited to frequency-domain methods: it is more effective in collinear geometries,[355] and the change in pointing while scanning light sources can introduce further experimental complications.[372]

<sup>&</sup>lt;sup>i</sup>We note that this is the case for resonance, i.e.  $\omega = \omega_{ba}$ ; otherwise, the detuning factor  $|\Delta_{ba}|^2 = |\omega_{ba} - \omega - i\Gamma_{ba}|^2$ should be used in lieu of  $\Gamma_{ab}^2$ . The other contributions to  $\alpha$  are  $(4\pi\omega F\Gamma_{ba})/(\hbar cn)$ .[99] Since the experiments being compared are electronic,  $\omega$  and  $\Gamma_{ba}$  should be of the same magnitude, and F/n should be similar for most organic solvents. Otherwise,  $4\pi/\hbar c$ are constants.

<sup>&</sup>lt;sup>ii</sup>Once again, we assume resonance  $(|\Delta_{ba}|^2 \approx \Gamma_{ba}^2)$  and the other contributing terms  $(F/(4D\hbar^3))$  should be similar or constant.
– CHAPTER 6 —

## Broadband Coherent Raman Spectroscopy

## 6.1 Background

Interaction between a medium and light can result in the scattering of photons by the constituent molecules. The majority of these photons will be scattered without any loss of energy through Rayleigh scattering. However, some of the photons will exchange energy with the medium through Raman scattering. The change in the energy of the photon represents the energy associated with a vibrational quantum of one of the molecule's Raman-active vibrational modes. For the energy of the vibrational mode,  $\Omega$ , the exchange of energy will yield a photons of frequencies,  $\omega_{out} = \omega_{in} \pm \Omega$ . The two cases reflect the two available directions for energy exchange: the frequency of the output photon corresponds to the sum in anti-Stokes Raman scattering, extracting energy from a vibrationally-excited molecule; the frequency will decrease in Stokes Raman scattering, in which the scattering results in vibrational excitation of the molecule. Identifying the frequency shifts can provide insight into the Raman-active vibrations of the molecules constituting the medium.[373]

Raman scattering can be considered to be a four-wave mixing process in which the interaction between the incident photons and the medium is described according to the third-order nonlinear susceptibility,  $\chi^{(3)}$ .[99] This third-order nonlinear susceptibility allows the molecules of the medium to mediate interactions between the incident photons and the vacuum field corresponding to the zero-point energy.[99, 374] The third-order susceptibility will exhibit a broad nonresonant background in addition to resonances associated with the vibrational modes; scattering will be enhanced when the energy difference between the incident photons and the vacuum field match these resonant frequencies.[120–122, 135, 139, 143, 147, 148, 186, 354, 360, 375, 376] The susceptibility is considered to be a complex quantity; according to the standard treatment, the nonresonant contributions are assumed to be only real, while the resonant parts will add to both the real and the imaginary components of  $\chi^{(3)}$ .[122, 123, 353, 355] Spontaneous Raman scattering is adequately described according to only the imaginary part of the susceptibility.[353] In addition to being linear with the molecules.[190]

Since spontaneous Raman scattering involves an interaction with the vacuum field, the scattered photons lack coherence, [139, 148, 186] yielding very low signal levels due to the low sampling efficiency. [186] The low efficiency limits the utility of spontaneous Raman scattering, preventing its application to media with high levels of background signal that may overwhelm the Raman scatter, or in which the acquisition times required may limit the ability to track the dynamics of short-lived species. [136, 377] Signal may be enhanced through electronic resonance, but electronic resonance may also enhance the background through fluorescence. [139, 378] It is also possible to increase the excitation power, but there are often limits to the extent that this is feasible.[148, 376]

Many of the disadvantages of spontaneous Raman scattering can be overcome through coherent Raman scattering, a broad category of different Raman-based nonlinear optical methods. These coherent methods yield significantly higher signal-to-noise ratios, [190, 377] allowing significantly faster acquisition. [136, 360] While coherent Raman scattering is generally less sensitive to fluorescence, [136, 375, 379] the higher signal-to-noise levels often allow the usage of weaker fields, reducing fluorescence and risk of damage to the sample. [148] Coherent Raman scattering methods still access the third-order nonlinear susceptibility, but are generally implemented with synchronized fields at different frequencies; [186, 190] the frequency difference between the fields determines the vibrational modes accessed during the experiment, as well as whether the process corresponds to anti-Stokes or Stokes Raman scattering. [186] While some authors suggest the improvement in signal arises from a higher cross-section, [375] most appear to attribute the improved signal-to-noise ratio in coherent Raman spectroscopies to the cooperative buildup and emission of signal: [139, 148, 186] the coherence of the scattered photons allows significant improvement is the sampling efficiency. [186]

Development of the various coherent Raman spectroscopies has been motivated by the potential benefits offered by Raman scattering. It is traditionally more straightforward to obtain broadband vibrational information by Raman scattering, [375, 378, 380] while still allowing the use of narrowband methods employing more sensitive detection. [380] Additionally, Raman-based methods also offer promise in microscopy applications: [381, 382] while the frequencies necessary for infrared absorbance measurements cannot be focused as effectively, Raman scattering can be achieved with visible or near infrared frequencies that can be focused more tightly. [139, 148] Finally, while femtosecond spectroscopy can be performed in the infrared, [383–387] there are additional challenges in effectively generating the required ultrashort infrared pulses. [137, 166] Coherent Raman spectroscopies have been realized in a number of ways, with methods having been applied to a variety of relevant systems, including transition-metal complexes, [388] chlorophyll a, [136] and proteins such as bacteriorhodopsin. [389]

A straightforward approach to coherent Raman spectroscopy is stimulated Raman spectroscopy. The general strategy for this method is to introduce two fields of different frequencies and overlap them in the sample.[190] The sample mediates the four-wave mixing process between these fields; when the frequency difference matches the frequency of a Raman-active vibration, the interaction will be enhanced. In stimulated Raman spectroscopies, the interaction is gauged by measuring the intensity of one of the fields: four-wave mixing will result in attenuation of the pump or gain of the probe.[136, 190, 378] Since the technique is self-heterodyned,[190, 390] it exhibits a linear dependence on the imaginary part of  $\chi^{(3)}$ . This has the advantage of

yielding information only associated with the vibrational resonances rather than the nonresonant background; it also makes the signal proportional to the number density of the Raman-active species.[190] However, the signal can be difficult to detect, as it requires resolving very small changes on a large background.[136, 137, 190, 391] It should be noted that in this context, we are only considering stimulated Raman spectroscopy to involve methods of the manner described above: this name has also been used for techniques in which a laser is focused into a sample, and Raman processes result in enhancement of new frequencies at the onset of self-focusing and filamentation or plasma generation. While these methods have been described to gain insight into condensed systems, the mechanism generally results in extreme or abnormal conditions.[392–396]

Of particular interest is femtosecond stimulated Raman spectroscopy (FSRS), a multiplexed version of stimulated Raman spectroscopy. [397] FSRS is generally accomplished by combining narrow- and broadband pulses in a single experiment.[137, 398] These experiments may employ either two or three beams.[137, 391, 399] The two beams incorporated in most FSRS implementations are the Stokes probe and Raman pump; typically the probe is a broadband, femtosecond pulse and the pump is a narrowband, picosecond pulse.[137, 391, 397, 400] Like the more general method of stimulated Raman spectroscopy, when the difference in frequency between the pump and probe match a vibrational resonance, gain is observed in the probe; [136, 190, 378] however, by implementing a broadband probe, FSRS allows the simultaneous measurement of many Raman shifts rather than requiring continuous tuning of the fields; [378, 400] this facilitates faster data acquisition, enabling more averaging and a correspondingly higher signal-to-noise ratio. [400] While the probe pulse is broadband, the spectral resolution of the method is limited by the convolution of the pump bandwidth and the line width of the vibrational mode. [136, 137, 378, 391, 399] Some authors also have incorporated a third beam into the FSRS experiment: the actinic pump which is an additional femtosecond pulse.[137, 391, 399, 400] The presence of this third beam allows the preparation of a particular excited state that is probed at some delay by the Raman pump and Stokes probe. [137, 391, 399] Despite the picosecond duration of the Raman pump in these experiments, the temporal resolution with respect to probing the excited state is determined solely by the cross-correlation of the actinic pump and Stokes probe, [137, 378, 391, 399, 401] allowing FSRS to simultaneously have spectral resolution on the scale of the vibrational modes and temporal resolution of tens of femtoseconds.[137, 390, 399] Due to the nature of the method, FSRS does not require any phase matching considerations, [136] permitting implementation in either a collinear or noncollinear geometry. [137, 398]

However, despite the advantages, there are also some drawbacks to FSRS. Just like the more general stimulated Raman spectroscopy, FSRS requires the ability to reliably measure small changes associated with the Raman gain on a large background.[136, 190, 391] Thus, while data can be acquired rapidly due to

the broadband nature of the technique, [400] achieving high signal-to-noise ratios does require averaging to overcome the otherwise inherent low signal-to-noise ratio. [137] There are also artifacts that can manifest in the data. While contamination from fluorescence is not considered to be a problem, [378, 390] transient absorption and stimulated emission can occur, manifesting as broad features in the Raman gain spectrum. [136, 137, 378] However, the application of shifted excitation Raman difference spectroscopy [402, 403] has been demonstrated to allow these artifacts to be corrected. [378] Trying to extend the method to resonant excitation or anti-Stokes configurations results in negative signals and complicated line shapes. [137, 398, 400] Finally, the use of a sufficiently strong pump has been shown to introduce cross-phase modulation in the probe; [136, 137] however, ensuring a smooth probe spectrum and shallower change in the pump's temporal envelope can minimize the consequences of phase modulation, [136] as can fibrillation of the pump[378] in a manner similar to correcting for scatter interference in other spectroscopies. [404, 405]

An alternative approach, Raman induced Kerr effect spectroscopy (RIKES), to broadband Raman has been proposed in the literature to overcome the large background inherent to FSRS.[397] This method is based on the optical Kerr effect: propagation of an intense, polarized pump results in an intensity-dependent change in the refractive index, inducing a temporary birefringence within the medium.[406] The birefringence is a result of the response of the medium, and the change may involve electronic, vibrational, rotational, or intermolecular contributions.[406–408] The induced birefringence will also affect other fields co-propagating through the medium by cross-phase-modulation. Specifically, if a linearly-polarized probe is introduced, the anisotropic change in refractive index will induce a change in the field's polarization state.[377, 406] Monitoring the change in the transmission of this probe through a crossed polarizer then informs on the resonant and non-resonant response of the medium through the third-order nonlinear susceptibility.[190, 377, 379, 397, 406–411] While this method has been applied to electronic resonances,[407, 410, 411] in RIKES, the goal is to specifically probe the resonance enhancement in the Kerr effect when the pump and probe frequencies differ by the frequency of one of the medium's Raman-active vibrational modes.[190, 377, 379, 397, 406]

Much like FSRS, RIKES does not require phase matching considerations, [377] allowing it to be performed in either a collinear or a noncollinear geometry. [190, 377, 379, 406] Broadband implementations of RIKES have also been reported. [379, 397] In contrast to FSRS, RIKES is dependent upon both the imaginary and real parts of the nonlinear susceptibility; [409, 410] thus nonresonant background is a concern, [190, 377, 397, 406] and the variety of mechanisms capable of contributing to the optical Kerr effect can yield background signals exhibiting dynamics. [407, 408, 410] However, circular polarization of the pump results in suppression of the nonresonant background when Kleinmann symmetry is valid. [190, 377, 397] An alternative approach to resolving challenges with nonresonant background is the introduction of a local oscillator for heterodyne-detection.[190, 379, 409, 410] This also has the added benefit of enhancing the sensitivity of the technique,[379] though it has been shown that RIKES can achieve similar signal-to-noise ratios to FSRS without optimization.[397] The sensitivity of RIKES can also be improved with enhancement through electronic resonance.[379]

While FSRS and RIKES have trivial phase matching, it is also possible to perform coherent Raman spectroscopy through a phase matched process: coherent (anti-)Stokes Raman spectroscopy. While the Stokes Raman based spectroscopy (CSRS) has been demonstrated,[186] the anti-Stokes process, CARS, is more common.[120–124, 139, 140, 143, 146–148, 353–357, 360, 375, 376, 380, 412, 413] While some authors have implemented pulse-shaping methods that allow CARS to be implemented with a single source,[122, 354] the general approach for CARS is based upon spatial and temporal overlap of two or three fields in a medium.[120, 121, 135, 139, 143, 147, 148, 186, 360, 375, 376] In the three pulse sequence, the pulses provide the Raman pump,  $\omega_{\rm P}$ , the Stokes field,  $\omega_{\rm S}$ , and the probe pulse,  $\omega_{\rm P'}$ ;[165, 354, 360] when reduced to two fields, the interaction with the probe field is replaced by a second interaction with the pump,  $\omega_{\rm P'} = \omega_{\rm P}$ .[120, 121, 135, 139, 143, 147, 148, 186, 360, 375, 376] The three interactions mediated by the medium will occur through the third-order susceptibility, yielding an output at the frequency  $\omega_{\rm AS} = \omega_{\rm P} - \omega_{\rm S} + \omega_{\rm P'}$ .[120–122, 135, 139, 143, 147, 148, 186, 354, 360, 375, 376] This illustrates the main advantage of CARS: since we require  $\omega_{\rm P} > \omega_{\rm S}$  for the anti-Stokes process,  $\omega_{\rm AS}$  will always be a unique frequency and blue-shifted relative to the excitation fields. This makes CARS detection robust to scatter and fluorescence.[139, 190]

It is most common to implement CARS with homodyne detection.[120, 121, 123, 124, 139, 140, 143, 146, 148, 186, 356, 357, 360, 375, 376, 380, 412, 413] However, homodyne detection of  $\omega_{\rm AS}$  will yield a signal proportional to  $|\chi^{(3)}|^2$ .[190] This introduces two main limitations to the method. First, because  $\chi^{(3)}$  is proportional to the number density (N), homodyne detection results in signal scaling as  $N^2$ .[99, 122, 148, 190, 354] While this is valuable when the molecule of interest is a majority species in the sample,[122] it makes detection challenging for dilute species,[190, 354] imposes a limit on the sensitivity of the technique,[355] and makes quantification more difficult.[122, 353, 355] Additionally, the scaling of the homodyne-detected signal with  $|\chi^{(3)}|^2$  results in the signal containing contributions from both the imaginary and real parts of the nonlinear susceptibility. Contributions from the nonresonant response results in a background over which it is ideally possible to observe resonance enhancement when  $\omega_{\rm P} - \omega_{\rm S}$  matches the frequency of a Raman-active vibrational mode.[120–122, 135, 139, 143, 147, 148, 186, 354, 360, 375, 376] This can still introduce significant limitations.[121, 123, 139, 148, 165, 190, 355, 357, 360] The interference between the real components from the resonant and nonresonant response of the medium distorts the vibrational line

shape. [136, 190, 353] Excitation at visible frequencies, [140, 148] especially when a two-photon resonance is available, [139, 140] can introduce significant levels of nonresonant background. While near infrared excitation can reduce this background, [140, 148] the scaling of the signal with  $N^2$  can still allow the nonresonant response to overwhelm the resonant signal through a larger number density in many applications. [121] Though femtosecond excitation allows more efficient delivery of higher peak powers without concern for damage or heating, [140, 376] it is also important to match the excitation bandwidth of at least one interaction to the line width of the vibrational modes: besides the spectral resolution made available by picosecond pulses, [139, 140] the increasing bandwidth of increasingly shorter pulses will eventually saturate the spectral width of the resonant process; beyond this point, the resonant process will be enhanced no further, while the nonresonant signal strength will continue to increase. [139, 140, 146, 356, 357] Many techniques have been shown to help suppress nonresonant background, though each has limitations and drawbacks. Phasing of the excitation fields by pulse shaping has been shown to help improve the resolution of the signal. [143] Polarization control can also be used by choosing polarizations for the excitation fields such that Kleinmann symmetry predicts the cancellation of nonresonant signal. [143, 146, 361] Time resolving the signal by delaying the final interaction (the probe field) also allows reduction of nonresonant background: by introducing an appropriate delay, the probe ideally only samples the free-induction decay of the induced vibrational coherence.[121, 165, 360] However, in highly scattering samples, efforts to control phase and polarization can be disrupted.[121] Meanwhile, changing the polarization of the excitation fields or delaying the arrival of the probe reduces the signal available.[355] Additionally, there are advantages to using high numerical aperture objectives in CARS; however, under such conditions, the paraxial approximation breaks down[139] and it is difficult to ensure polarization purity.[139, 414]

An alternative strategy to overcome the challenges of nonresonant background is the implementation of heterodyne detection[122, 353–355] due to its advantages. By interfering the CARS signal with a local oscillator, the heterodyned signal is proportional to the electric field of the CARS signal, rather than the intensity of the field;[122, 353–355] the signal is therefore proportional to  $\chi^{(3)}$ , rather than  $|\chi^{(3)}|^2$ . This leaves the signal proportional to the number density,[123] making quantification more straightforward[122, 353, 355] and simplifying the detection of weaker species.[190, 354, 355] The interference with the local oscillator can amplify the signal, improving sensitivity.[355] Heterodyne detection also allows the real and imaginary components of the nonlinear susceptibility to be resolved:[353–355] as the nonresonant background is predominantly associated with the real component of the nonlinear susceptibility, this allows the resonant response to be isolated.[122, 123, 353, 355] Despite these advantages, heterodyne detection is rarely implemented: while some authors have employed a pulse shaper to achieve heterodyne detection,[122, 354] it generally

143

requires an additional source to supply the local oscillator.[354] Additionally, due to the phase-matching of the process, it is really only practical to implement heterodyne detection in a collinear geometry.[355]

The phase matching requirements of CARS also provides the other challenge in implementing the technique. While RIKES and FSRS are satisfied by a trivial phase matching in the same manner as transient absorption, [136, 377] phase matching is important to consider in the CARS experiment. Like other nonlinear techniques, CARS must be performed such that we simultaneously satisfy the energy conservation condition,  $\omega_{AS} = \omega_P - \omega_S + \omega_{P'}, [120-122, 135, 139, 143, 147, 148, 186, 354, 360, 375, 376]$  and momentum conservation,  $k_{\rm AS} = k_{\rm P} - k_{\rm S} + k_{\rm P'}$ .[135, 361] Different approaches have been used to satisfy the phase-matching of CARS, but a common one is the BOXCARS geometry, in which the three input fields and the output occupy four corners of a rectangle [120, 121, 135, 147, 186] However, this alignment can require relatively significant angles.[186] Additionally, it can be difficult to simultaneously phase match the CARS processes corresponding to different vibrational frequencies over a broad range of Raman shifts, [120, 136, 137] as changing the frequencies of the fields necessitates the use of different input angles.[135] The effects of phase matching can also complicate the interpretation of observed line shapes; [120, 147] care must be taken in alignment and focusing, as the presence of higher-order transverse modes [205] and the Guoy phase shift [139, 203] can both complicate the interpretation of phase matching. However, there are some situations where phase matching can be neglected. If the interaction region is short, [139] phase matching can be ignored: in the limit of minimal depletion of the excitation fields, signal intensity is proportional to  $\operatorname{sinc}^{2}(L\Delta k/2)$ , where L is the interaction length and  $\Delta k$  is the phase mismatch; short pathlengths then make the experiment less sensitive to phase matching. Alternatively, if dispersion is sufficiently weak, such as in a gas, the phase matching condition reduces to the energy conservation condition. [135] Broadband phase matching has also been proposed by introducing a wide-angle probe to simultaneously satisfy the phase-matching of different processes; such an alignment necessitates moving the detector to measure the intensity of CARS signal associated with different Raman shifts, however.[135]

The concept of reducing the constraints of phase matching with sufficiently wide-angle excitation has lead to the implementation of CARS in microscopy.[139, 140, 146, 148, 165, 355–357, 375, 380, 412] In this application, CARS provides a sensitive, label-free method of imaging;[165, 357, 380, 412] additionally, the dependence of the signal upon the overlap of multiple beams improves the spatial resolution.[148] In microscopy, the limit of phase matching the CARS process can be overcome by the use of high numerical aperture objectives: this provides sufficiently wide-angle excitation so as to relax the phase matching condition,[139, 140, 146–148] as well as reducing the interaction length.[139] High numerical aperture objectives even allow collinear geometries,[146] facilitating the application of heterodyne detection.[355] The tighter focus also increases spatial selectively[139, 375] and improves the image quality.[139] However, high numerical aperture objectives cannot be implemented for all systems, and the filtering schemes necessary to isolate the CARS signal can result in a loss of sensitivity.[147]

While a few reported implementations of CARS employ only picosecond[139, 148, 355] or femtosecond[120, 122, 143, 147, 354, 357, 413] excitation, many other implementations of CARS[121, 146, 186, 353, 356, 360, 375, 376, 412] - as well as various implementations of FSRS[136, 137, 378, 389–391, 398, 399, 401, 415] and RIKES[379, 397] - have sought to achieve multiplexing while retaining spectral resolution on the order of the line width of vibrational modes.[390, 401] While this could be achieved by synchronization of femtosecond and picosecond lasers,[390] it is difficult to reliably achieve synchronization without timing jitter.[376, 415] While continuum generation is possible with picosecond pulses,[318, 320, 332, 416] the higher peak intensity of femtosecond pulses generally yields higher efficiencies[417] since the nonlinear processes associated with light generation are generally nonresonant.[139, 140, 146, 356, 357] As a consequence, rather than start with picosecond pulses and achieve multiplexing through continuum generation, most reports describe starting with femtosecond pulses and the application of a variety of strategies to improve spectral resolution.[136, 137, 165, 186, 356, 357, 376, 378, 380, 389–391, 397, 399–401, 412, 417–419] In addition to reporting strategies for narrowing their excitation pulses, some authors also give consideration to increasing the breadth of their pulses in order to maximize the range of Raman shifts accessible.[122, 354]

Many of the efforts to extend the accessible range of Raman shifts are based upon white light generation.[136, 137, 356, 375, 378–380, 397, 398, 420] A brief overview on the generation of white light has been provided in Ch. 5. However, an alternative strategy that has been employed for generating broadband pulses is cascaded frequency mixing processes.[400] By crossing two beams with wave vectors  $k_1$  and  $k_2$ , various mixing processes are cascaded to form spatially resolved orders satisfying various phase matching conditions generalized  $(n + 1)k_1 - nk_2$ .[282–284, 421] These cascaded side bands are broad and tunable,[283, 306, 400] providing an alternative to white light generation. However, unlike white light generation, these side-bands are reasonably well-compressed pulses with durations on the order of the driving fields.[283, 400]

Strategies for developing narrowband pulses from broadband sources are diverse, but can be considered to fall into three categories: filtering, in which only a small bandwidth from the original broadband source is retained; methods based on appropriate chirp of the broadband source; and, narrowing through parametric processes in a nonlinear crystal. Some of these methods also introduce picosecond optical parametric amplifiers that are pumped by the narrowed source, providing the capability to tune the narrowed pulse.[391, 400, 415]

Three main strategies for filtering broadband sources have been reported in the literature. All three are

inherently inefficient, with reported efficiencies less than 0.3%, since they are predicated on passing only a small bandwidth from the original source. [399, 417, 419] The simplest filtering scheme that has been reported is the use of a pair of commercial filters that together reduced the bandwidth to  $17 \text{ cm}^{-1}$ .[136, 137] This corresponds to a pulse duration slightly shorter than one picosecond (800 fs), but this strategy restricts the tuning of the pump according to the availability of appropriate filters. A limited ability to tune the Raman pump was introduced through grating-based filters; the design is based upon a 4f pulse shaper: two gratings separated by a distance of 4f and by two lenses (focal lengths of f), with the lenses positioned to have a shared image plane at which a slit is positioned. [121, 186, 378] Some authors have simplified the configuration by using a flat mirror at the image plane, allowing the use of a single grating. [380, 415] These grating-based filters allow significantly longer pulse durations, up to 8-9 picoseconds, and the ability to tune within the bandwidth of the broadband source.[378] However, in order to achieve such short pulses, it is necessary to use very small slits; this can introduce perturbations to the mode. [390] The symmetry of the resulting pulses have also been reported to introduce ringing, [390] though some authors have suggested it can be reduced for some temporal profiles. [400] Besides the inherent inefficiency due to the restricted bandwidth, grating-based filters suffer from further inefficiencies from diffraction off the grating.[390] The final filtering strategy that has been described is the use of Fabry-Pérot etalons.[390] The motivation for filtering using Fabry-Pérot etalons is their earlier application in other spectroscopies, such as sum-frequency generation.[422] Fabry-Pérot etalons offer high efficiency at the central color and are capable of very narrow bandwidths (as low as  $3 \text{ cm}^{-1}$  reported).[390] The narrowed pulses exhibit an exponential profile in the time domain: such a profile helps enhance the vibrationally resonant signal relative to the nonresonant background by matching the dephasing of the coherence. [390] While it can be difficult to simultaneously achieve a sufficient free spectral range and a narrow bandwidth affordably, etalons can be combined with other filters to compensate for the free spectral range. [423] The main additional limitation associated with Fabry-Pérot etalons is that the cost can be significant.[390]

The inefficiency associated with spectral filtering led a variety of attempts to improve efficiency through chirped pulses. The first of these methods has been termed "spectral focusing." [412] This method was motivated by an earlier observation that the presence of chirp in the excitation fields can actually improve the resolution of the CARS process. [146] An early implementation of applying chirp followed this observation by overlapping a broadly chirped pulse and a compressed pulse: as the interaction necessary for the CARS process is only possible during temporal overlap, the resolution is controlled by the "temporal slit" resulting from the chirp rate and the femtosecond pulse duration. [376] However, by applying an equal degree of chirp to both the pump and Stokes pulses, other authors have shown the potential to apply the entire bandwidth of broadband pulses to drive processes corresponding to a target Raman shift: with appropriate adjustment of chirp and delay, the instantaneous frequencies temporally overlapped correspond to excitation of a particular vibrational mode throughout the entire duration of the pulse.[356, 357, 401, 412] An alternative method for improving resolution through chirp involves spectral compression by the sum-frequency generation of counter-chirped pulses. In this scheme, a pulse is split to yield two pulses; each pulse is passed through a stretcher to apply chirp: however, one pulse has a positive chirp applied, while the other pulse is negatively chirped. These pulses are then mixed to yield a narrow band pulse.[391, 400, 424]

However, this method involving counter-chirped beams requires careful alignment of a pair of stretchers so as to add equal chirp to both beams; [419] a handful of authors have reported an alternative that bypasses this complication. Rather than apply chirp, these authors use the large group velocity mismatch between a fundamental and its second harmonic to impose a very narrow phase matching bandwidth on the secondharmonic generation process. [165, 399, 417–419] This allows tuning of the pulse-duration by changing crystal material - and therefore the group velocity mismatch - or by changing the crystal length.[399] However, the narrow phase-matching bandwidth does not significantly reduce the efficiency. Rather, the spectral compression can be considered to arise from a series of mixing processes between frequencies of  $\omega_0 + \Delta \omega$  and  $\omega_0 - \Delta \omega$ : for any value of  $\Delta \omega$ , the sum-frequency will be  $2\omega_0$ . [417] Even without the ability to simultaneously phase-match the sum-frequency generation for every value of  $\Delta \omega$ , the cooperation of the sum-frequency and second-harmonic generation processes towards spectral narrowing make this method very efficient, [399, 417] with efficiencies reportedly approaching 40%.[417] The main limitation to this method is in the resulting line shape: as the mixing process is most efficient at the beginning of the crystal - when the pulses are most intense - the pulses exhibit a temporal profile with a sharp edge and an approximately quadratic tail. [399, 417, 418] However, unlike Fabry-Pérot etalons, the tails of these spectrally-compressed pulses decay in the reverse direction relative to the vibrational dephasing: the edge of the pulse artificially attenuates the free-induction decay of the coherence, introducing ringing in the spectrum.[399] While a filtering scheme to improve the line shape and reduce this ringing has been proposed, it calls for a grating-based spectral filter. [399] Additionally, it has been reported that introducing some phase-mismatch to the second-harmonic generation can influence the line shape.[417] An alternative solution has been the introduction of periodically-poled crystals, such as periodically poled lithium niobate or stoichiometric lithium tantalate. [165, 417–419] These engineered materials exhibit attractive properties for spectral compression: high group velocity mismatch, high secondorder susceptibility, and periodic poling to avoid spatial walk-off.[417] However, by tuning the periodicity of the materials, it has been demonstrated that it is possible to engineer desired temporal profiles for the spectrally compressed pulse. [165, 418, 419]

A variety of methods have been employed to study the vibrational structure of the photosystem II and various co-factors of the protein,[57] including infrared absorption[60, 62, 84] and (resonance) Raman.[58, 59, 61, 79] However, the number of transitions available in photosystem II results in very complicated absorption spectra;[62] the resonance-enhancement yields simpler spectra, but the use of spontaneous Raman results in relatively weak signals[58, 59, 61] and fluorescence can be a concern.[61] Coherent Raman spectroscopies offer the potential to improve the sensitivity of these latter experiments and provide intrinsic resistance to fluorescence. For testing the application of broadband coherent Raman spectroscopy, benzene was used as an initial test sample due to the strong Raman-active mode at 992 cm<sup>-1</sup>.[406] Manganese tetraphenylporphine was selected as a model system for exploring resonance enhancement: previous reports indicate that at least one of the electronic transitions (Band V, near 480 nm) is believed to exhibit strong contributions from the manganese, and that excitation of this band results in resonance-enhancement of the Raman mode associated with the low frequency vibration between the manganese and the axial ligand.[309]

## 6.2 Experimental Deviations

The broadband CARS experiments reported in this section were performed as four color experiments. The two femtosecond TOPAS-C provided two femtosecond pulses as the Raman pump ( $\bar{\nu}_1$ ) and Stokes ( $\bar{\nu}_2$ ) pulses and were tuned so the frequency difference between the TOPAS-C was close to the frequency of the target vibrational mode. The Raman probe pulse was derived from the residual fundamental from the Spitfire Pro XP regenerative amplifier. In contrast to the standard alignment of the fundamental, the beam was not turned directly into the appropriate delay line. Rather, a set of three mirrors was introduced to allow the introduction of a volume Bragg grating for spectral narrowing with an appropriate angle between the incident and diffracted beams.

While the enhancement of vibrational dephasing attributed to the temporal profile of pulses narrowed with Fabry-Pérot etalons[390] was attractive, calculations suggested that it would be difficult for a reasonably affordable etalon to simultaneously provide wide tuning capability, high throughput, narrow spectral profiles, and a wide free spectral range. However, it has been reported that etalons with a narrow free spectral range may be combined with volume Bragg gratings to simultaneously achieve narrow spectral profiles and broad free spectral ranges.[423] These gratings are prepared by imaging a sinusoidal refractive index grating in a material; based on the period of these modulations ( $\Lambda$ ), the index of the material (n), and the incident angle  $(\theta)$ , a very narrow bandwidth centered around a wavelength,  $\lambda_0$ :

$$\lambda_0 = \cos\left(\arcsin\left(\frac{\sin(\theta)}{n}\right)\right)\frac{\Lambda}{2n} \tag{6.2.1}$$

is diffracted with high efficiency.[423]

The grating (Optigrate; Oviedo, FL) employed in these experiments was 2.1 mm x 5 mm x 5 mm and designed to operate around a central wavelength of 805.3 nm, yield a spectral width of  $\sim 0.25$  nm (FWHM), and achieve > 97% diffraction efficiency. Testing indicated a  $\sim 35^{\circ}$  angle between the incident beam and the diffracted beam optimized the overlap of the diffracted color ( $\sim 785$  nm at 30° and  $\sim 795$  nm at 40°) with the center of the regenerative amplifier output; it should be noted, though, that these measurements were collected with an OceanOptics USB-2000 that exhibits a sharp decrease in response near 800 nm and consequently reports distorted spectra for  $\sim 35$  fs pulses centered near 800 nm. The central color observed in these experiments was  $\sim$  794 nm; however, we are not confident that the Jobin Yvon MicroHR monochromator provided sufficient resolution to reliably characterize the spectrum. Consequently, Raman shifts reported in all CARS spectra have been determined by matching the observed vibrational features to the reported frequencies for the sample; sources for the frequencies are for benzene (993  $\text{cm}^{-1}$ ),[425] benzonitrile  $(1601 \text{ cm}^{-1})$ , [426] and pyridine (992 and 1031 cm<sup>-1</sup>). [186] The temporal profile of the diffracted pulse, shown in Fig. 6.1, was determined through a nonresonant CARS process in carbon tetrachloride with the TOPAS-C tuned to  $\bar{\nu}_1 = 8000 \text{ cm}^{-1}$  and  $\bar{\nu}_2 = 7000 \text{ cm}^{-1}$ . It should be noted that this spectrum does represent convolution of the temporal profile of the diffracted pulse with the TOPAS-C output; however, as the durations of the latter two pulses are  $\sim 40$  fs, the convolution should be a minor effect. With the conventions of delay on the table, the tail of the diffracted pulse should match the dephasing of the vibrational mode in a manner similar to an etalon.

Phase matching angles for these experiments were calculated within the cw approximation for the center of each pulse. These calculations indicated two key details. First, it proved necessary to use a faster focusing mirror for all experiments: the angles calculated for near infrared excitation required a f = 50 cm mirror, while for the experiments with visible excitation this had to be reduced further to f = 20 cm. The other key detail indicated by these calculations was the narrow phase matching bandwidths: within this cw approximation, it was anticipated that the phase matching bandwidth would be significantly narrowed in the visible.

The CARS experiments were performed as one- or two-dimensional acquisitions involving only the monochromator frequency and the relative delays. The TOPAS-C were set to generate particular colors



Figure 6.1: Temporal profile of the narrowed ~ 794 nm probe as determined by nonresonant CARS ( $\bar{\nu}_1 = 8000 \text{ cm}^{-1}$  and  $\bar{\nu}_2 = 7000 \text{ cm}^{-1}$ ) in carbon tetrachloride.

for  $\bar{\nu}_1$  and  $\bar{\nu}_2$ . The convolution of the spectral content of these pulses determined the accessible vibrational frequencies; [120–124] by scanning the monochromator, we determined the frequencies of the vibrational coherences created by this convolution based upon the intensity of the upconverted Raman probe at each frequency. As before, the preamplifier limited reliable application of available photomultiplier tubes; a silicon photodiode was used for detection instead. Reported usage of "spectral focusing" [356, 357, 401, 412] indicated the importance of considering any chirp or other frequency modulation present in the pump and Stokes pulses. Due to suspicions of the upconversion processes in the TOPAS-C, this was expected to be a particular concern in experiments using sum-frequency signal. Two-dimensional acquisitions monitoring the CARS spectrum as a function of relative delay between the pump and Stokes pulses suggested this was not a large concern in the near infrared (data not shown); similar scans collected with the pump and Stokes pulses derived from the sum-frequency signal process will be shown in the Results and Discussion.

## 6.3 Results and Discussion

Consistent with previous reports, [399, 417, 419] the use of filtering to achieve spectrally narrow pulses resulted in a dramatic reduction in beam energy. Prior to introducing the Bragg grating, we measured a per pulse energy of 40  $\mu$ J in the residual fundamental; with the grating in place, this energy was reduced to 370 nJ per pulse. Even with the reported > 97% diffraction efficiency, these losses correspond to 0.9% efficiency; based upon the reported efficiencies of other filtering strategies, [399, 417, 419] this observed efficiency is consistent with the reduction in bandwidth. However, besides the limits to spectral resolution arising from a femtosecond probe, Fig. 6.2 shows that the picosecond probe also provides a high degree of discrimination against the nonresonant background. A theoretical foundation for this has been previously reported: [140] as the pulse duration decreases, both nonresonant and resonant signal increase; however, while nonresonant signal continues to increase, the resonant response only scales so long as the excitation bandwidth is less than or comparable to the vibrational line width. Consequently, under purely femtosecond excitation, the ratio of resonant to nonresonant signal is predicted to be quite small. [139, 140, 146, 356, 357] In Fig. 6.2, we show that even a single picosecond interaction is sufficient to improve this ratio.

All scans in Fig. 6.2 were collected under conditions of near infrared excitation:  $\bar{\nu}_1 = 8000 \text{ cm}^{-1}$ and  $\bar{\nu}_2 = 7000 \text{ cm}^{-1}$ . Under the conditions of the initial table alignment (femtosecond probe pulse), we measured the dynamics associated with both carbon tetrachloride (CCl<sub>4</sub>) and benzene (C<sub>6</sub>H<sub>6</sub>). Most of the signal for both samples was consistent with the nonresonant response arising from cross-correlation of the three excitation pulses. However, in benzene, we do observe weak signal above the baseline at long delays, consistent with the longer dynamics expected for a vibrational coherence. Nevertheless, the ratio of this signal to the nonresonant response at zero delay is very low: as the scale in (A) shows, it is approximately one to one hundred, which is reasonable in the context of previous calculations.[140] This contrasts with the behavior observed when the narrowed, picosecond probe is used, (B). The signal from carbon tetrachloride suggests that the weaker excitation from the picosecond probe yields a lower sensitivity, as the nonresonant signal is only two orders of magnitude over the baseline, rather than the three orders of magnitude in (A); this is consistent with the higher contrast expected due to the peak intensity available from femtosecond excitation.[140] In exchange for the lower sensitivity, though, (B) shows that we can achieve much greater specificity for the vibrational dynamics: where the femtosecond probe yielded a resonant to nonresonant ratio of about one to one hundred, the picosecond probe achieves ratios of about seven to ten, a seventy-fold improvement.

With the spectrally-narrowed probe, we explored the reliability, sensitivity, and resolution of broadband CARS with near infrared excitation ( $\bar{\nu}_1 = 8000 \text{ cm}^{-1}$  and  $\bar{\nu}_2 = 7000 \text{ cm}^{-1}$ ). A sampling of these results is reported in Fig. 6.3, 6.4, and 6.5. All three figures present a two-dimensional acquisition (A) reporting signal on a logarithmic (base-10) scale as a function of delay versus Raman shift; the delay corresponds to the relative delay of the probe relative to the temporally overlapped pump and Stokes pulses. Zero delay is defined according to the sharp falling edge in the nonresonant signal due to the rising edge of the probe pulse. (B) and (C) present one-dimensional views of the data: (B) presenting vertical (delay) slices, and (C) showing horizontal (frequency) slices. All slices shown in (B) and (C) are color- and style-coded to correspond to the vertical and horizontal, respectively, slices shown in (A).

Consistent with the Raman properties of carbon tetrachloride,<sup>i</sup> Fig. 6.3 shows no evidence of vibrational resonances for Raman shifts between 850 and 1200 cm<sup>-1</sup>. This is evidenced by the sharp edge at a delay of 0 ps, and the broad, featureless spectral character across the frequency range. The dip near a Raman shift of 850 cm<sup>-1</sup> is attributed to noise. The slices in (B) and (C) show that the nonresonant response in carbon tetrachloride gives us sensitivity of approximately one and a half orders of magnitude (signal levels above baseline) in these experiments.

In contrast to Fig. 6.3, the presence of a vibrational resonance yields a very clear signature in the two-dimensional spectrum, as shown for neat benzene in Fig. 6.4. In the two-dimensional spectrum, the vibrational resonance at 993 cm<sup>-1</sup> is seen clearly above the baseline. Contrasting the vertical slices at 975 cm<sup>-1</sup> and 993 cm<sup>-1</sup>, the dashed and solid vertical slices respectively, we see in (B) that an exponential

<sup>&</sup>lt;sup>i</sup>Available at webbook.nist.gov.



Figure 6.2: Comparison of CARS signal with ps and fs probe pulses. All scans report normalized signal versus the delay between the probe pulse and the overlapped Raman pump ( $\bar{\nu} = 8000 \text{ cm}^{-1}$ ) and Stokes ( $\bar{\nu} = 7000 \text{ cm}^{-1}$ ) pulses; the monochromator was set to the wavelength at which the CARS signal associated with benzene's 993 cm<sup>-1</sup> vibrational mode was expected. (A) and (B) are the results for carbon tetrachloride and benzene with a femtosecond, 800 nm probe and the spectrally narrowed picosecond probe, respectively. The data presented for benzene represent all data accumulated over two scans.



Figure 6.3: Broadband CARS of carbon tetrachloride (CCl<sub>4</sub>) with Raman pump pulse  $\bar{\nu}_1 = 8000 \text{ cm}^{-1}$ , Stokes pulse  $\bar{\nu}_2 = 7000 \text{ cm}^{-1}$ , and narrowband probe (~ 794 nm). (A) Signal (base-10 logarithmic scale) as a function of delay and Raman shift. Delay is relative delay between the probe pulse and the overlapped Raman pump and Stokes pulses. Raman shifts are assigned on the basis of matching time-resolved vibrational resonances in benzene and pyridine to known values.[186, 425] The solid vertical line identifies the slice plotted in (B), and the solid and dashed horizontal lines represent the points from which the slices in (C) are derived. (B) and (C) are signal (base-10 logarithmic scale) versus relative probe delay (B) or Raman shift (C).

decay is clearly visible when the monochromator is set to detect resonant signal (Raman shift of  $993 \text{ cm}^{-1}$ ), whereas signal is only present during pulse overlap when off resonance. In (C), the solid trace shows a horizontal slice of the signal during pulse overlap. In this slice, we see clear evidence of interference between the nonresonant background and the vibrational resonance in the dip in signal near 1025 cm<sup>-1</sup> - this is quite close to the Raman shift at which Levenson and Bloembergen observe a minimum resulting from this interference  $(1020 \text{ cm}^{-1})$ .[110] However, we observe nearly an order of magnitude of contrast between the resonant and nonresonant response; the improvement over the signal in Fig. 6.2 is attributed to more accurate setting of the monochromator: in the data acquired for Fig. 6.2, the monochromator was set based upon the expected wavelength for the CARS signal rather than the wavelength of maximum signal. Recall, the dispersion of the MicroHR monochromator limits our ability to know the frequency of the probe with high precision. In contrast to pulse overlap, time resolution of the signal - given by the dashed line in (C) - allows isolation of the vibrational resonance; consistent with concerns over the use of time resolution for nonresonant background suppression, [355] we do observe some loss of signal ( $\sim 12\%$  on the logarithmic scale). While the strong response of benzene does leave us with high sensitivity after these losses - signal approximately one and a half orders of magnitude over baseline - this loss of signal could prove a limitation with dilute species or those with weaker response.[355]

As expected, the narrowed probe pulse provides significant improvements in resolution. Not only is the vibrational lineshape in Fig. 6.4 distinguishable, but we see in Fig. 6.5 that we can clearly resolve two close vibrational modes. In Fig. 6.5, we show the CARS spectrum of a mixture of benzene and pyridine under near infrared excitation. These compounds exhibit vibrational frequencies very close to each other: benzene exhibits a Raman-active vibration at 993 cm<sup>-1</sup>, [425] while pyridine exhibits a pair of resonances, one at 992 cm<sup>-1</sup> and the other at 1031 cm<sup>-1</sup>. [186] Signals consistent with the vibrational frequencies are present in the two-dimensional acquisition of (A). The time-resolved slice in (C), given by the spectrum in the dashed line, show that we achieve baseline resolution between the vibrations at 992/993 cm<sup>-1</sup> and the vibration at 1031 cm<sup>-1</sup>; this is comparable with the resolution in the time-resolved CSRS results reported by Pestov et al, and both data sets appear to exhibit better resolution than the spontaneous Raman spectrum reported by the same authors. [186] We will also point out the dip in nonresonant signal near a Raman shift of 1150 cm<sup>-1</sup>; we will address this feature later in this chapter.

In order to increase the applicability of these broadband Raman processes towards resonance enhancement, we sought to extend the methods towards visible excitation. The same methods were employed as were used to acquire the previously described near infrared excited CARS spectra, only substituting visible Raman pump and Stokes pulses generated through the TOPAS-C second-harmonic signal (SHS) or sum-



Figure 6.4: Broadband CARS of benzene (C<sub>6</sub>H<sub>6</sub>) with Raman pump pulse  $\bar{\nu}_1 = 8000 \text{ cm}^{-1}$ , Stokes pulse  $\bar{\nu}_2 = 7000 \text{ cm}^{-1}$ , and narrowband probe (~ 794 nm). (A) Signal (base-10 logarithmic scale) as a function of delay and Raman shift. Delay is relative delay between the probe pulse and the overlapped Raman pump and Stokes pulses. Raman shifts are assigned on the basis of matching the time-resolved vibrational resonance of benzene to its known value.[425] The solid and dashed vertical lines identify the slices plotted in (B), and the solid and dashed horizontal lines represent the points from which the slices in (C) are derived. (B) and (C) are signal (base-10 logarithmic scale) versus relative probe delay (B) or Raman shift (C). The faded red line in (C) marks the vibrational frequency of benzene, 993 cm<sup>-1</sup>.



Figure 6.5: Broadband CARS of a mixture of benzene and pyridine ( $C_5H_5N$ ) with Raman pump pulse  $\bar{\nu}_1 = 8000 \text{ cm}^{-1}$ , Stokes pulse  $\bar{\nu}_2 = 7000 \text{ cm}^{-1}$ , and narrowband probe (~ 794 nm). (A) Signal (base-10 logarithmic scale) as a function of delay and Raman shift. Delay is relative delay between the probe pulse and the overlapped Raman pump and Stokes pulses. Raman shifts are assigned on the basis of matching the time-resolved vibrational resonances of benzene and pyridine to known values.[186, 425] The vertical lines are color- and style-coded to match the slices plotted in (B), and the solid and dashed horizontal lines represent the points from which the slices in (C) are derived. (B) and (C) are signal (base-10 logarithmic scale) versus relative probe delay (B) or Raman shift (C). The faded red lines in (C) mark the vibrational frequencies of benzene and pyridine, 993 cm<sup>-1</sup> and 1031 cm<sup>-1</sup>.

frequency signal (SFS) processes. Fig. 6.6 shows the broadband CARS spectrum acquired for benzonitrile using SHS-derived pump and Stokes pulses. In this spectrum, we observe a feature consistent with the 1601  $\rm cm^{-1}$  vibrational mode of benzonitrile. The dynamics exhibited are short but can be clearly resolved. The sensitivity of this signal is also weaker, which is consistent with the excitation pulses being derived from a second harmonic process. More energy is available from the SFS process, which was used to acquire the CARS data in Fig. 6.7.

We have previously discussed the problems that manifest upon upconversion of the idler in the TOPAS-C (see Ch. 4), and we have consequently speculated that the accumulated group delay and pump-signal group velocity mismatch may also manifest in the SFS process. For example, in experimental applications of the SFS process, it is not uncommon to see temporal profiles significantly broader than would be anticipated (see for instance the vertical line width in (E) of Fig. 5.1). Consequently, rather than collect CARS spectra studying the dynamics of the vibrational coherence, the two-dimensional acquisition of Fig. 6.7 seemed for more important. The results of this two-dimensional acquisition are signal (presented on a logarithmic scale) as a function of relative delay between the pump and Stokes pulses and the Raman shift (A). After background subtraction, the first moment of the signal was calculated, with the result overlaid in (A) as the black line; it should be noted that the noise at Raman shifts less than 950  $\rm cm^{-1}$  yielded a handful of meaningless data points - such points have been ignored. As the remaining points show, there does appear to be some degree of frequency-modulation, with the first moment exhibiting a change of  $\sim 50$  fs between Raman shifts of 950 and 1220  $\rm cm^{-1}$ . The interpretation of the broadband CARS signal as the convolution of the pump and Stokes pulses [120–124] complicates the assignment of this result to a particular frequencymodulation profile, but it warrants consideration in other experiments involving light generation through this nonlinear process.

The other pane (B) in Fig. 6.7 is the signal (on a base-10 logarithmic scale) of (A) integrated along vertical slices. Within this treatment, a feature that may be attributed to the 993 cm<sup>-1</sup> vibrational mode of benzene is still apparent. However, the contrast in this scan is significantly lower than in Fig. 6.4: whereas we observed approximately one order of magnitude contrast between the resonant and nonresonant signal under near infrared excitation, with the pump and Stokes pulses generated by the SFS process, our contrast is only approximately half of one order of magnitude. This is consistent with previous reports that nonresonant background becomes increasingly important at higher frequencies, [140, 148] as well as our measurements of nonresonant susceptibilities in Ch. 5. We also note that the width of the nonresonant background appears reduced. Under near infrared excitation, the nonresonant background is above baseline across much of the  $\sim 350 \text{ cm}^{-1}$  window of Raman shifts in Fig. 6.3, 6.4, and 6.5 and is visible over a much broader range than

is shown in these figures. On the other hand, the nonresonant background appears to be above baseline over a range of only  $\sim 200\text{-}300 \text{ cm}^{-1}$ ; this is consistent with the steeper phase matching conditions observed for visible excitation under the cw approximation and in agreement with concerns over operational bandwidth in these phase matched experiments.[120, 136, 137]

With pump and Stokes pulses derived from SFS, we also tried to find signal associated with resonanceenhanced CARS in MnTPP consistent with the reported resonance Raman of the complex.[309] However, while realignment to a different mask allowed detection of signal at the appropriate Raman shifts (~ 400  $cm^{-1}$ ), we could not detect any signal consistent with MnTPP. Referring back to our concerns in Ch. 5 regarding the sensitivity of experiments involving these metal species, it is not too surprising that we had difficulty finding signal as we have replaced two electronic resonances with a nonresonant interaction (the Raman probe). It may be worth returning to these efforts if the capability to tune the Raman probe over a broader frequency range is developed or acquired. Under such a situation, these CARS processes could be performed with four resonant interactions: just as we managed to detect signal under some conditions with MnTPP, the four resonant interactions may be sufficient to enable electronically enhanced CARS experiments.

Recall we mentioned when discussing Fig. 6.5 the decrease in nonresonant signal near  $1150 \text{ cm}^{-1}$ . The behavior yielding this feature is shown more fully in Fig. 6.8, in which it is clear that we observe a sequence of modulations as a function of Raman shift. However, these features are only observed during pulse overlap. Characterization of these modulations along different available axes is shown in Fig. 6.9. (C) makes it very clear that these modulations are not an interference effect associated with the monochromator slits: for increasing slit size, the only change in the signal is broadening and an increase in amplitude, each consistent with increasing slit width. Meanwhile, changing the sample from carbon tetrachloride to pyridine or benzene has a pronounced effect (D), but there is no obvious trend associated with changing the sample composition with respect to benzene and pyridine. From panes (A) and (B), it is clear that changing the TOPAS-C set points has pronounced effects. When these effects were first observed, modulations were observed when the spectral output of the OPAs was checked; however, retuning removed the modulations in the OPA spectral output, but not in the CARS spectra. Realignment of the table following retuning also changed the position and spacing of the fringes with respect to the benzene resonance.

We were unable to find any descriptions or reports of similar modulations in the nonresonant background of CARS spectra in the literature. The only other mention of modulations of any sort is from Pontecorvo et al,[399] who report the observation of ringing in FSRS spectra when employing a narrowband pump with a sharp cutoff rather than a sharp rise as we observe. The authors attribute this ringing to the sudden cutoff



Figure 6.6: Broadband CARS of benzonitrile ( $C_7H_5N$ ) with Raman pump pulse  $\bar{\nu}_1 = 16400 \text{ cm}^{-1}$ , Stokes pulse  $\bar{\nu}_2 = 15100 \text{ cm}^{-1}$ , and narrowband probe (~ 794 nm). (A) Signal (base-10 logarithmic scale) as a function of delay and Raman shift. Delay is relative delay between the probe pulse and the overlapped Raman pump and Stokes pulses. Raman shifts are assigned on the basis of matching the time-resolved vibrational resonance of benzonitrile to its known value.[426] The dashed and solid vertical lines match the dashed and solid slices plotted in (B), and the solid and dashed horizontal lines represent the points from which the slices in (C) are derived. (B) and (C) are signal (base-10 logarithmic scale) versus relative probe delay (B) or Raman shift (C). The faded red line in (C) marks the vibrational frequency of benzonitrile,  $1601 \text{ cm}^{-1}$ .



Figure 6.7: Broadband CARS of benzene (C<sub>6</sub>H<sub>6</sub>) with Raman pump pulse  $\bar{\nu}_1 = 20800 \text{ cm}^{-1}$ , Stokes pulse  $\bar{\nu}_2 = 19800 \text{ cm}^{-1}$ , and narrowband probe (~ 794 nm). (A) Signal (base-10 logarithmic scale) as a function of delay and Raman shift. Delay is the relative delay between the pump and Stokes pulses; overlap of both pulses with the probe is expected for all relative delays. Raman shifts are assigned on the basis of matching the vibrational resonance of benzene to its known value.[425] The black trace is the first moment of the signal, excluding the values calculated for several Raman shifts < 950 cm<sup>-1</sup>. (B) Is the base-10 logarithm of the results of (A) integrated over all delay values at each Raman shift. The faded red line marks the vibrational frequency of benzene, 993 cm<sup>-1</sup>.



Figure 6.8: A spectrally broader broadband CARS of benzene ( $C_6H_6$ ) with Raman pump pulse  $\bar{\nu}_1 = 8000$  cm<sup>-1</sup>, Stokes pulse  $\bar{\nu}_2 = 7000$  cm<sup>-1</sup>, and narrowband probe (~ 794 nm). (A) Signal (base-10 logarithmic scale) as a function of delay and Raman shift. Delay is relative delay between the probe pulse and the overlapped Raman pump and Stokes pulses. Raman shifts are assigned on the basis of matching the time-resolved vibrational resonance of benzene to known value.[425] The horizontal lines are color- and style-coded to the slices in (B). (B) shows signal (base-10 logarithmic scale) versus Raman shift for different horizontal slices of (A). The faded red line in (B) marks the vibrational frequency of benzene, 993 cm<sup>-1</sup>.



Figure 6.9: Broadband CARS with near infrared Raman pump and Stokes pulses. (A) Monochromator slices of CARS signal from benzene versus Raman shift for different pairs of Raman pump/Stokes pulse frequencies with a constant detuning (1000 cm<sup>-1</sup>). (B) Monochromator slices of CARS signal from benzene versus Raman shift for constant Raman pump pulse frequency and variable Stokes pulse frequency. (C) Monochromator slices of CARS signal from benzene versus Raman shift for different monochromator slit widths for Raman pump  $\bar{\nu}_1 = 8000 \text{ cm}^{-1}$  and Stokes pulse  $\bar{\nu}_2 = 7000 \text{ cm}^{-1}$ . (D) Monochromator slices of CARS signal from different solvent systems versus Raman shift for Raman pump  $\bar{\nu}_1 = 8000 \text{ cm}^{-1}$  and Stokes pulse  $\bar{\nu}_2 = 7000 \text{ cm}^{-1}$ . Ratios of benzene:pyridine are volumetric. Data in (A) and (B) collected before OPA retuning; data in (C) and (D) collected after OPA retuning.

of the pump during the vibrational dephasing; the artificial attenuation of the natural free induction decay manifests as ringing with a modulation frequency and amplitude related to how early the cutoff occurs during the free induction decay.[399] A similar cause would be consistent with the observation of modulations in pyridine and benzene, but their absence in carbon tetrachloride (see (D) in Fig. 6.9): as the latter is purely nonresonant, there are no free induction decays to artificially attenuate. It would also be expected to occur regardless of the state of the system.

However, there are many reasons to dismiss the application of this theory to explain these modulations. First, the exponential tail of the temporal profile of the Raman probe used here is believed to follow the decay of the vibrational free induction decay. The sharp leading edge of the pulse gives the sharp edge observed in the CARS spectra when the probe is delayed relative to the pump and Stokes pulses. This temporal profile is the reverse of the envelope described by Pontecorvo et al.[399] Additionally, the mechanism for the ringing observed by Pontecorvo et al is inconsistent with our observation of the modulations only during pulse overlap, as well as the lack of change in the modulations throughout the pulse overlap region (see (B) in Fig. 6.8). Thirdly, while ringing would be consistent with the recurrence of the modulations after retuning and realignment, the ringing should be a function of the vibrational mode's dephasing and the attenuation of this decay: this should have little dependence upon the frequencies used to prepare the coherence, raising questions as to why we would observe such significant changes in the modulations as we changed the excitation frequencies.

In order to understand these modulations, we attempted to simulate nonresonant processes capturing as many of the relevant factors, including the phase of the Gaussian field and dispersion across the spectral profiles of the pump and Stokes pulses; the resulting script is included in Appendix D. However, in capturing these different effects, the script yielded a parameter space far broader than could be reasonably explored. Surveys were conducted exploring a few of these parameter spaces, illustrating the complexity of the system; however, a few key insights were gained.

First, the effects of dispersion on these experiments is very pronounced. When the propagation of the fields is described only by the central color, the nonresonant contribution to the signal is a very smooth envelope. This is consistent with the description of the signal inheriting character according to the convolution of the pump and Stokes pulse envelopes.[120–124] The resulting signal will match this envelope if the sample exhibits low dispersion,[135] such as a gas,[120] or arises from a thin sample,[139] as when scattering from a powder.[121] However, the simulations illustrate that when dealing with a bulk sample as in the present experiments, the dispersion across the spectrum adds a further degree of complexity to interpreting the anticipated spectral character of the signal. Rather than attempt to explore the several parameters that control the phase matching angles in the simulations, it may be more straightforward to perform future experiments within the previously described limit of thin samples.

The other effect that proved pronounced was the effect of mismatch between the focal planes of the excitation beams. Various authors have previously discussed the importance of the Guoy phase term in nonlinear experiments, [139, 203, 427] and this manifested in the simulations: depending upon the relative focal point of the pump and Stokes pulses, the spectrum of the output could exhibit both pronounced oscillations and significant changes in amplitude. Unfortunately, this is a very difficult axis to gauge proper values with any degree of accuracy due to the imaging conditions on the table at the time of these experiments. When performing these experiments, we performed knife edge measurements to gauge the beam sizes: the results were consistent with beam waists of 20  $\mu$ m for the Stokes and probe pulses, but 40  $\mu$ m for the pump. This suggests that the pump pulse was smaller than the other beams at the focusing mirror, which is consistent with estimates of the beam sizes at the mirror based upon the measurement of aperture diameters that passed half the beam energy. According to the optics of Gaussian beams, this smaller beam waist for the pump will not only cause the focal spot size to be larger, but will also shift the focus; [105] with our measurements, this could correspond to shifts in focus of hundreds of micrometers - well within the range the simulations suggest modulations could be introduced. It is clear that when progressing to threeand four-color experiments, it is crucial to not only consider our usual quality metrics - spectral content, energies, alignment, sample - but also consider whether the alignment achieves the correct imaging in order to minimize artifacts and unexpected behavior induced by phase effects such as the Guoy phase.

CHAPTER A	
Appendix A	

~	I A margane instead of one D. LA
Damage in Regen:	1000 H - I M LINCA CARE
- high pump power? NO	Den the August to NO newer
	22-23 A
- Bandwidth ? 45 nm	Should the Z
2	( 100% current - what is everent ? 39.5 A 5
not stratched optimally?	how much SHG? No power out - no
too much peak power?	even visible.
	SHG QUA. ( CUMENT - 10
IR VIEWER more sensitive thin 407 A	A
~ 36 am out of country 7	
~ 35 out of compressor	difference between coving and post compressor
المم بماجلهم والمراجع والمراجع	al court-y.
NIO AM LOST BELLEA SWELLING DI	
	the second second second second
Dove let him knows when / when	t we plan to do RE: is system (extra crystals, mixer install)
<i>V</i> <b>I</b>	
	SHE TERD I BOLK MINOR => MEXIMIZE POWER
Set to 4.0 w, set to that current	t, adjust and to the
Skye - Tsumani	
Skye - Tsumeni	on each mirror in turn 7 has durne
Skye - Tournani use previous murror to center pump o	on each murror in turn } hear during
Skye - Tournani use previous mirror to center pump o	on each murror in turn } hear during
Skye - Tsumani use previous murror to cento pump o P2 is important !!	on each mirror in turn } hear during
Skye Tsumeni use previous invoor to cento pump o P2 is important !!	on each mirror in turn } Hearn during
Skye - Tournani use previous murror to cento pump o P2 is important !! M4 -	on each mirror in turn } hear during
Skye - Tournani use previous murror to centro pump o P2 is important!! M4 - [0]	on each murror in turn } hear during
Skye Towneni use previous murror to centro pump o P2 is important!! M4 - [0]	ed on all opprices
Skye Tsumani use previous murror to centro pump o P2 is important!! M4 - [0] (1) make sure pump is centre	ed on all optics
Skye Towneni use previous mirror to center pump o P2 is important!! MH - [0] (1) make sure pump is centere (2) carry fluoregence and	ed on all optics mode centered on all optics }
Skye Towneni use previous mirror to centro pump o P2 is important!! M4 - [0] (1) make sure pump is centrole (2) carry fluorenseenes and	ed on all optics mode centered on all optics } Drisms + ADM -> M5??
Skye Towneni use previous murror to centro pump o P2 is important!! M4 - [0] (1) make sure pump is centrol (2) cavity fluorence and (3) beam straight thigh /	ed on all optics mode centered on all optics } prisms + ADM -> M5??
Skye Tournani use previous inverse to centro pump o P2 is important!! M4 - [0] () make sure pump is centrole (2) carving fluorenseenes and (3) becom straight thick f	ed on all optics modes centered on all optics } prisms + ADM -> M5??
Skye Tournani use previous averor to center pump o P2 is important!! M4 - [0] () make sure pump is centere (2) covery fluorence and (3) becam straight thish f	ed on all optics mode centered on all optics } prisms + ADM -> M5??
Skye Towneni use previous mirror to centro pump o P2 is important!! M4 - [0] () make sure pump is centrol (2) cavity fluorence and (3) becom straight thigh f	on each murror in turn } been during ed on all optics mode centered on all optics } prisms + ADM -> MS ??
Skyr Tsumani use previous mirror to centro pump o P2 is important!! M4 - [0] (1) make sure pump is centrol (2) carving fluorenseence and (3) becom straight thigh f (1) Clipping -> no photodiode si	ed on all optics mode centered on all optics } prisms + APM -> M5?? Identical instance on all optics of the state of t
Skye Towneni use previous murror to centro pump o P2 is important!! M4 - [0] (1) make sure pump is centrole (2) cavity fluorenseence and (3) beam straight thigh f (1) Clipping > no photodiode si M5 + move M5 to mexim	ed on all optics mode centered on all optics } prisms + ADM -> MS ?? ingual nize phonodiode signal and compensate w/ oc / TEP-Manue Fires TV.
Skye Towneni use previous murror to centor pump o P2 is important!! M4 - [0] () make sure pump is centere (2) carity fluorencence and (3) beam straight thick f (1) Clipping > no photodiode si M5 > more M5 to mexim	ed on all optics mode centered on all optics } prisms + APM -> M5?? injanal nize provodiode signed and compensate w/ oc
Skye Tournami use previous inverse to centro pump o P2 is important!! M4 - [0] () make sure pump is centrose (2) carring fluorenseenes and (3) becom straight thigh f (1) Clipping => no photodiode si M5 => more M5 to maxim (2) at xt week Dave's open!!!	ed on all optics mode centered on all optics } prisms + ADM -> M5?? signal nize provodiode signal and compensate w/ oc Here on Mon after noon
Skye Tournani use previous murror to centro pump o P2 is important!! M4 - [0] () make sure pump is centrole (2) cavity fluorenceans and (3) beam straight thick f (1) Clipping > no photodiode si M5 = move M5 to maxim (2) next week Dave's open!!!	ed on all optics mode centered on all optics } prisms + ADM -> M5?? mignal nize provodiode signal and compensate w/ oc Here on Mon after noon
Skye Tournani use previous murror to centor pump o P2 is important!! M4 - [0] () make sure pump is centre (2) cavity fluorencence and (3) beam straight thigh f (1) Clipping > no photodiode si MS > more MS to maxim (2) next week Dave's open!!!	ed on all optics mode centered on all optics } prisms + ADM -> M5?? highed nize providiode signal and compensate w/ oc Here on Mon after noon y

1009? (i), diodus: 809 m 808.5 m Dave visit - 2 4/2013 -Little Suse each fiber: ~13 W × 2 : 26 W 19 light bulb pump green for Milleria w] inelfacency and conversion rates get the ms w 6 CM 6839 Aussications drive LBO-> LBO can get domaged if air is motist Na. YVOy gan Trunomi : can charge angle of H2, M3 do NOT charge the translation - do not change the fraction distances Millenia: gray box: courty for IR ? Shar coljastnerk for HR on end clocer to when case is olf - yellow tob at the Contains HIR end DC piping "briggs" IR to LBO carry front is a HR external HR twenus adjust HR for green carry when There arrived 23.5 A -> 13 W D2 808.5 mm Z Diode temp shifte werelength 22.4 A -> 13 W DI 809.1 MM 5 Check where info Overer and J A So A > end of life { Should be 13 W screen and J A ( 30 A > end of life { 809 nm w/ ~3 nm bandwidth gover and " current is Z0000 is the nominal ZO R- Theginning BSK is everage ~ 38K Need-6 br E @ 100 % concret . should be 5.5 W & Dove needs to make adjustant to the is the conce in SP mode ] Diode. 15k a piece do not scratch film! will need didde replaced on the board inside the laser herd Switch diode temp - changer power because you are trains to spectrum to the NOy spectrum La du not adjuit abbe Dave not set. pin 4 on the DIP switch montos weren's it pump power ideally, only adjust: is when there is a significant charge in - end micror running current, take a look at thege -LBO temp log! may need seasonal adjust to LBO timp Millenia mode - check Some distance ( 20 Gt ? ) pombre Trunomi - when bendwidth or pome over - don't hote charge, check Millenia (soly at low poverwith @ ~ 4.5W.

(2)W/ Millenia : ( 13 W from each alsoale should have to w Dove taen sets the current volume, from which the Milleria deriver power europer to quis 14 why the discours will run as -elufrente arrients is sheet is how Done gos 6 W disaler - look ok tringer on the board - offset for internal protodisals "should " replace @ 20 k hrs; can lest "SP Current" Service Level control on overge ~ 35k hrs and maxic NSOK Dove has set the pome curve - 100% curent should now result in 5.5 W 4 22.2 A 23.4 \* "Passbank everent" is when Millenia needle to drive the didden @ 2 A over max set point ercor Trunami: ( don't pamp w) more than 5 W - 5 could domage anings Brouster windows: were during could have been the same of the problem fuse can be removed and cleaned - usually only subside needs cleaning, but the muche of to Brown window were also diving dirly window - would disrupt the mode sufficienting Alignant look at external card overlap back reflection from HR to get louis Sextendi . from M3 in The floorment we at the overlap back reflection between My + MB during slar Equart. to avisible of -> look at dich optes up=redde tune center layer on My of M3 down : bluere con secon you of recover lasing look @ MS is all pill to center lake on MS Gue day Meut or acetane recover live here MG to get output intered on OC Ly this is when you try to maximize protodiode signal pumps on fl: pump should be between the "clows" of the holder ADM: RE frequencies sent into queriz constals of the RE frequencies help no de lock were alignment : 3.8 he prof ; 4 2000 (he ; 379 min 61 mm 12

168

Slit width - controlled by a set screw between the A and AA knobs sets the width between the rozors is lets more light through for a given GrVD compansation when typing to adjust pump power to get appropriate bandwidth

is make adjustments of N 100-200 mW at a time

ADM: when knocking, altert the power of the CW mode - locking - will see movement when want to optimize coarse caljustment to () gain mode-locking - will see movement when getting close

(2) retain peak power

40.03-40.58 7

verge d'bregvenerer necessory as you add or comove stars,

to metch

brequery of the cavity change

the ADM needs to emorge

" phote " changer frequency of the ADM

prave - adjust phase of the ADM signal to the carriery arequeray

Trunami alignment:

Rees spectrometer

- ~440.450 mW

- 300 non, cesteran a w/ ~60+ non bendwidth available

- mode locked "not working" need to first on AOM, Feduce bendwidth to ~ So an

then well bandwidth back of after modelocked

Pumped w/ 4.2 W

65 MARENIA: DI= 21.2 A DZ= 22.4 A

The ocean space devices we nove do not agree w/ Daw's

is they show w 10 non difference in Trunomi center

our device calibrated for VIS (as per Pam), while "Dave's 16 calibrated for NIR (and un just recalibrated)

a more spectrum has seen convected ("2013 02 05 Dove Alisn Final") by this should be the guide to which future alignments are made.

She

3

Spitfire (4) Stretches : - use external mirrors to align thigh Faraday isolator - use first the stretchermore mirrors after isolator to alignment tool prior to Stretche a chi wovelengt in the middle of the bendwidth - get Trunami to ly if the four dats of the grating are not averlapped, adjust stretches angle to get them overlapped we meror after stretche cid to first seed meror outside of the "collin" to get the beam aligned alignment tool before the perscope use perscope and seed mirror hefore extre-canny pockels cell to dign the seed to the alignment while aligning ! Deve recentered + ool 1 ( between the perscope and find seed mirror ( before extra-country Pockel's cell ) the ertre control The error call the following the visit of the 3 center on Pockets cell 3) center on alignment tool includes the canty and marror (the one would the photodicale) seed algrement inversioning Poetuls cen - need Find R they the alignment and -The to realign TEP: placed alignment tool Pove did < this after read alignment into canty between miora cavity mimor ( mar but hefore pump olyment Dove - Pamp alignment: extra-canty Pockels (ell) and the ildere of a get pomp lacing weakly (n9.7 A Wetween rod - rotated TFP until beam was aligned Here 2 to alignment tool > vertical alignment: press on mounts breads a of mid Jan 2013) sheps b Dave then placed alignment tool in slighting # Need Findl o align to alignment tool: front of the country meror that i monedicately to the fred follows the rod ( the first averos that - use first proponeror to align Ywi<sup>5</sup> Juriag pump to the alignment tool el. snowalt the pump transmite through ) and ituck before the rod used to other pump - transmitting - use second prop moror to align Cauty mirror to align to the tool alignment tool pump to pe affer the rod Checked that the beam was centered on the mirror julit before the end meror w/ the - at you beck reflection if promp photodiade behind it's used this mirror mirror on the other side of to align the seed to the alignment tool in front of the end murar w/ protodiods the rod Genecked that been reflection off - At this point, Dove rotated more place to the end murror returned -> pleved Any to get CW losing alignment tool in frant of the small E cavity morror before the end morror Ly looked at cord here compressor to by and see being Connel al the photodisals and adjucted moved wave glace between exotre canty ( LALY Pichels all and TFI-s try to get the end mirror to get the beat su back reflection aligned were seed into courty for helled 10 the other end alignment. Scan visitly see seed - rotate wave plate MILLON to maximize light

W wave place between extra carry Pochel's cell and TFP: -> where ever you can get the wave place to (5) - placed the alignment tool between and mirror ( w out photodiade) and intracementy Pochels scell: used law seed mirror to align seed to tool iterate ( - placed the dignment tool before extra-carity Pochels cell - used perscope to align sead - redid carry alignment: o end mirror ( w lost photodiode ) to aputure heline TFP o moved pump treasmitting cavity mirror after TEP thigh road to alignment tool . front of the other pump-transmitting muror omede sure been centered on party morror before and morror of protodisals a diqued beck reflection of endminor 6 looked @ back reflection between in front of rod Po for the second next looked @ back of a perture to try to see back reflection is monutury pressed on end mirror (w) photodisale) to try to allistinguish back rollecton from incoming send w beck-reflection "hopefully close enorgh 3, returned were place Is could tell lasting because of the charge in the pump is shift from bright real road to urangish rod " wolking" the cavity mode - can see the county mode well enough on the end minors In mohe a small adjustment to one and monor, complexate with an opposite change on the 2/5/2013 other end mirror TP TFP replaced - could not really the digned is made nation carring mode not aligned to the first kelescope mirror - Dave rotated the extre anode not aligned to the more man . catching polarizer to bring it back to alignment to the minor ) did not have much effect 3 - Jold TFP in oppics cabinet by knew seed aligned in the carty of Dave fell both Poetek cells were a little ous a ligned - translated both Poehels cells w/ CW lasing: taching up the power: Is apiterate herwean the fino, one area at a time to want to get the forward pass and the iture ( baserveblechon overlapped in the net rod next titled contry and mersons - do not do notoever the internet mirrors - these more already mean aligned for the seed and: work we why we compressor -> Tour got 5.2 W Pere we not able to make significant abayen to the wetput forking of the carry & Moved the first telescope mirror
enousing a-switch : "Pove get Ch 3 to high value (~ 25 round trips) than enabled 3,2,1 Gimportant only when seed is entering cont Q-surveheal ~ 10 seed: 5.2 W w[ seed : ~ 5.1 W Ch.1 and CAZ. separated by only MOAS -> should be more like 20 As? is Dave made adjustments to the two channels -> trying to get the greatest signal for the man pulses and suppress the post pulses No difference between juke tran when Ch. I enabled and disabled -> Pocheck cell not working correctly? Dove has requested Buildup reduction ( Dave removed plastic cove time should be off least another beards ARD 68RDJE -> should be 75J2 across each -> one measured 75J2 30ns - could be a the othe measured DD yovernight? high as 60 ns here on the only for extra Usually the resistors are the first things to go - Check these camby Pochuls call & first if a facture call doorn it a preser to be normy correctly 6TH ? essor: - demay from recompressed pulsenormel - happens else a few year -grenny demage - estate grenny 180° so the ber him it instead of the intuit recompressed atput? Compressoe : is also notired: what looked live a thumb point b Dave did the damage where the uncompressed pulse hits the grating 2/5/2013 rotated grating and measured to work the acrossation - pay attention for studing lines in the made smalled by extention to be before abground - could denote the joint they are present in the mode Auto correlation:

- use the first selescope mirror to align to the alignment tool before the first square mirror preceding the comparessor - we the second telescope mirror to align to the alignment tool when placed between the square mirror of elignment tool when

Nelto Autororrelator Miniophic Technology, Inc. N & S.K.

Compressor alignment:

Commissor

(journeni alignment guide: AND to focus () M2+ M3 focus - when lasing is achieved, should small adjustment, be make to makimize power, or should we just lieve them as you have them?

(2) Mulence to 5 W? or only for instell? -> shouldn't be necessary

172

growing angle: 650201

spectrum: centrical @ 798 nm ~35 nm FWHM

is should be symmetric ~ adjustants to confirmer ground and get \$3.5 fs pulses

Yesterday, Dove had returned the grating tority orginal mouting arongenera (mailing on tay). however, he had shifted the grating one to avoid the "web neve" -> This cannot the basis spatially chipped bass in the compressor to clip on the v-minor assembly not on the mounted deley 4) Due moved the grating back over to ensure the bass did not dip on these murrore

from the strape of the w. yes in the auto correlation, you can see if there is clipping in the compressor

In ~ 1/2 trans then compensate up motor delay ] keep incoming

compressor: opennal angle given by where the shortent pulse is a churved

not can be get extendly stretches: align to get the \_ } extind control: only ~ 1/8 threas

Stretcher and compressor: there are been screws that provide external control for the graning angles

actually yield the grope pulse length

Ly guild up time reduced my on extra ~ 20 m

when one of Dese's humans cards as presed at the exit port, it now looks purple rather than real La it "lookes like an ultraface pulse" when Ch. I not working properly, the cauty in not herry properly seeded, limiting its ability to

New pulse picker board received - replaced 2/6/2012

wil wrong orale: just see bright spot w/ connect order and oligned Pockels cell:

shaded areas should be dark.

Proveresend

set up one polenser in front of the Pochels cell once the other after is the two followers should be crossed - order marken

the pieces of polonzer

Pochels cell alignment (extra carry):

two palaaner

Spitfice, continued

Dave meanined Spithne pulse width w/ autocorrelator - ~97 fs

Townami reprace is now 76.11 MMg Les to get more bandwidth, Deve had to give the Toecam more gian more glass increases the effective length of the cavity to lower vep rate (Ŧ)

٣ inside compressor : the pors of light all not have to oucles on the mirrors on the motorised delay on the compressor gran-y, should see 000 Compressor grating: ~ \$ 5k deter 0 see on the microis on the motorized delay, okay to compressor retroreflector 03 Lewold probety need to go beek to factory to replear Spitfire: pump oppies : Mirror the tenses in the black holders actually have the optical elements the block holder also the second pumpmissor actually contains both a polarser and a lens Is in order to clean these lens, need to remove the polarse to get at the second surface to the lens has the soft AR coating 13 passes for build up 2 30.40 as build up reduction time 11-12 packed for build up 3 30.40 as build up reduction time w further edgustante : 4.4 w 2 cw 5.35 W => Qrowsteen Chi 103 Eh2 123 Ch3 282 Settings 5.2 W >> seeded 4 W post compressor ~ 41.3 fs pulse initially w former adjustments to compressor: ~40 fs output : usually shifted from seed centering a few nom bandwidth ~35 nm? have factory speed for Spithire - see email from Dave will service reports Sptfice 35Fs on 243 Test Report. pdf single and pumping should increase build up time by 1-2 round trips when operating spitture in free running mode mostore } measure spectrum - will get better performance it we a-switch mode (but u/o seed) } measure spectrum - will get better performance it we metch the Tomami output to the spectral properties of the non-seeded carry.

> pulse proces soard PN 90022027

w new offices setup: Hang inched towning: the still (no ws) ~ (00 www STI (no ws)
TOPAS Herry: 1.8 W Solly initial touchup:
sail the white light aperture will end up aligned to the aperture in front of the
align to external apprint and the pre-amp BBO?
while light - do not work the concentric circles
work it to be circular and stable
is which delay the arroad mirror of the retroseflector has adjustments for alignment thigh the open-re
while light wanty for scher and BBO
the time in the and part the align the signal through the same that the and the aret the
1145 and the million of the the part (1900
appeare
if hocus rates the crystal (C2) is the tight, can get while light guardon - " OPET =>
Harry: ~670 mw St ( no WS)
to alight capital in mine stops. To the minet Sally: ~670-680 mW StI (no WS) ("Ist in
I halfe for the mixer behavor:
to touch ap overlap of the artput, There adjusted the post of part of partition in the
pump ourror just where c2 (a 390 nm
w/ adjustment for collinearity Harry : 340-350 mW
Hany: StI w/o WS: ~670 mW Sally: 220 mW > howevery loos and yellow then orange
the same moved to a new desaproched from the oph,
nower @ 690 for Harry: ~250 mW   actual frequency: ~ 250 nm
a the second second stated more power
G Pare didn't feel that things were right - prisons of filler 1
which (webser (current) DOB and Miker has files for both OPAS soved and should in
Solivare: Door motored entry control perkapp (2013 0206 Dave usit initial trong)
The installed the New WINTOPAS software (v 3.2.236)
your men unned the configuration file folder configuration ( 1074/x on the dove received
from light convesion to Program File (xSG) / WINTOPAS3
to get : hourds get installed on the men board that is when the USB and power connectors are - copies then run from the locale of
Devices: 1060 new the boards to the mixer connectoric at the rear of the OPA.
the TORAS in WINTORAS: Service > install license Cable across the bottom of the bread board
need to retodd -> here to reload all curves
TUNING CURVES "
Successederations DI has shifted - should it on a day to day basis? NO
Since your of there winto Pass installed on my
"send motors home = reace   scrass where the contract to recover positions   lap top

installed now mixers on both OPAs		
		Ô
50113: ~560 9mW S+I > 6	50 mW w/ for the adjustment (@ 1300	priane BBO : type
w   Mixed installed at	the head of Sully:	block all other processes except sead
undervices Herny:	fac 1300 mm is nee	ideal to send DI to ~6800
ning com	OPA 2 = CZ has	heen replaced, as pur Andres
	Usinstalled i	wrong?
configuration files - the allesses in Harry's	Foomw E finally got OTA 2 W	vorking w/ an old
conf. files have been changed . in prale	S+I Configuration hile	- why day org cregoore a
to use our prove files aread the correct	different cont hu	
elik xer		
la created hybrid conf. files that	ellow Herny: ~ 700 mw	5+I
Miker 2+3 to be used, but wh	·ch well ~ 100 mW	@ 590 nm   When myer skowered
w our bring europes	Dave E by tuning n wash't too	notor: NBBO mW attached to Harry
	@ 11 DP AM	:~ qo mw
Dave: nextweek : end of mextmeete?		
complete alignment of	Sally: ~620-640 mi	W StI
TOPAS	N230 of large & N175-180 @	590 mm => Shill more yellowish
damage in Spithre ?:	highill looks more yellow then ore	nge
- gaining		
- comp. deley retro		
and and anothe entitled		
HR demage : do not port apoints before the OPA's		
helpore Dave visit:		
-Skye - crystals?		
- filtos?		

	15.4 -	40	TT N	6.4	8-9	2.0	13																				(
	1917 -	~ 6		ic q	0-3	20																					
1																											
13-06-001	3103		te	mpe	-		e vr	npor	TCAT																		
			٦/.	1"	ade	quete	for	Tsu	nami	510	6111	1 /	10 1	eed	or 1	he er	ktren	~ ol	6	0.01	31	0.1	•				
			Jale 1	over 1	مرامه	rheri	14	the	Tern	ami	ð	1.41-0	ant	+-	Lett	L OU	e 4	00 A	ω.	when	cw	, ce	ante	e d		1.1.160	and l
			6.		2								7				1							- cu	6 1	(110	
			0,		<b>~</b> ′	000	1 ~ 0	140 M	IW .	ashe		D dul •	1000	rear													
				L <sub>o</sub>	51711	G	1446	Unst	roble	, 10 -	* De	ve el	lioln	من سة	cnt to	5 De	inal										
l	Dee																										
a pressue :	Vanc	/~~~	L 54		-a jui	4 Mellin		-17488	e up	çera	psim	3															
	CHECK	ed -h	u 104	ndw.	dth	detec	101 -	medi	eng	و دمام	dyne	ment,	s-al	100100	l et r	æ											
EMPOWER :	Dave	And A		0.111	20	<u>ه</u> در	+ 21	o. 1 A		E and	meel	20.	wa	r thu	custe	+ un	سماهب	11	-	checked	-14	Meas	vene	4-	Dave	1'5 401	FA and
			)														-1	10		4							
																			OULLI	40 M	ol Me	tch u	p -3	OVE	40++	4 13 4	1.116
	LBD	ten	p col	mst :															mis	alibrak	col, (	20	w,	diffe	ence	is 4	Ιω,
	(	there	ore -	wee	(olts -	n n	- 1eft	side	c of	-14	Empo	. Bre	The f	er left	port w	hen			b-+	whet	· v9 -t1	u d	Gerene	a @	low	e pou	ws?
6	~ ζ		C	h			1.2	2.	Cat	be 1	A	ver le	<i>ml</i> .		•											· .	
here is an	L		Part	) 14		indian.							.1.														
openn)		Can	Ne	lure	4	volt	ng o	f fu	que	MO CO	16	in the	LBO	oven	with 1	νD	-sub c	PUVUC	the l	nte	back	art	whe	e 14	e um	bilice	l attac
ATLE S. OL U			5	M	in	No.	(tage	00	7ms	4 m	d 14	f.															
he season																											
stal pump	2						0	-																			
chower "	mee (			Arri	n <sup>1</sup>	V		And	, de c	- h-			Gr	. her													
chowner or these thread	pon (	Ţ	)one	doer	lan	know	بمناص	, the	o flae	· fro	. pot	s ine	for -	hos	Vene	reode	ديل اله	cd	ju (+ )	ne m							
chomer in these three ore	nrel pon f	ر د	)orl	doer	lan 	know	بمن	, the	o flae	· fro	. pot	5 ine	for -	hos	Neve)	reo de	ديل له.	cd	jn (+ )	ne m							
choover u these three ore	nee pon f	ך 4	)oul	doer	nol 1	kaon	ياعمادن	, the	<sub>0</sub> flue	· fro	· por	s une	Gr-	hos	.∧e ve)	reo de	داباه.	cd	.ju (+ )	ne m							
netal pump character u glasse glass ore	nere por	- <del>-</del>	)ort	doer	nol 1	kaon	ياعيان	, the	o den A	, fue	. pot	s une	Gr-	. hos	Aeve	reo de	داب اله.	cd	just i	ne m		L.:	1 tem	0			
netal pump character u these three	nee pon f	-> - <u>7</u> Libe	)oue ove	daer - te	nol m()	know 15	peaire	. The	0.4tm	· fno	, pot	s une Tru	far -	hos becke	Neve A re:	con	دا اه. مرب	ed.	just t	1. (m. 1.	- 0	nille	s bea	ę			
cho mine v cho mine v fleed flored	nee polin rhe	-1 -7 -7 -7	Jour Oue	doer - te	not mp	15	peake	, the	0.4tm }	· fro	, pob hurgi	s une	for -	hos beebe	∧eve) A re:	con	دا له.	ed. over	juct d Emp	1.e /m 3.Latter 1	- c) - l	hille Do	s tem	ę kemę			
retal print charve v flass three are	rhe	<del>ب</del> د دئ	)one ove	doer te	.mp	kno	peaire	the col	~ 4in	· fno wo t	, pot	s ure Tre	for -	hos beebe	Aeve A re:	con	دی له. جری	ed over	just t	12 m	- c' - L	hille _05	s tem	ę remę			
netal pump charve u fleer three ore	nese point	-> - <u>7</u> Libo	ove	doci -	mot mp	know is Emp	peake	the ce	o den to	wo t	n pot	s ure	for -	. hos	Arve) A re:	con	دل اه. دسته	over over	juit à	1.c m 5.cm t	- c' - l	hille Do the	, tem oven	e reme al c		, nçide	
etal punt channer u these these	nere polin The allow	-> - <u>7</u> Libo	)one ove	daer te	not mp	kanon is Emp	peaire	the ce	o fire	wo t	n pot	s ire	for-	hos heete	Arve	con	دل له. ۲۰۰۰	over	juct t	nem suo:	- c' - l	hille Do the	s tem oven	p temp		, nside	
etal punt charre u flase flee	rhe		OVE	does	mp mp	Kaono 15 Emp	peake	the ce	o the	· fro wo t	r pob	s ine	for-	hos beebe	Arve	con	دل که دست	over	just i Emp	nem nuy d	- c' - l	hille -Do the	s tem oven	e seme		, ~<.de	
retal purit cho mire i fleer three ore Spither.	epote pote rhe subme	ب م لنگه اند (۱	Ove ove	doer	mp de tre	is Emp	peake	, the	o fin	two t	pot	The not	Br -	. hos herbe	Arve	contra	دل له دمنه ۲۰۰۰	cd over	just à Emp	nem e	- c) - l nside	hille Do the	s ten oven nets owy	f pemp del c incre		the pro	
etal punt cho aver the fleer the ore Spokhre.	sure parts sure sure un Gree	ے میں لنگوں الدی الدی الدی الدی الدی الدی الدی الدی	OVE OVE	daes te	not mp d the	is Emp ave	peake peake	n ree	o fine	two the second s	pot	s re Tru compre	for -	. hos becker .nt	Neves A re: ulock-	con the	id is	cd over eleck	Emp Emp	nem nem i sue:	- c' - l nside cal m	hille -Do the	s tem over nets only	e kune incre		nside the pre	
etal punt cho ave v free the	ener part	ب مر م ل تاریخ ال تاریخ ال م	Ove ove enve Dove (	daes te soore soore soore pol	mod mp at the ed th entropy	is Emp	peaks peaks initial anitati	I The contract of the contract		two d two d two d	pot	s re Tre more compre	for -	. hos her be in t	A re: A re: 	coni the place	ed to	over electronic	Emp Emp one c ofte	nem my d	- c) - l nside end m	hille the for	s ten oven nets omy	f fem f incse		the preside	
netal purit cho ane u these they ore Spatre.	enter parte part part part part part part part part	<u>م م</u> م ل 30 ال 30 ال 10 ال 10 ال 10	ove ove	doer te	mp a tre d tre en bal	is Emp interpolation	peake peake inihau angut	n phe	o the	wo the	poi	True not	for - ve c' ssoc ; assec	. hos here here	A re: Ulock- ng prom ( for	cont the place	ed to	ed over	Emp Emp one co	nem ny d	- ci - l nside no nol	hille Do the	i tem over over	l ken l incee	, , , , , , , , , , , , , , , , , , ,	nside the pre	
etal puri cho ave u freet free ore Spinhire -	ere por the	<u>م م</u> لرژه د هد (۱ د م	ove ove exerce pove ( the	a te conc nod envov pol	mp of the le, D ed th eribol	is Emp inconf then of	peake peake subscript state the	n rhe not in re non ri r and role	o the	two t	pot	The second secon	for - ne c' ssor ; asse ; asse ;	. hos here here there here here here here here here here	A re: Lock- re prom ( the	contractor	ed to	over	Emp en o olte	e pope	- ci - l nside end m no rol	the	s ten over	l kml incse antro		the pression (	
etal punt cho ave v freet for ore Spatice.	ense posis	<u>م م</u> لدگه هه (۱۱ له .	ove ove exerce powe the	daes te mod emov pol wher hulg	mod mp d the ed th en bol	is Emp are polition of enny	peake peake initial initial initial initial	rhe in re g som ris roan ris roan ris roan ris	o the	two t	pot	s re True compre lans compre lans compre lans compre lans compre	for - ne c' ssoc - asce to to cob	. hos int tweak	A re: L lock- ng prim ( tre i bi L hold	contre tre place	ed to	ed over eleck ad	Emp Emp are a alte	nuy d tures	- c) - l nside ind m ab nd	the provident of the	s tem outer only only	l kml incse antro loer	r). ( art b	the preside of the president of the pres	
etal puri character flass ore Spikhre.	euse pos	ب مرابع لیکه م ا	ove ove enre powe fhe co	doer te mod pol wher hully	mod mp of the end th end th end th end th end th end th	is Emp are pole bron of Sect of	preases preases inchations inchat	r the r the r the r the r the r the r the r the r	- the -	two it	pot	s re True compre lans fror bo he bit por por por por por por por por	for - re c' ssor ; asce whys whys	. hos here here house	A re: uloce- re prom ( fur i bi i bili a bin	con the place	ad po uns unt	over over na ide	juit + + Emp en o ofte	ener my d terre	- cs - l nside cod m lo nd b oni	hille Do the correct	s ten ouen nets ony	f kmf incre niro		the pre-	
etal puri cho ave v fleer fleer ore Spuhre.	ense pos	ـــ - ج لدی هـ - در م ل م	ove ove nous fle co	deer - to conc nod pol nulg cons	mod .mp .d the le, D .returned .returned .returned	is Employed	peaks peaks our inihisti inihi	r the code in re- r source r and room r c r and room r c r and r and	o flue	two it	potential pre a pre a pre a the cohor	The second secon	for - re c' ssor ; asser ; asser ; helps riss o	. hos here here here here hose here here here	A re: uloce- re prom ( que b b) b bolo	con the place	ad to	over	Emp Emp are a alter	e pope	- c) - l nside nd m 20 m	hille bo the correct correct	, ten over	e seme incee normo		the preside	
etal puri cho ave v fleer fleer ore Spatre -	rive	ـــ - ب لريکو الم	ove ove enve fle co Dare Dare	does to some pol hug pol hug hug	mp mp di tre te, D ed th en ball i retu el i retu el	is Emp ave romy Vect of Gov	peake peake one initial initia inita initia	I the I the I the I the I the I the I the	house	two t two two two two two two two two two two	pot munge pre n i the color g littl w m	s ne Tre compre lans lans la bit bit o po o po o po o po o po o po o po o p	for - ve c' ssoc ; asce or to cete helps clear	. hos hos here he hoseby . a d deg th ac ct	A re: L hold a his c his c hold	reade cont the power place	ens	over over one ide	Emp	e pope	- ci - l nside no not	hille bo the correct	1 tem over nets ony ony	e incse ncse	e (). ()	the president	
etal puri cha mine v flass flass ore Spinhre -	rive	ـــ	ove ove nous pove 1 the co Doure Doure	does . te conc pol nug pol nug	moi mp di tre te, D te, D ter ba- i ret criba- i ret criba- i ret	is Emp are polition of Farmy Vector of Good	peaker peaker ours initial initial initial initial initial initial initial	H (Ce H (Ce H (Ce H (Ce) H (	· flue ·	two it is the contract of the	pot	The second	for - i no assoc ; assec to the helps olean	. hos hos here he hoseby . ~ 4 hose the ac cit	A re: L hold a hin a hin a hin	reode cont tre plac b, j tou	ad to	ede over	Emp Emp one confe	e pope	- cs - l nside end m rel end m 20 mil	nille Do te Correct	s tem ouen nets onny C	l semp incse incse	(c)	the provides a construction of the provides of	
etal puri cho ave v flase flase ore Spatre -	aver (	<u>م م م م</u> <u>م م م م</u> <u>م م م م</u> <u>م م م م</u>	ove ove nmos fhe n Dove fhe ch Dove ehme	does - to - to	mos mp d the le. D i reth cl i reth cl i nved a free seed	Emp is emp enve man source is is is is is is is is is is is is is	peake peake our angut the angut the	I she in the interval in the interval i	o the	two it is in the second	pot	Too more compression compressi	for - le c' ssor ; asce belgs riss 2 clean 16	. hos . hos . n + . tweele . n + . n	A re: L hold a his a his a his a his a his b hold b ho	tous	ens unt roa k ho daving	ellecu and Ide	Emp Emp one c offer why on	ner i ner i ener c tertes f pope d pope	- ci - l nside cod mode e oni	nille Do the further thistop	s ten ouen net	e kane increa	encel i innel i ort b	the president of the pr	
etal puri chi ave v flase flase ore Spahre -	aver ( par)	<u>م</u> م م ال ال الم الم الم الم الم الم الم الم	ove ove nous fle base Dave ching	does - te mod pol nulg pol nulg pol where mulg pol where mulg pol mulg pol	mos mp d the en bos s cet s cet s cet	Enoro is Emp ave polition Sect o	peake peake our instatu angut the angut the core gi	vi ce vi ce vi vi ce vi ce vi c vi c vi ce vi ce vi ce vi ce vi ce vi ce vi ce	o the	two it	politica pol	s re Tre compre lans lans laber labe	for - le c' ssoe : asce belgs riss o clear 16	. hos 	A re: L hold a ha ha	con: the p muri plac b, tou	en se la	ed	Emp Emp and offer alter vuly on	ner i ner i ener c ter sca f pope d pone	- ci - l nside cod m d e oni	nille Do the correct correct correct correct correct this correct this correct	i ten over	e kme incsee inco incsee incsee incsee incse	ese s eneret ene	the president of the pr	
netal purit channe v flace they ore Spinkre	and a free of the other other of the other other other of the other	<u>م م م م</u> ل ککه د ل م م ل م م	ove ove enve fle un Dove ( fle un Dove ehing	daer - te mod pol mug pol mug pol mug pol mug pol mug	mp mp de tre te, D ten bait i ret i ret i ret seed	Enors is Emp ave polition is fect of is for is for is	peake peake our initial angut the angut stoke	n ce n ce n ce n ce n ce n ce n ce n ce	o the	two it	politica pol	s re Tru compre for for for for for for for for for for	for - re c' ssoc ; assec ros p ceb weigs olean 16	. hos 	A re: L hold a ha ha	con 	ed to	ed	Emp Emp and contraction offer why only only	ner i ner i ener c ter sca f pope d pone	- ct - l nside cod m rel e oni	nille Do te	, ten oven nets omy omy omy	e reme incre doer doer	ese s energy ( ort )	nside the processing coon (	
netal purit cho mine v flace they ore Spinhre	The order	<u>م</u> م ل که د ل که د ل که .	ove ove enve fle un Dove ( un Dove Dove ehing	daer te mod pol wier hug pi wijo	not mp d the enter sed th enter second free second	Enop is Enop ave a polition of Formy Vector of Gove	peake peake our initial antitic etobo the g the core gi	n ce n ce n ce n ce n ce n ce n ce n ce	o den	two it in the interview it is the interview it	pot hunge pre 1 is the cohor e for W dh	s re 	for - ssor - asser - asser - to che treips oche treips oche treips oche	. hos . hos . n + . tweele . n + . n	A re: L hold L hold L hold L hold L hold		ed do	elect over and ide up cr	Emp Emp en o offer user m vuly on vuly on	num i num i num d tersca e prope d prope d prope	- ci - l nside ind m rel e oni - - - - - - - - - - - - - - - - - - -	hille Do the Curron intot thisso	, ten over nets omy omy c ton.	e reme incess doer doer Apl 3	are i area area area area area area area area	n nsi de tre pre Goon ( nore e	

												(L)
Beforetucing up	seed pointing	Dave had	SW	w/ 13 roum	al trips							
for seed adjustnes	rt. Deve m	ed the first	seed music	or este th	Le stretch	ne (ort	de the	"calha"	) anal	the pe	-igcope	adjustmet
in tonder	to walk	at them be	eed close	- da dhe	carry m	use and	sprmise p	buen .				
a the beauty T	muy news	Deve the	real the	sens d	elay caly	Athent						
4 the	on the data	1. he has of 0.0	the ara	a from	and the	ceed ce	ld Choose e	all a se a	1 1 0 0 0 0 0	hand and a		the set in the set
		1 netroes		1		Je Je		GII COLT				ind and rout
10	the loor											
with his adjustn	ever, Dove g	pot up to 4	W post	compressor								
in regi	wei 13 row	nd trips										
Deve was reasonal	ny serviced	with the p	ump clisnow	ent hermes	in the Spit	fore coal .	na ofas					
Ally (10949)												
~1300 nm:	seed overlap		-3.867°	-these	losition	used from	Dan's 1	mongener	- should	he ~ 130	5 nm	
		DI: 2	1.222 mm	2								
												not sure whether
Dave means	Lol ~ 1.8 W	out of spi	thre to 1	10943								1 9.0
										for ceed a	ne	BOAWAA 1
@ 1300 nm 1	" seed "look	is methy red	i . konsta (*	the second	Justine ceal					RTN-P	Rever (	Dahue NOVA II
			_		,,				1	,	0	2202621
@ 1200											1	PALL TEOLEU
C. [ Dec. VW	Dave op	HAMIBER CO	whereas ,	cl, and	TD1 ++ M	LXIMIZC	ru psue	same q	ead on n	eer-reel 1	y-	
	where a	his god	war to r	reach ~ 1	1.5 mw	<b>c</b> .						
	only m	varaged ~ 1.	( ~ W w	1 quere a	degrees of	preceion ;	Dove he	gen opmmi	zing int	unal cla	gren o	( meedon
							ц	mostly ju	154 pre-	and brend	p optic	3
	Danke a	to adjuncted	the focus of	I the point	r= preamp	prop- u	ented to	m	e the	"superfu	onesien	Le" in the CI
	on the	o panc just	before c	1, Dove	well a	card to	mehr sur	e to no	mp end	white 1.	tht we	e eveledone
		4 + 0000	100 95.W	ante -	hume to	al white	lyht co	lineer			,	
		v v.	nac con		point		,					
	aliu m	e adjumma i	to the pump to	end pro me	my, card	i only ge	A ~ 250	μw				
		~ additional	y, Dove	wa no la	orge yee	-J the re	a seed	post pre	ang			
						0					to and t	he red back
		por no alling	justant to	frunt pocos	(returning	the focus i	o original	Corres	1 - 1200	manayear	0	
	w ( noe	with on all	justant to decrees (c	omercision.	DI CI) m	the focus i	break 1	nw; T	) contract	tweeting mir	0 0m - m	ngading manageal
	w ( me	with on adjust	justrumt to degrees (c +-~~1.1	ompression,	DI CI ) ~ reasing du W	no focus : renegeal to have light a	break 1 pethre o	nw; T	reaced in	tweeting mir	on - m gniheart	nget a menaged
	w ( ove get	with on with releving exitensis to the power	justrant to degrees (c to ~ 1.1	ompression	b (retrining DI CI ) ~ reasized w	the focus i	breck 1 pethre of	now ; T	) cove sneet	tweetery mir	on - m gnihean	ngeoting managed
	ω ( «τωε 	with on adj intering existences to the power decided to a	justrumt to degrees (c to ~ 1.1	omercision more focus now, - Inco	b (returning DI CI ) ~ realing du wi	the focus i	break 1 break 1 pethors of	new ; T.	rected in	tweeting mir	on - m gnihcant	nyndras menegeal. Hy
	w ( mee aget Deve	with on adj intering existences is the power decided to si	justant to degrees (c to ~ 1.1 hack ~ 1.1	neression nov Inc 19 now have	blict) r Dlict) r reasing du sui	the focus i	breck 1 breck 1	now ; T.	() -() ove	menagen mena mena mena mena mena mena mena me	on - m gnihcant	ngnobra menegaal. Ng
	w( eve get Dove	with on adj ching external to the power decided to si	justrume es descrees (c +- ~ 1.1 hc/2 ~ 1.1	omercission, now, - Inco 18 now have	b (reterming DI_CI) ~ reasing du w	the follows i	breck 1 pubre of	nw; I	ened the	threading on cr	on - m gniheant	ryatra menezial. ry
	w( the get Deve	with on adj ching exitend to the power decided to si	justrume 40 descrees (c 4- ~ 1.1 Incle ~ 1.1	0 promp focus 0 mp ((53),00 , 0 mb , - Inco 13 now how	b (rectanning DI CI ) m reasing du w	the follows i	o ongina break 1 pethre of	now ; I	con ined	managen hurenten nurr	) ore - ore grificant	rystra menegad.
in orders serv	w ( me get Dove ice fechnician	with on adj colling existences to the power decoded to si access : source	justrume en despeces (c e ~ 1.1 mck ~ 1.1	. prof focus omp(6551.00, mi) , 1000	blict) ~	the follows i	breck 1 breck 1 pubre of	nw ; I	1 -12000	monayers	) ors - or gnificant	ryatra menegad. hy
in plan Serv	w (-mue get Dove ies technicien	with on adj whing exitensis of the power decided to si access : sour	33354 mm 40 desgrees (c 4 ~ 1.1 mc/2 ~ 1.1 Cr	. prof bacus 	blict) ~	the folus i	breek 1 breek 1	nw ; T	, -, ) over	meneyen tweeter mir	ors - on gnificant	ryatra menagad.
in plan serv	be be Deve Deve ice fechnicien ice fechnicien	with on adjusting extends to the power decided to si access = ser toning curr	33354 min 400 deligiees (c 4- ~ 1.1 sincle ~ 1.1 Ca Ves	. prof bacus 	b (retroning DI CI ) re reasing du w	the follows in the fo	breek 1 breek 1 pethre of	now ; T	, -, ) ove	tweeting on or	on - on gnihean	ryatra menagad hy

										3
using the WINT	OPAS Fowhere	created a tun	ing curve her ?	Sally's pread	mp i					
power and	alignment	and tring								
0										
C 1300		ve used the	loss two fromp :	r cram to cent	e te pun	P 20 C2, 1	her adjusted	the menual co	mot for D2	+0
	find p	wanterne output	њ							
	C -									
while adjusting for	1 400	Incenty Pove	adjurned to	delay 50	the seed of	sd pump u	rese not ben	merally overla	ppeal	
collinearity, "Dove notical		is could se	e the doubled	pump (Blie)	) and th	e send (red	1) able the	HR		
that the second parring		Dove used	the last the	bowb ship	ice to ou	elep tuse	in the near fi	eld (just-out	ide to OPA)	·
we ob, so he adjusted		and the	fer hold (do	n the tab	06)					
it - only used the true		عيره و	overlapped, D	ove tixed i	ke sempar	el avelop	and touched	up the collineer	ity in the for 6	ield
comes marries ale										
СІ	1									
	114	0 nm 100 mb	<b>.</b>		after com	plabing the tran	ngenre, Dove	noticed ther	there was OF	G
	110	0 mm 240 mw	)		in th	preamp -	I hely the sou	rur of the inst	chility we	
	1.8	0 nm 300mb	<b>)</b>		were	seeing in t	the second her	s of the curre		
	(2)	00 mm 370 m	ω		4	from then D.	we changed 4	ne switch out	tomes?	
	121	lonn 400 m	<b>ы</b>							
	124	10 nm 420 n	.ω		to five the	presence of the	DØGr:			
	124	0 nm 440 m	w		Porc	adjusted the focus	of the preamp	fump - changed	the land to the	N40 I
	121	bonn your	.w				mize of Cr			
After this part, Dave	13	00 m 460 m	w -> 480mw			- beep	the pump goin	s my a aper	NTE.	
tried adjustice ne	151	10 mm 480 m	ພ							
switchout times on	134	lon 490 m	ω		thus	adjustness da	pered the pure at	1300 nm 40 00 49	~ 400 mw	
the spathine	134	on 460 m	ω.			is charged	the focus aga	n, stylidig		
	138	0 am 4301	wψ			power inc	reased to near	4 480 mW @ 130	Pom	
	1400	pom 440 mi	u l							
	147	lonm 430 m	J .							
	144	0 mm 450 m	w							
	14	10 mm 450 m	3							
	[48	on years	3							
	151	engen 140 mbi	<b>)</b>							
	151	lenn 420 mW								
	15	40 nm 420 mw								
	151	iona you who								
	151	10 nm 300 mW								
	[66	Wa 105. ma								
	101	0 mm 300 mw								

142										(4)
Harry Turing	-									
when celthing on	initial aligo	ant in Harr	y. Dark notic	ed souther on the	minor just holy	· CI- cu	cancel it to rea	nove an dire		
1 1					J ·					
() 1300 pm:	Motor Co	ntrols increased	seed power f	6 ~ 0.95 mW;	internal painting	adjustones	h increased	the power port	Imw	
	Dove col	junical the st	etroceus on c	in optic before	the white light	- delay ( T	)) - adjustn	me of neutral	density (v	encode A
	Ger	the prop go.	- 40 40 L	share light pl	ave .					
		ls it open	re is open i	all to suy, choul	al get white I.	ght - it wi	11 he sat-rate	al, ht should	he true	
		Dave m	a seen "	the chief highly in	has he opened +	e white her	it aperture cl	n the way. He	adjured the	neutral
		charlete	measure of	This. Haywar		N.000 MIND	n c cherge	ne con make,	they'r ne gi	nearly
		210000		•.						
used to	Creeke new	has the	e "optreal	WINTOMS 40	stor 0 4 9(.051)	for the are a	no poutaos.			
		,			C.v-	``				
Doue spisming	ed Collines	in which	another the	noves amp Sta	ye.					
re optimised	compression	to Sally Li	0943) - AD	, to a state mus (	2 1300 art of	Harry (1070	12)			
4	when sim	ry optimised	al compress	on to Harry	(10742) po	amp	got mybe	580 mw		
				,	'		, ,			
	to an	150		1520 m	550 mW					
	10 nm .	150 mw		1520 m	550 mW 510 mW					
. [/* .(\ .(\	10 nm . 10 nm .	150 mW 340 mW 420 mW		1520 nm 1540 nm 1660 nm	550 mW 510 mW 500 mW					
ן! ונק ננק	ti) am 10 am 10 am	150 mW 340 mW 420 mW 480 mW		1520 m 1540 m 1660 m 1580 m	550 mW 510 mW 500 mW 450 mW					
ןאי תנק 129 121	ti) am 40 nm . 10 nm . 10 nm . 10 nm .	150 mW 340 mW 420 mW 480 mW		1520 nm 1560 nm 1580 nm 1680 nm	550 mW 560 mW 500 mW 450 mW 490 mW					
(P (L) (15) (12) (12) (12) (12)	ti) am 10 am 10 am 10 am 10 am	150 mW 340 mW 420 mW 470 mW 500 mW 520 mW		1520 am 1540 am 1660 am 1580 am	550 mW 510 mW 500 mW 450 mW 440 mW					
(1 	10 am 10 am 10 am 0 am 10 am 10 am	150 mb 340 mw 420 mw 480 mw 500 mw 520 mw 520 mw		1520 nm 1540 nm 1660 nm 1580 nm	550 mW 560 mW 500 mW 450 mW 490 mW					
[[* 	ti) am 10 am 10 am 0 am 10 am 10 am 10 am 10 am 10 am	150 mb 340 mw 420 mw 470 mw 500 mw 520 mw 550 mw 540 mw		1520 am 1540 am 1660 am 1580 am	550 mW 500 mW 500 mW 450 mW 490 mW					
(/* 	10 am 10 am 10 am 10 am 10 am 10 am 10 am	150 mW 340 mW 420 mW 470 mW 500 mW 520 mW 550 mW 550 mW 580 mW		1520 nm 1540 nm 1560 nm 1580 nm	550 mW 510 mW 500 mW 450 mW 490 mW					
[/* 	t0 am 10 am 10 am 10 am 10 am 10 am 10 am 10 am 10 am 10 am	150 mW 340 mW 420 mW 480 mW 500 mW 520 mW 550 mW 580 mW 580 mW		1520 am 1540 am 1660 am 1580 am	550 mw 500 mw 450 mw 450 mw					
(/* ///////////////////////////////////	t0 am 10	150 mW 340 mW 420 mW 470 mW 500 mW 520 mW 520 mW 530 mW 580 mW 580 mW 580 mW 580 mW		1520 am 1540 am 1660 am 1580 am	550 mW 510 mW 500 mW 450 mW 440 mW					
*   15   20   21   21   21   24   24   340   340   340   340   340	10 am 10 am	150 mW 340 mW 420 mW 420 mW 500 mW 520 mW 550 mW 550 mW 580 mW 580 mW 580 mW 580 mW		1520 nm 1540 nm 1660 nm 1580 nm	550 mw 500 mw 450 mw 450 mw					
(*  ()  14  12  12  12  12  12  12  12  12  12  12	10 am 10	150 mW 340 mW 420 mW 420 mW 500 mW 500 mW 550 mW		1520 am 1540 am 1660 am 1580 am	550 mW 510 mW 500 mW 450 mW 490 mW					
(/*) 11.5 12.5 12.1 12.1 12.4 12.4 12.4 12.4 13.4 13.4 13.4 13.4 13.4 13.4 13.4 13	10 am 10	150 mW 340 mW 410 mW 410 mW 500 mW 520 mW 520 mW 530 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW		1520 nm 1540 nm 1560 nm 1580 nm	500 mW 510 mW 450 mW 450 mW					
(/* ///////////////////////////////////	10 am 10 am	150 mW 340 mW 420 mW 500 mW 520 mW 520 mW 550 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW		1520 am 1540 am 1660 am 1580 am	550 mW 500 mW 450 mW 450 mW					
(/* (/*) () () () () () () () () () () () () ()	10 am 10	150 mW 340 mW 420 mW 420 mW 500 mW 500 mW 500 mW 550 mW 550 mW 550 mW 550 mW 550 mW 550 mW 550 mW 550 mW		1520 am 1540 am 1560 am 1580 am	550 mW 510 mW 450 mW 450 mW					
(/*) (/*) (1) (1) (1) (1) (1) (1) (1) (1	10 am 10	150 mW 340 mW 420 mW 420 mW 500 mW 520 mW 550 mW 550 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW		1520 am 1540 am 1560 am 1580 am 1600 am	550 mW 510 mW 450 mW 450 mW 490 mW					
(/* (/* // // // // // // // // // // // // //	10 am 10 am 10 am 10 am 10 am 10 am 10 am 10 am 10 am 10 am 1	150 mW 340 mW 420 mW 500 mW 500 mW 520 mW 550 mW 550 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW 580 mW		1520 am 1540 am 1660 am 1580 am	550 mW 510 mW 450 mW 450 mW 470 mW	Image: Sector				
(/*) (/*) (1) (1) (1) (1) (1) (1) (1) (1	10         am           40         am           50         am	150 mW 340 mW 420 mW 500 mW 500 mW 500 mW 550 mW		1520 am 1540 am 1560 am 1580 am	550 mW 510 mW 450 mW 450 mW 470 mW					

									$(\mathbf{G})$
Miller Stage Ca	libration								
(0743									
SHI		SH S							
800 00	us ww	590 nm	24 mw						
810 000	50 mW	590 nm	s4 mw						
820 mm	(400 04)	590 mm	27 mw						
\$30 m	62 mw	600 mm	Kam 88						
840 000	61 mw	610 mm	91000						
840 000	لاياس 0ي	62000	84						
560 000	(0 mW	630 000	105 mw - 4	minute - needed	to proch as	(an of the o			
170 AM	ar mu	(elkD 0 en	الماسر كاترا						
Sto an	55	life an	(02 mbd						
\$4p	63 mW	(+D + m	(00 - 100						
din sm	fa mil	den um	90						
fin an	45	(470 am	06						
924	- 5 M(W	680 14	20						
44.0	46 MW	690 nm	Jo mu						
100 000	45 mb	700 nm	68 mw						
710 4.44	44 ~~~	410 nm	( ~						
950 nm	41	120	00 ~ 60						
700 14	4[ mW	130 000	62 mw						
7 10 mm	JE mio	1000	GO MID						
980 nm	56 mW	150 am	54 mW						
990 m	35 ~~~	fle na	49						
[000 hm	34 mw	270 nm	"U mw						
1010	32 mW	780 ~~	41 ~ W						
(0 20 4 m	32 mW	110 mg	28 MW						
1090 nm	32 mw	400 mm	22 mW						
1040 nm	32 mw						-		
1050 nm	31 0000								
1060 m	30 m.W								
1070	30 mw								
1080 nm	29 mw								
1040 nm	30 mw								
(100 n~	Waw								
1110 000	30 mW								
1120 m	30 ~W								
. 11 30 ora	32 0040								
198 m	32 mm								

The second M	worning the spiriture ourput	power was jower	(S.S W). Dure me	asured the Tourami,	and is hed not pow Soo not	6
. ب	abjuited the width and w	arelinete to match	matter - pourt	~490 mas		
		a de comes		A management from	2008 m 20 2 4 8 m 4 6 m 4 10	
4.	uning adjusticities and the	THE WA	South South Content	In the COTTONY FORM	provide porter power in the contract	
Dry 0 , 104,	to. and one doe the	we ; wi pointer c	harger got 500 m	,w		
ц. 	did not get emprovement	ul next round	trip in the Spothic	e ( int round traps in	inedial of 18)	
	UP when swatching	600 13-214 row	nd trips, needed t	a more compression	deley ternerd	
10743 Mixer	centraved					
SFI		SES		DEG		
535 nm	(so mw	470 nm	130 mW	2500 000	2 mu 1.7 mw	
540 nm	150 mbs	475 mm	235 mW	2750 nm	(.8 milli	
545	150 000	480	LSO MW	3000 000	(.4 mw ?	
140	152 614	485 000	135 mW	3150 nm	4.3 mW	
970 nm	150	440	2145W	100	4.5 mw	
640		445	446	700 000		
	ISS mw	â.	L'IS MO	2120 000	4.6 mW	
542 11	160 MW	380 AM	290 MW	4000 nm	3.1 mW	
\$10 nm	170 mw	505 nm	220 mW	4250 nm	2.6 m.W	
675 nm.	170 mW	510 am	210 MW	4500	2.4 MW	
580 mm	180.00	515 am	210 MW	6000 am	1.7 mw	
585 an	175 mW	520 non	210 mw	5500 nm	1.5 mW	
590 nm	(60 mW	525 nm	210 146	4000 000	(. L mw	
595 nm	170 000	530 am	190 mW	6500 am	0.7 mw	
600 nm	170 mw			9000 nm	0.7 mW	
605 nm	ino mo			9500 am	0.7 mw	
610 m	110 mw			8000 0-0-	0.6 mW	
				8500 Am	D.6 mw	
				4900 6.00	0.6 mW	
				6500 p.m.	1) le mai	
					0.5 mla)	
				10000 AM		
				liber om	0.4 MW	

							( <del>7</del> )
10242							
10,100							
(P 1) and	690						
( 1300 1.00	200 000						
Sombrut m [	1.9-6-						
15-21-12 AL 144	C.1.						
10 142 141 144	Campionon						
						144	
201				2H2		<u>&gt;F1</u>	
6.00	2.	uts		10-		144	1180
400 000	10 mbi	1140 0.00	34 mW	570 nm	50 mW	533 MM	Hamm Do wa
810 nm	74~D	1120 00	59 mW	590 M.M.	11 m W	540 100	120 MW
870 am	80 mW	1150 AM	4.5 MW	590 nm	س ج <u>ا</u>	545 NM	62) mU2
830 AM	78 mW	1140 000	45 ~~	600 AM	120 mW	\$30 A.M	155
Edo vu	78 mW	1159 0.00	SUMW	610 nm	125 mu	275 AM	لما ٨٨ ٥٥١
850 Am	76 ~W	160 000	57 mw	620 nm	130 mW	560 nm	Curr Cy]
\$60 mm	74 m W			630 nm	130 AW	565 nm	سم oli
870 nm	24 mw			640 nm	130 mW	570	170 MW
an 088	71 mw			650 nm	125 mW	575 nm	175 m12
890 nm	69 mW			660 nm	(20 mW	580 AM	ن میں 120
900 m	GUMW			630 nm	[LD ANK)	585 nm	(90 mW
910 m	(ji) mili			680 nm	90 mW	590 nm	200 mW
420 nm	59 mb			690 mm	لواد ساي	595 nm	190 MW
430 nm	ju nw			700 am	Ble mu	600 Am	Lgo mW
940 nm	SIMW			710 nm	L8 MW	605 AM	170 mw
950 nm	49 mw			720 nm	(p orli	GIO AM	135 m W
960 am	47mW			230 nm	26 mW		
970	44 mW			\$40 nm	52 mw		
980 m	4200			250 m	6L mb		
990 ~~	41 mW			760	62 mW		
[000 nm	40			370 pm	\$7 mW		
1010 200	38 mW			200	54 ~~		
1020	38 mW			790 ~~	44 MW		
1030 nm	3km W			800 am	42 pmW		
[040 nm	32 mw						
1050 nm	34 ~~						
1060	34 ~~~~			``			
1030 ~~	34 mW						
1080	34 ~~						
109000	35 ~~						
1100 000	35						

										3
10742	Mixer (	hbrate	n							
SFS				SHSHI		SHSHS		DFG		
								_		
440	0.00	2.50	s nu	395 nm	ACI	285 nm	LO mW	2400 nm	3.7 mW	
425	0.00	290	mw	398 am	NA	287 m	13 mW	2750 000	7.6 mW	
480	0.00	26	S MW	403 000	20 mW	292 000	18	3000 m	9.9 mw	
485	0.00	290	) mbi	40% 000	ZUMW	297 000	20 MW	3250 nm	10.1 mW	
490	am	2.64	b muu	413 000	2400	302 000	21 mil	3500 nm	2.2 mW	
2195		22	0	418 000	24 mw	307 000	21 1000	3750 m	S. G. mlal	
400		75	5	423 000	22.00	312 00	22 mbl	4000 m	7.7 mW	10.5100
505	400	140	i mu	428	21 ~~~	317	23 mls	4250 000	2.2 m (a)	110.0.0
510	A.F.	14.7	(unally)	425.00	1800	322	23	L/500 m	64	131.400
SIC	6.00	190		U42	الم ال	322 AM	a V mil	5000 mm	6.3	141-133
40	1.1	670		443	He who	432	26	then	1 4	123.666
106	P.4	260	2	444		432.	(1 mW	5700 00	9.3 MW	123.993*
200	(19)	12		4/2	10,00	277 N/A	24	(COU PR	2.7 MW	129 409
2.90	10	20	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	122 //	1	342 000	25 MW	6500 m	3.2 mW	120 400
				ma 6CP	(( mD	347 nm	29 000	1000 04	3.0 MW	130.404
				463 000	[0 mW	352 nm	23 MWS	7500 m	2.9 MW	
				468 nm	9	357 nm	23 mw	gap an	2.7 MW	
				473 am	fmb	362 nm	20 mW	6500 00	2.5 MW	
				478 m	7 mw	3107 am	19 mw	geos am	2.3 mW	
				183 m	8 mW	392 nm	17 mW	9500 mm	2.0 MW	
						377 nm	15 mW	10003 mm	1.7 mW	
						382 nm	13 mW	11000 nm	1.4 MW	
NOTE:	Tours	noiccard	s dowere 1	the \$00 nm		387 nm	11 100			
Ti:Sa	ph out	ent hec	man of a s	obige Chorescent		392 nm	8 mW			
olye	ncorps	roted,	nto to C	anal spock. The		397 00	7 mw			
dyc	also Flu	oroscan b	olve when	exposed to		402 nm	NA,			
UN,	You co	n we th	in to confi	rom the exittence		HOFIN	NA			
ول ر	LV whe	n have	( cligary m	k SHSHS						
6.00	ц.									
Mixero	libreh	on Dev	c took a su	ccord lose of Tsuno	mi. This fire, 1	he opened it up to	chrick the internation			
by high W	ned he	meunin	hold-1	and normed to the	crowber. Shill	observed cersition	٠.			
ملعمل	al to	Millena	- topping	on the fible near-	the Milleric .	end on the Mil	lence care - no	effect. Dave does	not support t	+ Millenic 11 the
	טמיין	m.,								
6 10	ofsc	other off	M2 - 1	Dove scienced the optic	, and the scat	he min gone.				
e managed	4	where the	Spulf To	mani e little puel	ic sensitive bu	is soll not perfer	t. He advised	that we keep the	mode locue on	br now.
			E.IN U. S. X.							

tes Dave	laft, collected power curve wil	our 407A				G
1+Fire stat	nus + 3.9 W					
	13 round trips (99.50	, 124.50 and 25	3.50)			
uched up co	mpression to 10743 @ 1300 n	m to maximized power				
0743				10742		
es.k	0.0					
ore.	211	SCore	uniher 1	S+I	Slipee-unifier)	
40	100 mW	45 m.W		100 m 13	60 MW	
0 nm	220 mw	130 mW		280 MW	200 mW	
50 nm	315 mW	180 mw		400 mW	290 mW	
90 am	380 mW	2.30 m	•	440 MM	300 mW	
20 em	400 mW	245 mV3		SOUMO	340mW	
to nm	390 m W ( a little unstrable )	225 MW		SHO WW	360 m W	
60 A.M.	440 mW (a little unstable)	270 mW		500 mW	360 mW	
0 nm	480 MD	285 mu	2	780 mW	330 mW	
09 nm	500 mW	290 mV	5	600 mw	380 mW	
20 om	490 mW	275 mW		600 mW	370 mW	
40 nm	430 ~~~	Sled mW		600 mW	360 ~~~	
60 nm	460 mile .	240 mW		5 80 mW	340 mW	
50 nm	420 mW	210 mW		600 mW	350 mW	
Pan	440 mw	220 mW		600 mW	340 mW	
20	370 mW	185 mW	)	GIOMW	340 mw	
0 000	420 ~~	210 mW		600 mW	340 mW	
10 nm	430 mw	215 mW		600 mW	330 MW	
80 am	440 MW	220 mW		680 mW	300 mW	
18 am	420 mw	190 mW		560 mw	300 mw	
lo nm	380 ~W	180 mW		Stomw	280 mW	
10 nm	400 ~10	180 mW		SIOnw	260 - 20	
10 nm	350 mp	170mW		Soomu	240	
se am	260 mW	180 00 60		410 mW	190 mW	
Inn	300 mW	90 m W		440 m W	150 m W	
0 am	260 mW	45 m13				

charter g.								
F.S fuse -> front	of power supply; qu	who there -	220 N 10	A? - 100	it in menus	Λ		
							MOC 3044	1
if your excluses	2 ?					1560	150 Com	
,	24?						T 57.2W	
			2.				h ) h l · ·	10
and the case into	15 - + MOHUS (	on red leads.	{ turn al	il porer sul	hy and unp	wy		6
	( _12 001 _		)					_
			( M.	legie · man	al is the and	y custome	menvel	
for in nower some	42		100	+ may ence		white & Am	chip / follow	, +
1	a deader and and	+				1		~ * *
ie yas fuil Fare	1 June 310 Car Star	COOLINE		CCO DO - M		00 00 13		
				FELKOW	Erel	/8 *	•• )	
				Start compres	yhar .			
plack heads :	thermal cut out	3	1	Interlock -	95			
			Ly y	turn heater	on ? check	e heater ben	p selvice conel	4
	red wires as	to each he	ere )	43 min	}.			
	La Collona	to save has	te l	b)(0 (20000	20 00 70	Week cooles	4000	
				for the second			7.0.0	
			· \	salety telt	= Second	c/ick_		
					7			
Joe - ask for	fire rod - 740	power supply	for M	Villenia V		4 screws	remore unde	
6					1	Front		
							a Gurand (C)	3)
	En sub subsc	2			2	10 8	E JADOGS De	me
SER SCRE O WORKS THE	int in place	2424	C + Unipres e	COUNTONN		Can .		
distro	chon between the c	JULIES J					alich	-
						i		101
						7	01.0)	1
a sh across - any	s themal ent out	) gwa	p m here:	?		remove	front games b.	0 96 0
			It the from	+ panel boar	el ?	( is cher	LUL CONNECTOR	· +>
		ξ.				here	sters	
		)					- Shewled real	57
dowl check the w	·····						a LTON M	L Co
10								
~7 min		le+ 72	geob combre	kely -> bo	the deader	should cool 4	~ ~ ~ ~ /	D
1	(2)					45 mir	?	
Z elicka P	La gavare	E big for -	white -> mi	delle of the p	one soppy			
Storing	1 looken vel	~ ~ ~ ~	onvessor V	es? - should	feel hot			
	real or bl	2:0 0000	La harris					
		on tol	700	p seize up	12 C	voter 11 TW	10 The Care	
word test -> W	ating test			1)	- Yes Con	heeron		
	Is just tries to have	and gets 18	1 it hears	Çar	- materia P 50			

Dave	VISIT - May	2 - 14 2014	
Empower	r filter: should be	vacuumed and a guartering	
	Gilteris	the back of the power supply - see photo.	
6000	out in Milleni	THO every once in a while	
		Checked HR+ SHG leng - both methy close	
Millen	a touch up	act updated acted a per (did act look & distance to the up the	1.1.1.1
i tottert.	in touch up t	and aparter from and the constant of a same output - Juit	(DOMED CY-
		<b>,</b>	
	Switch T	in 4 3 doesn't metter whether or not love is on	
	Men SCIEL	in con set	
	in SPC-	en's node:	
	- cor	set operating current on mon screen - direct control of DI and D?	riat
	- 910	laus both a course and the casisting on the SHGT temp	
		13 accessed in the same festion on in standard anoder	
		just promoter course adjustment an well on fine.	
		rescurements are node as a number of bits of resolution - no direct	physical
		mong is known the range of the temperature control	
		is more positive increase LBD temp	
		hor washed Appended I To be	
	- MA	n: can entry temps for disalus, character that give 13 W out of	Ol woll s
	ىسان -	new velues one entered in the new new in SP Correct node, swith	ren to
		normal power and let it sit ? the new powe curves will be written	
		is switch part to leave sported technician access after power	-me witten
		No the second se	
Uneri:	rod translate	: () There are six screwe that hold the rod mont in place	0.
	· · ·	to the cente one looke the sant, and to an ous on	it an
		rod Togg on allan translation	
		1,407	
	rol cleaning:	mot detany filvene etc any are fine	
	5		
	1		
	tuning slit we	you too have only area -3 mon or so ( maybe a (ittle more )	
	Dave contract or	opics wi the turing shi in place =) really doesn't matter whather the trong shit is	there or not
		· · · · · · · · · · · · · · · · · · ·	



- P2 is actually a focusing nurror there is a small screw todallow to act adjustment of the focus of the beam; should not need adjustment value to have out of the Milleria changes
- Dave an hers tracke getting sufficient bendwicht (to his trace) he transloted M3 I sag jage states along the aross that M3 her a small say screw that allow transloted
  - M3 his a line on it and on its noutry herdure to gauge tradition is m2 and M3 offer read to he naved in tanelow - if one is nound formand /bock,
    - the other likely needer to he norsel forwal / book
  - A NEVER TO THIS !!!
- Alw the adjustrates, Dave is able to get up to ~70 m band water (Further) reducing to ~60 mm, Dave could push to pomente pump pour to ~4.3 W -/s CW breakthrough
  - Howen, ofthe closery and reopeny the shotse, Dove rank not recover node-locky sometry on M7?
    - w 4.0 w grang : 480 and out, of 60 an bond udit
      - could top pretty aggressivery up love of mode locking

			(3
atches rive	Spitfire unstall:		
To gri the E	imposed output en	tuy correctly. Dove pr	t the alignment tool right inside the shift
Bendundt	detector: old	Phatodiodes: Pick of any	color 1.2 +
		Picosecond Spitfine, seem	te new photodioden are only certite to 300 a
Dove dow	at we free cou		
he	ofter burns the	gh the Thin Film Polen	shore in the systems, we again that that
			Grety - MEOH ?
charcod par	ches. le no cycle	?	drizzb
To align	intracanty Pockel	s cell;	
pice	differe alto.		
	1 polarze all	v	between extra carry pocula cell and TF
	loon holore he	re plate	) L'allows all seed pulses in to the casety
optimi	re focuels cell h	) optimited cross pettern	
		> frem looks were	
New TFP	- dow not suppo	ere low shill re	ided for as losy - needed to trap a fluorescence pro.
CLI: a	llare setal in	; CL. Z: roteter 90°,	treppy; Ch. 3: Noteren 90°, Zee
v gets too	poed in cervity	turns off after	, -> off my the
questo un	plete - light g	it istokal (80° w/ 2	parsies ] time the pulse cours back from the end mus
			The states for
TDG: De	elay 700: relan	re deleg hetween Kompone	and Truneni triggers optimile by water pulse from
anter:			25 Po#160?

							(4)
on: 3,8	2,1 2	or menual tu	n on				
off: 1,2	,3						
to over co	ne bud: c	hisoble check	of Ch. 1,2,	3+ of	TDG?		
		is have	to be in 3	ervice node			
		Timing	-> Use Bw	D - Check	50xec 1,2,3	· · · · · · · · · · · · · · · · · · ·	
Ase - sinu	the TFP doe	n not hold '	back lasy .	when you Poe	uels celle en	e off, there	is likely
50 A.	ASE compo	net to the o	uppet - get 1	ou , tu	compressor (	compressor	eppeer
6.0	decient	because of th	~)				
	C.A. Marker						
anpi	ILLES SPORTENE						
1	ts: need to	switch out Cl	.3 earlier	in order to	optimize si	ud pointy	
w New Unit					•	-	
~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	connot d	isony Ch.3	+ optmize	plus tren -	risk of de	mage	
~/ <u>Vi</u> D UN	connot o	115006 Ch.3	+ optmize	plu tren-	risk of de	meg	
	cannot d	15004 CL.3	+ optmize	plu tren -	risk of de	~ • <del>y</del>	
Chiller flord	cannot d	there pull b	+ optmize	pulse trein -	il ood temp	ik too high	comerstanty,
Ohiller fl-id	cannot d	the price of the p	t optmize	puter tream -	if not temp . now need on	is too high	concistenting , hiller for the
Ohiller fl-id	connet d	B halpi pull b Gool hyp	to optimize	plu tren -	if not temp namy need on Spitfices	ic too high	concistentes , hilles for the
Chiller fl-id	connet d e: chits to H.sh Voltage	the opt Ch.3 the halps pull b Gool high 1- TDG b	t optmize	puter piche	if nod temp many need on Spitfines	is too high	concistenting , huller for the
Ohiller fl-id turn off	connet d a: contres to H.sh Voltege	14004 Ch.3 the halps pull b Gool high 1- TDG b	to optimize wet every from speed drives abre checking	price treen -	if not temp now need on Spitfnes board	ic too high	consistentes >
chiller fl-id	connet o a: catter to High Voltage	the option of the second secon	t often.re net cung from speed driven	puter piche	if nod temp namy need an Spitfines	is too high additional c	comaistentas , halles for the
Ohiller fl-id turn off HV : il a	connet d a: chits to H.sh Voltage o voltage, t	to HV jergel in	to optimize wet any from speed drives effice checking to TDGr 19	price treen -	if nod temp now need on Spitfies board	il too high addithand c	concistenty > hille for the
chiller fl-id turn off HV : il a	connet o a: chits to High Voltage www.voltage, th Portuits ce	the syster Ch. 3 the halps poils b Gool high 1- TDG b to HV georget in 11 - 200	to optimize next cum from speed driven afre checking for TDGr is	puter piche	if nod temp namy need an Spitfines board	il too high a additional c	constanty, h.11 e for the each
Ohiller fl-id turn off HV: il a	connet o a: contres to High Voltage, t Portuis ce o current   Po	146004 Ch.3 He halps poll b Gool hyb 1- TDG b HV jengel m 11	to optimize weet every from speed drives abre checking to TDGr is pt from , these	price tren - the road } price picket proble.	if not temp many need on Spitfield board board in . there are	z - one for the HV ju	concistenty > hill e for the each
Chiller fl-id turn off HV : if a if a	connet o a: chits to High Voltage b voltage, t Portulis ca to current   Po the High SZ	the order Ch.3 The halps pull b Gool hyph I- TDG b to HV jumper in 11	to optimize next cum from speed driven the TDGr is the TDGr is at fum , these	puter tren -	if nod temp namy need an Spitfores board d. there are m w/ enter	il too high additional c Z - are for the HV ju	constanty, hille for the each
Ohiller fl-ia turn off HV : if a jf a	connet o a: connet o a: connet H.sh Voltage, t Portula ce o connent   Po the Hist SQ2 care the Unit	the property of the point of th	to optimize wet any from speed drives the TDGr is the TDGr is of from , these A and surrent	price treen - the road } price picker probably dec n e proble. the mate T	if not temp now need on Spitfnes board d. there are n w/ enter Dar / toma	Z - one for the HV ju	concistenty ; hill co for the cach
chiller fl-id turn off HV: if a if a con s	connot o d: chits re High Voltage, t Portues ce o current Po the High Sch sue the theo node	Le HV jungel in HV jungel in HV jungel in Le HV jungel	to optimize meet cum from speed drives the TDGr is the TDGr is of from, these ph and exercent	puter prode	if nod temp namy need on Spitfires board d. there are m w/ enter Dar / times	z - ore for the HV ju	constanty, hille for the each
Ohiller fl-ia turn off HV : if a jf a	connot o a: contros to H. Sh Voltage, t Portuls ce o corrent   Po the High Sce see the tho No de	the property of the point of th	to optimize wet cum from speed drives the TDGr is the TDGr is pt from , these p and current	price treen - the road } price picker problem dec n e proble. m the Th	if not temp now need on Spitfnes board board n ~ ( eiter DGr / time a	Z - one for the HV ju	constanty > h. II er for tu each
Chiller fl-ie turn off HV : if a if m	connet of d: chitty To High Voltage, t Portulis ce o current   Po the High Sp sue the Ymo No de	the property of the point of the property of the point of	to optimize wet any from speed drives to TDGr is at fry, these p and converse	puter prode	if nod temp namy need on Spitfires board d. there are m ~1 enter DGr / thmy a	il tou high additional c Z - one for the HV ju	consistenty, hille for the each
chiller fl-id turn off HV: if a if m can fegens: can	connet of a: contros to High Voltage, t Portulas ce o current   Po the High se sue the tho No de	HV (1 + 1) $HV (1 + 1)$ $HV$	to optimize weet any from speed drives to TDGr is to TDGr is pt from , these pt and answers HOWEVER:	price treen - the road } price picket probles n c probles m the Th il you row	if nod temp nam need an Spitfines board d. there are n -/ enter Dar / times a -/ under 20	il too high additional c Z - and for the HV ju	constanty, hille for the cach
Chiller fl-ie turn off HV : if a if m con a fegens: con	connet of d: chits to High Voltage, t Portuels ce o current   Po the High Sp see the Was no de 	the property of the second denses and the second denses of the second de	to optimize west cump from speed drives the TDGr is at fing, these at fing, these at cod current HowEVER: Slower - MOR	puter frem - the road } puter picker postering dec n a probler if you rom pesse will be a	if nod temp many need on Spitfines board d. there are m w/ enter DGr / times a c/ wn dw 20 sequired to reek y	il too high addithonol c Z - one for the HV ju	consistenty, hille for the each
chiller fl-id turn off HV: il a if m con fegons: con	connet of a: controls to High Voltage, t Portules ce o current   Po the High sce sue the stro no de no de - but b - then in	isover Ch.3 There for the form $h = \frac{1}{1000}$ for $h = \frac{1}{1000}$ in $\frac{1}{1000}$ for $h = \frac{1}{10000}$ in $\frac{1}{100000}$ for $h = \frac{1}{10000000000000000000000000000000000$	to optimize weet any from speed drives to TDGr is to TDGr is to TDGr is to a charrens to cod anorens HowEVER: slower - non ore nonlinear	price treen - the road } price picked probably_dea n a proble. m the Th il you rom pesses will be a chief that each	if nod temp naw need on Spitfines board d. there are n -   eiter DGr / times a equired to reek y not he compan	ic too high additional c z - one for the HV ju the HV ju	constanty, hill of for the cach
Chiller fl-ie turn off HV : if a if a con s fegons : con	connet of a: chits to High Voltage, t Portuels ce contrant   Po the High Sp sue the Woo No de - mul in - mul in	the property of the second sec	to optimize wet any from speed drives to TDGr is to TDGr is to TDGr is to a corrend to Cod corrend HowEVER: slower - non ore nonlineer = broade p	puter picker puter picker puter picker poter picker n c probler il you rom peter with he c chief there een	if not temp many need on Spitfines board id. there are in w/ eiter DGr / times a c/ when dw 20 required to reek y not he compan	il too high addithond c Z - one for the HV ju the HV ju can while n wi peak power	constanty, hille for the each npe or

																													(=		
	Em	Conte	LE	30 :	hec	red.	40	~3:	20° i	2																					
					held	4 .0	٢	leu	5	c	pare	ي.	લ્	· 5 :	on	ONI	side	and	20	-to(	2					C					
					~~~	ne	وليه	red		Meo	H is	6.	.7		ha	a	hu	Lise	2 14	610	ene	cnd	0	1120	fpe	(r 601	exp	15100			
									y	red	40	6+	4	cæ (	e.	ret-															
					LB	0 +	emp	adj		Ge	r u	64	trin	por	(0	f th	e 3	>													
,	~*~	een	+	Pa	-le el c			101	able			сь.	7																		
,			2	10	Charty	G			490		2		6																		
to	ali	gn	-tw	COM	press	Pr,	Dav	e	نه د ا	eles	~	* C	3 . ما	s e	24	50	the	+ +	he	pou	o i	s r	e d u	ced .							
		6	easie	r 4	н (	alisv	?																								
			655	1.10	5+	o d	cone	ye -1	ng	?																					
1	clure	6	1.00	4	- da	0.00		~	0.00		000	-h					d.			4	tre			mad							
(D	Crac	bee	n e	clk	chan	1	nyt		the	5.	q. M	-+-	1001	-	۲. ۱۲	.4 .	isn t	c	.56	*	¢.	the	4	ure		vtree	110	de	um	4	
											Ĺ										_		,					5			
the		out	ns	£	the	Co	ofre	50 C	Cr	d.	-he	1.70	14	40	44	co	~60	esso	r .	sho .	rd	he	ove	eac	Lo	hu:	if				
	-	they	04	• •	.04 1	ca	~ >	one	-	chi	ze .	flore.	٧-	miri	or	255	emb	y (	tre	10	~ /	ter	09	+1	du	elez	١.	~)			
to	cla	200	the	Cor	naces	hor	4.00	hq:		alaz	24	J	An	ne	DH	(6	~~~	c. 59	USE	+	٢٠٩	16)	, e	nal	510	000 0	ce.	-1 1			
to	put	h	ede,	,	nst	pus	L +	1	pech	0	٢,		+															, .	0		
								_																							
wh	un	-ne	Aew	Con	press	9 <b>7</b> 1	s pi	ered	. 14	•																					
			- ch	rch	ver	rcal	5	hec	h	ref	uch	~	th	J.L	*~	alu	sn ~	4	100	1											
			- ch	ech.	har	1200	itel	4	c	ree	Fy	che	reh		re	ove	ler	2	٢,	in .	and	s-t	5	Con-	pres	-					
											,																				
			to	tov	eh -	y v	rt.	al		42	ver	ed	cdi	h14	~*	- 20	3	ren 1	3	man	ntr.										
			to	tove	L .	- r L	erz	ente	.(;	use	14	. \	- a	1.110	< a	ssen	.bLy	) (	twis	+	h1	the	her	1 Fan	79	is c	~ (re	4)			
2400	ected	270	eren	2		- 81	°'l.																								
,				5																											

1 1	1				1		I		I.				1				1	1	1		I	I		ł	1	- 1			ł	1 <sup>1</sup>	ł	1	
					<u> </u>	+	ĺ	1~	ł		4	+ -	-		+ -			÷ -	+ - 1	+		i –		7	1					- +			
						i	<u> </u>				+	••••••			4	+		 		 			Lle	-2.1	fZe	्राम्				<u> </u> ;			
	<u>_Q</u> ;	est	pas			+	ł	¦	ł			-				1		-	† -		+ -	+ .						-		<u> </u>	ļ		
	-	Dri	<u>tı - "</u>	the	To				1 - 2	~	1 7	متة ؟ ده	<b>6</b>	15_+	<u>hin</u> (	expes	red?	ł	ł		-	TOPP	5 4	<b>5</b> 0	°t1	40-			łł	M	- v_4	·	5-1
		ADA	L <u>a</u>	ean :-	ج	1-00	ь <b>6</b> 0,	hte_	S-lue	142 Z	<b>7</b> *+ -	-		-	-	i +	-	1	ļ	ļ .	1-	run		r.16	" ~	1. 1	201				4	-	-
						ļ	ļ				·	<b>.</b>		•		<u> </u>				ļ	ļ							ļ					
				L			L	1.		<b>.</b>	+	+				i				ļ .											ļi		
		4.4	1 100	ec .		(ms:	1	dent.	6 5	un-t c	, <b>L</b> 2-	+ C	   :	i e	del e	┢.	L.	4.	Arain		na.	-sr.	. 1	220			~						
						Γ		Ţ				1	1	,	,				1	<b>N</b>			-7		-								
								1			+	·			1		İ.		1								<b>H</b> .	r					
		Spit	hre o	and	<u>- 19</u>	Sam	* :	+ +0	efar-	tr f	×~f	, <u>, ,                                </u>	1.5	. h.e	*	10	bela	1	100	<u>, , ,</u>	rcu	\$ 10	<b>`</b>	مع	pta:	2<	,0	ber		f.	se,		
								<u> </u>	110,	44	eft.	30	pu -	9a(	1 10	Trees	-	¦	<b> </b>											<b>}</b> ∤			
								on	mise	, h	1 th	<b>p 4</b>	re e	ærl	bd .	and .	100	my	<u> </u>		•	t e	100	o(	rate	* *		1.1	./.	0×	اصدا		
									ļ			•				; 		ļ	+ •	۱ ۱	ļ										i		
						ļ		<u> </u>	ļ	+		·	ļ					1 	ŧ	1 <del> </del>	<b> </b>	} 									L		
								1	ļ	;					1	•			 	ļ		; 			20	4 1	0 2	1			,		
	Qu	esto	ns								4	1			1	•				1											, , 		
		-	[	100	. c		6.0	d mi 1	-	м	ماله	ilia.	J	M			160	r	***	,25	$\prod$	M	Men		1.00	21 54		(8.1)					
		-	Ronal					1			<u></u>	d.h.	00	,		+, t,		1	C	μ	}	<b>[</b>		-	4.							- 1	
			c			1000	ļ		13/			1			10-12-0	1.00	n-g		1	 							<u>, , , , , , , , , , , , , , , , , , , </u>	900	000				
			foc # 3	10	<u> </u>	PLAS .		t .		+	+ 1	Ļ —	ł		-			•	ł	-		Ma	(rei)	- <u>-</u>		لمومدة	¥.1	1 201	زاهم	24	- 44	ł	-
		-	ROM		5010	<u>}-</u>	etto	-pnie	K S	joi ve	hts	<u>{ · ·</u>	Meon	.,_cce	tone	<del>: Jue</del>	4	د له ه	-+ 1	r tu	<u></u>			_50	trace	- cani		<u>a</u>					
		l				1			<u> </u>		1	·		·····	•	+		•													 		
										+	+			••••••		+		•	+	 											i		
				·				<u> </u>		· • • • • • • • • • • • • • • • • • • •	I	4	i	+	+				+											¦			
		Mil	1000	<u>ب</u>	set	, <del>to</del> :	4.96	္မ်ယ္	1	$\mathcal{D}$	۲	21.7	A	22.	4° <u>c</u>	-	.	ı _	-		Ļ.	Ι.				-				   	   - ,		-
					L	ļ.		1.	Ĺ	Ρ	Z	27.	A	29.	ų م ۲	+	L	! _	1	1		L .		_	-					+-	_	_	
						;		i .								4		+   	( ]														
						1		,				1	1	,		1		1	1														
						-	İ.	<u>.</u>	t	•	,	-	İ		1	-		1	† - 	† -	1	† -	-			-	-		ţ.		.	1	_
		12 83	<u>ب</u> تد ا	l.		<u>⊦_</u> >> ¦	1	<u> </u>	ļ		+		-	+ 1	1	1				t —									-				
		- 1		40-0	hear	-1	Car	<u></u> †~••	1 -	нк <u>.</u>	.+_ ⊂ 4	، ب			Ľ	+	1.	 1		÷ .	t				-			1	+		- 1		• ·
				mea	+=	1000		ice_	100-	a.		¥**-	Part 1	~~	60	10.00	5	<u> </u>	<u> </u>	sse	<b>*</b>	500	un	7					- 5 0	إحدا م	rel		
				cree	hed	No!	۴.	120 <b>1</b> 2	1 3.	*wş¥	<b>e</b>	1423	<b>†</b> . (	<u>44</u>	Pr	Ĺ		ł	ł		<del> </del>		+	-			-			<u> </u>	۲۰	-	-
								 		 		4	_	• +	ļ	ļ		 												<u> </u>			
				with	cu	they are	Ao	i m	(-	, ~~ ., 	مددود	4 4	. ب	4 5	\$-57	+ ~~		ļ	 														
				ale	cae	4	Bres	pre		<b>\</b>	<u>} </u>	يف	> <		<u>၂</u> ၂၂	ļ			1 +	ļ	ļ												
						L				-	-	i .		۱ +		1		ļ	1 -	ı	L						_				l		_
				alia.	Led.	hes a		Da	RI	4 te	fla		0	ha	3131	m					th	C1			. 10		(	× .	-	-	le fui	~	
				- )			i	•	ĺ			1		1					1	'		н.	٨٥	M							, i 		
						101	 h. s	?,	1	•	1	1	1		1	[	[	1					_,	•			_				Ē	-	
				2.000			¥~ œ_		1	•		1	1	· · · · · ·	1	<u> </u>			+ +	1	1	<u> </u>											
				éleor	ed_	чэ_ '		•	1-	r		1	1						۱ <u> </u>		İ	 								$\left  - \right $			
		 		tean	lerco	the-	å.,	, cha	ped.	have	• <b>\</b> 	ML	₽₽	19 1	et c	å.,~	<u>م ۲</u>	3	43.0	m	5 ~ 2	e s	<u>ns.h</u>	vr		ر <del>بر</del> ا	hein	4 1	one	er	terot	verd	<b>ć</b>
		¦		! 		,	 				1	1						, ,	U	\$ the	re is	0.	n.	١	¢ . e-	o <sup>s</sup> h	<b>~</b> •	<u>(</u> /	<u>ь ч</u>	1-0	the l		
		I	1	ad y	Lorm.	<u>b</u>	1 7	2 «	h 1	t do	<u>~ -</u>		<u>t - 1</u>	+m	k	₽ f	~				h	th new	5	fre	۱۰۰	e.	~ ~	m		ļi			
	T	OPS	=0 M	3504	4:(	<del>,</del> 4 4	فيه و	ito-	┥	 +	<u> </u>	۱ <del>۱</del>	ļ			• •		 	+	ļ									╞	ļ'	⊢•		
		1	i	1	l				1		ŧ	i	1			I.	1	1				ŀ								1 .			

					1	1			1		I	I		· · ·			1	1			I					÷.,		. 1	ŀ		1	. 1	
			-		-	+			<u> </u>									t	+	4	t		+	·					,	·+	- 1	ł	
															, ,			+	+			i	1										
	- 55	<u>^_</u>	bre .	وىعم	fr f	a_fs	કપુક	ten.		ļ	, 	14					+ -	-		+			1							Í		-	
	isine	<u> </u>	pre c	ihean	ed t	Le g	4-6	çue.	ļ		18 .	110	<u> ?</u>	<u>م ).</u>	+	•	5~	<b>a</b> N 1	مرًه	( <u></u> }=≥	$\dot{r}$	4.											
			. he	ac	veced	- 00	d	respt	ane		La-	<u>'</u> *	<b> </b>	! 				+ -	<b>-</b> -		<b> </b>							;	-	-			
					- ,	<u> </u>	R	i +	و، اھ	<u></u>	te_	£1.	feese	عما	<u>ا</u>	-t_	C 4	h_	çan	-	(10	4-		k id	e }								
			-			k-7-	M	Le	hie	-te	ممحد	has	-	<u>-</u>	per	. ما		ļ			ļ	ļ	ļ					L		-			
						1		İ			•								L.												_		
			R2	Ara	5			h.~.	L		110.	Lea	-	ane	4	مد		- 54	0,	Lω				:		}							
			•					<b>9</b>	F -	Γ				Γ		-					<b></b>	<b>—</b>	Ţ										
			1			1			1				1						İ			!	· · · · · ·	1				1					
	T					í		†	t.	-	† -	1	<u> </u>		h a .				1	1		+ -	÷			•		[]		_			١
		we	redu	real	100	~ <del>6</del> -	4	2.9	۲ <sup>ω</sup> ,	Tu	<u>[-</u> '	<u>qet</u>	nea	+3	3.91	<u>~</u>				+			-	1 -									7
								ł	ł			┼──	<u> </u>			, 					ŧ		+ - 1			·							
	Da	e h	ال	dec	ided	4-	tre	br_	M3	1000	-s	ļ	<u> </u>	ļ	+					-+		4	1	<u> </u>									
											~ ~ ~ ~			· • · · · · · · · · · · · · · · · · · ·	+			+	•		+ -	+	·	-				Ļ					
	μ	٥١.	sas	Ļ+	12	, <u>o-</u> r	sce	uc for	h	+= -	٩	المريح و	sφ	estr	<u>+00-</u>	here.	1.00	'pd	170	7 -	<u>h</u>	f ce	$\frac{1}{1}$	a)		┤ -┨ ·	_P-	~f	e	4	ω		
	L						I	+	<u> </u>		 			· ·	; †	-		<u> </u>	5	5-5	7-	Ϋ́	(E	-HA	<u>b</u>					 			
		L	,   	•		; <i>*</i>	•		ļ	ļ	+	, ,	ļ		 +		ļ			45	6-50	5 ~~	2	ļ									
			_					•			1900	kr:	~	500	- :	ہ د	ا_							İ		<u> </u>		1			ļ		•
			1 1	;		1				i 1	ľ											İ		1						L	i		
	[			1						1	Ī b	امما	Lat		Lu	he	lor.	. de	d.	- du		5	1						!	!			
						·	!				T T			Γ	Ĩ		1		Ţ	1		,	1	-	ļ	-							
i						1		1	1		1	1	1-	1 -				1		1			•	1	1	•	h	<u> </u>	İ				l
		-	İ	i –	<u> </u>							1	+	† -	t ,	+ 1		- <del> </del> ,	-i 1		1	-1		+	† -	•			+ I				
	Spal	re_		+	ł		+ -	<u>+</u>	-	1					 -  -	† —	+				-	-,	-1						-	1			
	- Ch	1.	nd 1	<u>€⊾. 7</u> 	4 ±0	5	حطن		fm	+ 1	ynt_	1	₽-	<u> </u>	9-54	<u>م_</u>	120	( <del>)</del> (	<b>0</b> m 1	<b>~</b> . 0	4•	فعلاس	10-1	h~	<u> </u>	ease	· bC		<u>+</u>				
						┿──					, <u>.</u> ., (		+-(	-154	<u>, tre</u>	<u>;- c</u>	ja_	<u>aa</u>	2 <b></b>	<u>.</u>	┼──	<u> </u>							╂	<u> </u>	ŀ+i		
	حما	\$2.44	]و حو	TF	7	y ve	3 -	ł ł		;	+			-+	l		+		1	+									+ -	ŧ	-		
	-	-	.			-l	 	•		+	+		+		r	1			1	+		1		+-		+	ļ .	+ .	ł —			•	
	Rus	e tr	ļ_ι	fer	10.1	4 1	4e	+		<u>                                     </u>	ifure	<u></u>	the		4- 7	<u>n</u>	f		dip	- 5	2 pre	fre	++	are	the	(a	e	He	d.	fee	<u>(.ce</u>	ю	
	ļ		 +	, 	<u> </u>		+	<u> </u>	41-	ļ	ļ	+ the	1 1	<u>k.</u>	4 5	4 mg	hu	00	<u>.</u>	4	.fm	the	lue	her	Pr.	1.50	 	ter	d c	bre	لحعلو	<b>_</b>	
		I	, 		l	1	+ <b>-</b>	<u>'</u> -	Ë		; 	1			ļ		4.		ļ		<b>_</b>						• •		Ļ	ļ .			ļ
							 	<u> </u>	1	14	nes	J	+ 04	-u	c14-		1_^	· rs	red	4	40	<b>↓</b> +.	<u>ل</u>	<u>4.</u>		<u>u</u> 7	+ +	1-		ļ			ļ
			1																	ľ			ſ	+		4				 			
	tike	aun		- se		20	أسواد	Cell												1		1	İ.	•									
			1					Ι			Ī	Τ						1	,			Ì									! ,		
				100		1	1.		0	1	1 Fair			1	010		J.			2	1.	de U	1	1 1 -	FFP	ר ו		Ln	]	. >			
	$\vdash$	<b>†</b>	1	+	† ¢	que	i na	Toce_	- voc	گ <sup>ر م</sup> مر ر	<u></u>	0	- and the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second	<u>مى</u> سر ر	-++``* १८००	†~-	Ī		· -	. h .		ļ	1	1			11	<u> </u>	] n				
						0	†		<b>.</b>	·		+	1		<u>, 79</u>	, <del></del> .	Top.	1	'`	5.4	Tro-S			1	t		14	PLACK	1	÷-(	а <i>ү</i> .	†	
			063	1~2	1	Jacu	<u></u>	بو «	101	<u> </u>	<u> </u>	<u>pe ++</u>	1	. <b>.</b>	1	<b>†</b>	+		+		1			+		1		<u>†</u>	$\square$		+	↓	
	1	1	+		-		·	 1			+-						-	+	+		+		+-	1	+ -				+		•	i	ł
	ł		'	·			+	1				+ -				, T			+			+	†	ł		-			+	-			
	15.	\$ <b>*</b> _*	$\mathcal{P}$	ф.e	puch	de v	¢ v	4 ~.	1.04	(~	<u>4)</u> .		40 ~	ψ.	+	ł				+-		+	+-	-						<u> </u>	 1	,	┝
				• •	1					 +	+	+	+		+	<u> </u>		+				÷.	ļ				+	L	–		;	*** ****	
-		025	+ DRM	350	41	MA 7	י נ אי ד	5 A ]	ļ -	i t	+-	<b>_</b>				 	ļ			1	L	ļ		 ;	ļ	+			+	T -	۰_		<b>-</b>
ĺ			1	1		1					!				i				1		1		i					i -	1	1	I.	1	

Dave 11514 11 21 2014	
Questions:AR cost on Pochels cells? if so how early dam	aged? + Des, has AR county
Campesson bread board ingover startling	J
separated from the new breadboard by a reven	of pools
New Southres: TDG has HU generators	the Behels Cells an anal
I I Ruise pierre board delage the HV's sond to extracanty S	chub cell_
- Brance - Pochet	
Battre instally Pockets all, Dave checked alignment - start the se	entre seed them
into to chang, and ploud croudd polarzers before and 	alte tre
Emport frage choose	· · · ·
- open pleitic cover - turn off Chiller	· · · · · · · · · · · · · · · · · · ·
the prop champer is in an exclosure whore hid is held don in 4 save	m. Renove friese.
two are for sorry the pump champer (he ld by 2 screws)	
La deservative de la deservative de la contra de la deservative de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra	the second second second second second second second second second second second second second second second s

					↓ · · · · · · · · · · · · · · · · · · ·			
-	there one	3 gold c	dored block _t.	gold clups. i t	diade banks	te_anoca		
	el	light though	feech disdy to	por nound	the gold clips	and introdic	- !	
	Cox	toet the	y tu sever					
	when we	renoved th	4 Hop disde -	trure m a	pleck got at _ +	n - ele	Strode on	*u
		Darie sound	Lit noyld be - (	1015.00 +0 . 5	and off the dan	lege + the	clip ; hut	tre
-		as for the	droche <u>blogu</u>	n sand off th	- corrosion, place		stand on the t	l fr.
		La la	he careful of t	nt disaler! d	to not touch .	· · · · · ·		· · · · · · · · · · · · · · · · · · ·
-		for both	cover fine s	and poper cha	-ld be - yed	`-	• •	
	p ni 	tt						
••••••	LBe	3.147	··· ··, ·			· ·		
	rued	~21.9 A	to' get 20 _1	<u>ا کار ا</u>	then clip corroded,	Smalle - d	intred optics	? '
		adjustnet	did help_	( -1	it_15-02/2017	clear dio	Ren - Meon	not acetoro
	· - · · · · · · · · · · · · · · · · · ·						-      	
-	· '		· · · · · · · ·	<u>+</u>	، -, پر			· 
•••••		• •   ••••••••	'	·				
		• •	· · · · · · · · · · · · · · · · · · ·					
-	· · · ·	-	, , , , ,	] 	1	 		
				+	<u></u>			· · ·
-	 + -			-			· · ·	· · · ·
		· · · · ·		• • •		1 - · ·		
		_ <sup>1</sup> 1		+ -			-	· ·

	 с. т. ант.,			 9	• · · · · · · · · · · · · · · · · · · ·	
	Dave visit - 1	6/15				
	when Poc	kets call_gets al	sinegeol , Ishely a	1990/ +10	send	
	i '	need ~ 20	o ps to evoid de	nege	3?	
		····	-	· · · · · · · · ·		· · · · · · · ·
 -	TSuhani ve	tical drift norm		the charge that	pessert3	utsoclard up determining 1
ر ۔	Nº.	good answer on	haden 'weg we need	to reduce pro	gener to reduce On breaktion	gh - A country syraling?
-	M·ileaie : _De	e wevellig tovere	up suc fro	the adjusts	HR pointing	
,						
	factory :	- nearly the se	he as criteri	Peak pomer	<b>A</b>	
-	-		•			
	Deste store	to cuching se	id band with	·		
	-		·	· , ·		
		~Go mi FwH Postistretcher :	M .n_	m FWMM ~/	Boo an confer this is a	marable to Bro @ install
		Mgo Inni Fiurn Postintretetuar :	M	m Flagm wi	300 am center   this is co	magarable == 343 @ install
		MGO INN FUH Postistieteluur : Postistieteluur :	M ~ ~ _ ~ 2.5 ~	a. Fuzyaz w/	300 am center ( this is co	rea, then records an
	high per	MGO MN FWH Postistretchar : er systems , wi _chense is	M .n m2x ~ 2.5 n in _ mmodus . ng in _ mmodus . ng	m Flagging w/ seed: classing	300 am canter   this is co all Pocheck calls, unspecks	eed, then reensole an
	us   haigh pou	MGO MN FWH Postistietcher : 	M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M in M	m Flaging w/ mad: dissource triant as possile peak problem naged when a	800 an carter   this is co en Pochete celle, unspecher le	ect, then records an
	us   high pour TEP : more Bochers car	MGO MN FWH Postistietchar : er systems , wi -chanac is -chanac is	M in 12.5 v 2.5 n her _ withdue ing be is case dameges when is get day when to Postu on the Ts	m FLEYM w/ med: disable time as posite pert problem magde when p its cell is why unani	300 am canter ( this is co en Pochete cells, unspoches le contes obren -1 seco you need to water, out	her cus ( cu breaktingh
	us   high pour TEP: moon Bochers and Lick datag: - w	MGO BAN FWH Postistietchar : 	M in 12.5 v 2.5 n hen _ minodus : ng be the as case damages when Ja: get dar mage to Poch on the Ts ca. to the pull	m Flaging w/ med: disable time as possible part problem magica when go the cell is why what	800 am carter ( this is co en Pocheck calls, unspocks 	her cw ( cw breaktingh
	vs   high pour TEP: moon - Poches call 2.cris dalas:-un	MGD BAN FWH Postistretchar : 	M in 102.5 v2.5 n len - minodus ng le va as case damegra wine je get dar mage to Poche on the Ts ca to the pull optimizes to pro-	m Fleyn w/ seed: clisable time as possib perf proprim naged when g the cells is hely unabli	800 am carter   this is co all Pocheck cells, unistocks 	her cw ( cw bacelettingh
	vs   high pour TEP: more Bochets cent Zerio denos: - un To check seed 1	MGO MN FWH Poststretchar : 	M in 12.5 v 2.5 n her without in a le vie as case demeges when is get der mage to Pocher on the Ts ca to the put optimizes to pro- Korpower lance Rochells cells all	m FLEYM w/ med: disable time as posite pert problem maged when p is call is when or and	300 am center ( this is co en Pochete cells, unspoches 	for cus ( cus bicestuttingly
	vs   high pour TEP: moon Pochess can 2.15 datas: m To chack seed 1	MGO MN FWH Postistietchar : 	M in 12.5 v 2.5 n her invidue ng be ve as case damages when is get dar mage to Poche on the put on the put optimizes to pro- Korponer lare Rochelis cells at read	m Flagging w/ med: disable times as possible people problem magical when ge its call is when what is true (16-16 A)_ monals = epsiver	BOD am carter   this is co an Pocheck calls, unspoch s 	her cus ( cu bracktingh entry to seed peopling at 043; estert

1 :	E.						2 4	· .		12.1	- i		· ·				· · · para		F	· - ·			1	1.2
	+ -	i.		,					-  -	. -	ł	ł	ł		 1			-			1		۱ <u> </u>	
	+-		•	1 -	* *****					┞──┼			''	·,	. <u></u>							-	-	
	t	Ch. 2	aa j	Jun -	د سريدسه	1. C. P.	+- &	* _p~	<del>ار</del> _علیا، ا	re-n l	مه م	ه در	loce	where a	_ as' n	men	40 tr	•_  →( 	i+ e1	, .Lee,	ملاب			1. All
-	1	-i: - -		1 4	. C. P C.	<i>س</i> ) <u>(</u>	<u>א ר</u>		he	<b>!</b> ¦	-		•	· · · ·	-			-		-	*		i	-*
	.;	-4 -1- 		1								-	- 1	1 1		<u>-</u>					•		1	
÷	+	1.1 2		+					0	i		410-	· •		+				+ +-		÷		+ 1	\$
-		<u>v</u> -"  <u>v</u>			en ge	s ou	ngca	be	tore_	rensia	P-3	هما و	ac, c	لفعم	7~ 5	n '	يىر <u>ت</u> م	ا منها	اليورن	eli ger	مله م	هرهر		-
÷			6	feer Ju	gen	are	a_ (	iews.		-540	++ +1	at	Coler	r tre	347	re4		~\	ھے	tect	1	. <b>4-6</b> . '	- i	3
-	Ē		40	دينه پ	0	ا	¥	,		-	*			-	· -	-r		1	,		1.			·
		The next		1 1 6 D 1 D 0											 					+	<u>†</u> -;	•		
	-	_ 11-2_ 14-4	20100	- 	nulse	train		1		l'int	1100	<b>.</b>	· ·		1	• L	land	Year			-51	+6	· _	۲.
	1		ي. ما	t the	Gre d				- !	earlie		. 1		+			, . , .			:	' - 1	,	• †	
-	ĩ			-	••••	3.	)								1			 -		5	; ,		Ì	<u>ہ</u> ۔۔۔۔
					• •		4	navi 1811an 1			,	1			 ,							-	1	- ;
		Empore	· ·	į.	Hernal		,		Lord		- loss		mude		. : •		- - :		-	اديم		1	1 +	9. *
	1	· ·	,						:	1	•	1		 			•-> >				, . ,		E E	
		· · ·	 	, ce	ber -	to c	lean	yur	nd		1.	۱ <sup>-</sup> ۱	1	'	_	- *		‡ •			-		·	-:
			l I		t	, 	<u> </u>							'		'			-		Ϊ.		' '	ţ
				-	+		_ '		_			ĺ	+	:				:					l	
	۱ ــــــــــــــــــــــــــــــــــــ	بن درما	aller	fivid	- 1.100	y 'm	exclern	يسا_لم	r, qn	Newy	n. fi	1+0	en or	the G	++-785	-							[	
			r		}		1		- <b>1</b>	_					-									
	¥			j	إاأهي _		<u>k</u> e	_ piede	<u>.</u>		<u></u>	ایرون	Smff	n i	k. G	her he	*2.5		te_f	H232.			+	
*	Í.			 	111eu 12	their is		- piente na k , _	44 <u>-</u>	nill g	مع ده معن	د معداء وداره	STNF4 544	n f	k. (c) 4 (d)	ل <del>مب</del> المب	- th		ten fi	16232- tr <u>e</u> 1	 	امەندى		
R R -		+			بيني بر بر	these is were	μ. 	_ piente ** k	44	6.111 9 ha.nk	se_co cen	ودانه -	strift 944	en it	4 5	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	مربله —- حود م	 ;×		th 1		looks,	-	
*	-	+	-		الأمي ۲ ۲	that is		- picete ener	<u>ku</u> +	<u>1119</u>	<u> </u>	<u>معرمة</u> مردده ر	5775A 694	9-14-	4 5	1 10	مربه حروم		€_£	tre 1	2	- -	-	
- -					الأحي ۲ ۲ ۲	their is		- picote *** ,   -	<u>tu</u>			-119 	57757 584	9		1 1-2 1 - 1-2			€æ_£	Habbar -		- -	-	
R R 	-				¥ .	their is		- picel:	<b>4</b> 4		<u>.</u>		<u>377</u> 64 44	********* ********		<u> </u>                                   			1	Horas -		- - 	+	
*			-		باللي ال بلا ال 	their is		- picele 			<u>.</u>		374 944	9		<u>                                      </u>				H):32 -		- - 	• • • • • • • • • • • • • • • • • • •	
R R 4								- picd.			••••••••••••••••••••••••••••••••••••••		<u></u>	****			2353 			H: 232		- - -	+	
R 5 5			-			- were		••• • • • • • • • • • • • • • • • • •			••••••••••••••••••••••••••••••••••••••		344 344	• • • • • • • • • • • • • • • • • • •		<u>                                    </u>				16:53 2		- - 	+	
k k			-								••••••••••••••••••••••••••••••••••••••			• • • • • • • • • • • • • • • • • • •						16:532 - tre 1		- - 		
-			-													· · · · · · · · · · · · · · · · · · ·			€	₩ <u>₽</u> 32				
											• • • •								<b>-</b>	₩ <u>₽</u> 3⊁ -				
			-					- pickt			<u>.</u>									<b>M</b> D32				
											<u>54. 00</u> 64. 0									₩ <u>₽</u> 3⊁ -				
								- prodet 			<u>34. 60</u> 6an 								<b>Aa a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b a b <b>a b a b a b <b>a b a b <b>a b a b <b>a b a b <b>a b a b a b a b a b a </b></b></b></b></b></b>	<b>M</b> D3+ - <b>·</b>				
											<u>54. co</u> <b>can</b>									<b>H D D A -</b>				
																				<b>H</b> D3+ 1				
											<u>.</u>									<b>M</b> D3+ -				
																				<b>H</b> , <b>D</b> , <b>S</b> , <b>C</b> , <b>C</b> , <b>C</b> , <b>C</b> , <b>C</b> , <b>C</b> , <b>C</b> , <b>C</b>				
																				<b>H</b> D3+				

·).	Pove 1134 - 4/9/2013						
-		Prochala call co	a an the St	Pahala call			
	usvally TF	P can get d	ralled, but it	yn Pochale c	ell gen drilled	it is a signif	cent .
		iciam That I	2 <u></u>	n Tu custy	(.( 104 , TFY , e		
	- Donitry is the Pochets	call han drille	t_ effects of d_on him	the seed _ 5	ed , so Day	has indicated	al abant
		how the E given the ci	ange vi err	ent necessar	g +- ger u	onsietent pe	borneres
		n <u>m</u> <u>sert</u> e is worder	tre_she_nd	tall.	ol. tr_Engo	- could have	changed.
		- apparents	Lene gets d	el the Empe	ere grine_c	an open headed	dust shield
		- Oct o	ne a lens	the med (	ensing ; this	can chong	ge the beat
		<u> </u>	Dave by price	ver seen grt	report inh	ne the coast	the soul literation
			from that	othe			
	Entre proving x Spine to				· · · · · · · · · · · · · · · · · · ·	· ,	+ 1
-	- + foretre	-joint la back	iclication off	of one of the c	the upper - not	buy to worry ch	out in this case -
		+ suppose it	estably the w	re plose slyhar	Long mare h	e recessing to r	
-   			_ ! 			, ,	, 1
	Kockis cell dianege : the beam mant	of the carry of	ade in the Pop	hab cell?	. then could	be whythe _	Beckels cell as
 	getty drilled	n at lease	where to freet	ny then to.	ietye the wanger	marsh, har dy	er not surpect
	the carry each -	when retri	role in detent	·~ tre fosition	, of the beam	ue. 54	
	Daisny sightered	tonchy of E	a voue form to	ty to company	, hug- 14 had a		the surpres
-	(S-uh on the	014 ps _ Spir	hre <u>) i you'd y</u>	ent to adjust	the basis of the	- (	peak.
	pour, the bech it a to Dave's not	lf - have the	bours of the	pump on the f	cee of the rod	could result	n demagn .
-	· · · · · · · · · · · · · · · · · · ·		I	-	- -	· · ·	

	· · · · · · · · · · · · · · · · · · ·				2
Davik in colum	try to faces for the	focung amoral al to	" prove beck reflects	an = the point	the screw hein -
on the a	on my soloch did	hit allow him to	get the position ho	washed so he moved	+re   minty
- While Dove		ngh beesky I spoke	to John to poils on	what Dave had said	Re: the factory's
thought	en whet a	constructions	ye + to Poch	where it. According 12.	Jann:
	it & common for the reason for	the beam who	t to not be in t	wrod fill the lexe	itation volme mude
	the rod	fall that the ca		e detter mate by the ima	peng of the end
	rod should	could not see a	early answer or	to why the boos of the	e Emposed mo the
	↑	·····	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	
- Bean worth - Co	cathet a caused .	, should be no t	re centra ol tra r d-daniza to a coat	ad creaty = re	a beer for from folley
c hat >q	of " - Donity sha	general the tar	definitive my to det	en na tra y to serve an	
- Dave sytemated 1	ng_trigus into ladgivent	the position of the	trong seconds these	the Engoint hean first	
- After exposis	he the end were	I to Petral call	suffere Dove in s	HI getting the real spot	н 1 - т т т т т т
Next, Dove + NV	90" rotate ~	stror that conse off	the TFP		
	the the one be we	ted to cotate 180°			
Dove spoke.	+ Prity agan.	From what it so	ends like - the f	etang in the the second	1, holy trinky ?
	eos_then forme en that we are a	of end on purpos	the rod	y different _ sp the p	-mp is 'getting
the	toched have the trajute been in y	elding a small	e carity and ,	with sufficient intersty	+ danige the
1	La gour by en	he canny in	ode looks short	the same size on ep very shollow theirs	ch offic, so the

-	ana an		4 M S				1-	~ i .		_	- 17-5 1	- 1					-   -	+-	· ·	, . 	•	• -	1		•	
		• • •	' 	1				<u>'</u>	ا ام		50			ļ.,	7	.00	<u> </u> i.							·	2	
	1	16,9	7.قد	* 		" 2MI4	3.5, 	thore and	1 24		т са 1		يمعا معاد	44 ( )	• •			<b>ins</b>		1			Land			
	1			بالم مالع	rye Lead		، جو عدا ا	,		ر ۲۰۹۱ ا	d 11.4 C			,	', <u></u> <u>₽</u> ?'' 14	سي م مع	1	-H			له و ديد 	د معي امرين		ite em		
1	ł	'		Ī		5		 :	-fam	7					- 1		Ĩ	104	, , ,	1	(144		Scieb	~ ~		-
		1	,	1		,								, 		,	-		·	-	!	1	1		1	
	904	 เริ่ม	eren	ne	لعم	' and	-		*	-	. eve			l Le om	***	Dove	-	- 		1 5.8	5	1 120		,	I	
		•	· · · ·		l	· 1_		3	,				- t f=					- 7- 	74	do a		ant al	. Ste en	e in i	ne :	•
	$\mathcal{D}^{\bullet}$	we'r	40	<b>A</b>		•~** ?	. where	in he	_M9	sund	4	16.5	ພ	w.e.	~	md	~16	<del>د</del> یا		ļ	-		i		• [	
- 1				, ,		¦ + +	1.		3	1			,				•		1		•	I			_	•
	٨o	da :	stre		(ao	ing at	ء ابت	- A	فمم	اللوذا	· * ,	وارع	ja nam	<b>₩</b>			• •	<b>.</b>		_			•		' +	۲ 
-	-	•		*	-	+	-	_,	,		-	+		†								-			ļ	
	اه ,	ijo		• 66.	-م	<u>oui</u>	n I	Leng.	tout		m_s	t	ہیں	54		helo"	600	~d.	<u>tu</u>	مله م	<u> </u>	_?:::n	-tret	tu	-	
1			9~	40	ĸ	que	6ملير	بط	- 1	-	•		: +					-					ł		ļ	
					-	-	- :	'-	ʻ.,	-	; †		•		- '		.   .		+	  -		-				
	Ţ	we	spoleo	ومد ا	ts	Printary	1.1	لاجيع	ayee	n	ໆ ່ ຢ	lon 4	المعما	. +	معتقل	e n	<b>د</b> . سا	ز م	- <b>`</b> +	H,	eploced	d <u>i</u> (ens		like	Ŋ-+	
-	<b></b>	-: -		÷				ا،بر	40.04	. <u>be</u>	۰. ۵	<u>اطم د</u> ر	4 101.	<u>ر -حا</u>	·	- <del> </del>		·'-			~		<u> </u>  -			
1		1. 1	÷Ē	5	يحر ا	s go	רי' <b>ז</b>			*	• •	e	e upe	لمع	واليان	<b>د. مله</b>	the	Se.+	fre; b	opeh	"y.'	tur ~	111 51-	• -	-	
I	1	-	t	ł	٥	~ * * * *	0. 	ne1 <u>"</u>	bea	~ 0	ا،مب	نعه	15.40	+ مم	101	•				1	r '		- ;		1	-
			•	retu	ined	th S	<u>00</u> ~	<u>-</u> (c	י לו	,	-					I.		 	-	-1	- ,	- ,	ļ			
	- <sup> </sup>	~	- 5	,è+ [	-8	te	~ <b>6</b> ⊅<	: <b>S</b>	<u>as_f</u>	بعالم	2						┼	<u> </u>			;					-
		,	I	ł	r.,	1. 1	1-			:			•	1					'	-	'		1		1	-
		'		+ -	1 CX	4C/n e/l 7			,		Inter	nel.		-	-	-		-	-	1						
		-		-	1-1	Ene		; -	1										. •	•		*				-
,		- <del>-</del>	-	$\uparrow$		1	Ť			- 1			<b></b>		<del></del>			-		• •					L+	
1				1		R	1		1	١l		~		; - ,			;- :		-				ł		,	
1					ŧ.	L.	ļ		-	Ĩ	Ĩ	Ł	-Ó				1-				+		1	1	1	
		de		1		1	4	÷		1	Ì	4	1							-			l	ı	Ì	Ŧ
			- a	ألب	tur	Char	<b>K</b> , T	Some	inter 1	عاطه	+_10	4 50	مد ا		from		Inte		s		ver	and 1	4 0			•
	-		-	,	_e ۱	use	يسرم	- ena	مورد	4	teren	. بو	contre	لى الى		_giat		i		Ì	5	• •				-
		-	-	a	w,	Deve	2	i4.4		  a	been	-	+•	، ام يح	ູ່ກ	the	ميا [	_ 0	h who	- h	1 141	- اوم	i po	ner .		
	- '-		anda a a			shu G	14 -1	hed_	the_b	الترجع ا	5.56	w.	<u>ب</u> م	usue !	- hen	وويل	k al	fe	4.1	<u>و او</u>	h he	em_ 41		lary	<u>ه</u>	
1		I	,	-	<b>at</b> -	the _ b	щ5 .		-							-		ł		Ţ	<u> </u>			v		
÷	-			1	i T	" De	. p.	tred_	N0112	y ti	<b>ne ç</b> aci	and	م بي	-rrat	•1 +	m h	ç'em '	expe	بطعه	form	nd.	and	beckn	ands,	ł	
	-		:	1	1	· -	h-1	1 12	had	· _ J	ļ.#L	e.ll	ect ·	he	5411	fe 14	the	n od	<u>[.</u>	k-8_	f0.	laze .	n the	rod	ļ	
	<b>.</b>		·· ·	1 1 -	i	·			- <b>f</b> re d'-anti-				+ ,	·	-	<del></del> .	'			-,	۱ ، 				•	
- 1	Do	we d	العديط	•	ليع ا	rentd.	. lea	e_th	<i>7</i> 5−¢	<u>r</u> rzł	· -1	~ 6	1) and	1.	* 14	ادويع		-	-							
Ļ .			لم_	thu	- * 4	need	1	20 A	and	L +	250	nš -	5-14	<u>ل</u> ۲	المسيد	-							,			
1				;		<u> </u>	1(1	LIA	he e	- 619	entr	٦ ^	210 .	)		<del></del>	;						• •		- '	-
									$\pm$	+		•			-									, ,	<del>!</del> -	
	out	· • • •	Nod	e.	c	lot sm	يلاب	• ** `	Zove		<u>م</u> د	men	eð l	e - 1	•••••	J "	ŕ	~~~	2 - 14	u p	<b>≁</b> ∖					
è						1	1	1	1 1	1		1					1			1			1 I	i	E	

Dave 's vient - 4 27 (2015 Millenie's down to 3.9 W why? -4 Blein aber + know; we kyle + + + terescope Newport sent: there are the minnes that in - Spitfee Ace 4 Dectory mosted to try these of case a different set of milrors had been installed when the Aco were set up " when the opter neve sweiped, Due said the redrus of curvature lipstud similar and the size of the beam on the lens didn't charge appreciably \_ se set up additional over or and the beam dow to the will (by the day create board.) starting from the 9.5" the tary indicated , Dave moved the felsecope separation , , and back (down + ~8" and out to ~17") and gauged the collimentary The beam we tasty - spin divisy at the full deteres for all instance separations; however, at the difterior across my take, the been appeared verybly calibrated ~13", que or take. Daug left the mirrors with experation of 13.5" 13.25 And touching up to canton of ew Dave switched to be Q-switched - hume appending on D-subter, there was a large envelope linder the obrivat ins pulce train profile !... Date back approxime to becauld call voltage (in source produe) to see it that could suppress, t in win concerned RE whether Pochety cell in fully rotating polarization and conveguenting not - holday bethen the cariting suckchy to a clean sufface on the existing Rechard cell HA . generated a the TDG and set down in the unbilities, along with a BAC. that carcies the throug information the good gold boxes in the new regard use to thing signed and the HU to generate the shirt pulses of HU to trigger the Pochel's cally extra caring Bechery cell . intracanter Pointer chil crossed polansus Lens poper (diffeese) TFP Were plate TFP PC 1 dl fre vertally vouplete extra Lavis "Horaso++11 condece the particit will end with e could see to polanzed polenze when you do it this way , 3-17 Dane's not sure the abbounte polarizat . he had to nove to diffice sclose to the Packels kell you should have a pattern 00 in order to get known light to

	يون ۽ ايون -	t and the	1	1 - 1 - 1			•	1999 - 19	• • • • •	e de la la			ر معم ا	· · ·	
				ļ	-,		<u>†</u>		l.						<u>)</u>
-	when bok	ing of th	e extra	carry "	Padels CI	<u>*:-</u>			1	I		-			-
	1	-	· - <sup>•</sup> -		<u></u>	t	ري	)	:	1	-'		- 1		
	the	lettein	· looked	e more	like	++c	(): • [	w	ricpi	n - T	Zone. di	dat them	6 <del>1</del> 6.,	ik the	
	İ			, <del> </del>	<u> </u>	1	0,		<b>-</b>		CO-SECO	<b>t_</b>	·· ···		'
	-		+						;				1		, - _, -,
	i +-	·	•	· · ·	L I .		-	F	_;	-	'				[
	Due cooses	ned by	<u>+- siza</u>	<u>en 44</u>	50d -	1-1-293		5-9647-	~ 5.5	₩ }_	thinks	tu epot i	ize H	too lage _	
÷			12 -8	hoens	also m	100 m 1	onge	, -'	-	7	on H	he rod -	}		
ł		;		·			† I		-	-,		;			-i÷
	Deve show	 		' his sine		1	دعالمه	ا مسلمہ	22-	,			-	' <u></u> -	
	4.	Dove _	der 10.44		لوارد بر الوارد بر	my ede	وجنبع _ ا	the feles	cope		while		te s	se of the	
		mde	<u>en 1</u>	the road	L- <u>1</u>	المولاط 1	ceep .	-45-+		tings 1	ooned .	لار زرب جمحات	ke A	- 4 -	. 4
		- <u><u></u></u>		family	ended	ing as the	some a	pur we	arrive	A _ At e	n to j	cases - of a	:011.0	at a state	
		e	med er	- f-1	120_ N	med	great	2 and 1	un se	areton?	)	ded a l	laze	Sp>+oq	
-		+*		₩. 					-	; -,	i 1		- !		ļi
			 Duc		 بدوسط	،	- 1 1	- 1	, ., (		t • • •	Antoin -	-	,	
_			La neu	ernal.	-3J. Dave	forse m	ode sr	_ به م	od i vn tie	ent in		Contrary	•( }	-	-
_	+ -	1	· _	fair	- fecas	cope me	به مر م	djuezan.	r 24	+ se.	مد ال	~'/8 "	!		
	• • •	. <b>İ</b>	`w\*	t tis	edjux	net (m	ا مہ	4o	مد مي ٢	بهم ( (د	begen	ينو ما .	ing	vent	
	~	<u> </u>		مول ما	re kept	haven adj	~*~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	. to the	مورعموه ا	re −sqe	et m	( دېددېسې	, ' <del>+</del>	separation	
	مىر مىر مىر مىر مىر مىر مىر مىر مىر مىر	+	   		onk 4	4.3_50	leaver	-+" iu	بسيم	م امد م	red	chan	<u> </u>	ld -p me	
-				·	the dia		سمعہ س سرحا م	e nere	donce	the hea	m. sit	به مورد مرود مورد مرود مرود	_ rodi	he _ "too s	ب العمر ا
						CLASS - 5-	Power 1	oneosed 4	- neally	⊤ ∼`(~ م سا}ه ∟	when the	Great M	- 400	formeral	
	l					5			· · · · ·			· · · · · · ·	المسما	- der -20	
	20 00 02	o. S_A _			Afre the	aoljusta	-+ , De	ne me	rething	~ 6.2-6	ન ખે	-1 20 A			·
	ین ما	they Do	ye up		ţ,	in build	~	~70 .	N Long	gu., He	fred	adjunt	24	o dary, w	*
	94	4.	ا <u>می</u> ت میر ا		:	~~	synifi	ant gains	-				-	-	
			مليولم	+ 1.be											
	20 20	A → 6			b 1	hid y .	• ~	lons in	، معمد مل	han b	efore	lessent	hally	-J. my	twin
				i			]		<b>8</b> - 1.			er De	ove v	ad had	
			1 (1	, <u> </u>	ļ	······		eneratede memorine m	ļ	-	· · •	ivəteil	; , ,,	·	
			;	•				t,	-		٠	_ لې روم	ן ביי		ſs
		1		t .			ł		-	1	ŗ,		tor	- Then fel	┺╴ <u>┆</u> ┊
	*			1 1	+ · ·		÷,	- -	.			_ <b>164</b>	the e	stre pour	Ê-
			L	ŧ	1 1	+'-		* *				<u> 1</u> 4	in se	en di Clam	ixprec.
*	· - ,	1 : -		1	4			 1				Sole	ot 1	- wishos	

		•			1	1	en en en en en en en en en en en en en e	* * * * * *	1	
-+		1	1				+- + ·	<b>+-</b>	t t	1.
	Pauls	CALLS -CARC	cut on a	anele - une		to wedar	in mediace	Lernes the		1
		he con	my - need	Alaman + 1	to can the	to elim	for two		0	ļ
		بر	alter in	Hellet - of	the new	Pour cer	Date al	wed to see		Ì
			the car	ing and the	unal ti	Lod moin	to conging	se for the	chenne	
	•	1	. introd	fred in th	focuen .	ien			-	1
	'	*	1			•	• •	· · ·		1
.	when o	away a	eger ns.Q	survigenced, no	de loom prett		· · · · · · · · · · · · · · · · · · ·	- + +		•
	··	u	· chen ; s	real is inprove	Lud wode s	hous a "hel	st and see	he nerrow	·	, ,
	· •		·	- +						
	وابد د	equery -	Beneti ce	11 - conty peak	d_~61w	⊨		·	ŧ .	;
	-	Ì	+		<u>~4.4</u> v	but of coopress	or	• <u> </u>		
, -	··· ··						· · · · · · · · · · · · · · · · · · ·	, 	;	+
1	Dene	finalizat	ve streten	elignent, incl.	er 6.2 403.23 40	get to BWI	ל <u>ה</u> ' <i>וי+</i> נע ו	are stoke		
	-		*				;			İ
		-			+ -		, , , , , , , , , , , , , , , , , , ,	i ·	1 1	-
j-	بد عدیت است. ۲۰۱۰ میلادا		<b>T</b>		4 Had 11					· -¦
-	, wint	la he	the the be	i i i		Calaria and	'itse ors	g spening the	2	† •
+		+	whiled do	and the second	<u> 207-2 - (2040</u> - 40	Sheens' work	in the circles of		<b></b>	1 -
			4 éven un	the the scores	are transened	pressing on th	in light still one	in the snithe	a little	
_	- '		have		: :	, ,	and the second second second second second second second second second second second second second second second	1 1	1	1
	,	De	e were to	adjuse this sign	hy so that the	. Id_don open	the shorton Ri	14	•	ſ
-	ł	,			J -	• •	•	• ]		
+	Dave .	also stope	D and	manin - man	to check the	seed bandrights	+ st th -	ا+، "مونيقت ا	might not	
	he	' shy	stretchad	energy , here	the smaller	been size	bat-of when g	reded	1	
-	÷ -	60 100k	and of the	they band	dn~_~2.2	-2.3 nm FWHM	; Dave fert tim	should be find		
		+	· · ·	is ownide the	Spittine 4	factory feel	1.5 nm for	a 1.2 ps pull		ļ.
-+			•	'					l'	
	To five	11e :15504	e with the	entrance shatte	Dave created	a shim out of	a most of dec	k type and b	yed it on	I
	He	liel _		·					-	L
-	,	مين <del>ين</del> ه ه	M. Dave	i would have he	to shift the	mode down	and_realism	the Spithe	<b>K</b> • -	1
	+ 1			the second			· · · · · ·			·'
	- 1 1	, <u>, , , , , , , , , , , , , , , , , , </u>	1 01 Olver	teres server	- 00 10 Hak	· <b>·····</b>	•	- ·	I	
	. Even the	and the	been size	-	ender Dove		accered the	+ down to t		
		4 46	filel 'sia	of the nod	i i chant r		, 	f	~ confrenssr	
		1			, , , , , , , , , , , , , , , , , , , ,	try, "+5			·	• ·
•		1.						, ,	ŧ ,	• •
	•	· · ·					I		-	, i _
		 	·	 	· · · · · · · · ·			, 		1
			ı	1	1		1		<b>5</b> 1	1
		- :		ł	·		1		•	

		+
÷	Dave #1914 09' 08 2015	·
		_
	Per Dove:	
	we're the only group he's amore of using a 30 fs oscillator	
1	abcert other trans is one that were a 30 fe oscillator to seed a 30 fe and 100 fe reserves	
	to they had to he as the entre subtron in one on a the faction developed a belearned in and the	·
	in the second share a mark of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second	
	Seed site carrier	1
	New T box - switch is on a board in the power supply	1
	R12. trimpot for the diodle	i
	in new Millenia, freeze 3 trimports, but 1 it is still R18	1
		11
	th SP Current rode the Millenia "remembrack the old get pont" ( want 13 10 @ 809, and	
	should get 6 N 2-4	1
	$24.14 \Rightarrow D^2$ $\sim 4.7 \omega$	
e.		
-		1 1
	on the meth screen in St Current node: reports current power	1 ‡
~	allow you to directly see the current on each divade	1.
	1 23 W dudus	
	a cannot bet the current range too high	i
	in indida & hurd is undate, and allow even has an in the fact of the little	
	because the part of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of some set a contract of so	I
	Dist Draw in the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the second construction of the seco	
	Objects: 1 - both entress and teng need to be set in the Setting succes	.
-	pocess: _ 1	
-	pocess: _ 1	
-	plocess: _ 1 both enrens and temp need! to be set in the Setup suren in SP Current mode : set arrent and temp to get ~66 is and proper divide reformance sultch 1+ Power mode : will reset the current to get ~3.5 is	
-	pocess:	
-	pocess:	
-	poless:	
-	plocess: plocess:	
-	podess: podess: n SP Current node: set current and temp to get 16 W and proper divide performance subtch to Powe node: will reset the current to get ~5.5 W 	
-	pocess: pocess: n Sl Current node: set arrent and temp to get 16 W and proper divide performance suitch to Powe node: will reset the current to get ~3.5 W Ned to wat us min to recalibrate alter, go to 100° 10 in current mode - 100° 100° 100° 500 m Recolibration Recolibration page 351 m FWHM	-
-	poless: poless: n Sl Current mode : set arrent and temp to get 16 W and proper divide performance suitch to Power node : will read the current to get ~5.5 W Ned to with any allow you to go ~ 2 A one Ned to with any allow you to go ~ 2 A one Ned to with any allow you to go ~ 2 A one Ned to with any allow you to go ~ 2 A one Ned to with any allow you to go ~ 2 A one Ned to with a performance Recalibration D1 24.8°C 2.8 -2.9 nm 4 Long on the symmetrized	
-	poless: poless: n Sl Current mode : set arrent and temp to get 46 W and proper divide performance sultain to Power mode : will reset the current to get ~5.5 W 	
-	Disert   D2 set poless:	
-	Diset: D2 set process: In Sl Current node: set current and temp to get 160 and proper divide performance suitch to Power node: will reset the current to get ~3.5 D La vill only allow you to go 1~2 A one need to vert w5 mm to ge collibrate alter, go to 100% in current mode - 100% should yield ~55 W Recalibration D1 24.8°C 2.8-2.9 mm D2: 24.8°C 13 N @ 809 mm 24.90 and 13 N @ 809 mm 21.28 A D2: 24.2°C 13 N @ 809 mm 24.90 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 21.29 and 13 N @ 809 mm 23.1-3.2 and 24.10 and 24.80 and 24.10 and 24.80 and 24.10 and 24.80 and 24.10 and 25.10 and 26.10 and 27.10 and 27.10 and 28.10 and 29.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 and 20.10 an	
	District 1 2 years point and the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the	
	District 1 D2 set process:	
	Diset:   D2 set posess:	
	Diset: D2 set posess:	
	Docessi - production of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set o	
	Decent in the place and conjugate with the set of the local is a planet in the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of th	

/			 17 = H									]_		- ! -					,	- ,		 
	•		<u>                                     </u>		<u> </u>												: 7		+	، ا	, 	
-	with	3P aver	ا_حو	mede :	~4.2	ພ		-			-				,	<b>-</b>				;		
-		4 after	Dig	مه مه	Juta	<b>+</b>	-		1	-	1 - 1 -	1.	¦							· '		
				''.				;			—	-		·	• 				:		· - ·	-
		touch		e HR	00	Mon	sful	- <u>Ber</u>	\$	3F	<u>-</u> 4	<u> </u>	<u>4'</u> 2	·			<u> </u>					
-	·							·			'		'				: [		; -		ı	
	Alte	-cssidur-	\$_ <u>14</u>	"  set- (	, atries	- Short Ch	مبىرە	- 4	Pouro_	1 ~00	:_ عد 	(No	τ_50_	Pom	امض – ر	ہ_ (م	s∧d_ y ∶	<u></u>	5 m-	<b>^</b> , -	, -	
-				, , , , , , , , , , , , , , , , , , , ,	🕂 -			1		1		-	-		~~ ~~ ~							
	22	- 21.	s{		<u>سا +ر</u> ا	eek :	22	<b>A</b>		abe	102	<u>محود</u>	F-7	2	5 <u>A</u>							
	23.2	→ 12.°	3			• •	23.2	LA						12.	ŦĄ	-	• 4			-	- ' ' -	
	• -		-'-	'				'- 					,				'		1		-   1	٤ 1
	-	_'4 w	10 96	7.3 C.	rent	-	-	! ; - !	·'	.¦			,		+		·		- '	-	'- 	·
	.4 4	<u>5.3</u> W	C	00%	Curren	*				+					÷		•		,		· - +-	
	-				-							÷	-	-	-					,		
		cald	ve wa	the peak	~ - c	SHG		MELLE	~ ?		*_5*4	<u>ہ م</u>	nd gru	4_h∾ !	• • •	تعامير	rate		-'		Ï	
	 ,	hetto		R. 00	the g	yde ol	اد وه	، دىس	an	ן <b>ז</b> י ו	ا يعا	دئىمىر ا	<u>بدہ</u> :	i-			·					
	•				····· ,····			more	vede	1		¦		+			·					
	-	,		- ' -		-	-	· -		 1		 		-; -		-			,	- ,		
-	1 FWILLEN.		ent c	loss shi		وبهميعالم	00		<b>~</b>	Dave		~ œ́	1_5 - C	ه ب	7.		-	-	•_			
		-+ 			•			(				-		-		-	•	r				٠
	Noaci	wann a	1.4	Jojn ver	<u> </u>	و ترديد	ودير	<u>يد _د</u>	- <del>1</del> 4	ADM	:_•_• 	ν				<u>le</u> (>r	, ,	400 9 404	J .		,	
-	†	•••	.'			 		• _ • • • •		:						-	-				'-	
-		CLICK 4	م ده ر د		• <b>***</b>	15000	~	He.	<u>ر</u> ب ب	Yand :		K-41	لودي م. ا	20							-, ,	
	-	<u></u>	+	0	wrat,	1190	~ U				' - : .		4.0 6	۔ م∽`د			, ,			-		
	•	10 200	lacks		Anni	<u></u>	ې•		, water						·T		· · ·		•		1	
		10	1					i						• • •					 i			
	•	Sama	ا مدر مما	- 	-+1 C	-' - 0 0 1	, 	المدا	, - 	~ •			2.0	1	.08.0	lv. de	۰ +	н <sup>н</sup>	•		' i	
-	•			-ye vi					,			• •		.9.04			1	1 1	1		· .	
	E a home	. en 3		2013 1	, - 1 00 1		1			,				· ·	-		· · · · ·			• •	1	••
Ē	· · · ·	7			-100 FL	. اه	cre +	vo le	54 a	bran	1		-		-			. 1			. 1	
	· · · ·		[		- 75 FL	<b>.</b>	' [ <sup></sup>		0						_		1				- i	
					1						•			1			to do	the	cdjim	trat	~ I	
	ander	the the h	,	un clso	clissin	t on the		stre .	1 44	Far	ado-	4 . 5	olator				16 00	J color	,j- clək	to the ,	opertury	
	<b>n</b>					)		1			• •	د		•					1	Cols		
	There	wes a	ship	00055	their m	obe Life	, the	street	w.	<sup>t</sup> Dev	س م	~ c	معم	ed ,	bout	align	nunt q	is he	sent 1	L.		
	Towna	~! (w	: Hue	>lop	did <sup>1</sup> no	of line	' vy.			•	İ.			+	-	-	,					
		ocme sould.	misch	samet a	ord n	لمتصليف	shoul	her ,	-t' of	eng		مارمو	y not	m -	too si	he	~t	+ +	1		1	
L _	in	order	6 9	ex the	zi hed	the al	igned	l Do	we h	ad	, , 4-o.	chan	, cr tu	, engi	4 2	the	'greti	~ + · ·	ه ااجب		-	
	•	rolate	the	V·m.		> 61 -						-	U	5			-	5	1			
_	l , +-					1		3		,	i _										· '	
1	1				1									1							•	
	1							•			;		-	1				-	- '			

				J										·** · ·			<u> </u>	!		!		-	 ł
			; 					····· ·			'.											3	
	Dove	0/75	wea	the	line	deron 1	مٰہ ما	ه مع	even_	alw	ţre a	celju	*~~•	۴	-					-	<u> </u>		-
		6 h	 	suelly	do	esn #	see	kn	mt	- tu	. 15	600 g	source	<i>a</i> ro	+~51	_	-	-			ļ. '		
		1	i	picose	cond	-5454	m_h	••	e 2		9~~~	ve gr	oty.	_!		_ !	-		ا 	 	↓!		
			ļ	انه مع	d th	" extr	o dist	<u>يەر، ،</u>	<u>n 1</u>	<u> </u>	هْد نْع	<u>- 8±</u>	dow		<u>-</u> 2	ICO Sec	inha.	band	lung.	rs_	 		
			-			!' _		i			;				'-				!				
	Deve	-ta-	-	İ	-		'.	Ļ			!						 	<u> </u>		· -		-,	-
	cfru	مغزنه	shn	g the	streta	cher, D	save 4	rook	۰ ۵	small		is and	ا س	والدهط	l it;	محتصعه	tre.	med	•				-   -
· ,	'	vertical	hy c	بل بلو	n tent	diy:		, , .										<u> </u>		• ••••••			
· ·		,	w	wetch	ad t	he spec	tim	or h	4 14	ved -	tla i	uries :	-				- ! -	-	, 		<u> </u>		
		,		l lo	hange	ا ينه ا	ne spe	etro	<u>n</u>	s ne	reve	ed the	e iris	vert	العال	1-1in	form_	<u>v_n</u>	A	<u>slignn</u>	<u>ا هې</u>	S_tre_	
,						fool	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	7 -										<u> </u> '		-, -	1 1		1
<u> </u>				c	<u>ne 'ye</u>		the s	rect	~ <u>m</u>	وم	he !	-ove d	<u>tre</u>	1015	ho	n Zon	<u> </u>	- inf	<u> </u>	<u>, en</u>	10.500	<u>.</u>	!
		·			'Yu	- fo <u>c</u> ur	_o( th	۰ <u>۰</u>	•••••••	heh	•nd'	the	skei	che	•	- ' -				1	· ·		
+	-			 		- 6-	ned_4	_ ad	ljure	tran	داديله	<u>e</u> ~		_1	L		· '	ļ_ '	-		:		-
		;	he	่นคร้า	t s	vre u	-+	دمار	red	tes	•	effect	ь, ti	<i>م</i> رج	-, e	Her.	than	the	Ū,	checks	tet	_	! -
- ,		÷	ļ	yve c	eed	مه ا	<u>درود</u> به	₹_'.C	hv pe	<b>d</b>	¦-					· · · · · ·	- <u> </u>	<u> </u>				, ,	
:			1			,					·		,	: '		-							
-:	Canto	803	hur	, 24	5 • 2.6	na ben	dur att	-	-			-	-	, 1			· -	: <sup>1</sup>		'			!
- •		-,			-	•					:		- ,	1		1		1			· ·		1
			<u>  ^0 _</u> (	~ē%r - 0	than gi	<u>k - a</u>	ethy s	ignik	scrt.	بييما.	m_!	0.1 1	<u>~_i</u> ł	નુન્	- 6.1.	<u>eu :</u>	₩. f	Jet	etme	.)	1	-	
	-	+ ~		- ب	Dave	سربعا	<u>א</u> איי	ې يە	few	_n <b>m</b>	of	chan	۲ -, ۲	they	+ +	~ *		ay	-1	fente	، میں دڈا	a	:
$\vdash$					v^ '	ņ	-	-		!	-	' -	- ! !		-	'	-'					'	•
	-		1					-, -	-		- 1		¦	'	-	-	-	;					
- true	sere'n	ف هميخ	due .	peroly	dt a	etetre .	can_b	<u>اما _</u>	eune		s	mæ	the pr	ang	<u>。(</u>	يددي	طر و	<u>k</u>					
	,	De	re 5 (	(LA -	tia a	lieds o	~~ _e	- 44	mu.	يددطعه	-	¦	' —					·		-	-	' - I	• .
			;			, -	- !	-	'		-			'								,	
-	,	-					1	-			-	1		- 				I				,	•
' <b>Ta</b> w	e _t	epter	<u>. 6</u> .	the Yoc	<u>whi</u> c	all					!-			' '			-	. –	••		• •		
	<u>- h h</u>	star het	K_ <u>1</u> 1	<u>~~</u> ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	nime	hier	to rup	olu ce	the	عدم ا	and	ula '	the year	a to ! ;	align	1~4 (4	rugh	:			•		1
-						+	- :	- -	-1 -	-'		¦		· ,			-	-				-	1
12	11. SU	4+6	hely	ull <u>.</u>				i j.						;		-							' •
'	<u>  *</u>	wet f				pers,	110 10	<u>to</u> P(	0_CK	; ;													 1
			' -									alle s	. /	,	И.	6			he	,			ı.
- 70	e 1501 J	• • • · · •	~	1 <sup>01</sup> 7	_w		S'COACE	K.		م مه، ا	a che	~ <i>}</i> @rn	~ <i>/</i> ~		,.	UR A 540,	, ~,			"Y.	.,		
- + -	The	at civel	~ ~	···· /4	41.j	our of the	•~ •		~~~~~~		1		1	ן ה ניי		,					•		
+ L 1	 	 	 	 	- .1	4	<b>D</b> - /	Å.,	ىر،	1.6	1	۲.	۲. هر ۱۰. ۱۰		н эл "И	۳ <u>م</u>	-	•					
- 10		·· <b>·)</b> . ~'	- <b>-</b> -	(	• <b>+</b> •	i Kuri	1.04	ر ان ان ان ان ان ان ان ان ان ان ان ان ان	<b>x</b>	the s		89"" Ro F	ared-	19 1 2	y no International	GEON	7 	A	le an		-		
		''	<b>'"</b> } •	Rettor	-7-		. + . . + .	ache.	J	ر	2		7			. "	, /~	1	ere	<b>γ</b> <sup>2</sup>	r		
	Deve	browel	-71' .i	te le	"Y	alate	، مصل ال	ette		a'100	seal	Ŀ	m-h	د ا	<b>.</b> .	Jahi-	<b>.</b>	er 1 -			•		
	. Vrvc		••••	 						1		-		- r	_ ]			~ ~		-			
			1	i	I			Ì			i	1		-		-							,
						,																	

la se a construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la construction de la c	'_			
			. 4	
looking at the mode of the Towner's over an extended pathlength (1.30 (24) the mode out of the	. τ <u>.</u>	neni		; 
10963 OK	;		· 	 
arean regardy the focusing within the stritcher?		<u> </u>		1.
		;	1	·
Greassily the mode of the Trunching and pullect		1	1	;
is the mode of the reason in controlled more strikely by the quality of the certity the more	tre	etc.		
ie. the reas not the seed				1 -
		:	-	1
They would be such the analy of the areas in the second of the	er min Lassi	<u>(</u> )	· · ·	
more ( the an when the must i) is also have the state characteristic the the		Deu	1	
and the way is a set is a set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of		Dave.	{ ,	
		-' -		1.
				+
The fectory decigned the love for rege with a 10 nm seed? - 11.3 nm bond width	'			- 1 - 1
is intended a bordown filt bits und to cut the fordualty to loom, but Dave	fand	. that	+	- :
cut and the mean every	<sup>j</sup>			! - '
			·	-;
- Looking into the solar, we notual it looked "signified" hefore the extra country Pocheli cell, hit	~=	- '	-	-
belie the polense	- '-			; +
In it looks like Dane hidrent realigned the personge after tity ching because the	• •	• .	•	
bedry in thigh on the alignment tool	' -		-	ļ
+ considering, it we choppy on the polante perscope, which we causing the she	pe '	1		
			·	
Ere Portuely cell alignment:			·	-,
Deve had a yack of 3 potensen (meigher I 40 I); promy the grace hefor the Bochila cer	<u>ech</u>	-eun		<u> </u>
the sun regular on the two polarious (one inform and one after)			-	
	;	'		!
Dive swegged back to the old Pockel's cell	ł		i 1	; .
is fried to ace lowing	1			1 1
had the align that county + the seed - year evid where h only .	•			
when he related the were never and a little low - a higher TEM- many on the differe	NV .7	FM	I	;
have delegat charby by a catcher descent				, -,
			1 i i	!
	;		,	\$
ger som polo ; sinnen vor te gersonnen	.'			,
a more because (concer 20 TC "bad informer "but around the	1	,		,
	1			
when we old returns cut Dave un gritting is to be une on a bed Spoil , who is a good spoil	'			
is an time foreiton, the mode appeared to be children on the edged the Packets the	:	,		;
Dove worked to a new point on the Power cerr				
in agreen ~ 5.8 W w/ a-surrected and sudded	i			; .
when seeded, setty a Burt BURT of 40 ins instead of 20 ins			-	
Ly due _ + the mode of the seed being _ smaller?			'	
				13
#### - -- --5 1 · · fectory : 40 ns build of reduction time . 1 . 1 1 1 . . mode looks sman; ring is inside around the mode to seed usually doesn't impace the lite of the model nee of the but we're every that with the exect inproduced, the one pro node is maybe 80% of the abon 20 · | | l i 1 + 1 + 1decrease the longer of the \_\_\_\_ the hold we are seeing is due to overfreening of the prop !! is Donin' were Date its change the trung of the a switch in the Empower of Empowed pulse in order to in to be lefted my RS 292' coble and started the Empone officiers ] decreased the every Deve connected the Empo - ( und in group to spithere · . . . . S More effectively prop Dial and secure commiscation from the Empower rod.? is made Dave's computer, and it and could not communicate Dove had a change fre channel from COME to COM 1 in order to get communication remained to the complex inclusion to done the Ace - still no communication ; the complex is directly recognizing the Empower of CONI . also restarting the conquest we got communication Lo Dave set 1 the Empower public width to 2000 he (from 5000 he) \_ ------the girt is then you want to delive the energy to to rod on efficiently \_\_\_\_! is possible - if the pulse is either the long or the short is will not pump the rod en effectently . . . . . . . . . . . . to change in the potennesses - in feet, power may be a little "lower, but not a bignificant change --- -'---the halo is still there . -. Durish' said that the mode when shaded a expected to be smaller than the mode when Q-switched Dove mentil sugar muster the stade laked a little brighter on the security music aller the TEP. is be translated the worror, then peaked if the carridy again and wh. 3' w w1 the seed (was w5.6-5.8 w before translating the a street) reached 6.4 0 w/s the seed Dove replaced the Pockets cell and touched up the carry flignment again ... -Pattern when Pockels cell is aligned: isoger (inagingi' isogyre) \_

	, ;	-		1	1		1 · 1 ·	1	Р. <sup>1</sup> ,			•			· ·		1	1	,	•			:							1
	-; ;	4	is lo	~~ ~	400	mak	sure	 		+~	 nun et	 • •+	1.14	a tra	 M, (		çar	snd				i cuə	+•	alışa	. +•	cent		ŋ-		
		-			yre	- A.G	'. h ¦	~	مدوه	1 . hr	<u>nvef</u> -I	<b>3</b>	<b>~</b> .	ما	d	10	' <b>s</b> 	i	s _		-		,	-			:			!
			Ling	ەر بە	ولوير	. <u>un</u>	-too!	<u>۔</u>	•	<u>a</u> 6600	هبع		p. 14.	: _وروم	yere.	L_	¦ 		<u> </u>	i 	· ·						; 	 		
	;		وم	لهيه. اسانية	-1	a.			yren	<sup>34</sup> ~ J		3 0	<b></b>	1 0	.31	31	ins.	( e**55			n h	fore	) . 1				, ' , '		-	
	; ; -,	-	د	ent_1	ابنه ا	+_0	2-3-	tch		، <u>س</u> م	own_t	۲ م م	<u>.</u>					 	 -	•	-		•				1	ı	-	, , , ,
	-,			na tog	السبان دسران	·	chair'	<b>ce</b> 11_		- <u>bri</u>	shire			 	2 6		 			  succ	,							 	-	
	-,		-	-	d	יי <u>ר א</u> יי <u>ר</u> יי <u>ר</u> אי	0-34		- 0pe	eton	1	, ,						1		נרי י	-	42-42				<b></b> )	, ,			
	ţ		i.	1	the	mer	us an s-rb	۲ . در م	ten	دسون های ه	then .	لمعدم	-	Jonet U e	-	5039	l han l soct	remain	- 7 105	veln	larg	r 18	•h 1	ba.for	e ; •	hen 1	بغ دلعه	- <b>.</b>		
-	-	-	•		1			•	-2		2,	•			1			ļ	1	 !	-			-			•	'		-
	•		<b>م</b> '	eve † 	rical	e ne.	ా బింగ	. ~	nd, on	ن دې	e-n 41	n Pa	che l	L CAN	, clbe	er	÷. ŀ	eve ge	. <b>He</b> .	'dra	ad a	1 YOG	n e.	the	sand	اسمه رمة	, odn. Led	L I		
						1									י 1		L .	 	 	-			'			-				1
	٠				-				• _ •		•	•	ı			:			-	,			1				•	,	_	,   ;
				a da							,	:			;	: : :	•	'	,	:			ł				-	-		· · -
-	- 		ſ	+	L				* *					-	, <b>-</b> ,	 		-	-	·	•				-					·++
-	╞	1	'		,							-		r	1		-	1	-											
-	- 	!								• •		i 			י ו י	 				•	-		-		-				, 	
	•	-	ļ	4						•	1							,	,	•			•				•	;		
	!	,			ł			-	 -	' I		-	!	1	-	•	-		-	•		-	-	 1	-		· -	-		
	-' 				.l :				;	·	<u> </u> !	 1		 	 	 	·	.'					 !							
Ī	ļ	'	Ì	1	, ,		1			- 1 1	•	•		ı	,	! 	-			-		-	1	-			•			
-	:	1	1	+	•				'	1		!	!			, , ,				•				i			(		, 	-
	:								1		-,		1		-	• •				•		-				-			!	۰ ۲۰۰۰
		-	-	-	•				-	-		_1	-			• •					 '			-			•		ı ı	4
										'-		, 	, 					. <u> </u>		 	 		'.				, 		, -	· · ·
4 4	:	l	4		•		· · ·					t ,				,	-	-	-		- 1		:	,	_ ' 	۰.				
~ ~ ~ ~			,		•	1				1		, ; ,				•			-	1	, <sup>1</sup>		:			- 1	•		, I	•
s my st	-	, ,				 '	·i		,							;				 I I	·'		¦   		<u> </u>	<u> </u>	1		·····'····	*
1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1					İ.	;	1									Ł				1	، بر ا	'	Ì				1		, 	i

,				· · · ·	. · · · · · · · · · · · · · · · · · · ·	•	14 - 11 - 11 - 11 - 11 - 11 - 11 - 11 -
		ł	•			1 1	1.
		· ·			·	- +	
Dave visit September 2	3,29 2015	•		к 1			Ì
		· ·	- 1			-	
Air flow in the Spitfice	¥P:	;	in the root of	hanne. Dave n	enced white	Inder on the	
		· · · · · · · ·	bes plok	thet we	signs of we	<b>بد</b> ی	
- goes from the T	che into the rod	chem bes	, 1		·   ·		· ·
- offer the cad the	mber the future	much at the a	board - the	is the hum	dite sinsor		
	ferra the second	ه میر ماد <sup>ر</sup> ممناطع					
		ening to the	icu _ i	-			1
				· - ; -			
, to check for an from	Dave duconnecto	d the import in	the two code 'cl	nember and	connected o	shart stretch a	•
hole to the bod -	no digged time	end into the	a besive of	wayar 4 way	ch i for all	د دا ۲	ł
Lair flow	slow - abort	I bubble pe	second; Dave	send the is	e 4-+ -3		ļ
				<u> </u>		, 	
inside the TCu		ş					ļ
	sta wasa' sa Gar a				, 1		1
(apent_s is epper			۳۰ - ۲۰ ۱۹	-		-	1 -
- Dow chacked	~ _ cirllow _ before _	the disiccant	the end elies	ينه إسمالهما ي	mich fange	then observed _	-i-
				-			· ¦
- Dark replayed , and	stretce at they	right after the	desiccent tob	1 · · ·	1 1	1	1
- ,				1	1		
After the desiccing to the	the small metal n	wount is a med	sange; Dave	becked the	if to increase	e bir flow.	1
In Initially housing d	deed but the	increased ages	-		, , , ,	;	1
6 Streebert D	an chadred the are	Their one rage	••••••••••••••••••••••••••••••••••••••	and the first		;	,
, <b> </b>		···			· · · ·		:
						,	
	1						 
- we considered prove the pro	ge gen generer one	put and less No	to the promotivity	senser - hum	diting strill died	not go down	 -
. we conserve prove the pro	ge gen generator ont	put and les No	to to monolity	seren - hum	۵. عبر stril did فلط مرً عبو R	not go down	 -   -
. we conserve purge the pur 	ge gon grunner out	put and less Non the humdethe <5%	to to homelity	- senser - hum.	a، بس shill did مليط س جوE	noi go down	
	ge gen generator out	put and lab No	to the hornalisty	senser - hum	a، عمر stril did الطامر عن per	nod go down	
we conserve proje the pro- la Dave hybride ofter they, we saw Dave eventue the	ge gen generator out	put and leb by <u>the humdity</u> <5%	to to homelity	565500 - hum	۵. ۲۰۰ shill did الط مر بي يو لا	nod go down	
we considered proje the pro	ge gen generator onthe he' connector or humostity reading	put and les by the humdling C5%	to to homelity	545600 - hum.	۵. ۵۰۰ shill did ملط س + بع _ Pe R 	nod go down	
ue consective prime the pri-	ge gen generator out <u>the</u> <u>connector</u> or <u>humohity</u> reading <u>flow</u> through the	put and les his <u>the humdity</u> <5% humdity curver	40 to homelity 50350	- 40 see - hum	۵. ۵۰ shill did مطرح م <u>و</u> <u>م</u> وال 	nod go down	
	ge gen generator out <u>the</u> <u>connector</u> or humaniting reading flow through the to tubing humanities	put and les his <u>the humder</u> <5% humdering custor 1 5 lot up to <u>A</u>	40 to homelity 50300 on 100 in and to ref of		۵. ۵۰ sh1 did مطر <u>به p</u> et	nod go down	
we conserve proje the pro	ge gen generator out	put and les his <u>the humdets</u> <5% humdets of cantor 1 5 lost of the All a canfle number	40 40 homestik	State _ hum	A.t. 5411 did het. <u>t.</u> <u>p</u> e F 	nod go down	
we conserve prote the prot	ge gen grunner om he connector or humaling ceading flow through the te tubing humality	put and leb by <u>the humder</u> <i>CS %</i> <u>b</u> detre <u>b</u> detre <u>b</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u> detre <u>c</u>	40 40 homeslikk <u>Sensor</u> <u>on pork</u> <u>inv</u> and <u>in</u> <u>one</u> <u>one</u> <u>b</u> <u>(</u>	54565 - hum	4.+4 5411 did heter - +2 _ PC [ 	nod go down	
we conserve prop the pro-	ge gen grunner om he <u>Connector</u> or hu <u>connector</u> or hunnering reading flow thrungh the te tubing hunnering te tubing hunnering flow before ond aft	put and les by the humders C5%	to to promotive	54560 - hum. - the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second in the second ine	a.t.n str11 did hat.n str11 did hat.n pc 1  	nod go down	
We conserve prop the pro	y gen grunner out tu Connector or hu Connector or humanity reading flow through tu u tubing humanity tu tooking flow thefore and effe	put and les his the humder C576	to to promotive <u>source</u> of the pro- <u>source</u> of the pro- <u>ber</u> humaning to <u>ber</u> humaning to <u>cont</u> flow before cont flow efter	545600 - hum. - flu seasur h - flu seasur h 	4.44 5471 did 4.44 5471 did 4.44 Pc F 	nod go down	
ve consected proje the pro	et gen generator out the <u>consector</u> or humality ceading [flow through the tubing humality tubing humality flow before ord aff human the cod and fibm	put and less the the humdets <5%	to to homelity and to wind the box homelity to box homelity to and from helper art from helper to TCU directu	44. 365 - 4	4.4 5471 did 4.4 5471 did 4.4 <u>P</u> ER 	nod go down	
we conserve proje the pro	et gen generator out the connector or humating ceading flow through the te tubing humating flow the out of flow the for and fiber we the cod and fiber	put and less his - the humderty - fre humderty - fre humderty 	to to humality <u>ensor</u> in rid ope <u>and</u> is rid ope <u>b</u> ( <u>b</u> ( <u>b</u> ( <u>b</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> ( <u>c</u> () ( <u>c</u> () ( <u>c</u> () ()))))))))))))))))))))))))))))))))	5445 - hum. - the sees we have - the sees we have 	4.4 5471 did here Per                                                                                                                                                                                                                                                                                                                                  -	nod go down	
We conserve proje the pro	the gen grunner out the connector or humaning reading flow through the te tubing humaning the connector ord of flow before ord of humaning the te tubing humaning the connector of the te tubing humaning the connector of the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning the te tubing humaning humaning the te tubing humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning humaning huma	put and leb by the humdets <5% humdets c5% humdets confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confie confi	to to promotive the form	54565 - hum. - the section in the section in the section in the section in the section in the section is a section in the section in the section in the section is a section in the section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section is a section in the section in the section is a section in the section in the section is a section in the section is a section in the section is a section in the section in the section is a section in the section in the section in the section is a section in the section in the section in the section is a section in the section in the section in the section is a section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in the section in	4. +4 5411 did heter - +2 Pc R 	nod go down	
We conserve proje the pro	ge gen grunner out tu' <u>connector</u> or hunneling reading flow through tu <u>tubing hunneling</u> <u>tubing tubing</u> <u>tubing tubing</u> <u>tubing tubing</u> <u>tubing tubing</u> <u>tubing tubing</u> <u>tubing tubing</u> <u>tubing tubing</u>	put and less his the humders CS 7/0	to to promotive and to with OK 	545600 - hum. 2. the season - hum. 2. the season - hum. 3. the season - hum. (b-+ abort c. 1. the the hum.al. 3. with	4.4 5411 did hele - 40 - 70 F	nod go down	· · · · · · · · · · · · · · · · · · ·
We conserve proje the pro	the gen grunner out the connector or humaning reading flow through the tubing humaning flow through the content of the flow the fore and fiber the bypen, heaving	put and les by the humders CS76	to to promotist <u>source</u> and the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>source</u> of the <u>sour</u>	shister - hum. - the seasur he - the seasur he 	4. + 1 5411 did hele - +	nod go down	
We conserve prop the pro	y gen grunner out te to here the content of here the content of here the content of the tober the content te tober the test of te tober the test of flow the for out off here by pars, he are and here by pars, he are and here the desticement here the desticement	put and les by the humder CS76	to to promotive source of the promotive and to unit off or humanity or from before and from before and from before to TCU directur to the go down shi to the used one to the used one	susser _ hum. to stard _ olropo. (bt abot c. ) to the hum.al. swill humality ge contrained dec	4.4 5471 did 4.4 5471 did 4.4 <u>P</u> EF 	nod go down	
We conserve prop the pro-	the gen generator out the connector or humanity ceading flow through the te tubing humanity the 	put and les by the humdity CS 76	to to provide the fore the the provide the fore the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the provide the pr	the second by get	4.4 5471 did het +=                                                                                                                                                                                                                                                                                                                                  	nod go down	
We conserve proje the pro	the gen grunnes out	put and leb by the humdets <5% 	to to promotive the form	Shiser - hum. - the sector hum. - the sector hum. 	4. ty shill did here Pe F                                                                                                                                                                                                                                                                                                               -	nod go down	
We considered proofs the pro-	the gen grunner out	put and less his the humdets <5%	to to promotive for and to wind other and to wind the and to wind the and the wind the and flow helper and flow helper and flow helper and flow helper and flow helper to ToU direction to the go down shi to the wind and	Shiser - hum. Shiser - hum. - the sector - hum. - the sector - hum. 	4. ty shill did here _ the _ Pell 	nod go down	

ļ	} 1	• •		ar s T	e tos	· 1		-	-	ł'   -					<b>;</b> ~~ .		-,	-	۱ -	1		-			-				1
	J		 hered			 L		-				       		;	 	··•••		 140	<u>.</u>	   					1			- 2	<u>'`</u> 
	-	,	يعاود ها		Carty		Spil	ele	420 <u>3</u> 4	Pur	•تە ئ	in the			-		- -		· -	-	 -	-	, ,	٠	-		ı I		
i	1	-	;	;						 				. <b>-</b>						 			'-	- ;					
-		Day	<u>به به ا</u>	<u>••`+</u> i	Cont	<u>ميم م</u>	<u>1_3</u> 6	 	<u>+n</u> (	at k	<u>of</u> (	ar flow	<u>+</u>	•[+	v_( <u>od</u>	<u>chen</u>	<u>be</u>	 		1				1	¦				-
İ			n the	.i.a	certe	ndge ,	•	i hrm	جار بر	ch	: - "es 4	, -  -  - ~ (*	 5 %		 ~_~	 ج.م_5						_ _	alter	~3			ا المناسب		- ¦ - _
				ŀ	'	-	-	f-34	· <b>\</b> a~ \a	مالع	_#	45	<b>م،رما</b> ۲.	end	,	sl•	<u></u>	-ftu	مه	ead		Ļ	< 10	<u>/</u> -	_		- 1	, 	<u> </u>
- !		· '	- '	 1		''		¦	<u></u>		1	 			; ;			-											
		i	· ·			'					 	i							/	'   			- 	- ,		-			ļ
-   -	(	Ace				- ,			-		 		-		!			-	ļ	   			-	- '		- -	-		
		'	Dave	دلبودليه	d	be!	<u>&gt;</u> *-2	2:d		ئ ا	**	Strete	<u>ee</u> {	₩ _ <u>+</u>	<u>w T</u> e	20401	_టు_	and		<del>ها · ولم</del>	<b>رسم</b>								
!	ł	'		1		*o¥	~•	0	!						; ; -	- 1.	.6 ^~	ور م ا	end -	1	; c		y.erd	المع	-	1 - 4 - 5 1	· ~ ·	-	
•	,		The a	i Linerae I	.r -	tro~g1	4		tre	, • .•	i me	y ne	~~~~	•••4 -	+-	ايين ميور	he_q	*****	يسه	35	ِ ا	ريعا	ليعيم		_				;_
	. L	;		: 6	the	the-	<u>+</u> בר	<u>ب</u>		<u>- f</u>	<b>*</b>	<u>he</u>	Cour _	<u>4.44</u>	Den	e he	<u>et a</u>	يياممه	♣+ '	•	mau	· +	u co	<u>~</u> <u></u>	، ر <u>ي</u> حر	; ,			
	1			*	Jele	1 •'			-5	' ~r	ب	7	AI	,	,		- ' -	, –	_		_			,		-	-		•
	•						1	1	1			1			-			• _	-	-			-	-	: ر	- I			1
ļ.	÷ .	<b>.</b> . '	Dove	وسميد	d +	, ر	~ e lle	. ~	10		้าง	Towns	C	ບຸ່	u sed	-tre	free	LINC	m-tra	~ ~	المعد	÷.	( u (		r 4	•		-	i
											•	1						Ś				• -			,				•
 †		-	alı Dave 17	5~ ~ ~	the state	ipe sse tre	n d	ام . لوست	tre for	•tu wrd	, for	~ 2" .	cnd	، د د ه به ۱۰ ۵۴۰۰	nt i i uszed	the	ال مرام مرمولد	ر سر مراد	بدما	on t	se .	near	•••		' -  focu	6106	مەر 1	<b></b>	1
 + :	•	-	alı Dave 17 c'	5~ &~ ~ 10 64.6	tre the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second s	iperse du tu	stred Stred		+4 61	o tru nard	e foi	~ 2" 、	cnd	(clova 084n	nte L i ingedi	tre tre	ت مرام ه مع له	رسی ام ا		on t		noge	•^		-  focu	sing ,	<b>مە</b> - 1	n <b>e</b> r	
         	•	-	alı Dave 17 cl	20 هم مع اه معده لي يي	the sead to to	tu tu	stred Stred E	leun leun	he for 	o true nard treed	, for	~ 2" \       <u>11</u> e.se	cnd cnd	د ده به ۱۹۹۵ ک ۱۹۹۰ ک	d for	the the using	E ~ fr a~gla	(سر). المانية المانية المانية		on 4		nage	•••	<b>ha</b> '	-  freu	\$ ing	ран — С ( ) ) 	n <b>ə</b> r	
                                   	-	-	alı Dave 17 cl	<b>3</b> م هم مع اه بعده این چ س:	the the the the the the the the the the	tu tu tu be	Shree E E Lon (	leur leur	te for 	o true vard toech	e for k - on iterm iorre	the se	cnd cnd rretchd rg cd correc	(clove ophn 	ant is insect a foc	the the <u>wisting</u> the	Erge angle	).   ba 		on 4	بد را فلی	nage  D	0 n . 	+u. 		sins ge <u>t</u>		nər 	
	-	-	alı Dave 17 c)	وم هم مع اه وهده او یو او	the the the the the the the the the the	perse tu tu be por Dove	Stree E E ion ( iven	how how how how how	te for _ces de tr cel	o true truect	, for , on iterm iocra djur	~ 2" , ' <u>44</u> , 50 J gret Jack	cnd cnd rretchd my cd correc on	(clow option 	d for	tu tu tu tu tu tu tu tu tu tu tu tu tu t	Erro ergle <u>werro</u> firet	i be be foren		0 +	44.4	mage  Da	on ne c	+u. 		درمع ه <u>ب</u>		n <b>ə</b> r	- - -
		-	alı Dave 17 c -	مم مع مد مع ان معد ان ان ان ان ا	the sound to the the the the the the the the	perse tu tu be por Dove	Stree E E Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon ( Loon (	here here here here here here here here	te for cas de tr	o true nard tweet stree	e for djue	+ mm	cnd cnd metchd my cd correc on osurt	(clove option option Atuan Atuan Atuan	and for cel for h on fur ca	tu tu tu tu tu tu	E ~fr a~gle frat	be be be			444		ne c	+u. 		sins 8-4	مد - ۲۰ : جمد :	nər 	
			alı Deve 17 	مم مد مع ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل ان تعمل الما الما الما الما الما الما الما ا	the the the the the the the the the the	tu tu tu porte port Dove	LI d Stree E. Loon ( Loon ( Loon (	here here here here here here here here	the for de the the sussist	e tur unard treed streed	, foi k brin brin djue	- ( cl	cond cond may ca connection on or onthe connection	(clove option  Aread Area	d for fursed	the the the the free connege	E ~ fr a ~ gle frat			in t	tor,	nage 	on we c	+u - 		sins 8-±	بعد - ۲۰ 		
			ali Deve 17 c'  Deve	<b>3</b> م ده مه ده ده د د د د د د د	the the the the the the the the the the	pesse tu tu be po' Dave	Li d Shred E. Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon ( Lon	and we have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have and have	te for ees te tu tu	etue tuere tue tue tu tu tu tu tu tu	s for k become orre djue		cnd cnd my ca correc on osurt here	(clove option <u>prine</u> true true true		44 44 44 44 44 44 44 44 44 44 44 44 44	Enfo engle first muny zever			1 20	ter ,		00 			<i>د</i> ، در در در در در در در در در در در در در	<b>مد - ۲</b> ۰ ۲۰ ۳۰ ۲۰		
		- 	ali Dave 17  Dave	<b>3</b> ~ 4 ∩ ∧ 40 1 ∪ (1 ∪ (1 ∪ (1 ∪ (1 ∪ (1 ∪ (1 ∪ (1 ∪	the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second secon	tu tu tu be por Dore Set	E Shree E h h con ( iven iven iven iven	apour leans tron leans tron	te for cession two cussion	etued treed street street	s for k - on between - - - - - - - - - - - - -	the should have	cod cod my ca correc on out	(clove opthn <u>y</u> minud tru tru tru tru tru tru tru tru tru tru	and he we	the the the the the the the the the the	E ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	be for for hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores hores h		1 can 1 can 1 can 1 can	ter,		••• •••• •••••	+u 		\$1~5 84.4			
			ali Dave 47 cl	<b>3</b> م ده مه ده بعده لاچ سیا سوه ب	the seal	perse tu tu be por Sore	Li d Shred E La h La h La h La h La h La h La h La h	here here here here here here here here	the for the case	etui uerd beed street street	s for & - on between        -	the se	overtee cnd metchild overtee on overte have	(clove ophn <u>y</u> on trud tru tru tru tru tru tru tru tru tru tru	a wa a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a na b a n b a n b a n b a n b a n b a n b a n b a n b a n b a n b a n b a n b a n a n	44 44 44 44 44 64 64 64 74 74 75 75 70 70 70 70 70	Eropangle brengle first response			1 cen 1 cen 1 cen 1 cen	4 or ,		on . 			5 ing 95-2	•••• C		
	, - 		alı Dave 47  Dave 	5~ an na lo sast (lo sast lo sast 	the seat	perse tu tu be por Dove Sort	11 d 34re- E 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 ( 1000 (	in in in in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its in its	the for de cas	street	, for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for ,	turny ~ ~ 2", deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres deres der	overtectual cond restances contector overtectual have a south have a south have	(clow ophn 	and here a	44 44 44 44 44 44 64 64 64 64 64 64 10 70 70 70	Eropa angle first score			1 can 	tor ,		va c			sins 9.4	•••• • • • • • • • • • • • • • • • • •	~~ 	
	-		alı Dave 47 Cl  Dave Hesk a	5~ en ne is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb is carb i	the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of the sealer of th	perse tu tu perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse p	E Shred E L L L L L L L L L L L L L L L L L L	and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and here and	the for	stured streed streed streed	, for k - or orra djuc 	the share	overee cod wetebut or correc or or or or the the the the the the the the	(clow option <u>option</u> <u>option</u> <u>option</u> <u>option</u> <u>option</u> <u>option</u> <u>option</u>	a ve	44 44 44 44 44 44 44 44 44 44 44 44 44	E ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	be for for here		1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1 cen 1	the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contraction of the contracti		••• ••• •••			sins g.t	•••• 61		
	, 		alı Dave 47 cl  Dave Hesk a Dave	5~ an no is cart (is cart (is cart (is cart (is cart (is cart (is cart) (is cart (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart) (is cart	the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second secon	perse tu tu be perse pore Sort	E Shree E - - - - - - - - - - - - -	in a government of the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second	the for	trued	, for , for , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or , or	turny ~ ~ 2", fte.se get trans - (cl trans should have	overee cnd reteled or correc on overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe over over	(clove opthon y one when the the the the the the the the the the	a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we a we	44 44 44 44 44 44 44 44 44 44 44 44 44	E ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~			can 	delug		en i ve c tream			suns 96-4			
			alı Due 17 cl  Dave Hesk a Dave	۲۰ ۵۰ مده ۵۰ مده ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰	the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the seal of the se	perse tu tu tu perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse perse pers	E Sheet E The Sheet Con ( Sheet E conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Conde Co	in the set	the for case	e tui tured street constant	, for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for ,	turny ~ ~ 2" , descent to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a sector to a se	overee cod meteled over ee on overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe overe over over	(clove opthon 	a ve a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b a ve b	44 44 44 44 44 44 44 44 44 44 44 44 44	E ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	be be be be be be be be be be be be be b	· · · · · · · · · · · · · · · · · · ·	00 + 100 + 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - -	tor , , , , , , , , , , , , , , , , , , ,		en i			s.ns 96-4	••••••••••••••••••••••••••••••••••••••		
			ali Dave 17 Cl Dave Dave Dave	مد مع مع الم المع المع المع المع المع المع المع ا	the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second secon	perse tu tu tu be for Dave so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so + so	sheet E a h ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ( ton ())))))))))))))))))))))))))))))))))))	Action of the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second se	the for all the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the case of the ca	o true true de serve s ac s ac s ac s ac s ac s ac s ac s ac	, for a become become a become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become become b	he he	overteel cond meterete on overte on overte have a have a have a have a have a have a have a have a have a have a have a have a have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have b have have b have have b have have b have have have have have have have have	(clove option 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 2 0000 2 0000 2 000 2 000 2 000 2 000 2 000 2 000 2 000 20	the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of	44 44 44 44 44 44 44 44 44 44 44 44 44	Engle angle first surg versor 11+	be be be be be be be be be be be be be b		on +	tor		on ves c tream tre			sins 94.4			
			Ali Dave 17 Cl  Dave Hesk a Dave Dave	۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰ ۲۰	Are seed b criticate from the from the criticate the the the the the the the t	perse tu tu tu perse pore pore soit soit soit soit soit soit		and the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second s	the for de de de de de de de de de de de de de d	e true true de sources sources to a sources to a sources to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a to a t	, for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for , for ,	the se	overeer cond weeter on content overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer overeer oveer overeer oveer oveer overeer overeer oveer oveer oveer ove	(clove ophn <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u>_</u> <u></u>		44 44 44 44 44 44 44 44 44 44 44 44 44	E ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	be be be be be be be be be be		100 + 100 + 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 -	4444		en ve c tream			5			

	а о о • • •		· · · · · · · · · · · · · · · · · · ·			•		· · · · · · · · · · · · · · · · · · ·		1.
-	- Still and	ee=12-7 -	~ 2.3.2.5	OM FWHM		**************************************		╸ <mark>╎╴╴╶╶╌╸┆╶╴╴</mark> ╎╺╴╸╎╼╸╸╎╼╸╴		
			4 gette	) ,~ 1'8.1 d v	m FWHM Fre	ູ ພາ	•			1
,		first para	· the st	المتعاديس معمدوه	ser band- at	-t- ~11 2m	1			
	1 1 1						* i *			-, -
:	formy	in of a costed	Dove	could june	Leave the stre	the graty whe	e he had m	over it lent	. week	
	'	werked .		Dere (rox	to more to	grating back	in order to	Not alip_ and _b	<u>م ماراد الم</u>	
		here the	been 1	better contered			; ; ;			
-1	د (اور میه	~	tet	Dave glaud	the elizan		from of the		·	
	ا د میر رب	ud-ral	الا د اسدا	e track the	e vertel c	ejuit ~+ !=(	the great and	estrect		-
		collad el	in_ look	et the beck	reliecton wh	the disant	tool is neer	+++ ++++++++++++++++++++++++++++++++++	~~~~	
		l·	- <u></u>	you put to	* teo 1.5 ti	int of the focu	دیم ، <u>محمد معم</u> ر - ا	بدميم الم		
	cit one	point when	distry it	ne strétern. I	bove the Town	ants cas mode	un junion and	ind on Day	·e '	-
, i					' (		·	:	1	-
	*	•						:	1 .	1
	Dave rec Seco	pleded the	ta	inter - unoc	'dhe the IEP.	The an close of	A The freek a	l Horrar that alur	eats the	
a simulation of	Dave sco Seco	pleded the nd pass in floor		88 - 002	alter the IEP		A _7	i viorrar that alur	e abr the	
	Deve rea Seco -+ Ring anon	pleded the ad pass by fr/w and the	G 05	2014 ~110( _ 8 \$ - 002	after the IEP				¢ • ← ↓ ↓	
1	Dave (cr Seco  Ring anon	plecked the ad pess by P(1) and the	G 05	2014 ~110[ 3\$ -002 40 - 002	after the JED					-
, , , , , , , , , , ,	Deve (cr seco  Ring anon	plefed the -d pess -d p(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -(x) -	G 05	2014 ~10[ 8\$ -002 40 - 002	after the IED	or the seed			e a fa y fa	
	Dave red Seco Ring anon	ple fe d' the -d, pess -, p(2) -, -, -, -, -, -, -, -, -, -, -, -, -, -	G 05	2014 ~10C	offer the IEP	or the seed				
	Deve (cr seco  Ring anon	plecked the ad pess by P/20 	G 05	1014 ~		or to seed				
	Deve (cr Seco Ring anon	ple fe of the ad pess by p/20 	G 05	2014 ~10C						
	Deve (cr Seco Ring anon	ple fe ol ' the ad , p css by p/100 	G 05	2014 ~10C						
	Deve (cr Seco Ring anon	ple fe of the of pess y p/ys 	G 05	2014 ~						
	Dave (cd 5600 Ring 2007	ple fa d' the ad p ess y p (1) 	G 05	2014 ~10( 3\$ -002						
	Dave (cd 5600 Ring and 	ple fa d' the ad pess y p(1) 		2014 ~o( 35002 						
	Dave (cd 5600 Ring and 	ple fa d' the ad p ess y p(1) 		2014 ~o( 35						

CHAPTER B	
Appendix B	

In this section, we will present the evaluation of the derivatives  $\frac{\partial n(\lambda)}{\partial \lambda}$  and  $\frac{\partial^2 n(\lambda)}{\partial \lambda^2}$  for a variety of systems employed in the calculations presented in different chapters. These systems include benzene, a representative glass (BK7), air, and  $\beta$ -barium borate, BBO (at  $\theta = 0^{\circ}, 90^{\circ}$  and  $\theta \neq 0^{\circ}, 90^{\circ}$ ).

### **B.1** Dispersion of Benzene

The dispersion curve for benzene follows the model reported by Moutzouris et al: [428]

$$n(\lambda) = \sqrt{2.170184597 + 0.00059399\lambda^2 + \frac{0.02303464}{\lambda^2} - \frac{0.000499485}{\lambda^4} + \frac{0.000178796}{\lambda^6}}$$
(B.1.1)

The wavelengths from which this dispersion has been determined range from 450 nm to 1550nm. From Eq. B.1.1, the first derivative with respect to  $\lambda$  necessary to calculate the group velocity can be found:

$$\frac{\partial}{\partial\lambda}n(\lambda) = \frac{1}{2}\frac{1}{n(\lambda)}\left(0.00118798\lambda - \frac{0.04606928}{\lambda^3} + \frac{0.00199794}{\lambda^5} - \frac{0.001072776}{\lambda^7}\right)$$
(B.1.2)

where  $n(\lambda)$  is as calculated from Eq. B.1.1. The derivative of Eq. B.1.2 yields the second-order derivative necessary for calculating the group velocity dispersion:

$$\frac{\partial^2}{\partial\lambda^2}n(\lambda) = \frac{1}{2}\frac{1}{n(\lambda)} \left( 0.00118798 + \frac{0.13820784}{\lambda^4} - \frac{0.0099897}{\lambda^6} + \frac{0.007509432}{\lambda^8} \right) - \frac{1}{n(\lambda)} \left(\frac{\partial}{\partial\lambda}n(\lambda)\right)^2 \quad (B.1.3)$$

where  $n(\lambda)$  is again calculated according to Eq. B.1.1, and  $\partial/\partial \lambda n(\lambda)$  is calculated using Eq. B.1.2.

## B.2 Dispersion of BK7 Glass

When the dispersion information of glass is required, BK7 is used as a representative glass. The dispersion is determined according to the dispersion reported by Schott.<sup>i</sup> The index of refraction can be calculated:

$$n(\lambda) = \sqrt{1 + \frac{1.03961212\lambda^2}{\lambda^2 - 0.00600069867} + \frac{0.231792344\lambda^2}{\lambda^2 - 0.0200179144} + \frac{1.01046945\lambda^2}{\lambda^2 - 103.560653}}$$
(B.2.1)

The first derivative of  $n(\lambda)$  is then:

$$\frac{\partial}{\partial\lambda}n(\lambda) = \frac{1}{2}\frac{1}{n(\lambda)} \left(\frac{2.07922424\lambda}{\lambda^2 - 0.00600069867} - \frac{2.07922424\lambda^3}{(\lambda^2 - 0.00600069867)^2} + \frac{0.463584688\lambda}{\lambda^2 - 0.0200179144} - \frac{0.463584688\lambda^3}{(\lambda^2 - 0.0200179144)^2} + \frac{2.0209389\lambda}{\lambda^2 - 103.560653} - \frac{2.0209389\lambda^3}{(\lambda^2 - 103.560653)^2}\right)$$
(B.2.2)

<sup>i</sup>The documentation and formula are available at refractiveindex.info.

where  $n(\lambda)$  is calculated according to Eq. B.2.1. The second derivative can be evaluated from Eq. B.2.2:

$$\frac{\partial^2}{\partial\lambda^2}n(\lambda) = \frac{1}{n(\lambda)} \left[ \frac{1}{2} \left( \frac{2.07922424}{\lambda^2 - 0.00600069867} - \frac{10.3961212\lambda^2}{(\lambda^2 - 0.00600069867)^2} + \frac{8.31689696\lambda^4}{(\lambda^2 - 0.00600069867)^3} \right. \\ \left. + \frac{0.463584688}{\lambda^2 - 0.0200179144} - \frac{2.31792344\lambda^2}{(\lambda^2 - 0.0200179144)^2} + \frac{1.854338752\lambda^4}{(\lambda^2 - 0.0200179144)^3} \right]$$
(B.2.3)  
$$\left. + \frac{2.0209389}{\lambda^2 - 103.560653} - \frac{10.1046945\lambda^2}{(\lambda^2 - 103.560653)^2} + \frac{8.0837556\lambda^4}{(\lambda^2 - 103.560653)^3} \right) - \left( \frac{\partial}{\partial\lambda} n(\lambda) \right)^2 \right]$$

where  $n(\lambda)$  is again determined by Eq. B.2.1, and  $\partial/\partial \lambda n(\lambda)$  is calculated with Eq. B.2.2.

## **B.3** Dispersion of Air

The dispersion for air is derived from Ciddor. [429] The index of refraction is calculated:

$$n(\lambda) = 1 + \frac{0.05792105}{238.0185 - \lambda^{-2}} + \frac{0.00167917}{57.362 - \lambda^{-2}}$$
(B.3.1)

While in calculations we will generally neglect the dispersion from air, the group velocity and group velocity dispersion are usually still made available, using the first and second derivatives of the index:

$$\frac{\partial}{\partial\lambda}n(\lambda) = -\frac{0.1158421}{\lambda^3(238.0185 - \lambda^{-2})^2} - \frac{0.00335834}{\lambda^3(57.362 - \lambda^{-2})^2} \tag{B.3.2}$$

$$\frac{\partial^2}{\partial\lambda^2}n(\lambda) = \frac{0.3475263}{\lambda^4(238.0185 - \lambda^{-2})^2} + \frac{0.4633684}{\lambda^6(238.0185 - \lambda^{-2})^3} + \frac{0.01007502}{\lambda^4(57.362 - \lambda^{-2})^2} + \frac{0.01343336}{\lambda^6(57.362 - \lambda^{-2})^3}$$
(B.3.3)

#### B.4 Dispersion of β-Barium Borate

The final material system that will be discussed is BBO. This is also the most difficult, as it is necessary to consider the dispersion of the ordinary index of refraction, the extraordinary index of refraction, and the effective index of refraction when the phase-matching angle is neither  $0^{\circ}$  nor  $90^{\circ}$ . Further, there are multiple sources for dispersion formulae with subtle differences. The first two sources - Eimerl et al[287] and Kato[288] - appear to be the most established; both use the same general formula for the dispersion of BBO:

$$n(\lambda) = \sqrt{A + \frac{B}{\lambda^2 + C} + D\lambda^2}$$
(B.4.1)

The coefficients for an ordinary beam are A = 2.7405, B = 0.0184  $\mu$ m<sup>2</sup>, C = -0.0179  $\mu$ m<sup>2</sup>, and D = -0.0155  $\mu$ m<sup>-2</sup> according to Eimerl et al and A = 2.7359, B = 0.01878  $\mu$ m<sup>2</sup>, C = - 0.01822  $\mu$ m<sup>2</sup>, D = -0.01354  $\mu$ m<sup>-2</sup>

according to Kato, while those for an extraordinary beam are A = 2.3730,  $B = 0.0128 \ \mu m^2$ ,  $C = -0.0156 \ \mu m^2$ , and  $D = -0.0044 \ \mu m^{-2}$  according to Eimerl et al and A = 2.3753,  $B = 0.01224 \ \mu m^2$ ,  $C = -0.01667 \ \mu m^2$ , and  $D = -0.01516 \ \mu m^{-2}$  according to Kato.[287, 288] Unfortunately, these coefficients were only determined based upon characterization of BBO out to 1 $\mu$ m. However, Eimerl et al are among those that report absorption by BBO beginning around 2  $\mu$ m;[106, 208, 287] Zhang et al[208] indicate that this absorption arises from phonon modes and causes corresponding changes to the index of refraction at these longer wavelengths. While Lu and Liu indicate the formulae of Eimerl et al and Kato are reasonably consistent with sumfrequency generation involving wavelengths in this region,[289] Zhang et al modeled the behavior of NIR optical parametric processes in order to expand the model for the refractive index to incorporate higher-order terms accounting for the phonon modes.[208] The result is a model of the form:

$$n = \sqrt{A + \frac{B}{\lambda^2 + C} + D\lambda^2 + E\lambda^4 + F\lambda^6}$$
(B.4.2)

In modeling the parametric performance of an optical parametric generator/optical parametric amplifier system, Zhang et al found the following coefficients to be appropriate: for the ordinary refractive index, A = 2.7359, B = 0.01878  $\mu$ m<sup>2</sup>, C = - 0.01822  $\mu$ m<sup>2</sup>, D = - 0.01471  $\mu$ m<sup>-2</sup>, E = 0.0006081  $\mu$ m<sup>-4</sup>, and F = -0.00006740  $\mu$ m<sup>-6</sup>; and, for the extraordinary refractive index, A = 2.3753, B = 0.01224  $\mu$ m<sup>2</sup>, C = -0.01667  $\mu$ m<sup>2</sup>, D = -0.01627  $\mu$ m<sup>-2</sup>, E = 0.0005716  $\mu$ m<sup>-4</sup>, and F = -0.00006305  $\mu$ m<sup>-6</sup>.[208]

The derivation of the form in Eq. B.4.2 from the original dispersion formula, Eq. B.4.1, fortunately causes the derivatives of Eq. B.4.1 and Eq. B.4.2 to be very similar. The first derivative of Eq. B.4.1 is:

$$\frac{\partial}{\partial\lambda}n(\lambda) = \frac{\lambda}{n(\lambda)} \left[ D - \frac{B}{(\lambda^2 + C)^2} \right]$$
(B.4.3)

while the first derivative of Eq. B.4.2 is:

$$\frac{\partial}{\partial\lambda}n(\lambda) = \frac{\lambda}{n(\lambda)} \left[ -\frac{B}{(\lambda^2 + C)^2} + D + 2E\lambda^2 + 3F\lambda^4 \right]$$
(B.4.4)

Extending these to the second derivatives yields:

$$\frac{\partial^2}{\partial\lambda^2}n(\lambda) = \frac{1}{n(\lambda)} \left[ \frac{4B\lambda^2}{(\lambda^2 + C)^3} - \left(\frac{\partial}{\partial\lambda}n(\lambda)\right)^2 \right] + \frac{1}{\lambda}\frac{\partial}{\partial\lambda}n(\lambda)$$
(B.4.5)

for the model employed by Kato and Eimerl et al (Eq. B.4.1), or:

$$\frac{\partial^2}{\partial\lambda^2}n(\lambda) = \frac{1}{n(\lambda)} \left[ 4\lambda^2 \left( \frac{B}{(\lambda^2 + C)^3} + E + 3F\lambda^2 \right) - \left( \frac{\partial}{\partial\lambda}n(\lambda) \right)^2 \right] + \frac{1}{\lambda} \frac{\partial}{\partial\lambda}n(\lambda)$$
(B.4.6)

for the extended model of Zhang et al (Eq. B.4.2).

However, the derivatives  $\partial/\partial \lambda n(\lambda)$  and  $\partial^2/\partial \lambda^2 n(\lambda)$  in Eq. B.4.3 and Eq. B.4.5 or Eq. B.4.4 and Eq. B.4.6 only hold for ordinary or extraordinary polarizations within the crystal, namely when the phase-matching angle,  $\theta$ , is either 0° or 90°. When the phase-matching angle is at neither 0° nor 90°, the effective index of refraction,  $n_{\theta}(\lambda)$ , is determined based upon the phase-matching angle and the ordinary and extraordinary dispersion curves,  $n_o(\lambda)$  and  $n_e(\lambda)$ , respectively. The functional dependence of  $n_{\theta}(\lambda)$  upon these quantities is:[106]

$$n_{\theta}(\lambda) = n_0(\lambda) \sqrt{\frac{1 + \tan^2(\theta)}{1 + [n_o^2(\lambda)/n_e^2(\lambda)] \tan^2(\theta)}}$$
(B.4.7)

Therefore, when we evaluate the derivative, we end up with a much more complicated expression. While the proper value of  $\theta$  will be dependent upon the wavelengths, it will be constant for a given experimental condition; therefore, we can treat it as a constant and only consider the evaluation of the derivatives based upon the dispersion curves of  $n_o(\lambda)$  and  $n_e(\lambda)$ . Evaluating the first derivative yields:

$$\frac{\partial}{\partial\lambda}n_{\theta}(\lambda) = \frac{n_{\theta}(\lambda)}{n_{o}(\lambda)}\frac{\partial}{\partial\lambda}n_{o}(\lambda) + \frac{n_{\theta}^{3}(\lambda)\tan^{2}(\theta)}{n_{o}(\lambda)n_{e}^{2}(\lambda)[1+\tan^{2}(\theta)]} \left[\frac{n_{o}(\lambda)}{n_{e}(\lambda)}\frac{\partial}{\partial\lambda}n_{e}(\lambda) - \frac{\partial}{\partial\lambda}n_{o}(\lambda)\right]$$
(B.4.8)

where  $\partial/\partial\lambda n_o(\lambda)$  and  $\partial/\partial\lambda n_e(\lambda)$  are derivatives evaluated according to Eq. B.4.3 or Eq. B.4.4 with the appropriate coefficients for the ordinary index  $(\partial/\partial\lambda n_o(\lambda))$  or for the extraordinary index  $(\partial/\partial\lambda n_e(\lambda))$ . We can continue with the result of Eq. B.4.8 to also determine the second derivative of  $n_{\theta}(\lambda)$  with respect to  $\lambda$ :

$$\frac{\partial^{2}}{\partial\lambda^{2}}n_{\theta}(\lambda) = \left[\frac{1}{n_{o}(\lambda)} - \frac{\tan^{2}(\theta)}{1 + \tan^{2}(\theta)} \frac{3n_{\theta}^{2}(\lambda)}{n_{o}(\lambda)n_{e}^{2}(\lambda)}\right] \left(\frac{\partial}{\partial\lambda}n_{o}(\lambda)\right) \left(\frac{\partial}{\partial\lambda}n_{\theta}(\lambda)\right) \\
+ \left[\frac{\tan^{2}(\theta)}{1 + \tan^{2}(\theta)} \frac{n_{\theta}^{3}(\lambda)}{n_{o}^{2}(\lambda)n_{e}^{2}(\lambda)} - \frac{n_{\theta}(\lambda)}{n_{o}^{2}(\lambda)}\right] \left(\frac{\partial}{\partial\lambda}n_{o}(\lambda)\right)^{2} \\
+ \left[\frac{n_{\theta}(\lambda)}{n_{o}(\lambda)} - \frac{\tan^{2}(\theta)}{1 + \tan^{2}(\theta)} \frac{n_{\theta}^{3}(\lambda)}{n_{o}(\lambda)n_{e}^{2}(\lambda)}\right] \left(\frac{\partial^{2}}{\partial\lambda^{2}}n_{o}(\lambda)\right) \\
+ \frac{\tan^{2}(\theta)}{1 + \tan^{2}(\theta)} \left[\frac{2n_{\theta}^{3}(\lambda)}{n_{o}(\lambda)n_{e}^{3}(\lambda)} \left(\frac{\partial}{\partial\lambda}n_{o}(\lambda)\right) \left(\frac{\partial}{\partial\lambda}n_{e}(\lambda)\right) \\
+ \frac{3n_{\theta}^{2}(\lambda)}{n_{e}^{3}(\lambda)} \left(\frac{\partial}{\partial\lambda}n_{e}(\lambda)\right) \left(\frac{\partial}{\partial\lambda}n_{\theta}(\lambda)\right) \\
- \frac{3n_{\theta}^{3}(\lambda)}{n_{e}^{4}(\lambda)} \left(\frac{\partial}{\partial\lambda}n_{e}(\lambda)\right)^{2} + \frac{n_{\theta}^{3}(\lambda)}{n_{e}^{3}(\lambda)} \left(\frac{\partial^{2}}{\partial\lambda^{2}}n_{e}(\lambda)\right)\right]$$
(B.4.9)

In Eq. B.4.9, the partial derivatives  $\partial/\partial \lambda n_o(\lambda)$  and  $\partial/\partial \lambda n_e(\lambda)$  are calculated as in Eq. B.4.8. The partial derivative of  $n_{\theta}(\lambda)$ ,  $\partial/\partial \lambda n_{\theta}(\lambda)$ , can be calculated using the result of Eq. B.4.8. Finally,  $\partial^2/\partial \lambda^2 n_o(\lambda)$  and  $\partial^2/\partial \lambda^2 n_e(\lambda)$  are the second derivatives of the ordinary and extraordinary indices, respectively, with respect to  $\lambda$ ; these may be calculated by Eq. B.4.5 or B.4.6.

We will note that the results of Eq. B.4.8 and B.4.9 are general for any material for which the index at the phase-matching angle,  $\theta$ , is determined as in Eq. B.4.7. As this relationship holds for any uniaxial crystal, the results of Eq. B.4.8 and Eq. B.4.9 can be applied to any such material. However, the appropriate forms of the necessary derivatives -  $\partial^n/\partial\lambda^n n_\alpha(\lambda)$ ,  $n \in \{1, 2\}$  and  $\alpha \in \{o, e\}$  - would need to be determined. While there are other crystals - such as lithium niobate[106] - for which dispersion relations of the same form as Eq. B.4.1 or B.4.2 are used, many other common crystals adopt other models for calculating the refractive index.

CHAPTEI	a c
Appendiz	c C

In this appendix, we present the code developed for the calculations in Ch. 4. This code has been developed in Python for operation on MacOS and other UNIX-based systems. There are some aspects of operation, such as the implementation of multiprocessing, that would likely make this code unsuitable for operation directly on Windows-based systems. The code consists of ten individual scripts, each containing a subset of the functions and classes. These scripts have been reproduced in the sections that follow, with some modifications for presentation.

Basic operation of this script requires several packages to be loaded, then a sequence of functions from the other scripts to be executed. This sequence is outlined in the function **basic\_operation()** below, which would generate a list **fields** that may be manipulated further. The list of "developed packages" in the preamble are the ten scripts mentioned above. These are presented in the subsequent sections of this appendix.

# Native python packages import numpy as np import matplotlib.pyplot as plt import scipy.interpolate as inter import multiprocessing, threading, os, itertools # Developed python packages import parameters, efield, algorithm, controls import user\_settings as user import material\_params as mat import gain\_formulae as gain import plotting\_routines as pltr import phasematching as pm import nl\_crystal as nl

def basic\_operation():

- # Initialize user specifications and parameters
- u = user.user\_specs()
- p = parameters.parameters(u)

```
# Modify the parameters appropriately
controls.modify_params(p)
# Initialize the fields
fields = controls.initialize(p)
# Save and plot the fields as appropriate
if p.u.save:
        controls.save_output(p, fields, 'initial')
elif p.u.show:
        pltr.plot2d(fields, p, 'initial_fields')
# Perform the numerical integration
fields = controls.drive_algorithm(p, fields)
# Save and plot the resulting fields as appropriate
if p.u.save:
        controls.save_output(p, fields, 'final')
elif p.u.show:
        pltr.plot2d(fields, p, 'final_fields')
```

# C.1 user\_settings.py

The script user\_settings.py contains the definitions of two classes that provide parameters for the operation of different parts of the script. The class gain\_selections defines the options available to the user for enabling and disabling the different contributions to the gain functions. The class user\_specs outlines the various settings that define the details of the simulation to be performed.

# Native python packages import numpy as np import socket import os # Developed python packages # (none)

```
def __init__(self):
                # Options for calculations within the
                # slowly varying amplitude approximation:
                self.sva_group_velocity = True
                self.sva_GVD = True
                self.sva_absorption = True
                self.sva_amplification = True
                self.sva_in_lab_frame = True
                # Options for second order calculations
                # (no longer maintained)
                self.so_space_time_coupling = True
                self.so_gvm = True
                self.so_gvd = True
                self.so_gvm_sq = True
                self.so_absorption = True
                self.so_amplification = True
class user_specs:
```

def \_\_init\_\_(self):

```
# Birefringent Materials:
```

# 'BBO (Kato)', 'BBO (Eimerl)', 'LiNbO3'

# Isotropic Materials:

# <none>

self.phi = 0. # for a birefringent material

# the azimuthal angle in deg

# The following define the process that should be simulated: self.parametric\_process = 'parametric amplification' # valid options are: # 'self-phase modulation': uses field 1 # 'second-harmonic generation': uses fields 1 and 2 # 'parametric amplification': fields 1, 2, and 3 # 'sum-frequency generation': fields 1, 2, and 3 self.process = 'Type II' # For parametric amplification, specify the process. # Currently only negative uniaxial crystals are # supported, so options are: # 'Type I' # 'Type II' # 'Type III' # This is over-ridden for SHG. # The following define the properties of the fields # Vacuum wavelengths:

> > '3' : 0.}

# field wavelengths (nm)

```
self.energy_conservation = 20.
```

# tolerance in testing energy conservation

# 10 tolerates ~0.05nm in error from MIR idler

# 80 tolerates ~0.05nm in error from a 800nm pump

# 10 tolerates ~0.05nm for SHG

```
# Initial field conditions:
self.energy = {
                      '1': 0.,
                       '2': 0.,
                       '3' : 0.}
       # field energies (in uJ)
self.e2_size = {
                      '1': 0.,
                       '2': 0.,
                       '3' : 0.}
        # 1/e2 beam size (cm)
self.field_envelopes = {
                         '1' : 'Gaussian',
                                       '2' : 'Gaussian',
                                       '3' : 'Gaussian'}
        # Identifies field source
        # Options are:
               # 'zero':
                                Defines a field that evaluates to
               #
                                O V/m at all points in time
               # 'Gaussian':
                                   Defines a field described
               #
                                        by a Gaussian intensity
                                        envelope
               #
               # 'file': Uses the contents of a file to
               #
                                define the field. load_params
               #
                                must be populated.
self.Gauss_widths = {
                         '1': 0.,
                                       '2': 0.,
                                       '3' : 0.}
```

# width of Gaussians (fs)

```
self.Gauss_centers = { '1' : 0.,
                                                '2': 0.,
                                                '3' : 0.}
                # centers of Gaussians (fs)
        self.load_params = {
                        '1' : ['file', ['re_column', 'im_column']],
                        '2' : ['file', ['re_column', 'im_column']],
                        '3' : ['file', ['re_column', 'im_column']]}
                # identifies file (including subdirectories) to be
                # used as the source for the field. (Untested)
        self.field_phase = {
                                   '1': 0.,
                                        '2': 0.,
                                        '3' : 0.}
                # pulse envelope phase (deg)
        self.loaded_time_scale = 35.
                # When loading fields from file, use this to
                # estimate the characteristic time of the field.
# The following provide optional modifications or settings to the
# calculations and/or operation of the script.
# The standard "main" function will only save if .save is set to True.
# .save_dir identifies the desired directory.
        self.save = True
        self.save_dir = ' '
# The standard "main" function will only show the results if
# .show is set to True.
```

self.show = False

# .silent limits the output to terminal during operation.
 self.silent = False

# .timer controls whether time stamps are printed.

self.timer = True

# .threading controls whether multiprocessing is used. .cores allows
# user-defined number of processes. If .cores is 'Default', N-1
# processes will be initiated, where N is the cores available.
# If .limit\_override = False and .cores is greater than N, 'Default'
# behavior will be used.

self.threading = True
self.cores = 0
self.limit\_override = True
self.radial\_samples = 0

# .adaptive\_steps controls whether the simulation is iterated over a
# fixed number of spatial steps or whether an adaptive algorithm is
# employed to use the fewest steps while ensuring accuracy.
 self.adaptive\_steps = False # NOT YET IMPLEMENTED
 self.steps = 1000

```
self.modify_T_window = False
self.alt_T_window = 800. #fs
```

# The following allow optional modification of derived parameters: self.slowly\_varying\_amp = True

> # apply\_contrast allows suppression of the wings of the field # any points with magnitude less than .contrast will be set to 0. self.apply\_contrast = False self.contrast = 1.e-13

# Modify the phase matching angle

```
self.modify_theta = False
self.alt_theta = 22.2
```

# Modify the effective nonlinearity

```
self.modify_chi2eff = False
self.modify_chi3eff = False
self.alt_chi2eff = 0.
self.alt_chi3eff = 0.
```

# Modify the group velocities

```
self.modify_ug = False
self.alt_ug = {'1' : 0., '2' : 0., '3' : 0.}
```

# These are group velocities and cannot be set to 0.

```
# Modify the time sampling. Default is 512 samples.
# More than ~5k-7.5k is not advised; numpy broadcasting is
# used throughout the calculations, and this becomes less
# efficient around grids of ~10k points.
```

```
self.modify_sampling = True
self.alt_time_samples = 128
```

# Simulate noncollinearity by changes in the effective
# group velocity according to .gamma.

```
self.noncollinear = False
self.gamma = 0.5
```

## C.2 parameters.py

The code in parameters.py defines a class parameters. This class takes the settings defined by the user in an instance of user\_settings.user\_specs (see  $\S$  C.1) and manipulates them: these values are used to define other necessary parameters and modify existing settings.

```
# Native python packages
import numpy as np
import matplotlib.pyplot as plt
import sys
# Non-native python packages
import user_settings as user
import material_params as mat
```

class parameters:

```
def __init__(self, user_settings):
```

else:

```
print('Initializing parameters')
```

```
self.user_settings_conversions()
```

# Initialize parameters describing the material environment

# and phase-matching process

```
if self.u.parametric_process == 'second-harmonic generation' \
    or self.u.parametric_process == 'parametric amplification' \
    or self.u.parametric_process == 'sum-frequency generation':
```

```
self.m = mat.birefringent_params(self.u)
self.mat_type_marker = 'birefringent'
```

```
elif self.u.parametric_process == 'self-phase modulation':
    self.m = mat.isotropic_params(self.u)
    self.mat_type_marker = 'isotropic'
```

else:

```
print('Unsupported process. Ending process.')
exit()
```

```
self.material_conversions()
```

# Identify the time-scales involved in the interacting pulses

```
if 'file' in self.u.field_envelopes:
```

```
self.time_scale = min(self.u.loaded_time_scale, np.min(\
    np.asarray(self.u.Gauss_widths.values())[np.nonzero(\
    self.u.Gauss_widths.values())]))
```

else:

self.time\_scale = np.min(np.asarray(\
 self.u.Gauss\_widths.values())[np.nonzero(\
 self.u.Gauss\_widths.values())])

# Initialize time and frequency windows. Determining the ranges # follows the method described by Gale et al (JOSAB 1998 15(2) 702)

if self.u.modify\_sampling:

self.time\_samples = self.u.alt\_time\_samples

else:

self.time\_samples = 512.

```
if self.u.modify_T_window:
        self.T_window = self.u.alt_T_window*1e-15
else:
        self.T_window = ((self.time_scale)\
                        *np.sqrt(self.time_samples/0.44))
self.time = np.linspace(-self.T_window/2, self.T_window/2, \
                        self.time_samples)
self.freq_samples = self.time_samples
self.freq = np.linspace(-np.pi/abs(self.time[1]-self.time[0]), \
                        np.pi/abs(self.time[1]-self.time[0]),\
                        self.freq_samples + 1)[:-1]
self.freq_window = abs(self.freq[-1] - self.freq[0])
# Describe the TEM mode of each field
self.r_max = np.max(np.asarray(self.u.e2_size.values()))
if self.u.radial_samples == 'Default':
        self.r_samples = 15
else:
        self.r_samples = self.u.radial_samples
self.r = np.linspace(0., 1.25*self.r_max, self.r_samples)
self.fluence = {}
for j in self.u.e2_size:
```

```
uence = {}
n self.u.e2_size:
if self.u.e2_size[j] == 0:
    self.fluence[j] = np.zeros(len(self.r))
else:
    self.fluence[j] = (self.u.energy[j]/((np.pi/2)\
    *((0.5*self.u.e2_size[j])**2)))\
```

```
*np.exp(-2*((self.r/(0.5\
```

```
*self.u.e2_size[j]))**2))
```

# Define initial position of the fields self.z = 0.def n(self, wavelength, field): if self.mat\_type\_marker == 'birefringent': if self.m.polarization[field] == 'extraordinary': return self.m.cry.eff(wavelength, self.m.theta) elif self.m.polarization[field] == 'ordinary': return self.m.cry.ord(wavelength) elif self.mat\_type\_marker == 'isotropic': exit() else: exit()

def user\_settings\_conversions(self):

self.c = 3.E8
 # speed of light, m/s
self.hbar = 1.054571E-34
 # Planck's constant, J\*s
self.u.L \*= 0.01
 # convert material length from cm to m

231

for i in self.u.Gauss\_widths:

```
self.u.Gauss_widths[i]*=1.E-15
```

```
# convert from fs to s
```

```
self.u.Gauss_centers[i]*=1.E-15
```

```
# convert from fs to s
```

```
self.u.e2_size[i]*=0.01
```

# convert from cm to m

self.u.energy[i]\*=1.E-6

# convert from uJ to J

```
self.u.loaded_time_scale*=1.E-15
```

 $\ensuremath{\texttt{\#}}$  convert from fs to s

```
self.cutoff = 0.
if self.u.material == 'BBO (Eimerl)':
        self.cutoff = (2*np.pi*self.c*1.e9)/13290.
elif self.u.material == 'BBO (Kato)':
        self.cutoff = (2*np.pi*self.c*1.e9)/12500.
elif self.u.material == 'BBO (Zhang)':
        self.cutoff = (2*np.pi*self.c*1.e9)/5830.
elif self.u.material == 'LiNbO3':
        self.cutoff = (2*np.pi*self.c*1.e9)/13400.
```

def material\_conversions(self):

```
# convert pm/V to m/V
```

self.m.dk\*=100.

# convert from 1/cm to 1/m

```
def plot_tem(self):
    plt.clf()
    for j in self.fluence:
        plt.plot(self.r, self.fluence[j])
    plt.show()
```

C.3 material\_params.py

We previously referenced material\_params in § C.2. The classes in this script, birefringent\_params and isotropic\_params, are called according to the process being simulated. At this time, it has not been necessary to develop isotropic\_params, but birefringent\_params performs the calculations and manipulations necessary to identify the phase matching angle and effective nonlinearity for the process.

self.cry = nl.NLcrystals()
self.cry.selection = user\_sett.material

```
# populate the local wavelength and field property dictionaries
self.def_mask(user_sett)
```

# redefine the keys to match internal notation

i[j] = i.pop(self.mask[j])

ini\_keys = i.keys()

for j in ini\_keys:
 if j in self.mask:
 pass
 else:

```
i.pop(j,None)
```

```
self.chi3eff = 0.
```

```
if user_sett.parametric_process == 'second-harmonic generation':
    self.shg_pm(user_sett)
elif user_sett.parametric_process == 'parametric amplification' \
    or user_sett.parametric_process == 'sum-frequency generation':
    if user_sett.process == 'Type I':
        self.type_one_pm(user_sett)
    elif user_sett.process == 'Type II':
        self.type_two_pm(user_sett)
```

```
elif user_sett.process == 'Type III':
    self.type_three_pm(user_sett)
```

else:

```
print('Unsupported process requested. Ending script.')
exit()
```

```
def shg_pm(self, u):
        if math.isnan(pm.type_one(self.cry, u.wavelengths['SHG'],\
                        u.wavelengths['Fundamental'])):
                print('No phase matching angle is available.\
                                Setting theta to 90deg.')
                self.theta = 90.
        else:
                self.theta = pm.type_one(self.cry, u.wavelengths['SHG'], \
                                        u.wavelengths['Fundamental'])
        self.ug = {
                'Fundamental' : self.cry.ord_vg(u.wavelengths['Fundamental']),
                'SHG' : self.cry.ex_vg(u.wavelengths['SHG'], self.theta)}
        self.gvd = {
                'Fundamental' : self.cry.ord_gvd(u.wavelengths['Fundamental']),
                'SHG' : self.cry.ex_gvd(u.wavelengths['SHG'], self.theta)}
        self.rho = {
                'Fundamental' : 0,
                'SHG' : (180/np.pi)*np.arctan((((self.cry.ord(\
                                u.wavelengths['SHG']))**2)\
                                /((self.cry.ex(u.wavelengths['SHG']))**2))\
                                *np.tan((np.pi/180)*self.theta)) - self.theta}
        self.dk = pm.type_one_mismatch(self.cry, u.wavelengths['SHG'], \
                        u.wavelengths['Fundamental'], self.theta)
        self.chi2eff = 2*self.cry.deff(self.theta, u.phi, 'Type I')
```

```
def type_one_pm(self, u):
        if math.isnan(pm.type_one(self.cry, u.wavelengths['Pump'],\
                        u.wavelengths['Signal'])):
                print('No phase matching angle is available.\
                                Setting theta to 90deg.')
                self.theta = 90.
        else:
                self.theta = pm.type_one(self.cry, u.wavelengths['Pump'],\
                                        u.wavelengths['Signal'])
        self.ug = {
                'Pump' : self.cry.ex_vg(u.wavelengths['Pump'], self.theta),
                'Signal' : self.cry.ord_vg(u.wavelengths['Signal']),
                'Idler' : self.cry.ord_vg(u.wavelengths['Idler'])}
        self.gvd = {
                'Pump' : self.cry.ex_gvd(u.wavelengths['Pump'], self.theta),
                'Signal' : self.cry.ord_gvd(u.wavelengths['Signal']),
                'Idler' : self.cry.ord_gvd(u.wavelengths['Idler'])}
        if u.noncollinear:
                self.rho = {
                'Pump' : (180/np.pi)*np.arctan((((self.cry.ord(\
                                u.wavelengths['Pump']))**2)\
                                /((self.cry.ex(u.wavelengths['Pump']))**2))\
                                *np.tan((np.pi/180)*self.theta)) - self.theta,
                'Signal' : 0.,
                'Idler' : 0.}
        self.dk = pm.type_one_mismatch(self.cry, u.wavelengths['Pump'],\
                        u.wavelengths['Signal'], self.theta)
        self.chi2eff = 2*self.cry.deff(self.theta, u.phi, 'Type I')
```

```
def type_two_pm(self, u):
        if math.isnan(pm.type_two(self.cry, u.wavelengths['Pump'],\
                        u.wavelengths['Signal'])):
                print('No phase matching angle is available.\
                                Setting theta to 90deg.')
                self.theta = 90.
        else:
                self.theta = pm.type_two(self.cry, u.wavelengths['Pump'],\
                                        u.wavelengths['Signal'])
        self.ug = {
                'Pump' : self.cry.ex_vg(u.wavelengths['Pump'], self.theta),
                'Signal' : self.cry.ord_vg(u.wavelengths['Signal']),
                'Idler' : self.cry.ex_vg(u.wavelengths['Idler'], self.theta)}
        self.gvd = {
                'Pump' : self.cry.ex_gvd(u.wavelengths['Pump'], self.theta),
                'Signal' : self.cry.ord_gvd(u.wavelengths['Signal']),
                'Idler' : self.cry.ex_gvd(u.wavelengths['Idler'], self.theta)}
        if u.noncollinear:
                self.rho = {
                'Pump' : (180/np.pi)*np.arctan((((self.cry.ord(\
                                u.wavelengths['Pump']))**2)/((self.cry.ex(\
                                u.wavelengths['Pump']))**2))*np.tan((np.pi\
                                /180)*self.theta)) - self.theta,
                'Signal' : 0.,
                'Idler' : (180/np.pi)*np.arctan((((self.cry.ord(\
                                u.wavelengths['Idler']))**2)/((self.cry.ex(\
                                u.wavelengths['Idler']))**2))*np.tan((np.pi\
                                /180)*self.theta)) - self.theta}
        self.dk = pm.type_two_mismatch(self.cry, u.wavelengths['Pump'],\
                        u.wavelengths['Signal'], self.theta)
```

```
self.chi2eff = 2*self.cry.deff(self.theta, u.phi, 'Type II')
```

```
def type_three_pm(self, u):
        if math.isnan(pm.type_three(self.cry, u.wavelengths['Pump'],\
                        u.wavelengths['Signal'])):
                print('No phase matching angle is available.\
                                Setting theta to 90deg.')
                self.theta = 90.
        else:
                self.theta = pm.type_three(self.cry, u.wavelengths['Pump'],\
                                        u.wavelengths['Signal'])
        self.ug = {
                'Pump' : self.cry.ex_vg(u.wavelengths['Pump'], self.theta),
                'Signal' : self.cry.ex_vg(u.wavelengths['Signal'], self.theta),
                'Idler' : self.cry.ord(u.wavelengths['Idler'])}
        self.gvd = {
                'Pump' : self.cry.ex_gvd(u.wavelengths['Pump'], self.theta),
                'Signal' : self.cry.ex_gvd(u.wavelengths['Signal'], self.theta),
                'Idler' : self.cry.ord_gvd(u.wavelengths['Idler'])}
        if u.noncollinear:
                self.rho = {
                'Pump' : (180/np.pi)*np.arctan((((self.cry.ord(\
                                u.wavelengths['Pump']))**2)/((self.cry.ex(\
                                u.wavelengths['Pump']))**2))*np.tan((np.pi\
                                /180)*self.theta)) - self.theta,
                'Signal' : (180/np.pi)*np.arctan((((self.cry.ord(\
                                u.wavelengths['Signal']))**2)/((self.cry.ex(\
                                u.wavelengths['Signal']))**2))*np.tan((np.pi\
                                /180)*self.theta)) - self.theta,
                'Idler' : 0.}
```

self.chi2eff = 2\*self.cry.deff(self.theta, u.phi, 'Type III')

def def\_mask(self, u):

```
if u.parametric_process == 'second-harmonic generation':
    if abs(u.wavelengths['1'] - 2*u.wavelengths['2'])\
         *180. < u.energy_conservation \
        or abs(u.wavelengths['2'] - 2*u.wavelengths['1'])\
         *180. < u.energy_conservation:</pre>
```

pass

else:

```
print('The chosen colors exceed the specified tolerance \
    for energy conservation.')
```

exit()

```
if u.wavelengths['1'] < u.wavelengths['2']:</pre>
```

# Field 2 is the fundamental and Field 1 is the SHG

```
self.polarization = {
    'Fundamental' : 'ordinary',
    'SHG' : 'extraordinary'}
```

else:

sys.exit()

if u.wavelengths['1'] == 0 or u.wavelengths['2'] == 0 \
 or u.wavelengths['3'] == 0:

exit()

# Field 1 is the pump

```
pass
```

else:

print('The chosen colors exceed the specified  $\$ 

tolerance for energy conservation.')

```
exit()
```

```
if u.wavelengths['2'] < u.wavelengths['3'] or\
                u.wavelengths['2'] == u.wavelengths['3']:
                # Field 2 is the signal and Field 3 is the idler
                self.mask = {
                         'Pump' : '1',
                         'Signal' : '2',
                         'Idler' : '3'}
        elif u.wavelengths['3'] < u.wavelengths['2']:</pre>
                # Field 3 is the signal and Field 2 is the idler
                self.mask = {
                         'Pump' : '1',
                         'Signal' : '3',
                         'Idler' : '2'}
elif u.wavelengths['2'] < u.wavelengths['1'] and\</pre>
        u.wavelengths['2'] < u.wavelengths['3']:</pre>
        # Field 2 is the pump
        if abs((1E9/u.wavelengths['2'])-(1E9/u.wavelengths['1'])\
                -(1E9/u.wavelengths['3']))<u.energy_conservation:
                pass
        else:
                print('The chosen colors exceed the specified \
                          tolerance for energy conservation.')
                exit()
        if u.wavelengths['1'] < u.wavelengths['3'] or \
```

```
u.wavelengths['1'] == u.wavelengths['3']:
```

```
# Field 1 is the signal and Field 3 is the idler
                self.mask = {
                         'Pump' : '2',
                         'Signal' : '1',
                         'Idler' : '3'}
        elif u.wavelengths['3'] < u.wavelengths['1']:</pre>
                # Field 3 is the signal and Field 1 is the idler
                self.mask = {
                         'Pump' : '2',
                         'Signal' : '3',
                         'Idler' : '1'}
elif u.wavelengths['3'] < u.wavelengths['1'] and\</pre>
         u.wavelengths['3'] < u.wavelengths['2']:</pre>
        # Field 3 is the pump
        if abs((1E9/u.wavelengths['3'])-(1E9/u.wavelengths['2'])\
                  -(1E9/u.wavelengths['1']))<u.energy_conservation:
```

```
pass
```

```
else:
    print('The chosen colors exceed the specified\
        tolerance for energy conservation.')
    exit()
```

# Field 1 is the signal and Field 2 is the idler

```
self.mask = {
                        'Pump' : '3',
                        'Signal' : '1',
                        'Idler' : '2'}
        elif u.wavelengths['2'] < u.wavelengths['1']:</pre>
                # Field 2 is the signal and Field 1 is the idler
                self.mask = {
                        'Pump' : '3',
                        'Signal' : '2',
                        'Idler' : '1'}
if u.process == 'Type I':
        self.polarization = {
                        'Pump' : 'extraordinary',
                        'Signal' : 'ordinary',
                        'Idler' : 'ordinary'}
elif u.process == 'Type II':
        self.polarization = {
                        'Pump' : 'extraordinary',
                        'Signal' : 'ordinary',
                        'Idler' : 'extraordinary'}
elif u.process == 'Type III':
        self.polarization = {
                        'Pump' : 'extraordinary',
```

243
'Signal' : 'extraordinary', 'Idler' : 'ordinary'}

return

class isotropic\_params:

def \_\_init\_\_(self, user\_sett):

print('Sorry, this hasn\'t been developed yet. Exiting...')
exit()

C.4 nl\_crystal.py

In § C.3, the code references two non-native Python scripts: phasematching.py and nl\_crystal.py. Here, we present the latter (nl\_crystal.py), as the class this script contains (NLcrystals) is also referenced by phasematching.py (§ C.5). The class of nl\_crystal.py (NL\_crystals) contains the optical properties of a variety of nonlinear crystals and performs the calculations necessary to determine the index, group velocity, and group velocity dispersion for both ordinary and extraordinary polarizations and the effective nonlinearity.

# Native python packages import numpy as np # Non-native python packages # (none)

class NLcrystals:

```
selection = 'LiNb03'
process = 'Type I'
```

```
def available_crystals(self):
```

```
print('\'LiNbO3\', \'BBO (Kato)\', \'BBO (Eimerl)\',\
```

\'BBO (Zhang)\', \'AgGaS2\', \'AgGaSe2\'')

```
def available_processes(self):
```

```
print('\'Type I\', \'Type II\', \'Type III\'')
```

```
def Selmeier(self, ex_or_ord):
```

```
# LiNbO3 Selmeier coefficients from Lambda Photometrics data
# coefficients from refractiveindex.info did not give correct performance
# format of Selmeier equation for these coefficients consistent with BBO
```

```
if self.selection == 'LiNb03':
    if ex_or_ord == 'extraordinary':
        return [4.582, 0.099169, -0.044432, -0.02195]
    if ex_or_ord == 'ordinary':
        return [4.9048, 0.11768, -0.0475, -0.027169]
```

```
if self.selection == 'BBO (Kato)':
    if ex_or_ord == 'extraordinary':
        return [2.3753, 0.01224, -0.01667, -0.01516]
    elif ex_or_ord == 'ordinary':
        return [2.7359, 0.01878, -0.01822, -0.01354]
```

```
if self.selection == 'BBO (Eimerl)':
    if ex_or_ord == 'extraordinary':
        return [2.3730, 0.0128, -0.0156, -0.0044]
    elif ex_or_ord == 'ordinary':
```

```
return [2.7405, 0.0184, -0.0179, -0.0155]
        if self.selection == 'BBO (Zhang)':
                if ex_or_ord == 'extraordinary':
                        return [2.3753, 0.01224, -0.01667, -0.01627, \
                                                 0.0005716, -0.00006305]
                elif ex_or_ord == 'ordinary':
                        return [2.7359, 0.01878, -0.01822, -0.01471,\
                                                 0.0006081, -0.00006740]
        if self.selection == 'AgGaS2':
                if ex_or_ord == 'extraordinary':
                        return [3.5873, 1.9533, -0.11066, 2.3391, -1030.7]
                elif ex_or_ord == 'ordinary':
                        return [3.3970, 2.3982, -0.09311, 2.1640, -950]
        if self.selection == 'AgGaSe2':
                if ex_or_ord == 'extraordinary':
                        return [3.3132, 3.3616, -(0.38201*0.38201),\
                                                 1.7677, -1600]
                elif ex_or_ord == 'ordinary':
                        return [3.9362, 2.9113, -(0.38821*0.38821),\
                                                1.7954, -1600]
def ord(self, wavelength):
        Sel = self.Selmeier('ordinary')
        l = wavelength/1000.
        if self.selection == 'LiNbO3' or self.selection == 'BBO (Kato)'
```

```
return np.sqrt(Sel[0]+(Sel[1]/(1*1+Sel[2]))+(Sel[3]*1*1))
        if self.selection == 'AgGaS2' or self.selection == 'AgGaSe2':
                return np.sqrt(Sel[0] + ((Sel[1]*l*l)/(l*l+Sel[2])) \
                                + ((Sel[3]*1*1)/(1*1+Sel[4])))
        if self.selection == 'BBO (Zhang)':
                return np.sqrt(Sel[0] + (Sel[1]/(1*1 + Sel[2])) + (Sel[3]*1*1) \
                                + (Sel[4]*1*1*1) + (Sel[5]*1*1*1*1*1))
def ex(self, wavelength):
        Sel = self.Selmeier('extraordinary')
        l = wavelength/1000.
        if self.selection == 'LiNb03' or self.selection == 'BBO (Kato)' \
                                or self.selection == 'BBO (Eimerl)':
                return np.sqrt(Sel[0]+(Sel[1]/(1*1+Sel[2]))+(Sel[3]*1*1))
        if self.selection == 'AgGaS2' or self.selection == 'AgGaSe2':
                return np.sqrt(Sel[0] + ((Sel[1]*l*l)/(1*l+Sel[2])) \
                                + ((Sel[3]*1*1)/(1*1+Sel[4])))
        if self.selection == 'BBO (Zhang)':
                return np.sqrt(Sel[0] + (Sel[1]/(1*1 + Sel[2])) + (Sel[3]*1*1) \
                                + (Sel[4]*1*1*1) + (Sel[5]*1*1*1*1*1))
def eff(self, wavelength, theta):
       no = self.ord(wavelength)
       ne = self.ex(wavelength)
        q = (np.pi/180)*theta
       return no*np.sqrt((1+np.tan(q)*np.tan(q))/(1 \
```

+ ((no\*no)/(ne\*ne))\*np.tan(q)\*np.tan(q)))

```
def ord_vg(self, wavelength):
        if self.selection == 'LiNbO3' or self.selection == 'BBO (Kato)' \
                                or self.selection == 'BBO (Eimerl)' \
                                or self.selection == 'BBO (Zhang)':
                c = 3E10 \ \text{#cm/s}
                n = self.ord(wavelength)
                l = wavelength/1000.
                Sel = self.Selmeier('ordinary')
                if self.selection == 'LiNbO3' or self.selection == 'BBO (Kato)' \
                                        or self.selection == 'BBO (Eimerl)':
                        B = Sel[1]
                        C = Sel[2]
                        D = Sel[3]
                        dndl = (1/n)*(D-(B/((1*1+C)*(1*1+C))))
                elif self.selection == 'BBO (Zhang)':
                        B = Sel[1]
                        C = Sel[2]
                        D = Sel[3]
                        E = Sel[4]
                        F = Sel[5]
                        dndl = (1/n)*(D+2*E*1*1+3*F*1*1*1\
                                         -(B/((l*l+C)*(l*l+C))))
```

```
def ex_vg(self, wavelength, theta):
        if theta == 90 or theta == 0:
                if theta == 90:
                         if self.selection == 'LiNbO3' \setminus
                                          or self.selection == 'BBO (Kato)' \
                                          or self.selection == 'BBO (Eimerl)' \
                                          or self.selection == 'BBO (Zhang)':
                                 c = 3E10 \ \text{#cm/s}
                                 n = self.ex(wavelength)
                                 l = wavelength/1000.
                                 Sel = self.Selmeier('extraordinary')
                                 if self.selection == 'LiNbO3' \setminus
                                          or self.selection == 'BBO (Kato)'\
                                          or self.selection == 'BBO (Eimerl)':
                                          B = Sel[1]
                                          C = Sel[2]
                                          D = Sel[3]
                                          dndl = (1/n)*(D-(B/((1*1+C)*(1*1+C))))
                                 elif self.selection == 'BBO (Zhang)':
                                          B = Sel[1]
                                          C = Sel[2]
                                          D = Sel[3]
                                          E = Sel[4]
```

```
F = Sel[5]
                                dndl = (1/n)*(D + 2*E*1*1 + 3*F*1*1*1*1 \
                                         - (B/((1*1 + C)*(1*1 + C))))
                        return c*(1E-14)*(1/(n-1*dndl))
                                                 # in mm/fs
        elif theta == 0:
                return self.ord_vg(wavelength)
else:
        if self.selection == 'LiNbO3' or self.selection == 'BBO (Kato)' \
                        or self.selection == 'BBO (Eimerl)' \
                        or self.selection == 'BBO (Zhang)':
                q = (np.pi/180)*theta
                c = 3E10 \ \text{#cm/s}
                no = self.ord(wavelength)
                ne = self.ex(wavelength)
                n = no*np.sqrt(((1+np.power(np.tan(q),2))/(1))
                                +np.power(((no/ne)*np.tan(q)),2))))
                l = wavelength/1000
                Sel_ord = self.Selmeier('ordinary')
                Sel_ex = self.Selmeier('extraordinary')
                if self.selection == 'LiNbO3' \
                                or self.selection == 'BBO (Kato)'\
                                or self.selection == 'BBO (Eimerl)':
                        Bo = Sel_ord[1]
                        Co = Sel_ord[2]
```

Do = Sel\_ord[3]

250

```
Be = Sel_ex[1]
Ce = Sel_ex[2]
De = Sel_ex[3]
dnedl = (1/ne)*(De-(Be/((1*1+Ce)*(1*1+Ce))))
dnodl = (1/no)*(Do-(Bo/((1*1+Co)*(1*1+Co))))
```

elif self.selection == 'BBO (Zhang)':

```
Bo = Sel_ord[1]
Co = Sel_ord[2]
Do = Sel_ord[3]
Eo = Sel_ord[4]
Fo = Sel_ord[5]
Be = Sel_ex[1]
Ce = Sel_ex[2]
De = Sel_ex[3]
Ee = Sel_ex[4]
Fe = Sel_ex[5]
```

```
dndl = (n/no)*dnodl \
    + ((n**3)/(no*ne*ne))*(((np.tan(q))**2)/(1\
    +((np.tan(q))**2)))*((dnodl) - ((no*dnedl)/(ne)))
```

def ord\_gvd(self, wavelength):

```
if self.selection == 'LiNbO3' or self.selection == 'BBO (Kato)' \
                        or self.selection == 'BBO (Eimerl)' \
                        or self.selection == 'BBO (Zhang)':
        c = 3E10 \ \text{#cm/s}
       n = self.ord(wavelength)
        l = wavelength/1000.
        Sel = self.Selmeier('ordinary')
        if self.selection == 'LiNbO3' or self.selection == 'BBO (Kato)'
                                or self.selection == 'BBO (Eimerl)':
                B = Sel[1]
                C = Sel[2]
                D = Sel[3]
                dndl = (1/n)*(D-(B/((1*1+C)*(1*1+C))))
                d2nd12 = (dnd1/1) - (((dnd1)**2)/n) 
                                + ((4.*1*1*B)/(n*((1*1+C)**3)))
        elif self.selection == 'BBO (Zhang)':
                B = Sel[1]
                C = Sel[2]
                D = Sel[3]
                E = Sel[4]
                F = Sel[5]
                dndl = (1/n)*(D + 2*E*1*1 + 3*F*1*1*1*1 \
                                - (B/((1*1 + C)*(1*1 + C))))
                d2nd12 = (dnd1/1) - ((dnd1**2)/n) + ((4.*1*1)/n) 
                                *(E + 3.*F*1*1 + (B/((1*1+C)**3)))
```

```
def ex_gvd(self, wavelength, theta):
        if theta == 90 or theta == 0:
                if theta == 0:
                         return self.ord_gvd(self, wavelength)
                if theta == 90:
                         if self.selection == 'LiNbO3' \
                                 or self.selection == 'BBO (Kato)' \
                                 or self.selection == 'BBO (Eimerl)' \
                                 or self.selection == 'BBO (Zhang)':
                                 c = 3E10 \ \text{#cm/s}
                                 n = self.ex(wavelength)
                                 l = wavelength/1000.
                                 Sel = self.Selmeier('extraordinary')
                                 if self.selection == 'LiNbO3' \setminus
                                         or self.selection == 'BBO (Kato)' \
                                         or self.selection == 'BBO (Eimerl)':
                                         B = Sel[1]
                                         C = Sel[2]
                                         D = Sel[3]
                                         dndl = (1/n)*(D-(B/((1*1+C)*(1*1+C))))
                                         d2ndl2 = (dndl/l) - (((dndl)**2)/n) \
                                                  + ((4.*1*1*B)/(n*((1*1+C)**3)))
```

```
return ((l**3)/(2*np.pi*c*c))*d2ndl2*(1.E25)
# in fs2/mm
```

else:

```
if self.selection == 'LiNb03' \
            or self.selection == 'BBO (Kato)'\
            or self.selection == 'BBO (Eimerl)':
            Bo = Sel_ord[1]
            Co = Sel_ord[2]
            Do = Sel_ord[3]
            Be = Sel_ex[1]
            Ce = Sel_ex[2]
            De = Sel_ex[3]
            dnedl = (1/ne)*(De-(Be/((1*1+Ce)*(1*1+Ce))))
            d2nedl2 = (dnedl/1) - (((dnedl)**2)/ne) \
                 + ((4.*1*1*Be)/(ne*((1*1+Ce)**3)))
            dnodl = (1/no)*(Do-(Bo/((1*1+Co)*(1*1+Co))))
            d2nodl2 = (dnodl/1) - (((dnodl)**2)/no) \
```

elif self.selection == 'BBO (Zhang)':

```
Bo = Sel_ord[1]
Co = Sel_ord[2]
Do = Sel_ord[3]
Eo = Sel_ord[4]
Fo = Sel_ord[5]
Be = Sel_ex[1]
Ce = Sel_ex[2]
De = Sel_ex[3]
Ee = Sel_ex[4]
Fe = Sel_ex[5]
```

dnedl = (1/ne)\*(De + 2\*Ee\*1\*1 + 3\*Fe\*1\*1\*1\

dndl = (n/no)\*dnodl + ((n\*\*3)/(no\*ne\*ne))\*(((np.tan(q))\*\*2)/(1.+((np.tan(q))\*\*2)))\ \*((dnodl) - ((no\*dnedl)/(ne))) # Because of the length of the expression, the # calculation of d2ndl2 is broken into pieces  $A = ((1/no) - (((np.tan(q))**2)/(1.+((np.tan(q))**2))) \setminus$ \*((3.\*n\*n)/(no\*ne\*ne)))\*(dnodl)\*(dndl)  $B = ((n/no) - (((np.tan(q))**2)/(1.+((np.tan(q))**2))) \setminus$ \*((n\*n\*n)/(no\*ne\*ne)))\*(d2nodl2) C = -((n/(no\*no)) - (((np.tan(q))\*\*2)/(1))+((np.tan(q))\*\*2)))\*((n\*n\*n)/(no\*no\*ne\*ne)))\ \*(dnodl)\*(dnodl) # D, E, F, and G need to be multiplied # by the factor tan2/1+tan2 D = ((2.\*n\*n\*n)/(no\*ne\*ne\*ne))\*(dnodl)\*(dnedl) E = ((3.\*n\*n)/(ne\*ne\*ne))\*(dnedl)\*(dndl)F = - ((3.\*n\*n\*n)/(ne\*\*4))\*((dned1)\*\*2) G = ((n\*n\*n)/(ne\*ne\*ne))\*(d2ned12) $d2nd12 = A + B + C \setminus$ + ((((np.tan(q))\*\*2)/(1+((np.tan(q))\*\*2)))\ \*(D + E + F + G))

```
return ((1**3)/(2*np.pi*c*c))*d2ndl2*(1.E25)
                                                # in fs2/mm
def deff(self, theta, phi, process):
        if self.selection == 'LiNbO3' or self.selection == 'BBO (Kato)'
                                or self.selection == 'BBO (Eimerl)' \
                                or self.selection == 'BBO (Zhang)':
                if self.selection == 'LiNbO3':
                        d22 = 2.4
                        d31 = -4.52
                elif self.selection == 'BBO (Kato)' \
                        or self.selection == 'BBO (Eimerl)'
                        or self.selection == 'BBO (Zhang)':
                        d22 = 2.3
                        d31 = -0.16
                if process == 'Type I':
                        return abs(d31*np.sin((np.pi/180)*theta)\
                                        -d22*np.cos((np.pi/180)*theta)
                                        *np.sin(3*(np.pi/180)*phi))
                elif process == 'Type II' or process == 'Type III':
                        return abs(d22*np.cos((np.pi/180)*theta)\
                                        *np.cos((np.pi/180)*theta)\
                                        *np.cos(3*(np.pi/180)*phi))
```

return

## C.5 phasematching.py

The other non-native script called by material\_params.py is presented here. Note that while it is not listed as a dependency, phasematching.py does require nl\_crystal.py, as the functions phasematching.py contains require an object of class NLcrystals.

```
# Native python packages
import numpy as np
# Non-native python packages
# (none)
# phase matching calculation for e --> oo
# requires:
# - an object of Class nl_crystal.NLcrystals (crystal)
# - two floats (pump and signal)
def type_one(crystal, pump, signal):
```

```
idler = 1/((1/pump) - (1/signal))
```

wp = 2\*np.pi\*3E17\*(1/pump)
ws = 2\*np.pi\*3E17\*(1/signal)
wi = 2\*np.pi\*3E17\*(1/idler)

```
nop = crystal.ord(pump)
nep = crystal.ex(pump)
ns = crystal.ord(signal)
ni = crystal.ord(idler)
```

```
return (180/np.pi)\
```

```
*np.arccos(np.sqrt((1/(nep*nep-nop*nop)))\
                *(np.power(((wp*nop*nep)/(ni*wi+ns*ws)),2)-nop*nop)))
# phase matching calculation for e --> oe
# requires:
         - an object of Class nl_crystal.NLcrystals (crystal)
#
         - two floats (pump and signal)
#
def type_two(crystal, pump, signal):
        idler = 1/((1/pump)-(1/signal))
        wp = 2*np.pi*3E17*(1/pump)
        ws = 2*np.pi*3E17*(1/signal)
        wi = 2*np.pi*3E17*(1/idler)
        nop = crystal.ord(pump)
       nep = crystal.ex(pump)
       ns = crystal.ord(signal)
       noi = crystal.ord(idler)
       nei = crystal.ex(idler)
```

A = np.power(ns\*ns\*ws\*ws\*(nei\*nei-noi\*noi)\*(nep\*nep-nop\*nop),2)

B = 2\*((np.power(ns\*ns\*ws\*ws\*nop,2)\

-np.power(wp\*ns\*ws\*nop\*nep,2))\*(nep\*nep-nop\*nop)\

\*(nei\*nei-noi\*noi)\*(nei\*nei-noi\*noi)+(np.power(ns\*ns\*ws\*ws\*noi,2)\

-np.power(ns\*ws\*wi\*noi\*nei,2))\*(nei\*nei-noi\*noi)\

\*(nep\*nep-nop\*nop)\*(nep\*nep-nop\*nop))

C = ((np.power(wp\*nop\*nep,4)+np.power(ns\*ws\*nop,4)\

-2\*np.power(wp\*ns\*ws\*nop\*nop\*nep,2))\*(nei\*nei-noi\*noi)\

\*(nei\*nei-noi\*noi))+((np.power(wi\*nei\*noi,4)+np.power(ns\*ws\*noi,4)\

-2\*np.power(ns\*ws\*wi\*noi\*nei,2))\*(nep\*nep-nop\*nop)\

\*(nep\*nep-nop\*nop))+((4\*np.power(ns\*ns\*ws\*ws\*noi\*nop,2)\

-2\*np.power(wi\*wp\*nop\*nep\*noi\*nei,2)\

-4\*np.power(wp\*ns\*ws\*nop\*nep\*noi,2)\

-4\*np.power(ns\*ws\*wi\*noi\*nei\*nop,2))\

\*(nep\*nep-nop\*nop)\*(nei\*nei-noi\*noi))

D = ((2\*np.power(wp\*nop\*nep,4)\*noi\*noi\

+2\*np.power(ns\*ws\*nop,4)\*noi\*noi\

-2\*np.power(wi\*wp\*nop\*nep\*noi\*nei\*nop,2)\

-4\*np.power(wp\*ns\*ws\*nop\*nep\*noi\*nop,2)\

-2\*np.power(ns\*ws\*wi\*noi\*nei\*nop\*nop,2))\*(nei\*nei-noi\*noi))\

+((2\*np.power(wi\*nei\*noi,4)\*nop\*nop\

+2\*np.power(ns\*ws\*noi,4)\*nop\*nop\

-2\*np.power(wi\*wp\*nop\*nep\*noi\*nei\*noi,2)\

-2\*np.power(wp\*ns\*ws\*nop\*nep\*noi\*noi,2)\

-4\*np.power(ns\*ws\*wi\*noi\*nei\*nop\*noi,2))\*(nep\*nep-nop\*nop))

E = np.power(noi\*wp\*nop\*nep,4)+np.power(wi\*nei\*noi\*nop,4)\

+np.power(ns\*ws\*noi\*nop,4)\

-2\*np.power(wi\*wp\*nop\*nep\*noi\*nei\*nop\*noi,2)\

-2\*np.power(wp\*ns\*ws\*nop\*nep\*nop\*noi\*noi,2)\

-2\*np.power(ns\*ws\*wi\*noi\*nei\*noi\*nop\*nop,2)

roots = np.roots([A,B,C,D,E])

for i in roots:

if i.imag == 0:

if 0<= i.real <= 1:

return (180/np.pi)\*np.arccos(np.sqrt(i.real))

# phase matching calculation for e --> eo

```
# requires:
         - an object of Class nl_crystal.NLcrystals (crystal)
#
#
         - two floats (pump and signal)
def type_three(crystal, pump, signal):
        idler = 1/((1/pump)-(1/signal))
        return type_two(crystal, pump, idler)
# the following return the phase mismatch (in cm-1) for
# the given process at a given crystal angle
# all require:
         - an object of Class nl_crystal.NLcrystals (crystal)
#
         - three floats (pump, signal, and theta)
#
# Type I: e --> oo
def type_one_mismatch(crystal, pump, signal, theta):
        idler = 1/((1/pump)-(1/signal))
        wp = 2*np.pi*3E17*(1/pump)
        ws = 2*np.pi*3E17*(1/signal)
        wi = 2*np.pi*3E17*(1/idler)
        nop = crystal.ord(pump)
        nep = crystal.ex(pump)
        ns = crystal.ord(signal)
        ni = crystal.ord(idler)
        npu = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nep),2))\
                +(np.power(((np.cos((np.pi/180)*theta))/nop),2))))
```

```
return np.abs((npu*wp-ns*ws-ni*wi)*(1/(3E10)))
```

```
# Type II: e --> oe
def type_two_mismatch(crystal, pump, signal, theta):
    idler = 1/((1/pump)-(1/signal))
    wp = 2*np.pi*3E17*(1/pump)
    ws = 2*np.pi*3E17*(1/signal)
    wi = 2*np.pi*3E17*(1/signal)
    wi = 2*np.pi*3E17*(1/idler)
    nop = crystal.ord(pump)
    nep = crystal.ord(pump)
    ne = crystal.ex(pump)
    ns = crystal.ord(signal)
    noi = crystal.ord(idler)
    nei = crystal.ex(idler)
    npu = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nep),2))\
            +(np.power(((np.sin((np.pi/180)*theta))/nep),2))\
            ni = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nep),2))\
```

```
+(np.power(((np.cos((np.pi/180)*theta))/noi),2))))
```

```
return np.abs((npu*wp-ns*ws-ni*wi)*(1/(3E10)))
```

```
# Type III: e --> eo
```

```
def type_three_mismatch(crystal, pump, signal, theta):
```

```
idler = 1/((1/pump)-(1/signal))
```

```
wp = 2*np.pi*3E17*(1/pump)
```

```
ws = 2*np.pi*3E17*(1/signal)
```

```
wi = 2*np.pi*3E17*(1/idler)
```

```
nop = crystal.ord(pump)
nep = crystal.ex(pump)
ni = crystal.ord(idler)
nos = crystal.ord(signal)
nes = crystal.ex(signal)
```

```
npu = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nep),2))\
        +(np.power(((np.cos((np.pi/180)*theta))/nop),2))))
ns = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nes),2))\
        +(np.power(((np.cos((np.pi/180)*theta))/nos),2))))
```

return np.abs((npu\*wp-ns\*ws-ni\*wi)\*(1/(3E10)))

```
# Type IV: e --> ee
def type_four_mismatch(crystal, pump, signal, theta):
```

```
idler = 1/((1/pump)-(1/signal))
```

```
wp = 2*np.pi*3E17*(1/pump)
ws = 2*np.pi*3E17*(1/signal)
```

```
wi = 2*np.pi*3E17*(1/idler)
```

```
nop = crystal.ord(pump)
```

```
nep = crystal.ex(pump)
```

```
noi = crystal.ord(idler)
```

```
nei = crystal.ex(idler)
```

```
nos = crystal.ord(signal)
```

```
nes = crystal.ex(signal)
```

```
npu = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nep),2))\
+(np.power(((np.cos((np.pi/180)*theta))/nop),2))))
```

ns = np.sqrt(1/((np.power(((np.sin((np.pi/180)\*theta))/nes),2))\

```
+(np.power(((np.cos((np.pi/180)*theta))/nos),2))))
ni = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nei),2)))
+(np.power(((np.cos((np.pi/180)*theta))/noi),2))))
```

```
return np.abs((npu*wp-ns*ws-ni*wi)*(1/(3E10)))
```

```
# Type V: o --> oo
def type_five_mismatch(crystal, pump, signal, theta):
```

```
idler = 1/((1/pump)-(1/signal))
```

```
wp = 2*np.pi*3E17*(1/pump)
ws = 2*np.pi*3E17*(1/signal)
wi = 2*np.pi*3E17*(1/idler)
```

npu = crystal.ord(pump)

ni = crystal.ord(idler)

```
ns = crystal.ord(signal)
```

```
return np.abs((npu*wp-ns*ws-ni*wi)*(1/(3E10)))
```

```
# Type VI: o --> oe
def type_six_mismatch(crystal, pump, signal, theta):
```

```
idler = 1/((1/pump)-(1/signal))
```

```
wp = 2*np.pi*3E17*(1/pump)
ws = 2*np.pi*3E17*(1/signal)
wi = 2*np.pi*3E17*(1/idler)
```

```
npu = crystal.ord(pump)
noi = crystal.ord(idler)
```

```
nei = crystal.ex(idler)
ns = crystal.ord(signal)
```

```
ni = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nei),2))\
+(np.power(((np.cos((np.pi/180)*theta))/noi),2))))
```

```
return np.abs((npu*wp-ns*ws-ni*wi)*(1/(3E10)))
```

```
# Type VII: o --> eo
```

```
def type_seven_mismatch(crystal, pump, signal, theta):
```

```
idler = 1/((1/pump)-(1/signal))
```

wp = 2\*np.pi\*3E17\*(1/pump)
ws = 2\*np.pi\*3E17\*(1/signal)

wi = 2\*np.pi\*3E17\*(1/idler)

```
npu = crystal.ord(pump)
ni = crystal.ord(idler)
nos = crystal.ord(signal)
nes = crystal.ex(signal)
```

```
ns = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nes),2))\
+(np.power(((np.cos((np.pi/180)*theta))/nos),2))))
```

```
return np.abs((npu*wp-ns*ws-ni*wi)*(1/(3E10)))
```

```
# Type VIII: o --> ee
```

```
def type_eight_mismatch(crystal, pump, signal, theta):
```

idler = 1/((1/pump)-(1/signal))

```
wp = 2*np.pi*3E17*(1/pump)
ws = 2*np.pi*3E17*(1/signal)
wi = 2*np.pi*3E17*(1/idler)
npu = crystal.ord(pump)
noi = crystal.ord(idler)
nei = crystal.ex(idler)
nos = crystal.ex(idler)
nos = crystal.ord(signal)
nes = crystal.ex(signal)
ns = np.sqrt(1/((np.power(((np.sin((np.pi/180)*theta))/nes),2)))
+(np.power(((np.cos((np.pi/180)*theta))/nos),2))))
```

return np.abs((npu\*wp-ns\*ws-ni\*wi)\*(1/(3E10)))

## C.6 efield.py

The script efield.py contains the electric\_field class; this class has been defined in order to not only contain the electric fields, but also to provide functionalities and properties necessary throughout the script.

# Native python packages import numpy as np import matplotlib.pyplot as plt import math import sys import os import datetime from scipy import interpolate import glob # Non-native python packages import phasematching as  ${\tt pm}$ 

import parameters as p

class electric\_field:

def \_\_init\_\_(self, field, params):

self.E = (np.empty(len(params.freq)))
self.id = str(field)
self.l = params.u.wavelengths[str(field)]
self.n0 = params.n(self.l, str(field))
self.w0 = 2\*np.pi\*params.c\*(1E9/self.l)

def fill(self, params, fluence):

if params.u.field\_envelopes[self.id] == 'Gaussian':
 self.gaussian(params, fluence)

elif params.u.field\_envelopes[self.id] == 'zero':

self.zero()

elif params.u.field\_envelopes[self.id] == 'file':

# Not implemented
exit()

elif params.u.field\_envelopes[self.id] == 'rectangular':

self.rectangular(params, fluence)

```
return
```

```
def int(self):
```

```
return (1.33E-3)*(self.n0)*(np.real(self.E)**2 + np.imag(self.E)**2)
def gaussian(self, params, fluence):
        reEt = np.empty(len(params.time))
        imEt = np.empty(len(params.time))
        sum = ((np.exp((-4.*np.log(2))*(((params.time \
                - params.u.Gauss_centers[self.id])
                /params.u.Gauss_widths[self.id])**2)))\
                *(params.time[1]-params.time[0])).sum()
        reA = ((np.cos((np.pi/180)*params.u.field_phase[self.id]))**2)\
                        *(fluence/sum)
        imA = ((np.sin((np.pi/180)*params.u.field_phase[self.id]))**2)\
                        *(fluence/sum)
        reEt = np.sqrt((reA/((1.33E-3)*self.n0))*np.exp((-4.*np.log(2))\
                        *(((params.time - params.u.Gauss_centers[self.id])\
                        /params.u.Gauss_widths[self.id])**2)))
        imEt = np.sqrt((imA/((1.33E-3)*self.n0))*np.exp((-4.*np.log(2))\
                        *(((params.time - params.u.Gauss_centers[self.id])\
                        /params.u.Gauss_widths[self.id])**2)))
```

self.ft(reEt, imEt, params)

if params.u.apply\_contrast:

retemp = np.real(self.E)

```
imtemp = np.imag(self.E)
```

```
gmax = max(max(np.abs(retemp)), max(np.abs(imtemp)))
retemp[np.where(np.abs(retemp)<gmax*params.u.contrast)] = 0.
imtemp[np.where(np.abs(imtemp)<gmax*params.u.contrast)] = 0.</pre>
```

self.E = retemp + 1j\*imtemp

return

def zero(self):

self.E = np.zeros(len(self.E)) + 1j\*np.zeros(len(self.E))

def rectangular(self, params, fluence):

self.E = (np.empty(len(params.freq),dtype=complex))

```
for i in range(len(params.freq)):
    if i < 5 or i > len(params.freq) - 5:
        self.E[i] = complex(0.)
    else:
```

self.E \*= np.sqrt(fluence/(self.int()).sum())\*self.E

self.E[i] = 1.+1j\*0.

return

def ft(self, reEt, imEt, params):

```
reE = np.empty(len(params.freq))
imE = np.empty(len(params.freq))
```

```
self.E = (1/np.sqrt(len(params.time)))*np.sum( (reEt+1j*imEt)[None,:]\
    *np.exp(-1j*params.freq[:,None]*params.time[None,:]), axis = 1)
```

return

def ift(self, params):

```
return (1/np.sqrt(len(params.time)))*np.sum( self.E[None,:]\
    *np.exp(1j*params.freq[None,:]*params.time[:,None]), axis = 1)
```

def pulse\_energy(self, params):

Et = self.ift(params)

C.7 controls.py

The script controls.py contains many of the central functions of the script: once the parameters are defined through parameters.py, many of the manipulations and executions are executed through functions in controls.py.

```
# Native python packages
import numpy as np
import matplotlib.pyplot as plt
import scipy.interpolate as interp
```

import multiprocessing, threading, datetime, os, functools, math
# Non-native python packages
import parameters, efield, algorithm
import user\_settings as user
import material\_params as mat
import gain\_formulae as gain
import plotting\_routines as pltr
import phasematching as pm
import nl\_crystal as nl

# modify\_params modifies constants and values in the parameters object
# appropriate to user specifications

```
def modify_params(p):
```

if p.u.modify\_theta: p.m.theta = p.u.alt\_theta if p.u.parametric\_process == 'second-harmonic generation': p.m.dk = pm.type\_one\_mismatch(p.m.cry, p.u.wavelengths['SHG'],\ p.u.wavelengths['Fundamental'], p.m.theta) elif p.u.parametric\_process == 'parametric amplification': if p.u.process == 'Type I': p.m.dk = pm.type\_one\_mismatch(p.m.cry,\ p.u.wavelengths['Pump'],\ p.u.wavelengths['Signal'], p.m.theta) elif p.u.process == 'Type II': p.m.dk = pm.type\_two\_mismatch(p.m.cry,\ p.u.wavelengths['Pump'],\ p.u.wavelengths['Signal'], p.m.theta) elif p.u.process == 'Type III': p.m.dk = pm.type\_three\_mismatch(p.m.cry,\ p.u.wavelengths['Pump'],\

```
p.m.dk*=100.
if p.u.modify_chi2eff:
        p.m.chi2eff = p.u.alt_chi2eff
if p.u.modify_chi3eff:
        p.m.chi3eff = p.u.alt_chi3eff
if p.u.modify_ug:
        for i in [p.u.alt_ug]:
                for j in p.m.mask:
                        i[j] = i.pop(p.m.mask[j])
                ini_keys = i.keys()
                for j in ini_keys:
                        if j in p.m.mask:
                                pass
                        else:
                                i.pop(j,None)
        for i in p.u.alt_ug:
                p.m.ug[i] = p.u.alt_ug[i]*1.e12
```

p.u.wavelengths['Signal'], p.m.theta)

def initialize(p):

```
if p.u.silent:
```

pass

else:

print('Initializing fields')

fields = []

for i in range(len(p.r)):

fields.append({})

for j in p.u.wavelengths:

fields[i][j] = efield.electric\_field(j,p)
fields[i][j].fill(p, p.fluence[j][i])
fields[i][j].E[np.where((fields[i][j].w0+p.freq)<=\</pre>

p.cutoff)] = complex(0.)

return fields

def drive\_algorithm(p, fields):

if p.u.silent:

pass

else:

print('Starting algorithm')

if p.u.slowly\_varying\_amp:

if p.u.adaptive\_steps:

# Adaptive stepping has not yet been developed; when this capability is developed,

# this is how it may be implemented

#

fields = adaptive\_stepping(p,fields)

pass

else:

fields = fixed\_stepping(p, fields)

else:

if p.u.adaptive\_steps:

# Adaptive stepping has not yet been developed; when this capability is developed,

# this is how it may be implemented

else:

```
fields = second_order_fixed_stepping(p, fields)
```

return fields

def fixed\_stepping(p, fields):

step\_size = p.u.L/p.u.steps

parameter\_dict = {

```
'curr_position' : p.z,
'step_size' : step_size,
'c' : p.c,
'dw' : p.freq,
'w' : {},
'k' : {},
'n0' : {},
'w0' : {},
'chi2eff' : p.m.chi2eff,
'chi3eff' : p.m.chi3eff,
'ug' : {},
'gvd' : {},
'parametric_process' : p.u.parametric_process,
'dk' : p.m.dk,
'cutoff' : p.cutoff,
'time' : p.time,
'NaN_flag' : False}
```

field\_iter = []

```
for i in fields[0]:
```

for j in range(len(parameter\_dict['w'][fields[0][i].id])):

else:

```
parameter_dict['w'][fields[0][i].id][j] = 0.
parameter_dict['k'][fields[0][i].id][j] = 0.
```

```
for i in range(len(p.r)):
```

```
field_iter.append({})
```

```
for j in fields[0]:
```

```
field_iter[i][j] = fields[i][j].E.copy()
```

counter = 0

```
if p.u.timer:
```

start\_time = str(datetime.datetime.now()).split(' ')[-1]

while counter < p.u.steps:

```
counter += 1
```

if p.u.silent:

pass

else:

```
print('Starting iteration '+str(counter)+' of '+str(p.u.steps)\
```

```
+' at '+str(datetime.datetime.now()))
```

if p.u.threading:

```
if p.u.cores != 'Default' \
    and p.u.cores <= multiprocessing.cpu_count() \
    and p.u.cores > 0:
```

```
pool = multiprocessing.Pool(processes = p.u.cores)
chunk_size = int(math.ceil(len(field_iter)/p.u.cores))
```

elif p.u.cores != 'Default' and p.u.limit\_override == True:

```
pool = multiprocessing.Pool(processes = p.u.cores)
chunk_size = int(math.ceil(len(field_iter)/p.u.cores))
```

else:

if p.u.cores == 'Default':
 pass

```
elif p.u.cores > multiprocessing.cpu_count() \
    or p.u.cores<1:
        print('An invalid number of cores was requested.\
            Defaulting to '\
            +str(multiprocessing.cpu_count()-1)\
            +' cores.')

pool = multiprocessing.Pool(processes = \
                (multiprocessing.cpu_count()-1))
chunk_size = int(math.ceil(len(field_iter)\
                /(multiprocessing.cpu_count()-1)))
results = pool.map(functools.partial(algorithm.Runge_Kutta, \)</pre>
```

```
parameter_dict), field_iter, \
```

```
chunksize = chunk_size)
```

pool.close()

```
for i in range(len(results)):
```

```
for j in results[i]:
```

```
field_iter[i][j] = results[i][j].copy()
```

del results

else:

```
for i in range(len(p.r)):
```

```
for j in results:
            field_iter[i][j] = results[j].copy()
```

parameter\_dict['curr\_position'] += step\_size

```
if parameter_dict['NaN_flag']:
```

break

```
for i in range(len(p.r)):
    for j in fields[0]:
        fields[i][j].E = field_iter[i][j].copy()
```

```
p.z = parameter_dict['curr_position']
```

```
if p.u.timer:
```

return fields

def second\_order\_fixed\_stepping(p, fields):

step\_size = p.u.L/p.u.steps

parameter\_dict = {

'curr\_position' : p.z,

```
'step_size' : step_size,
'c' : p.c,
'dw' : p.freq,
'w' : {},
'k' : {},
'n0' : {},
'w0' : {},
'chi2eff' : p.m.chi2eff,
'chi3eff' : p.m.chi3eff,
'ug' : {},
'gvd' : {},
'parametric_process' : p.u.parametric_process,
'dk' : p.m.dk,
'cutoff' : p.cutoff,
'NaN_flag' : False}
```

field\_iter = []

for i in fields[0]:

for j in range(len(parameter\_dict['w'][fields[0][i].id])):

if parameter\_dict['w'][fields[0][i].id][j] >= p.cutoff:

wavelength = ((2\*np.pi\*p.c)\

/(parameter\_dict['w'][fields[0][i].id][j]))\*1.e9
```
parameter_dict['k'][fields[0][i].id][j] = \
    (p.n(wavelength, fields[0][i].id)\
    *parameter_dict['w'][fields[0][i].id][j])/p.c
```

else:

```
parameter_dict['w'][fields[0][i].id][j] = 0.
parameter_dict['k'][fields[0][i].id][j] = 0.
```

```
for i in range(len(p.r)):
```

field\_iter.append({})

```
for j in fields[0]:
    field_iter[i][j] = {}
    field_iter[i][j]['field'] = fields[i][j].E.copy()
    field_iter[i][j]['first derivative'] = \
        np.zeros(len(fields[i][j].E)) \
        + 1j*np.zeros(len(fields[i][j].E))
```

counter = 0

if p.u.timer:

```
start_time = str(datetime.datetime.now()).split(' ')[-1]
```

while counter < p.u.steps:

counter += 1

if p.u.silent:

pass

else:

print('Starting iteration '+str(counter)+' of '\

+str(p.u.steps)+' at '+str(datetime.datetime.now()))

```
if p.u.threading:
```

```
if p.u.cores != 'Default' \
    and p.u.cores <= multiprocessing.cpu_count() \
    and p.u.cores > 0:
```

```
pool = multiprocessing.Pool(processes = p.u.cores)
chunk_size = int(math.ceil(len(field_iter)/p.u.cores))
```

elif p.u.cores != 'Default' and p.u.limit\_override == True:

```
pool = multiprocessing.Pool(processes = p.u.cores)
chunk_size = int(math.ceil(len(field_iter)/p.u.cores))
```

else:

```
if p.u.cores == 'Default':
    pass
```

```
elif p.u.cores > multiprocessing.cpu_count() \
    or p.u.cores<1:
        print('An invalid number of cores was requested.\
            Defaulting to '\
            +str(multiprocessing.cpu_count()-1)\
            +' cores.')</pre>
```

```
chunk_size = int(math.ceil(len(field_iter)\
```

```
/(multiprocessing.cpu_count()-1)))
```

```
results = pool.map(functools.partial(\
```

```
algorithm.second_order_Runge_Kutta, \
parameter_dict), field_iter, \
chunksize = chunk_size)
```

```
pool.close()
```

for i in range(len(results)):

```
for j in results[i]:
```

field\_iter[i][j]['field'] = \

results[i][j]['field'].copy()

```
field_iter[i][j]['first derivative'] = \
```

results[i][j]['first derivative'].copy()

del results

else:

```
for i in range(len(p.r)):
```

for j in results:

field\_iter[i][j]['field'] = \

results[j]['field'].copy()

```
field_iter[i][j]['first derivative'] = \
```

```
results[j]['first derivative'].copy()
```

parameter\_dict['curr\_position'] += step\_size

```
if parameter_dict['NaN_flag']:
```

break

```
for i in range(len(p.r)):
```

for j in fields[0]:

```
fields[i][j].E = field_iter[i][j]['field'].copy()
```

```
p.z = parameter_dict['curr_position']
```

if p.u.timer:

return fields

def save\_output(p, fields, state):

```
curr = os.getcwd()
```

if state == 'initial':

```
folder_name = ''
```

for i in ((str(datetime.datetime.now()).split('.')[0]).split(':')):

```
folder_name += i+'_'
```

for i in p.u.wavelengths:

folder\_name += str(p.u.wavelengths[i])+'\_'

folder\_name+=p.u.parametric\_process+'\_'+p.u.material+'\_'+p.u.process

if p.u.save\_dir[-1]=='/':

p.u.save\_dir += folder\_name

else:

p.u.save\_dir += '/'+folder\_name

if os.path.exists(p.u.save\_dir):

pass

else:

os.makedirs(p.u.save\_dir)

os.chdir(p.u.save\_dir)

if state == 'initial':

write\_parameters(p)

write\_fields(p, fields, 'initial')

pltr.plot2d(fields, p, 'initial\_fields\_freq')
pltr.plot2dtime(fields, p, 'initial\_fields\_time')
pltr.plot2dweighted(fields, p, 'initial\_fields\_freq\_weighted')

elif state == 'final':

write\_fields(p, fields, 'final')

pltr.plot2d(fields, p, 'final\_fields\_freq')

pltr.plot2dtime(fields, p, 'final\_fields\_time')

pltr.plot2dweighted(fields, p, 'final\_fields\_freq\_weighted')

```
os.chdir(curr)
```

return

```
def write_parameters(p):
```

```
f = open('simulation_parameters.txt', 'w+')
```

```
f.write('Simulation Parameters\n\n')
```

```
f.write('Wavelengths\n')
```

for i in p.u.wavelengths:

```
f.write(str(i)+':\t'+str(p.u.wavelengths[i])+' nm\n')
```

```
f.write('\n')
```

```
f.write('Material Parameters\n')
```

```
if p.mat_type_marker == 'birefringent':
```

```
f.write(p.u.material+': '+str(p.u.L*100)+' cm, theta: '\
```

```
+str(p.m.theta)+' deg, phi: '+str(p.u.phi)+' deg\n')
```

```
elif p.mat_type_marker == 'isotropic':
```

```
f.write(p.u.material+': '+str(p.u.L*100)+' cm\n')
```

f.write('Number of steps: {0}\n'.format(p.u.steps))

```
if p.u.parametric_process == 'parametric amplification':
```

```
f.write('Process: '+p.u.parametric_process+', '+p.u.process+'\n')
else:
```

f.write('Process: '+p.u.parametric\_process+'\n')

```
f.write('Beam Parameters\n')
for i, j in [('Initial Delay', p.u.Gauss_centers), ('Beam Size', p.u.e2_size), \
        ('Pulse Width', p.u.Gauss_widths), ('Initial Phase', p.u.field_phase)]:
        curr_string = i+':'
        if i == 'Initial Delay' or i == 'Pulse Width':
                modifier = 1.e15
                unit = 'fs'
        if i == 'Beam Size':
                modifier = 100.
                unit = 'cm'
        if i == 'Initial Phase':
                modifier = 1.
                unit = 'deg.'
        for k in j:
                curr_string += '\t'+k+': '+str(j[k]*modifier)+' '+unit+';'
        curr_string+= '\n'
        f.write(curr_string)
        del curr_string
```

```
f.write('\n')
```

f.write('\n')

```
if p.u.noncollinear or p.u.modify_ug or p.u.modify_chi2eff or p.u.modify_chi3eff:
    f.write('Modifications\n')
```

if p.u.noncollinear:

```
f.write('Noncollinear angle: '+str(p.u.gamma)+' deg\n')
if p.u.modify_chi2eff:
    f.write('Alternate chi2: '+str(p.m.chi2eff*1E12)+' pm/V\n')
if p.u.modify_chi3eff:
    f.write('Alternate chi3: '+str(p.m.chi3eff)+'\n')
if p.u.modify_ug:
    f.write('Group velocities:\n')
    for i in p.m.ug:
        f.write(str(i)+':\t'+str(p.m.ug[i]*1e-12)+' mm/fs\n')
f.close()
return
```

```
def write_fields(p, fields, state):
    if state == 'initial':
        f = open('initial_fields.txt', 'w+')
    elif state == 'final':
        f = open('final_fields.txt', 'w+')
    for i in range(len(p.r)):
        f.write('Fields at r = '+str(p.r[i]*1000)+' mm\n\n')
        f.write('Detuning (PHz)')
        for j in fields[i]:
            f.write('\tRe('+str(j)+')\tIm('+str(j)+')')
        f.write('\n')
        f.write('\n')
        for j in range(len(p.freq)):
```

```
f.write(str(p.freq[j]*1e-15))
```

for k in fields[i]:

```
f.write('\t'+str(np.real(fields[i][k].E[j]))+'\t'\
+str(np.imag(fields[i][k].E[j])))
```

```
f.write('\n')
```

```
f.write('\n\n')
```

f.close()

del f

```
if state == 'initial':
    f = open('initial_integrated_spectrum.txt', 'w+')
elif state == 'final':
    f = open('final_integrated_spectrum.txt', 'w+')
int_Int = {}
for j in fields[0]:
    Int = []
    for i in range(len(p.r)):
        Int.append(fields[i][j].int())
    int_Int[j] = []
    for i in range(len(p.freq)):
        sum = 0
```

```
for k in range(len(p.r)):
    sum+=2*np.pi*Int[k][i]*p.r[k]*(p.r[1]-p.r[0])
int_Int[j].append(sum)
```

del Int

```
energy = {}
```

for j in fields[0]:

sum = 0

```
for i in range(len(p.r)):
    sum += 2*np.pi*(fields[i][j].pulse_energy(p))\
    *p.r[i]*(p.r[1]-p.r[0])
```

```
energy[j] = sum
```

```
f.write('Final Pulse Energies\n')
```

for j in energy:

```
f.write(j+':\t'+str(energy[j]*1.e6)+' uJ\n')
```

f.write('\n\n')

f.write('Integrated Spectra\n\n')

```
f.write('Detuning (PHz)')
```

for j in int\_Int:

f.write('\t'+str(j))

f.write('\n')

for j in range(len(p.freq)):

f.write(str(p.freq[j]\*1e-15))

for k in int\_Int:

f.write('\t'+str(int\_Int[k][j]))

f.write('\n')

f.close()

C.8 algorithm.py

The two functions in algorithm.py execute one step of the Runge-Kutta algorithm when called.

```
# Native python packages
import numpy as np
import matplotlib.pyplot as plt
# Non-native python packages
import gain_formulae as gain
import efield
```

def Runge\_Kutta(pd, fields):

fm = {}
fm['z'] = pd['curr\_position']

for i in fields:

```
fm[i] = fields[i].copy()
KI = {}
for i in fields:
    KI[i] = gain.gain(pd, i, fm)
    KI[i] *= pd['step_size']
del fm
fm = {}
```

```
fm['z'] = pd['curr_position'] + 0.5*pd['step_size']
```

```
for i in fields:
    fm[i] = (fields[i] + 0.5*KI[i]).copy()
```

 $KII = \{\}$ 

```
for i in fields:
    KII[i] = gain.gain(pd, i, fm)
    KII[i] *= pd['step_size']
```

del fm
fm ={}
fm['z'] = pd['curr\_position'] + 0.5\*pd['step\_size']

for i in fields: fm[i] = (fields[i] + 0.5\*KII[i]).copy()

```
KIII = \{\}
for i in fields:
        KIII[i] = gain.gain(pd, i, fm)
        KIII[i] *= pd['step_size']
del fm
fm ={}
fm['z'] = pd['curr_position'] + pd['step_size']
for i in fields:
        fm[i] = (fields[i] + KIII[i]).copy()
KIV = {}
for i in fields:
        KIV[i] = gain.gain(pd, i, fm)
        KIV[i] *= pd['step_size']
output = {}
for i in fields:
        output[i] = (fields[i]+(1./6.)*KI[i]+(1./3.)*KII[i]+(1./3.)*KIII[i]\
                                + (1./6.)*KIV[i]).copy()
        if np.isnan(output[i].sum()):
                print('nan present in '+i)
del fm, KI, KII, KIII, KIV
return output
```

def second\_order\_Runge\_Kutta(pd, fields):

```
fm = {}
fm['z'] = pd['curr_position']
```

```
for i in fields:
    fm[i] = {}
    fm[i]['field'] = fields[i]['field'].copy()
    fm[i]['first derivative'] = fields[i]['first derivative'].copy()
```

KI = {}

```
for i in fields:
    KI[i] = gain.second_order_gain(pd, i, fm)
    KI[i]['field'] *= pd['step_size']
    KI[i]['first derivative'] *= pd['step_size']
```

```
del fm
fm = {}
fm['z'] = pd['curr_position'] + 0.5*pd['step_size']
```

 $KII = \{\}$ 

```
for i in fields:
    KII[i] = gain.second_order_gain(pd, i, fm)
    KII[i]['field'] *= pd['step_size']
```

```
KII[i]['first derivative'] *= pd['step_size']
del fm
fm ={}
fm['z'] = pd['curr_position'] + 0.5*pd['step_size']
for i in fields:
        fm[i] = {}
        fm[i]['field'] = (fields[i]['field'] + 0.5*KII[i]['field']).copy()
        fm[i]['first derivative'] = (fields[i]['first derivative'] \
                                + 0.5*KII[i]['first derivative']).copy()
KIII = \{\}
for i in fields:
        KIII[i] = gain.second_order_gain(pd, i, fm)
        KIII[i]['field'] *= pd['step_size']
        KIII[i]['first derivative'] *= pd['step_size']
del fm
fm ={}
fm['z'] = pd['curr_position'] + pd['step_size']
for i in fields:
        fm[i] = \{\}
        fm[i]['field'] = (fields[i]['field'] + KIII[i]['field']).copy()
        fm[i]['first derivative'] = (fields[i]['first derivative'] \
                                + KIII[i]['first derivative']).copy()
```

KIV = {}

for i in fields:

```
KIV[i] = gain.second_order_gain(pd, i, fm)
        KIV[i]['field'] *= pd['step_size']
        KIV[i]['first derivative'] *= pd['step_size']
output = {}
for i in fields:
        output[i] = {}
        output[i]['field'] = (fields[i]['field'] + (1./6.)*KI[i]['field'] \
                                + (1./3.)*KII[i]['field'] \
                                + (1./3.)*KIII[i]['field'] \
                                + (1./6.)*KIV[i]['field']).copy()
        output[i]['first derivative'] = (fields[i]['first derivative'] \
                        + (1./6.)*KI[i]['first derivative'] \
                        + (1./3.)*KII[i]['first derivative'] \
                        + (1./3.)*KIII[i]['first derivative'] \
                        + (1./6.)*KIV[i]['first derivative']).copy()
        if np.isnan(output[i]['field'].sum()) \
                or np.isnan(output[i]['first derivative'].sum()):
                print('nan present in '+i)
                pd['NaN_flag'] = True
```

del fm, KI, KII, KIII, KIV

return output

## C.9 gain\_formulae.py

While the code in § C.8 performs the sequence of calculations necessary for one iteration of the Runge-Kutta algorithm, it is programmed such that it invokes an external function to provide the gain formulae representing the system of differential equations. For the parametric processes being simulated, these formulae are

provided in gain\_formulae.py. It should be noted that - as written - the functions defining the gain formulae look to user\_settings.py (the gain\_selections class) to determine which terms from the differential equation should be incorporated.

# Native python packages import numpy as np import scipy.interpolate as interp import matplotlib.pyplot as plt # Non-native python packages import user\_settings as user

def gain(pd, id, fm):

u = user.gain\_selections()

if u.sva\_absorption:

# Enables accounting of absorption by BBO

1.7605, 1.7605, 1.7605]

wl\_array = [150, 170, 190, 195, 220, 235, 255, 280, 300.78125, 335, 375,\
 421.875, 500, 578.125, 644.53125, 700, 1000, 1500, 1600, 1660,\
 2000, 2085.9375, 2310, 2355, 2400, 2460, 2500, 2610, 2655,\
 2725, 2775, 2785.15625, 2899.15966, 2921.875, 3000, 3020,\
 3050, 3060, 3100, 3120, 3150, 3170, 3200, 3300, 40000]
alpha\_array = [0.2808, 0.195909, 0.14617, 0.129779, 0.089947, \
 0.07212557, 0.05990443, 0.04796, 0.125, 0.102, 0.0788, 0.06827,\
 0.05797, 0.04194, 0.03528, 0.02354, 0.02518, 0.02518, 0.02518, \
 0.01305, 0.01305, 0.025177, 0.3999, 0.3629, 0.3999, 0.1547,\
 0.2043, 0.4533, 0.6349, 0.4533, 0.6297, 0.7851, 0.7851, 0.8935,\
 1.0038, 1.3202, 1.2058, 1.5426, 1.5426, 1.2808, 1.2808, 1.5931,\

```
freq_array = (2*np.pi*3.e8*1.e9*(1/np.asarray(wl_array))).copy()
alpha_array = np.asarray(alpha_array)*100.
```

```
# It should be noted that interp.interp1d may require
```

```
# 'np.asarray(freq_array)[::-1]' and
```

```
if pd['parametric_process'] == 'second-harmonic generation' \
    or pd['parametric_process'] == 'parametric amplification' \
    or pd['parametric_process'] == 'sum-frequency generation':
```

```
if pd['parametric_process'] == 'second-harmonic generation':
    ref = 'Fundamental'
elif pd['parametric_process'] == 'parametric amplification':
    ref = 'Pump'
elif pd['parametric_process'] == 'sum-frequency generation':
    ref = 'Signal'
```

gain = np.zeros(len(pd['w'][id])) + 1j\*np.zeros(len(pd['w'][id]))

k0 = (pd['n0'][id]\*pd['w0'][id])/pd['c']

if u.sva\_group\_velocity:

```
if id == ref:
pass
```

else:

if u.sva\_in\_lab\_frame:

else:

if u.sva\_GVD:

if u.sva\_in\_lab\_frame:

else:

if u.sva\_absorption:

if u.sva\_in\_lab\_frame:

gain += - (alpha\_func(pd['w'][id])/2.)\*fm[id]

else:

if u.sva\_amplification:

if u.sva\_in\_lab\_frame:

```
coeff = (1./(2.*k0))
```

else:

if pd['parametric\_process'] == 'second-harmonic generation':

if id == 'Fundamental':

```
*pd['dw'][None,:]\
 *pd['time'][:,None]),axis=1)
shg_time = (1/np.sqrt(len(pd['dw'])))\
 *np.sum(np.exp(-1j\
 *pd['k']['SHG'][None,:]\
 *fm['z'])*temp_SHG[None,:]*np.exp(1j\
 *pd['dw'][None,:]\
 *pd['time'][:,None]),axis=1)
```

polzn = ((2\*pd['chi2eff'])\

/np.sqrt(len(pd['dw'])))\
\*np.sum(np.conj(fund\_time)[None,:]\
\*shg\_time[None,:]\
\*np.exp(-1j\*pd['dw'][:,None]\
\*pd['time'][None,:]), axis=1)
polzn[np.where(pd['w']['Fundamental'] \

<= pd['cutoff'])] = complex(0.)

del temp\_fund, temp\_SHG, fund\_time, shg\_time

elif id == 'SHG':

```
fund_time = (1/np.sqrt(len(pd['dw'])))\
 *np.sum(np.exp(-1j\
 *pd['k']['Fundamental'][None,:]\
 *fm['z'])*temp_fund[None,:]*np.exp(1j\
 *pd['dw'][None,:]\
 *pd['time'][:,None]),axis=1)
```

```
polzn = (pd['chi2eff']/np.sqrt(len(pd['dw'])))\
    *np.sum(fund_time[None,:]\
    *fund_time[None,:]*np.exp(-1j\
    *pd['dw'][:,None]\
    *pd['time'][None,:]), axis=1)
polzn[np.where(pd['w']['SHG'] \
    <= pd['cutoff'])] = complex(0.)</pre>
```

del temp\_fund, fund\_time

if id == 'Pump':

```
signal_time = (1/np.sqrt(len(pd['dw'])))\
    *np.sum(np.exp(-1j\
    *pd['k']['Signal'][None,:]\
    *fm['z'])*temp_signal[None,:]*np.exp(1j\
    *pd['dw'][None,:]\
    *pd['time'][:,None]),axis=1)
idler_time = (1/np.sqrt(len(pd['dw'])))\
    *np.sum(np.exp(-1j\)
```

```
*pd['k']['Idler'][None,:]\
*fm['z'])*temp_idler[None,:]*np.exp(1j\
*pd['dw'][None,:]\
*pd['time'][:,None]),axis=1)
```

```
polzn=((2*pd['chi2eff'])/np.sqrt(len(pd['dw'])))\
 *np.sum(signal_time[None,:]\
 *idler_time[None,:]*np.exp(-1j\
 *pd['dw'][:,None]\
 *pd['time'][None,:]), axis=1)
polzn[np.where(pd['w']['Pump'] \
 <= pd['cutoff'])] = complex(0.)</pre>
```

```
del temp_signal, temp_idler,\
    signal_time, idler_time
```

```
elif id == 'Signal':
```

```
pump_time = (1/np.sqrt(len(pd['dw'])))\
    *np.sum(np.exp(-1j\
    *pd['k']['Pump'][None,:]\
    *fm['z'])*temp_pump[None,:]\
    *np.exp(1j*pd['dw'][None,:]\
    *pd['time'][:,None]),axis=1)
idler_time = (1/np.sqrt(len(pd['dw'])))\
```

```
*np.sum(np.exp(-1j\
*pd['k']['Idler'][None,:]*fm['z'])\
*temp_idler[None,:]*np.exp(1j\
*pd['dw'][None,:]\
*pd['time'][:,None]),axis=1)
```

```
polzn = ((2*pd['chi2eff'])\
    /np.sqrt(len(pd['dw'])))\
    *np.sum(pump_time[None,:]\
    *np.conj(idler_time)[None,:]\
    *np.exp(-1j*pd['dw'][:,None]\
    *pd['time'][None,:]), axis=1)
polzn[np.where(pd['w']['Signal'] \
    <= pd['cutoff'])] = complex(0.)</pre>
```

del temp\_pump, temp\_idler, pump\_time, idler\_time

```
elif id == 'Idler':
```

```
pump_time = (1/np.sqrt(len(pd['dw'])))\
 *np.sum(np.exp(-1j\
 *pd['k']['Pump'][None,:]*fm['z'])\
 *temp_pump[None,:]*np.exp(1j\
 *pd['dw'][None,:]\
 *pd['time'][:,None]),axis=1)
```

```
signal_time = (1/np.sqrt(len(pd['dw'])))\
 *np.sum(np.exp(-1j\
 *pd['k']['Signal'][None,:]\
 *fm['z'])*temp_signal[None,:]\
 *np.exp(1j*pd['dw'][None,:]\
 *pd['time'][:,None]),axis=1)
```

```
del temp_pump, temp_signal, \
    pump_time, signal_time
```

```
gain += -1j*((coeff)/(pd['c']*pd['c']))*np.exp(1j\
            *pd['k'][id]*fm['z'])*(pd['w'][id]*pd['w'][id])*polzn
```

del polzn

return gain

else:

exit()

def second\_order\_gain(pd, id, fm):

# This has not been maintained...

```
C.10 plotting_routines.py
```

The final script, plotting\_routines.py, defines the functions previously called for plotting the results of the simulations.

```
# Native python packages
import matplotlib.pyplot as plt
import numpy as np
# Non-native python packages
# (none)
```

def plot2d(fields, p, filename):

```
Refields = {}
Imfields = {}
Int = {}
int_Int = {}
```

```
for i in range(len(p.r)):
    Refields[j].append(np.real(fields[i][j].E))
    Imfields[j].append(np.imag(fields[i][j].E))
    Int[j].append(fields[i][j].int())
```

```
int_Int[j] = []
for i in range(len(p.freq)):
    sum = 0
    for k in range(len(p.r)):
        sum+=2*np.pi*Int[j][k][i]*p.r[k]*(p.r[1]-p.r[0])
        int_Int[j].append(sum)
```

```
plt.clf()
plt.figure(1, figsize=(12,8))
subplots = 1
show_title = True
units_y = False
label_x = False
counter = 1
for j in fields[0]:
```

```
globalmax = max(np.amax(np.asarray(Refields[j])), \
    np.amax(np.asarray(Imfields[j])), \
    abs(np.amin(np.asarray(Refields[j]))), \
    abs(np.amin(np.asarray(Imfields[j]))))
```

if globalmax == 0:

```
pass
```

```
else:
```

```
globalmin = min(np.amin(np.asarray(Refields[j])), \
    np.amin(np.asarray(Imfields[j])), \
    abs(np.amax(np.asarray(Refields[j]))), \
    abs(np.amax(np.asarray(Imfields[j])))
```

```
plt.subplot(len(fields[0]), 4, subplots)
```

```
if globalmax == 0:
```

else:

```
plt.contourf(p.r*1000., freqs, np.asarray(Refields[j]).T\
    /globalmax,200, cmap='seismic', \
    levels=np.linspace(-1,1,200))
```

if label\_x:

```
plt.xlabel('cross section (mm)')
```

else:

plt.tick\_params(axis='x', which='both', labeltop='off',  $\backslash$ 

labelbottom='off')

if units\_y:

plt.ylabel(j+'\nangular frequency (x10-15)')

else:

```
plt.ylabel(j+'\n')
```

if show\_title:

```
plt.title('Re(E)')
plt.clim(-1,1)
```

```
plt.ylim(freqs[0], freqs[-1])
subplots+=1
```

```
plt.subplot(len(fields[0]), 4, subplots)
```

```
if globalmax == 0:
```

```
plt.contourf(p.r*1000., freqs, np.asarray(Imfields[j]).T,200, \
```

```
cmap='seismic', levels=np.linspace(-1,1,200))
```

else:

```
plt.contourf(p.r*1000., freqs, np.asarray(Imfields[j]).T\
    /globalmax,200, cmap='seismic', \
    levels=np.linspace(-1,1,200))
```

if label\_x:

```
plt.xlabel('cross section (mm)')
```

else:

```
plt.tick_params(axis='y', which='both', labelright='off', \
```

labelleft='off')

if show\_title:

```
plt.title('Im(E)')
```

```
plt.clim(-1,1)
plt.ylim(freqs[0], freqs[-1])
subplots+=1
```

```
plt.subplot(len(fields[0]), 4, subplots)
plt.contourf(p.r*1000., freqs, np.asarray(Int[j]).T, 200, cmap='seismic')
```

```
if label_x:
        plt.xlabel('cross section (mm)')
else:
        plt.tick_params(axis='x', which='both', labeltop='off', \
                labelbottom='off')
plt.tick_params(axis='y', which='both', labelright='off', \
        labelleft='off')
if show_title:
        plt.title('Intensity')
if np.max(Int[j]) == 0:
        plt.clim(-1,1)
else:
        plt.clim(-np.max(Int[j]), np.max(Int[j]))
plt.ylim(freqs[0], freqs[-1])
subplots+=1
binned_scaling = 3
plt.subplot(len(fields[0]), 4*binned_scaling, \
        binned_scaling*subplots-(binned_scaling-1))
plt.plot(np.asarray(int_Int[j]), freqs)
plt.xticks([])
plt.tick_params(axis='y', which='both', labelright='off', \
        labelleft='off')
if max(int_Int[j])==0:
        plt.xlim(-0.2,1)
else:
        plt.xlim(-0.1*max(int_Int[j]), max(int_Int[j])\
                +0.1*max(int_Int[j]))
plt.ylim(freqs[0], freqs[-1])
```

```
subplots+=1
counter += 1
if counter != 1:
    show_title = False
if counter == int(len(fields[0])/2):
    units_y = True
else:
    units_y = False
if counter == len(fields[0]):
```

label\_x = True

if p.u.save:

```
plt.savefig(filename+'.png', format='png', transparent=True)
```

if p.u.show:

plt.show()

plt.close()

return

def plot2dtime(fields, p, filename):

Refields = {}

```
Imfields = {}
Int = {}
int_Int = {}
for j in fields[0]:
    Refields[j] = []
    Imfields[j] = []
    Int[j] = []
    for i in range(len(p.r)):
        Refields[j].append(np.real(fields[i][j].ift(p)))
        Imfields[j].append(np.imag(fields[i][j].ift(p)))
        Int[j].append((1.33E-3)*(fields[i][j].n0)\
          *(Refields[j][i]**2 + Imfields[j][i]**2))
```

```
for i in range(len(p.freq)):
    sum = 0
    for k in range(len(p.r)):
        sum+=2*np.pi*Int[j][k][i]*p.r[k]*(p.r[1]-p.r[0])
    int_Int[j].append(sum)
```

```
plt.clf()
plt.figure(1, figsize=(12,8))
subplots = 1
```

int\_Int[j] = []

show\_title = True
units\_y = False
label\_x = False
counter = 1

```
globalmax = max(np.amax(np.asarray(Refields[j])), \
    np.amax(np.asarray(Imfields[j])), \
    abs(np.amin(np.asarray(Refields[j]))), \
    abs(np.amin(np.asarray(Imfields[j])))
```

```
if globalmax == 0:
```

pass

else:

```
globalmin = min(np.amin(np.asarray(Refields[j])), \
    np.amin(np.asarray(Imfields[j])), \
    abs(np.amax(np.asarray(Refields[j]))), \
    abs(np.amax(np.asarray(Imfields[j]))))
```

plt.subplot(len(fields[0]), 4, subplots)

```
if globalmax == 0:
```

```
plt.contourf(p.r*1000., p.time*1e15, np.asarray(Refields[j]).T,\
200, cmap='seismic', levels=np.linspace(-1,1,200))
```

else:

```
plt.contourf(p.r*1000., p.time*1e15, np.asarray(Refields[j]).T\
    /globalmax,200, cmap='seismic', \
    levels=np.linspace(-1,1,200))
```

if label\_x:

```
plt.xlabel('cross section (mm)')
```

else:

plt.tick\_params(axis='x', which='both', labeltop='off', \

```
labelbottom='off')
```

if units\_y:

plt.ylabel(j+'\ntime (fs)')

else:

```
plt.ylabel(j+'\n')
```

if show\_title:

```
plt.title('Re(E)')
```

```
plt.clim(-1,1)
plt.ylim(p.time[0]*1e15, p.time[-1]*1e15)
```

subplots+=1

```
plt.subplot(len(fields[0]), 4, subplots)
```

```
if globalmax == 0:
    plt.contourf(p.r*1000., p.time*1e15, np.asarray(Imfields[j]).T,\
        200, cmap='seismic', levels=np.linspace(-1,1,200))
```

else:

```
plt.contourf(p.r*1000., p.time*1e15, np.asarray(Imfields[j]).T\
    /globalmax,200, cmap='seismic',\
```

levels=np.linspace(-1,1,200))

if label\_x:

plt.xlabel('cross section (mm)')

else:

```
labelleft='off')
if show_title:
    plt.title('Im(E)')
plt.clim(-1,1)
plt.ylim(p.time[0]*1e15, p.time[-1]*1e15)
subplots+=1
```

if label\_x:

```
plt.xlabel('cross section (mm)')
```

else:

```
plt.tick_params(axis='y', which='both', labelright='off', \
```

labelleft='off')

if show\_title:

```
plt.title('Intensity')
```

```
if np.max(Int[j]) == 0:
```

```
plt.clim(-1,1)
```

else:

```
plt.clim(-np.max(Int[j]), np.max(Int[j]))
```

plt.ylim(p.time[0]\*1e15, p.time[-1]\*1e15)

subplots+=1

```
plt.xticks([])
plt.tick_params(axis='y', which='both', labelright='off', \
        labelleft='off')
if max(int_Int[j])==0:
       plt.xlim(-0.2,1)
else:
        plt.xlim(-0.1*max(int_Int[j]), max(int_Int[j])\
                +0.1*max(int_Int[j]))
plt.ylim(p.time[0]*1e15, p.time[-1]*1e15)
subplots+=1
counter += 1
if counter != 1:
        show_title = False
if counter == int(len(fields[0])/2):
       units_y = True
else:
        units_y = False
if counter == len(fields[0]):
        label_x = True
```

if p.u.save:

```
plt.savefig(filename+'.png', format='png', transparent=True)
```

if p.u.show:

plt.show()
plt.close()
return

```
def plot2dweighted(fields, p, filename):
```

```
Refields = {}
Imfields = {}
Int = \{\}
int_Int = {}
Int_for_int = {}
for j in fields[0]:
        Refields[j] = []
        Imfields[j] = []
        Int[j] = []
        Int_for_int[j] = []
        for i in range(len(p.r)):
                Refields[j].append(np.real(fields[i][j].E)*2*np.pi*p.r[i])
                Imfields[j].append(np.imag(fields[i][j].E)*2*np.pi*p.r[i])
                Int[j].append(fields[i][j].int()*2*np.pi*p.r[i])
                Int_for_int[j].append(fields[i][j].int())
        int_Int[j] = []
        for i in range(len(p.freq)):
                sum = 0
```

for k in range(len(p.r)):

```
sum+=2*np.pi*Int_for_int[j][k][i]*p.r[k]*(p.r[1]-p.r[0])
```

```
int_Int[j].append(sum)
```

del Int\_for\_int

```
plt.clf()
plt.figure(1, figsize=(12,8))
subplots = 1
show_title = True
units_y = False
label_x = False
counter = 1
for j in fields[0]:
        globalmax = max(np.amax(np.asarray(Refields[j])), \
                np.amax(np.asarray(Imfields[j])), \
                abs(np.amin(np.asarray(Refields[j]))), \
                abs(np.amin(np.asarray(Imfields[j]))))
        freqs = (fields[0][j].w0 + p.freq)*(1/(2.*np.pi*3.e10))
        for i in range(len(freqs)):
                if freqs[i] < 1.4227e14*(1/(2*np.pi*3.e10)):
                        freqs[i] = 0.
        if globalmax == 0:
                pass
        else:
                cmin = min(np.amin(np.asarray(Refields[j]).T/globalmax), \
                        np.amin(np.asarray(Imfields[j]).T/globalmax))
                cmax = max(np.amax(np.asarray(Refields[j]).T/globalmax), \
```

np.amax(np.asarray(Imfields[j]).T/globalmax))

```
globalmin = min(np.amin(np.asarray(Refields[j])), \
    np.amin(np.asarray(Imfields[j])), \
    abs(np.amax(np.asarray(Refields[j]))), \
    abs(np.amax(np.asarray(Imfields[j]))))
```

plt.subplot(len(fields[0]), 4, subplots)

```
if globalmax == 0:
```

else:

```
plt.contourf(p.r*1000., freqs, np.asarray(Refields[j]).T\
    /globalmax,200, cmap='seismic', \
    levels=np.linspace(-1,1,200))
```

if label\_x:

plt.xlabel('cross section (mm)')

else:

if units\_y:

plt.ylabel(j+'\nangular frequency (x10-15)')

else:

plt.ylabel(j+'\n')

if show\_title:

plt.title('Re(E)')

```
plt.clim(-1,1)
plt.ylim(freqs[0], freqs[-1])
```

```
subplots+=1
```

```
plt.subplot(len(fields[0]), 4, subplots)
```

```
if globalmax == 0:
```

else:

```
plt.contourf(p.r*1000., freqs, np.asarray(Imfields[j]).T\
    /globalmax,200, cmap='seismic', \
```

levels=np.linspace(-1,1,200))

if label\_x:

plt.xlabel('cross section (mm)')

else:

```
plt.tick_params(axis='y', which='both', labelright='off', \
```

labelleft='off')

if show\_title:

plt.title('Im(E)')

```
plt.clim(-1,1)
plt.ylim(freqs[0], freqs[-1])
subplots+=1
```

```
plt.subplot(len(fields[0]), 4, subplots)
plt.contourf(p.r*1000., freqs, np.asarray(Int[j]).T, 200, cmap='seismic')
```

if label\_x:

```
plt.xlabel('cross section (mm)')
else:
        plt.tick_params(axis='x', which='both', labeltop='off', \
                labelbottom='off')
plt.tick_params(axis='y', which='both', labelright='off', \
        labelleft='off')
if show_title:
        plt.title('Intensity')
if np.max(Int[j]) == 0:
       plt.clim(-1,1)
else:
        plt.clim(-np.max(Int[j]), np.max(Int[j]))
plt.ylim(freqs[0], freqs[-1])
subplots+=1
binned_scaling = 3
plt.subplot(len(fields[0]), 4*binned_scaling, \
        binned_scaling*subplots-(binned_scaling-1))
plt.plot(np.asarray(int_Int[j]), freqs)
plt.xticks([])
```

```
plt.tick_params(axis='y', which='both', labelright='off', \
    labelleft='off')
```

if max(int\_Int[j])==0:

```
plt.xlim(-0.2,1)
```

else:

```
plt.xlim(-0.1*max(int_Int[j]), max(int_Int[j])\
+0.1*max(int_Int[j]))
```

```
plt.ylim(freqs[0], freqs[-1])
```

```
counter += 1

if counter != 1:
    show_title = False

if counter == int(len(fields[0])/2):
    units_y = True
else:
    units_y = False
if counter == len(fields[0]):
    label_x = True
```

subplots+=1

if p.u.save:

```
plt.savefig(filename+'.png', format='png', transparent=True)
```

if p.u.show:

plt.show()

plt.close()

return

CHAPTER D	
Appendix D	

In this appendix, we present code developed to simulate the evolution of signal arising from an arbitrary polarization. This code has been implemented for operation on MacOS and other UNIX-based systems. We have sought to incorporate as many considerations in simulating signal generation as could be considered, including the phase of the driving fields and pulse propagation effects. This code consists of three scripts: parameters.py, operations.py, and main.py.

Once again, the code has been formatted for this presentation.

## D.1 parameters.py

The code in parameters.py defines a class that outlines the parameters for the system to be simulated and the functions used to calculate and determine the optical properties of the material. It should be noted that in these functions, it would be trivial to add additional materials and solvents as desired.

import numpy as np
from numpy import power as pow
import os
import matplotlib.pyplot as plt
from scipy.interpolate import interp1d
from functools import partial

class parameters:

```
def __init__(self):
```

### Options

# parameters.diagnostic is a Boolean that controls whether diagnostic plots are
# displayed during operation
 self.diagnostic = False

# parameters.smoothing is a Boolean the controls whether the loaded signal files

# are baseline subtracted and smoothed (True) or are just modified by having the # baseline subtracted (False). parameters.baseline\_percentage is a scalar value, # N, that defines all points within N% of the maximum difference from the initial # point in the loaded arrays as "baseline". parameters.boxcar is a scalar value # that defines the half-width of the smoothing boxcar, and only has an impact # when smoothing is enabled. parameters.cutoffs defines the threshold of # smoothing: all points in the arrays for which y(kx)y\*(kx) is below the cutoff # will be smoothed. parameters.smooth\_mode has been added to control whether the # smoothing is performed in the loaded domain or in the Fourier domain; it seems # that smoothing in the Fourier domain is more effective for suppressing baseline # noise, whereas smoothing in the loaded domain appears to be better when the # feature is noisy.

self.smoothing = True self.smooth\_mode = {'w1' : 'fourier', 'w2' : 'fourier', 'w3' : 'loaded'} self.baseline\_percentage = 2. self.boxcar = 8 self.cutoffs = {'w1' : 2.e-4, 'w2' : 4.e-4, 'w3' : 9.e-5}

# parameters.characteristic\_length is an estimate for the temporal width of the # pulses. This value needs to be a reasonable estimate for the shortest pulses # involved in the experiment, but does not have to be exact; it is just used to # define time/frequency grids with approximately equal sampling of the # polarization in both domains. parameters.time\_points defines the number of # points should be used in constructing the grids.

self.characteristic\_length = 75.

self.time\_points = 128

# parameters.z\_points defines the number of transverse slices that should be
# taken for the integration

self.z\_points = 701

# parameters.x\_points and parameters.y\_points defines the desired number of # points along the x and y axes of each transverse slice. The program has some # flexibility to override this if it is necessary to ensure sufficient bounds of # the transverse slices.

self.x\_points = 128
self.y\_points = 128

# parameters.absorption, parameters.gvm\_in, parameters.gvm\_out, parameters.gvd\_in
# and parameters.gvd\_out control what absorptive and dispersive properties of the
# system should be considered. The dispersion of the air is neglected, but
# the dispersion of windows and material are both considered. The GVM and GVD
# can be enabled/disabled on the input and output fields separately.

self.absorption = True
self.gvm\_in = True
self.gvm\_out = True
self.gvd\_in = True
self.gvd\_out = True
self.mono\_convolution = False

# parameters.dispersive\_k controls whether the wave vector of the fields
# should be treated according to the propagation at the carrier frequency (False)
# of that each frequency should exhibit a unique wave vector (True)

## self.dispersive\_k = False

```
# parameters.save controls whether the polarization slices are saved during
# operation. parameters.save_final controls whether the final field is saved.
# parameters.folder directs the program to the desired save location.
# Note that even ~128x128x128 slices will yield ~180MB files, and double the time
# required for the polarization_slice function to execute.
    self.save = False
```

```
self.multiprocessing = True
self.processes = 8
self.chunksize = 1
```

# Modification to the nonresonant electronic nonlinear susceptibility reported by # Levenson and Bloembergen.

self.elec\_modifier = 150000.

### Experimental Configuration

```
# Beam placement and distance of the focusing optics from the sample
self.mask = {'w1' : [-0.5, 2.0], 'w2' : [0., -2.], 'w3' : [1, 1.75]}
self.d = 24.*2.54
# Phase-matching coefficients and experimental degeneracy
self.phase_matching = {'w1' : 1, 'w2' : -1, 'w3' : 1}
self.degeneracy = 6.
# Monochromator characteristics
self.N = 300. # grooves/mm
self.f_len = 140. # mm
# Slit width
self.slits = 100. # microns
```

### Sample and window characteristics

```
# Sample path length in mm
self.L = 1.
# Window path length in mm
self.win_len = (12.5-1.)/2.
# Materials
self.sample = {'benzene' : 1.0}
self.windows = 'glass'
```

```
self.focus_shift = {'w1' : 0., 'w2' : 0., 'w3' : 0.}
self.focus_offset = {'w1' : [0., 0.], 'w2' : [0., 0.], 'w3' : [0., 0.]}
self.waist = {'w1' : 40.5, 'w2' : 23., 'w3' : 23.}
```

### Field Properties and Parameters

Parameters for file loading and processing

###

### Other:

```
self.c = 3.e8
self.L *= 1000. # Convert from mm to microns
self.win_len *= 1000. # Convert from mm to microns
# Calculate external angles
self.theta_ext = {}
```

```
for i in self.mask:
```

```
# time_freq_grids()
```

```
def spatial_grids(self):
```

```
# Calculate the half-width bounds for the x and y axes.
xy_hw = [(0.5 * i * min(self.waist.itervalues()) * np.sqrt(np.pi/i)) \
                        for i in [self.x_points, self.y_points]]
# Define the x,y grids
self.x = np.linspace(-xy_hw[0], xy_hw[0], self.x_points)
self.y = np.linspace(-xy_hw[1], xy_hw[1], self.y_points)
self.kx = np.linspace(-1/(2*abs(self.x[1] - self.x[0])),\
                1/(2*abs(self.x[1] - self.x[0])), self.x_points+1)[:-1]
self.ky = np.linspace(-1/(2*abs(self.y[1] - self.y[0])), \
                1/(2*abs(self.y[1] - self.y[0])), self.y_points+1)[:-1]
ds = [self.L*np.tan(self.theta['w4'][0]), \
                self.L*np.tan(self.theta['w4'][1]), \
                self.L]
ds = np.sqrt(np.dot(np.asarray(ds), np.asarray(ds)))
deltaz = max(np.amax(abs(self.x))*np.tan(self.theta['w4'][0]),\
               np.amax(abs(self.y))*np.tan(self.theta['w4'][1]))
self.z = np.linspace(-(0.5* + deltaz), (0.5*ds + deltaz), self.z_points)
return
```

# This method calculates the optical properties at the relevant frequencies# It adds a property to the class for each optical property, with each property# being a dictionary containing the values of those properties for each field.

```
def calculate_optical_properties(self, w = 'default'):
```

```
if w == 'default':
    del w
    w = self.w
self.n0, self.nw, self.nwin, self.nair = {}, {}, {}, {}, {}
    # indices of refraction
self.ug = {}
```

# group velocity
self.gvd, self.gvdw = {}, {}
 # group velocity dispersions
self.a0, self.aw = {}, {}
 # absorption coefficients

for i in self.w0:

# The index of refraction of the sample will be calculated as the sum of the # products of each component's refractive index and volume fraction (the # Arago-Biot model. While it seems that this model is not as accurage as the # Lorentz-Lorentz model, it is reasonably accurate and much simpler.

self.n0[i] = sum( ((self.calc\_index( ((2 \* np.pi \* 3.e17)\
 /self.w0[i]), j)\* self.sample[j]) for j in self.sample) )
nw\_temp = []
nw\_temp.extend( (sum( ((self.calc\_index(((2 \* np.pi \* 3.e17)\
 /(self.w0[i] + self.w[j])), k) \* self.sample[k]) \
 for k in self.sample))) for j in range(len(w)))
self.nw[i] = np.asarray(nw\_temp)

# The Arago-Biot model, i.e. n = sum ni\*fi where ni and fi are the index and # volume fraction of the i'th sample component, can be extended to calculate # the group velocity of a mixture. From the definition of the group velocity # ug = (dk/dw)^1: given k = nw/c, dk/dw = d/dw (nw/c). If we treat n as the # index of a mixture: dk/dw = d/dw ((sum ni\*fi) w/c) = sum d/dw ((ni\*fi) w/c) # But fi is independent of frequency, so: dk/dw = sum fi \* d/dw (ni w/c) # We can recognize d/dw (ni w/c) as dk/dw of the i'th component, so dk/dw of

```
# the mixture is the sum of fi*(dk/dw)i, or ug^-1 = sum (fi/ugi).
self.ug[i] = (1.e18)/(sum((((1/self.calc_gv(((2 * np.pi * 3.e17)\
/self.w0[i]), j))*self.sample[j]) for j in self.sample)))
```

# This method calculates the internal angles of each field

def calculate\_internal\_angles(self):

self.theta = {}

# For each key in the property describing external angles (i.e. 'w1', 'w2', 'w3')
# calculate the internal angles for each of these fields using Snell's law.

for i in self.theta\_ext:

self.theta[i] = [(np.arcsin((self.nair[i]\*\

np.sin(self.theta\_ext[i][j])) \

/self.n0[i])) for j in [0,1] ]

del i

# If a key exists in theta for 'w1', 'w2', and 'w3', calculate the internal # angles for the output field according to the phase-matching conditions.

if 'w1' in self.theta and 'w2' in self.theta and 'w3' in self.theta:

 $\ensuremath{\texttt{\#}}$  Calculate the magnitude of the k-vector for each field in 1/micron

```
kmag = {}
for i in self.w0:
    kmag[i] = (self.w0[i]*self.n0[i]/(3.e8))*1.e-6
    # in 1/micron
```

del i

# Find thetax and thetay for the output field that satisfy the relationships: # k4x = c1\*k1x + c2\*k2x + c3\*k3x

# k4y = c1\*k1y + c2\*k2y + c3\*k3y

# This may not necessarily be the exact solution since it neglects kz but # hopefully should be close for most cases.

sums = [sum( ((self.phase\_matching[i]\*kmag[i]\
 \*np.sin(self.theta[i][j])) \
 for i in self.theta)) for j in [0,1]]
self.theta['w4'] = [ (np.arcsin(sums[i]/kmag['w4'])) \
 for i in [0,1]]

# Use the calculated angles to calculate the actual k-vectors

k = {}

for i in self.theta:

k[i] = kmag[i]\*np.asarray([np.sin(self.theta[i][0]),

np.sin(self.theta[i][0]), np.sqrt(1 \

```
# Calculate the phase-mismatch as k4' - k4
```

else:

return

def calc\_index(self, wavelength, material):

```
# while nm is taken as the default for wavelengths, most forms of the Selmeier
# or Cauchy's equations uses microns
# convert from nm to micron by dividing by 1000
wl = wavelength/1000.
```

```
elif material == 'glass':
```

```
# material_properties.gv(WAVELENGTH, MATERIAL)
# Takes WAVELENGTH (in nm) and the name of a material (as a string)
# Returns the group velocity of MATERIAL at WAVELENGTH in units mm/fs
# Group velocity is dw/dk, or (dk/dw)^-1
```

def calc\_gv(self, wavelength, material):

# Defined as (dw/dk) (or  $(dk/dw)^{(-1)}$ )

wl = wavelength/1000.

```
- 6*e*pow(wl, -7))
```

```
elif material == 'glass':
                a1, b1 = 1.03961212, -0.00600069867
                a2, b2 = 0.231792344, -0.0200179144
                a3, b3 = 1.01046945, -103.560653
                dndl = 0.5*pow(self.calc_index(wavelength, material), -1) \
                                * sum(((2 * i * wl * pow((pow(wl,2) + j), -1) \
                                - 2 * i * pow(wl,3) * pow((pow(wl,2) + j),-2)) \
                                for i,j in [(a1,b1), (a2,b2), (a3,b3)]))
        elif material == 'air':
                a1, b1 = 0.05792105, 238.0185
                a2, b2 = 0.00167917, 57.362
                dndl = sum( ((-2 * i * pow(wl,-3) * pow((j - pow(wl,-2)), -2)) \
                                                for i,j in [(a1,b1), (a2,b2)]))
        return self.c * pow((self.calc_index(wavelength, material) - wl*dndl), \
                -1)*1.e-12
                        # in mm/fs
# parameters.gvd(WAVELENGTH, MATERIAL)
# Takes WAVELENGTH (in nm) and the name of a material (as a string)
# Returns the group velocity dispersion of MATERIAL at WAVELENGTH
# in units fs<sup>2</sup>/mm
# Group velocity dispersion is d^2k/dw^2
def calc_gvd(self, wavelength, material):
        wl = wavelength/1000.
        if material == 'benzene':
                a, b, c, d, e = 2.170184597, 0.00059399, 0.02303464,
                        -0.000499485, 0.000178796
                dndl = 0.5*pow(self.calc_index(wavelength, material), -1) \
```

```
* (2*b*wl - 2*c*pow(wl, -3) - 4*d*pow(wl, -5) \
                        - 6*e*pow(wl, -7))
        d2ndl2 = 0.5*pow(self.calc_index(wavelength, material), -1) \
                        * (2*b + 6*c*pow(wl, −4) + 20*d*pow(wl, −6) \
                        + 42 \times (wl, -8)) + -0.5
                        *pow(self.calc_index(wavelength, material), -2) \
                        * dndl * (2*b*wl - 2*c*pow(wl, -3) \
                        - 4*d*pow(wl, -5) - 6*e*pow(wl, -7))
elif material == 'glass':
        a1, b1 = 1.03961212, -0.00600069867
        a2, b2 = 0.231792344, -0.0200179144
        a3, b3 = 1.01046945, -103.560653
        dndl = 0.5*pow(self.calc_index(wavelength, material), -1) \
                        * sum(((2 * i * wl * pow((pow(wl,2) + j), -1) \
                        - 2 * i * pow(wl,3) * pow((pow(wl,2) + j),-2)) \
                        for i,j in [(a1,b1), (a2,b2), (a3,b3)]))
        d2ndl2 = 0.5 * pow(self.calc_index(wavelength, material), -1) \
                        * sum(((2 * i * pow((pow(wl,2) + j),-1) \
                        - 10 * i * pow(wl,2) * pow((pow(wl,2) + j),-2) \
                        + 8 * i * pow(wl,4) * pow((pow(wl,2) + j),-3)) \
                        for i,j in [(a1,b1), (a2,b2), (a3,b3)])) \
                        - 0.5*pow(self.calc_index(wavelength, material),
                        -2) * dndl * sum( ((2 * i * wl * pow((pow(wl,2) \
                        + j),-1) - 2 * i * pow(wl,3) \
                        * pow((pow(w1,2) + j),-2)) \
                        for i,j in [(a1,b1), (a2,b2), (a3,b3)]))
elif material == 'air':
        a1, b1 = 0.05792105, 238.0185
        a2, b2 = 0.00167917, 57.362
        d2ndl2 = sum(((6 * i * pow(wl,-4) * pow((j - pow(wl,-2)),-2))
                        + 8 * i * pow(wl,-6) * pow((j - pow(wl,-2)),-3))
                        for i,j in [(a1,b1), (a2,b2)]))
```

return (((wl)\*\*3)/(2\*np.pi\*(self.c\*\*2))\*d2ndl2)\*(1e21) #fs2/mm

# parameters.alpha(WAVELENGTH)

# Takes WAVELENGTH (in nm)

# Returns the absorbance coefficient of the sample at WAVELENGTH in units 1/cm

def alpha(self, wavelength):
 return 0.

# parameters.chi(p, FREQUENCY, INDICES)

# Takes angular frequency (in 1/s)

# Currently set to return the non-resonant electronic and Raman

# response of benzene's 992cm-1 mode in a CARS experiment

# in which (w1-w2) prepare the vibrational coherence

# Returns the chi3 value in units of m2/V2

def calc\_chi(self, freq, material):

if material == 'benzene':

# Convert to the frequency shift in wavenumbers from frequency of w3
wn = ((freq - self.w0['w3'])/(2\*np.pi))\*(1/3.e10)

# Terms from Levenson and Bloembergen, J Chem Phys 60 1323 with a
# modifier to increase the electronic response under fs excitation. Note

- $\ensuremath{\texttt{\#}}$  that we neglect the imaginary contribution described by Levenson and
- # Bloembergen. (They do not give a fixed value, and it is >200 times

chi\_vib = 1.74e-35 \* (992./((992.\*\*2) - (wn\*\*2) + 1j\*2.\*wn\*1.15))

# Local field enhancement factor

- # Chi3 in cgs units; factor of 0.5 accounts for difference in degeneracy chi\_cgs = 6.75e21\*(chi\_elec+chi\_vib)\*F\*0.5
- # Convert chi3 from cgs units to SI units

return chi\_cgs\*4\*np.pi\*(1.e-4)\*((299.79)\*\*(-2))

## D.2 operations.py

The second script associated with these calculations, operations.py, contains many of the functions necessary for proper execution.

import numpy as np
from numpy import power as pow
import os, itertools, datetime
import matplotlib.pyplot as plt
from scipy.interpolate import interp1d
from functools import partial
from itertools import product

# operations.load\_field (FOLDER, FILE, COLUMNS)

# Takes two strings (FOLDER and FILE) and a 2-element list (COLUMNS). FOLDER and FILE # identify the location and name of the file containing the raw tuning data for one # field. COLUMNS is a two-element list that specifies the indices (with indexing starting # at 0) of the columns containing the data for the x (COLUMNS[0]) and y (COLUMNS[1]) data. def load\_field(folder, file, columns):

# Initialize the arrays.

x\_axis, y\_axis = [], []

# Open the target file.

f = open(folder + '/' + file)

# Load each line and append the values in the appropriate columns to x and y arrays. for line in f:

x\_axis.append(float(line.split('\t')[columns[0]]))
y\_axis.append(float(line.split('\t')[columns[1]]))

f.close()

del f, line

```
# Return the loaded files as a tuple of arrays (x,y)
```

```
return np.asarray(x_axis), np.asarray(y_axis)
```

```
# Initialize the matplotlib figure and axes objects necessary to ensure proper operation
```

# for diagnostics enabled or disabled.

def initialize\_diagnostics(modes, diagnostic, smoothing):

```
# Initialize a matplotlib figure object and a dictionary for storing the axes
```

```
fig1, fig2 = plt.figure(), plt.figure()
```

 $axes1, axes2 = \{\}, \{\}$ 

# If diagnostics are not enabled, just populate axes1 and axes2 as dummy variables.

if not diagnostic:

```
for i in ['w1', 'w2', 'w3']:
```

axes1[i] = [None,None]
axes2[i] = [None,None]

# Otherwise, populate axes according to whether smoothing is enabled or not.

else:

## return fig1, axes1, fig2, axes2

# operations.smooth(X,Y,CUTOFF, BOXCAR, BASE\_PER, DIAG, AX, MODE)
# Takes two numpy arrays (X,Y), three scalar values (CUTOFF, BOXCAR, and BASE\_PER), a
# boolean (DIAG), and an axis object from matplotlib.pyplot (only necessary if
# DIAG is True). Performs baseline subtraction and smoothing on the Y array. CUTOFF
# specifies the threshold under which smoothing is performed, and BOXCAR specifies the
# limit for the half-width of the smoothing boxcar. BASE\_PER specifies the percentage
# used to define which points in the Y array constitute "baseline". DIAG indicates
# the routine making the function call is performing diagnostic plotting; if True, smooth
# requires an axis object on which it can plot the results of processing.
def smooth(x, y, cutoff, boxcar= 6, base\_per= 2., diag= False, ax=None, mode='fourier'):
# Baseline subtraction.

y = baseline\_subtract(y = y, perc = base\_per)

# If the smoothing mode is to operate in the Fourier domain:

if mode == 'fourier':

len(x)+1)[:-1]

# Perform the 1DFT

 $y_ft = (1/np.sqrt(len(x))) * (y[:,None] \setminus$ 

\* np.exp(1j\*2\*np.pi\*x[:,None]\*kx[None,:])).sum(axis=0)

# If diagnostics are enabled, plot y(kx)

if diag:

ax.plot(kx, y\_ft\*np.conj(y\_ft), 'k-')
ax.set\_yscale('log')

# Initialize the smoothed array as a copy of the original array.

y\_ft\_sm = y\_ft.copy()

# Only perform smoothing for points where y(kx)y\*(kx) are below the cutoff. It

# should be noted that this algorithm assumes a feature in the middle of the data
# with continuous background that needs to be smoothed.

for i in np.where(y\_ft\*np.conj(y\_ft) < cutoff)[0]:</pre>

# Define initial averaging window as the span of i +/- boxcar.

# If lower < 0, the boxcar reaches the beginning of the array; in this case, only
# average over the points up to the upper bound.</pre>

if lower < 0:

y\_ft\_sm[i] = np.mean(y\_ft[:upper])

# If upper extends the range of the boxcar beyond the end of the array, the # average is taken from the lower bound to the end of the array, incorporating # the first point, due to the symmetry of the FT.

elif upper >  $(len(y_ft) - 1):$ 

y\_ft\_sm[i] = np.mean(np.concatenate((y\_ft[lower:],\

y\_ft[:1])))

# Otherwise, perform the average between upper and lower. However, so we are # only averaging out noise, we check to make sure that the bounds of the boxcar do # not significantly extend into the main features of the array; if the bounds do # extend into the main features of the array (i.e. those above the cutoff level), # the bounds are gradually decreased as the index approaches the main feature(s).

else:

```
if check2 > boxcar:
                                                 break
                                upper += 2
                                lower += -2
                                y_ft_sm[i] = np.mean(y_ft[lower:upper])
# If diagnostics are enabled, plot the smoothed y(kx) data.
                if diag:
                        ax.plot(kx, y_ft_sm*np.conj(y_ft_sm), 'r-')
# Perform the 1DFT to convert the smoothed y(kx) to a smoothed y(x)
                y_sm = (1/np.sqrt(len(kx))) * (y_ft_sm[:,None] \setminus
                        * np.exp(-1j*2*np.pi*kx[:,None]*x[None,:])).sum(axis=0)
                del y_ft, y_ft_sm
# Ensure the smoothed array is real-valued.
                y_sm = np.real(np.sqrt(y_sm*np.conj(y_sm)))
# Otherwise, just perform smoothing on the background subtracted raw data.
        elif mode == 'loaded':
                y_sm = np.empty(np.shape(y))
                for i in range(len(y)):
                        if i < boxcar:
                                y_sm[i] = np.mean(y[:i+boxcar+1])
                        elif i > (len(y) - boxcar - 1):
                                y_sm[i] = np.mean(y[i-boxcar:])
                        else:
                                y_sm[i] = np.mean(y[i-boxcar:i+boxcar+1])
# Since the Fourier smoothing algorithm will yield all values greater than or equal to
# zero, we will perform one more baseline subtraction, then return the result.
        return baseline_subtract(y = y_sm, perc = base_per)
```

# Takes a numpy array Y and a scalar value PERC (corresponding to the desired cutoff # percentage). Averages all points within PERC% of the maximum deviation from the

# operations.baseline\_subtract(Y, PERC)

# initial point in Y, subtracts this value from the array, and returns the result. def baseline\_subtract(y, perc, \*\*kwargs):

baseline = np.mean(y[np.where(abs(y-y[0]) < (perc/100.)\</pre>

\*np.amax(abs(y - y[0])))[0]])

return y - baseline

```
# operations.center_freq(GUESS, W, WL, SIG)
```

# Takes a scalar value (GUESS) and three arrays(W, WL, SIG). WL and SIG define the # spectrum of the field in question. Starting from the GUESS of the central frequency, # center\_freq interpolates SIG onto a regular frequency grid, W (centered at 0), and # calculates the true central frequency and returns the result.

def center\_freq(guess, w, wl, sig):

```
# Calculate an angular frequency grid centered at guess from the wavelength grid
```

f = 2\*np.pi\*(3.e17/wl) - guess

# Generate a model for interpolating the signal onto an arbitrary frequency grid.

if f[1] > f[0]:

model = interp1d(f, sig, bounds\_error = False, fill\_value = 0.)
else:

model = interp1d(f[::-1], sig[::-1], bounds\_error = False, \
 fill value = 0.)

# Calculate the central frequency sum(I\*w)/sum(I)

w0 = ((((model(w))\*(w + guess)).sum())/((model(w)).sum()))

# Ensure the value is real-valued, and return it

return float(np.real(np.sqrt(w0\*np.conj(w0))))

```
# operations.grid_spectrum(WO, W, WL, SIG)
```

# Takes a scalar value (WO) and three arrays (W, WL, SIG). WL and SIG define the # spectrum of a field that may not necessarily be gridded regularly. grid\_spectrum # takes the WL grid and converts it to a frequency grid centered at 0 (where the # frequency is WO). A 1D interpolation is then performed to map the SIG array # onto the provided frequency grid, W. def grid\_spectrum(wO, w, wl, sig): # Calculate an angular frequency grid from the loaded wavelength grid

f = (2\*np.pi\*(3.e17/wl)) - w0

# Generate a model for interpolating the signal onto an arbitrary frequency grid.
 if f[1] > f[0]:

model = interp1d(f, sig, bounds\_error = False, fill\_value = 0.)
else:

model = interp1d(f[::-1], sig[::-1], bounds\_error = False, \
 fill\_value = 0.)

# Return the interpolation of the raw data onto the provided frequency grid.

return model(w)

# operations.calculate\_field(X\_AX, SIG, NO, WAIST, ENERGY, T\_OR\_W, diagnostic, returnt)
# Takes two arrays (X\_AX and SIG), three scalars (NO, WAIST, and ENERGY), a
# string (T\_OR\_W), and a boolean (diagnostic). SIG is the intensity-level spectrum on
# the X\_AX grid, which should either be an angular frequency grid in 1/s or a time grid
# in femtoseconds. T\_OR\_W informs whether X\_AX is a frequency grid (T\_OR\_W='frequency')
# or a time grid (T\_OR\_W = 'time'). NO, WAIST, and ENERGY are used to determine the
# proper scaling factor for converting SIG to an electric field gridded onto X\_AX. If
# diagnostic is True, the function will also print out a line comparing the energy
# calculated with the resulting scaling factor to the provided energy. returnt allows the
# user to request calculate\_field to return the internal t grid.

# The electric field is returned in units of V/micron\*s (the per second is essentially
# per unit frequency/per unit time).

def calculate\_field(x\_ax,sig,n0,waist,energy,t\_or\_w,perc,diagnostic=False,returnt=False):

if t\_or\_w == 'frequency':

# If x\_ax is frequency, define a time grid and perform the 1DFT to get I(t)

t = np.linspace(-np.pi/(abs(x\_ax[1]-x\_ax[0])), np.pi/(abs(x\_ax[1]

-x\_ax[0])), len(x\_ax)+1)[:-1]

intensity = (1/np.sqrt(len(x\_ax))) \* (sig[:,None] \

\* np.exp(1j\*x\_ax[:,None]\*t[None,:])).sum(axis=0)

# Ensure I(t) is real-valued with a mean baseline of 0.

```
intensity = np.real(np.sqrt(intensity*np.conj(intensity)))
                intensity = baseline_subtract(y = intensity, perc = perc)
                t *= 1.e15
                                  # convert the resulting time axis to fs
        elif t_or_w == 'time':
# If x_ax is in time, only need to copy the x_ax and sig variables.
                t = x_ax
                intensity = sig.copy()
# Define temporary x,y grids
        xy_hw = 64.*waist*np.sqrt(np.pi/128.)
        x,y = np.linspace(-xy_hw, xy_hw, 128), np.linspace(-xy_hw, xy_hw, 128)
# Calculate the amplitude spectrum.
        amp = np.where(intensity < 0, -np.sqrt(-intensity), np.sqrt(intensity))</pre>
# Define the electric field at the waist E(t,x,y)
        E = amp[:,None,None] \
                * np.exp(-(((x[None,:,None])**2) + ((y[None,None,:])**2))/(waist**2))
# Calculate the integral of EE* over t,x,y, i.e. total energy for current amp
        integral = (1.33e-3) * n0 * ((((np.real(E*np.conj(E)) * abs(t[1] - t[0]) \
                        * 1.e-15 * abs(x[1] - x[0]) * 1.e-6 * abs(y[1] - y[0]) \setminus
                        * 1.e-6).sum(axis = 2)).sum(axis = 1)).sum(axis = 0))
        del E
# If diagnostic is true, the provided energy and integrated energy will be printed.
        if diagnostic:
                E = (amp*np.sqrt(energy/integral))[:,None,None] \
                        * np.exp(-(((x[None,:,None])**2) \
                        + ((y[None,None,:])**2))/(waist**2))
                integral2 = (1.33e-3) * n0 * ((((np.real(E*np.conj(E)) \
                                * abs(t[1] - t[0]) * 1.e-15 * abs(x[1] - x[0]) \
                                * 1.e-6 * abs(y[1] - y[0]) \
                                * 1.e-6).sum(axis = 2)).sum(axis = 1)).sum(axis = 0))
                print 'Calculated energy: {0}; Reported energy: {1}'\
                        .format(integral2, energy)
# Return the normalized amplitude-level signal and - if requested - the time grid
```

```
if returnt:
```

```
return amp*np.sqrt(energy/integral)*1.e-6, t
```

else:

```
return amp*np.sqrt(energy/integral)*1.e-6
```

# The factor of 1.e-6 returns the field in unis of V/micron instead of V/m.

# operations.field\_slice is a function called by polarization\_slice

# \*\*kwargs is expected to provide values associated with specific keywords, so it is not # advised to use this function directly without reviewing the behavior.

def field\_slice(\*\*kwargs):

# Generate rotation matrices to rotate by thetax around the y axis (A) and by -thetay
# around the x-axis (B) (the negative is required due to sign conventions).

A = np.asarray([[np.cos(kwargs['theta'][0]), 0, np.sin(kwargs['theta'][0])],\

 $[0,1,0], \$ 

[-np.sin(kwargs['theta'][0]), 0, np.cos(kwargs['theta'][0])]])

 $B = np.asarray([[1,0,0], \])$ 

[0, np.cos(-kwargs['theta'][1]), -np.sin(-kwargs['theta'][1])],

[0, np.sin(-kwargs['theta'][1]), np.cos(-kwargs['theta'][1])]])

# Calculate the field's local coordinates [x',y',z'] as A (dot) B (dot) [x,y,z]

xp = np.empty(np.shape(kwargs['x']))

yp = np.empty(np.shape(kwargs['x']))

zp = np.empty(np.shape(kwargs['x']))

# For whatever reason, it proved necessary to calculate the product elementwise.

for i,j in product(range(np.size(kwargs['x'],axis=0)),\

range(np.size(kwargs['x'],axis=1))):

 $[xp[i,j], yp[i,j], zp[i,j]] = np.dot(A, np.dot(B, \$ 

np.asarray([kwargs['x'][i,j], kwargs['y'][i,j], \

kwargs['z'][i,j]])))

# For linear effects, we need the pathlength from the start of the sample.

```
[x0,y0,z0] = np.dot(A,np.dot(B,np.asarray([kwargs['x'],kwargs['y'],\
```

-kwargs['L']/2])))

dzp = zp - z0 del x0, y0, z0

# Apply the offsets and shift to the sample coordinates.

# For 'long' pulses, we can calculate the field in time, since dispersion is minor. # We also do not need to be concerned about the separability of A and psi, as # neglecting dispersion and treating absorption only according to w0 removes the # frequency dependence of psi.

if kwargs['rel\_length'] == 'long':

# The spatial effects of TEM00 mode propagation are calculated.

# Absorptive effects are applied according to the absorption at w0.

if kwargs['absorption']:

E \*= np.exp(-0.5\*kwargs['a0']\*dzp[None,:,:])

# For 'short' pulses, dispersive effects need to be applied in the frequency domain.

elif kwargs['rel\_length'] == 'short':

# 1DFT of E(t) to E(w)

E = (1/np.sqrt(len(kwargs['t']))) \* (kwargs['Et'][:,None] \

\* np.exp(-1j\*kwargs['t'][:,None]\*1.e-15\

```
*kwargs['w'][None,:])).sum(axis=0)
```

# Calculation of the spatial effects of TEMO0 mode propagation

```
Ew = E[:,None,None] * (1/np.sqrt(1 + ((zp[None,:,:]/z0)**2))) \
 * np.exp(-(((xp[None,:,:])**2) + ((yp[None,:,:])**2))\
 /((wzp[None,:,:])**2)) * np.exp(1j\
 *np.arctan(zp[None,:,:]/z0)) \
 * np.exp(-1j*(((xp[None,:,:])**2) + ((yp[None,:,:])**2))\
 *(np.pi/wl) * Rzpinv[None,:,:])
```

# Apply the effects of longitudinal phase

```
if kwargs['disp']:
```

Ew \*= np.exp(-1j \* ((kwargs['nw'][:,None,None] \* (kwargs['w0']\

else:

del E

# Absorptive effects are applied, accounting for the wavelength-dependence.

```
if kwargs['absorption']:
```

Ew \*= np.exp(-0.5 \* kwargs['aw'][:,None,None] \* dzp[None,:,:])

# GVM effects are applied.

if kwargs['gvm\_in']:

Ew \*= np.exp(-1j \* ((1/kwargs['ug']) - (1/kwargs['ugr'])) \

\* kwargs['w'][:,None,None] \* dzp[None,:,:])

# GVD effects are applied, accounting effects of both the sample and the window.

if kwargs['gvd\_in']:

if kwargs['win\_len'] > 0:

```
thetaw = []
                        for i in [0,1]:
                                thetaw.append(np.arcsin((kwargs['n0'] * \
                                        np.sin(kwargs['theta'][i]))\
                                        /kwargs['nwin']))
                        del i
                        d = [kwargs['win_len'] * np.tan(thetaw[0]), \
                                        kwargs['win_len'] * np.tan(thetaw[1]), \
                                        kwargs['win_len']]
                        d = np.sqrt(np.dot(np.asarray(d), np.asarray(d)))
                else:
                        d = 0.
                Ew *= np.exp(-1j * 0.5 * kwargs['gvd'] \
                        * ((kwargs['w'][:,None,None])**2) \
                        * dzp[None,:,:]) * np.exp(-1j * 0.5 * kwargs['gvdw'] \
                        * ((kwargs['w'][:,None,None])**2) * d)
        # Perform the 1DFT to get E(t,x,y,z) from E(w,x,y,z).
        E = np.empty(np.shape(Ew), dtype=complex)
        for i in range(np.size(Ew, 2)):
                E[:,:,i] = (1/np.sqrt(len(kwargs['w']))) * (Ew[:,None,:,i] \
                        * np.exp(1j * kwargs['w'][:,None,None] \
                        * kwargs['t'][None,:,None] * 1.e-15)).sum(axis=0)
return E
                # Return the 3D E(t,x,y,z)
```

def polarization\_slice(z, \*\*kwargs):

# Rotation matrices calculate x',y',z' (z' \_|\_ the sample) from x,y,z (z||k4); since # w4's internal angles are measured relative to sample normal, need their negatives. # A is the rotation matrix around the y-axis by an angle of -thetax, while B is the # rotation matrix around the x-axis by an angle of -thetay (actually need --thetay). A = np.asarray([[np.cos(-kwargs['theta']['w4'][0]), 0,  $\$ 

np.sin(-kwargs['theta']['w4'][0])],\

```
[0, 1, 0], \
```

```
[-np.sin(-kwargs['theta']['w4'][0]), 0, \
```

np.cos(-kwargs['theta']['w4'][0])]])

 $B = np.asarray([[1,0,0], \])$ 

```
[0, np.cos(kwargs['theta']['w4'][1]), \
```

```
-np.sin(kwargs['theta']['w4'][1])],\
```

```
[0, np.sin(kwargs['theta']['w4'][1]), \
```

np.cos(kwargs['theta']['w4'][1])]])

# Apply the rotation matrices to calculate the sample coordinates x',y',z'. These are # 2D arrays with dimensions inherited from the transverse slice's x and y coordinates.

[xp, yp, zp] = np.dot(A,  $\setminus$ 

np.dot(B, np.asarray([kwargs['x'][:,None],\

kwargs['y'][None,:], z])))

# Calculate the TEMOO mode for each field. This employs a temporary function that # reduces the function call to a single field identifier 'fld'.

```
lam = lambda fld: field_slice(x = xp, y = yp, z = zp, t = kwargs['t'], \
    w = kwargs['w'], theta = kwargs['theta'][fld], \
    focus_offset = kwargs['focus_offset'][fld], \
    focus_shift = kwargs['focus_shift'][fld], L = kwargs['L'], \
    w0 = kwargs['w0'][fld], waist = kwargs['waist'][fld], \
    n0 = kwargs['n0'][fld], nw = kwargs['nw'][fld], \
    rel_length = kwargs['rel_length'][fld], \
    Et = kwargs['Et'][fld], a0 = kwargs['a0'][fld], \
    ugr = kwargs['aw'][fld], ug = kwargs['gvd'][fld], \
    win_len = kwargs['win_len'], nwin = kwargs['nwin'][fld], \
    gvdw = kwargs['gvdw'][fld], absorption = kwargs['absorption'], \
    disp = kwargs['disp'])
```

E1 = lam('w1')

```
E2 = lam('w2')
        E3 = lam('w3')
        del lam
# Multiply the time-domain electric fields according to the phase-matching scheme.
       prod = 1.
       for i,j in [('w1', E1), ('w2', E2), ('w3',E3)]:
                if kwargs['phase_matching'][i] > 0:
                        prod *= (j**kwargs['phase_matching'][i])
                elif kwargs['phase_matching'][i] < 0:</pre>
                        prod *= (np.conj(j)**abs(kwargs['phase_matching'][i]))
        del i, j#, E1, E2, E3
# Perform the 1DFT to take the product from the time domain to the frequency domain.
        P = np.empty(np.shape(prod), dtype=complex)
        for i in range(np.size(prod, axis=2)):
                P[:,:,i] = (1/np.sqrt(len(kwargs['t']))) * (prod[:,None,:,i] \
                                * np.exp(1j * kwargs['t'][:,None,None] * 1.e-15\
                                * kwargs['w'][None,:,None])).sum(axis=0)
        del prod, i
        Pk = np.empty(np.shape(P), dtype=complex)
        for i in range(len(kwargs['w'])):
# Multiply the field product by the nonlinear susceptibility and degeneracy.
                P[i,:,:] = kwargs['degeneracy'] * np.where(zp <= kwargs['L']/2., \</pre>
                                np.where(zp >= -kwargs['L']/2., kwargs['chi'][i],\
                                complex(0.)), complex(0.)) * P[i,:,:]
# Perform the 1DFT from x --> kx, generating the 2D array temp(y,kx)
                temp = (1/np.sqrt(len(kwargs['x']))) * (P[i,:,:,None] \
                                * np.exp(-1j * 2 * np.pi * kwargs['x'][:,None,None] \
                                * kwargs['kx'][None,None,:])).sum(axis=0)
# Populate the ith 2D slice of Pk with the result of the 1DFT of the array temp,
# yielding the final slice in (kx,ky)
```
del temp

del i

# If the polarization slices are to be saved, this is accomplished here.

if kwargs['save']:

# Save the current directory so that the program can return to it.

curr\_dir = os.getcwd()

# If the subdirectories don't exist, create them.

if not os.path.exists((kwargs['folder']+'/slices/')\

.split(os.path.commonprefix( \

[os.getcwd(), kwargs['folder']])+'/')[-1]):

os.makedirs((kwargs['folder']+'/slices/')\

.split(os.path.commonprefix(\

[os.getcwd(), kwargs['folder']])+'/')[-1])

# Change to the save directory

os.chdir(kwargs['folder']+'/slices')

# Create the file and write the array

```
f = open('slice_at_z_{0}_2.txt'.format(z), 'w')
```

f.write('Frequency (Hz) \t x-pos (microns) \

\t y-pos (microns) \t Polarization \n')

f.writelines('{0}\t{1}\t{2}\t{3}\n'.format(kwargs['w'][i[0]]\

+kwargs['w0']['w4'], kwargs['x'][i[1]],\

kwargs['y'][i[2]], P[i[0], i[1], i[2]]) \

for i in itertools.product(range(len(kwargs['w'])), \

```
range(len(kwargs['x'])), range(len(kwargs['y']))))
```

# Close the file and return to the original directory.

f.close()
del f
os.chdir(curr\_dir)

# To calculate the contribution of this polarization slice to the output field, it

# will be more convenient to have the wave vector magnitude available.

```
k0 = ((kwargs['n0']['w4']*kwargs['w0']['w4'])/(3.e14))
```

# Now, multiply the polarization by the various relevant contributions:

# Multiply the polarization by the longitudinal phase term, exp(ikz)

if kwargs['disp']:

Pk \*= np.exp(1j \* ((kwargs['nw']['w4'][:,None,None] \

\* (kwargs['w0']['w4'] + kwargs['w'][:,None,None]))/(3.e14)) \* z)

else:

Pk \*= np.exp(1j \*((kwargs['n0']['w4']\*kwargs['w0']['w4'])/(3.e14)) \* z)

# Multiply the polarization by the frequency, speed of light, and wavenumber

Pk \*= -1j\* ((kwargs['w0']['w4'] + kwargs['w'][:,None,None])\*\*2) \

\* (1/(2 \* ((3.e14)\*\*2) \* k0))

# Multiply the polarization by the exponential accounting for transverse evolution of # the plane wave.

# Now, we will multiply the polarization by the various terms describing the # dispersive character of the medium, if enabled:

if kwargs['gvm\_out']:

Pk \*= np.exp(-1j \* ((1/kwargs['ug']['w4']) - (1/kwargs['ug']['w3'])) \

\* kwargs['w'][:,None,None] \* (kwargs['z\_end'] - z))

if kwargs['gvd\_out']:

Pk \*= np.exp(-1j \* 0.5 \* kwargs['gvd']['w4'] \

\* ((kwargs['w'][:,None,None])\*\*2) \* (kwargs['z\_end'] - z))

if kwargs['absorption']:

Pk \*= np.exp(-0.5 \* (kwargs['aw']['w4'][:,None,None]) \

\* (kwargs['z\_end'] - z))

def intensity(E, mono\_conv, w, t, N, f, slits, n0, w0):

# Intensity of an electric field when the field is in mks units

I = 1.33e-3 \* n0 \* E \* np.conj(E)

if mono\_conv:

rec\_lin\_disp = (1./N)\*(1./f)\*1.e-3

# reciprocal linear dispersion, nm/microns

# Determine the wavelength grid corresponding to the frequency grid.

dwl = ((2\*np.pi\*3.e17)/((w0)\*\*2))\*w

# Define the instrument function of the monochromator as a triangle function.

(1/(rec\_lin\_disp\*slits))\*abs(dwl)), 0.)

# Since convolving functions in a domain is equivalent to multiplication of their

# Fourier transforms, 1DFT both the intensity and instrument functions

monot = (1/np.sqrt(len(w))) \* (mono[:,None] \* np.exp(1j \* w[:,None] \

\* t[None,:] \* 1.e-15)).sum(axis=0)

# Convert the Fourier transforms to real-valued functions.

It,monot = np.real(np.sqrt(It\*np.conj(It))),\

np.real(np.sqrt(monot\*np.conj(monot)))

# Multiply the intensity by the instrument function.

It \*= np.sqrt(len(w))\*monot

# FT the result back to the frequency domain.

del It, monot, mono, dwl, rec\_lin\_disp

# Return the real-valued result.

```
return np.real(np.sqrt(I*np.conj(I)))
```

def save\_output(E, E\_slice, I, folder, file, w, x, y):
# Save the 3D array E(w, x, y) and integrated 1D slice E(w) to a file.

```
curr_dir = os.getcwd()
# If the target save directory doesn't exist, create it and any higher directories
        if not os.path.exists(folder.split(os.path.commonprefix( \
                                        [os.getcwd(), folder])+'/')[-1]):
                os.makedirs(folder.split(os.path.commonprefix( \
                                        [os.getcwd(), folder])+'/')[-1])
# Navigate to the target directory
       os.chdir(folder)
       now = datetime.datetime.now()
# Format file names
       file1d, file3d = file + '_1d', file + '_3d'
       for i in str(now).replace('-',' ').replace(':',' ').split('.')[0].split(' '):
                file1d, file3d = file1d + '_' + i, file3d + '_' + i
       file1d, file3d = file1d+'.txt', file3d+'.txt'
       # Write data to files
       for i,j in [(file1d, '1d'), (file3d,'3d')]:
                f = open(i, 'w')
                if j == '3d':
                        f.write('Frequency (Hz)\tx-pos (microns)\ty-pos (microns)\
                                tField (V/m) n'
                        f.writelines('{0}\t{1}\t{2}\t{3}\n'.format(w[i[0]], x[i[1]],\
                                 y[i[2]], E[i[0], i[1], i[2]]*1.e6) for i in \
                                 itertools.product(range(len(w)), \
                                range(len(x)), range(len(y))))
                if j == '1d':
                        f.write('Frequency (Hz) \t Field (V/m) \t Power \n')
                        f.writelines('{0}\t{1}\t{2}\n'.format(w[i], E_slice[i], I[i]) \
                                        for i in range(len(w)))
                f.close()
                del f
        # Return to the original directory, and return to the function call
       os.chdir(curr_dir)
```

## D.3 main.py

The final script - main.py - initializes an instance of the class from parameters.py, then calls the various functions from operations.py as necessary. For different applications, this script may need to be heavily modified according to the desired fields.

```
import numpy as np
from numpy import power as pow
import os, datetime, itertools
import matplotlib.pyplot as plt
from scipy.interpolate import interp1d
from functools import partial
from multiprocessing import Pool
import parameters, operations
reload(parameters), reload(operations)
def main(p = 'default', ret_I = False):
# Clear and close existing files; ensure interactive mode is on
        plt.clf(), plt.close(), plt.ion()
# Initialize an object of the parameters class.
        if p == 'default':
                p = parameters.parameters()
# Define a temporary function for loading files, which calls operations.load_field,
# but reduces the call to a single variable, a string that serves as field identifier.
        lam = lambda w: operations.load_field(folder = p.folder, \
                        file = p.files[w][p.colors[w]], columns = p.columns[w])
        w1_col, w1_sig = lam('w1')
        w2_col, w2_sig = lam('w2')
```

```
w3_del, w3_sig = lam('w3')
```

del lam

```
# I know that my w3_sig data is a negative signal, so I multiply by a factor of -1. w3_sig *= -1
```

# Initialize matplotlib figure and axes objects for diagnostic plots.

# Generate a temporary function for data processing, whether this is just baseline # subtraction or baseline subtraction and smoothing. The function generalizes the # calls of operations.smooth and operations.baseline\_subtract, filling an unused kwarg # with extraneous variables in operations.baseline\_subtract.

if p.smoothing:

lam = lambda x,y,w,ax: operations.smooth(x = x, y = y, \
 cutoff = p.cutoffs[w], boxcar = p.boxcar, \
 base\_per = p.baseline\_percentage, \
 diag = p.diagnostic, ax = ax, mode = p.smooth\_mode[w])

elif not p.smoothing:

lam = lambda x,y,w,ax: operations.baseline\_subtract(y = y, \
 perc = p.baseline\_percentage, x = x, w = w, ax = ax)

# Processed data is generated by calling the temporary function for each field.

w1\_sig\_sm = lam(w1\_col, w1\_sig, 'w1', ax1['w1'][1])

w2\_sig\_sm = lam(w2\_col, w2\_sig, 'w2', ax1['w2'][1])

w3\_sig\_sm = lam(w3\_del, w3\_sig, 'w3', ax1['w3'][1])

# Plot the processed and unprocessed data if diagnostics are enabled.

if p.diagnostic:

else: plt.close(fig1), plt.close(fig2)

# Overwrite the signal variables with the smoothed arrays.

w1\_sig, w2\_sig, w3\_sig = w1\_sig\_sm.copy(), w2\_sig\_sm.copy(), w3\_sig\_sm.copy()

del w1\_sig\_sm, w2\_sig\_sm, w3\_sig\_sm, lam

if p.diagnostic: del i, j, k, l

# Create time and frequency grids that should yield even sampling of the polarization # on each grid. These grids are saved in the existing parameters.parameters object.

p.time\_freq\_grids()

# Calculate the central frequencies for w1 and w2 fields based on the loaded tuning # files and save the results in p.w0, then calculate the central frequency for w4.

for i, j, k in [('w1',w1\_col,w1\_sig), ('w2', w2\_col, w2\_sig)]:

```
p.w0[i] = operations.center_freq(guess = (float(p.colors[i])*(3.e10)\
```

```
*2*np.pi), wl = j, sig = k, w = p.w)
```

```
del i
```

# Create dictionaries that will contain the optical properties at each w0.

p.calculate\_optical\_properties()

```
# Calculate the internal angles for each field
```

p.calculate\_internal\_angles()

```
# Interpolate w1_sig and w2_sig onto the previously calculated w grid.
```

```
w1_sig = operations.grid_spectrum(w0 = p.w0['w1'], w = p.w, \
```

```
wl = w1_col, sig = w1_sig)
```

w2\_sig = operations.grid\_spectrum(w0 = p.w0['w2'], w = p.w, \

```
wl = w2_col, sig = w2_sig)
```

del w1\_col, w2\_col

```
# Convert the loaded spectra to electric field amplitudes, with units V/micron.
```

```
lam = lambda x, y, w, tw, returnt: operations.calculate_field(x_ax = x, sig = y,\
    n0 = p.n0[w], waist = p.waist[w], energy = p.energy[w][p.colors[w]], \
    t_or_w = tw, perc = p.baseline_percentage,\
    diagnostic = p.diagnostic, returnt = returnt)
```

```
E1t, t1 = lam(p.w, w1_sig, 'w1', 'frequency', True)
        E2t, t2 = lam(p.w, w2_sig, 'w2', 'frequency', True)
        E3t = lam(w3_del, w3_sig, 'w3', 'time', False)
        del lam, w1_sig, w2_sig, w3_sig
# Grid the resulting electric fields, incorporating assumed initial delays and phase.
       Et = \{\}
        for i, j, k, l in [('w1', t1, E1t, p.delay['t31']), \
                        ('w2', t2, E2t, p.delay['t32']),\
                        ('w3', w3_del, E3t, 0.)]:
                if j[1] > j[0]:
                        model = interp1d(j - 1, k*np.exp(1j*p.phase[i]*np.pi/180), \
                                        bounds_error = False, fill_value = complex(0.))
                else:
                        model = interp1d(j[::-1] - 1, k[::-1]\
                                        *np.exp(1j*p.phase[i]*np.pi/180), \
                                        bounds_error = False, fill_value = complex(0.))
                Et[i] = model(p.t)
                del model
                if p.diagnostic:
                        ax2[i][0].plot(p.t, np.real(Et[i]), 'k-', p.t, \
                                np.real(Et[i]), 'ko')
                        ax2[i][1].plot(p.t, np.imag(Et[i]), 'k--', p.t, \
                                np.imag(Et[i]), 'ko')
        del i, j, k, l, E1t, E2t, E3t, t1, t2, w3_del
# Determine the bounds and spacing of the z grid and the x, y, kx, and ky grids
       p.spatial_grids()
       polzn_call = partial(operations.polarization_slice, x = p.x, y = p.y, kx = p.kx,\
                                ky = p.ky, t = p.t, w = p.w, Et = Et, \setminus
                                focus_offset = p.focus_offset, \
                                focus_shift = p.focus_shift,\
                                rel_length = p.rel_lengths, waist = p.waist, \
```

```
phase_matching = p.phase_matching, theta = p.theta, \
w0 = p.w0, n0 = p.n0, nw = p.nw, a0 = p.a0, aw = p.aw,
ug = p.ug, gvd = p.gvd, nwin = p.nwin, gvdw = p.gvdw, \
chi = p.chi, L = p.L, win_len = p.win_len, \
deltak = p.deltak, degeneracy = p.degeneracy, \
gvm_in = p.gvm_in, gvm_out = p.gvm_out, \
gvd_in = p.gvd_in, gvd_out = p.gvd_out, \
absorption = p.absorption, save = p.save, \
folder = p.save_folder, z_end = p.z[-1], \
disp = p.dispersive_k)
```

```
# Initialize an array to perform the summation.
    sum = np.zeros((len(p.w), len(p.kx), len(p.ky)), dtype=complex)
```

if not p.multiprocessing:

# For each point in the z grid:

for i in p.z:

# Add the result of calculating the polarization slice to the running sum,

# modified by the integration constant dz

 $sum += polzn_call(z = i) * abs(p.z[1] - p.z[0])$ 

elif p.multiprocessing:

# initialize the multiprocessing pool

pool = Pool(processes = p.processes)

# Perform the calculations asynchronously, adding each result to the running sum

# (modified by the integration factor dz) as that slice is available.

for i in pool.imap\_unordered(polzn\_call, p.z, chunksize=p.chunksize):

sum+=i \* abs(p.z[1] - p.z[0])

pool.close()

pool.join()

del i

# Perform the 2DFT necessary to convert the sum from the (kx,ky) domain to the (x,y)

# domain.

E = np.empty(np.shape(sum), dtype=complex)

for i in range(len(p.w)):

# 1DFT of kx  $\rightarrow$  x on the resulting sum.

# Perform the second 1DFT (ky --> y) on the temporary array temp and save the # result to the initialized E array.

```
E[i,:,:] = (1/np.sqrt(len(p.ky))) * (temp[:,:,None] * np.exp(1j * 2 \
```

del temp

del sum

# The field as a function of frequency is then the integral of E(w,x,y) over the x,y # dimensions.

# Calculate the intensity, convolving with the monochromator if relevant.

I = operations.intensity(E = E\_slice, mono\_conv = p.mono\_convolution, w = p.w, \
 t = p.t, N = p.N, f = p.f\_len, slits = p.slits, \
 n0 = p.n0['w4'], w0 = p.w0['w4'])

if p.save\_final:

if ret\_I:

```
return p.w+p.w0['w4'], I
```

else:

return p.w+p.w0['w4'], E\_slice

References

- Dau, H.; Haumann, M. The manganese complex of photosystem II in its reaction cycle Basic framework and possible realization at the atomic level. *Coordination Chemistry Reviews* 2008, 252, 273.
- McEvoy, J. P.; Brudvig, G. W. Water-Splitting Chemistry of Photosystem II. Chemical Reviews 2006, 106, 4455.
- (3) Dau, H.; Liebisch, P.; Haumann, M. The structure of the manganese complex of Photosystem II in its dark-stable S<sub>1</sub>-state - EXAFS results in relation to recent crystallographic data. *Physical Chemistry Chemical Physics* 2004, 6, 4781.
- (4) Glöckner, C.; Kern, J.; Broser, M.; Zouni, A.; Yachandra, V.; Yano, J. Structural Changes of the Oxygen-evolving Complex in Photosystem II during the Catalytic Cycle. *The Journal of Biological Chemistry* 2013, 288, 22607.
- (5) Penner-Hahn, J. E.; Fronko, R. M.; Pecoraro, V. L.; Yocum, C. F.; Betts, S. D.; Bowlby, N. R. Structural Characterization of the Manganese Sites in the Photosynthetic Oxygen-Evolving Complex Using X-ray Absorption Spectroscopy. *Journal of the American Chemical Society* **1990**, *112*, 2549.
- (6) Pushkar, Y.; Yano, J.; Sauer, K.; Boussac, A.; Yachandra, V. K. Structural changes in the Mn<sub>4</sub>Ca cluster and the mechanism of photosynthetic water splitting. *Proceedings of the National Academy of Sciences of the United States of America* 2008, 105, 1879.
- (7) Weng, T.-C.; Hsieh, W.-Y.; Uffelman, E. S.; Gordon-Wylie, S. W.; Collins, T. J.; Pecoraro, V. L.; Penner-Hahn, J. E. XANES Evidence Against a Manganyl Species in the S<sub>3</sub> State of the Oxygen-Evolving Complex. *Journal of the American Chemical Society* **2004**, *126*, 8070.
- (8) Yano, J.; Kern, J.; Irrgang, K.-D.; Latimer, M. J.; Bergmann, U.; Glatzel, P.; Pushkar, Y.; Biesiadka, J.; Loll, B.; Sauer, K.; Messinger, J.; Zouni, A.; Yachandra, V. K. X-ray damage to the Mn<sub>4</sub>Ca complex in single crystals of photosystem II: A case study for metalloprotein crystallography. *Proceedings of the National Academy of Sciences of the United States of America* 2005, 102, 12047.
- (9) Yano, J.; Kern, J.; Sauer, K.; Latimer, M. J.; Pushkar, Y.; Biesiadka, J.; Loll, B.; Saenger, W.; Messinger, J.; Zouni, A.; Yachandra, V. K. Where Water Is Oxidized to Dioxygen: Structure of the Photosynthetic Mn<sub>4</sub>Ca Cluster. *Science* **2006**, *314*, 821.
- (10) Yano, J.; Yachandra, V. K. Where Water Is Oxidized to Dioxygen: Structure of the Photosynthetic Mn<sub>4</sub>Ca Cluster from X-ray Spectroscopy. *Inorganic Chemistry* **2008**, 47, 1711.

- Bergmann, U.; Grush, M. M.; Horne, C. R.; DeMarois, P.; Penner-Hahn, J. E.; Yocum, C. F.; Wright,
  D. W.; Dubé, C. E.; Armstrong, W. H.; Christou, G.; Eppley, H. J.; Cramer, S. P. Characterization of the Mn Oxidation States in Photosystem II by KβX-ray Fluorescence Spectroscopy. *The Journal of Physical Chemistry B* 1998, 102, 8350.
- (12) Umena, Y.; Kawakami, K.; Shen, J.-R.; Kamiya, N. Crystal structure of oxygen-evolving photosystem II at a resolution of 1.9 Å. *Nature* 2011, 473, 55.
- (13) Glatzel, P.; Bergmann, U.; Yano, J.; Visser, H.; Robblee, J. H.; Gu, W.; de Groot, F. M.; Christou, G.; Pecoraro, V. L.; Cramer, S. P.; Yachandra, V. K. The Electronic Structure of Mn in Oxides, Coordination Complexes, and the Oxygen-Evolving Complex of Photosystem II Studied by Resonant Inelastic X-ray Scattering. *Journal of the American Chemical Society* **2004**, *126*, 9946.
- (14) Glatzel, P.; Schroeder, H.; Pushkar, Y.; Boron III, T.; Mukherjee, S.; Christou, G.; Pecoraro, V. L.; Messinger, J.; Yachandra, V. K.; Bergmann, U.; Yano, J. Electronic Structural Changes of Mn in the Oxygen-Evolving Complex of Photosystem II during the Catalytic Cycle. *Inorganic Chemistry* 2013, 52, 5642.
- (15) Boussac, A.; Girerd, J.-J.; Rutherford, A. W. Conversion of the Spin State of the Manganese Complex in Photosystem II Induced by Near-Infrared Light. *Biochemistry* **1996**, *35*, 6984.
- (16) Boussac, A. Inhomogeneity of the EPR multiline signal from the S<sub>2</sub>-state of the photosystem II oxygen-evolving enzyme. *Journal of Biological Inorganic Chemistry* **1997**, *2*, 580.
- (17) Boussac, A.; Un, S.; Horner, O.; Rutherford, A. W. High-Spin States  $(S \ge 5/2)$  of the Photosystem II Manganese Complex. *Biochemistry* **1998**, *37*, 4001.
- (18) Boussac, A.; Kuhl, H.; Un, S.; Rögner, M.; Rutherford, A. W. Effect of Near-Infrared Light on the S<sub>2</sub>-State of the Manganese Complex of Photosystem II from Synechococcus elongatus. Biochemistry 1998, 37, 8995.
- (19) Boussac, A.; Kuhl, H.; Ghibaudi, E.; Rögner, M.; Rutherford, A. W. Detection of an Electron Paramagnetic Resonance Signal in the S<sub>0</sub> State of the Manganese Complex of Photosystem II from *Synechococcus elongatus. Biochemistry* **1999**, *38*, 11942.
- (20) Boussac, A.; Sugiura, M.; Kirilovsky, D.; Rutherford, A. W. Near-infrared-induced Transitions in the Manganese Cluster of Photosystem II: Action Spectra for the S<sub>2</sub> and S<sub>3</sub> Redox States. *Plant & Cell Physiology* **2005**, *46*, 837.

- (21) Boussac, A.; Sugiura, M.; Lai, T.-L.; Rutherford, A. W. Low-temperature photochemistry in photosystem II from *Thermosynechococcus elongatus* induced by visible and near-infrared light. *Philo*sophical Transactions of the Royal Society B 2008, 363, 1203.
- (22) Charlot, M.-F.; Boussac, A.; Blondin, G. Towards a spin coupling model for the Mn<sub>4</sub> cluser in Photosystem II. *Biochimica et Biophysica Acta* 2005, 1708, 120.
- (23) Chrysina, M.; Zahariou, G.; Ioannidis, N.; Petrouleas, V. Conversion of the g = 4.1 EPR signal to the multiline conformation during the S<sub>2</sub> to S<sub>3</sub> transition of the oxygen evolving complex of Photosystem II. Biochimica et Biophysica Acta 2010, 1797, 487.
- (24) Cox, N.; Rapatskiy, L.; Su, J.-H.; Pantazis, D. A.; Sugiura, M.; Kulik, L.; Dorlet, P.; Rutherford, W. A.; Neese, F.; Boussac, A.; Lubitz, W.; Messinger, J. Effect of Ca<sup>2+</sup>/Sr<sup>2+</sup> Substitution on the Electronic Structure of the Oxygen-Evolving Complex of Photosystem II: A Combined Multifrequency EPR, <sup>55</sup>Mn-ENDOR, and DFT Study of the S<sub>2</sub> State. *Journal of the American Chemical Society* **2011**, *133*, 3635.
- (25) Horner, O.; Rivière, E.; Blondin, G.; Un, S.; Rutherford, A. W.; Girerd, J.-J.; Boussac, A. SQUID Magnetization Study of the Infrared-Induced Spin Transition in the S<sub>2</sub> State of Photosystem II: Spin Value Associated with the g = 4.1 EPR Signal. Journal of the American Chemical Society 1998, 120, 7924.
- (26) Hsieh, W.-Y.; Campbell, K. A.; Gregor, W.; Britt, R. D.; Yoder, D. W.; Penner-Hahn, J. E.; Pecoraro, V. L. The first spectroscopic model for the S<sub>1</sub> state multiline signal of the OEC. *Biochimica et Biophysica Acta* 2004, 1655, 149.
- (27) Ioannidis, N.; Petrouleas, V. Electron Paramagnetic Resonance Signals from the S<sub>3</sub> State of the Oxygen-Evolving Complex. A Broadened Radical Signal Induced by Low-Temperature Near-Infrared Light Illumination. *Biochemistry* 2000, 39, 5246.
- (28) Kulik, L. V.; Epel, B.; Lubitz, W.; Messinger, J. <sup>55</sup>Mn Pulse ENDOR at 34 GHz of the S<sub>0</sub> and S<sub>2</sub> States of the Oxygen-Evolving Complex in Photosystem II. Journal of the American Chemical Society 2005, 127, 2392.
- (29) Kulik, L. V.; Epel, B.; Lubitz, W.; Messinger, J. Electronic Structure of the Mn<sub>4</sub>O<sub>x</sub>Ca Cluster in the S<sub>0</sub> and S<sub>2</sub> States of the Oxygen-Evolving Complex of Photosystem II Based on Pulse <sup>55</sup>Mn-ENDOR and EPR Spectroscopy. *Journal of the American Chemical Society* **2007**, *129*, 13421.

- Peloquin, J. M.; Campbell, K. A.; Randall, D. W.; Evanchik, M. A.; Pecoraro, V. L.; Armstrong, W. H.; Britt, R. D. <sup>55</sup>Mn ENDOR of the S<sub>2</sub>-State Multiline EPR Signal of Photosystem II: Implications on the Structure of the Tetranuclear Mn Cluster. *Journal of the American Chemical Society* 2000, 122, 10926.
- (31) Rapatskiy, L.; Cox, N.; Savitsky, A.; Ames, W. M.; Sander, J.; Nowaczyk, M. M.; Rögner, M.; Boussac, A.; Neese, F.; Messinger, J.; Lubitz, W. Detection of the Water-Binding Sites of the Oxygen-Evolving Complex of Photosystem II Using W-Band <sup>17</sup>O Electron-Electron Double Resonance-Detected NMR Spectroscopy. Journal of the American Chemical Society **2012**, 134, 16619.
- (32) Sanakis, Y.; Ioannidis, N.; Sioros, G.; Petrouleas, V. A Novel S = 7/2 Configuration of the Mn Cluster of Photosystem II. Journal of the American Chemical Society 2001, 123, 10766.
- (33) Su, J.-H.; Cox, N.; Ames, W.; Pantazis, D. A.; Rapatskiy, L.; Lohmiller, T.; Kulik, L. V.; Dorlet, P.; Rutherford, W. A.; Neese, F.; Boussac, A.; Lubitz, W.; Messinger, J. The electronic structures of the S<sub>2</sub> states of the oxygen-evolving complexes of photosystem II in plants and cyanobacteria in the presence and absence of methanol. *Biochimica et Biophysica Acta* **2011**, *1807*, 829.
- (34) Baxter, R.; Krausz, E.; Wydrzynski, T.; Pace, R. J. Identification of the Near-Infrared Absorption Band from the Mn Cluster of Photosystem II. *Journal of the American Chemical Society* 1999, 121, 9451.
- (35) Boussac, A.; Zimmermann, J.-L.; Rutherford, A. W.; Lavergne, J. Histidine oxidation in the oxygenevolving photosystem-II enzyme. *Nature* **1990**, *347*, 303.
- (36) Dekker, J. P.; van Gorkom, H. J.; Brok, M.; Ouwehand, L. Optical Characterization of Photosystem II Electron Donors. *Biochimica et Biophysica Acta* 1984, 764, 301.
- (37) Dekker, J. P.; van Gorkom, H. J.; Wensink, J.; Ouwehand, L. Absorbance Difference Spectra of the Successive Redox States of the Oxygen-Evolving Apparatus of Photosynthesis. *Biochimica et Biophysica Acta* 1984, 767, 1.
- (38) Dekker, J. P.; Plijter, J. J.; Ouwehand, L.; van Gorkom, H. J. Kinetics of Manganese Redox Transitions in the Oxygen-Evolving Apparatus of Photosynthesis. *Biochimica et Biophysica Acta* 1984, 767, 176.
- (39) Dismukes, G. C.; Mathis, P. A near infrared electronic transition associated with conversion between S-states of the photosynthetic O<sub>2</sub>-evolving complex. *FEBS Letters* **1984**, *178*, 51.

- (40) Hillmann, B.; Schlodder, E. Electron transfer reactions in Photosystem II core complexes from Synechococcus at low temperature - difference spectrum of P680<sup>+</sup>Q<sub>A</sub><sup>-</sup>/P680 Q<sub>A</sub> at 77 K. Biochimica et Biophysica Acta 1995, 1231, 76.
- (41) Kretschmann, H.; Dekker, J. P.; Saygin, Ö.; Witt, H. T. An agreement on the quaternary oscillation of ultraviolet absorption changes accompanying the water splitting in isolated Photosystem II complex from the cyanobacterium Synechococcus sp. Biochimica et Biophysica Acta 1988, 932, 358.
- (42) Lavergne, J. Optical-difference spectra of the S-state transitions in the photosynthetic oxygen-evolving complex. *Biochimica et Biophysica Acta* 1987, *894*, 91.
- (43) Lavergne, J. Difference Spectra of the Oxidized Intermediates in the Photosynthetic Oxygen-Evolving System: Evidence for a Small S<sub>0</sub> → S<sub>1</sub> Absorption Change. *Photochemistry and Photobiology* **1989**, 50, 235.
- (44) Morton, J.; Akita, F.; Nakajima, Y.; Shen, J.-R.; Krausz, E. Optical Identification of the longwavelength (700-1700 nm) electronic excitations of the native reaction centre, Mn<sub>4</sub>CaO<sub>5</sub> cluster and cytochromes of photosystem II in plants and cyanobacteria. *Biochimica et Biophysica Acta* 2015, 1847, 153.
- (45) Pace, R. J.; Stranger, R. What spectroscopy reveals concerning the Mn oxidation levels in the oxygen evolving complex of photosystem II: X-ray to near infra-red. *Dalton Transactions* **2012**, *41*, 11145.
- (46) Saygin, O.; Witt, H. T. Evidence for the electrochromic identification of the change of charges in the four oxidation steps of the photoinduced water cleavage in photosynthesis. *FEBS Letters* 1985, 187, 224.
- (47) Saygin, Ö.; Witt, H. T. Optical characterization of intermediates in the water-splitting enzyme system of photosynthesis - possible states and configurations of manganese and water. *Biochimica et Biophysica Acta* 1987, 893, 452.
- (48) Schatz, G. H.; van Gorkom, H. J. Absorbance difference spectra upon charge transfer to secondary donors and acceptors in Photosystem II. *Biochimica et Biophysica Acta* **1985**, *810*, 283.
- (49) Stewart, D. H.; Nixon, P. J.; Diner, B. A.; Brudvig, G. W. Assignment of the Q<sub>Y</sub> Absorbance Bands of Photosystem II Chromophores by Low-Temperature Optical Spectroscopy of Wild-Type and Mutant Reaction Centers. *Biochemistry* 2000, *39*, 14583.
- (50) Van Gorkom, H. J. Identification of the Reduced Primary Electron Acceptor of Photosystem II as a Bound Semiquinone Anion. *Biochimica et Biophysica Acta* 1974, 347, 439.

- (51) Van Leeuwen, P. J.; Vos, M. H.; van Gorkom, H. J. Photosynthetic oxygen evolution. 1990, 1018, 173.
- (52) Van Leeuwen, P. J.; van Gorkom, H. J.; Dekker, J. P. Analysis of UV absorption changes with periodicity four in oxygen-evolving photosystem II membranes. *Journal of Photochemistry and Photobiology* B: Biology 1992, 15, 33.
- (53) Van Leeuwen, P. J.; Heimann, C.; Gast, P.; Dekker, J. P.; van Gorkom, H. J. Flash-induced redox changes in oxygen-evolving spinach Photosystem II core particles. *Photosynthesis Research* 1993, 38, 169.
- (54) Van Leeuwen, P. J.; Heimann, C.; van Gorkom, H. J. Absorbance difference spectra of the S-state transitions in Photosystem II core particles. *Photosynthesis Research* **1993**, *38*, 323.
- (55) Velthuys, B. R. Spectroscopy characterization of the acceptor state Q<sup>-</sup><sub>A</sub> and the donor state S2 of Photosystem II of spinach in the blue, red and near-infrared. *Biochimica et Biophysica Acta* 1988, 933, 249.
- (56) Witt, K. Further Evidence of X-320 as a Primary Acceptor of Photosystem II in Photosynthesis. FEBS Letters 1973, 38, 116.
- (57) Barry, B. A.; Cooper, I. B.; De Riso, A.; Brewer, S. H.; Vu, D. M.; Dyer, R. B. Time-resolved vibrational spectroscopy detects protein-based intermediates in the photosynthetic oxygen-evolving cycle. Proceedings of the National Academy of Sciences of the United States of America 2006, 103, 7288.
- (58) Chen, J.; Barry, B. A. Ultraviolet Resonance Raman Microprobe Spectroscopy of Photosystem II. Photochemistry and Photobiology 2008, 84, 815.
- (59) Chen, J.; Yao, M.; Pagba, C. V.; Zheng, Y.; Fei, L.; Fend, Z.; Barry, B. A. Directly probing redoxlinked quinones in photosystem II membrane fragments via UV resonant Raman scattering. *Biochimica et Biophysica Acta* **2015**, *1847*, 558.
- (60) Chu, H.-A.; Hillier, W.; Law, N. A.; Babcock, G. T. Vibrational spectroscopy of the oxygen-evolving complex and of manganese model compounds. *Biochimica et Biophysica Acta* 2001, 1503, 69.
- (61) Cua, A.; Stewart, D. H.; Reifler, M. J.; Brudvig, G. W.; Bocian, D. F. Low-Frequency Resonance Raman Characterization of the Oxygen-Evolving Complex of Photosystem II. *Journal of the American Chemical Society* 2000, 122, 2069.
- (62) Noguchi, T. Coordination Chemistry Reviews 2008, 252, 336.

- (63) Ames, W.; Pantazis, D. A.; Krewald, V.; Cox, N.; Messinger, J.; Lubitz, W.; Neese, F. Theoretical Evaluation of Structural Models of the S<sub>2</sub> State in the Oxygen Evolving Complex of Photosystem II. Journal of the American Chemical Society 2011, 133, 19743.
- (64) Gatt, P.; Petrie, S.; Stranger, R.; Pace, R. J. Rationalizing the 1.9 Å Crystal Structure of Photosystem
   II A Remarkable Jahn-Teller Balancing Act Induced by a Single Proton Transfer. Angewandte
   Chemie International Edition 2012, 51, 12025.
- (65) Kurashige, Y.; Chang, G. K.-L.; Yanai, T. Entangled quantum electronic wavefunctions of the Mn<sub>4</sub>CaO<sub>5</sub> cluster in photosystem II. *Nature Chemistry* **2013**, *5*, 660.
- (66) Luber, S.; Rivalta, I.; Umena, Y.; Kawakami, K.; Shen, J.-R.; Kamiya, N.; Brudvig, G. W.; Batista,
   V. S. S<sub>1</sub>-State Model of the O<sub>2</sub>-Evolving Complex of Photosystem II. *Biochemistry* 2011, 50, 6308.
- (67) Petrie, S.; Stranger, R.; Pace, R. J. Structural, Magnetic Coupling and Oxidation State Trends in Models of the CaMn<sub>4</sub> Cluster in Photosystem II. *Chemistry: A European Journal* **2008**, *14*, 5482.
- (68) Siegbahn, P. E. M. Structures and Energetics for O<sub>2</sub> Formation in Photosystem II. Accounts of Chemical Research 2009, 42, 1871.
- (69) Siegbahn, P. E. M. An Energetic Comparison of Different Models for the Oxygen Evolving Complex of Photosystem II. Journal of the American Chemical Society 2009, 131, 18238.
- (70) Siegbahn, P. E. M. The Effect of Backbone Constraints: The Case of Water Oxidation by the Oxygen-Evolving Complex in PSII. *ChemPhysChem* **2011**, *12*, 3274.
- (71) Sproviero, E. M.; Gascon, J. A.; McEvoy, J. P.; Brudvig, G. W.; Batista, V. S. Characterization of synthetic oxomanganese complexes and the inorganic core of the O<sub>2</sub>-evolving complex in photosystem II: Evaluation of the DFT/B3LYP level of theory. *Journal of Inorganic Biochemistry* 2006, 100, 786.
- (72) Sproviero, E. M.; Gascón, J. A.; McEvoy, J. P.; Brudvig, G. W.; Batista, V. S. QM/MM Models of the O<sub>2</sub>-Evolving Complex of Photosystem II. *Journal of Chemical Theory and Computation* 2006, 2, 1119.
- (73) Sproviero, E. M.; Gascón, J. A.; McEvoy, J. P.; Brudvig, G. W.; Batista, V. S. Quantum Mechanics/Molecular Mechanics Study of the Catalytic Cycle of Water Splitting in Photosystem II. *Journal* of the American Chemical Society **2008**, 130, 3428.
- (74) Sproviero, E. M.; Gascón, J. A.; McEvoy, J. P.; Brudvig, G. W.; Batista, V. S. Computational studies of the O<sub>2</sub>-evolving complex of photosystem II and biomimetic oxomanganese complexes. *Coordination Chemistry Reviews* 2008, 252, 395.

- (75) Yamanaka, S.; T., S.; Kanda, K.; Isobe, H.; Umena, Y.; Kawakami, K.; Shen, J.-R.; Kamiya, N.; Okumura, M.; Nakamura, H.; Yamaguchi, K. Structure and Reactivity of the Mixed-Valence CaMn<sub>4</sub>O<sub>5</sub> (H<sub>2</sub>O)<sub>4</sub> and CaMn<sub>4</sub>O<sub>4</sub> (OH) (H<sub>2</sub>O)<sub>4</sub> Clusters at Oxygen Evolution Catalyst of Photosystem II. Hybrid DFT (UB3LYP and UBHandHLYP) Calculations. *International Journal of Quantum Chemistry* 2012, 112, 321.
- (76) Boussac, A.; Sugiura, M.; Rutherford, A. W.; Dorlet, P. Complete EPR Spectrum of the S<sub>3</sub>-State of the Oxygen-Evolving Photosystem II. *Journal of the American Chemical Society* **2009**, *131*, 5050.
- (77) Bensasson, R.; Land, E. J. Optical and Kinetic Properties of Semireduced Plastoquinone and Ubiquinone: Electron Acceptors in Photosynthesis. *Biochimica et Biophysica Acta* 1973, 325, 175.
- (78) Candeias, L. P.; Turconi, S.; Nugent, J. H. A. Tyrosine Y<sub>Z</sub> and Y<sub>D</sub> of photosystem II Comparison of optical spectra to those of tyrosine oxidised by pulsed radiolysis. *Biochimica et Biophysica Acta* 1998, 1363, 1.
- (79) Cua, A.; Stewart, D. H.; Brudvig, G. W.; Bocian, D. F. Selective Resonance Raman Scattering from Chlorophyll Z in Photosystem II via Excitation into Near-Infrared Absorption Band of the Cation. *Journal of the American Chemical Society* **1998**, *120*, 4532.
- (80) Dashdorj, N.; Xu, W.; Martinsson, P.; Chitnis, P. R.; Savikhin, S. Electrochromic Shift of Chlorophyll Absorption in Photosystem I from *Synechocystis* sp. PCC 6803: A Probe of Optical and Dielectric Properties around the Secondary Electron Acceptor. *Biophysical Journal* 2004, *86*, 3121.
- (81) Dau, H.; Sauer, K. Electric field effect on chlorophyll fluorescence and its relation to Photosystem II charge separation reactions studied by a salt-jump technique. *Biochimica et Biophysica Acta* 1991, 1098, 49.
- (82) Guo, Z.; Lin, S.; Woodbury, N. W. Utilizing the Dynamic Stark Shift as a Probe for Dielectric Relaxation in Photosynthetic Reaction Centers During Charge Separation. *The Journal of Physical Chemistry B* 2013, 117, 11383.
- (83) Melis, A.; Brown, J. S. Stoichiometry of system I and system II reaction centers and of plastoquinone in different photosynthetic membranes. *Proceedings of the National Academy of Sciences of the United States of America* 1980, 77, 4712.
- (84) Razeghifard, M. R.; Kim, S.; Patzlaff, J. S.; Hutchison, R. S.; Krick, T.; Ayala, I.; Steenhuis, J. J.; Boesch, S. E.; Wheeler, R. A.; Barry, B. A. The Journal of Physical Chemistry B. **1999**, *103*, 9790.

- (85) Albela, B.; Chottard, G.; Girerd, J.-J. Biomimetic approach to the oxygen evolving center: resonance Raman investigation of a manganese μ-oxo dimer in three oxidation states. Journal of Biological Inorganic Chemistry 2001, 6, 430.
- (86) Blondin, G.; Davydov, R.; Philouze, C.; Charlot, M.-F.; Styring, S.; Åkermark, B.; Girerd, J.-J.; Boussac, A. Electron paramagnetic resonance study of the S = <sup>1</sup>/<sub>2</sub> ground state of a radiolysis-generated manganese(III)-trimanganese(IV) form of [Mn<sup>IV</sup><sub>4</sub>O<sub>6</sub>(bipy)<sub>6</sub>]<sup>4+</sup> (bipy = 2,2'-bipyridine). Comparison with the photsynthetic Oxygen Evolving Complex. Journal of the Chemical Society, Dalton Transactions **1997**, 6, 4069.
- (87) Collomb, M.-N.; Deronzier, A.; Richardot, A.; Peécaut, J. Synthesis and characterization of a new kind of Mn<sub>2</sub><sup>III,IV</sup> μ-oxo complex: [Mn<sub>2</sub>O<sub>2</sub>(terpy)<sub>2</sub>(H<sub>2</sub>O)<sub>2</sub>](NO<sub>3</sub>)<sub>3</sub> · 6 H<sub>2</sub>O, terpy = 2,2':6'2"-terpyridine. New Journal of Chemistry **1999**, 23, 351.
- (88) Vincent, J. B.; Christou, G. Model complexes suggest certain S-state changes of the photosynthetic water-oxidation enzyme may involve an Mn(II)-Mn(III) transition. FEBS Letters 1986, 207, 250.
- (89) Chen, H.; Tagore, R.; Olack, G.; Vrettos, J. S.; Weng, T.-C.; Penner-Hahn, J.; Crabtree, R. H.; Brudvig, G. W. Speciation of the Catalytic Oxygen Evolution System: [Mn<sup>III/IV</sup><sub>2</sub>(μ-O)<sub>2</sub>(terpy)<sub>2</sub> (H<sub>2</sub>O)<sub>2</sub>](NO<sub>3</sub>)<sub>3</sub> + HSO<sub>5</sub><sup>-</sup>. Inorganic Chemistry **2007**, 46, 34.
- (90) Limburg, J.; Brudvig, G. W.; Crabtree, R. H. O<sub>2</sub> Evolution and Permanganate Formation from High-Valent Manganese Complex. *Journal of the American Chemical Society* **1997**, *119*, 2761.
- (91) Limburg, J.; Vrettos, J. S.; Liable-Sands, L. M.; Rheingold, A. L.; Crabtree, R. H.; Brudvig, G. W. A Functional Model for O-O Bond Formation by the O<sub>2</sub>-Evolving Complex in Photosystem II. *Science* 1999, 283, 1524.
- (92) Limburg, J.; Vrettos, J. S.; Chen, H.; de Paula, J. C.; Crabtree, R. H.; Brudvig, G. W. Characterization of the O<sub>2</sub>-Evolving Reaction Catalyzed by [(terpy)(H<sub>2</sub>O)Mn<sup>III</sup>(O)<sub>2</sub>Mn<sup>IV</sup>(OH<sub>2</sub>)(terpy)](NO<sub>3</sub>)<sub>3</sub> (terpy = 2,2':6,2"-Terpyridine). Journal of the American Chemical Society **2001**, 123, 423.
- (93) Lundberg, M.; Blomberg, M. R. A.; Siegbahn, P. E. M. Oxyl Radical Required for O-O Bond Formation in Synthetic Mn-Catalyst. *Inorganic Chemistry* 2004, 43, 264.
- (94) Tagore, R.; Crabtree, R. H.; Brudvig, G. W. Oxygen Evolution Catalysis by a Dimanganese Complex and Its Relation to Photosynthetic Water Oxidation. *Inorganic Chemistry* 2008, 47, 1815.

- (95) Yagi, M.; Narita, K. Catalytic O<sub>2</sub> Evolution from Water Induced by Adsorption of [(OH<sub>2</sub>) (Terpy) Mn (μ-O)<sub>2</sub> Mn (Terpy) (OH<sub>2</sub>)]<sup>3+</sup> Complex onto Clay Compounds. Journal of the American Chemical Society 2004, 126, 8084.
- (96) Yamazaki, H.; Igarashi, S.; Nagata, T.; Yagi, M. Substituent Effects on Core Structures and Heterogeneous Catalytic Activities of Mn<sup>III</sup>(μ-O)<sub>2</sub>Mn<sup>IV</sup> Dimers with 2,2':6',2"-Terpyridine Derivative Ligands for Water Oxidation. *Inorganic Chemistry* **2012**, *51*, 1530.
- (97) Young, K. J.; Brennan, B. J.; Tagore, R.; Brudvig, G. W. Photosynthetic Water Oxidation: Insights from Manganese Model Chemistry. Accounts of Chemical Research 2015, 48, 567.
- (98) Pakoulev, A. V.; Rickard, M. A.; Kornau, K. M.; Mathew, N. A.; Yurs, L. A.; Block, S. B.; Wright, J. C. Mixed Frequency-/Time-Domain Coherent Multidimensional Spectroscopy: Research Tool or Potential Analytical Method? Accounts of Chemical Research 2009, 42, 1310.
- (99) Wright, J. Lecture Notes: Experimental Spectroscopy, Chemistry 623.; 2012.
- (100) Wright, J. C. Multiresonant Coherent Multidimensional Spectroscopy. Annual Review of Physical Chemistry 2011, 62, 209.
- (101) Griffiths, D. J., Introduction to Electrodynamics, Third; Prentice-Hall: Upper Saddle River, NJ, 1999.
- (102) Boyd, R. W., Nonlinear Optics, Second; Academic Press: San Diego, 2003.
- (103) Akhmanov, S. A.; Vysloukh, V. A.; Chirkin, A. S., Optics of Femtosecond Laser Pulses; American Institute of Physics: New York, 1992.
- (104) Svelto, O., Principles of Lasers, Second; Plenum Press: New York, 1982.
- (105) Verdeyen, J. T., Laser Electronics; Prentice-Hall: Englewood Cliffs, NJ, 1981.
- (106) Dmitriev, V. G.; Gurzadyan, G. G.; Nikogosyan, D. N., Handbook of Nonlinear Optical Crystals, Second; Springer Series in Optical Sciences; Springer: Berlin, 1997.
- (107) Kittel, C., Introduction to Solid State Physics, Eighth; John Wiley & Sons: Hoboken, NJ, 2005.
- (108) Lax, M.; Louisell, W. H.; McKnight, W. B. From Maxwell to paraxial wave optics. *Physical Review A* 1975, 11, 1365.
- (109) Shen, Y. R., The Principles of Nonlinear Optics; John Wiley & Sons: New York, 1984.
- (110) Levenson, M. D.; Bloembergen, N. Dispersion of the nonlinear optical susceptibilities of organic liquids and solutions. *The Journal of Chemical Physics* **1974**, *60*, 1323.

- (111) Iga, K., Fundamentals of Laser Optics; Lasers, Photonics, and Electro-Optics; Plenum Press: New York, 1994.
- (112) Poularikas, A. D., The Handbook of Formulas and Tables for Signal Processing; The Electrical Engineering Handbook Series; CRC Press: Boca Raton, FL, 1999.
- (113) Press, W. H.; Teukolsky, S. A.; Vetterling, W. T.; Flannery, B. P., Numerical Recipes in C: The Art of Scientific Computing, Third; Cambridge University Press: New York, 2007.
- (114) Bracewell, R. N., The Fourier Transform and Its Applications, Third; McGraw Hill: Boston, 2000.
- (115) Brabec, T.; Krausz, F. Nonlinear Optical Pulses in the Single-Cycle Regime. *Physical Review Letters* 1997, 78, 3282.
- (116) Stewart, J., Calculus: Early Transcendentals, Fifth; Thomson Brooks/Cole: Mason, OH, 2007.
- (117) Arfken, G. B.; Weber, H. J., Mathematical Methods for Physicists, Sixth; Elsevier: Burlington, MA, 2005.
- (118) Wang, J.-C.; Wang, J.-K. Experimental and theoretical analysis of white-light seeded, collinear phasematching, femtosecond optical parametric amplifiers. *Journal of the Optical Society of America B* 2004, 21, 45.
- (119) Midwinter, J. E.; Warner, J. The effects of phase matching method and of uniaxial crystal symmetry on the polar distribution of second-order non-linear optical polarization. British Journal of Applied Physics 1965, 16, 1135.
- (120) Compton, R.; Filin, A.; Romanov, D. A.; Levis, R. J. Elucidating the spectral and temporal contributions from the resonant and nonresonant response to femtosecond coherent anti-Stokes Raman scattering. *The Journal of Chemical Physics* **2008**, *128*, 154517.
- (121) Pestov, D.; Murawski, R. K.; Ariunbold, G. O.; Wang, X.; Zhi, M.; Sokolov, A. V.; Sautenkov, V. A.; Rostovtsev, Y. V.; Dagariu, A.; Huang, Y.; Scully, M. O. Optimizing the Laser-Pulse Configuration for Coherent Raman Spectroscopy. *Science* **2007**, *316*, 265.
- (122) Von Vacano, B.; Buckup, T.; Motzkus, M. Highly sensitive single-beam heterodyne coherent anti-Stokes Raman scattering. *Optics Letters* **2006**, *31*, 2495.
- (123) Wang, X.; Zhang, A.; Zhi, M.; Sokolov, A. V.; Welch, G. R.; Scully, M. O. Heterodyne coherent anti-Stokes Raman scattering for spectral phase retrieval and signal amplification. *Optics Letters* 2010, 35, 721.

- (124) Zeytunyan, A.; Crampton, K. T.; Zadoyan, R.; Apkarian, V. A. Supercontinuum-based three-color three-pulse time-resolved coherent anti-Stokes Raman scattering. *Optics Express* **2015**, *23*, 24019.
- (125) Barnes, N. P.; Corcoran, V. J. Parametric generation processes: spectral bandwidth and acceptance angles. Applied Optics 1976, 15, 696.
- (126) Cerullo, G.; De Silvestri, S. Ultrafast optical parametric amplifiers. *Review of Scientific Instruments* 2003, 74, 1.
- (127) Gloster, L. A. W.; Jiang, Z. X.; King, T. A. Characterization of an Nd:YAG-Pumped β-BaB<sub>2</sub>O<sub>4</sub> Optical Parametric Oscillator in Collinear and Noncollinear Phase-Matched Configurations. *IEEE Journal of Quantum Electronics* 1994, 30, 2961.
- (128) Milton, M. J. T.; McIlveen, T. J.; Hanna, D. C.; Woods, P. T. High-efficiency infrared generation by difference-frequency mixing using tangential phase matching. *Optics Communications* 1992, *87*, 273.
- (129) Nautiyal, A.; Bisht, P. B.; Bindra, K. S.; Oak, S. M. Effects of thickness of β-barium borate and angle of non-collinearity of the fs pulse generation by optical parametric amplification. Optics & Laser Technology 2009, 41, 539.
- (130) Maker, P. D.; Terhune, R. W.; Nisenoff, M.; Savage, C. M. Effects of Dispersion and Focusing on the Production of Optical Harmonics. *Physical Review Letters* **1962**, *8*, 21.
- (131) Joosen, W.; Bakker, H. J.; Noordam, L. D.; Muller, H. G.; van Linden van den Heuvell, H. B. Parametric generation in β-barium borate of intense femtosecond pulses near 800 nm. Journal of the Optical Society of America B 1991, 8, 2537.
- (132) Song, L.; Leng, Y.; Zhang, C.; Wang, J.; Li, C.; Li, R. Carrier-envelope phase offset for pulses from a tunable optical parametric amplifier. *Optics Communications* **2011**, *284*, 3047.
- (133) Seilmeier, A.; Spanner, K.; Laubereau, A.; Kaiser, W. Narrow-band tunable infrared pulses with sub-picosecond time resolution. *Optics Communications* 1978, 24, 237.
- (134) Guo, X.; Wang, C.; Leng, Y.; Li, R. Bandwidth analysis of type-I optical parametric chirped pulse amplification systems. *Journal of the Optical Society of America B* **2014**, *31*, 2615.
- (135) Laufer, G.; Miles, R. B. Angularly Resolved Coherent Raman Spectroscopy. Optics Communications 1979, 28, 250.
- (136) McCamant, D. W.; Kukura, P.; Mathies, R. A. Femtosecond Broadband Stimulated Raman: A New Approach for High-Performance Vibrational Spectroscopy. *Applied Spectroscopy* **2003**, *57*, 1317.

- (137) McCamant, D. W.; Kukura, P.; Mathies, R. A. Femtosecond Time-Resolved Stimulated Raman Spectroscopy: Application to the Ultrafast Internal Conversion in β-Carotene. The Journal of Physical Chemistry A 2003, 107, 8208.
- (138) Tzankov, P.; Buchvarov, I.; Fiebig, T. Broadband optical parametric amplification in the near UV-VIS. Optics Communications 2002, 203, 107.
- (139) Cheng, J.-X.; Volkmer, A.; Sunney Xie, X. Theoretical and experimental characterization of coherent anti-Stokes Raman scattering microscopy. *Journal of the Optical Society of America B* 2002, 19, 1363.
- (140) Cheng, J.-X.; Volkmer, A.; Book, L. D.; Sunney Xie, X. An Epi-Detected Coherent Anti-Stokes Raman Scattering (E-CARS) Microscopy with High Spectral Resolution and High Sensitivity. *The Journal of Physical Chemistry B* 2001, 105, 1277.
- (141) Garbacik, E. T.; Korterik, J. P.; Otto, C.; Herek, J. L.; Offerhaus, H. L. Epi-detection of vibrational phase contrast coherent anti-Stokes Raman scattering. *Optics Letters* **2014**, *39*, 5814.
- (142) Wang, P.; Slipchenko, M. N.; Zhou, B.; Pinal, R.; Cheng, J.-X. Mechanisms of Epi-Detected Stimulated Raman Scattering Microscope. *IEEE Journal of Selected Topics in Quantum Electronics* 2012, 18, 384.
- (143) Oron, D.; Dudovich, N.; Yelin, D.; Silderberg, Y. Narrow-Band Coherent Anti-Stokes Raman Signals from Broad-Band Pulses. *Physical Review Letters* **2002**, *88*, 063004.
- (144) Shim, S.-H.; Strasfeld, D. B.; Ling, Y. L.; Zanni, M. T. Automated 2D IR spectroscopy using a mid-IR pulse shaper and application of this technology to the human islet amyloid polypeptide. *Proceedings of the National Academy of Sciences of the United States of America* 2007, 104, 14197.
- (145) Tian, P.; Keusters, D.; Suzaki, Y.; Warren, W. S. Femtosecond Phase-Coherent Two-Dimensional Spectroscopy. Science 2003, 300, 1553.
- (146) Cheng, J.-X.; Volkmer, A.; Book, L. D.; Sunney Xie, X. Multiplex Coherent Anti-Stokes Raman Scattering Microspectroscopy and Study of Lipid Vesicles. *The Journal of Physical Chemistry B* 2002, 107, 8493.
- (147) Romanov, D.; Filin, A.; Compton, R.; Levis, R. Phase matching in femtosecond BOXCARS. Optics Letters 2007, 32, 3161.
- (148) Zumbusch, A.; Holtom, G. R.; Sunney Xie, X. Three-Dimensional Vibrational Imaging by Coherent Anti-Stokes Raman Scattering. *Physical Review Letters* 1999, *82*, 4142.

- (149) Gaydardzhiev, A.; Nikolov, I.; Buchvarov, I.; Petrov, V.; Noack, F. Ultrabroadband operation of a femtosecond optical parametric generator based on BiB<sub>36</sub> in the near-IR. Optics Express 2008, 16, 2363.
- (150) Ghotbi, M.; Beutler, M.; Petrov, V.; Gaydardzhiev, A.; Noack, F. High-energy, sub-30 fs near-IR pulses from a broadband optical parametric amplifier based on collinear interaction in BiB<sub>3</sub>O<sub>6</sub>. Optics Letters 2009, 34, 689.
- (151) Nikolov, I.; Gaydardzhiev, A.; Buchvarov, I.; Tzankov, P.; Noack, F.; Petrov, V. Ultrabroadband continuum amplification in the near infrared using BiB<sub>36</sub> nonlinear crystals pumped at 800 nm. *Optics Letters* **2007**, *32*, 3342.
- (152) Cerullo, G.; Nisoli, M.; De Silvestri, S. Generation of 11 fs pulses tunable across the visible by optical parametric amplification. *Applied Physics Letters* **1997**, *71*, 3616.
- (153) Hao, Q.; Huang, Y.-C. Two-octave polarized supercontinuum generated from a Q-switched laser pumped doubly resonant parametric oscillator. Optics Letters 2013, 38, 1863.
- (154) Kobayashi, T.; Baltuska, A. Sub-5 fs pulse generation from a noncollinear optical parametric amplifier.
   Measurement Science & Technology 2002, 13, 1671.
- (155) Virgillito, D., Personal Communication.
- (156) Wang, H.; Weiner, A. M. A Femtosecond Waveform Transfer Technique Using Type II Second Harmonic Generation. *IEEE Journal of Quantum Electronics* 2004, 40, 937.
- (157) Yu, L. et al. Optimization for high-energy and high-efficiency broadband optical parametric chirpedpulse amplification in LBO near 800 nm. Optics Letters 2015, 40, 3412.
- (158) Baer, T. Large-amplitude fluctuations due to longitudinal mode coupling in diode-pumped intracavity doubled Nd:YAG lasers. *Journal of the Optical Society of America B* **1986**, *3*, 1175.
- (159) Kafka, J. D.; Baer, T. Prism-pair dispersive delay lines in optical pulse compression. Optics Letters 1987, 12, 401.
- (160) Akhmanov, S. A.; Chirkin, A. S.; Drabovich, K. N.; Kovrigin, A. I.; Khokhlov, R. V.; Sukhorukov, A. P. Nonstationary Nonlinear Optical Effects and Ultrashort Light Pulse Formation. *IEEE Journal* of Quantum Electronics **1968**, 4, 598.
- (161) Comly, J.; Garmire, E. Second Harmonic Generation from Short Pulses. Appled Physics Letters 1968, 12, 7.

- (162) Glenn, W. H. Second-Harmonic Generation by Picosecond Optical Pulses. IEEE Journal of Quantum Electronics 1969, 5, 284.
- (163) Stabinis, A.; Valiulis, G.; Ibragimov, E. A. Effective sum frequency pulse compression in nonlinear crystals. Optics Communications 1991, 86, 301.
- (164) Falk, J.; Murray, J. E. Single-Cavity Noncollinear Optical Parametric Oscillation. Applied Physics Letters 1969, 14, 245.
- (165) Selm, R.; Winterhalder, M.; Zumbusch, A.; Krauss, G.; Hanke, T.; Sell, A.; Leitenstorfer, A. Ultrabroadband background-free coherent anti-Stokes Raman scattering microscopy based on a compact Er:fiber laser system. *Optics Letters* **2010**, *35*, 3282.
- (166) Fecko, C. J.; Loparo, J. J.; Tokmakoff, A. Generation of 45 femtosecond pulses at 3 μm with a KNbO<sub>3</sub> optical parametric amplifier. *Optics Communications* **2004**, *241*, 521.
- (167) Bhar, G. C.; Datta, P. K.; Rudra, A. M.; Chatterjee, U. Tangentially phase-matched efficient difference frequency generation in beta barium borate crystal. *Optics Communications* **1994**, *105*, 95.
- (168) Nisoli, M.; Stagira, S.; De Silvestri, S.; Svelto, O.; Valiulis, G.; Varanavicius, A. Parametric generation of high-energy 14.5-fs light pulses at 1.5 µm. *Optics Letters* **1998**, *23*, 630.
- (169) Harris, S. E.; Oshman, M. K.; Byer, R. L. Observation of Tunable Optical Parametric Fluorescence. *Physical Review Letters* 1967, 18, 732.
- (170) Danielius, R.; Piskarskas, A.; Stabinis, A.; Banfi, G. P.; Di Trapani, P.; Righini, R. Traveling-wave parametric generation of widely tunable, highly coherent femtosecond light pulses. *Journal of the Optical Society of America B* 1993, 10, 2222.
- (171) Di Trapani, P.; Andreoni, A.; Solcia, C.; Foggi, P.; Danielius, R.; Dubietis, A.; Piskarskas, A. Matching of group velocities in three-wave parametric interaction with femtosecond pulses and application to traveling-wave generators. *Journal of the Optical Society of America B* **1995**, *12*, 2237.
- (172) Reed, M. K.; Steiner-Shepard, M. K.; Armas, M. S.; Negus, D. K. Microjoule-energy ultrafast optical parametric amplifiers. *Journal of the Optical Society of America B* **1995**, *12*, 2229.
- (173) Banfi, G. P.; Di Trapani, P.; Danielius, R.; Piskarskas, A.; Righini, R.; Sa'nta, I. Tunable femtosecond pulses close to the transform limit from traveling-wave parametric conversion. *Optics Letters* **1993**, 18, 1547.
- (174) Banfi, G. P.; Danielius, R.; Piskarskas, A.; Di Trapani, P.; Foggi, P.; Righini, R. Femtosecond traveling-wave parametric generator with lithium triborate. *Optics Letters* **1993**, *18*, 1633.

- (175) Petrov, V.; Seifert, F.; Noack, F. High repetition rate traveling wave optical parametric generator producing nearly bandwidth limited 50 fs infrared light pulses. *Applied Physics Letters* 1994, 65, 268.
- (176) Seifert, F.; Petrov, V.; Noack, F. Sub-100-fs optical parametric generator pumped by a high-repetition-rate Ti:sapphire regenerative amplifier system. *Optics Letters* **1994**, *19*, 837.
- (177) Nisoli, M.; De Silvestri, S.; Magni, V.; Svelto, O.; Danielius, R.; Piskarskas, A.; Valiulis, G.; Varanavicius, A. Highly efficient parametric conversion of femtosecond Ti:sapphire laser pulses at 1 kHz. *Optics Letters* **1994**, *19*, 1973.
- (178) Huber, R.; Satzger, H.; Zinth, W.; Wachtveitl, J. Noncollinear optical parametric amplifiers with output parameters improved by the application of a white light continuum generated in CaF<sub>2</sub>. Optics Communications **2001**, 194, 443.
- (179) Karthick Kumar, S. K.; Goswami, T.; Bhattacharyya, I.; Goswami, D. Visible 20-femtosecond pulse generation by double-pass non-collinear optical parametric amplifier. *Current Science* 2009, *96*, 1496.
- (180) Kobayashi, T.; Shirakawa, A. Sub-10-fs tunable pulses in visible and NIR and visible sub-5-fs pulses generated by noncollinear OPA. *Journal of Luminescence* 2000, 87-89, 119.
- (181) Oien, A. L.; McKinnie, I. T.; Jain, P.; Russell, N. A.; Warrington, D. M.; Gloster, L. A. W. Efficient, low-threshold collinear and noncollinear β-barium borate optical parametric oscillators. *Optics Letters* **1997**, 22, 859.
- (182) Shirakawa, A.; Kobayashi, T. Applied Physics Letters 1998, 72, 147.
- (183) Shirakawa, A.; Sakane, I.; Takasaka, M.; Kobayashi, T. Sub-5-fs visible pulse generation by pulsefront-matched noncollinear optical parametric amplification. *Applied Physics Letters* **1999**, 74, 2268.
- (184) Tzankov, P.; Fiebig, T.; Buchvarov, I. Tunable femtosecond pulses in the near-ultraviolet from ultrabroadband parametric amplification. *Applied Physics Letters* 2003, *82*, 517.
- (185) Kohler, D. D.; Block, S. B.; Kain, S.; Pakoulev, A. V.; Wright, J. C. Ultrafast Dynamics within the 1S Exciton Band of Colloidal PbSe Quantum Dots Using Multiresonant Coherent Multidimensional Spectroscopy. *The Journal of Physical Chemistry C* 2014, *118*, 5020.
- (186) Pestov, D.; Ariunbold, G. O.; Wang, X.; Murawski, R. K.; Sautenkov, V. A.; Sokolov, A. V.; Scully, M. O. Coherent versus incoherent Raman scattering: molecular coherence excitation and measurement. Optics Letters 2007, 32, 1725.
- (187) Prior, Y. Three-dimensional phase matching in four-wave mixing. Applied Optics 1980, 19, 1741.

- (188) Trebino, R.; DeLong, K. W.; Fittinghoff, D. N.; Sweetser, J. N.; Krumbügel, M. A.; Richman, B. A.; Kane, D. J. Measuring ultrashort laser pulses in the time-frequency domain using frequency-resolving optical gating. *Review of Scientific Instruments* **1997**, *68*, 3277.
- (189) Winker, B. K.; Wright, J. C. Nonlinear Atomic Spectroscopy of Flames. Analytical Chemistry 1988, 60, 2599.
- (190) Kumar, V.; Casella, M.; Molotokaite, E.; Gatti, D.; Kukura, P.; Manzoni, C.; Polli, D.; Marangoni, M.; Cerullo, G. Balanced-detection Raman-induced Kerr-effect spectroscopy. *Physical Review A* 2012, 86, 053810.
- (191) An, Y. Q.; Cundiff, S. T. Phase inversion in rotational anisotropy of second harmonic generation at Si(001) interfaces. *Physical Review B* 2003, 67, 193302.
- (192) Li, D.; Xiong, W.; Jiang, L.; Xiao, Z.; Golgir, H. R.; Wang, M.; Huang, X.; Zhou, Y.; Lin, Z.; Song, J.; Ducharme, S.; Jiang, L.; Silvain, J.-F.; Lu, Y. Multimodal Nonlinear Optical Imaging of MoS<sub>2</sub> and MoS<sub>2</sub>-Based van der Waals Heterostructures. ACS Nano 2016, 10, 3766.
- (193) Zhou, X.; Cheng, J.; Zhou, Y.; Cao, T.; Hong, H.; Liao, Z.; Wu, S.; Peng, H.; Liu, K.; Yu, D. Strong Second-Harmonic Generation in Atomic Layered GaSe. *Journal of the American Chemical Society* 2015, 137, 7994.
- (194) Veitas, G., Personal Communication.
- (195) Kleinman, D. A. Nonlinear Dielectric Polarization in Optical Media. *Physical Review* 1962, 126, 1977.
- (196) Powers, P. E., Fundamentals of Nonlinear Optics; CRC Press: Boca Raton, FL, 2011.
- (197) Johnson, L. W.; Riess, R. D.; Arnold, J. T., Introduction of Linear Algebra, Fifth; Addison Wesley: Boston, 2002.
- (198) Akhmanov, S. A.; Khokhlov, R. V., Problems of Nonlinear Optics; Electromagnetic Waves in Nonlinear Dispersive Media; Gordon and Breach: New York, 1972.
- (199) Bakker, H. J.; Planken, P. C. M.; Muller, H. G. Numerical calculation of optical frequency-conversion processes: a new approach. *The Journal of the Optical Society of America B* **1989**, *6*, 1665.
- (200) Gale, G. M.; Hache, F.; Cavallari, M. Broad-Bandwidth Parametric Amplification in the Visible: Femtosecond Experiments and Simulations. *IEEE Journal of Selected Topics in Quantum Electronics* 1998, 4, 224.

- (201) Gale, G. M.; Cavallari, M.; Hache, F. Femtosecond visible optical parametric oscillator. Journal of the Optical Society of America B 1998, 15, 702.
- (202) Moosmüller, H.; Anderson, S. B.; She, C. Y. Generation of four-wave mixing and stimulated Raman signals in focused Gaussian beams. *Physical Review A* 1989, 40, 6977.
- (203) Bjorklund, G. C. Effects of Focusing on Third-Order Nonlinear Processes in Isotropic Media. IEEE Journal of Quantum Electronics 1975, QE-11, 287.
- (204) Herring, G. C.; Hillard Jr., M. E. Effect of crossed beams on Stokes gain and interaction length in stimulated Raman gain spectroscopy. *Applied Optics* **1995**, *34*, 6837.
- (205) Teets, R. E. CARS signals: phase matching, transverse modes, and optical damage effects. Applied Optics 1986, 25, 855.
- (206) Demirdöven, N.; Khalil, M.; Golonzka, O.; Tokmakoff, A. Dispersion compensation with optical materials for compression of intense sub-100-fs mid-infrared pulses. *Optics Letters* 2002, 27, 433.
- (207) Fork, R. L.; Brito Cruz, C. H.; Becker, P. C.; Shank, C. V. Compression of optical pulses to six femtoseconds by using cubic phase compensation. *Optics Letters* **1987**, *12*, 483.
- (208) Zhang, D.; Kong, Y.; Zhang, J.-y. Optical parametric properties of 532-nm-pumped beta-bariumborate near the infrared absorption edge. Optics Communications 2000, 184, 485.
- (209) Kenyon, I. R., The Light Fantastic: A Modern Introduction to Classical and Quantum Optics, Second; Oxford University Press: Oxford, 2011.
- (210) Azimov, B. S.; Karamzin, Y. N.; Sukhorukov, A. P.; Sukhorukova, A. K. Interaction of weak pulses with a low-frequency high-intensity wave in a dispersive medium. *Soviet Physics Journal of Experimental and Theoretical Physics* 1980, 51, 40.
- (211) Gale, G. M.; Cavallari, M.; Driscoll, T. J.; Hache, F. Generation of highly coherent tunable midinfrared femtosecond pulses by parametric frequency-mixing at 82 MHz. Optics Communications 1995, 119, 159.
- (212) Manassah, J. T. Effects of velocity dispersion on a generated second harmonic signal. Applied Optics 1988, 27, 4365.
- (213) Manassah, J. T. Amplitude and phase of a pulsed second-harmonic signal. Journal of the Optical Society of America B 1987, 4, 1235.
- (214) Manassah, J. T.; Cockings, O. R. Induced phase modulation of a generated second-harmonic signal. Optics Letters 1987, 12, 1005.

- (215) Nishikawa, T.; Uesugi, N. Effects of walk-off and group velocity difference on the optical parametric generation in KTiOPO<sub>4</sub> crystals. *Journal of Applied Physics* **1995**, *77*, 4941.
- (216) Pang, D.; Zhang, R.; Wang, Q. Theoretical Analysis of noncollinear phase-matched optical parametric amplifier seeded by a white-light continuum. *Optics Communications* **2001**, *196*, 293.
- (217) Reisner, S.; Gutmann, M. Numerical treatment of UV-pumped, white-light-seeded single-pass noncollinear parametric amplifiers. *Journal of the Optical Society of America B* **1999**, *16*, 1801.
- (218) Press, W. H.; Teukolsky, S. A.; Vetterling, W. T.; Flannery, B. P., Numerical Recipes in C: The Art of Scientific Computing, Second; Cambridge University Press: Cambridge, 1992.
- (219) Moulton, P. F. Spectroscopic and laser characteristics of Ti:Al<sub>2</sub>O<sub>3</sub>. Journal of the Optical Society of America B 1986, 3, 125.
- (220) Cundiff, S. F.; Knox, W. H.; Ippen, E. P.; Haus, H. A. Frequency-dependent mode size in broadband Kerr-lens mode locking. *Optics Letters* **1996**, *21*, 662.
- (221) Fluck, R.; Jung, I. D.; Zhang, G.; Kärtner, F. X.; Keller, U. Broadband saturable absorber for 10-fs pulse generation. Optics Letters 1996, 21, 743.
- (222) Jung, I. D.; Kärtner, F. X.; Matuschek, N.; Sutter, D. H.; Morier-Genoud, F.; Zhang, G.; Keller, U. Self-starting 6.5-fs pulses from a Ti:sapphire laser. *Optics Letters* **1997**, *22*, 1009.
- (223) Kasper, A.; Witte, K. J. 10-fs pulse generation from a unidirectional Kerr-lens mode-locked Ti:sapphire ring laser. Optics Letters 1996, 21, 360.
- (224) Xu, L.; Spielmann, C.; Krausz, F.; Szipöcs, R. Ultrabroadband ring oscillator for sub-10-fs pulse generation. Optics Letters 1996, 21, 1259.
- (225) Brabec, T.; Spielmann, C.; Krausz, F. Mode locking in solitary lasers. Optics Letters 1991, 16, 1961.
- (226) Curley, P. F.; Spielmann, C.; Brabec, T.; Krausz, F.; Wintner, E.; Schmidt, A. J. Operation of a femtosecond Ti:sapphire solitary laser in the vicinity of zero group-delay dispersion. *Optics Letters* 1993, 18, 54.
- (227) Knox, W. H. In situ measurement of complete intracavity dispersion in an operating Ti:sapphire femtosecond laser. Optics Letters 1992, 17, 514.
- (228) Zhou, J.; Taft, G.; Huang, C.-P.; Murnane, M. M.; Kapteyn, H. C.; Christov, I. P. Pulse evolution in a broad-bandwidth Ti:sapphire laser. *Optics Letters* **1994**, *19*, 1149.
- (229) Matos, L.; Mücke, O. D.; Jian, C.; Kärtner, F. X. Carrier-envelope phase dynamics and noise analysis in octave-spanning Ti:sapphire lasers. *Optics Express* **2006**, *14*, 2497.

- (230) Fork, R. L.; Martinez, O. E.; Gordon, J. P. Negative dispersion using pairs of prisms. *Optics Letters* 1984, 9, 150.
- (231) Pelouch, W. S.; Powers, P. E.; Tang, C. L. Self-starting mode-locked ring-cavity Ti:sapphire laser. Optics Letters 1992, 17, 1581.
- (232) Spence, D. E.; Kean, P. N.; Sibbett, W. 60-fsec pulse generation from a self-mode-locked Ti:sapphire laser. Optics Letters 1991, 16, 42.
- (233) Salin, F.; Grangier, P.; Georges, P.; Brun, A. Pulse propagation near zero group-velocity dispersion in a femtosecond dye laser. *Optics Letters* **1990**, *15*, 1374.
- (234) Brée, C.; Demircan, A.; Steinmeyer, G. Saturation of the All-Optical Kerr Effect. Physical Review Letters 2011, 106, 183902.
- (235) Roy, R.; Schulz, P. A.; Walther, A. Acousto-optic modulator as an electronically selectable unidirectional device in a ring laser. *Optics Letters* 1987, 12, 672.
- (236) Huggett, G. Mode-Locking of CW Lasers by Regenerative RF Feedback. Applied Physics Letters 1968, 13, 186.
- (237) French, P. M. W.; Williams, J. A. R.; Taylor, J. R. Femtosecond pulse generation from a titaniumdoped sapphire laser using nonlinear external cavity feedback. *Optics Letters* **1989**, *14*, 686.
- (238) Haus, H. A. Mode-Locking of Lasers. *IEEE Journal on Selected Topics in Quantum Electronics* 2000, 6, 1173.
- (239) Hou, D.; Lee, C.-C.; Yang, Z.; Schibli, T. R. Timing jitter characterization of mode-locked lasers with <1 zs/\_/Hz resolution using a simple optical heterodyne technique. *Optics Letters* **2015**, *40*, 2985.
- (240) Haus, H. A.; Shank, C. V.; Ippen, E. P. Shape of Passively Mode-Locked Laser Pulses. Optics Communications 1975, 15, 29.
- (241) Martinez, O. E.; Fork, R. L.; Gordon, J. P. Theory of passively mode-locked lasers including self-phase modulation and group-velocity dispersion. *Optics Letters* **1984**, *9*, 156.
- (242) Sarukura, N.; Ishida, Y.; Nakano, H.; Yamamoto, Y. cw passive mode locking of a Ti:sapphire laser. Applied Physics Letters 1990, 56, 814.
- (243) Salin, F.; Squire, J.; Piché, M. Mode locking of Ti:Al<sub>2</sub>O<sub>3</sub> lasers and self-focusing: a Gaussian approximation. Optics Letters 1991, 16, 1674.
- (244) Piché, M.; Salin, F. S. Self-mode locking of solid-state lasers without apertures. Optics Letters 1993, 18, 1041.

- (245) Rudd, J. V.; Korn, G.; Kane, S.; Squier, J.; Mourou, G. Chirped-pulse amplification of 55-fs pulses at a 1-kHz repetition rate in a Ti:Al<sub>23</sub> regenerative amplifier. *Optics Letters* **1993**, *18*, 2044.
- (246) Barty, C. P. J.; Korn, G.; Raksi, F.; Rose-Petruck, C.; Squier, J.; Tien, A.-C.; Wilson, K. R.; Yakovlev,
   V. V.; Yamakawa, K. Regenerative pulse shaping and amplification of ultrabroadband optical pulses.
   Optics Letters 1996, 21, 219.
- (247) Roth, U.; Loewenthal, F.; Balmer, J. E.; Weber, H. P. Compensation of Nonlinear Self-Focusing in High-Power Lasers. *IEEE Journal of Quantum Electronics* 2000, *36*, 687.
- (248) Lenzner, M.; Spielmann, C.; Wintner, E.; Krausz, F.; Schmidt, A. J. Sub-20-fs, kilohertz-repetitionrate Ti:sapphire amplifier. *Optics Letters* 1995, 20, 1397.
- (249) Lemoff, B. E.; Barty, C. P. J. Quintic-phase-limited, spatially uniform expansion and recompression of ultrashort optical pulses. *Optics Letters* 1993, 18, 1651.
- (250) Pessot, M.; Maine, P.; Mourou, G. 1000 Times Expansion/Compression of Optical Pulses for Chirped Pulse Amplification. Optics Communications 1987, 62, 419.
- (251) Strickland, D.; Mourou, G. Compression of Amplified Chirped Optical Pulses. Optics Communications 1985, 56, 219.
- (252) Perry, M. D.; Ditmire, T.; Stuart, B. C. Self-phase modulation in chirped pulse amplification. Optics Letters 1994, 19, 2149.
- (253) Wynne, K.; Reid, G. D.; Hochstrasser, R. M. Regenerative amplification of 30-fs pulses in Ti:sapphire at 5 kHz. Optics Letters 1994, 19, 895.
- (254) Konoplev, O. A.; Meyerhofer, D. D. Cancellation of B-Integral Accumulation for CPA Lasers. IEEE Journal of Selected Topics in Quantum Electronics 1998, 4, 459.
- (255) Wong, K. S.; Qui, Z. R.; Wang, H.; Wong, G. K. L. Efficient visible femtosecond optical parametric generator and amplifier using tilted pulse-front pumping. *Optics Letters* **1997**, *22*, 898.
- (256) Fraser, J. M.; Hall, K. C. Interferometric nonlinear mixing in multiple-pass femtosecond optical parametric amplification. *Optics Express* **1999**, *5*, 21.
- (257) Chien, C. Y.; Korn, G.; Coe, J. S.; Squier, J.; Mourou, G.; Craxton, R. S. Highly efficient secondharmonic generation of ultraintense Nd:glass laser pulses. *Optics Letters* **1995**, 20, 353.
- (258) Fendt, A.; Kranitzky, W.; Laubereau, A.; Kaiser, W. Efficient Generation of Tunable Subpicosecond Pulses in the Infrared. Optics Communications 1979, 28, 142.

- (259) Wang, Y.; Luther-Davies, B. Frequency-doubling pulse compressor for picosecond high-power neodymium laser pulses. *Optics Letters* **1992**, *17*, 1459.
- (260) Chen, Y.-C.; Fang, X.-J.; Li, J.; Liang, X.-Y.; Zhang, H.-L.; Feng, B.-H.; Zhang, X.-L.; Xu, Z.-Y. Efficient femtosecond optical parametric generator with a birefringent delay compensator. *Applied Optics* 2001, 40, 2579.
- (261) Danielius, R.; Piskarskas, A.; Di Trapani, P.; Andreoni, A.; Solcia, C.; Foggi, P. Matching of group velocities by spatial walk-off in collinear three-wave interaction with tilted pulses. *Optics Letters* 1996, 21, 973.
- (262) Danielius, R.; Dubietis, A.; Piskarskas, A.; Valiulis, G.; Varanavičius, A. Generation of compressed 600-720-nm tunable femtosecond pulses by transient frequency mixing in a β-barium borate crystal. Optics Letters 1996, 21, 216.
- (263) Greenfield, S. R.; Wasielewski, M. R. Optical parametric amplification of femtosecond pulses tunable from the blue to the infrared with microjoule energies. *Applied Optics* **1995**, *34*, 2688.
- (264) Nisoli, M.; De Silvestri, S.; Valiulis, G.; Varanavicius, A. Fivefold femtosecond pulse compression by sum frequency generation. *Applied Physics Letters* **1996**, *68*, 3540.
- (265) Yakovlev, V. V.; Kohler, B.; Wilson, K. R. Broadly tunable 30-fs pulses produced by optical parametric amplification. *Optics Letters* **1994**, *19*, 2000.
- (266) Wachman, E. S.; Edelstein, D. C.; Tang, C. L. Continuous-wave mode-locked and dispersion-compensated femtosecond optical parametric oscillator. *Optics Letters* **1990**, *15*, 136.
- (267) Wasylczyk, P.; Walmsley, I. A.; Wasilewski, W.; Radzewicz, C. Broadband noncollinear optical parametric amplifier using a single crystal. *Optics Letters* 2005, *30*, 1704.
- (268) Wei, P.-F.; Li, C.; Zhang, C.-M.; Wang, J.-L.; Leng, Y.-X.; Li, R.-X. High-Conversion-Efficiency and Broadband Tunable Femtosecond Noncollinear Optical Parametric Amplifier. *Chinese Physical Letters* 2008, 25, 2514.
- (269) Wei, P.; Liu, J.; Li, C. Temporal and spectral characterizations of tunable noncollinear optical parametric amplifier. Optics Communications 2013, 286, 334.
- (270) Gloster, L. A. W.; McKinnie, I. T.; King, T. A. Noncollinear phase matching in a type I barium borate optical parametric oscillator. *Optics Communications* **1994**, *112*, 328.
- (271) Laenan, R.; Graener, H.; Lauberau, A. Broadly tunable femtosecond pulses generated by optical parametric oscillation. Optics Letters 1990, 15, 971.

- (272) Danielius, R.; Piskarskas, A.; Di Trapani, P.; Andreoni, A.; Solcia, C.; Foggi, P. Visible pulses of 100 fs and 100 µJ from an upconverted parametric generator. *Applied Optics* **1996**, *35*, 5336.
- (273) Joosen, W.; Agostini, P.; Petite, G.; Chambaret, J. P.; Antonetti, A. Broadband femtosecond infrared parametric amplification in β-BaB<sub>2</sub>O<sub>4</sub>. Optics Letters **1992**, 17, 133.
- (274) Dubietis, A.; Valiulis, G.; Tamošsauskas, G.; Danielius, R.; Piskarskas, A. Nonlinear second-harmonic pulse compression with tilted pulses. *Optics Letters* **1997**, *22*, 1071.
- (275) Bromage, J.; Rothhardt, J.; Hädrich, S.; Dorrer, C.; Jocher, C.; Demmler, S.; Limpert, J.; Tünnermann, A.; Zuegel, J. D. Analysis and suppression of parasitic processes in noncollinear optical parametric amplifiers. *Optics Express* **2011**, *19*, 16797.
- (276) Gehr Russell, J.; Kimmel, M. W.; Smith, A. V. Simultaneous spatial and temporal walk-off compensation in frequency-doubling femtosecond pulses in β-BaB<sub>2</sub>O<sub>4</sub>. Optics Letters **1998**, 23, 1298.
- (277) Wilhelm, T.; Piel, J.; Riedle, E. Sub-20-fs pulses tunable across the visible from a blue-pumped single-pass noncollinear parametric converter. *Optics Letters* **1997**, *22*, 1494.
- (278) Osvay, K.; Ross, I. N. Broadband sum-frequency generation by chirp-assisted group-velocity matching. Journal of the Optical Society of America B 1996, 13, 1431.
- (279) Danielius, R.; Dubietis, A.; Valiulis, G.; Piskarskas, A. Femtosecond high-contrast pulses from a parametric generator pumped by the self-compressed second harmonic a Nd:glass laser. Optics Letters 1995, 20, 2225.
- (280) Grün, A.; Austin, D. R.; Cousin, S. L.; Biegert, J. Three-wave mixing mediated femtosecond pulse compression in β-barium borate. *Optics Letters* **2015**, 40, 4679.
- (281) Ziegler, L. D.; Morais, J.; Zhou, Y.; Constantine, S.; Reed, M. K.; Steiner-Shepard, M. K.; Lommel,
   D. Tunable 50-fs Pulse Generation in the 250-310-nm Ultraviolet Range. *IEEE Journal of Quantum Electronics* 1998, 34, 1758.
- (282) Danielius, R.; Di Trapani, P.; Dubietis, A.; Piskaraskas, A.; Podenas, D.; Banfi, G. P. Self-diffraction through cascaded second-order frequency-mixing effects in β-barium borate. Optics Letters 1993, 18, 574.
- (283) Liu, W.; Zhu, L.; Fang, C. Observation of sum-frequency-generation-induced cascaded four-wave mixing using two crossing femtosecond laser pulses in a 0.1 mm beta-barium-borate crystal. Optics Letters 2012, 37, 3783.

- (284) Liu, W.; Zhu, L.; Wang, L.; Fang, C. Cascaded four-wave mixing for broadband tunable laser sideband generation. Optics Letters 2013, 38, 1772.
- (285) Petrov, V.; Noack, F. Frequency upconversion of tunable femtosecond pulses by parametric amplification and sum-frequency generation in a single nonlinear crystal. *Optics Letters* **1995**, *20*, 2171.
- (286) Sueptitz, M.; Kaindl, R. A.; Lutgen, S.; Woerner, M.; Riedle, E. 1 kHz solid state laser system for the generation of 50 fs pulses tunable in the visble. *Optics Communications* **1996**, *131*, 195.
- (287) Eimerl, D.; Davis, L.; Velsko, S.; Graham, E. K.; Zalkin, A. Optical, mechanical, and thermal properties of barium borate. *Journal of Applied Physics* **1987**, *62*, 1968.
- (288) Kato, K. Second-Harmonic Generation to 2048 Å in β-BaB<sub>2</sub>O<sub>4</sub>. *IEEE Journal of Quantum Electronics* 1986, *QE-22*, 1013.
- (289) Lu, M.-H.; Liu, Y.-M. Infrared up-conversion with beta barium borate crystal. Optics Communications 1991, 84, 193.
- (290) Freund, J. E. Mathematical Statistics. 1992.
- (291) Engel, T.; Reid, P., *Physical Chemistry*; Prentice Hall: Upper Saddle River, NJ, 2006.
- (292) Nautiyal, A.; Bisht, P. B. Broadly tunable parametric line emission from β-barium borate on pumping with picosecond pulses. Optics Communications 2008, 281, 3351.
- (293) Davis, T. S.; Fackler, J. P.; Weeks, M. J. Spectra of Manganese (III) Complexes. The Origin of the Low-Energy Band. *Inorganic Chemistry* 1968, 7, 1994.
- (294) Dingle, R. The Visible and Near Infrared Spectrum of Manganese (III) Complexes. Acta Chemica Scandinavica 1966, 20, 33.
- (295) Dingle, R. The Jahn-Teller Effect in Manganic Acetylacetonate. Journal of Molecular Spectroscopy 1962, 9, 426.
- (296) Diaz-Acosta, I.; Baker, J.; Hinton, J. F.; Pulay, P. Calculated and experimental geometries and infrared spectra of metal tris-acetylacetonates: vibrational spectroscopy as a probe of molecular structure for ionic complexes. Part II. Spectrochimica Acta Part A 2003, 59, 363.
- (297) Stults, B. R.; Marianelli, R. S.; Day, V. W. Distortions of the Coordination Polyhedron in High-Spin Manganese(III) Complexes. I. Synthesis and Characterization of a Series of Five- and Six-Coordinate Bis(acetylacetonato) manganese(III) Complexes. Crystal Structure of Azidobis(acetyl-acetonato) manganese(III). Inorganic Chemistry 1975, 14, 722.

- (298) Stults, B. R.; Day, R. O.; Marianelli, R. S.; Day, V. W. Distortions of the Coordination Polyhedron in High-Spin Manganese(III) Complexes. 2. Crystal Structure of Thiocyanatobis(acetylacetonato) manganese(III). *Inorganic Chemistry* 1979, 18, 1847.
- (299) Stults, B. R.; Marianelli, R. S.; Day, V. W. Distortions of the Coordination Polyhedron in High-Spin Manganese(III) Complexes. 3. Crystal and Molecular Structure of γ-Tris(acetylacetonato) man-ganese(III): A Tetragonally Elongated Octahedral Form. *Inorganic Chemistry* **1979**, *18*, 1853.
- (300) Juban, E. A.; McCusker, J. K. Ultrafast Dynamics of <sup>2</sup>E State Formation in Cr(acac)<sub>3</sub>. Journal of the American Chemical Society 2005, 127, 6857.
- (301) McCusker, J. K. Femtosecond Absorption Spectroscopy of Transition Metal Charge-Transfer Complexes. Accounts of Chemical Research 2003, 36, 876.
- (302) Schrauben, J. N.; Dillman, K. L.; Beck, W. F.; McCusker, J. K. Vibrational coherence in the excited state dynamics of Cr(acac)<sub>3</sub>: probing the reaction coordinate for ultrafast intersystem crossing. *Chemical Science* **2010**, *1*, 405.
- (303) Fielding, P. E.; MacKay, A. G. N.I.R. Absorption Spectra of Phthalocyanines. Australian Journal of Chemistry 1975, 28, 1445.
- (304) Grobosch, M.; Schmidt, C.; Kraus, R.; Knupfer, M. Electronic properties of transition metal phthalocyanines: The impact of the central metal atom (d<sup>5</sup>-d<sup>10</sup>). Organic Electronics **2010**, *11*, 1483.
- (305) Kraus, R.; Grobosch, M.; Knupfer, M. Full electronic excitation spectrum of condensed manganese phthalocyanine. *Chemical Physics Letters* 2009, 469, 121.
- (306) Liu, K.; Lei, Y.; Wang, G. Correlation between oxygen adsorption energy and electronic structure of transition metal macrocyclic complexes. *The Journal of Chemical Physics* **2013**, *139*, 204306.
- (307) Ghani, F.; Kristen, J.; Riegler, H. Solubility Properties of Unsubstituted Metal Phthalocyanines in Different Types of Solvents. *Journal of Chemical & Engineering Data* 2012, 57, 439.
- (308) Boucher, L. J. Manganese Porphyrin Complexes. III. Spectroscopy of Chloroaquo Complexes of Several Porphyrins. Journal of the American Chemical Society 1970, 92, 2725.
- (309) Gaughan, R. R.; Shriver, D. F.; Boucher, L. J. Resonance Raman Spectra of Manganese(III) Tetraphenylporphin Halides. Proceedings of the National Academy of Sciences of the United States of America 1975, 72, 433.
- (310) Gale, R.; Peacock, R. D.; Samori, B. The Linear Dichroism Spectra of Free Base and Manganese(III) Porphyrins. *Chemical Physics Letters* 1976, 37, 430.
- (311) Humphrey, J. L.; Kuciauskas, D. Charge Transfer Enhances Two-Photon Absorption in Transition Metal Porphyrins. *Journal of the American Chemical Society* 2006, 128, 3902.
- (312) Ha, J.-H.; Chae, W.-S.; Kim, J. Tripmultiplet-state dynamics of manganese(III) porphyrin fabricated into PVC film. Journal of Analytical Science & Technology 2011, 2, 69.
- (313) Gonçalves, P. J.; De Boni, L.; Borissevitch, I. E.; Zílio, S. C. Excited State Dynamics of meso-Tetra(sulphonatophenyl) Metalloporphyrins. *The Journal of Physical Chemistry A* 2008, 112, 6522.
- (314) Irvine, M. P.; Harrison, R. J.; Strahand, M. A.; Beddard, G. S. Picosecond Spectroscopy and Kinetics of Metalloporphyrins. *Berichte der Bunsengesellschaft fur Physikalische Chemie* **1985**, *86*, 226.
- (315) Yanghee, K.; Choi, J. R.; Yoon, M.; Furube, A.; Asahi, T.; Masuhara, H. Excited-State Dynamics of 5,10,15,20-Tetraphenyl-21H,23H-porphine Manganese(III) Chloride Encapsulated in TiMCM-41 and MCM-41; Proved by fs-Diffuse Reflectance Laser Photolysis. *The Journal of Physical Chemistry B* 2001, 105, 8513.
- (316) Yan, X.; Kirmaier, C.; Holten, D. A Picosecond Study of Rapid Multistep Radiationless Decay in Manganese(III) Porphyrins. *Inorganic Chemistry* **1986**, 25, 4774.
- (317) Jeoung, S. C.; Kim, D.; Cho, D. W. Transient resonance Raman spectroscopic studies of some paramagnetic metalloporphyrins: effects of axial ligand on charge-transfer and photoreduction processes. *Journal of Raman Spectroscopy* 2000, *31*, 319.
- (318) Brodeur, A.; Chin, S. L. Band-Gap Dependence on the Ultrafast White-Light Continuum. Physics Review Letters 1998, 80, 4406.
- (319) Nagura, C.; Suda, A.; Kawano, K.; Obara, M.; Midorikawa, K. Generation and characterization of ultrafast white-light continuum in condensed media. *Applied Optics* 2002, 41, 3735.
- (320) Brodeur, A.; Chin, S. L. Ultrafast white-light continuum generation and self-focusing in transparent condensed media. *Journal of Optical Society of America B* 1999, 16, 637.
- (321) Baltuška, A.; Fuji, T.; Kobayashi, T. Controlling the Carrier-Envelope Phase of Ultrashort Light Pulses with Optical Parametric Amplifiers. *Physical Review Letters* 2002, *88*, 133901.
- (322) Baum, P.; Lochbrunner, S.; Piel, J.; Riedle, E. Phase-coherent generation of tunable visible femtosecond pulses. *Optics Letters* 2003, 28, 185.
- (323) Baum, P.; Riedle, E.; Greve, M.; Telle, H. R. Phase-locked ultrashort pulse trains at separate and independently tunable wavelengths. *Optics Letters* 2005, 30, 2028.

- (324) Bellini, M.; Hänsch, T. W. Phase-locked white-light continuum pulses: toward a universal optical frequency-comb synthesizer. Optics Letters 2000, 25, 1049.
- (325) Corsi, C.; Tortora, A.; Bellini, M. Mutual coherence of supercontinuum pulses collinearly generated in bulk media. *Applied Physics B* **2003**, *77*, 285.
- (326) Gaundalini, A.; Eckle, P.; Anscombe, M.; Schlup, P.; Biegert, J.; Keller, U. 5.1 fs pulses generated by filamentation and carrier envelope phase stability analysis. *Journal of Physics B: Atomic, Molecular* and Optical Physics 2006, 39, S257.
- (327) Béjot, P.; Kasparian, J.; Henin, S.; Loriot, V.; Vieillard, T.; Hertz, E.; Faucher, O.; Lavorel, B.;
   Wolf, J.-P. Higher-Order Kerr Terms Allow Ionization-Free Filamentation in Gases. *Physical Review Letters* 2010, 104, 103903.
- (328) Bradler, M.; Baum, P.; Riedle, E. Femtosecond continuum generation in bulk laser host materials with sub-μJ pump pulses. Applied Physics B 2009, 97, 561.
- (329) Couairon, A.; Franco, M.; Mysyrowicz, A.; Biegert, J.; Keller, U. Pulse self-compression to the singlecycle limit by filamentation in a gas with a pressure gradient. *Optics Letters* **2005**, *30*, 2657.
- (330) Kasparian, J.; Sauerbrey, R.; Chin, S. L. The critical laser intensity of self-guided light filaments in air. Applied Physics B 2000, 71, 877.
- (331) Méjean, G.; Kasparian, J.; Yu, J.; Frey, S.; Salmon, E.; Ackermann, R.; Wolf, J. P.; Bergé, L.; Skupin,
   S. UV-Supercontinuum generated by femtosecond pulse filamentation in air: Meter-range experiments
   versus numerical simulations. *Applied Physics B* 2006, *82*, 341.
- (332) Mikalauskas, D.; Dubietis, A.; Danielius, R. Observation of light filaments induced in air by visible picosecond laser pulses. *Applied Physics B* **2002**, *75*, 899.
- (333) Nguyen, N. T.; Saliminia, A.; Liu, W.; Chin, S. L.; Vall'ee, R. Optical Breakdown versus filamentation in fused silica by use of femtosecond infrared laser pulses. *Optics Letters* 2003, 28, 1591.
- (334) Petrarca, M.; Petit, Y.; Delagrange, R.; Béjot, P.; Kasparian, J. Higher-order Kerr improve quantitative modeling of laser filamentation. *Optics Letters* 2012, *37*, 4347.
- (335) Wang, H.; Fan, C.; Shen, H.; Zhang, P. F.; Qiao, C. H. Relative contributions of higher-order Kerr effect and plasma in laser filamentation. *Optics Communications* **2013**, *293*, 113.
- (336) Aközbek, N.; Scalora, M.; Bowden, C. M.; Chin, S. L. White-light continuum generation and filamentation during the propagation of ultra-short laser pulses in air. Optics Communications 2001, 191, 353.

- (337) Kohl-Landgraf, J.; Nimsch, J.-E.; Wachtveitl, J. LiF, an underestimated supercontinuum source in femtosecond transient absorption spectroscopy. *Optics Express* **2013**, *21*, 17060.
- (338) Hauri, C. P.; Guandalini, A.; Eckle, P.; Kornelis, W.; Biegert, J.; Keller, U. Generation of intense few-cycle laser pulses through filamentation parameter dependence. *Optics Express* **2005**, *13*, 7541.
- (339) Jang, D. G.; Nam, I. H.; Kim, M. S.; Kang, K.; Suk, H. Generation of broadband supercontinuum light by double-focusing of a femtosecond laser pulse in air. *Applied Physics Letters* **2015**, *107*, 131105.
- (340) Liu, X.-L.; Lu, X.; Liu, X.; Feng, L.-B.; Ma, J.-L.; Li, Y.-T.; Chen, L.-M.; Dong, Q.-L.; Wang, W.-M.;
  Wang, Z.-H.; Wei, Z.-Y.; Sheng, Z.-M.; Jie, Z. Broadband supercontinuum generation in air using tightly focused femtosecond laser pulses. *Optics Letters* 2011, *36*, 3900.
- (341) Spokoyny, B.; Koh, C. J.; Harel, E. Stable and high-power few cycle supercontinuum for 2D ultrabroadband electronic spectroscopy. *Optics Letters* 2015, 40, 1014.
- (342) Fork, R. L.; Shank, C. V.; Hirlimann, C.; Yen, R.; Tomlinson, W. J. Femtosecond white-light continuum pulses. *Optics Letters* **1983**, *8*, 1.
- (343) Ashcom, J. B.; Gattass, R. R.; Schaffer, C. B.; Mazur, E. Numerical aperture dependence of damage and supercontinuum generation from femtosecond laser pulses in bulk fused silica. *Journal of the Optical Society of America B* 2006, 23, 2317.
- (344) Nikolaitchik, A. V.; Korth, O.; Rodgers, M. A. J. Crown Ether Substituted Monomeric and Cofacial Dimeric Metallophthalocyanines. 1. Photophysical Studies of the Free Base, Zinc(II), and Copper(II) Variants. The Journal of Physical Chemistry A 1999, 103, 7587.
- (345) Hybl, J. D.; Albrecht, A. W.; Gallagher Faeder, S. M.; Jonas, D. M. Two-dimensional electronic spectroscopy. *Chemical Physics Letters* 1998, 297, 307.
- (346) Cong, P.; Deuel, H. P.; Simon, J. D. Using optical coherence to measure the ultrafast dephasing of large molecules in room-temperature liquids. *Chemical Physics Letters* **1993**, *212*, 367.
- (347) Maçôas, E. M. S.; Kananavicius, R.; Myllyperkiö, P.; Pettersson, M.; Kunttu, H. Relaxation Dynamics of Cr(acac)<sub>3</sub> Probed by Ultrafast Infrared Spectroscopy. *Journal of the American Chemical Society* 2007, 129, 8934.
- (348) Nie, Z.; Long, R.; Sun, L.; Huang, C.-C.; Zhang, J.; Xiong, Q.; Hewak, D. W.; Shen, Z.; Prezhdo,
   O. V.; Loh, Z.-H. Ultrafast Carrier Thermalization and Cooling Dynamics in Few-Layer MoS<sub>2</sub>. ACS Nano 2014, 8, 10931.

- (349) Palfrey, S. L.; Heinz, T. F. Coherent interactions in pump-probe absorption measurements: the effect of phase gratings. *Journal of the Optical Society of America B* **1985**, *2*, 674.
- (350) Tomita, M.; Matsuoka, M. Ultrafast pump-probe measurement using intensity correlation of incoherent light. *Journal of the Optical Society of America B* **1986**, *3*, 560.
- (351) Wolfseder, B.; Seidner, L.; Domcke, W.; Stock, G.; Seel, M.; Engleitner, S.; Zinth, W. Vibrational coherence in ultrafast electron-transfer dynamics of oxazine 1 in N,N-dimethylaniline: simulation of a femtosecond pump-probe experiment. *Chemical Physics* **1998**, *1998*, 323.
- (352) Carlson, R. J.; Wright, J. C. Absorption and Coherent Interference Effects in Multiply Resonant Four-Wave Mixing Spectroscopy. Applied Spectroscopy 1989, 43, 1195.
- (353) Evans, C. L.; Potma, E. O.; Sunney Xie, X. Coherent anti-Stokes Raman scattering spectral interferometry: determination of the real and imaginary components of nonlinear susceptibility  $\chi^{(3)}$  for vibrational microscopy. *Optics Letters* **2004**, *29*, 2923.
- (354) Müller, C.; Buckup, T.; von Vacano, B.; Motzkus, M. Heterodyne single-beam CARS microscopy. Journal of Raman Spectroscopy 2009, 40, 8099.
- (355) Potma, E. O.; Evans, C. L.; Sunney Xie, X. Heterodyne coherent anti-Stokes Raman scattering (CARS) imaging. Optics Letters 2006, 31, 241.
- (356) Pegoraro, A. F.; Ridsdale, A.; Moffatt, D. J.; Jia, Y.; Pezacki, J. P.; Stolow, A. Optimally chirped multimodal CARS microscopy based on a single Ti:sapphire oscillator. Optics Express 2009, 17, 2984.
- (357) Brückner, L.; Buckup, T.; Motzkus, M. Enhancement of coherent anti-Stokes Raman signal via tailored probing in spectral focusing. *Optics Letters* 2015, 40, 5204.
- (358) Kohler, D. D., Personal Communication.
- (359) Krishna, M. B. M.; Rao, D. N. Influence of solvent contribution on nonlinearities of near infra-red absorbing croconate and squaraine dyes with ultrafast laser excitation. *Journal of Applied Physics* 2013, 114, 133103.
- (360) Shen, Y.; Voronine, D. V.; Sokolov, A. V.; Scully, M. O. A versatile setup using femtosecond adaptive spectroscopic techniques for coherent anti-Stokes Raman scattering. *Review of Scientific Instruments* 2015, 86, 083107.
- (361) Song, J. J.; Eesley, G. L.; Levenson, M. D. Applied Physics Letters 1976, 29, 567.

- (362) Brixner, T.; Mančal, T.; Stiopkin, I. V.; Fleming, G. R. Phase-stabilized two-dimensional electronic spectroscopy. *The Journal of Chemical Physics* **2004**, *121*, 4221.
- (363) Christensson, N.; F., M.; J., H.; Sperling, J.; Bixner, O.; Nemeth, A.; Kauffmann, H. F. High Frequency Vibrational Modulations in Two-Dimensional Electronic Spectra and Their Resemblance to Electronic Coherence Signatures. *The Journal of Physical Chemistry B* 2011, 115, 5383.
- (364) Harel, E.; Long, P. D.; Engel, G. S. Single-shot ultrabroadband two-dimensional electronic spectroscopy of the light-harvesting complex LH2. *Optics Letters* **2011**, *36*, 1665.
- (365) Harel, E.; Rupich, S. M.; Schaller, R. D.; Talapin, D. V.; Engel, G. S. Measurement of electronic splitting in PbS quantum dots by two-dimensional nonlinear spectroscopy. *Physical Review B* 2012, 86, 075412.
- (366) Lynch, M. S.; Slenkamp, K. M.; Cheng, M.; Khalil, M. Coherent Fifth-Order Visible-Infrared Spectroscopies: Ultrafast Nonequilibrium Vibrational Dynamics in Solution. *The Journal of Physical Chemistry A* 2012, *116*, 7023.
- (367) Mančal, T.; Nemeth, A.; Milota, F.; Lukeš, V.; Kauffmann, H. F.; Sperling, J. Vibrational wave packet induced oscillations in two-dimensional electronic spectra. II. Theory. *The Journal of Chemical Physics* 2010, 132, 184515.
- (368) Nemeth, A.; Milota, F.; Mančal, T.; Lukeš, V.; Kauffmann, H. F.; Sperling, J. Vibronic modulation of lineshapes in two-dimensional electronic spectra. *Chemical Physics Letters* 2008, 459, 94.
- (369) Nemeth, A.; Lukeş, V.; Sperling, J.; Milota, F.; Kauffmann, H. F.; Mančal, T. Two-dimensional electronic spectra of an aggregating dye: simultaneous measurement of monomeric and dimeric lineshapes. *Physical Chemistry Chemical Physics* 2009, 11, 5986.
- (370) Tekavec, P. F.; Myers, J. A.; Lewis, K. L. M.; Ogilvie, J. P. Two-dimensional electronic spectroscopy with a continuum probe. *Optics Letters* **2009**, *34*, 1390.
- (371) Zhang, Y.; Meyer, K.; Ott, C.; Pfeifer, T. Passively phase-stable, monolithic, all-reflective twodimensional electronic spectroscopy based on a four-quandrant mirror. *Optics Letters* 2013, 38, 356.
- (372) Block, S. B., Personal Communication.
- (373) Robinson, J. W.; Skelly Frame, E. M.; Frame II, G. M., Undergraduate Instrumental Analysis, Sixth; Marcel Dekker: New York, 2005.
- (374) Fox, M., Quantum Optics: An Introduction; Oxford University Press: Oxford, 2006.

- (375) Kee, T. W.; Cicerone, M. T. Simple approach to one-laser, broadband coherent anti-Stokes Raman scattering microscopy. Optics Letters 2004, 29, 2701.
- (376) Knutsen, K. P.; Johnson, J. C.; Miller, A. E.; Petersen, P. B.; Saykally, R. J. High spectral resolution multiplex CARS spectroscopy using chirped pulses. *Chemical Physics Letters* 2004, 387, 436.
- (377) Heiman, D.; Hellwarth, R. W.; Levenson, M. D.; Martin, G. Raman-Induced Kerr Effect. Physical Review Letters 1976, 36, 189.
- (378) McCamant, D. W.; Kukura, P.; Yoon, S.; Mathies, R. A. Femtosecond broadband stimulated Raman spectroscopy: Apparatus and methods. *Review of Scientific Instruments* **2004**, *75*, 4971.
- (379) Klenerman, D. Multiplexed optically heterodyned Raman-induced Kerr-effect spectroscopy in the blue. Optics Letters 1991, 16, 838.
- (380) Brinkmann, M.; Dobner, S.; Fallnich, C. Light source for narrow and broadband coherent Raman scattering microspectroscopy. *Optics Letters* **2015**, *40*, 5447.
- (381) Duncan, M. D.; Reintjes, J.; Manuccia, T. J. Scanning coherent anti-Stokes Raman microscope. Optics Letters 1982, 7, 350.
- (382) Duncan, M. D.; Reintjes, J.; Manuccia, T. J. Imaging biological compounds using the coherent anti-Stokes Raman scattering microscope. *Optical Engineering* **1985**, *24*, 352.
- (383) Demirdöven, N.; Khalil, M.; Tokmakoff, A. Correlated Vibrational Dynamics Revealed by Two-Dimensional Infrared Spectroscopy. *Physical Review Letters* 2002, *89*, 237401.
- (384) Khalil, M.; Demirdöven, N.; Tokmakoff, A. Obtaining Absorptive Line Shapes in Two-Dimensional Infrared Vibrational Correlation Spectra. *Physical Review Letters* 2003, 90, 047401.
- (385) Khalil, M.; Demirdöven, N.; Tokmakoff, A. Vibrational coherence transfer characterized with Fouriertransform 2D IR spectroscopy. *The Journal of Chemical Physics* **2004**, *121*, 362.
- (386) Nee, M. J.; Baiz, C. R.; Anna, J. M.; McCanne, R.; Kubarych, K. J. Multilevel vibrational coherence transfer and wavepacket dynamics probed with multidimensional IR spectroscopy. *The Journal of Chemical Physics* 2008, 129, 084503.
- (387) Osborne, D. G.; Kubarych, K. J. Rapid and Accurate Measurement of the Frequency-Frequency Correlation Function. *The Journal of Physical Chemistry A* 2013, 117, 5891.
- (388) Smeigh, A. L.; Creelman, M.; Mathies, R. A.; McCusker, J. K. Femtosecond TIme-Resolved Optical and Raman Spectroscopy of Photoinduced Spin Crossover: Temporal Resolution of Low-to-High Spin Optical Switching. *Journal of the American Chemical Society* **2008**, *130*, 14105.

- (389) Shim, S.; Dasgupta, J.; Mathies, R. A. Femtosecond Time-Resolved Stimulated Raman Reveals the Birth of Bacteriorhodopsin's J and K Intermediates. *Journal of the American Chemical Society* 2009, 131, 7592.
- (390) Hoffman, D. P.; Valley, D.; Ellis, S. R.; Crewman, M.; Mathies, R. A. Optically shaped narrowband picosecond pulses for femtosecond stimulated Raman spectroscopy. *Optics Express* 2013, 21, 21685.
- (391) Kovalenko, S. A.; Dobryakov, A. L.; Ernsting, N. P. An efficient setup for femtosecond stimulated Raman spectroscopy. *Review of Scientific Instruments* **2011**, *82*, 063102.
- (392) Li, Z.; Li, Z.; Zhou, M.; Wang, Y.; Men, Z.; Sun, C. Stimulated Raman scattering of lattice translational modes in liquid heavy water. *Optics Letters* 2012, *37*, 1319.
- (393) Li, Z.; Shan, X.; Li, Z.; Cao, J.; Zhou, M.; Wang, Y.; Men, Z.; Sun, C. Estimating the pressure of laser-induced plasma shockwave by stimulated Raman shift of lattice translational modes. *Applied Physics Letters* 2012, 101, 029108.
- (394) Rakesh Kumar, V.; Prem Kiran, P. Onset of ice VII phase of liquid water: role of filamentation in stimulated Raman scattering. Optics Letters 2015, 40, 2802.
- (395) Men, Z.; Li, Z.; Zhou, M.; Lu, G.; Zou, B.; Li, Z.; Sun, C. Stimulated Raman scattering from ice-VIII by shock-induced compression in liquid water. *Physical Review B* **2012**, *85*, 092101.
- (396) Yui, H.; Sawada, T. Interaction of Excess Electrons with Water Molecules at the Early Stage of Laser-Induced Plasma Generation in Water. *Physical Review Letters* 2000, 85, 3512.
- (397) Shim, S.; Mathies, R. A. Femtosecond Raman-Induced Kerr effect spectroscopy. Journal of Raman Spectroscopy 2008, 39, 1526.
- (398) Frontiera, R. R.; Shim, S.; Mathies, R. A. Origin of negative and dispersive features in anti-Stokes and resonance femtosecond stimulated Raman spectroscopy. *The Journal of Chemical Physics* 2008, 129, 064507.
- (399) Pontecorvo, E.; Ferrante, C.; Elles, C. G.; Scopigno, T. Spectrally tailored narrowband pulses for femtosecond stimulated Raman spectroscopy in the range 330-750 nm. Optics Express 2013, 21, 6886.
- (400) Zhu, L.; Liu, W.; Fang, C. A versatile femtosecond stimulated Raman spectroscopy setup with tunable pulses in the visible to near infrared. *Applied Physics Letters* **2014**, *105*, 041106.
- (401) Dunlap, B.; Richter, P.; McCamant, D. W. Stimulated Raman spectroscopy using chirped pulses. Journal of Raman Spectroscopy 2014, 45, 918.

- (402) Shreve, A. P.; Cherepy, N. J.; Franzen, S.; Boxer, S. G.; Mathies, R. A. Rapid-flow resonance Raman spectroscopy of bacterial photosynthetic reaction centers. *Proceedings of the National Academy of Sciences of the United States of America* 1991, 88, 11207.
- (403) Shreve, A. P.; Cherepy, N. J.; Mathies, R. A. Effective Rejection of Fluorescence in Raman Spectroscopy Using a Shifted Excitation Difference Technique. *Applied Spectroscopy* **1992**, *46*, 707.
- (404) McClain, B. L.; Finkelstein, I. J.; Fayer, M. D. Vibrational echo experiments on red blood cells: Comparison of the dynamics of cytoplasmic and aqueous hemoglobin. *Chemical Physics Letters* 2004, 392, 324.
- (405) Spector, I. C.; Olson, C. M.; Huber, C. J.; Massari, A. M. Simple fully reflective method of scatter reduction in 2D-IR spectroscopy. *Optics Letters* 2015, 40, 1850.
- (406) Borysow, J.; Taylor, R. H.; Keto, J. W. Raman-Induced Kerr Effect Spectroscopy of Solutions. Journal of Raman Spectroscopy 1989, 20, 203.
- (407) Foggi, P.; Righini, R.; Torre, R.; Kamalov, V. F. Molecular dynamics of β-carotene in solution measured by subpicosecond transient optical Kerr effect. *Chemical Physics Letters* **1992**, *193*, 23.
- (408) Foggi, P.; Kamalov, V. F.; Righini, R.; Torre, R. Resonance-enhanced time-resolved optical Kerr effect of β-carotene in solution. Optics Letters 1992, 17, 775.
- (409) Orczyk, M. E.; Samoc, M.; Swiatkiewicz, J.; Manickam, N.; Tomoaia-Cotisel, M.; Prasad, P. N. Optical heterodyning of the phase-tuned femtosecond optical Kerr gate signal for the determination of complex third-order susceptibilities. *Applied Physics Letters* **1992**, *60*, 2837.
- (410) Orczyk, M. E.; Samoc, M.; Swiatkiewicz, J.; Prasad, P. N. Dynamics of third-order nonlinearity of canthaxanthin carotenoid by the optically heterodyned phase-tuned femtosecond optical Kerr gate. *The Journal of Chemical Physics* **1993**, *98*, 2524.
- (411) Segal, J.; Huppert, D.; Fainberg, B. Resonance heterodyne optical Kerr effect spectroscopy of population wave packets with intense chirped pulses. *Journal of Luminescence* 2003, 102-103, 669.
- (412) Hellerer, T.; Eneider, A. M. K.; Zumbusch, A. Applied Physics Letters. 2004, 85, 25.
- (413) Oron, D.; Dudovich, N.; Silderberg, Y. Femtosecond Phase-and-Polarization Control for Background-Free Coherent Anti-Stokes Raman Spectroscopy. *Physical Review Letters* 2003, 90, 213902.
- (414) Bremard, C.; Dhamelincourt, P.; Laureyns, J.; Turrell, G. The Effect of High Numerical-Aperture Objectives on Polarization Measurements in Micro-Raman Spectrometry. *Applied Spectroscopy* 1985, 39, 1036.

- (415) Shim, S.; Mathies, R. A. Generation of narrow-bandwidth picosecond visible pulses from broadband femtosecond pulses for femtosecond stimulated Raman. *Applied Physics Letters* 2006, *89*, 121124.
- (416) Xing, Q.; Yoo, K. M.; Alfano, R. R. Conical emission by four-photon parametric generation by using femtosecond laser pulses. *Applied Optics* **1993**, *32*, 2087.
- (417) Marangoni, M.; Frida, D.; Quintavalle, M.; Cirmi, G.; Pigozzo, F. M.; Manzoni, C.; Baronio, F.; Capobianco, A. D.; Cerullo, G. Narrow-bandwidth picosecond pulses by spectral compression of femtosecond pulses in a second-order nonlinear crystal. *Optics Express* **2007**, *15*, 8884.
- (418) Conforti, M.; Baronio, F.; De Angelis, C. From femtosecond infrared to picosecond visible pulses: temporal shaping with high-efficiency conversion. *Optics Letters* **2007**, *32*, 1779.
- (419) Marangoni, M.; Brida, D.; Conforti, M.; Capobianco, A. D.; Manzoni, C.; Baronio, F.; Nalesso, G. F.; De Angelis, C.; Ramponi, R.; Cerullo, G. Synthesis of picosecond pulses by spectral compression and shaping of femtosecond pulses in engineered quadratic nonlinear media. *Optics Letters* **2009**, *34*, 241.
- (420) Liebel, M.; Schnedermann, C.; Wende, T.; Kukura, P. Principles and Applications of Broadband Impulse Vibrational Spectroscopy. *The Journal of Physical Chemistry A* 2015, 119, 9506.
- (421) Zhu, L.; Liu, W.; Fang, C. Tunable sideband laser from cascaded four-wave mixing in thin glass for ultra-broadband femtosecond stimulated Raman spectroscopy. *Applied Physics Letters* 2013, 103, 061110.
- (422) Lagutchev, A.; Hambir Selezion, A.; Dlott, D. D. Nonresonant Background Suppression in Broadband Vibrational Sum-Frequency Generation Spectroscopy. *The Journal of Physical Chemistry C* 2007, 111, 13645.
- (423) Lumeau, J.; Glebov, L. B.; Smirnov, V. Tunable narrowband filter based on a combination of Fabry-Perot etalon and volume Bragg grating. *Optics Letters* **2006**, *31*, 2417.
- (424) Raoult, F.; Boscheron, A. C. L.; Husson, D.; Sauteret, C.; Modena, A.; Malka, V.; Dorchies, F.; Migus,
   A. Efficient generation of narrow-bandwidth picosecond pulses by frequency doubling of femtosecond chirped pulses. *Optics Letters* 1998, 23, 1117.
- (425) Donaldson, P. M.; Guo, R.; Fournier, F.; Gardner, E. M.; Barter, L. M. C.; Barnett, C. J.; Gould, I. R.; Klug, D. R.; Palmer, D. J.; Willison, K. R. Direct identification and decongestion of Fermi resonances by control of pulse time ordering in two-dimensional IR spectroscopy. *The Journal of Chemical Physics* 2007, 127, 114513.

- (426) Hendra, P. J.; Loader, E. J. Laser Raman Spectra of Adsorbed Species. Transactions of the Faraday Society 1971, 67, 828.
- (427) Ghavami Sabouri, S.; Khorsandi, A. Multisegment oven scheme for Guoy phase shift compensation in second-harmonic generation. *Applied Optics* **2015**, *54*, 8580.
- (428) Moutzouris, K.; Papamichael, M.; Betsis, S. C.; Stavrakas, I.; Hloupis, G.; Triantis, D. Refractive, dispersive, and thermo-optic properties of twelve organic solvents in the visible and near-infrared. *Applied Physics B* 2014, 116, 617.
- (429) Ciddor, P. E. Refractive index of air: new equations for the visible and near infrared. Applied Optics 1996, 35, 1566.